UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM S-1 REGISTRATION STATEMENT

UNDER
THE SECURITIES ACT OF 1933

CORE SCIENTIFIC, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 7370 (Primary Standard Industrial Classification Code Number) 86-1243837 (I.R.S. Employer Identification No.)

106 East 6th Street, Suite 900-145 Austin, TX 78701 (425) 998-5300

(Address, including zip code, and telephone number, including area code, of registrant's principal executive offices)

Michael Levitt Chief Executive Officer Core Scientific, Inc. 106 East 6th Street, Suite 900-145 Austin, TX 78701 (425) 998-5300

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copies to:

Nicolas H.R. Dumont Daniel Peale Cooley LLP 55 Hudson Yards New York, New York 10001 Tel: (212) 479-6000 Fax: (212) 479-6275 Todd M. DuChene EVP, General Counsel Core Scientific, Inc. 106 East 6th Street, Suite 900-145 Austin, TX 78701 (425) 998-5300

Approximate date of commencement of proposed sale to the public:

As soon as practicable after this Registration Statement is declared effective.

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933, check the following box.

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. \Box

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. \Box

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer	Accelerated filer	
Non-accelerated filer	Smaller reporting company	Σ
	Emerging growth company	Σ

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 7(a)(2)(B) of the Securities Act. \Box

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the Registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933 or until the Registration Statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a), may determine.

The information in this preliminary prospectus is not complete and may be changed. These securities may not be sold until the registration statement filed with the Securities and Exchange Commission is effective. This preliminary prospectus is not an offer to sell these securities nor does it seek an offer to buy these securities in any jurisdiction where the offer or sale is not permitted.

Subject to Completion, Dated February 8, 2022

PRELIMINARY PROSPECTUS



Up to 186,172,423 Shares of Common Stock Up to 14,891,667 Shares of Common Stock Issuable Upon Exercise of Warrants Up to 6,266,667 Warrants to Purchase Common Stock

This prospectus relates to the issuance by us of an aggregate of up to 14,891,667 shares of our common stock, par value \$0.0001 per share (the "common stock"), which consists of (i) up to 6,266,667 shares of common stock issuable upon the exercise of warrants (the "Private Placement Warrants") originally issued in a private placement to XPDI Sponsor LLC, a Delaware limited liability company (the "Sponsor"), and certain funds and accounts managed by subsidiaries of BlackRock, Inc. in connection with the initial public offering of Power & Digital Infrastructure Acquisition Corp. ("XPDI") and (ii) up to 8,625,000 shares of common stock issuable upon the exercise of warrants (the "Public Warrants" and, together with the Private Placement Warrants, the "Warrants") originally issued in the initial public offering of XPDI. We will receive the proceeds from any exercise of any Warrants for cash.

This prospectus also relates to the offer and sale from time to time by the selling securityholders named in this prospectus or their permitted transferees (the "selling securityholders") of (i) up to 186,172,423 shares of common stock consisting of (a) up to 8,625,000 Founder Shares issued in a private placement in connection with the initial public offering of XPDI, (b) up to 6,266,667 shares of common stock issuable upon exercise of the Private Placement Warrants, (c) up to 91,052,355 shares of common stock (including shares issuable upon the exercise of convertible securities) held by certain affiliates of our company, (d) up to 80,228,401 shares of common stock issuable upon conversion of certain Convertible Notes and (ii) up to 6,266,667 Private Placement Warrants. We will not receive any proceeds from the sale of shares of common stock or Warrants by the selling securityholders pursuant to this prospectus.

The selling securityholders may offer, sell or distribute all or a portion of the securities hereby registered publicly or through private transactions at prevailing market prices or at negotiated prices. We will not receive any of the proceeds from such sales of the shares of common stock or Warrants, except with respect to amounts received by us upon exercise of the Warrants. We will bear all costs, expenses and fees in connection with the registration of these securities, including with regard to compliance with state securities or "blue sky" laws. The selling securityholders will bear all commissions and discounts, if any, attributable to their sale of shares of common stock or Warrants. See the section titled "Plan of Distribution."

Our common stock and Public Warrants are listed on the Nasdaq Global Select Market under the symbols "CORZ" and "CORZW," respectively. On February 4, 2022, the last reported sales price of our common stock was \$9.44 per share and the last reported sales price of our Public Warrants was \$2.90 per warrant.

We are an "emerging growth company" as defined under U.S. federal securities laws and, as such, have elected to comply with reduced public company reporting requirements. This prospectus complies with the requirements that apply to an issuer that is an emerging growth company.

Investing in our securities involves a high degree of risk. You should review carefully the risks and uncertainties described in the section titled "Risk Factors" beginning on page 7 of this prospectus, and under similar headings in any amendments or supplements to this prospectus.

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved of these securities, or passed upon the accuracy or adequacy of this prospectus. Any representation to the contrary is a criminal offense.

Prospectus dated

, 2022

ABOUT THIS PROSPECTUS

This prospectus is part of a registration statement on Form S-1 that we filed with the Securities and Exchange Commission (the "SEC") using the "shelf" registration process. Under this shelf registration process, the selling securityholders may, from time to time, sell the securities offered by them described in this prospectus. We will not receive any proceeds from the sale by such selling securityholders of the securities offered by them described in this prospectus. This prospectus also relates to the issuance by us of the shares of common stock issuable upon the exercise of any Warrants. We will not receive any proceeds from the sale of shares of common stock underlying the Warrants pursuant to this prospectus, except with respect to amounts received by us upon the exercise of the Warrants for cash.

Neither we nor the selling securityholders have authorized anyone to provide you with any information or to make any representations other than those contained in this prospectus or any applicable prospectus supplement or any free writing prospectuses prepared by or on behalf of us or to which we have referred you. Neither we nor the selling securityholders take responsibility for, and can provide no assurance as to the reliability of, any other information that others may give you. Neither we nor the selling securityholders will make an offer to sell these securities in any jurisdiction where the offer or sale is not permitted.

We may also provide a prospectus supplement or post-effective amendment to the registration statement to add information to, or update or change information contained in, this prospectus. You should read both this prospectus and any applicable prospectus supplement or post-effective amendment to the registration statement together with the additional information to which we refer you in the sections of this prospectus titled "Where You Can Find More Information."

Pursuant to the terms of (a) that certain Agreement and Plan of Reorganization and Merger, dated as of July 20, 2021, as amended on October 1, 2021, and as further amended on December 29, 2021, by and among Core Scientific Holding Co., a Delaware corporation ("Legacy Core"), XPDI Merger Sub Inc., a Delaware corporation and wholly owned subsidiary of XPDI ("Merger Sub"), and XPDI, and (b) that certain Agreement and Plan of Merger, dated as of October 1, 2021, as amended on January 14, 2022, by and among XPDI, Legacy Core, XPDI Merger Sub 3, LLC, a Delaware limited liability company and wholly owned subsidiary of XPDI ("Merger Sub 3"), and Blockcap, Inc., a Nevada corporation and wholly owned subsidiary of Legacy Core ("Blockcap"), the Business Combination (as defined below) was effected by (i) the merger of Merger Sub with and into Legacy Core (the "First Merger"), which occurred on January 19, 2022, with Legacy Core surviving the First Merger as a wholly owned subsidiary of XPDI, (ii) the merger of Legacy Core with and into XPDI (the "Second Merger"), which occurred on January 20, 2022, with XPDI surviving the Second Merger, and (iii) following the closing of the Second Merger on January 20, 2022, the merger of Blockcap with and into Merger Sub 3 (the "Third Merger"), with Merger Sub 3 surviving the Third Merger as a wholly owned subsidiary of XPDI under the name "Core Scientific Acquired Mining LLC." Immediately prior to the effective time of the First Merger, XPDI filed a Second Amended and Restated Certificate of Incorporation with the Secretary of State of the State of Delaware pursuant to which XPDI changed its name from "Power & Digital Infrastructure Acquisition Corp." to "Core Scientific, Inc.".

Unless the context indicates otherwise, references in this prospectus to "Core," "Legacy Core," "we," "us," "our" and similar terms refer to Core Scientific Holding Co. and its subsidiaries prior to the consummation of the Business Combination and Core Scientific, Inc. (f/k/a Power & Digital Infrastructure Acquisition Corp.) and its subsidiaries after the consummation of the Business Combination. References to "XPDI" refer to the predecessor registrant prior to the consummation of the Business Combination.

SPECIAL NOTE REGARDING FORWARD-LOOKING STATEMENTS

Certain statements in this prospectus may constitute "forward-looking statements" for purposes of the federal securities laws. Our forward-looking statements include, but are not limited to, statements regarding our and our management team's expectations, hopes, beliefs, intentions or strategies regarding the future. In addition, any statements that refer to projections, forecasts or other characterizations of future events or circumstances, including any underlying assumptions, are forward-looking statements. The words "anticipate," "believe," "continue," "could," "estimate," "expect," "intends," "may," "might," "plan," "possible," "potential," "predict," "project," "should," "will," "would" and similar expressions may identify forward-looking statements, but the absence of these words does not mean that a statement is not forward-looking. Forward-looking statements in this prospectus may include, for example, statements about:

- execute its business strategy, including monetization of services provided and expansions in and into existing and new lines of business;
- realize the benefits expected from the acquisition of Blockcap, including any related synergies;
- anticipate the uncertainties inherent in the development of new business lines and business strategies;
- · retain and hire necessary employees;
- anticipate the impact of the COVID-19 pandemic, including variant strains of COVID-19, and its effect on business and financial conditions:
- manage risks associated with operational changes in response to the COVID-19 pandemic, including the emergence of variant strains of COVID-19;
- increase brand awareness:
- attract, train and retain effective officers, key employees or directors;
- · upgrade and maintain information technology systems;
- acquire and protect intellectual property;
- meet future liquidity requirements and comply with restrictive covenants related to long-term indebtedness;
- effectively respond to general economic and business conditions, including the price of bitcoin;
- maintain the listing on, or to prevent the delisting of our securities from, Nasdaq or another national securities exchange;
- · obtain additional capital, including use of the debt market;
- enhance future operating and financial results;
- successfully execute expansion plans;
- anticipate rapid technological changes;
- comply with laws and regulations applicable to its business, including tax laws and laws and regulations related to data privacy and the
 protection of the environment;
- stay abreast of modified or new laws and regulations applicable to its business or withstand the impact of any new laws and regulations related to its industry;
- anticipate the impact of, and response to, new accounting standards;
- anticipate the significance and timing of contractual obligations;
- maintain key strategic relationships with partners and distributors;
- respond to uncertainties associated with product and service development and market acceptance;
- anticipate the impact of changes in U.S. federal income tax laws, including the impact on deferred tax assets;

- successfully defend litigation; and
- successfully deploy the proceeds from the Business Combination.

These forward-looking statements are based on information available as of the date of this prospectus, and current expectations, forecasts and assumptions, and involve a number of judgments, risks and uncertainties. Accordingly, forward-looking statements should not be relied upon as representing our views as of any subsequent date, and we do not undertake any obligation to update forward-looking statements to reflect events or circumstances after the date they were made, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws.

You should read this prospectus and the documents that we reference in this prospectus and have filed as exhibits to the registration statement, of which this prospectus is a part, completely and with the understanding that our actual future results may be materially different from what we expect. We qualify all of our forward-looking statements by these cautionary statements.

In addition, statements that "we believe" and similar statements reflect our beliefs and opinions on the relevant subject. These statements are based upon information available to us as of the date of this prospectus and while we believe such information forms a reasonable basis for such statements, such information may be limited or incomplete, and such statements should not be read to indicate that we have conducted an exhaustive inquiry into, or review of, all potentially available relevant information. These statements are inherently uncertain, and investors are cautioned not to unduly rely upon these statements.

TABLE OF CONTENTS

	Page
Prospectus Summary	1
Risk Factors	7
Market, Industry and Other Data	55
Use of Proceeds	56
Determination of Offering Price	57
Market Information for Securities and Dividend Policy	58
Management's Discussion and Analysis of Financial Condition and Results of Operations	59
<u>Business</u>	97
<u>Management</u>	109
Executive Compensation	116
Certain Relationships and Related Party Transactions	134
Principal Stockholders	142
Selling Securityholders	144
Description of Capital Stock	145
Material U.S. Federal Income Tax Consequences	156
Plan of Distribution	162
<u>Legal Matters</u>	165
<u>Experts</u>	165
Where You Can Find More Information	166
Unaudited Pro Forma Combined Financial Information	

You should rely only on the information contained in this prospectus, any supplement to this prospectus or in any free writing prospectus, filed with the Securities and Exchange Commission. Neither we nor the selling securityholders have authorized anyone to provide you with additional information or information different from that contained in this prospectus filed with the Securities and Exchange Commission. We take no responsibility for, and can provide no assurance as to the reliability of, any other information that others may give you. The selling securityholders are offering to sell, and seeking offers to buy, our securities only in jurisdictions where offers and sales are permitted. The information contained in this prospectus is accurate only as of the date of this prospectus, regardless of the time of delivery of this prospectus or any sale of our securities. Our business, financial condition, results of operations and prospects may have changed since that date.

For investors outside of the United States: Neither we nor the selling securityholders, have done anything that would permit this offering or possession or distribution of this prospectus in any jurisdiction where action for that purpose is required, other than in the United States. Persons outside the United States who come into possession of this prospectus must inform themselves about, and observe any restrictions relating to, the offering of our securities and the distribution of this prospectus outside the United States.

i

FREQUENTLY USED TERMS

As used in this prospectus, unless otherwise noted or the context otherwise requires, references to:

- "anchor investors" means the following funds and accounts managed by subsidiaries of BlackRock, Inc.: The Obsidian Master Fund, HC NCBR Fund and Blackrock Credit Alpha Master Fund L.P.;
 - "Blockcap" means Blockcap, Inc., a Nevada corporation;
 - "Business Combination" means the transactions contemplated by the merger agreement and related documents;
 - "Common Stock" means the common stock, par value \$0.0001 per share, of Core;
- "Convertible Notes" means the convertible promissory notes issued pursuant to (i) that certain Secured Convertible Note Purchase Agreement, dated as of April 19, 2021, by the among Legacy Core and the other parties thereto, and (ii) that certain Convertible Note Purchase Agreement, dated as of August 20, 2021, by the among Legacy Core and the other parties thereto;
- "Core" means Core Scientific, Inc. (formerly Power & Digital Infrastructure Acquisition Corp.), after giving effect to the Business Combination and the effectiveness of the Charter;
 - "Core/Blockcap merger" means the merger and the other transactions contemplated by the Core/Blockcap merger agreement;
- "Core/Blockcap merger agreement" means that certain Agreement and Plan of Merger, dated as of July 15, 2021, by and among Legacy Core, Blockcap and the other parties thereto;
 - "Legacy Core common stock" means the common stock, par value \$0.00001 per share, of Legacy Core;
 - "DGCL" means the Delaware General Corporation Law;
 - "Effective Time" means the effective time of the First Merger;
 - "Founder Shares" means the 8,625,000 shares of common stock of XPDI that were issued prior to the IPO of XPDI;
 - "GAAP" means generally accepted accounting principles in the United States;
 - "IPO" or "Initial Public Offering" means the initial public offering of XPDI completed on February 12, 2021;
 - "Legacy Core" means Core Scientific Holding Co., a Delaware corporation, and its consolidated subsidiaries, prior to the Effective Time;
- "merger agreement" means that certain Agreement and Plan of Reorganization and Merger, dated as of July 20, 2021, as amended on October 1, 2021, and as further amended on December 29, 2021, by and among Legacy Core, XPDI Merger Sub Inc. and XPDI;
 - "Nasdaq" means The Nasdaq Global Select Market;
- "Private Placement Warrants" means the 6,266,667 warrants held by the Sponsor and anchor investors that were issued to the Sponsor and anchor investors at the closing of the IPO of XPDI, each of which is exercisable for one share of Common Stock, in accordance with its terms;

"Public Warrants" means the 8,625,000 warrants included as a component of the XPDI units sold in the IPO of XPDI, each of which is exercisable for one share of Common Stock, in accordance with its terms;

"SEC" means the U.S. Securities and Exchange Commission;

"Sponsor" means XPDI Sponsor LLC, a Delaware limited liability company;

"Transactions" means the transactions contemplated by the merger agreement, including the merger;

"Trust Account" means the trust account into which \$345.0 million of the net proceeds of the IPO and certain proceeds from the concurrent sale of the Private Placement Warrants were deposited for the benefit of the public stockholders of XPDI, the balance of which account was released and distributed in connection with the Business Combination;

"Warrants" means the Private Placement Warrants and the Public Warrants; and

"XPDI" means Power & Digital Infrastructure Acquisition Corp., a Delaware corporation (which was renamed Core Scientific, Inc. in connection with the Business Combination).

PROSPECTUS SUMMARY

This summary highlights information contained elsewhere in this prospectus and does not contain all of the information that you should consider in making your investment decision. Before investing in our securities, you should carefully read this entire prospectus, including our consolidated financial statements and the related notes thereto and the information set forth in the sections titled "Risk Factors" and "Management's Discussion and Analysis of Financial Condition and Results of Operations." Unless the context otherwise requires, we use the terms "Core," "company," "we," "us" and "our" in this prospectus to refer to Core Scientific, Inc. and our consolidated subsidiaries.

Overview

Core is a best-in-class large-scale operator of dedicated, purpose-built facilities for digital asset mining and a premier provider of blockchain infrastructure, software solutions and services. We mine digital assets for our own account and provide hosting services for other large-scale miners. We are one of the largest blockchain infrastructure, hosting provider and digital asset mining companies in North America, with approximately 497MW of operating electric power as of January 31, 2022. We mine Bitcoin, Ethereum and other digital assets for third-party hosting customers and for our own account at our six fully operational data centers in North Carolina (2), Georgia (2), North Dakota (1) and Kentucky (1). In addition, in October 2021, we announced the entry of an agreement with the City of Denton, Texas and an affiliate of Tenaska Energy, Inc. to develop our seventh facility, a blockchain data center in Denton, Texas, which is expected to have 300MW of power capacity. In February 2022, the Muskogee City-County Port Authority announced an agreement with us to develop a 500MW data center at the Port of Muskogee John T. Griffin Industrial Park.

In July 2021 we completed the acquisition of Blockcap, one of our largest hosting customers. Blockcap is a blockchain technology company with industrial scale digital asset mining operations. Blockcap's primary historical business was the mining of digital asset coins and tokens, primarily Bitcoin and, to a lesser extent, Siacoin and Ethereum. At the time of its acquisition, Blockcap claimed to be the largest independent cryptocurrency mining operator in North America. While Blockcap did sell or exchange the digital assets it mined to fund its growth strategies or for general corporate purposes from time to time, it generally retained its digital assets as investments in anticipation of continued adoption of digital assets as a "store of value" and a more accessible and efficient medium of exchange than traditional fiat currencies. In addition to mining, holding and exchanging digital assets, Blockcap also evaluated and completed investments in related technologies and ancillary businesses, including RADAR, an early stage company focused on technology enhancement and development in the digital asset industry that it acquired on July 1, 2021. The acquisition of Blockcap significantly expands our self-mining operations and increases the number of miners we own. We intend to utilize RADAR's business assets and the technical expertise of its principals in enhancing our existing blockchain mining technology and software and in further strengthening our leadership position and value creation potential through the development of decentralized finance (DeFi) products and services.

Our total revenue was \$242.7 million and \$36.6 million for the nine months ended September 30, 2021 and 2020, respectively. We had operating income of \$29.3 million and an operating loss of \$5.8 million for the nine months ended September 30, 2021 and 2020, respectively. We had a net loss of \$13.2 million and \$9.9 million for the nine months ended September 30, 2021 and 2020, respectively. As of September 30, 2021, we had an accumulated deficit of \$87.9 million.

Corporate Information

We were originally known as Power & Digital Infrastructure Acquisition Corp. On January 19, 2022, Legacy Core, XPDI, and Merger Sub consummated the transactions contemplated under the merger agreement,

following the approval at the special meeting of the stockholders of XPDI held on January 19, 2022. In connection with the closing of the Business Combination, we changed our name from Power & Digital Infrastructure Acquisition Corp. to Core Scientific, Inc.

Our principal executive offices are located at 106 East 6th Street, Suite 900-145, Austin, Texas 78701, and our telephone number is (425) 998-5300. Our corporate website address is www.corescientific.com. Information contained on or accessible through our website is not a part of this prospectus, and the inclusion of our website address in this prospectus is an inactive textual reference only.

"Core Scientific" and our other registered and common law trade names, trademarks and service marks are property of Core Scientific, Inc. This prospectus contains additional trade names, trademarks and service marks of others, which are the property of their respective owners. Solely for convenience, trademarks and trade names referred to in this prospectus may appear without the ® or ™ symbols.

Emerging Growth Company Status

We are an "emerging growth company," as defined in the Jumpstart Our Business Startups Act of 2012 (the "JOBS Act"). As an emerging growth company, we are exempt from certain requirements related to executive compensation, including the requirements to hold a nonbinding advisory vote on executive compensation and to provide information relating to the ratio of total compensation of our President and Chief Executive Officer to the median of the annual total compensation of all of our employees, each as required by the Investor Protection and Securities Reform Act of 2010, which is part of the Dodd-Frank Act.

Section 102(b)(1) of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies are required to comply with the new or revised financial accounting standards. The JOBS Act provides that a company can choose not to take advantage of the extended transition period and comply with the requirements that apply to non-emerging growth companies, and any such election to not take advantage of the extended transition period is irrevocable. Core has elected to avail itself of the extended transition period. We will be an emerging growth company at least until December 31, 2022 and intend to take advantage of the benefits of the extended transition period emerging growth company status permits. During the extended transition period, it may be difficult or impossible to compare our financial results with the financial results of another public company that complies with public company effective dates for accounting standard updates because of the potential differences in accounting standards used.

We will remain an emerging growth company under the JOBS Act until the earliest of (a) December 31, 2026, (b) the last date of our fiscal year in which we have a total annual gross revenue of at least \$1.07 billion, (c) the date on which we are deemed to be a "large accelerated filer" under the rules of the SEC with at least \$700.0 million of outstanding securities held by non-affiliates or (d) the date on which we have issued more than \$1.0 billion in non-convertible debt securities during the previous three years.

Summary of Risk Factors

Investing in our securities involves a high degree of risk. You should carefully consider the risks and uncertainties described below together with all of the other information contained in this prospectus. The occurrence of one or more of the events or circumstances described in the section titled "Risk Factors," alone or in combination with other events or circumstances, may harm our business, financial condition and operating results. Such risks include, but are not limited to:

Our business is highly dependent on a small number of digital asset mining equipment suppliers.

- Our business is capital intensive, and failure to obtain the necessary capital when needed may force us to delay, limit or terminate our
 expansion efforts or other operations, which could have a material adverse effect on our business, financial condition and results of
 operations.
- We may not be able to obtain new hosting and transaction processing hardware or purchase such hardware at competitive prices during times of high demand, which could have a material adverse effect on our business, financial condition and results of operations.
- Our business is heavily impacted by social, political, economic and other events and circumstances in countries outside of the United States, most particularly China and other non-Western countries. China's shifting position on mining activity within its borders could reduce our revenue and profitability.
- A significant portion of our assets are pledged to our senior secured noteholders. This obligation may limit our ability to obtain
 additional capital to grow our business and failure to repay obligations to our noteholders when due will have a material adverse effect
 on our business and could result in foreclosure on our assets.
- We are subject to risks associated with our need for significant electric power and the limited availability of power resources, which
 could have a material adverse effect on our business, financial condition and results of operations. An inability to purchase and
 develop additional sources of low-cost renewable sources of energy effectively will have a material adverse effect on our business,
 financial condition and results of operations.
- We plan to continue to acquire other businesses or receive offers to be acquired, which could require significant management attention, disrupt our business or dilute stockholder value.
- If we do not successfully integrate Blockcap, Blockcap's subsidiary, RADAR, or future acquisitions or strategic partnerships that we may enter into, we may not realize the anticipated benefits of any such acquisitions or partnerships, which could have a material adverse effect on our business, financial condition and results of operations.
- We generate significant revenue from a limited number of hosting facilities in Kentucky, Georgia, North Carolina and North Dakota
 and a significant disruption to operations in this region could have a material adverse effect our business, financial condition and
 results of operations.
- Our future success depends on our ability to keep pace with rapid technological changes that could make our current or future technologies less competitive or obsolete.
- The further development and acceptance of cryptographic and algorithmic protocols governing transaction validation and the issuance
 of, and transactions in, digital assets are subject to a variety of factors that are difficult to evaluate. The slowing or stoppage of
 development or acceptance of blockchain networks and digital assets would have an adverse material effect on the successful
 development of the mining operation and value of mined digital assets.
- Our ability to use net operating losses to offset future taxable income may be subject to limitations.
- We operate in a rapidly developing industry and have an evolving business model with a limited history of generating revenue from
 our services. In addition, our evolving business model increases the complexity of our business, which makes it difficult to evaluate
 our future business prospects and could have a material adverse effect on our business, financial condition and results of operations.
- We have experienced difficulties in establishing relationships with banks, leasing companies, insurance companies and other financial
 institutions that are willing to provide us with customary financial products and services, which could have a material adverse effect
 on our business, financial condition and results of operations.

- Digital assets exchanges and other trading venues are relatively new and, in some cases, partially unregulated and may therefore be
 more exposed to fraud and failure.
- We may not have adequate sources of recovery if the digital assets held by us are lost, stolen or destroyed due to third-party digital asset services, which could have a material adverse effect on our business, financial condition and results of operations.
- Losses relating to our business may be uninsured, or insurance may be limited.
- Diversification of our business by investing in additional digital assets, financial instruments and businesses could require significant investment or expose us to trading risks.
- As more processing power is added to a network, our relative percentage of total processing power on that network is expected to
 decline absent significant capital investment, which has an adverse impact on our ability to generate revenue from processing
 transactions on that network and could have a material adverse effect on our business, financial condition and results of operations.
- Our reliance on third-party mining pool service providers for our mining revenue payouts may have a negative impact on our operations.
- Malicious actors or botnet may obtain control of more than 50% of the processing power on the Bitcoin or other network.
- Digital assets are subject to extreme price volatility. The value of digital assets is dependent on a number of factors, any of which could have a material adverse effect on our business, financial condition and results of operations.
- Any loss or destruction of a private key required to access a digital asset of ours is irreversible. We also may temporarily lose access to our digital assets.
- The digital assets held by us are not subject to FDIC or SIPC protections.
- Our interactions with a blockchain may expose us to SDN or blocked persons or cause us to violate provisions of law that did not
 contemplate distribute ledger technology.
- We have identified material weaknesses in our internal control over financial reporting and may identify additional material
 weaknesses in the future or otherwise fail to maintain an effective system of internal control, which may result in material
 misstatements of Core's financial statements or cause it to fail to meet its periodic reporting obligations.
- We have identified a material weakness in our internal control over financial reporting. This material weakness could continue to adversely affect our ability to report our results of operations and financial condition accurately and in a timely manner.

The Offering				
Issuance of Common Stock				
Shares of common stock offered by us	14,891,667 shares of common stock, consisting of (i) 6,266,667 shares of common stock that are issuable upon exercise of the Private Placement Warrants and (ii) 8,625,000 shares of common stock that are issuable upon exercise upon the exercise of the Public Warrants.			
Shares of common stock outstanding prior to the exercise of all Warrants	320,124,058 (as of February 4, 2022).			
Shares of common stock outstanding assuming exerci of all Warrants	ise 335,015,725 (based on the total shares outstanding as of February 4, 2022).			
Exercise price of the Public and Private Placement Warrants	\$11.50 per share, subject to adjustment as described herein.			
Use of proceeds	We will receive up to an aggregate of approximately \$171.3 million from the exercise of the Public and Private Placement Warrants. We expect to use the net proceeds from the exercise of the Public and Private Placement Warrants for general corporate purposes. See "Use of Proceeds."			
Resale of Common Stock and Warrants				
Shares of common stock offered by the selling securityholders	We are registering the resale by the selling securityholders named in this prospectus, or their permitted transferees, an aggregate of 186,172,423 shares of common stock, consisting of:			
	• up to 8,625,000 Founder Shares;			
	 up to 6,266,667 shares of common stock issuable upon exercise of the Private Placement Warrants; 			
	 up to 91,052,355 shares of common stock (including shares issuable upon the exercise of convertible securities) held by certain affiliates of our company; and 			
	 up to 80,228,401 shares of common stock issuable upon conversion of certain Convertible Notes. 			
Warrants offered by the selling securityholders	Up to 6,266,667 Private Placement Warrants.			
Redemption	The Public Warrants are redeemable in certain circumstances. See "Description of Capital Stock—Public Warrants."			
Terms of the offering	The selling securityholders will determine when and how they will dispose of the securities registered for resale under this prospectus.			

Lock-Up Restrictions Certain of our securityholders are subject to certain restrictions on transfer until the

termination of applicable lock-up periods. See the section titled "Certain Relationships and Related Party Transactions—Lock-Up Restrictions."

Use of proceeds We will not receive any of the proceeds from the sale of the shares of common stock or

Private Placement Warrants by the selling securityholders, except with respect to amounts

received by us due to the exercise of the Warrants.

Before investing in our securities, you should carefully read and consider the information Risk factors

set forth in "Risk Factors."

Nasdaq ticker symbols "CORZ" and "CORZW."

For additional information concerning the offering, see "Plan of Distribution."

RISK FACTORS

Investing in our securities involves a high degree of risk. You should carefully consider the risks and uncertainties described below together with all of the other information contained in this prospectus, including our financial statements and related notes appearing at the end of this prospectus and in the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations," before deciding to invest in our securities. If any of the events or developments described below were to occur, our business, prospects, operating results and financial condition could suffer materially, the trading price of our common stock could decline, and you could lose all or part of your investment. The risks and uncertainties described below are not the only ones we face. Additional risks and uncertainties not presently known to us or that we currently believe to be immaterial may also adversely affect our business.

Risks Related to Core's Business and Industry

Our business is highly dependent on a small number of digital asset mining equipment suppliers.

Our business is highly dependent upon digital asset mining equipment suppliers such as Bitmain Technologies, Ltd ("Bitmain") providing an adequate supply of new generation digital asset mining machines at economical prices to customers intending to purchase our hosting and other solutions. The growth in our business is directly related to increased demand for hosting services and digital assets such as bitcoin which is dependent in large part on the availability of new generation mining machines offered for sale at a price conducive to profitable digital asset mining, as well as the trading price of digital assets such as bitcoin. The market price and availability of new mining machines fluctuates with the price of bitcoin and can be volatile. Higher bitcoin prices increase the demand for mining equipment and increases the cost. In addition, as more companies seek to enter the mining industry, the demand for machines may outpace supply and create mining machine equipment shortages. There are no assurances that digital asset mining equipment suppliers, such as Bitmain, will be able to keep pace with any surge in demand for mining equipment. Further, manufacturing mining machine purchase contracts are not favorable to purchasers and we may have little or no recourse in the event a mining machine manufacturer defaults on its mining machine delivery commitments. If we and our customers are not able to obtain a sufficient number of digital asset mining machines at favorable prices, our growth expectations, liquidity, financial condition and results of operations will be negatively impacted.

Our business is capital intensive, and failure to obtain the necessary capital when needed may force us to delay, limit or terminate our expansion efforts or other operations, which could have a material adverse effect on our business, financial condition and results of operations.

The costs of constructing, developing, operating and maintaining digital asset mining and hosting facilities, and owning and operating a large fleet of the latest generation mining equipment are substantial.

Our mining operations can only be successful and ultimately profitable if the costs, including hardware and electricity costs, associated with mining digital assets are lower than the price of the digital assets we mine when we sell them. Our miners experience ordinary wear and tear from operation and may also face more significant malfunctions caused by factors which may be beyond our control. Additionally, as the technology evolves, we may acquire newer models of miners to remain competitive in the market. Over time, we replace those miners which are no longer functional with new miners purchased from third-party manufacturers, who are primarily based in China.

As miners become obsolete or degrade due to ordinary wear and tear from usage, or are lost or damaged due to factors outside of our control, these miners will need to be repaired or replaced along with other equipment from time to time for us to stay competitive. This upgrading process requires substantial capital investment, and we may face challenges in doing so on a timely and cost-effective basis based on availability of new miners and our access to adequate capital resources. If we are unable to obtain adequate numbers of new and replacement miners at scale, we may be unable to remain competitive in our highly competitive and evolving industry.

Moreover, in order to grow our hosting business, we need additional hosting facilities to increase our capacity for more miners. The costs of constructing, developing, operating and maintaining hosting facilities and growing our hosting operations may increase in the future, which may make it more difficult for us to expand our business and to operate our hosting facilities profitably.

We will need to raise additional funds through equity or debt financings in order to meet our operating and capital needs. Additional debt or equity financing may not be available when needed or, if available, may not be available on satisfactory terms. An inability to generate sufficient cash from operations or to obtain additional debt or equity financing would adversely affect our results of operations. Additionally, if this happens, we may not be able to mine digital assets as efficiently or in similar amounts as our competition and, as a result, our business and financial results could suffer.

We may not be able to obtain new hosting and transaction processing hardware or purchase such hardware at competitive prices during times of high demand, which could have a material adverse effect on our business, financial condition and results of operations.

Historically, an increase in interest and demand for digital assets has led to a shortage of hosting and transaction processing hardware and increased prices. We and our customers and potential customers have experienced, and may in the future experience, difficulty in obtaining new equipment or replacement components for our and their existing equipment, including graphics processing units and application-specific integrated circuit chipsets and computer servers, which has had, and in the future may have, a material impact on the demand for our services and associated revenue. Currently, restrictions on digital asset mining in China have increased availability of used mining equipment and decreased prices of new mining equipment. In addition, these restrictions have decreased available mining facilities in China and increased demand for hosting in countries outside of China including the U.S. To the extent miners view this used equipment as a viable alternative to purchasing new miners from us our equipment sales may suffer, which could have a material adverse effect on our business, financial condition and results of operations.

If future prices of bitcoin are not sufficiently high, our business, results of operations and financial condition could be materially and adversely affected, which may have a negative impact on the trading price of our securities.

Our financial condition and results of operations are, and are expected to increasingly be, reliant on our ability to sell the bitcoin we mine at a price greater than our costs to produce that bitcoin. As the price for new miners we buy increases, our cost to produce a single bitcoin also increases, therefore requiring a corresponding increase in the price of bitcoin for us to maintain our results of operations. If future prices of bitcoin are not sufficiently high, we may not realize the benefit of the capital expenditures we incur each time we acquire new miners. If this occurs, our business, results of operations and financial condition could be materially and adversely affected, which may have a negative impact on the trading price of our securities, which may have a materially adverse impact on investors' investment in our Company.

Our success depends in large part on our ability to mine digital assets profitably and to attract customers for our hosting capabilities. Increases in power costs or our inability to mine digital assets efficiently and to sell digital assets at favorable prices will reduce our operating margins, impact our ability to attract customers for our services and harm our growth prospects and could have a material adverse effect on our business, financial condition and results of operations.

Our growth depends in large part on our ability to successfully mine digital assets and to attract customers for our hosting capabilities. We may not be able to attract customers to our hosting capabilities for a number of reasons, including if:

there is a reduction in the demand for our services due to macroeconomic factors in the markets in which we operate;

- we fail to provide competitive pricing terms or effectively market them to potential customers;
- we provide hosting services that are deemed by existing and potential customers or suppliers to be inferior to those of our competitors, or
 that fail to meet customers' or suppliers' ongoing and evolving program qualification standards, based on a range of factors, including
 available power, preferred design features, security considerations and connectivity;
- businesses decide to host internally as an alternative to the use of our services;
- we fail to successfully communicate the benefits of our services to potential customers;
- we are unable to strengthen awareness of our brand;
- we are unable to provide services that our existing and potential customers desire; or
- · our customers are unable to secure an adequate supply of new generation digital asset mining equipment to host with us.

If we are unable to obtain hosting customers at favorable pricing terms or at all, it could have a material adverse effect on our business, financial condition and results of operations.

A slowdown in the demand for blockchain technology or blockchain hosting resources and other market and economic conditions could have a material adverse effect on our business, financial condition and results of operations.

Adverse developments in the blockchain industry, and in the blockchain hosting market could lead to a decrease in the demand for hosting resources, which could have a material adverse effect on our business, financial condition and results of operations. We face risks including those related to:

- a decline in the adoption and use of bitcoin and other similar digital assets within the technology industry or a decline in value of digital assets;
- increased costs of complying with existing or new government regulations applicable to digital assets and other factors;
- a downturn in the market for blockchain hosting space generally, which could be caused by an oversupply of or reduced demand for blockchain space;
- any transition by our customers of blockchain hosting from third-party providers like us to customer-owned and operated facilities;
- the rapid development of new technologies or the adoption of new industry standards that render our or our customers' current products
 and services obsolete or unmarketable and, in the case of our customers, that contribute to a downturn in their businesses, increasing the
 likelihood of a default under their service agreements or their becoming insolvent;
- a slowdown in the growth of the Internet generally as a medium for commerce and communication;
- availability of an adequate supply of new generation digital asset mining equipment to enable us to mine digital assets at scale and for customers who want to host with us to be able to do so; and
- the degree of difficulty in mining digital assets and the trading price of such assets.

To the extent that any of these or other adverse conditions exist, they are likely to have an adverse impact on our mining rewards and market demand and pricing for our services, which could have a material adverse effect on our business, financial condition and results of operations.

Additionally, we and our customers are affected by general business and economic conditions in the United States and globally. These conditions include short-term and long-term interest rates, inflation, money supply,

political issues, legislative and regulatory changes, including the imposition of new tariffs affecting our or our customers' products and services, fluctuations in both debt and equity capital markets and broad trends in industry and finance, all of which are beyond our control. Macroeconomic conditions that affect the economy and the economic outlook of the United States and the rest of the world could adversely affect our customers and vendors, which could have a material adverse effect on our business, financial condition and results of operations.

Our business is heavily impacted by social, political, economic and other events and circumstances in countries outside of the United States, most particularly China and other non-Western countries. China's shifting position on mining activity within its borders could reduce our revenue and profitability.

Our business is heavily impacted by social, political, economic and other events and circumstances in countries outside of the United States, most particularly in China and other non-Western countries. These events and circumstances are largely outside of our influence and control. We are heavily dependent on the Chinese manufacture of equipment, much of which has historically been for sale within China and other countries outside the United States. We believe that historically China was a location of significant digital asset mining at low electric power rates. Recently, China and other foreign governments have taken action to prohibit or significantly restrict digital asset mining. For example, in May and June 2021, in their efforts to curb digital asset trading and mining, regulators in several Chinese Provinces, including Qinghai, Inner Mongolia and Sichuan, announced policies to curb or ban local digital asset mining operations. Following the ban announcement, the price of bitcoin experienced a drop of over 30% in May. The long-term impact of such restrictions is unknown and could be detrimental to our business and profitability. Currently, the restrictions in China have enhanced our business by reducing the number of operating digital asset miners and decreasing the mining difficulty which has increased our digital asset yield and increased revenue. Whether or not the lack of mining activity in China will negatively impact Chinese miner manufacturing and the development, price, availability of new and enhanced mining equipment is unknown. Should China or other countries that currently restrict digital asset mining eliminate such restrictions or actually seek to enhance such mining activity, the likely increase in mining activity would likely reduce our revenue and profitability.

Continuing coronavirus outbreaks may have a material adverse impact on our business, liquidity, financial condition and results of operations.

COVID-19 was first reported in December 2019 in the City of Wuhan, Hubei, China and was recognized as a pandemic by the World Health Organization on March 11, 2020. In response to the pandemic, governmental authorities around the World, including the United States, Canada, China and elsewhere, introduced various measures to limit the spread of the pandemic, including travel restrictions, border closures, business closures, quarantines, self- and forced isolations, shelter-in-place orders and social distancing. COVID-19 reduced the number of new generation machines available for purchase by prospective customers of our blockchain hosting services, reduced demand for our services and delayed and continues to frustrate and delay global supply chains that has impacted and will continue to impact the pace at which new mining machines are added to our facilities. The continued impact or a resurgence of COVID-19, including the emergence of variant strains of COVID-19, could have a material impact on our business, liquidity, financial condition and results of operations and any such impact will be determined by the severity and duration of the continuing pandemic.

Changes in tariffs or import restrictions could have a material adverse effect on our business, financial condition and results of operations.

Equipment necessary for digital asset mining is almost entirely manufactured outside of the United States. There is currently significant uncertainty about the future relationship between the United States and various other countries, including China, the European Union, Canada, and Mexico, with respect to trade policies, treaties, tariffs and customs duties, and taxes. For example, since 2019, the U.S. government has implemented significant changes to U.S. trade policy with respect to China. These tariffs have subjected certain digital asset

mining equipment manufactured overseas to additional import duties of up to 25%. The amount of the additional tariffs and the number of products subject to them has changed numerous times based on action by the U.S. government. These tariffs have increased costs of digital asset mining equipment, and new or additional tariffs or other restrictions on the import of equipment necessary for digital asset mining could have a material adverse effect on our business, financial condition and results of operations.

Our historical financial results may not be indicative of our future performance.

In 2018, we generated limited revenue and incurred substantial losses and may continue to incur losses for the foreseeable future. We had a net loss of \$11.9 million in for the fiscal year ended December 31, 2019 and a net loss of \$12.2 million for the fiscal year ended December 31, 2020. As of September 30, 2021, our accumulated deficit was \$87.9 million. Our historical results are not indicative of our future performance. If we are not able to successfully develop our business, it will have a material adverse effect on our business, financial condition and results of operations.

We may be required to record goodwill or other long-lived asset impairment charges, which could result in a significant charge to earnings.

Under GAAP, we review our long-lived assets, such as goodwill, intangible assets and fixed assets, for impairment when events or changes in circumstances indicate the carrying value may not be recoverable. A significant portion of our total assets consists of goodwill and intangible assets. Goodwill is assessed for impairment at least annually. Factors that may be considered in assessing whether goodwill or other long-lived assets may not be recoverable include reduced estimates of future cash flows and slower growth rates in our industry. We may experience unforeseen circumstances that adversely affect the value of our goodwill or other long-lived assets and trigger an evaluation of the recoverability of the recorded goodwill and other long-lived assets. Separately, we assess our digital assets, which currently predominately consists of bitcoin, for impairment on a daily basis. Digital assets are currently considered long-lived intangible assets under GAAP, meaning that any decrease in their fair values below our carrying values for such assets at any time subsequent to their acquisition will require us to recognize impairment charges, whereas we may make no upward revisions for any market price increases until a sale, which may adversely affect our operating results in any period in which such impairment occurs. Moreover, there is no guarantee that future changes in GAAP will not require us to change the way we account for digital assets held by us. Due principally to variability in the price of bitcoin, we have recently recorded related impairment charges, and may continue to do so in future reporting periods. Our results of operations may be materially impacted if we are required to record a significant charge due to an impairment of our goodwill, intangible assets or long-lived assets.

A significant portion of our assets are pledged to our senior secured noteholders. This obligation may limit our ability to obtain additional capital to grow our business and failure to repay obligations to our noteholders when due will have a material adverse effect on our business and could result in foreclosure on our assets.

As of September 30, 2021, we owed our senior secured convertible noteholders \$220.9 million. The notes have a maturity date of April 19, 2025, accrue interest at a rate of 10% per annum (of which 4% is payable in cash and 6% is payable in kind), and are convertible under certain circumstances into shares of our capital stock. In August through November of 2021, an additional \$300 million of senior unsecured convertible notes were issued under substantially the same terms and conditions as the original \$215 million of secured convertible notes issued in April 2021, except that such notes are unsecured until the occurrence of a) an initial public offering or SPAC merger, b) a private placement of equity securities with gross proceeds to the Company of at least \$50 million or c) a change in control at which time they become secured on a pari passu basis with the secured convertible notes. At maturity, the secured convertible notes (but not the senior unsecured notes) not converted will be owed two times the face value of such notes plus accrued interest. The terms of the convertible notes include numerous restrictions and covenants, which significantly limit our flexibility in obtaining additional indebtedness while the convertible notes are outstanding. It is necessary for us to grow our business in order to

generate the free cash flow necessary to repay the principal and interest on our indebtedness. If we were to default on the amounts owed or other terms and conditions of the convertible notes, the noteholders would have the right to exercise rights and remedies to collect, which would include foreclosing on most of our assets. A default would have a material adverse effect on our business and our stockholders could lose their entire investment in us.

Our revenue comes from a small number of customers, and the loss of, or significant decrease in business from, a number of these customers or our failure to continually attract new customers could have a material adverse effect on our business, financial condition and results of operations.

We have generated a significant portion of our historical revenue from a small number of hosting customers. Historically, Blockcap was one of our largest hosting customers and represented a significant portion of our revenue. Any failure to meet our end-users' expectations, including, but not limited to, any inability to meet their requirements for increased hosting capacity at attractive rates, could result in cancellation or non-renewal of our business relationships. Our increased focus on self-mining could be interpreted by our current and prospective customers as being competitive or inconsistent with our third-party hosting operations. If these customers reduced spending on our services, or changed their outsourcing strategy by moving to in-house facilities or outsourcing to other service providers, and we are not able to offset that lost revenue or replace the reduced capacity utilization with our own mining equipment, it could have a material adverse effect on our business, financial condition and results of operations. We have made significant investments in our business, such as acquiring additional hosting facilities and equipment, and incurring additional costs in connection with the expansion of our business to meet our anticipated mining needs as well as the anticipated needs of both current and future customers. Accordingly, if we fail to obtain significant additional customers or fail to increase our self-mining operations, it could have a material adverse effect on our business, financial condition and results of operations.

Delays in the expansion of existing hosting facilities or the construction of new hosting facilities or significant cost overruns could present significant risks to our business and could have a material adverse effect on our business, financial condition and results of operations.

The servers used for digital asset transaction processing and colocation hosting require the use of facilities ("hosting facilities") with a highly specialized infrastructure and considerable, reliable power in order to compete effectively. Our growth strategy is to increase our mining capacity and increase substantially the number of miners we operate. In order to meet our financial plan, we need to expand our existing hosting facilities or obtain suitable land to build new hosting facilities. We may face challenges in obtaining suitable land to build new hosting facilities, as we need to work closely with the local power suppliers and local governments of the places where our proposed hosting facilitates are located. Delays in actions that require the assistance of such third parties, in receiving required permits and approvals or in mediations with local communities, if any, may negatively impact our construction timelines and budget or result in any new hosting facilities not being completed at all.

Our two newest facilities in Georgia and North Dakota became operational during the fourth quarter of 2021. In October 2021, we announced the entry of an agreement with the City of Denton, Texas and an affiliate of Tenaska Energy, Inc. to develop a new facility in Texas. Additional expansion of existing hosting facilities and construction of new hosting facilities is also being contemplated. Such expansion and construction require us to rely on the experience of one or more designers, general contractors and subcontractors, and such designers or contractors may experience financial or other problems during the design or construction process. We may also experience quality control issues as we implement any upgrades in our hosting capacity through the installation and maintenance of chipsets and servers or new cooling technologies such as immersion and water curtain cooling. Our business will be negatively impacted if we are unable to run our mining operations in a way that is technologically advanced, economically and energy efficient and temperature controlled. If we are unsuccessful, we will damage our miners and the miners of third parties and the profitability of our mining operations.

If we experience significant delays in the supply of power required to support any hosting facility expansion or new construction, the progress of such projects could deviate from our original plans, which could cause material and negative effects on our revenue growth, profitability and results of operations. Any material delay in completing these projects, or any substantial cost increases or quality issues in connection with these projects, could materially delay our ability to deliver our hosting capacity, cause us to incur penalties under hosting contracts, result in reduced order volume and materially adversely affect our business, financial condition and results of operations.

We are subject to risks associated with our need for significant electric power and the limited availability of power resources, which could have a material adverse effect on our business, financial condition and results of operations. An inability to purchase and develop additional sources of low-cost renewable sources of energy effectively will have a material adverse effect on our business, financial condition and results of operations.

Our mining and hosting services require a significant amount of electric power. The costs of electric power account for a significant portion of our cost of revenue. We require a significant electric power supply to conduct our mining activity and to provide many hosting services we offer, such as powering and cooling our and our customers' servers and network equipment and operating critical mining and hosting facility and equipment infrastructure.

The amount of power required by us and our customers will increase commensurate with the demand for our services and the increase in miners we operate for ourselves and our hosting customers. Energy costs and availability are vulnerable to seasonality, with increased costs primarily in the summer months and risks of outages and power grid damage as a result of inclement weather, animal incursion, sabotage and other events out of our control. Although we aim to build and operate energy efficient hosting facilities, there can be no assurance such facilities will be able to deliver sufficient power to meet the growing needs of our business. The cost of power at our hosting facilities is dependent on our ability to perform under the terms in the power contracts we are a party to, which we may be unable to do successfully. Pursuant to these power contracts, if we fail to curtail our power usage when called upon or fail to satisfy certain eligibility requirements for monthly bill credits, our power costs would increase. Any system downtime resulting from insufficient power resources or power outages could have a material adverse effect on our business, financial condition and results of operations. Our operations do not run on back-up generators in the event of a power outage. Increased power costs and limited availability and curtailment of power resources will reduce our revenue and have a material and adverse effect on our cost of revenue and results of operations.

Any system downtime resulting from insufficient power resources or power outages could have a material adverse effect on our business, financial condition and results of operations. Because the mining portion of our business consumes a large amount of energy, it is not practical or economical for our operations to run on back-up generators in the event of a power outage.

Governments and government regulators may potentially restrict the ability of electricity suppliers to provide electricity to hosting and transaction processing operations such as ours, which could have a material adverse effect on our business, financial condition and results of operations.

Although we have not experienced it since our inception, governments or government regulators may potentially restrict electricity suppliers from providing electricity to hosting facilities and hosting and transaction processing operators in times of electricity shortage or may otherwise potentially restrict or prohibit the provision of electricity to transaction process operators like us. For example, on May 14, 2018, the Chelan County Public Utility District in Washington approved a three-month extension of a moratorium on the approval of electric service for new digital asset transaction operators in Chelan County. In March 2018, the City of Plattsburgh, New York, placed an 18-month moratorium on transaction processing to preserve natural resources, the health of its residents and the "character and direction" of the city after residents complained about significantly higher electricity bills.

In the event government regulators issue moratoriums or impose bans or restrictions involving hosting operations or transaction processing in jurisdictions in which we operate, we will not be able to continue our operations in such jurisdictions. A moratorium, ban or restriction could have a material adverse effect our business, financial condition and results of operations.

Power outage in our hosting facilities could have a material adverse effect on our business, financial condition and results of operations.

Although we control, operate and have access to our servers and all of the other components of our network, we are still vulnerable to disruptions and power outages resulting from weather, animal incursions, accidents, equipment failures, curtailments, acts of war, sabotage and other events. We do not have backup power generators for our blockchain operations in the event of a power outage. This could impact our ability to generate and maintain contractually specified power levels to our contractual counterparties, which could have a material adverse effect on our business, financial condition and results of operations.

If we do not accurately predict our hosting facility requirements, it could have a material adverse effect on our business, financial condition and results of operations.

The costs of building out, leasing and maintaining our hosting facilities constitute a significant portion of our capital and operating expenses. In order to manage growth and ensure adequate capacity for our digital mining operations and new and existing hosting customers while minimizing unnecessary excess capacity costs, we continuously evaluate our short- and long-term data center capacity requirements. If we overestimate our business' capacity requirements or the demand for our services and therefore secure excess data center capacity, our operating margins could be materially reduced. If we underestimate our data center capacity requirements, we may not be able to service the expanding needs of our existing customers and may be required to limit new customer acquisition, which could have a material adverse effect on our business, financial condition and results of operations.

We plan to continue to acquire other businesses or receive offers to be acquired, which could require significant management attention, disrupt our business or dilute stockholder value.

As part of our business strategy, we have made and in the future intend to make acquisitions of other companies, products and technologies. We have limited experience in acquisitions. We may not be able to find suitable acquisition candidates and we may not be able to complete acquisitions on favorable terms in the future, if at all. In July 2021, we acquired Blockcap, one of our largest hosting customers for digital asset mining, and its subsidiary, Radar Relay, Inc. ("RADAR"), an early stage company focused on technology enhancement and development in the digital asset industry. The acquisition of Blockcap and any future acquisitions may not ultimately strengthen our competitive position or achieve our goals, and could ultimately be viewed negatively. In addition, we may not be able to integrate Blockcap and RADAR or other acquired businesses successfully or effectively manage our company following an acquisition. We intend to utilize RADAR's business assets and the technical expertise of its principals in enhancing our existing blockchain mining technology and software and in further strengthening our leadership position and value creation potential through the development of DeFi products and services. If we fail to successfully integrate Blockcap, RADAR or other future acquisitions, or the people or technologies associated with those acquisitions, into our company, the results of our operations could be adversely affected. Any integration process will require significant time and resources, require significant attention from management and disrupt the ordinary functioning of our business, and we may not be able to manage the process successfully, which could harm our business. Moreover, we may not successfully evaluate or utilize the acquired technology and accurately forecast the financial impact of an acquisition transaction, including accounting charges. Additionally, we may receive indications of interest from other parties interested in acquiring some or all of our business and adversely affect our op

We may have to pay cash, incur debt or issue equity securities to pay for any such acquisition, each of which could affect our financial condition or the value of our capital stock. The sale of equity to finance any such acquisitions could result in dilution to our stockholders. If we incur more debt, it would result in increased fixed obligations and could also subject us to covenants or other restrictions that would impede our ability to flexibly operate our business.

If we do not successfully integrate Blockcap, Blockcap's subsidiary, RADAR, or future acquisitions or strategic partnerships that we may enter into, we may not realize the anticipated benefits of any such acquisitions or partnerships, which could have a material adverse effect on our business, financial condition and results of operations.

On July 30, 2021, we acquired Blockcap, one of our largest hosting customers, and Blockcap's subsidiary RADAR, an early stage company focused on technology enhancement and development in the digital asset industry. In the future we expect to contemplate and pursue acquisitions to expand and diversify our business. We may also form strategic partnerships with third parties that we believe will complement or augment our existing business. We cannot, however, provide assurance that we will be able to identify any potential acquisition or strategic partnership candidates, consummate any additional acquisitions or enter into any strategic partnerships in the future or that any such future acquisitions or strategic partnerships will be successfully integrated or advantageous to us. Blockcap, RADAR and other entities we acquire may not achieve the revenue and earnings we anticipate, or their liabilities may exceed our expectations. We could face integration issues pertaining to the internal controls and operational functions of Blockcap and RADAR, and we also could fail to realize cost efficiencies or synergies that we anticipated from the Blockcap acquisition or other acquisition candidates. The pursuit of potential acquisitions or the integration of Blockcap and RADAR could divert our management's attention and cause us to incur expenses in identifying, investigating and pursuing suitable acquisitions, whether or not they are consummated. Client dissatisfaction or performance problems with Blockcap or RADAR could have a material adverse effect on our reputation as a whole. We may be unable to profitably manage Blockcap, RADAR or other acquired entities, or we may fail to integrate them successfully without incurring substantial expenses, delays or other problems. We may not achieve the anticipated benefits from the acquisition of Blockcap, RADAR or future acquisitions or strategic partnerships due to a number of factors, including:

- inability or difficulty integrating and benefiting from acquired technologies or solutions in a profitable manner, including as a result of reductions in operating income, increases in expenses, failure to achieve synergies or otherwise;
- unanticipated costs or liabilities associated with Blockcap and RADAR or another acquisition or strategic partnership;
- difficulty integrating the accounting systems, operations and personnel of Blockcap and RADAR;
- adverse effects to our existing business relationships and clients or to Blockcap's business relationships and clients as a result of the
 acquisition;
- loss of key employees, particularly those of Blockcap and RADAR;
- assumption of potential liabilities of Blockcap and RADAR, including regulatory noncompliance or acquired litigation, and expenses
 relating to contractual disputes of the acquired business for, infringement of intellectual property rights, data privacy violations or other
 claims;
- difficulty in acquiring suitable businesses, including challenges in predicting the value an acquisition will ultimately contribute to our business; and
- · use of substantial portions of our available cash or assumption of additional indebtedness to consummate an acquisition.

If we fail to successfully integrate Blockcap and RADAR or other businesses that we may acquire or strategic partnerships that we may enter into, we may not realize any of the benefits we anticipate in connection

with any such acquisitions or partnerships, which could have a material adverse effect on our business, financial condition and results of operations.

If there are significant changes to the method of validating blockchain transactions, such changes could reduce demand for our blockchain hosting services.

New digital asset transaction protocols are continuously being deployed, and existing and new protocols are in a state of constant change and development. While certain validation protocols currently employ a "proof of work" consensus algorithm, whereby transaction processors are required to expend significant amounts of electrical and computing power to solve complex mathematical problems in order to validate transactions and create new blocks in a blockchain, there may be a shift towards adopting alternative validating protocols. These protocols may include a "proof of stake" algorithm or an algorithm based on a protocol other than proof of work, which may decrease the reliance on computing power as an advantage to validating blocks. Our transaction processing operations, and, to our knowledge, the operations of our potential hosting customers, are currently designed to primarily support a proof of work consensus algorithm. Should the algorithm shift from a proof of work validation method to a proof of stake method, mining would require less energy and may render any company that maintains advantages in the current climate (for example, from lower priced electricity, processing, real estate or hosting) less competitive. As a result of our efforts to optimize and improve the efficiency of our digital asset mining operations, we may be exposed to the risk in the future of losing the benefit of our capital investments and the competitive advantage we hope to gain from this as a result, and may be negatively impacted if a switch to proof of stake validation were to occur. Any such change to transaction validating protocols could have a material adverse effect on our business, financial condition and results of operations.

If we fail to accurately estimate the factors upon which we base our contract pricing, we may generate less profit than expected or incur losses on those contracts, which could have a material adverse effect on our business, financial condition and results of operations.

Our hosting contracts are generally priced on the basis of estimated power consumption by our clients, along with other costs of service, as adjusted for actual costs. Our ability to earn a profit on such contracts requires that we accurately estimate the costs involved and outcomes likely to be achieved and assess the probability of generating sufficient hosting and colocation capacity within the contracted time period. In addition, we may not be able to obtain all expected benefits, including tax abatements or government incentives offered in opportunity zones. The inability to accurately estimate the factors upon which we base our contract pricing could have a material adverse effect on our business, financial condition and results of operations.

Any failure in the critical systems of our hosting facilities or services we provide could lead to disruptions in our and our customers' businesses and could harm our reputation and result in financial penalty and legal liabilities, which would reduce our revenue and have a material adverse effect on our business, financial condition and results of operations.

The critical systems of the hosting facilities we operate and the services we provide are subject to failure. Any failure in the critical systems of any hosting facility we operate or services that we provide, including a breakdown in critical plant, equipment or services, routers, switches or other equipment, power supplies or network connectivity, whether or not within our control, could result in service interruptions impacting our customers as well as equipment damage, which could significantly disrupt the normal business operations of our customers, harm our reputation and reduce our revenue. Any failure or downtime in one of the facilities that we operate impact mining rewards generated by us and reduce the profitability of our customers. The total destruction or severe impairment of any of the facilities we operate could result in significant downtime of our services and loss of customer data. Since our ability to attract and retain customers depends on our ability to provide highly reliable service, even minor interruptions in our service could harm our reputation and negatively

impact our revenue and profitability. The services we provide are subject to failures resulting from numerous factors, including:

- power loss;
- equipment failure;
- human error or accidents;
- theft, sabotage and vandalism;
- failure by us or our suppliers to provide adequate service or maintain our equipment;
- network connectivity downtime and fiber cuts;
- service interruptions resulting from server relocation;
- · security breaches of our infrastructure;
- improper building maintenance by us;
- · physical, electronic and cybersecurity breaches;
- · animal incursions;
- fire, earthquake, hurricane, tornado, flood and other natural disasters;
- extreme temperatures;
- water damage;
- · public health emergencies; and
- terrorism.

Moreover, service interruptions and equipment failures may expose us to potential legal liability. As our services are critical to our customers' business operations, any disruption in our services could result in lost profits of or other indirect or consequential damages to our customers. Although our customer contracts typically contain provisions limiting our liability for breach of such agreements, there can be no assurance that a court would enforce any contractual limitations on our liability in the event that one of our customers brings a lawsuit against us as the result of a service interruption that they may ascribe to us. The outcome of any such lawsuit would depend on the specific facts of the case and any legal and policy considerations that we may not be able to mitigate. In such cases, we could be liable for substantial damage awards, which would as a result have a material adverse effect on our business, financial condition and results of operations.

We generate significant revenue from a limited number of hosting facilities in Kentucky, Georgia, North Carolina and North Dakota and a significant disruption to operations in this region could have a material adverse effect our business, financial condition and results of operations.

A significant portion of our critical business operations are concentrated in Kentucky, Georgia, North Carolina and North Dakota. A significant disruption to facilities in this region could materially and adversely affect our operations. Additional facilities and development of new facilities, however, will require additional capital investment to reach full "build out." An event beyond our control, including, but not limited to, an act of God (including, but not limited to, fires, explosions, earthquakes, drought, hurricanes, tidal waves, floods and other extreme weather events), war, hostilities (whether war is declared or not), acts of foreign enemies, embargo, rebellion, revolution, insurrection, military or usurped power, civil war, contamination by radioactivity or chemical weapon, riot, strikes, protests, lockouts, disorder, acts or threats of terrorism, pandemics or other catastrophic events that result in the destruction or disruption of any of our critical business or IT systems could severely affect our ability to conduct normal business operations, and, as a result, could have a material adverse effect on our business, financial condition and results of operations.

Our success is dependent on the ability of our management team and our ability to attract, develop, motivate and retain other well-qualified employees, which may be more difficult, costly or time-consuming than expected.

Our success depends largely on the development and execution of our business strategy by our senior management team. We cannot assure you that our management will work well together, work well with our other existing employees or successfully execute our business strategy in the near-term or at all, which could have a material adverse effect on our business, financial condition and results of operations.

Our future success also depends on our continuing ability to attract, develop, motivate and retain highly qualified and skilled directors and other employees. In particular, it is difficult to locate experienced executives in our industry and offer them competitive salaries at this stage in our development. We may be unable to retain our directors, senior executives and key personnel or attract and retain new directors, senior executives and key personnel in the future, any of which could have a material adverse effect on our business, financial condition and results of operations.

Competition for employees is intense, and we may not be able to attract and retain the qualified and skilled employees needed to support our business, which in turn could have a material adverse effect on our business, financial condition and results of operation.

We believe our success depends on the efforts and talent of our employees, including hosting facility design, construction management, operations, data processing, engineering, IT, risk management and sales and marketing personnel. Our future success depends on our continued ability to attract, develop, motivate and retain qualified and skilled employees. Competition for highly skilled personnel is extremely intense. We may not be able to hire and retain these personnel at compensation levels consistent with our existing compensation and salary structure. Some of the companies with which we compete for experienced employees have greater resources than we have and may be able to offer more attractive terms of employment.

In addition, we invest significant time and expenses in training our employees, which increases their value to competitors who may seek to recruit them. If we fail to retain our employees, we could incur significant expenses in hiring and training their replacements, and the quality of our services and our ability to serve our customers could diminish, resulting in a material adverse effect on our business, financial condition and results of operations.

We may be vulnerable to security breaches, which could disrupt our operations and have a material adverse effect on our business, financial condition and results of operations.

A party who is able to compromise the physical security measures protecting our hosting facilities could cause interruptions or malfunctions in our operations and misappropriate our property or the property of our customers. As we provide assurances to our customers that we provide the highest level of security, such a compromise could be particularly harmful to our brand and reputation. We may be required to expend significant capital and resources to protect against such threats or to alleviate problems caused by breaches in security. As techniques used to breach security change frequently and are often not recognized until launched against a target, we may not be able to implement new security measures in a timely manner or, if and when implemented, we may not be certain whether these measures could be circumvented. Any breaches that may occur could expose us to increased risk of lawsuits, regulatory penalties, loss of existing or potential customers, harm to our reputation and increases in our security costs, which could have a material adverse effect on our business, financial condition and results of operations.

In addition, any assertions of alleged security breaches or systems failure made against us, whether true or not, could harm our reputation, cause us to incur substantial legal fees and have a material adverse effect on our business, financial condition and results of operations. Whether or not any such assertion actually develops into

litigation, our management may be required to devote significant time and attention to dispute resolution (through litigation, settlement or otherwise), which would detract from our management's ability to focus on our business. Any such resolution could involve the payment of damages or expenses by us, which may be significant. In addition, any such resolution could involve our agreement with terms that restrict the operation of our business. Any such resolution, including the resources exhausted in connection therewith, could have a material adverse effect on our business, financial condition and results of operations.

Furthermore, security breaches, computer malware and computer hacking attacks have been a prevalent concern in the Bitcoin exchange market since the launch of the Bitcoin network. Any security breach caused by hacking, which involves efforts to gain unauthorized access to information or systems, or to cause intentional malfunctions or loss or corruption of data, software, hardware or other computer equipment, and the inadvertent transmission of computer viruses, could harm our business operations or result in loss of our assets.

We are subject to litigation risks.

We may be subject to litigation arising out of our operations. Damages claimed under such litigation may be material, and the outcome of such litigation may materially impact our operations, and the value of the common shares. While we will assess the merits of any lawsuits and defend such lawsuits accordingly, we may be required to incur significant expense or devote significant financial resources to such defenses. In addition, the adverse publicity surrounding such claims may have a material adverse effect on our operations.

We may be exposed to cybersecurity threats and hacks, which could have a material adverse effect on our business, financial condition and results of operations.

The threats to network and data security are increasingly diverse and sophisticated. Despite our efforts and processes to prevent breaches, our computer servers and computer systems may be vulnerable to cybersecurity risks, including denial-of-service attacks, physical or electronic break-ins, employee theft or misuse and similar disruptions from unauthorized tampering with our computer servers and computer systems. The preventive actions we take to reduce the risk of cyber incidents and protect our information technology and networks may be insufficient to repel a major cyber-attack in the future. To the extent that any disruption or security breach results in a loss or damage to our network, in unauthorized disclosure of confidential information or in a loss of our digital assets, it could cause significant damage to our reputation, lead to claims against us and ultimately have a material adverse effect on our business, financial condition and results of operations. Additionally, we may be required to incur significant costs to protect against damage caused by these disruptions or security breaches in the future.

Our future success depends on our ability to keep pace with rapid technological changes that could make our current or future technologies less competitive or obsolete.

Rapid, significant and disruptive technological changes continue to impact our industry. The infrastructure at our hosting facilities may become less marketable due to demand for new processes and technologies, including, without limitation: (i) new processes to deliver power to, or eliminate heat from, computer systems; (ii) customer demand for additional redundancy capacity; (iii) new technology that permits higher levels of critical load and heat removal than our hosting facilities are currently designed to provide; (iv) an inability of the power supply to support new, updated or upgraded technology; and (v) a shift to more power-efficient transaction validation protocols. In addition, the systems that connect our hosting facilities to the Internet and other external networks may become insufficient, including with respect to latency, reliability and diversity of connectivity. We may not be able to adapt to changing technologies, identify and implement new alternatives successfully or meet customer demands for new processes or technologies in a timely and cost-effective manner, if at all, which would have a material adverse effect on our business, financial condition and results of operations.

Even if we succeed in adapting to new processes and technologies, there is no assurance that our use of such new processes or technology would have a positive impact on our financial performance. For example, we could

incur substantial additional costs if we needed to materially improve our hosting center infrastructure through the implementation of new systems or new server technologies that require levels of critical load and heat removal that our facilities are not currently designed to provide. In addition, if one of our new offerings were competitive to our prior offerings and represented an adequate or superior alternative, customers could decide to abandon prior offerings that produce higher revenue or better margins for the new offering. Therefore, the adaptation to new processes and technologies could result in lower revenue, lower margins and/or higher costs, which could have a material adverse effect on our business, financial condition and results of operations.

In addition, our competitors or others might develop technologies that are more effective than our current or future technologies, or that render our technologies less competitive or obsolete. Further, many of our competitors may have superior financial and human resources deployed toward research and development efforts. We may not be able to effectively keep pace with relevant technological changes. If competitors introduce superior technologies for hosting operations or transaction processing, and we cannot make upgrades to our hardware or software to remain competitive, it could have a material adverse effect on our business, financial condition and results of operations.

Our compliance and risk management methods might not be effective and may result in outcomes that could adversely affect our reputation, operating results, and financial condition.

Our ability to comply with applicable complex and evolving laws, regulations, and rules is largely dependent on the establishment and maintenance of our compliance, audit, and reporting systems, as well as our ability to attract and retain qualified compliance and other risk management personnel. While we plan to devote significant resources to develop policies and procedures to identify, monitor and manage our risks, we cannot assure you that our policies and procedures will always be effective against all types of risks, including unidentified or unanticipated risks, or that we will always be successful in monitoring or evaluating the risks to which we are or may be exposed in all market environments.

We may infringe on third-party intellectual property rights or other proprietary rights, which could have a material adverse effect on our business, financial condition and results of operations.

Our commercial success depends on our ability to operate without infringing third-party intellectual property rights or other proprietary rights. For example, there may be issued patents of which we are not aware that our services or products infringe on. Also, there may be patents we believe we do not infringe on, but that we may ultimately be found to by a court of law or government regulatory agency. Moreover, patent applications are in some cases maintained in secrecy until patents are issued. Because patents can take many years to issue, there may be currently pending applications of which we are unaware that may later result in issued patents that our services or products allegedly infringe on.

If a third party brings any claim against us based on third-party intellectual property rights and/or other proprietary rights, we will be required to spend significant resources to defend and challenge such claim, as well as to invalidate any such rights. Any such claim, if initiated against us, whether or not it is resolved in our favor, could result in significant expense to us, and divert the efforts of our technical and management personnel, which could have a material adverse effect on our business, financial condition and results of operations.

The further development and acceptance of cryptographic and algorithmic protocols governing transaction validation and the issuance of, and transactions in, digital assets are subject to a variety of factors that are difficult to evaluate. The slowing or stoppage of development or acceptance of blockchain networks and digital assets would have an adverse material effect on the successful development of the mining operation and value of mined digital assets.

The use of digital assets to, among other things, buy and sell goods and services, is part of a new and rapidly evolving industry that employs digital assets based upon a computer-generated mathematical and/or

cryptographic protocol. The future of this industry is subject to a high degree of uncertainty. The factors affecting the further development of this industry include, but are not limited to:

- continued worldwide growth in the adoption and use of digital assets and blockchain technologies;
- government and quasi-government regulation of digital assets and their use, or restrictions on or regulation of access to and operations of digital asset transaction processing;
- changes in consumer demographics and public tastes and preferences;
- the maintenance and development of the open-source software protocols or similar digital asset systems;
- the availability and popularity of other forms or methods of buying and selling goods and services, or trading assets including new means of using fiat currencies;
- general economic conditions and the regulatory environment relating to digital assets; and
- negative consumer perception of digital assets, including digital assets specifically and digital assets generally.

A decline in the popularity or acceptance of digital assets could materially impact us or our potential hosting customers, which could have a material adverse effect on our business, financial condition and results of operations.

Our ability to use net operating losses to offset future taxable income may be subject to certain limitations.

As of December 31, 2020, we had U.S. federal and state net operating losses ("NOLs") of approximately \$46.6 million and \$32.8 million, respectively, available to offset future taxable income, some of which begin to expire in 2033. U.S. federal NOLs incurred in taxable years beginning after December 31, 2017, can be carried forward indefinitely, but the deductibility of U.S. federal NOLs in taxable years beginning after December 31, 2020, is subject to certain limitations. For instance, a lack of future taxable income would adversely affect our ability to utilize such NOLs before they expire.

In addition, under the Internal Revenue Code of 1986, as amended (the "Code") substantial changes in our ownership may limit the amount of pre-change NOLs that can be utilized annually in the future to offset taxable income. Section 382 of the Code imposes limitations on a company's ability to use its NOLs if one or more stockholders or groups of stockholders that own at least 5% of the company's stock increase their ownership by more than 50 percentage points over their lowest ownership percentage within a rolling three-year period. Similar rules may apply under state tax laws. As such, prior changes in our ownership or future changes in our ownership may limit our ability to utilize our NOLs. We have not yet determined the cumulative change in our ownership resulting from the Business Combination or any resulting limitations on our ability to utilize our NOLs or other tax attributes. Subsequent statutory or regulatory changes in respect of the utilization of NOLs for U.S. federal or state tax purposes, such as suspensions on the use of NOLs or limitations on the deductibility of NOL carryforwards, or other unforeseen reasons, may result in our existing NOLs expiring or otherwise being unavailable to offset future taxable income. For these reasons, we may not be able to utilize a material portion of our NOLs, even if we have taxable income.

We may not be able to adequately protect our intellectual property rights and other proprietary rights, which could have a material adverse effect on business, financial condition and results of operations.

We may not be able to obtain broad protection in the United States or internationally for all of our existing and future intellectual property and other proprietary rights, and we may not be able to obtain effective protection for our intellectual property and other proprietary rights in every country in which we operate. Protecting our intellectual property rights and other proprietary rights may require significant expenditure of our financial,

managerial and operational resources. Moreover, the steps that we may take to protect our intellectual property and other proprietary rights may not be adequate to protect such rights or prevent third parties from infringing or misappropriating such rights. Any of our intellectual property rights and other proprietary rights, whether registered, unregistered, issued or unissued, may be challenged by others or invalidated through administrative proceedings and/or litigation.

We may be required to spend significant resources to secure, maintain, monitor and protect our intellectual property rights and other proprietary rights. Despite our efforts, we may not be able to prevent third parties from infringing upon, misappropriating or otherwise violating our intellectual property rights and other proprietary rights. We may initiate claims, administrative proceedings and/or litigation against others for infringement, misappropriation or violation of our intellectual property rights or other proprietary rights to enforce and/or maintain the validity of such rights. Any such action, if initiated, whether or not it is resolved in our favor, could result in significant expense to us, and divert the efforts of our technical and management personnel, which may have a material adverse effect on our business, financial condition and results of operations.

Risks Related to Core's Limited Operating History and Early Stage of Growth

We operate in a rapidly developing industry and have an evolving business model with a limited history of generating revenue from our services. In addition, our evolving business model increases the complexity of our business, which makes it difficult to evaluate our future business prospects and could have a material adverse effect on our business, financial condition and results of operations.

Our business model has evolved in the past and continues to do so. We previously were engaged primarily in providing infrastructure hosting services to our commercial counterparties after being founded to engage in the business of verifying and confirming transactions on a blockchain, also known as transaction processing, or "mining." Recently, we decided to substantially increase our focus on mining blockchain for our own account. As a result of our recent acquisition of Blockcap, we significantly expanded our self-mining operations. We may adjust our business model further from time to time, including trying to offer additional types of products or services, such as a blockchain application designed by us, blockchain services and other related businesses, or entering into strategic partnerships or acquisitions. We have generated limited revenue from such services, and we do not know whether any of them will be successful. The evolution of and modifications to our business strategy will continue to increase the complexity of our business and placed significant strain on our management, personnel, operations, systems, technical performance and financial resources. Future additions to or modifications of our business strategy are likely to have similar effects. Further, any new services that we offer that are not favorably received by the market could damage our reputation or our brand. There can be no assurance that we will ever generate sufficient revenues or achieve profitably in the future or that we will have adequate working capital to meet our obligations as they become due.

We cannot be certain that our current business strategy or any new or revised business strategies will be successful or that we will successfully address the risks we face. In the event that we do not effectively evaluate future business prospects, successfully implement new strategies or adapt to our evolving industry, it will have a material adverse effect on our business, financial condition and results of operations.

We may not be able to compete effectively against our current and future competitors, which could have a material adverse effect on our business, financial condition and results of operations.

The digital asset mining industry is highly innovative, rapidly evolving and characterized by healthy competition, experimentation, frequent introductions of new products and services and uncertain and evolving industry and regulatory requirements. We expect competition to further intensify in the future as existing and new competitors introduce new products or enhance existing products. We compete against a number of companies operating both within the United States and abroad, that have greater financial and other resources and that focus on digital asset mining, including businesses focused on developing substantial Bitcoin mining

operations. If we are unable to compete successfully, or if competing successfully requires us to take costly actions in response to the actions of our competitors, our business, operating results and financial condition could be adversely affected.

We compete with a range of hosting providers and blockchain providers for some or all of the services we offer. We face competition from numerous developers, owners and operators in the blockchain industry, including technology companies, such as hyperscale cloud players, managed service providers and real estate investment trusts ("REITs"), some of which own or lease properties similar to ours, or may do so in the future, in the same submarkets in which our properties are located. Cloud offerings may also influence our customers to move workloads to cloud providers, which may reduce the services they obtain from us. Our current and future competitors may vary from us in size, service offerings and geographic presence.

Competition is primarily centered on reputation and track record; design, size, quality, available power and geographic coverage of hosting space; quality of installation and customer equipment repair services; relationships with equipment manufacturers and ability to obtain replacement parts; technical and software expertise; and financial strength and price. Some of our current and future competitors may have greater brand recognition, longer operating histories, stronger marketing, technical and financial resources and access to greater and less expensive power than we do.

In addition, many companies in the industry are consolidating, which could further increase the market power of our competitors. As a result, some of our competitors may be able to:

- identify and acquire desirable properties that we are interested in from developers;
- offer hosting services at prices below current market rates or below the prices we currently charge our customers;
- bundle colocation services with other services or equipment they provide at reduced prices;
- develop superior products or services, gain greater market acceptance and expand their service offerings more efficiently or rapidly;
- adapt to new or emerging technologies and changes in customer requirements more quickly;
- take advantage of acquisition and other opportunities more readily; and
- adopt more aggressive pricing policies and devote greater resources to the promotion, marketing and sales of their services.

We operate in a competitive market, and we face pricing pressure with respect to our hosting services. Prices for our hosting services are affected by a variety of factors, including supply and demand conditions and pricing pressures from our competitors. We may be required to lower our prices to remain competitive, which may decrease our margins and could have a material adverse effect on our business, financial condition and results of operations.

In addition, we also face significant competition from other users and/or companies that are processing transactions on one or more digital asset networks, as well as other potential financial vehicles, including securities, derivatives or futures backed by, or linked to, digital assets through entities similar to us, such as exchange-traded funds. Market and financial conditions, and other conditions beyond our control, may make it more attractive to invest in other financial vehicles, or to invest in digital assets directly. Such events could have a material adverse effect on our business, financial condition and results of operations and potentially the value of any digital assets we hold or expect to acquire for our own account.

We have experienced difficulties in establishing relationships with banks, leasing companies, insurance companies and other financial institutions that are willing to provide us with customary financial products and services, which could have a material adverse effect on our business, financial condition and results of operations.

As an early stage company with operations focused in the digital asset transaction processing industry, we have in the past experienced, and may in the future experience, difficulties in establishing relationships with banks, leasing companies, insurance companies and other financial institutions that are willing to provide us with customary leasing and financial products and services, such as bank accounts, lines of credit, insurance and other related services, which are necessary for our operations. To the extent a significant portion of our business consists of digital asset transaction mining, processing or hosting, we may in the future continue to experience difficulty obtaining additional financial products and services on customary terms, which could have a material adverse effect on our business, financial condition and results of operations.

Risks Related to Regulatory Framework

If we were deemed an "investment company" under the Investment Company Act of 1940, as amended (the "1940 Act"), applicable restrictions could make it impractical for us to continue our business as contemplated and could have a material adverse effect on our business.

An issuer will generally be deemed to be an "investment company" for purposes of the 1940 Act if:

- it is an "orthodox" investment company because it is or holds itself out as being engaged primarily, or proposes to engage primarily, in the business of investing, reinvesting or trading in securities; or
- it is an inadvertent investment company because, absent an applicable exemption, it owns or proposes to acquire "investment securities" having a value exceeding 40% of the value of its total assets (exclusive of U.S. government securities and cash items) on an unconsolidated basis.

We believe that we are not and will not be primarily engaged in the business of investing, reinvesting or trading in securities, and we do not hold ourselves out as being engaged in those activities. We intend to hold ourselves out as a digital asset mining business. Accordingly, we do not believe that we are an "orthodox" investment company as described in the first bullet point above.

While certain digital assets may be deemed to be securities, we do not believe that certain other digital assets, in particular Bitcoin, are securities; therefore, we believe that less than 40% of our total assets (exclusive of U.S. government securities and cash items) on an unconsolidated basis will comprise digital assets that could be considered investment securities. Accordingly, we do not believe that we are an inadvertent investment company by virtue of the 40% inadvertent investment company test as described in the second bullet point above. Although we do not believe any of the digital assets we may own, acquire or mine are securities, there is still some regulatory uncertainty on the subject, see "—There is no one unifying principle governing the regulatory status of digital assets nor whether digital assets are securities in any particular context. Regulatory changes or actions in one or more countries may alter the nature of an investment in us or restrict the use of digital assets in a manner that adversely affects our business, prospects or operations." If certain digital assets, including Bitcoin, were to be deemed securities, and consequently, investment securities by the SEC, we could be deemed an inadvertent investment company. Similarly, if we were to acquire digital assets deemed investment securities to hold for our own account or to engage in certain transactions, such as loan or repurchase transactions, we could be deemed an inadvertent investment company.

If we were to be deemed an inadvertent investment company, we may seek to rely on Rule 3a-2 under the 1940 Act, which allows an inadvertent investment company a grace period of one year from the earlier of (a) the date on which the issuer owns securities and/or cash having a value exceeding 50% of the issuer's total assets on either a consolidated or unconsolidated basis or (b) the date on which the issuer owns or proposes to acquire investment securities having a value exceeding 40% of the value of such issuer's total assets (exclusive of U.S.

government securities and cash items) on an unconsolidated basis. We are putting in place policies that we expect will work to keep the investment securities held by us at less than 40% of our total assets, which may include acquiring assets with our cash, liquidating our investment securities or seeking no-action relief or exemptive relief from the SEC if we are unable to acquire sufficient assets or liquidate sufficient investment securities in a timely manner. As Rule 3a-2 is available to an issuer no more than once every three years, and assuming no other exclusion were available to us, we would have to keep within the 40% limit for at least three years after we cease being an inadvertent investment company. This may limit our ability to make certain investments or enter into joint ventures that could otherwise have a positive impact on our earnings. In any event, we do not intend to become an investment company engaged in the business of investing and trading securities.

Finally, we believe we are not an investment company under Section 3(b)(1) of the 1940 Act because we are primarily engaged in a non-investment company business.

The 1940 Act and the rules thereunder contain detailed parameters for the organization and operations of investment companies. Among other things, the 1940 Act and the rules thereunder limit or prohibit transactions with affiliates, impose limitations on the issuance of debt and equity securities, prohibit the issuance of stock options, and impose certain governance requirements. We intend to continue to conduct our operations so that we will not be deemed to be an investment company under the 1940 Act. However, if anything were to happen that would cause us to be deemed to be an investment company under the 1940 Act, requirements imposed by the 1940 Act, including limitations on our capital structure, ability to transact business with affiliates and ability to compensate key employees, could make it impractical for us to continue our business as currently conducted, impair the agreements and arrangements between and among us and our senior management team and materially and adversely affect our business, financial condition and results of operations.

Any change in the interpretive positions of the SEC or its staff with respect to digital asset mining firms could have a material adverse effect on us.

We intend to conduct our operations so that we are not required to register as an investment company under the 1940 Act. Specifically, we do not believe that digital assets, are securities. The SEC Staff has not provided guidance with respect to the treatment of these assets under the 1940 Act. To the extent the SEC Staff publishes new guidance with respect to these matters, we may be required to adjust our strategy or assets accordingly. There can be no assurance that we will be able to maintain our exclusion from registration as an investment company under the 1940 Act. In addition, as a consequence of our seeking to avoid the need to register under the 1940 Act on an ongoing basis, we may be limited in our ability to engage in digital asset mining operations or otherwise make certain investments or engage in certain transactions, and these limitations could result in our holding assets we may wish to sell or selling assets we may wish to hold, which could materially and adversely affect our business, financial condition and results of operations.

If regulatory changes or interpretations of our activities require our registration as a money services business ("MSB") under the regulations promulgated by the Financial Crimes Enforcement Network ("FinCEN") under the authority of the U.S. Bank Secrecy Act, or otherwise under state laws, we may incur significant compliance costs, which could be substantial or cost-prohibitive. If we become subject to these regulations, our costs in complying with them may have a material negative effect on our business and the results of our operations.

To the extent that our activities cause us to be deemed an MSB under the regulations promulgated by FinCEN under the authority of the U.S. Bank Secrecy Act, we may be required to comply with FinCEN regulations, including those that would mandate us to implement anti-money laundering programs, make certain reports to FinCEN and maintain certain records.

To the extent that our activities would cause us to be deemed a "money transmitter" ("MT") or equivalent designation, under state law in any state in which we may operate, we may be required to seek a license or

otherwise register with a state regulator and comply with state regulations that may include the implementation of anti-money laundering programs, maintenance of certain records and other operational requirements. For example, in August 2015, the New York State Department of Financial Services enacted the first U.S. regulatory framework for licensing participants in "virtual currency business activity." The regulations, known as the "BitLicense," are intended to focus on consumer protection and regulate the conduct of businesses that are involved in "virtual currencies" in New York or with New York customers and prohibit any person or entity involved in such activity to conduct activities without a license.

Such additional federal or state regulatory obligations may cause us to incur extraordinary expenses. Furthermore, we may not be capable of complying with certain federal or state regulatory obligations applicable to MSBs and MTs. If we are deemed to be subject to and determine not to comply with such additional regulatory and registration requirements, we may act to dissolve and liquidate.

There is no one unifying principle governing the regulatory status of digital assets nor whether digital assets are securities in any particular context. Regulatory changes or actions in one or more countries may alter the nature of an investment in us or restrict the use of digital assets in a manner that adversely affects our business, prospects or operations.

As digital assets have grown in both popularity and market size, governments around the world have reacted differently, with certain governments deeming digital assets illegal, and others allowing their use and trade without restriction. In some jurisdictions, such as in the U.S., digital assets are subject to extensive, and in some cases overlapping, unclear and evolving regulatory requirements.

Bitcoin is the oldest and most well-known form of digital asset. Bitcoin and other forms of digital assets have been the source of much regulatory consternation, resulting in differing definitional outcomes without a single unifying statement. Bitcoin and other digital assets are viewed differently by different regulatory and standards setting organizations globally as well as in the United States on the federal and state levels. For example, the Financial Action Task Force considers a digital asset as currency or an asset, and the U.S. Internal Revenue Service (the "IRS") considers a digital asset as property and not currency. Further, the IRS applies general tax principles that apply to property transactions to transactions involving virtual currency.

Furthermore, in the several applications to establish an exchange traded fund ("ETF") of digital assets, and in the questions raised by the Staff under the 1940 Act, no clear principles emerge from the regulators as to how they view these issues and how to regulate digital assets under the applicable securities acts. It has been widely reported that the SEC has recently issued letters and requested various ETF applications be withdrawn because of concerns over liquidity and valuation and unanswered questions about absence of reporting and compliance procedures capable of being implemented under the current state of the markets for exchange traded funds. On April 20, 2021, the U.S. House of Representatives passed a bipartisan bill titled "Eliminate Barriers to Innovation Act of 2021" (H.R. 1602). If passed by the Senate and enacted into law, the bipartisan bill would create a digital assets working group to evaluate the current legal and regulatory framework around digital assets in the United States and define when the SEC may have jurisdiction over a particular token or digital asset (i.e., when it is a security) and when the U.S. Commodity Futures Trading Commission (the "CFTC") may have jurisdiction (i.e., when it is a commodity).

If regulatory changes or interpretations require the regulation of Bitcoin or other digital assets under the securities laws of the United States or elsewhere, including the Securities Act of 1933, as amended (the "Securities Act"), the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the 1940 Act or similar laws of other jurisdictions and interpretations by the SEC, the CFTC, the IRS, Department of Treasury or other agencies or authorities, we may be required to register and comply with such regulations, including at a state or local level. To the extent that we decide to continue operations, the required registrations and regulatory compliance steps may result in extraordinary expense or burdens to us. We may also decide to cease certain operations and change our business model. For example, while we do not anticipate engaging in digital asset-

denominated loan transactions, we would be required to assess the application of, and comply with, federal securities laws in connection with those or similar transactions. Any disruption of our operations in response to the changed regulatory circumstances may be at a time that is disadvantageous to us

Current and future legislation and SEC-rulemaking and other regulatory developments, including interpretations released by a regulatory authority, may impact the manner in which Bitcoin or other digital assets are viewed or treated for classification and clearing purposes. In particular, Bitcoin and other digital assets may not be excluded from the definition of "security" by SEC rulemaking or interpretation requiring registration of all transactions unless another exemption is available, including transacting in Bitcoin or digital assets among owners and require registration of trading platforms as "exchanges."

Furthermore, the SEC may determine that certain digital assets or interests, for example tokens offered and sold in initial coin offerings ("ICO"), may constitute securities under the "Howey" test as stated by the United States Supreme Court. As such, ICO offerings would require registration under the Securities Act or an available exemption therefrom for offers or sales in the United States to be lawful. Section 5(a) of the Securities Act provides that, unless a registration statement is in effect as to a security, it is unlawful for any person, directly or indirectly, to engage in the offer or sale of securities in interstate commerce. Section 5(c) of the Securities Act provides a similar prohibition against offers to sell, or offers to buy, unless a registration statement has been filed. Although, we do not intend to be engaged in the offer or sale of securities in the form of ICO offerings, and we do not believe our planned mining activities would require registration for us to conduct such activities and accumulate digital assets. The SEC, CFTC, Nasdaq, IRS or other governmental or quasi-governmental agency or organization may conclude that our activities involve the offer or sale of "securities," or ownership of "investment securities," and we may be subject to regulation or registration requirements under various federal laws and related rules. Such regulation or the inability to meet the requirements to continue operations, would have a material adverse effect on our business and operations. We may also face similar issues with various state securities regulators who may interpret our actions as subjecting us to regulation, or requiring registration, under state securities laws, banking laws, or money transmitter and similar laws, which are also an unsettled area or regulation that exposes us to risks.

Regulatory changes or actions may restrict the use of digital assets or the operation of digital asset networks in a manner that may require us to cease certain or all operations, which could have a material adverse effect on our business, financial condition and results of operations.

Recently, there has been a significant amount of regulatory attention directed toward digital assets, digital asset networks and other industry participants by United States federal and state governments, foreign governments and self-regulatory agencies. For example, as digital assets such as bitcoin have grown in popularity and in market size, the Federal Reserve Board, U.S. Congress and certain U.S. agencies (e.g., FinCEN, the SEC, the CFTC and the Federal Bureau of Investigation) have begun to examine the operations of the Bitcoin network, Bitcoin users and Bitcoin exchange markets.

In addition, local state regulators such as the Texas State Securities Board, the Massachusetts Securities Division of the Office of the Secretary of the Commonwealth, the New Jersey Bureau of Securities, the North Carolina Secretary of State's Securities Division and the Vermont Department of Financial Regulation have initiated actions against, and investigations of, individuals and companies involved in digital assets.

Also, in March 2018, the South Carolina Attorney General Office's Security Division issued a cease-and-desist order against Genesis Mining and Swiss Gold Global, Inc., stating that both companies were to stop doing business in South Carolina and are permanently barred from offering securities in the state in the future since they offered unregistered securities via cloud mining contracts under the South Carolina Uniformed Securities Act of 2005, S.C. Code Ann. § 35-1-101, et seq. (the order against Genesis Mining was subsequently withdrawn).

Further, the North Carolina Secretary of State's Securities Division issued in March 2018 a Temporary Cease and Desist Order against Power Mining Pool (made permanent pursuant to a Final Order on April 19, 2018), ordering it to cease and desist, among other things, offering "mining pool shares," which were deemed "securities" under N.C. Gen. Stat. 78A-2(11), in North Carolina until they are registered with the North Carolina Secretary of State or are offered for sale pursuant to an exemption from registration under the North Carolina Securities Act, N.C. Gen. Stat. Chapter 78A.

Additionally, we rely on third-party mining pool service providers for mining revenue payouts from our mining operation, and certain of our potential hosting customers could be involved in, or could issue, cloud mining contracts or mining pool shares, and any regulatory restrictions on their practices could significantly reduce demand for our hosting services. Furthermore, it is possible that laws, regulations or directives that affect digital assets, digital asset transaction processing or blockchain server hosting may change in a manner that may adversely affect our ability to conduct our business and operations in the relevant jurisdiction.

In addition, various foreign jurisdictions either have adopted or may adopt laws, regulations or directives that affect digital assets, digital asset networks and their users and hosting service providers that fall within such jurisdictions' regulatory scope. Such laws, regulations or directives may conflict with those of the United States, may negatively impact the acceptance of digital assets by users, merchants and service providers outside of the United States and may therefore impede the growth of digital asset use. A number of countries, including India, China, South Korea and Russia, among others, currently have a more restrictive stance toward digital assets and, thereby, have reduced the rate of expansion of digital asset use, as well as digital asset transaction processing, in each of those countries. For example, in January 2018, several media publications reported that a Chinese multiagency government task force overseeing risk in Internet finance issued a notice ordering local authorities to guide the shutdown of digital asset transaction processing in China. However, the People's Bank of China immediately refuted such reports, indicating that digital asset transaction processing is still permitted in China. As a result of such conflicting positions taken within the Chinese government, a number of digital asset transaction processing operators have moved their operations from China to other jurisdictions in order to build in more regulatory certainty in their operations.

Governments may in the future take regulatory actions that prohibit or severely restrict the right to acquire, own, hold, sell, use or trade digital assets or to exchange digital assets for fiat currency. Ownership of, holding or trading in digital assets may then be considered illegal and subject to sanction. Governments may also take regulatory action that may increase the cost and/or subject digital asset mining companies to additional regulation.

By extension, similar actions by governments may result in the restriction of the acquisition, ownership, holding, selling, use or trading in the capital stock of digital asset mining companies, including our common stock. Such a restriction could result in us liquidating our digital asset inventory at unfavorable prices and may adversely affect our shareholders. The effect of any regulatory change, either by federal, state, local or foreign governments or any self-regulatory agencies, on us or our potential hosting customers is impossible to predict, but such change could be substantial and may require us or our potential hosting customers to cease certain or all operations and could have a material adverse effect on our business, financial condition and results of operations.

Current and future legislation and rulemaking regarding digital assets may result in extraordinary, non-recurring expenses and could have a material adverse effect on our business, financial condition and results of operations.

Current and future legislation and rulemaking by the CFTC and SEC or other regulators, including interpretations released by a regulatory authority, may impact the manner in which digital assets are treated. For example, digital assets derivatives are not excluded from the definition of "commodity future" by the CFTC. Furthermore, according to the CFTC, digital assets fall within the definition of a commodity under the Commodities Exchange Act (the "CEA") and as a result, we may be required to register and comply with

additional regulations under the CEA, including additional periodic reporting and disclosure standards and requirements. We may also be required to register as a commodity pool operator and to register as a commodity pool with the CFTC through the National Futures Association. If we are required to register with the CFTC or another governmental or self-regulatory authority, the scope of our business and operations may be constrained by the rules of such authority and we may be forced to incur additional expenses in the form of licensing fees, professional fees and other costs of compliance.

The SEC has issued guidance and made numerous statements regarding the application of securities laws to digital assets. For example, on July 25, 2017, the SEC issued a Report of Investigation (the "Report") which concluded that tokens offered and sold by the Decentralized Autonomous Organization ("DAO"), a digital decentralized autonomous organization and investor-directed venture capital fund for digital assets, were issued for the purpose of raising funds. The Report concluded that these tokens were "investment contracts" within the meaning of Section 2(a)(1) of the Securities Act and Section 3(a)(10) of the Exchange Act, and therefore securities subject to the federal securities laws. In December 2017, the SEC issued a cease-and-desist letter to Munchee Inc., ordering that the company stop its initial coin offering of MUN Tokens on the grounds that it failed to file a registration statement or qualify for an exemption from registration. Similar to the tokens issued by the DAO, the SEC found that the MUN Tokens satisfied the definition of an "investment contract," and were therefore subject to the federal securities laws. In February 2018, both the SEC and CFTC further reiterated their concerns regarding digital assets in written testimony to the Senate Banking, Housing and Urban Affairs Committee. On March 7, 2018, the SEC released a "Statement on Potentially Unlawful Online Platforms for Trading Digital Assets," and reiterated that, if a platform "offers trading of digital assets that are securities" and "operates as 'exchange,' as defined by the federal securities laws," the platform must register with the SEC as a national securities exchange or be exempt from registration. The SEC's statement serves as a notice to operators of any platforms, including secondary market trading platforms, which the SEC is actively monitoring for potentially fraudulent or manipulative behavior in the market for security tokens, as the SEC has cautioned recently in the context of ICOs. On November 16, 2018, the SEC released a "Statement on Digital Asset Securities Issuance and Trading," and emphasized that market participants must adhere to the SEC's well-established and well-functioning federal securities law framework when dealing with technological innovations, regardless of whether the securities are issued in certificated form or using new technologies, such as blockchain. This has all been followed by additional statements and guidance form the SEC including no-action letters relating to specific blockchain-based projects, and a Framework for "Investment Contract" Analysis of Digital Assets published by the Division of Corporation Finance on April 3, 2019. In an August 2021 interview, SEC Chairman Gensler signaled the SEC is contemplating a robust regulatory regime for digital assets and reiterated the SEC's position that many digital assets are unregulated securities.

The SEC has been active in asserting its jurisdiction over ICOs and digital assets and in bringing enforcement cases. The SEC has directed enforcement activity toward digital assets, and more specifically, ICOs. In September 2017, the SEC created a new division known as the "Cyber Unit" to address, among other things, violations involving distributed ledger technology and ICOs, and filed a civil complaint in the Eastern District of New York charging a businessman and two companies with defrauding investors in a pair of so-called ICOs purportedly backed by investments in real estate and diamonds (*see* Securities and Exchange Commission v. REcoin Group Foundation, LLC, et al., Civil Action NO. 17-cv- 05725 (E.D.N.Y, filed Sept. 29, 2017)). Subsequently, the SEC has filed several orders instituting cease-and-desist proceedings against (i) Carrier EQ, Inc., d/b/a AirFox and Paragon Coin, Inc. in connection with their unregistered offerings of tokens (*see* CarrierEQ, Inc., Rel. No. 33-10575 (Nov. 16, 2018) and Paragon Coin, Rel. No. 33-10574 (Nov. 16, 2018), respectively), (ii) Crypto Asset Management, LP for failing to register a hedge fund formed for the purpose of investing in digital assets as an investment company (*see* Crypto Asset Management, LP and Timothy Enneking, Rel. No. 33-10544 (Sept. 11, 2018)), (iii) TokenLot LLC for failing to register as a broker-dealer, even though it did not meet the definition of an exchange (*see* Tokenlot LLC, Lenny Kugel, and EliL. Lewitt, Rel. No. 33-10543 (Sept. 11, 2018)) and (iv) EtherDelta's founder for failing either to register as a national securities exchange or to operate pursuant to an exemption from registration as an exchange after creating a platform that clearly fell within the definition of an exchange (*see* Zachary Coburn, Rel. No. 34-84553 (Nov. 8, 2018)).

On June 4, 2019, the SEC filed a complaint in the U.S. District Court for the Southern District of New York against Kik Interactive, Inc. with respect to its September 2017 offering of Kin. According to articles published by various news outlets, the SEC has allegedly issued numerous subpoenas and information requests to technology companies, advisers and individuals involved in the digital asset space and ICOs, as part of a broad inquiry into the digital asset market.

Recently, a number of proposed ICOs have sought to rely on Regulation A and have filed with the SEC a Form 1-A covering a distribution of a digital token. Two such offerings were qualified in July 2019. In addition, some token offerings have been commenced as private securities offerings intended to be exempt from SEC registration. Further, the SEC has yet to approve for listing and trading any exchange-traded products (such as ETFs) holding digital assets. The SEC has taken various actions against persons or entities that have allegedly misused digital assets, engaged in fraudulent schemes (i.e., Ponzi scheme) and/or engaged in the sale of tokens that were deemed securities by the SEC.

Although our activities are not focused on raising capital or assisting others that do so, the federal securities laws are very broad. We cannot provide assurance as to whether the SEC will continue or increase its enforcement with respect to digital assets or ICOs, including taking enforcement action against any person engaged in the sale of unregistered securities in violation of the Securities Act or any person acting as an unregistered investment company in violation of the 1940 Act. Because the SEC has held that certain digital assets are securities based on the current rules and law, we may be required to register and comply with the rules and regulations under federal securities laws.

We cannot be certain as to how future regulatory developments will impact the treatment of digital assets under the law, including, but not limited to, whether digital assets will be classified as a security, commodity, currency and/or new or other existing classification. Such additional regulations may result in extraordinary, non-recurring expenses, thereby materially and adversely affecting an investment in us. If we determine not to comply with such additional regulatory and registration requirements, we may seek to cease certain or all of our operations. Any such action could have a material adverse effect on our business, financial condition and results of operations.

Federal or state agencies may impose additional regulatory burdens on our business. Changing laws and regulations and changing enforcement policies and priorities have the potential to cause additional expenditures, restrictions, and delays in connection with our business operations.

Federal and state laws and regulations may be subject to change or changes in enforcement policies or priorities, including changes that may result from changes in the political landscape and changing technologies. Future legislation and regulations, changes to existing laws and regulations, or interpretations thereof, or changes in enforcement policies or priorities, could require significant management attention and cause additional expenditures, restrictions, and delays in connection with our business operations.

Increasing scrutiny and changing expectations from investors, lenders, customers, government regulators and other market participants with respect to our Environmental, Social and Governance ("ESG") policies may impose additional costs on us or expose us to additional risks.

Companies across all industries and around the globe are facing increasing scrutiny relating to their ESG policies. Investors, lenders and other market participants are increasingly focused on ESG practices and in recent years have placed increasing importance on the implications and social cost of their investments. In February 2021, the Acting Chair of the SEC issued a statement directing the Division of Corporation Finance to enhance its focus on climate-related disclosure in public company filings and in March 2021 the SEC announced the creation of a Climate and ESG Task Force in the Division of Enforcement. The increased focus and activism related to ESG may hinder our access to capital, as investors and lenders may reconsider their capital investment allocation as a result of their assessment of our ESG practices. If we do not adapt to or comply with investor,

lender or other industry shareholder expectations and standards and potential government regulations, which are evolving but may relate to the suitable deployment of electric power, or which are perceived to have not responded appropriately to the growing concern for ESG issues, our reputation suffer which would have a material adverse effect on our business, financial condition and results of operations.

We may be subject to risks associated with misleading and/or fraudulent disclosure or use by the creators of digital assets.

Generally, we rely primarily on a combination of white papers and other disclosure documents prepared by the creators of applicable digital assets, as well as on our management's ability to obtain adequate information to evaluate the potential implications of transacting in these digital assets. However, such white papers and other disclosure documents and information may contain misleading and/or fraudulent statements (which may include statements concerning the creators' ability to deliver in a timely fashion the product and/or service disclosed in their white papers and other disclosure documents) and/or may not reveal any unlawful activities by the creators. Recently, there has been an increasing number of investigations and lawsuits by the SEC and the CFTC involving digital asset creators for fraud and misappropriation, among other charges. Additionally, FinCEN has increased its enforcement efforts involving digital asset creators regarding compliance with anti-money laundering and Know-Your-Customer laws.

To the extent that any of these creators make misleading and/or fraudulent disclosures or do not comply with federal, state or foreign laws, or if we are unable to uncover all material information about these digital assets and/or their creators, we may not be able to make a fully informed business decision relating to our transacting in or otherwise involving such digital assets, which could have a material adverse effect on our business, financial condition and results of operations.

Risks Related to Digital Assets

Because there has been limited precedent set for financial accounting for bitcoin and other digital assets, the determinations that we have made for how to account for digital assets transactions may be subject to change.

Because there has been limited precedent set for the financial accounting for bitcoin and other digital assets and related revenue recognition and no official guidance has yet been provided by the Financial Accounting Standards Board or the SEC, it is unclear how companies may in the future be required to account for digital asset transactions and assets and related revenue recognition. A change in regulatory or financial accounting standards could result in the necessity to change the accounting methods we currently intend to employ in respect of our anticipated revenues and assets and restate any financial statements produced based on those methods. Such a restatement could adversely affect our business, prospects, financial condition and results of operation.

Digital assets exchanges and other trading venues are relatively new and, in some cases, partially unregulated and may therefore be more exposed to fraud and failure.

To the extent that digital asset exchanges or other trading venues are involved in fraud or experience security failures or other operational issues, a reduction in digital asset prices could occur. Digital asset market prices depend, directly or indirectly, on the prices set on exchanges and other trading venues, which are new and, in most cases, largely unregulated as compared to established, regulated exchanges for securities, derivatives and other currencies. For example, during the past three years, a number of Bitcoin exchanges have been closed due to fraud, business failure or security breaches. In many of these instances, the customers of the closed Bitcoin exchanges were not compensated or made whole for the partial or complete losses of their account balances in such Bitcoin exchanges. While smaller exchanges are less likely to have the infrastructure and capitalization that provide larger exchanges with additional stability, larger exchanges may be more likely to be appealing targets for hackers and "malware" (i.e., software used or programmed by attackers to disrupt computer operation, gather sensitive information, or gain access to private computer systems) and may be more likely to be targets of regulatory enforcement action.

Digital asset transactions are irrevocable and, if stolen or incorrectly transferred, digital assets may be irretrievable. As a result, any incorrectly executed digital asset transactions could have a material adverse effect on our business, financial condition and results of operations.

Typically, digital asset transactions are not, from an administrative perspective, reversible without the consent and active participation of the recipient of the transaction or, in theory, control or consent of a majority of the processing power on the applicable network. Once a transaction has been confirmed and verified in a block that is added to the network blockchain, an incorrect transfer of a digital asset or a theft of a digital asset generally will not be reversible and we may not be capable of seeking compensation for any such transfer or theft. Although transfers of any digital assets we hold will regularly be made to or from vendors, consultants, services providers, etc., it is possible that, through computer or human error, or through theft or criminal action, our digital assets could be transferred from us in incorrect amounts or to unauthorized third parties. To the extent that we are unable to seek a corrective transaction with such third party or are incapable of identifying the third party that has received our digital assets through error or theft, we will be unable to revert or otherwise recover our incorrectly transferred digital assets. To the extent that we are unable to seek redress for such error or theft, such loss could have a material adverse effect on our business, financial condition and results of operations.

We may not have adequate sources of recovery if the digital assets held by us are lost, stolen or destroyed due to third-party digital asset services, which could have a material adverse effect on our business, financial condition and results of operations.

Certain digital assets held by us are stored using Bittrex, Inc. ("Bittrex") and Coinbase Global, Inc. ("Coinbase"), each a third-party digital asset service. We believe that the security procedures that Bittrex and Coinbase utilize, such as dual authentication security, secured facilities, segregated accounts and cold storage, are reasonably designed to safeguard our bitcoin and other digital assets from theft, loss, destruction or other issues relating to hackers and technological attack. Nevertheless, the security procedures cannot guarantee the prevention of any loss due to a security breach, software defect or act of God that may be borne by us. In addition, Bittrex and Coinbase's limited liability under its services agreement with us may limit our ability to recover losses relating to our bitcoin. If such digital assets are lost, stolen or destroyed under circumstances rendering a third party liable to us, it is possible that the responsible third party may not have the financial resources or insurance sufficient to satisfy any or all of our claims against the third party, or have the ability to retrieve, restore or replace the lost, stolen or destroyed digital assets due to governing network protocols and the strength of the cryptographic systems associated with such digital assets. To the extent that we are unable to recover on any of our claims against any such third party, such loss could have a material adverse effect on our business, financial condition and results of operations.

Losses relating to our business may be uninsured, or insurance may be limited.

Our hosting and colocation operations are subject to hazards and risks normally associated with the daily operations of hosting facilities. Currently, we maintain various insurance policies for business interruption for lost profits, property and casualty, public liability, commercial employee, workers' compensation, personal property and auto liability. Our business interruption insurance for lost profits includes coverage for business interruptions, our property and casualty insurance includes coverage for equipment breakdowns and our commercial employee insurance includes employee group insurance. We believe our insurance coverage adequately covers the risks of our daily business operations. However, our current insurance policies may be insufficient in the event of a prolonged or catastrophic event. The occurrence of any such event that is not entirely covered by our insurance policies may result in interruption of our operations, subject us to significant losses or liabilities and damage our reputation as a provider of business continuity services.

Additionally, it may not be possible, either because of a lack of available policies, limits on coverage or prohibitive cost, for us to obtain insurance of any type that would cover losses associated with our digital asset portfolio. In general, we anticipate that certain losses related to our business may be uninsurable, or the cost of

insuring against these losses may not be economically justifiable. We have obtained some limited coverage regarding our business, but if an uninsured loss occurs or a loss exceeds policy limits, it could have a material adverse effect on our business, financial condition and results of operations.

The digital assets held by us are not insured. Therefore, a loss may be suffered with respect to our digital assets which is not covered by insurance and for which no person is liable in damages which could adversely affect our operations and, consequently, an investment in us.

The impact of geopolitical, economic or other events on the supply of and demand for digital assets is uncertain, but could motivate large-scale sales of digital assets, which could result in a reduction in the price of such digital asset and could have a material adverse effect on our business, financial condition and results of operations.

As an alternative to fiat currencies that are backed by central governments, digital assets, which are relatively new, are subject to supply and demand forces based upon the desirability of an alternative, decentralized means of buying and selling goods and services. It is unclear how this supply and demand will be impacted by geopolitical events. Nevertheless, political or economic crises may motivate large-scale acquisitions or sales of digital assets either globally or locally. Large-scale sales of digital assets likely would result in a reduction in the price of the subject digital asset and could have a material adverse effect on our business, financial condition and results of operations.

In addition, the price of digital assets may be affected by the buying and selling of a significant amount of digital assets by a holder, or a group of holders. For example, Mt. Gox's bankruptcy trustee sold approximately \$400 million of bitcoin and Bitcoin Cash between December 18, 2017 and February 5, 2018 in order to generate proceeds to repay bitcoin owners and creditors that experienced losses as a result of cyber-attacks against Mt. Gox. Similarly, Satoshi Nakamoto, the pseudonymous person or persons who developed Bitcoin, could sell a significant portion of his estimated 1 million bitcoin representing 5% of the total bitcoin in circulation today. Many believe that such sales created a downward pressure on the price of bitcoin. Any such similar events, or other unforeseen actions by holders of a significant amount of digital assets, could have a material adverse effect on our business, financial condition and results of operations.

Digital assets, including bitcoin, face significant scaling obstacles that can lead to high fees or slow transaction settlement times and any mechanisms of increasing the scale of digital asset settlement may significantly alter the competitive dynamics in the market.

Digital assets face significant scaling obstacles that can lead to high fees or slow transaction settlement times, and attempts to increase the volume of transactions may not be effective. Scaling digital assets, and particularly bitcoin, is essential to the widespread acceptance of digital assets as a means of payment, which is necessary to the growth and development of our business.

Many digital asset networks face significant scaling challenges. For example, digital assets are limited with respect to how many transactions can occur per second. In this respect, bitcoin may be particularly affected as it relies on the "proof of work" validation, which due to its inherent characteristics may be particularly hard to scale to allow simultaneous processing of multiple daily transactions by users. Participants in the digital asset ecosystem debate potential approaches to increasing the average number of transactions per second that the network can handle and have implemented mechanisms or are researching ways to increase scale, such as "sharding," which is a term for a horizontal partition of data in a database or search engine, which would not require every single transaction to be included in every single miner's or validator's block. For example, the Ethereum network is in the process of implementing software upgrades and other changes to its protocol, the so-called Ethereum 2.0, which are intended to be a new iteration of the Ethereum network that changes its consensus mechanism from "proof of work" to "proof of stake" and incorporate the use of "sharding." This version aims to address: a clogged network that can only handle limited number of transactions per second and

the large consumption of energy that comes with the "proof of work" mechanism. This new upgrade is envisioned to be more scalable, secure, and sustainable, although it remains unclear whether and how it may ultimately be implemented.

There is no guarantee that any of the mechanisms in place or being explored for increasing the scale of settlement of digital asset transactions will be effective, how long they will take to become effective or whether such mechanisms will be effective for all digital assets. There is also a risk that any mechanisms of increasing the scale of digital asset settlements, such as the ongoing upgrades as part of Ethereum 2.0, may significantly alter the competitive dynamics in the digital asset market and may adversely affect the value of bitcoin and the price of our common stock. Any of which could have a material adverse effect on our business, prospects, financial condition, and operating results.

The IRS and certain states have taken the position that digital assets are "property" for income tax purposes.

In early 2014, the IRS issued basic guidance on the tax treatment of digital assets. The IRS has taken the position that a digital asset is "property" instead of "currency" for income tax purposes. As such, general tax principles applicable to property transactions apply to the acquisition, ownership, use, and disposition of digital assets. This overall treatment creates a potential tax liability for, and potential tax reporting requirements applicable to, us in any circumstance where we mine or otherwise acquire, own, use, or dispose of a digital asset. In 2019, the IRS issued additional guidance specifically relating to the income tax consequences that could arise from a digital asset hard fork event in which a new unit of digital asset may or may not be received, and released frequently asked questions to address certain digital asset topics such as tax basis, gain, or loss on the sale or exchange of certain kinds of digital assets and how to determine the fair market value of such digital assets.

There is no guarantee that the IRS will not alter its position with respect to the taxation of digital assets, or that legislation or judicial determinations in the future will not result in a tax treatment of digital assets and transactions in digital assets for tax purposes that differs from the treatment described above. You are urged to consult your own tax advisor as to the tax implications of our acquisition, ownership, use, and disposition of digital assets. The taxation of digital assets for state, local, or non-U.S. tax purposes may not be the same as the taxation of digital assets for U.S. federal income tax purposes.

In addition, under the Tax Cuts and Jobs Act of 2017, as of January 1, 2018, "like-kind exchange" treatment does not apply to digital assets. This means that gain from the sale or exchange of digital assets cannot be deferred by undertaking an exchange of one type of virtual currency for another.

Certain states, including New York and New Jersey, generally follow IRS guidance with respect to the treatment of digital assets for state income tax purposes, but it is unclear if other states will do so. Transactions involving digital assets for other goods and services may also be subject to sales and use or similar taxes under barter transaction treatment or otherwise. The treatment of digital assets for state income tax and sales tax purposes may have negative consequences, including the imposition of a greater tax burden on investors in digital assets or a higher cost with respect to the acquisition, ownership, use, and disposition of digital assets generally. In either case, this could have a negative effect on prices in the relevant digital asset exchange market and could have a material adverse effect on our business, financial condition, and results of operations.

Non-U.S. jurisdictions may also elect to treat digital assets in a manner that results in adverse tax consequences. To the extent a non-U.S. jurisdiction with a significant share of the market of digital asset owners or users imposes onerous tax burdens on such owners or users, or imposes sales, use, or value added tax on acquisitions and dispositions of digital assets for fiat currency, such actions could result in decreased demand for digital assets in such jurisdiction, which could impact the price of digital assets and have a material adverse effect on our business, financial condition, and results of operations.

Changes to, or changes to interpretations of, tax laws could have a material adverse effect on our business, financial condition, and results of operations.

All statements contained herein concerning U.S. federal income tax (or other tax) consequences are based on existing laws and interpretations thereof. The tax regimes to which we are subject or under which we operate, including income and non-income taxes, are unsettled and may be subject to significant change. While some of these changes could be beneficial, others could negatively affect our after-tax returns. Accordingly, no assurance can be given that the currently anticipated tax treatment will not be modified by legislative, judicial, or administrative changes, possibly with retroactive effect. In addition, no assurance can be given that any tax authority or court will agree with any particular interpretation of the relevant laws.

In 2021, significant changes to U.S. federal income tax laws were proposed, including increasing the U.S. federal income tax rate applicable to corporations from 21% to 28% and changes implicating information reporting with respect to digital assets. Congress may include some of or all these proposals in future legislation. There is uncertainty regarding whether these proposals will be enacted and, if enacted, their scope, when they would take effect, and whether they would have retroactive effect.

State, local, and non-U.S. jurisdictions could impose, levy, or otherwise enforce tax laws against us. Tax laws and regulations at the state and local levels frequently change, especially in relation to the interpretation of existing tax laws for new and emerging industries, and we cannot always reasonably predict the impact from, or the ultimate cost of compliance with, current or future taxes, which could have a material adverse effect on our business, financial condition, and results of operations.

Concerns about greenhouse gas emissions and global climate change may result in environmental taxes, charges, assessments or penalties and could have a material adverse effect on our business, financial condition and results of operations.

The effects of human activity on global climate change have attracted considerable public and scientific attention, as well as the attention of the United States and other foreign governments. Efforts are being made to reduce greenhouse gas emissions, particularly those from coal combustion power plants, some of which plants we may rely upon for power. The added cost of any environmental taxes, charges, assessments or penalties levied on such power plants could be passed on to us, increasing the cost to run our hosting facilities. Any enactment of laws or promulgations of regulations regarding greenhouse gas emissions by the United States, or any domestic or foreign jurisdiction in which we conduct business, could have a material adverse effect on our business, financial condition or results of operations.

Latency in confirming transactions on a network could result in a loss of confidence in the network, which could have a material adverse effect on our business, financial condition and results of operations.

Latency in confirming transactions on a network can be caused by a number of factors, such as transaction processors ceasing to support the network and/or supporting a different network. To the extent that any transaction processors cease to record transactions on a network, such transactions will not be recorded on the blockchain of the network until a block is solved by a transaction processor that does not require the payment of transaction fees or other incentives. Currently, there are no known incentives for transaction processors to elect to exclude the recording of transactions in solved blocks. However, to the extent that any such incentives arise (for example, with respect to Bitcoin, a collective movement among transaction processors or one or more transaction processing pools forcing Bitcoin users to pay transaction fees as a substitute for, or in addition to, the award of new bitcoin upon the solving of a block), transaction processors could delay the recording and verification of a significant number of transactions on a network's blockchain. If such latency became systemic, and sustained, it could result in greater exposure to double-spending transactions and a loss of confidence in the applicable network, which could have a material adverse effect on our business, financial condition and results of operations.

In addition, increasing growth and popularity of digital assets, ICOs and security token offerings, as well as non-digital asset related applications that utilize blockchain technology on certain networks, can cause congestion and backlog, and as result, increase latency on such networks. An increase in congestion and backlogs could result in longer transaction confirmation times, an increase in unconfirmed transactions (that is, transactions that have yet to be included in a block on a network and therefore are not yet completed transactions), higher transaction fees and an overall decrease in confidence in a particular network, which could ultimately affect our ability to transact on that particular network and, in turn, could have a material adverse effect on our business, financial condition and results of operations.

Significant or unexpected changes to our transaction processing operations may have a material adverse effect on our business, financial condition and results of operations.

We and our potential customers are engaged in the business of verifying and confirming transactions on a blockchain, also known as transaction processing, or "mining." We may have to make changes to the specifications of our transaction processing operations for any number of reasons beyond our control (e.g., increased governmental and quasi-governmental regulation of blockchain-related digital assets; changes in methods of validating digital asset transactions; creation of new digital assets; general economic conditions; changes in consumer demographics and public tastes and preferences; and rising energy costs, among other reasons), or we may be unable to develop our transaction processing operations in a manner that realizes those specifications or any form of functioning and profitable transaction processing operations. Furthermore, it is still possible that our transaction processing operations may experience malfunctions, electrical power failure, hacking, cybersecurity breaches or otherwise fail to be adequately developed or maintained. Any of the above risks, which could also impact our potential hosting customers, may have a material adverse effect on our business, financial condition and results of operations.

Currently, we believe there is relatively limited use of digital assets in the retail and commercial marketplace in comparison to relatively sizable use by speculators, thus contributing to price volatility that could adversely affect an investment in our stock.

We believe digital assets have not yet gained widespread acceptance as a means of payment for goods and services by any major retail or commercial outlets. We believe a significant portion of the demand for digital assets is generated by speculators and investors, some of whom may have no knowledge of the inner workings of those assets. Certain of these investors may seek to profit from the short-term or long-term holding of digital assets, and thus, may contribute to digital asset price volatility. A lack of expansion in the use of digital assets in retail and commercial markets, or a contraction of such use, may result in increased price volatility of digital assets or a reduction in the market price of digital assets or in the demand for digital assets which would reduce the demand of our hosting and colocation services or in the value of the digital assets held by us, any of which could have a material adverse effect on our business, financial condition and results of operations.

Diversification of our business by investing in additional digital assets, financial instruments and businesses could require significant investment or expose us to trading risks.

The field of digital assets is constantly expanding with over 4,000 digital assets in existence as of January 2021. We intend to evaluate the potential for mining or investing in existing, new and alternative digital assets. To the extent we elect to commence activities to generate digital assets, we would be required to invest our assets either to obtain mining equipment configured to generate digital assets based on a "proof of work" protocol or to post "stakes" to generate digital assets based on a "proof of stake" protocol. In addition, or in the alternative, we may trade its digital assets for other digital assets on centralized or decentralized exchanges. Optimization of such trades may vary depending on the exchange on which the trade is conducted because we may not have access to all exchanges on which such trades are available. Further, trading on centralized and decentralized exchanges may expose us to additional risks if such exchanges experience breaches of security measures, system errors or vulnerabilities, software corruption, hacking or other irregularities. Any new digital asset obtained

through generation or trading may be more volatile or fail to increase in value compared to digital assets we currently hold. As a result, any investment in different digital assets may not achieve our goals, may be viewed negatively by analysts or investors and may negatively affect our revenues and results of operations.

If the transaction fees for recording digital assets in a blockchain increase, demand for digital assets may be reduced and prevent the expansion of the networks to retail merchants and commercial business, resulting in a reduction in the acceptance or price of digital assets.

As the number of digital assets awarded for solving a block in a blockchain decreases, the incentive for mining participants to contribute processing power to networks will transition from a set reward to transaction fees. In order to incentivize mining participants to continue to contribute processing power to the networks, the network may transition from a set reward to transaction fees earned upon solving for a block. If mining participants demand higher transaction fees to record transactions in a blockchain or a software upgrade automatically charges fees for all transactions, the cost of using digital assets may increase and the marketplace may be reluctant to accept digital assets as a means of payment. Existing users may be motivated to switch from one digital asset to another or back to fiat currency. Decreased use and demand for digital assets may adversely affect their value and result in a reduction in the value of our common stock.

If the award of new digital assets and/or transaction fees for solving blocks is not sufficiently high to incentivize transaction processors, such processors may reduce or cease expending processing power on a particular network, which could negatively impact the utility of the network, reduce the value of its digital assets and have a material adverse effect on our business, financial condition and results of operations.

As the number of digital assets rewarded to transaction processors for validating blocks in a network decreases, the incentive for transaction processors to continue contributing processing power to the network may shift toward transaction fees. Such a shift may increase the transaction fees on a network. Higher transaction fees may reduce the utility of a network for an end user, which may cause end users to reduce or stop their use of that network. In such case, the price of the relevant digital asset may decline substantially and could go to zero. Such reduced price and demand for, and use of, the relevant digital asset and network, either as it applies to our transaction processing services or to those of our potential hosting customers, may have a material adverse effect on our business, financial condition and results of operations.

As more processing power is added to a network, our relative percentage of total processing power on that network is expected to decline absent significant capital investment, which has an adverse impact on our ability to generate revenue from processing transactions on that network and could have a material adverse effect on our business, financial condition and results of operations.

Processing power on networks has been increasing rapidly over time while the rewards and transaction fees available on those networks tends to decline over time. In order to grow or maintain the revenue we generate from processing transactions on such networks, we are required to invest significant capital to acquire new computer servers, expand our power capacity and otherwise increase our effective processing power on such networks. In the event we are unable to invest sufficient capital to grow or maintain the level of our processing power on a network relative to the total processing power of such network, our revenue from the applicable network will decline over time and as a result, it could have a material adverse effect on our business, financial condition and results of operations.

In addition, a decrease in the price of computer servers may result in an increase in transaction processors, which may lead to more competition for fees in a particular network. We are unable to predict the pace at which processing power is added to the network. In the event we are unable to realize adequate fees on a network due to increased competition, our revenue from the applicable network will decline over time and in turn, it could have a material adverse effect on our business, financial condition and results of operations.

We may only have limited control over our mining operation.

Our mining operation comprises blockchain mining technologies that depend on a network of computers to run certain software programs to solve complex transactions in competition with other mining operations and to process transactions. Because of this less centralized model and the complexity of our mining operation, we have limited control over the success of our mining operations. While we participate in mining pools to combine our mining operations with other mining participants to increase processing power to solve blocks, there can be no assurance that such pools will adequately address this risk

Our reliance on third-party mining pool service providers for our mining revenue payouts may have a negative impact on our operations.

We utilize third party mining pools to receive our mining rewards from a given network. Mining pools allow mining participants to combine their processing power, which increases the chances of solving a block and getting paid by the network. The rewards are distributed by the pool operator, proportionally to our contribution to the pool's overall mining power used to generate each block. We are dependent on the accuracy of the mining pool operator's record keeping to accurately record the total processing power provided to the pool for a given bitcoin or other digital asset mining application in order to assess the proportion of that total processing power we provided. While we have internal methods of tracking both our power provided and the total power used by the pool, the mining pool operator uses its own record-keeping to determine our proportion of a given reward. We have little means of recourse against the mining pool operator if we determine the proportion of the reward paid out to us by a mining pool operator is incorrect, other than leaving the pool. If we are unable to consistently obtain accurate proportionate rewards from our mining pool operators, we may experience reduced reward for our efforts, which would have an adverse effect on our business and operations.

Malicious actors or botnet may obtain control of more than 50% of the processing power on the Bitcoin or other network.

If a malicious actor or botnet (a volunteer or hacked collection of computers controlled by networked software coordinating the actions of the computers) obtains a majority of the processing power dedicated to mining on the Bitcoin or other network, it may be able to alter the blockchain on which the Bitcoin or other network and most Bitcoin or other digital asset transactions rely by constructing fraudulent blocks or preventing certain transactions from completing in a timely manner, or at all. The malicious actor or botnet could control, exclude, or modify the ordering of transactions, though it could not generate new bitcoin or digital assets or transactions using such control. The malicious actor could "double-spend" its own bitcoin or digital assets (i.e., spend the same bitcoin or digital assets in more than one transaction) and prevent the confirmation of other users' transactions for so long as it maintained control. To the extent that such malicious actor or botnet did not yield its control of the processing power on the Bitcoin or other network, or the Bitcoin or other community did not reject the fraudulent blocks as malicious, reversing any changes made to the blockchain may not be possible.

Although there are no known reports of malicious activity or control of the Bitcoin blockchain achieved through controlling over 50% of the processing power on the network, it is believed that certain mining pools may have exceeded the 50% threshold. The possible crossing of the 50% threshold indicates a greater risk in that a single mining pool could exert authority over the validation of Bitcoin transactions. To the extent that the Bitcoin or other digital asset ecosystems, including developers and administrators of mining pools, do not act to ensure greater decentralization of Bitcoin or other digital asset mining processing power, the feasibility of a malicious actor obtaining control of the processing power on the Bitcoin or other network will increase, which may adversely affect an investment us.

Transaction processing operators may sell a substantial amount of digital assets into the market, which may exert downward pressure on the price of the applicable digital asset and, in turn, could have a material adverse effect on our business, financial condition and results of operations.

Transaction processing requires the investment of significant capital for the acquisition of hardware, leasing or purchasing space, involves substantial electricity costs and requires the employment of personnel to operate the data facilities, which may lead transaction processing operators to liquidate their positions in digital assets to fund these capital requirements. In addition, if the reward of new digital assets for transaction processing declines, and/or if transaction fees are not sufficiently high, profit margins for transaction processing operators may be reduced, and such operators may be more likely to sell a higher percentage of their digital assets. Whereas it is believed that individual operators in past years were more likely to hold digital assets for more extended periods, the immediate selling of newly transacted digital assets by operators may increase the supply of such digital assets on the applicable exchange market, which could create downward pressure on the price of the digital assets and, in turn, could have a material adverse effect on our business, financial condition and results of operations.

To the extent that the profit margins of digital asset mining operations are not high, mining participants are more likely to sell their earned bitcoin, which could constrain bitcoin prices.

Over the past few years, digital asset mining operations have evolved from individual users mining with computer processors, graphics processing units and first-generation application-specific integrated circuit ("ASIC") servers. Currently, new processing power is predominantly added by incorporated and unincorporated "professionalized" mining operations. Professionalized mining operations may use proprietary hardware or sophisticated ASIC machines acquired from ASIC manufacturers. They require the investment of significant capital to acquire this hardware, to lease operating space (often in data centers or warehousing facilities), and to pay the costs of electricity and labor to operate the mining farms. As a result, professionalized mining operations are of a greater scale than prior mining operations and have more defined and regular expenses and liabilities. These regular expenses and liabilities require professionalized mining operations to maintain profit margins on the sale of digital assets. To the extent the price of digital assets decline and such profit margin is constrained, professionalized mining participants are incentivized to more immediately sell digital assets earned from mining operations, whereas it is believed that individual mining participants in past years were more likely to hold newly mined digital assets for more extended periods. The immediate selling of newly mined digital assets greatly increases the trading volume of the digital assets, creating downward pressure on the market price of digital asset rewards. The extent to which the value of digital assets mined by a professionalized mining operation exceeds the allocable capital and operating costs determines the profit margin of such operation. A professionalized mining operation may be more likely to sell a higher percentage of its newly mined digital assets rapidly if it is operating at a low profit margin and it may partially or completely cease operations if its profit margin is negative. In a low profit margin environment, a higher percentage could be sold more rapidly, thereby potentially depressing digital asset prices. Lower digital asset prices could result in further tightening of profit margins for professionalized mining operations creating a network effect that may further reduce the price of digital assets until mining operations with higher operating costs become unprofitable forcing them to reduce mining power or cease mining operations temporarily. Such circumstances could have a material adverse effect on our business, prospects or operations and potentially the value of bitcoin and any other digital assets we mine or otherwise acquire or hold for our own account

The "halvening" of rewards available on the Bitcoin network, or the reduction of rewards on other networks, has had and in the future could have a negative impact on our ability to generate revenue as our customers may not have an adequate incentive to continue transaction processing and customers may cease transaction processing operations altogether, which could have a material adverse effect on our business, financial condition and results of operations.

Under the current protocols governing the Bitcoin network, the reward for validating a new block on that network is cut in half from time to time, which has been referred to in our industry as the "halvening." When the

Bitcoin network was first launched, the reward for validating a new block was 50 bitcoin. In 2012, the reward for validating a new block was reduced to 25 bitcoin. In July 2016, the reward for validating a new block was reduced to 12.5 bitcoin, and in May 2020, the reward was further reduced to 6.25 bitcoin. In addition, other networks may operate under rules that, or may alter their rules to, limit the distribution of new digital assets. We, and to our knowledge, our potential hosting customers, currently rely on these rewards to generate a significant portion of our total revenue. If the award of digital assets for solving blocks and transaction fees are not sufficiently high, neither we nor our customers may have an adequate incentive to continue transaction processing and may cease transaction processing operations altogether, which as a result may significantly reduce demand for our hosting services. As a result, the halvening of available rewards on the Bitcoin network, or any reduction of rewards on other networks, would have a negative impact on our revenues and may have a material adverse effect on our business, financial condition and results of operations.

In addition, the reduction of rewards may reduce our profit margins, which could result in us selling a substantial portion of our digital assets, which are subject to high volatility. If we are forced to sell digital assets at low prices, it could have a material adverse effect on our business, financial condition and results of operations.

We intend to hold varying amounts of digital assets over time, and their value could be subject to inherent volatility. From time to time we have sold, and we may sell in the future, a portion of our digital assets to pay for costs and expenses, which has reduced, and may reduce in the future, the amount of digital assets we hold, thus preventing us from recognizing any gain from the appreciation in value of the digital assets we have sold and may sell in the future.

From time to time, we have sold, and we may sell in the future, a portion of our digital assets to pay for costs and expenses incurred, capital expenditures and other working capital, irrespective of then-current digital asset prices. Our past sales of digital assets have reduced the amount of digital assets we currently hold. We intend to hold varying amounts of digital assets over time, and their value could be subject to inherent volatility. When we sell a digital asset, we are unable to benefit from any future appreciation in the underlying value of that digital asset.

Consequently, our digital assets may be sold at a time when the price is lower than it otherwise might be in the future, which could reduce the gain we might have realized on the sale of that digital asset at a different time. If we sell any digital assets in the future, the loss of potential realized gains from the sale of such digital assets could have a material adverse effect on our business, financial condition and results of operations.

Digital assets are subject to extreme price volatility. The value of digital assets is dependent on a number of factors, any of which could have a material adverse effect on our business, financial condition and results of operations.

We currently generate a small portion of our revenue from processing blockchain transactions, in return for which receive digital assets. In connection with our recent merger with Blockcap, we anticipate that a much larger portion of our revenue will come from processing blockchain transactions in the form of digital assets like Bitcoin. We believe the value of digital assets related to our business is dependent on a number of factors, including, but not limited to:

- global digital asset supply;
- global digital asset demand, which can be influenced by the growth of retail merchants' and commercial businesses' acceptance of digital
 assets as payment for goods and services, the security of online digital asset exchanges and digital wallets that hold digital assets, the
 perception that the use and holding of digital assets is safe and secure, and the regulatory restrictions on their use;
- investors' expectations with respect to the rate of inflation of fiat currencies;

- investors' expectations with respect to the rate of deflation of digital assets;
- cyber theft of digital assets from online wallet providers, or news of such theft from such providers or from individuals' online wallets;
- the availability and popularity of businesses that provide digital asset-related services;
- fees associated with processing a digital asset transaction;
- changes in the software, software requirements or hardware requirements underlying digital assets;
- changes in the rights, obligations, incentives, or rewards for the various participants in digital asset mining;
- interest rates;
- currency exchange rates, including the rates at which digital assets may be exchanged for fiat currencies;
- fiat currency withdrawal and deposit policies on digital asset exchanges and liquidity on such exchanges;
- interruptions in service or failures of major digital asset exchanges;
- investment and trading activities of large investors, including private and registered funds, that may directly or indirectly invest in digital assets;
- momentum pricing;
- monetary policies of governments, trade restrictions, currency devaluations and revaluations;
- regulatory measures, if any, that affect the use of digital assets, restrict digital assets as a form of payment, or limit the purchase of digital assets;
- global or regional political, economic or financial events and conditions;
- expectations that the value of digital assets will change in the near or long term. A decrease in the price of a single digital asset may cause
 volatility in the entire digital asset industry and may affect other digital assets. For example, a security breach that affects investor or user
 confidence in bitcoin, ethereum, litecoin or another digital asset may affect the industry as a whole and may also cause the price of other
 digital assets to fluctuate; or
- · with respect to bitcoin, increased competition from other forms of digital assets or payments services.

Bitcoin and other digital assets have historically experienced significant intraday and long-term price volatility, significantly impacted by momentum pricing. Momentum pricing typically is associated with growth stocks and other assets whose valuation, as determined by the investing public, accounts for anticipated future appreciation in value. The market price for digital assets is determined using data from various digital asset exchanges, over-the-counter markets, digital asset futures markets, derivative platforms and other digital asset investment vehicles. We believe that momentum pricing may have resulted, and may continue to result, in significant and rampant speculation regarding future appreciation (or depreciation) in the value of digital assets, inflating and making their market prices more volatile, even more so than with traditional asset classes, such as equities. In addition, there is currently growing but limited acceptance of digital assets in the retail and commercial marketplace, as compared to the demand generated by investors seeking a long-term value retention or by speculators seeking to profit from the short- or long-term holding of such digital assets, which may contribute to their extreme levels of price volatility.

Even if shareholders are able to hold their common stock for the long-term, their common stock may never generate a profit, since digital asset markets have historically experienced extended periods of flat or declining prices, in addition to sharp fluctuations. Investors should be aware that there is no assurance that bitcoin or other

digital assets will maintain their long-term value in terms of future purchasing power or that the acceptance of digital asset payments by mainstream retail merchants and commercial businesses will continue to grow. If the price of bitcoin or other digital assets declines, we expect our profitability to decline

Any loss or destruction of a private key required to access a digital asset of ours is irreversible. We also may temporarily lose access to our digital assets.

Digital assets are each accessible and controllable only by the possessor of both the unique public key and private key associated with the digital asset, wherein the public and private keys are held in an offline or online digital wallet. To the extent a private key is lost, destroyed or otherwise compromised and no backup of the private key is available, we will be unable to access the applicable digital asset associated with that private key and the private key cannot be restored. As a result, any digital assets associated with such key could be irretrievably lost. Any loss of private keys relating to digital wallets used to store the applicable digital assets could have a material adverse effect on our business, financial condition and results of operations.

In addition, we may temporarily lose access to our digital assets as a result of software or systems upgrades or maintenance. In this case, we would likely rely on third parties to assist in restoring our access, and we cannot provide any assurance that such third parties will be able to restore access on a timely basis, or at all. Any temporary loss, if it occurs, could have a material adverse effect on our business, financial condition and results of operations.

Intellectual property rights claims may adversely affect the operation of any or all of the networks.

Third parties may assert intellectual property rights claims relating to the operation of digital assets and the holding and transfer of such assets. Regardless of the merit of any intellectual property rights claims or other legal action, any threatened action that reduces confidence in the long-term viability of any or all of the networks or other similar peer-to-peer networks, or in the ability of end-users to hold and transfer digital assets, may have a material adverse effect on our business, results of operations and financial condition. Additionally, a meritorious intellectual property rights claim could prevent us and other end-users from holding or transferring the digital assets, which could have a material adverse effect on our business, financial condition and results of operations.

A soft or hard fork on a network could have a material adverse effect on our business, financial condition and results of operations.

The rules governing a network's protocol are subject to constant change and, at any given time, there may be different groups of developers that can modify a network's protocol. As network protocols are not sold and their use does not generate revenues for their development teams, the core developers are generally not compensated for maintaining and updating the network protocols. Consequently, there is a lack of financial incentive for developers to maintain or develop networks and the core developers may lack the resources to adequately address emerging issues with network protocols. Although the Bitcoin and other leading networks are currently supported by core developers, there can be no guarantee that such support will continue or be sufficient in the future. To the extent that material issues arise with the Bitcoin or another network protocol and the core developers and open-source contributors are unable to address the issues adequately or in a timely manner, the networks may be adversely affected.

Any individual can download the applicable network software and make any desired modifications that alter the protocols and software of the network, which are proposed to developers, users and transaction processors on the applicable network through software downloads and upgrades, typically posted to development forums such as GitHub.com. Such proposed modifications can be agreed upon, developed, adopted and implemented by a substantial majority of developers, transaction processors and users, which, in such event, results in a "soft fork" or "hard fork" on the relevant network. A "soft fork" occurs when an updated version of the validating protocol

is still "backwards compatible" with previous versions of the protocol. As a result, non-upgraded network participants with an older version of the validating protocol will still recognize new blocks or transactions and may be able to confirm and validate a transaction; however, the functionality of the non-upgraded network participant may be limited. Thus, non-upgraded network participants are incentivized to adopt the updated version of the protocol. The occurrence of a soft fork could potentially destabilize transaction processing and increase transaction and development costs and decrease trustworthiness of a network.

A "hard fork" occurs when the updated version of the validating protocol is not "backwards compatible" with previous versions of the protocol, and therefore, requires forward adoption by network participants in order to recognize new blocks, validate and verify transactions and maintain consensus on the relevant blockchain. Since the updated version of the protocol is not backwards compatible, a hard fork can cause the relevant blockchain to permanently diverge into two separate blockchains on a network. For example, in the case of Bitcoin, a hard fork created two new digital assets: Bitcoin Cash and Bitcoin Gold. The value of a newly created digital asset from a hard fork ("forked digital asset") may or may not have value in the long-run and may affect the price of other digital asset if interest and resources are shifted away from previously existing digital assets to the forked digital asset. The value of a previously existing digital asset after a hard fork is subject to many factors, including the market reaction and value of the forked digital asset and the occurrence of other soft or hard forks in the future. As such, the value of certain digital assets could be materially reduced if existing and future hard forks have a negative effect on their value.

If a soft fork or hard fork occurs on a network, which we or our hosting customers are processing transactions or hold digital assets in, we may be required to upgrade our hardware or software in order to continue our transaction processing operations, and there can be no assurance that we may be able to make such upgrades. A soft fork or hard fork in a particular digital asset that we process could have a negative effect on the value of that digital asset and could have a material adverse effect on our business, financial condition and results of operations.

The digital assets held by us may be subject to loss, damage, theft or restriction on access, which could have a material adverse effect on our business, financial condition or results of operations.

There is a risk that some or all of the digital assets held or hosted by us could be lost, stolen or destroyed. We believe that the digital assets held or hosted by us and our mining operation will be an appealing target to hackers or malware distributors seeking to destroy, damage or steal our digital assets. Our security procedures and operational infrastructure may be breached due to the actions of outside parties, error or malfeasance of one of our employees, or otherwise, and, as a result, an unauthorized party may obtain access to our digital asset accounts, private keys, data or digital assets. Although we implement a number of security procedures with various elements such as two-factor verification, segregated accounts and secured facilities and plan to implement the maintenance of data on computers and/or storage media that is not directly connected to, or accessible from, the internet and/or networked with other computers ("cold storage") to minimize the risk of loss, damage and theft, and we update such security procedures whenever reasonably practicable, we cannot guarantee the prevention of such loss, damage or theft, whether caused intentionally, accidentally or by an act of God.

Additionally, outside parties may attempt to fraudulently induce our employees to disclose sensitive information in order to gain access to our infrastructure. As the techniques used to obtain unauthorized access, disable or degrade service, or sabotage systems change frequently, or may be designed to remain dormant until a predetermined event, and often are not recognized until launched against a target, we may be unable to anticipate these techniques or implement adequate preventative measures. As technological change occurs, the security threats to our bitcoin will likely adapt and previously unknown threats may emerge. Our ability to adopt technology in response to changing security needs or trends may pose a challenge to the safekeeping of our digital assets. To the extent we are unable to identify and mitigate or stop new security threats, our digital assets may be subject to theft, loss, destruction or other attack.

Any of these events could expose us to liability, damage our reputation, reduce customer confidence in our services and otherwise have a material adverse effect on our business, financial condition and results of operations. Furthermore, we believe that as our assets grow, we may become a more appealing target for security threats, such as hackers and malware. If an actual or perceived breach of our digital asset accounts occurs, the market perception of our effectiveness could be harmed.

The digital assets held by us are not subject to FDIC or SIPC protections.

We do not hold our digital assets with a banking institution or a member of the Federal Deposit Insurance Corporation ("FDIC") or the Securities Investor Protection Corporation ("SIPC"), and to date, neither the FDIC nor the SIPC has extended any such protections to depositors of digital assets. Accordingly, our digital assets are not subject to the protections by FDIC or SIPC member institutions and any loss of our digital assets could have a material adverse effect on our business, financial condition and results of operations.

We may not be able to maintain our competitive position as digital asset networks experience increases in total network hash rate.

As the relative market prices of a digital asset, such as Bitcoin, increases, more companies are encouraged to mine for that digital asset and as more miners are added to the network, its total hash rate increases. In order for us to maintain its competitive position under such circumstances, we must increase our total hash rate by acquiring and deploying more mining machines, including new miners with higher hash rates. There are currently only a few companies capable of producing a sufficient number of machines with adequate quality to address the increased demand. If we are not able to acquire and deploy additional miners on a timely basis, our proportion of the overall network hash rate will decrease and we will have a lower chance of solving new blocks which will have an adverse effect on our business and results of operations.

To the extent that any miners cease to record transactions in solved blocks, transactions that do not include the payment of a transaction fee will not be recorded on the blockchain until a block is solved by a miner who does not require the payment of transaction fees. Any widespread delays in the recording of transactions could result in a loss of confidence in that digital asset network, which could adversely impact an investment in us.

To the extent that any miners cease to record transactions in solved blocks, such transactions will not be recorded on the blockchain. Currently, there are no known incentives for miners to elect to exclude the recording of transactions in solved blocks; however, to the extent that any such incentives arise (e.g., a collective movement among miners or one or more mining pools forcing Bitcoin users to pay transaction fees as a substitute for or in addition to the award of new Bitcoins upon the solving of a block), actions of miners solving a significant number of blocks could delay the recording and confirmation of transactions on the blockchain.

Any systemic delays in the recording and confirmation of transactions on the blockchain could result in greater exposure to double-spending transactions and a loss of confidence in certain or all digital asset networks, which could have a material adverse effect on our business, prospects, financial condition, and operating results.

Our interactions with a blockchain may expose us to SDN or blocked persons or cause us to violate provisions of law that did not contemplate distribute ledger technology.

The Office of Financial Assets Control of the U.S. Department of Treasury ("OFAC") requires us to comply with its sanction program and not conduct business with persons named on its specially designated nationals ("SDN") list. However, because of the pseudonymous nature of blockchain transactions, we may inadvertently and without our knowledge engage in transactions with persons named on OFAC's SDN list. Our internal policies prohibit any transactions with such SDN individuals, but we may not be adequately capable of determining the ultimate identity of the individual with whom we transact with respect to selling digital assets. In addition, in the future, OFAC or another regulator, may require us to screen transactions for OFAC addresses or

other bad actors before including such transactions in a block, which may increase our compliance costs, decrease our anticipated transaction fees and lead to decreased traffic on our network. Any of these factors, consequently, could have a material adverse effect on our business, prospects, financial condition, and operating results.

Moreover, federal law prohibits any U.S. person from knowingly or unknowingly possessing any visual depiction commonly known as child pornography. Recent media reports have suggested that persons have imbedded such depictions on one or more blockchains. Because our business requires us to download and retain one or more blockchains to effectuate our ongoing business, it is possible that such digital ledgers contain prohibited depictions without our knowledge or consent. To the extent government enforcement authorities literally enforce these and other laws and regulations that are impacted by decentralized distributed ledger technology, we may be subject to investigation, administrative or court proceedings, and civil or criminal monetary fines and penalties, all of which could harm our reputation and could have a material adverse effect on our business, prospects, financial condition, and operating results.

We have identified material weaknesses in our internal control over financial reporting and may identify additional material weaknesses in the future or otherwise fail to maintain an effective system of internal control, which may result in material misstatements of our financial statements or cause it to fail to meet its periodic reporting obligations.

As a public company, we are required to maintain internal control over financial reporting and to report any material weaknesses in such internal control over financial reporting. We are currently a private company with limited accounting and financial reporting personnel and other resources with which to address its internal control and procedures. In connection with the audit of our consolidated financial statements for the year ended December 31, 2020, we and our independent registered public accounting firm identified material weaknesses in our internal control over financial reporting related to insufficient accounting and supervision with respect to appropriate processes and procedures to assess and apply the relevant accounting framework and a lack of appropriate communication and recordkeeping, particularly related to equity transactions. A material weakness is a deficiency, or a combination of deficiencies, in internal control over financial reporting such that there is a reasonable possibility that a material misstatement of our annual or interim financial statements will not be prevented or detected on a timely basis. With the oversight of our senior management and audit committee, we have instituted plans to remediate the material weakness and will continue to take remediation steps, including hiring additional key supporting accounting personnel with public company reporting and accounting operations experience. In addition, we are formalizing inter-departmental communication, including establishing appropriate standing and ad hoc committees and enhancing electronic document storage for key financial transactions. We believe the measures described above will remediate the material weaknesses identified and strengthen our internal control.

While we implement our plan to remediate the material weaknesses described above, we cannot predict the success of such plan or the outcome of its assessment of these plans at this time. If our steps are insufficient to remediate the material weaknesses successfully and otherwise establish and maintain an effective system of internal control over financial reporting, the reliability of our financial reporting, investor confidence, and the value of its common stock could be materially and adversely affected. We can give no assurance that the implementation of this plan will remediate these deficiencies in our internal control over financial reporting or that additional material weaknesses or significant deficiencies in our internal control over financial reporting will not be identified in the future. The failure to implement and maintain effective internal control over financial reporting could result in errors in our financial statements that could result in a restatement of its financial statements, causing us to fail to meet its reporting obligations.

If we fail to maintain an effective system of disclosure controls and internal control over financial reporting, our ability to produce timely and accurate financial statements or comply with applicable regulations could be impaired.

As a public company, we are subject to the reporting requirements of the Exchange Act, the Sarbanes-Oxley Act and the rules and regulations of the applicable listing standards of the Nasdaq. We expect that the requirements of these rules and regulations will continue to increase our legal, accounting and financial compliance costs, make some activities more difficult, time-consuming and costly and place significant strain on our personnel, systems and resources.

The Sarbanes-Oxley Act requires, among other things, that we maintain effective disclosure controls and procedures and internal control over financial reporting. We are continuing to develop and refine our disclosure controls and other procedures that are designed to ensure that information required to be disclosed by us in the reports that we file with the SEC is recorded, processed, summarized and reported within the time periods specified in SEC rules and forms and that information required to be disclosed in reports under the Exchange Act is accumulated and communicated to our principal executive and financial officers. We are also continuing to improve our internal control over financial reporting. In order to maintain and improve the effectiveness of our disclosure controls and procedures and internal control over financial reporting, we have expended, and anticipate that we will continue to expend, significant resources, including accounting-related costs and significant management oversight.

Our current controls and any new controls that we develop may become inadequate because of changes in conditions in our business. In addition, changes in accounting principles or interpretations could also challenge our internal controls and require that we establish new business processes, systems and controls to accommodate such changes. We have limited experience with implementing the systems and controls that are necessary to operate as a public company, as well as adopting changes in accounting principles or interpretations mandated by the relevant regulatory bodies. Additionally, if these new systems, controls or standards and the associated process changes do not give rise to the benefits that we expect or do not operate as intended, it could adversely affect our financial reporting systems and processes, our ability to produce timely and accurate financial reports or the effectiveness of internal control over financial reporting. Moreover, our business may be harmed if we experience problems with any new systems and controls that result in delays in their implementation or increased costs to correct any post-implementation issues that may arise.

Further, in addition to the material weaknesses we currently have, weaknesses in our disclosure controls and internal control over financial reporting may be discovered in the future. Any failure to develop or maintain effective controls or any difficulties encountered in their implementation or improvement could harm our business or cause us to fail to meet our reporting obligations and may result in a restatement of our financial statements for prior periods. Any failure to implement and maintain effective internal control over financial reporting also could adversely affect the results of periodic management evaluations and annual independent registered public accounting firm attestation reports regarding the effectiveness of our internal control over financial reporting that we will eventually be required to include in our periodic reports that will be filed with the SEC. Ineffective disclosure controls and procedures and internal control over financial reporting could also cause investors to lose confidence in our reported financial and other information, which would likely have a negative effect on the trading price of our common stock. In addition, if we are unable to continue to meet these requirements, we may not be able to remain listed on the Nasdaq. We are not currently required to comply with the SEC rules that implement Section 404 of the Sarbanes-Oxley Act and are therefore not required to make a formal assessment of the effectiveness of our internal control over financial reporting for that purpose. As a public company, we will be required to provide an annual management report on the effectiveness of our internal control over financial reporting commencing with our second annual report on Form 10-K.

Our independent registered public accounting firm is not required to formally attest to the effectiveness of our internal control over financial reporting until our first annual report filed with the SEC where we are an

accelerated filer or a large accelerated filer, and do not qualify as an emerging growth company or smaller reporting company with revenues of less than \$100 million. At such time, our independent registered public accounting firm may issue a report that is adverse in the event it is not satisfied with the level at which our internal control over financial reporting is documented, designed or operating. Any failure to maintain effective disclosure controls and internal control over financial reporting could harm our business and could cause a decline in the trading price of our common stock. For more information as it relates to the risk controls, please see "—We have identified material weaknesses in our internal control over financial reporting and may identify additional material weaknesses in the future or otherwise fail to maintain an effective system of internal control, which may result in material misstatements of our financial statements or cause it to fail to meet its periodic reporting obligations."

Risks Related to Ownership of Our Securities and Other General Matters

An active trading market for our common stock may never develop or be sustained.

Our common stock is listed on the Nasdaq under the symbol "CORZ." However, we cannot assure you that an active trading market for our common stock will develop on that exchange or elsewhere or, if developed, that any market will be sustained. Accordingly, we cannot assure you of the likelihood that an active trading market for our common stock will develop or be maintained, your ability to sell your shares of our common stock when desired or the prices that you may obtain for your shares.

The trading price of our common stock may be volatile, and you could lose all or part of your investment.

The trading price of our common stock is likely to be volatile and could be subject to fluctuations in response to various factors, some of which are beyond our control. These fluctuations could cause you to lose all or part of your investment in our common stock as you might be unable to sell your shares at or above the price you paid for those shares. Factors that could cause fluctuations in the trading price of our common stock include the following:

- price and volume fluctuations in the overall stock market from time to time;
- volatility in the trading prices and trading volumes of technology stocks;
- · volatility in the price of bitcoin and other digital assets;
- changes in operating performance and stock market valuations of other technology companies generally, or those in our industry in particular;
- sales of shares of our common stock by us or our stockholders, including sales under this prospectus;
- failure of securities analysts to maintain coverage of us, changes in financial estimates by securities analysts who follow our company, or our failure to meet these estimates or the expectations of investors;
- the financial projections we may provide to the public, any changes in those projections, or our failure to meet those projections;
- announcements by us or our competitors of new products, features, or services;
- the public's reaction to our press releases, other public announcements and filings with the SEC;
- rumors and market speculation involving us or other companies in our industry;
- actual or anticipated changes in our results of operations or fluctuations in our results of operations;
- actual or anticipated developments in our business, our competitors' businesses or the competitive landscape generally;
- · litigation involving us, our industry, or both, or investigations by regulators into our operations or those of our competitors;

- developments or disputes concerning our intellectual property or other proprietary rights;
- announced or completed acquisitions of businesses, products, services or technologies by us or our competitors;
- new laws or regulations or new interpretations of existing laws or regulations applicable to our business;
- changes in accounting standards, policies, guidelines, interpretations or principles;
- any significant change in our management; and
- · general economic conditions and slow or negative growth of our markets.

In addition, in the past, following periods of volatility in the overall market and in the market price of a particular company's securities, securities class action litigation has often been instituted against these companies. This litigation, if instituted against us, could result in substantial costs and a diversion of our management's attention and resources.

Future sales and issuances of our capital stock or rights to purchase capital stock could result in additional dilution of the percentage ownership of our stockholders and could cause our stock price to decline.

We may issue additional securities, including shares of common stock underlying warrants or as a result of the conversion of convertible notes or the exercise of options or RSUs. Future sales and issuances of our capital stock or rights to purchase our capital stock could result in substantial dilution to our existing stockholders. We may sell common stock, convertible securities and other equity securities in one or more transactions at prices and in a manner as we may determine from time to time. If we sell any such securities in subsequent transactions, investors may be materially diluted. New investors in such subsequent transactions could gain rights, preferences and privileges senior to those of holders of our common stock.

Future sales, or the perception of future sales, could cause the market price of our common stock to drop significantly, even if our business is doing well.

The sale of our securities in the public market, or the perception that such sales could occur, could harm the prevailing market price of our securities. These sales, or the possibility that these sales may occur, also might make it more difficult for us to sell equity securities in the future at a time and at a price that we deem appropriate.

Pursuant to our second amended and restated bylaws (the "Bylaws") and certain lock-up agreements entered into prior to the consummation of the Business Combination by and among Core and the stockholders and employees signatories thereto, certain stockholders of Core, including Legacy Core's stockholders and the Sponsor, which held approximately 90.7% and 2.6%, respectively, of our outstanding common stock as of the closing of the Business Combination, have agreed that, with respect to our common stock (including securities convertible into our common stock) held by Legacy Core's stockholders, through the date that is 180 days after the closing of the Business Combination, with respect to the Private Placement Warrants and any of our common stock issuable upon the exercise of the Private Placement Warrants, through the date that is 30 days after the closing of the Business Combination, and, with respect to the Founder Shares, through the date that is one year after the closing of the Business Combination, subject to certain exceptions, to not, without the prior written consent of the our board of directors, among other things, sell, offer to sell, contract or agree to sell, hypothecate, pledge, grant any option to purchase or otherwise dispose of or agree to dispose of, directly or indirectly any shares of our common stock, the Private Placement Warrants, our common stock issuable upon the exercise of the Private Placement Warrants, as applicable, held by the respective parties. Our board of directors can terminate these restrictions at any time and may do so to increase the number of shares eligible for resale on the public market.

However, following the expiration of such lock-up periods, these lock-up parties will not be restricted from selling our securities held by them, other than by applicable securities laws.

In connection with the Business Combination, XPDI's existing registration rights agreement was amended and restated to: (i) provide that we will file the registration statement of which this prospectus forms a part to register for resale of the securities held by the parties thereto under the Securities Act and (ii) afford each such party "piggyback" registration rights with respect to any underwritten offerings by the other stockholders and by us.

Sales of a substantial number of shares of our common stock in the public market could occur at any time, particularly after expiration of the above-mentioned lock-up periods and the registration of the resale of our securities discussed above. These sales, or the perception in the market that members of our management or holders of a large number of shares intend to sell shares, could reduce the market price of our common stock and the Warrants

Because we have not conducted an underwritten offering of our securities, no underwriter has conducted due diligence of our business, operations or financial condition or reviewed the disclosure in this prospectus.

Section 11 of the Securities Act ("Section 11") imposes liability on parties, including underwriters, involved in a securities offering if the registration statement contains a materially false statement or material omission. To effectively establish a due diligence defense against a cause of action brought pursuant to Section 11, a defendant, including an underwriter, carries the burden of proof to demonstrate that he or she, after reasonable investigation, believed that the statements in the registration statement were true and free of material omissions. In order to meet this burden of proof, underwriters in a registered offering typically conduct extensive due diligence of the registrant and vet the registrant's disclosure. Such due diligence may include calls with the issuer's management, review of material agreements, and background checks on key personnel, among other investigations.

Because we became publicly traded through a business combination with XPDI, a SPAC (as defined below), rather through an underwritten offering of its ordinary shares, no underwriter has conducted diligence on Legacy Core or XPDI in order to establish a due diligence defense with respect to the disclosure presented in this prospectus. If such investigation had occurred, certain information in this prospectus may have been presented in a different manner or additional information may have been presented at the request of such underwriter.

We have identified a material weakness in our internal control over financial reporting. This material weakness could continue to adversely affect our ability to report our results of operations and financial condition accurately and in a timely manner.

Our management is responsible for establishing and maintaining adequate internal control over financial reporting designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP. Our management is likewise required, on a quarterly basis, to evaluate the effectiveness of our internal controls and to disclose any changes and material weaknesses identified through such evaluation in those internal controls. A material weakness is a deficiency, or a combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of our annual or interim financial statements will not be prevented or detected on a timely basis.

As described elsewhere in this prospectus, we identified a material weakness in our internal control over financial reporting relating to our classification of a portion of the Class A Common Stock in permanent equity rather than temporary equity, as further described herein. Specifically, the Company's management has concluded that our control around the interpretation and accounting for certain complex features of the shares of Class A Common Stock and warrants issued by the Company was not effectively designed or maintained. This

material weakness resulted in the restatement of the Company's balance sheet as of February 12, 2021 and its interim financial statements for the quarters ended March 31, 2021 and June 30, 2021. Additionally, this material weakness could result in a misstatement of the warrant liability, shares of Class A Common Stock and related accounts and disclosures that would result in a material misstatement of the financial statements that would not be prevented or detected on a timely basis. As a result of this material weakness, our management concluded that our internal control over financial reporting was not effective as of March 31, 2021, June 30, 2021 and September 30, 2021.

Any failure to maintain such internal control could adversely impact our ability to report our financial position and results of operations on a timely and accurate basis. If our financial statements are not accurate, investors may not have a complete understanding of our operations. Likewise, if our financial statements are not filed on a timely basis, we could be subject to sanctions or investigations by the stock exchange on which our common stock is listed, the SEC or other regulatory authorities. In either case, there could result a material adverse effect on our business. Failure to timely file will cause us to be ineligible to utilize short form registration statements on Form S-3 or Form S-4, which may impair our ability to obtain capital in a timely fashion to execute our business strategies or issue shares to effect an acquisition. Ineffective internal controls could also cause investors to lose confidence in our reported financial information, which could have a negative effect on the trading price of our stock.

We can give no assurance that any additional material weaknesses or restatements of financial results will not arise in the future due to a failure to implement and maintain adequate internal control over financial reporting or circumvention of these controls. In addition, even if we are successful in strengthening our controls and procedures, in the future those controls and procedures may not be adequate to prevent or identify irregularities or errors or to facilitate the fair presentation of our financial statements.

Our warrants are accounted for as liabilities and the changes in value of our warrants could have a material effect on our financial results.

On April 12, 2021, the staff of the SEC (the "SEC Staff") issued a public statement entitled "Staff Statement on Accounting and Reporting Considerations for Warrants issued by Special Purpose Acquisition Companies ("SPACs")" (the "Statement"). The Statement focused on certain settlement terms and provisions related to certain tender offers following a business combination, which terms are similar to those contained in the Warrant Agreement (as defined below) governing our warrants. As a result of the SEC Statement, we reevaluated the accounting treatment of our 8,625,000 Public Warrants and 6,266,667 Private Placement Warrants, which were initially issued by XPDI, a SPAC, and determined to classify the warrants as derivative liabilities measured at fair value, with changes in fair value each period reported in earnings.

As a result, included on XPDI's unaudited condensed balance sheets previously filed with the SEC are derivative liabilities related to embedded features contained within our warrants. Accounting Standards Codification 815, Derivatives and Hedging ("ASC 815"), provides for the remeasurement of the fair value of such derivatives at each balance sheet date, with a resulting non-cash gain or loss related to the change in the fair value being recognized in earnings in the statement of operations. As a result of the recurring fair value measurement, our condensed financial statements and results of operations may fluctuate quarterly, based on factors, which are outside of our control. Due to the recurring fair value measurement, we expect that we will recognize non-cash gains or losses on our warrants each reporting period and that the amount of such gains or losses could be material.

We may redeem your unexpired public warrants prior to their exercise at a time that is disadvantageous to you, thereby making your public warrants worthless.

We have the ability to redeem the outstanding public warrants at any time after they become exercisable and prior to their expiration, at a price of \$0.01 per warrant, if, among other things, the last reported sales price of our common stock equals or exceeds \$18.00 per share (as adjusted for adjustments to the number of shares issuable

upon exercise or the exercise price of a public warrant). Please see "Description of Capital Stock—Public Warrants." If and when the public warrants become redeemable by us, we may exercise its redemption right even if it is unable to register or qualify the underlying securities for sale under all applicable state securities laws. Redemption of the outstanding public warrants as described above could force you to (i) exercise your public warrants and pay the exercise price therefor at a time when it may be disadvantageous for you to do so, (ii) sell your public warrants at the then-current market price when you might otherwise wish to hold your public warrants or (iii) accept the nominal redemption price which, at the time the outstanding public warrants are called for redemption, we expect would be substantially less than the market value of your public warrants. None of the Private Placement Warrants will be redeemable by us so long as they are held by the Sponsor, the anchor investors or their permitted transferees.

We may be required to take write-downs or write-offs, or may be subject to restructuring, impairment or other charges that could have a significant negative effect on our financial condition, results of operations and the price of our common stock, which could cause you to lose some or all of your investment.

Factors outside of Legacy Core's control may arise at any time. As a result of these factors, Core may be forced to later write-down or write-off assets, restructure operations or incur impairment or other charges that could result in Core reporting losses. Even though these charges may be non-cash items and therefore not have an immediate impact on Core's liquidity, the fact that Core could report charges of this nature could contribute to negative market perceptions about Core or its securities. In addition, charges of this nature may cause Core to be unable to obtain future financing on favorable terms or at all

Provisions in our corporate charter documents and under Delaware law may prevent or frustrate attempts by our stockholders to change our management or hinder efforts to acquire a controlling interest in us, and the market price of our common stock may be lower as a result.

Certain provisions of our second amended and restated certificate of incorporation (the "Charter") and our Bylaws may have an anti-takeover effect and may delay, defer or prevent a merger, acquisition, tender offer, takeover attempt or other change of control transaction that a stockholder might consider in its best interest, including those attempts that might result in a premium over the market price for the shares held by our stockholders.

These provisions provide for, among other things:

- the ability of our board of directors to issue one or more series of preferred stock;
- advance notice for nominations of directors by stockholders and for stockholders to include matters to be considered at our annual meetings;
- certain limitations on convening special stockholder meetings;
- limiting the persons who may call special meetings of stockholders;
- limiting the ability of stockholders to act by written consent; and
- Our board of directors have the express authority to make, alter or repeal the Bylaws.

Moreover, because we are incorporated in Delaware, we are governed by the provisions of Section 203 of the DGCL, which prohibit a person who owns 15% or more of our outstanding voting stock from merging or combining with us for a period of three years after the date of the transaction in which the person acquired in excess of 15% of our outstanding voting stock, unless the merger or combination is approved in a prescribed manner.

These anti-takeover provisions could make it more difficult or frustrate or prevent a third party to acquire our company, even if the third party's offer may be considered beneficial by many of our stockholders. Additionally, the

provisions may frustrate or prevent any attempts by our stockholders to replace or remove its current management by making it more difficult for stockholders to replace members of our board of directors, which is responsible for appointing the members of its management. As a result, our stockholders may be limited in their ability to obtain a premium for their shares. These provisions could also discourage proxy contests and make it more difficult for you and other stockholders to elect directors of your choosing and to cause us to take other corporate actions you desire. See "Description of Capital Stock."

Our common stock market price and trading volume could decline if securities or industry analysts do not publish research or publish inaccurate or unfavorable research about our business.

The trading market for our common stock will depend in part on the research and reports that securities or industry analysts publish about us or our business. The analysts' estimates are based upon their own opinions and are often different from our estimates or expectations. If one or more of the analysts who cover us downgrade our common stock or publish inaccurate or unfavorable research about our business, the price of our securities would likely decline. If few securities analysts commence coverage of us, or if one or more of these analysts cease coverage of us or fail to publish reports on us regularly, demand for our securities could decrease, which might cause the price and trading volume of our common stock to decline.

Core's management has limited experience in operating a public company. The requirements of being a public company may strain Core's resources and divert management's attention, and the increases in legal, accounting and compliance expenses may be greater than Core anticipates.

As a public company, and particularly after we are no longer an "emerging growth company" or a "smaller reporting company," we will continue to incur significant legal, accounting and other expenses that Legacy Core did not incur as a private company. Core is subject to the reporting requirements of the Exchange Act, and is required to comply with the applicable requirements of the Sarbanes-Oxley Act, and the Dodd-Frank Wall Street Reform and Consumer Protection Act, as well as the rules and regulations subsequently implemented by the SEC and the listing standards of the Nasdaq, including changes in corporate governance practices and the establishment and maintenance of effective disclosure and financial controls. Compliance with these rules and regulations can be burdensome. Core's management and other personnel will need to devote a substantial amount of time to these compliance initiatives. Moreover, these rules and regulations have increased, and will continue to increase, Legacy Core's historical legal and financial compliance costs and will make some activities more time-consuming and costly. For example, Core expects to incur significant expenses and evote substantial management effort toward ensuring compliance with the requirements of Section 404 of the Sarbanes-Oxley Act, which will increase when Core is no longer an "emerging growth company" or a "smaller reporting company" with revenues of less than \$100 million. Core will need to hire additional accounting and financial staff, and engage outside consultants, all with appropriate public company experience and technical accounting knowledge and maintain an internal audit function, which will increase its operating expenses. Moreover, Core could incur additional compensation costs in the event that it decides to pay cash compensation closer to that of other publicly listed companies, which would increase its general and administrative expenses and could materially and adversely affect its profitability. Core will evaluate these rules and regulations,

Core's executive officers have limited experience in the management of a publicly traded company. Their limited experience in dealing with the increasingly complex laws pertaining to public companies could be a significant disadvantage in that it is likely that an increasing amount of their time may be devoted to these activities, which will result in less time being devoted to the management and growth of the post-combination company. Core may not have adequate personnel with the appropriate level of knowledge, experience and training in the accounting policies, practices or internal control over financial reporting required of public companies. Core's management will need to continually assess its staffing and training procedures to improve its internal control over financial reporting. Further, the development, implementation, documentation and assessment of appropriate processes, in addition to the need to remediate any potential deficiencies, will require

substantial time and attention from management. The development and implementation of the standards and controls necessary for us to achieve the level of accounting standards required of a public company may require costs greater than expected. It is possible that Core will be required to expand its employee base and hire additional employees to support its operations as a public company which will increase our operating costs in future periods.

We do not intend to pay dividends for the foreseeable future.

We have never declared nor paid cash dividends on our capital stock. We currently intend to retain any future earnings to finance the operation and expansion of our business, and we do not expect to declare or pay any dividends in the foreseeable future. As a result, stockholders must rely on sales of their common stock after price appreciation as the only way to realize any future gains on their investment.

The provision of our Charter requiring exclusive venue in the Court of Chancery in the State of Delaware and the federal district courts of the United States for certain types of lawsuits may have the effect of discouraging lawsuits against directors and officers.

The Charter provides that the Court of Chancery of the State of Delaware (or, if and only if, the Court of Chancery of the State of Delaware lacks subject matter jurisdiction, any state court located within the State of Delaware or, if and only if, all such state courts lack subject matter jurisdiction, the federal district court for the District of Delaware) and any appellate court therefrom shall be the sole and exclusive forum for the following claims or causes of action brought under Delaware statutory or common law: (1) any derivative claim or action brought on our behalf; (2) any claim or cause of action asserting a breach of fiduciary duty by any of our current or former director, officer or other employee; (3) any claim or cause of action asserting a claim against us arising out of, or pursuant to, the DGCL, the Charter or the Bylaws; (4) any claim or cause of action seeking to interpret, apply, enforce or determine the validity of the Charter or the Bylaws (including any right, obligation, or remedy thereunder); (5) any claim or cause of action as to which the DGCL confers jurisdiction to the Court of Chancery of the State of Delaware; or (6) any claim or cause of action asserting a claim against us or any of its directors, officers or other employees, that is governed by the internal affairs doctrine, in all cases to the fullest extent permitted by law and subject to the court having personal jurisdiction over the indispensable parties named as defendants. The aforementioned provision will not apply to claims or causes of action brought to enforce a duty or liability created by the Securities Act, the Exchange Act or any other claim for which the federal courts have exclusive jurisdiction. However, as Section 22 of the Securities Act creates concurrent jurisdiction for federal and state courts over all suits brought to enforce any duty or liability created by the Securities Act, and an investor cannot waive compliance with the federal securities laws and the rules and regulations thereunder, there is uncertainty as to whether a court would enforce such a provision. To prevent having to litigate claims in multiple jurisdictions and the threat of inconsistent or contrary rulings by different courts, among other considerations, the Charter provides that the U.S. federal district courts will be the exclusive forum for resolving any complaint asserting a cause or causes of action arising under the Securities Act, including all causes of action asserted against any defendant to such complaint.

While the Delaware courts have determined that such choice of forum provisions are facially valid, a stockholder may nevertheless seek to bring a claim in a venue other than those designated in the exclusive forum provisions. In such instance, we would expect to vigorously assert the validity and enforceability of the exclusive forum provisions of the Charter. This may require significant additional costs associated with resolving such action in other jurisdictions and there can be no assurance that the provisions will be enforced by a court in those other jurisdictions.

These exclusive forum provisions may limit a stockholder's ability to bring a claim in a judicial forum that it finds favorable for disputes with our company or our directors, officers, or other employees, which may discourage lawsuits against us or our directors, officers and other employees. If a court were to find either exclusive-forum provision in the Charter to be inapplicable or unenforceable in an action, we may incur further

significant additional costs associated with resolving the dispute in other jurisdictions, all of which could seriously harm our business.

Core qualifies as an "emerging growth company" within the meaning of the Securities Act, and if it takes advantage of certain exemptions from disclosure requirements available to emerging growth companies, such decision could make Core's securities less attractive to investors and may make it more difficult to compare Core's performance to the performance of other public companies.

Core qualifies as an "emerging growth company" as defined in Section 2(a)(19) of the Securities Act, as modified by the Jumpstart Our Business Startups Act of 2012 (the "JOBS Act"). As such, Core is eligible for and intends to take advantage of certain exemptions from various reporting requirements applicable to other public companies that are not "emerging growth companies" for as long as it continues to be an emerging growth company, including, but not limited to, (a) not being required to comply with the auditor attestation requirements of Section 404 of the Sarbanes-Oxley Act, (b) exemptions resulting in reduced disclosure obligations regarding executive compensation in Core's periodic reports and proxy statements and (c) exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved. As a result, Core's stockholders may not have access to certain information they may deem important. Core will remain an emerging growth company until the earliest of (1) the last day of the fiscal year (a) following February 12, 2026, the fifth anniversary of the Initial Public Offering, (b) in which Core has total annual gross revenue of at least \$1.07 billion or (c) in which Core is deemed to be a large accelerated filer, which means the market value of its common stock that is held by non-affiliates exceeds \$700.0 million as of the prior June 30th and (2) the date on which Core has issued more than \$1.0 billion in non-convertible debt securities during the prior three-year period. We cannot predict whether investors will find Core's securities less attractive because of its reliance on these exemptions. If some investors find Core's securities less attractive as a result of Core's reliance on these exemptions, the trading prices of Core's securities may be lower than they otherwise would be, there may be a less active trading market for Core's securities and the trading prices of Core's securities may be more

Further, Section 102(b)(1) of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies (that is, those that have not had a Securities Act registration statement declared effective or do not have a class of securities registered under the Exchange Act) are required to comply with the new or revised financial accounting standards. The JOBS Act provides that a company can elect to opt out of the extended transition period and comply with the requirements that apply to non-emerging growth companies, but that any such an election to opt out is irrevocable. Core has elected not to opt out of such extended transition period, which means that when a standard is issued or revised and it has different application dates for public or private companies, it, as an emerging growth company, can adopt the new or revised standard at the time private companies adopt the new or revised standard. This may make comparison of our financial statements with those of another public company, which is neither an emerging growth company nor an emerging growth company which has opted out of the extended transition period, difficult or impossible because of the potential differences in accounting standards used.

MARKET, INDUSTRY AND OTHER DATA

Market, ranking and industry data used throughout this prospectus is based on the knowledge of the industry and the good faith estimates of our management. We also relied, to the extent available, upon the review of independent industry surveys and other publicly available information prepared by a number of third-party sources. All of the market data and industry information used in this prospectus involves a number of assumptions and limitations, and you are cautioned not to give undue weight to such estimates. Although we believe these sources are reliable, they cannot guarantee the accuracy or completeness of this information, and they have not independently verified this information. This prospectus contains our projected financial and operational information. Such projected financial and operational information is forward-looking and is for illustrative purposes only. The information should not be relied upon as being indicative of future results. Projections, assumptions and estimates of our future performance and the future performance of the industry in which we operate are necessarily subject to a high degree of uncertainty and risk due to a variety of factors, including, among other things, the sufficiency of infrastructure, including electricity sources, the price of bitcoin, the global hash rate and the shipment of manufacturers' units on a timely basis. While we are not aware of any misstatements regarding the industry data presented herein, our estimates involve risks and uncertainties and are subject to change based on various factors, including those discussed under the headings "Risk Factors" and "Management's Discussion and Analysis of Financial Condition and Results of Operations" in this prospectus.

Certain monetary amounts, percentages and other figures included in this prospectus have been subject to rounding adjustments. Accordingly, figures shown as totals in certain tables or charts may not be the arithmetic aggregation of the figures that precede them, and figures expressed as percentages in the text may not total 100% or, as applicable, when aggregated may not be the arithmetic aggregation of the percentages that precede them

USE OF PROCEEDS

All of the shares of common stock and Private Warrants, offered by the selling securityholders pursuant to this prospectus will be sold by the selling securityholders for their respective accounts. We will not receive any of the proceeds from these sales.

We will receive up to an aggregate of approximately \$171.3 million from the exercise of the Warrants, assuming the exercise in full of all such warrants for cash. We expect to use the net proceeds from the exercise of the Warrants for general corporate purposes, including to fund potential future investments and acquisitions of companies that we believe are complementary to our business and consistent with our growth strategy. We will have broad discretion over the use of proceeds from the exercise of the Warrants. There is no assurance that the holders of the Warrants will elect to exercise any or all of such Warrants. To the extent that the Warrants are exercised on a "cashless basis," the amount of cash we would receive from the exercise of the Warrants will decrease.

DETERMINATION OF OFFERING PRICE

The offering price of the shares of common stock underlying the Warrants offered hereby is determined by reference to the exercise price of the Warrants, including \$11.50 per share for each Private Placement Warrant and Public Warrant. The Public Warrants are listed on the Nasdaq Global Select Market under the symbol "CORZW."

We cannot currently determine the price or prices at which shares of common stock or the Private Warrants may be sold by the selling securityholders under this prospectus.

MARKET INFORMATION FOR SECURITIES AND DIVIDEND POLICY

Market Information

Our common stock and Public Warrants are currently listed on the Nasdaq Global Select Market under the symbols "CORZ" and "CORZW," respectively. Prior to the closing of the Business Combination, our common stock and our Public Warrants were listed on the Nasdaq Capital Market under the symbols "XPDI" and "XPDIW," respectively. On February 4, 2022, there were 600 holders of record of the common stock and one holder of record of our Public Warrants. We currently do not intend to list the Private Placement Warrants offered hereby on any stock exchange or stock market.

Dividend Policy

We have not paid dividends on our common stock to date and do not intend to pay cash dividends. The payment of cash dividends in the future will be dependent upon revenues and earnings, if any, capital requirements and general financial condition. The payment of any dividends will be within the discretion of our board of directors. It is the present intention of our board of directors to retain all earnings, if any, for use in our business operations and, accordingly, the board of directors does not anticipate declaring any dividends in the foreseeable future.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Unless the context otherwise requires, all references in this section to "we," "us," "our," the "Company" or "Legacy Core" refer to Core Scientific Holding Co. and its subsidiaries prior to the consummation of the Business Combination.

The following discussion and analysis of our financial condition and results of operations together with our consolidated financial statements and the related notes appearing elsewhere in this prospectus. The discussion and analysis should also be read together with the section entitled "Business" and our pro forma financial information. See "Unaudited Pro Forma Combined Financial Information." This discussion and analysis contains forward-looking statements that reflect our plans, estimates and beliefs that involve risks and uncertainties that may be outside our control. As a result of many factors, such as those set forth under the headings "Risk Factors" and "Forward-Looking Statements" and elsewhere in this prospectus, our actual results may differ materially from those anticipated in these forward-looking statements. All dollar amounts are in thousands, unless otherwise specified.

Overview

Core is a best-in-class large-scale operator of dedicated, purpose-built facilities for digital asset mining and a premier provider of blockchain infrastructure, software solutions and services. We mine digital assets for our own account and provide hosting services for other large-scale miners. We are one of the largest blockchain infrastructure, hosting provider and digital asset mining companies in North America, with approximately 497MW of operating electric power as of January 31, 2022. We mine Bitcoin, Ethereum and other digital assets for third-party hosting customers and for our own account at our six fully operational data centers in North Carolina (2), Georgia (2), North Dakota (1) and Kentucky (1). In addition, in October 2021, we announced the entry of an agreement with the City of Denton, Texas and an affiliate of Tenaska Energy, Inc. to develop our seventh facility, a blockchain data center in Denton, Texas, which is expected to have 300MW of power capacity. In February 2022, the Muskogee City-County Port Authority announced an agreement with us to develop a 500MW data center at the Port of Muskogee John T. Griffin Industrial Park.

In July 2021 we completed the acquisition of Blockcap, one of our largest hosting customers. Blockcap is a blockchain technology company with industrial scale digital asset mining operations. Blockcap's primary historical business was the mining of digital asset coins and tokens, primarily Bitcoin and, to a lesser extent, Siacoin and Ethereum. At the time of its acquisition, Blockcap claimed to be the largest independent cryptocurrency mining operator in North America. While Blockcap did sell or exchange the digital assets it mined to fund its growth strategies or for general corporate purposes from time to time, it generally retained its digital assets as investments in anticipation of continued adoption of digital assets as a "store of value" and a more accessible and efficient medium of exchange than traditional fiat currencies. In addition to mining, holding and exchanging digital assets, Blockcap also evaluated and completed investments in related technologies and ancillary businesses, including RADAR, an early stage company focused on technology enhancement and development in the digital asset industry that it acquired on July 1, 2021. The acquisition of Blockcap significantly expands our self-mining operations and increases the number of miners we own. We intend to utilize RADAR's business assets and the technical expertise of its principals in enhancing our existing blockchain mining technology and software and in further strengthening our leadership position and value creation potential through the development of DeFi products and services.

Our total revenue was \$242.7 million and \$36.6 million for the nine months ended September 30, 2021 and 2020, respectively. We had operating income of \$29.3 million and an operating loss of \$5.8 million for the nine months ended September 30, 2021 and 2020, respectively. We had a net loss of \$13.2 million and \$9.9 million for the nine months ended September 30, 2021 and 2020, respectively. Our Adjusted EBITDA was \$88.0 million and \$3.1 million for the nine months ended September 30, 2021 and 2020, respectively. Adjusted EBITDA is a

non-GAAP financial measure. See "Key Business Metrics and Non-GAAP Financial Measure" below for our definition of, and additional information related to, Adjusted EBITDA.

Our total revenue was \$60.3 million and \$59.5 million for the years ended December 31, 2020 and 2019, respectively. We had an operating loss of \$6.3 million and \$11.7 million for the years ended December 2020 and 2019, respectively. We had a net loss of \$12.2 million and \$11.9 million for the years ended December 31, 2020 and 2019, respectively. Our Adjusted EBITDA was \$6.1 million for the fiscal year ended December 31, 2020 and we had an Adjusted EBITDA loss of \$1.9 million for the fiscal year ended December 31, 2019.

For the three-month period ended September 30, 2021, the Company self-mined 1,588 Bitcoin; held 2,816 Bitcoin as of such date; and achieved a self-mining hash rate of 2.64 EH/s and a hosting hash rate of 4.48 EH/s (which figure takes into account Blockcap's hosting hash rate prior to the closing of the Core/Blockcap merger).

Our Business Model

Company Overview

Core is a blockchain technology company with industrial scale digital asset mining, equipment sales and hosting operations. Our operations are currently conducted in the United States at state-of-the-art facilities specifically designed and constructed for housing advanced mining equipment, which utilize carbon free energy sources for over 50% of our energy requirements as of September 30, 2021. The Company's primary business is self-mining and hosting third-party equipment used in mining of digital asset coins and tokens, including bitcoin. Core continues to evaluate investments in related blockchain technologies and ancillary businesses.

Our rapidly growing digital asset mining operation is focused on the generation of digital assets by solving complex cryptographic algorithms to validate transactions on specific digital asset network blockchains, which is commonly referred to as "mining." Our digital asset self-mining activity competes with myriad mining operations throughout the world to complete new blocks in the blockchain and earn the reward in the form of an established unit of a digital asset. While the Company sells or exchanges a portion of the digital assets it mines to fund its growth strategies or for general corporate purposes, we will hold a portion of our digital assets as investments in anticipation of continued adoption of digital assets as a "store of value" and a more efficient medium of exchange than traditional fiat currencies. Following the Blockcap acquisition, Core significantly expanded its self-mining operation and consequently reevaluated its digital asset investment policy. Moving forward, the Company intends to adopt an investment policy pursuant to which an investment committee comprised of corporate officers use common risk management techniques to manage our assets in light of specified liquidity criteria. Liquidity will be maintained through management of a portfolio of money market instruments, obligations of the U.S. government, bank deposits, commercial paper, and certain digital asset currencies and digital asset instruments, including digital asset currencies and instruments, in accordance with the investment policy, which may involve opportunistic sales or conversions of digital asset currencies and instruments in light of market and other conditions.

As one of the largest blockchain hosting providers in North America, we focus on clients with large- scale deployments and provide power, racks, proprietary thermodynamic management (heat dissipation and airflow management), redundant connectivity, 24/7 security as well as our proprietary software platforms, MinderTM and MinderOSTM, which provide infrastructure management and custom firmware that boost performance and energy efficiency. Our blockchain business is one of the only large-scale vertically integrated digital asset mining and blockchain infrastructure and hosting solutions business in North America.

Our proprietary data centers in North Carolina, Georgia, Kentucky and North Dakota are purpose-built facilities optimized for the unique requirements of high density blockchain computer servers. These facilities have attractive long-term power contracts at stable rates with approximately 497MW of operating electrical

power as of January 31, 2022. In addition, we expect to add 300MW of power capacity from the development of our Denton, Texas Facility. In February 2022, the Muskogee City-County Port Authority announced an agreement with us to develop a 500MW data center at the Port of Muskogee John T. Griffin Industrial Park. Our existing completed facilities leverage our specialized construction proficiency by employing high-density, low-cost engineering and power designs. Our North Dakota facility is expected to benefit from our operational mining experience and techniques to maximize operational efficiency. As the demand for digital assets increases and digital assets become more widely accepted, there is an increasing demand for professional-grade, scalable infrastructure to support growth of the blockchain ecosystem. We continually evaluate our mining performance, including our ability to access additional megawatts of electric power and to expand our total self-mining and customer and related party hosting hash rates. We may explore additional mining facilities and mining arrangements in connection with our short-, medium- and long-term strategic planning.

Segments

We have two reportable segments: "Hosting and Equipment Sales" consists primarily of our blockchain infrastructure and third-party hosting business and equipment sales, and "Mining" consisting of digital asset mining for our own account. The blockchain hosting business generates revenue through the sale of consumption-based contracts for our hosting services which are recurring in nature. Equipment sales revenue is derived from the company's ability to leverage our partnership with leading equipment manufacturers to secure equipment in advance, which is then sold to our customers. The digital asset mining segment generates revenue from operating our computer equipment as part of a pool of users that process transactions conducted on one or more blockchain networks. In exchange for these services, we received digital currency assets.

Mining Equipment

We own and host specialized computers ("miners") configured for the purpose of validating transactions on multiple digital asset network blockchains (referred to as, "mining"), predominantly the Bitcoin network. Substantially all of the miners we own and host were manufactured by Bitmain and incorporate application-specific integrated circuit ("ASIC") chips specialized to solve blocks on the bitcoin blockchains using the 256-bit secure hashing algorithm ("SHA-256") in return for bitcoin digital asset rewards.

We have entered into and facilitated agreements with vendors to supply mining equipment for our and our users' digital asset mining operations. We pre-pay a significant portion of the purchase price for these new miners as partially refundable deposits, with delivery scheduled to occur in monthly installments through November 2022, and the remainder of the purchase price for these new miners is payable in installments, with payment due in advance of the scheduled delivery dates set forth in the applicable purchase agreement.

As of September 30, 2021, we had deployed 78,998 miners, which number consists of 28,004 self- miners and 50,994 hosted miners, with 125,661 and 38,007 additional self- miners and hosted miners, respectively, ordered for deployment, which is expected to increase our hash rate and our customers and related parties hash rate by approximately 12.57 EH/s and 3.80 EH/s respectively. As of September 30, 2021, the remaining payments due on the bitcoin miners on order is approximately \$320.0 million, of which we paid \$162.1 million in 2021 and expect to pay \$157.9 million in 2022. Core allocates in advance the mining equipment it orders between its self-mining operations and hosting operations conducted on behalf of customers based on its estimates of where such equipment can most profitably and efficiently be used and in accordance with contractual arrangements with its customers

The first table below summarizes the total number of self- and hosted miners deployed as of September 30, 2021. The second table below summarizes the total number of self- and hosted miners ordered for delivery during the remainder of 2021 and 2022.

	Bitcoin Miner	Bitcoin Miners in deployed	
Mining Equipment	Hash rate (EH/s)	Number of Miners	
Self-miners ¹	2.64	28,004	
Hosted miners	4.48	50,994	
Total mining equipment	7.12	78,998	

	Bitcoin Mine	Bitcoin Miners On Order	
Mining Equipment	Hash rate (EH/s)	Number of Miners	
Self-miners1	12.57	125,661	
Hosted miners	3.80	38,007	
Total mining equipment	16.37	163,668	
Total in operation and on order	23.49	242,666	

Blockcap's hash rate and number of miners is included in self-miners in the table above.

Performance Metrics

Hash Rate

Miners perform computational operations in support of digital asset blockchains measured in "hash rate" or "hashes per second." A "hash" is the computation run by mining hardware in support of the blockchain; therefore, a miner's "hash rate" refers to the rate at which it is capable of solving such computations. The original equipment used for mining bitcoin utilized the Central Processing Unit ("CPU") of a computer to mine various forms of digital assets. Due to performance limitations, CPU mining was rapidly replaced by the Graphics Processing Unit ("GPU"), which offers significant performance advantages over CPUs. General purpose chipsets like CPUs and GPUs have since been replaced as the standard in the mining industry by ASIC chips such as those found in the S17 and S19 miners Core and its customers use to mine bitcoin. These ASIC chips are designed specifically to maximize the rate of hashing operations.

Network Hash Rate

In digital assets mining, hash rate is a measure of the processing speed by a mining computer for a specific digital asset. A participant in a blockchain network's mining function has a hash rate total of its miners seeking to mine a specific digital asset and, system-wide, there is a total hash rate of all miners seeking to mine each specific type of digital asset. A higher total hash rate relative to the system-wide total hash rate generally results over time in a corresponding higher success rate in digital asset rewards as compared to mining participants with relatively lower total hash rates.

However, as the relative market price for a digital asset, such as bitcoin, increases, more users are incentivized to mine that digital asset, which increases the network's overall hash rate. As a result, a mining participant must increase its total hash rate in order to maintain its relative possibility of solving a block on the network blockchain. Achieving greater hash rate power by deploying increasingly sophisticated miners in ever greater quantities has become one of the Bitcoin mining industry's great sources of competition. Our goal is to deploy a powerful fleet of self- and hosted-miners, while operating as energy-efficiently as possible.

The Merger and Public Company Costs

We entered into a merger agreement with XPDI and the Merger Sub on July 20, 2021. Pursuant to the merger agreement, Legacy Core became a wholly owned subsidiary of XPDI at the closing of the Business Combination on January 19, 2022. Refer to the "Unaudited Pro Forma Combined Financial Information—Description of the Merger" for additional information included elsewhere in this prospectus. The merger is accounted for as a reverse recapitalization and XPDI is treated as the "acquired" company for financial reporting purposes. Legacy Core has been deemed the predecessor and Core, the post-combination company, is the successor SEC registrant, meaning that Legacy Core's financial statements for periods prior to the consummation of the merger will be disclosed in Core's future periodic reports.

As a consequence of the merger, we and XPDI collectively incurred an aggregate of \$38.0 million in professional fees associated with, legal services, M&A advisor fees, financial advice, due diligence, and other deal-related costs. Of our total transaction costs, \$1.0 million was expensed.

Core is registered with the SEC and listed on Nasdaq, which requires us to hire additional personnel and implement procedures and processes to address public company regulatory requirements and customary practices. We expect to incur additional annual expenses as a public company for, among other things, internal controls compliance and public company reporting obligations, directors' and officers' liability insurance, director fees and additional internal and external accounting and legal and administrative resources, including increased audit and legal fees.

Impact of COVID-19

In March 2020, the World Health Organization declared the global outbreak of COVID-19 to be a pandemic. We continue to closely monitor the impact of COVID-19. COVID-19 has had and continues to have an adverse impact on our business and operations, particularly as a result of preventive and precautionary measures that we, other businesses, and governments are taking. Refer to "Risk Factors" included elsewhere in this prospectus for more information.

On March 27, 2020, President Trump signed into law the "Coronavirus Aid, Relief, and Economic Security ("CARES") Act." The CARES Act, among other things, includes provisions relating to refundable payroll tax credits, deferment of employer side social security payments, net operating loss carryback periods, alternative minimum tax credit refunds, modifications to the net interest deduction limitations, increased limitations on qualified charitable contributions, and technical corrections to tax depreciation methods for qualified improvement property.

It also allocated funds for the U.S. Small Business Administration's ("SBA") Paycheck Protection Program ("PPP") loans that are forgivable in certain situations to promote continued employment, as well as Economic Injury Disaster Loans to provide liquidity to small businesses harmed by COVID-19. In April 2020, we received a loan of \$2.2 million from the PPP through the SBA. The loan was unsecured and bore interest at a rate per annum of 1% and monthly payments of principal were to begin in July 2021. The loan was due in full in April 2022, however in July 2021, we repaid the loan in full.

We are unable to predict the full impact that the COVID-19 pandemic, including variant strains of COVID-19, will have on our future results of operations, liquidity and financial condition due to numerous uncertainties, including the duration of the pandemic and the actions that may be taken by government authorities across the United States. However, COVID-19, including variant strains of COVID-19, is not expected to result in any significant changes in costs going forward. We will continue to monitor the performance of our business and assess the impacts of COVID-19 and the emergence of new variant strains of COVID-19, including potential constraints on the supply of new miners.

Key Factors Affecting Our Performance

Market Price of Digital Assets

Our business is heavily dependent on the spot price of bitcoin, as well as other digital assets. The prices of digital assets, specifically bitcoin, have experienced substantial volatility, which may reflect "bubble" type volatility, meaning that high or low prices may have little or no relationship to identifiable market forces, may be subject to rapidly changing investor sentiment, and may be influenced by factors such as technology, regulatory void or changes, fraudulent actors, manipulation, and media reporting. Bitcoin (as well as other digital assets) may have value based on various factors, including their acceptance as a means of exchange by consumers and others, scarcity, and market demand.

Our financial performance and continued growth depend in large part on our ability to mine for digital assets profitably and to attract customers for our hosting services. Increases in power costs, inability to mine digital assets efficiently and to sell digital assets at favorable prices will reduce our operating margins, impact our ability to attract customers for our services, may harm our growth prospects and could have a material adverse effect on our business, financial condition and results of operations. Over time, we have observed a positive trend in the total market capitalization of digital assets which suggests increased adoption. However, historical trends are not indicative of future adoption, and it is possible that the adoption of digital assets and blockchain technology may slow, take longer to develop, or never be broadly adopted, which would negatively impact our business and operating results.

Network Hash Rate

Our business is not only impacted by the volatility in digital asset prices, but also by increases in the digital asset, primarily Bitcoin blockchain's network hash rate resulting from the growth in the overall quantity and quality of miners working to solve blocks on the Bitcoin blockchain and the difficulty index associated with the secure hashing algorithm employed in solving the blocks.

Difficulty

The increase in digital assets, primarily, Bitcoin difficulty and hash rate reduces the mining proceeds of the equipment proportionally and eventually requires Bitcoin miners to upgrade their mining equipment to remain profitable and compete effectively with other miners.

The table below provides a summary of the impact to revenue from the increase or decrease in the market price of Bitcoin, difficulty and our hash rate. The impact to revenue in each scenario assumes only one driver increases or decreases and all others are held constant.

	Impact to R	Impact to Revenue	
<u>Driver</u>	Increase in Driver	Decrease in Driver	
Market Price of Bitcoin	Favorable	Unfavorable	
Difficulty	Unfavorable	Favorable	
Core Scientific Hash Rate	Favorable	Unfavorable	

Halvening

Further affecting the industry, and particularly for the Bitcoin blockchain, the digital asset reward for solving a block is subject to periodic incremental halvening. Halvening is a process designed to control the overall supply and reduce the risk of inflation in digital assets using a proof of work consensus algorithm. At a predetermined block, the mining reward is reduced by half, hence the term "halvening."

For bitcoin, our most significant digital asset to which the majority of our mining power is devoted, the reward was initially set at 50 bitcoin currency rewards per block. The Bitcoin blockchain has undergone halvening three times since its inception as follows: (1) on November 28, 2012 at block 210,000; (2) on July 9, 2016 at block 420,000; (3) on May 11, 2020 at block 630,000, when the reward was reduced to its current level of 6.25 bitcoin per block. The next halvening for the Bitcoin blockchain is anticipated to occur in early 2024 at block 840,000. This process will repeat until the total amount of bitcoin currency rewards issued reaches 21 million and the theoretical supply of new bitcoin is exhausted, which is expected to occur around 2140. Many factors influence the price of bitcoin and the other digital assets we mine for, and potential increases or decreases in prices in advance of or following a future halvening are unknown.

Electricity Costs

Electricity cost is the major operating cost for the mining fleet, as well as for the hosting services provided to customers and related parties. See "Power Providers and Facility Development" for additional information related to Electricity Costs.

Equipment Costs

As the market value of digital assets has increased, the demand for the newest, most efficient miners has also increased, leading to scarcity in the supply of and thereby a resulting increase in the price of miners. As a result, the cost of new machines can be unpredictable, and could also be significantly higher than our historical cost for new miners.

Similarly, as bitcoin prices have risen, we have observed a significant increase in the demand for miners. As a result, at times, we may obtain Bitmain miners and other hardware from Bitmain or from third parties at higher prices, to the extent they are available. For example, in the second half of 2020 and continuing into 2021, we have observed a significant appreciation in the market price of bitcoin, as well as an increase in the per-unit price of the new Bitmain Antminer model S19-Pro and S19j-Pro miners we purchased during this same period. While we cannot know definitively if these two phenomena are linked, we have seen a measurable increase in the prices for new miners offered by Bitmain.

Our Customers

In addition to factors underlying our mining business growth and profitability, our success greatly depends on our ability to retain and develop opportunities with our existing customers and to attract new customers. On July 30, 2021, we acquired an existing hosting customer, Blockcap, and thereby increased our self-mining operations.

Our business environment is constantly evolving, and digital asset miners can range from individual enthusiasts to professional mining operations with dedicated data centers. The Company competes with other companies that focus all or a portion of their activities on mining activities at scale. We face significant competition in every aspect of our business, including, but not limited to, the acquisition of new miners, the ability to raise capital, obtaining low-cost electricity, obtaining access to energy sites with reliable sources of power, and evaluating new technology developments in the industry.

At present, the information concerning the activities of these enterprises may not be readily available as the vast majority of the participants in this sector do not publish information publicly or the information may be unreliable. Published sources of information include "bitcoin.org" and "blockchain.info"; however, the reliability of that information and its continued availability cannot be assured.

We believe, based on available data, that the trend of increasing market prices for bitcoin and other major digital assets we observed beginning in the third fiscal quarter of calendar year 2020 has resulted in an increase

in the scale and sophistication of competition in the digital asset mining industry, with new entrants and existing competitors gaining access to substantial capital resources to build larger and larger mining operations. If this trend of increasing market prices for bitcoin and other digital assets continues, which we believe has occurred (though with significant volatility) throughout calendar year 2021, we believe many new and existing competitors may be encouraged to build or expand their Bitcoin mining operations.

Despite this trend, we believe, based on available data and assuming full deployment of the miners we have ordered from Bitmain, we have and will continue to maintain a competitive hash rate capacity among both public and private Bitcoin miners. However, to remain competitive in our evolving industry, both against new entrants into the market and existing competitors, we anticipate that we will have to continue to expand our existing miner fleet by purchasing the latest generation of miners, as well as innovating to develop and implement new technologies and mining solutions.

We believe that our integrated blockchain service portfolio, as well as our differentiated customer experience and technology, are keys to retaining and growing revenue from existing customers and to acquiring new customers. For example, we believe our significant build-out and ready power along with our MinderTM software layer represent meaningful competitive advantages favorable to our business.

Differentiation, Innovation and Expansion of Our Platform

Our investments in research and development drive differentiation of our service offerings, core technology innovation and our ability to bring new products to market.

We believe that we differentiate ourselves by offering premium products and services including our ability to manage our electricity sourcing, construct proprietary passive cooled data centers, and enable the efficient performance of commercially available mining equipment through our management software.

We intend to continue to invest in our research and development capabilities to extend our platform management and software solutions across the blockchain in order to manage our mining fleet more efficiently, expand within existing accounts, and gain new customers by offering differentiated blockchain products and services.

Grow Our Go-to-Market and Partnership Ecosystem

In addition to the activities of our sales organization, our success in our mining business as well as in retaining and attracting new customers will depend on our ability to expand our ecosystem of strategic partners.

For blockchain, strategic partners include OEM manufacturers of mining equipment, including Bitmain and others. Our blockchain business requires access to the latest generation miners. Our management team constantly evaluates current and future hardware for reliability, performance, and cost efficiency. These partnerships enable at-scale access to new equipment at competitive prices, which in turn helps secure the profitability of our fleet for the long term as well as provide our customers access to leading-edge mining technology.

Regulation

Our financial prospects and continued growth depend in part on our ability to continue to operate in a compliant manner with all rules and regulations. Our business is subject to the oversight of numerous regulatory agencies in the United States and other jurisdictions. Our strategy is to continue to invest in our finance, legal, compliance, and security functions in order to remain at the forefront of crypto policy initiatives and regulatory trends. As the industry matures, we may experience fluctuations in our operating results as a result of changes in the law and regulations that are applicable to our business, which may limit our ability to support new blockchains and digital assets, onboard customers, and offer our products and services across jurisdictions.

Strategic Acquisitions and Investments

We intend to continue growing our platform through strategic acquisitions and investments. We plan to acquire and invest in companies with complementary products and technologies. Our strategic acquisitions may affect our future financial results. We will also continue to enter into strategic partnerships with various companies to scale our business, provide best-in-class blockchain infrastructure and hosting to our clients, grow our digital asset mining practice, and pioneer innovative financial products and services through blockchain technology.

Merger Agreement

On July 20, 2021, we entered into the merger agreement with XPDI and the Merger Sub. Pursuant to the merger agreement, and subject to the terms and conditions set forth therein, XPDI acquired Legacy Core through a series of transactions, including (x) Merger Sub merging with and into Legacy Core, with Legacy Core surviving the First Merger as a wholly owned subsidiary of XPDI, and (y) following the closing of the First Merger, Legacy Core merging with and into XPDI, with XPDI surviving the Second Merger. As a result of the merger, among other things, each outstanding share of Legacy Core common stock was cancelled in exchange for the right to receive a number of shares of Core common stock in an amount that is equal to the quotient obtained by dividing (a) an amount equal to (x) \$4.0 billion, divided by (y) the number of shares of Core common stock on a fully-diluted basis, by (b) \$10.00.

2019 and 2020 Acquisitions

Through December 31, 2020 we have completed and substantially integrated three asset acquisitions primarily related to software and patents, Stax, Atrio, and RStor. We routinely evaluate potential acquisitions that align with our growth strategy. Our acquisitions in any period may impact the comparability of our results with prior and subsequent periods. The integration of acquisitions also requires dedication of substantial time and resources, and we may never fully realize synergies and other benefits that we expect.

2021 Acquisitions

In July 2021, we acquired Blockcap, a Nevada corporation engaged in the business of digital asset mining and through its subsidiary, RADAR, an early stage company focused on technology enhancement and development in the digital asset industry. Pursuant to the Core/Blockcap merger agreement, each eligible share of Blockcap was converted into shares of Core common stock in accordance with a determined ratio, and Blockcap was merged with a wholly owned subsidiary of Legacy Core. As a result of the merger we acquired for self-mining all of the digital asset mining machines owned by Blockcap and hosted by us in our facilities. As a result, the existing hosting agreement between Legacy Core and Blockcap was terminated. We intend to utilize RADAR's business assets and the technical expertise of its principals in enhancing our existing blockchain mining technology and software and in further strengthening our leadership position and value creation potential through the development of DeFi products and services.

Condensed Statement of Operations

The following table presents a condensed statement of operations for the nine months ended September 30, 2021 and 2020 and the years ended December 31, 2020 and 2019:

		Nine Months Ended September 30,		Ended ber 31,
	2021	2020	2020	2019
	(in thou	(in thousands)		usands)
Total Revenue	\$242,688	\$36,633	\$ 60,320	\$ 59,523
Cost of revenue	145,193	31,906	50,928	48,996
Gross profit	97,495	4,727	9,392	10,527
Gain on legal settlements	(2,603)	5,814	5,814	
Gain from sales of digital currency assets	405	52	69	387
Impairment of digital currency assets	(12,552)	(4)	(4)	419
Total operating expenses	53,409	16,382	21,598	23,020
Operating Income (Loss)	29,336	(5,793)	(6,327)	(11,687)
Total non-operating income (expense), net	(43,227)	(4,104)	(5,879)	(235)
Income (loss) before income taxes	(13,891)	(9,897)	(12,206)	(11,922)
Income tax benefit	(697)			
Net Income (Loss)	<u>\$(13,194)</u>	\$ (9,897)	\$(12,206)	\$(11,922)

Key Business Metrics and Non-GAAP Financial Measure

In addition to our financial results, we use the following business metrics and non-GAAP financial measures to evaluate our business, measure our performance, identify trends affecting our business, and make strategic decisions. For a definition of these key business metrics, see the section titled "Self-Mining Hash rate and Adjusted EBITDA."

	Nine Mon Septem		Year I Decem	Ended ber 31,
	2021	2020	2020	2019
Self-Mining Hash rate (Exahash), EoP	2.64	0.12	0.35	0.08
Adjusted EBITDA (\$ Millions)	\$ 88.0	\$ 3.1	\$ 6.1	\$ (1.9)

Self-Mining Hash rate

We operate mining hardware which performs computational operations in support of the blockchain measured in "hash rate" or "hashes per second." A "hash" is the computation run by mining hardware in support of the blockchain; therefore, a miner's "hash rate" refers to the rate at which the hardware is capable of solving such computations. Our hash rate represents the hash rate of our miners as a proportion of the total Bitcoin network hash rate and drives the number of digital asset rewards that will be earned by our fleet. We calculate and report our hash rate in exahash per second ("EH/s"). One exahash equals one quintillion hashes per second.

We measure the hash rate produced by our mining fleet through our management software MinderTM, which consolidates the reported hash rate from each miner. The method by which we measure our hash rate may differ from how other operators present such measure.

Our self-mining hash rate was 2.64 EH/s and 0.12 EH/s for the nine months ended September 30, 2021 and 2020, respectively.

Our self-mining hash rate, inclusive of Blockcap ("combined self-mining") and customer and related party hosting hash rate grew 388%, to 7.12 EH/s for the nine months ended September 30, 2021 from 1.46 EH/s for the nine months ended September 30, 2020.

As a result of the growth in our self-mining machines and increased efficiency, we have seen substantial improvement in our self-mining hash rate over the last two years, growing the rate to 0.35 EH/s for the year ended December 31, 2020 from 0.08 EH/s for the year ended December 31, 2019.

Our combined self-mining hash rate, for the year ended December 31, 2020 and 2019, was 1.13 EH/s and 0.08 EH/s, respectively.

Our combined self-mining and customer and related party hosting hash rate grew 68%, to 2.27 EH/s for the year ended December 31, 2020 from 1.35 EH/s for the year ended December 31, 2019.

Adjusted EBITDA

Adjusted EBITDA is a non-GAAP financial measure defined as our net income or (loss), adjusted to eliminate the effect of (i) interest income, interest expense, and other income (expense), net; (ii) provision for income taxes; (iii) depreciation and amortization; (iv) stock-based compensation expense; and (v) certain additional non-cash and non-recurring items. For additional information, including the reconciliation of net income (loss) to Adjusted EBITDA, please refer to the table below. We believe Adjusted EBITDA is an important measure because it allows management, investors, and our board of directors to evaluate and compare our operating results, including our return on capital and operating efficiencies, from period-to-period by making the adjustments described above. In addition, it provides useful information to investors and others in understanding and evaluating our results of operations, as well as provides a useful measure for period-to-period comparisons of our business, as it removes the effect of net interest income (expense), taxes, certain non-cash items, variable charges, and timing differences. Moreover, we have included Adjusted EBITDA in this prospectus because it is a key measurement used by our management internally to make operating decisions, including those related to operating expenses, evaluate performance, and perform strategic and financial planning.

The above items are excluded from our Adjusted EBITDA measure because these items are non-cash in nature, or because the amount and timing of these items is unpredictable, not driven by core results of operations and renders comparisons with prior periods and competitors less meaningful. However, you should be aware that when evaluating Adjusted EBITDA, we may incur future expenses similar to those excluded when calculating these measures. Our presentation of this measure should not be construed as an inference that its future results will be unaffected by unusual or non-recurring items. Further, this non-GAAP financial measure should not be considered in isolation from, or as a substitute for, financial information prepared in accordance with GAAP. We compensate for these limitations by relying primarily on GAAP results and using Adjusted EBITDA on a supplemental basis. Our computation of Adjusted EBITDA may not be comparable to other similarly titled measures computed by other companies because not all companies calculate this measure in the same fashion. You should review the reconciliation of net income (loss) to Adjusted EBITDA below and not rely on any single financial measure to evaluate our business.

The following table presents a reconciliation of net income (loss) to Adjusted EBITDA for the Nine months ended September 30, 2021 and 2020 and the years ended December 31, 2020 and 2019:

	Nine Months Ended September 30,		Year Ended December 31,	
	2021	2020	2020	2019
	(in thou	sands)	(in thou	isands)
Adjusted EBITDA				
Net income (loss)	\$(13,194)	\$(9,897)	\$(12,206)	\$(11,922)
Adjustments:				
Interest expense, net	34,566	4,104	5,879	235
Income tax expense (benefit)	(697)	_	_	_
Depreciation and amortization	12,886	6,613	9,403	6,118
Stock-based compensation expense	31,012	2,446	3,037	2,880
Impairment of digital curreny assets	12,552	5	4	(419)
Legal settlement	2,603	_	_	_
Fair value adjustment on convertible note payable	8,661	_	_	_
Other items	(388)	(144)	(66)	1,166
Adjusted EBITDA	\$ 88,001	\$ 3,128	\$ 6,051	\$ (1,942)

Components of Results of Operations

Revenue

Our revenue consists primarily of returns from our hosting operations, including the sales of mining equipment to be hosted in our data centers and proceeds related to digital currency transaction processing (digital asset mining income) fees.

- Hosting revenue from customers and related parties. Hosting revenue from customers and related parties is based on consumption-based contracts with our customers and related parties. Most contracts are renewable, and our customers are generally billed on a fixed and recurring basis each month for the duration of their contract, which vary from one to three years in length. See "Certain Relationships and Related Party Transactions."
- Equipment sales to customers and related parties. Equipment sales to customers and related parties is derived from our ability to leverage our partnerships with leading equipment manufacturers to secure equipment in advance, which is then sold to our customers and related parties. Our equipment sales are typically in connection with a hosting contract.
- Digital asset mining income. We operate a digital asset mining operation using specialized computers equipped with application-specific integrated circuit (ASIC) chips (known as "miners") to solve complex cryptographic algorithms in support of the Bitcoin blockchain (in a process known as "solving a block") in exchange for digital asset rewards (primarily bitcoin). The Company participates in "mining pools" organized by "mining pool operators" in which we share our mining power (known as "hash rate") with the hash rate generated by other miners participating in the pool to earn digital asset rewards. The mining pool operator provides a service that coordinates the computing power of the independent mining enterprises participating in the mining pool. Fees are paid to the mining pool operator to cover the costs of maintaining the pool. The pool uses software that coordinates the pool members' mining power, identifies new block rewards, records how much hash rate each participant contributes to the pool, and assigns digital asset rewards earned by the pool among its participants in proportion to the hash rate each participant contributed to the pool in connection with solving a block. Revenues from digital asset mining are impacted by volatility in bitcoin prices, as well as increases in the Bitcoin blockchain's network hash rate resulting from the growth in the overall quantity and quality of miners working to solve blocks on the Bitcoin blockchain and the difficulty index associated with

the secure hashing algorithm employed in solving the blocks. The diagram below provides a simple illustration of the calculation of our annual digital asset mining income.



1 Amount represents the average number of blocks mined per year, e.g., blocks are mined on average every 10 minutes, or 144 per day, 52,560 per year.

Cost of Revenue

Cost of revenue consists of direct production costs of hosting and mining operations and costs of equipment sold to customers and related parties. Direct production costs of hosting and mining operations primarily includes electricity costs, employee wages and benefits, certain professional fees, other direct costs of facilities operations and depreciation and amortization expenses on property, plant and equipment used to provide hosting services and in support of self-mining.

Operating expense

Operating expense consists of research and development, sales and marketing, general and administrative and impairment of digital currency asset expenses. Each is outlined in more detail below.

- Research and development. We invest in research and development to build capabilities to extend our blockchain platform management
 and software solutions, in order to manage our mining fleet more efficiently, expand within existing accounts, and to gain new customers
 by offering differentiated blockchain hosting services. Research and development costs include compensation and benefits, other personnel
 related costs and professional fees.
- Sales and marketing. Sales and Marketing expenses consist of marketing expenses, trade shows and events, professional fees, compensation and benefits and other personnel related costs.
- General and administrative. General and administrative expenses include compensation and benefits expenses for employees, who are
 not part of the research and development and sales and marketing organization, professional fees, and other personnel related expenses.
 Also included is stock-based compensation, insurance, amortization of intangibles, gain (loss) on disposals of property, plant and
 equipment, asset impairments, franchise taxes, and bank fees.

(Loss) gain on legal settlements

(Loss) gain on legal settlements represent settlement losses related to amounts paid to former customers and gains related to amounts received as part of the resolution of legal actions for damages resulting from the early termination of agreements by former customers.

Gain (loss) from sales of digital currency assets

Gain (loss) from sales of digital currency assets are recorded when realized upon sale(s). In determining the gain (loss) to be recognized upon sale, we calculate the difference between the sales price and carrying value of the digital assets sold immediately prior to sale.

Impairment of digital currency assets.

We initially recognize digital currency assets that are received as digital asset mining income based on the fair value of the digital currency assets. Digital currency assets that are purchased in an exchange of one digital currency asset for another digital currency asset are recognized at the fair value of the asset received.

These assets are adjusted to fair value only when an impairment is recognized, or the underlying asset is held for sale. Impairment exists when the carrying amount exceeds its fair value, which is measured using the quoted price of the digital currency asset at the time its fair value is being measured, which is measured on a daily basis. To the extent that an impairment loss is recognized, the loss establishes the new costs basis of the digital asset.

Impairment losses are recognized in the period in which the impairment is identified. The impaired digital assets are written down to their fair value at the time of impairment and this new cost basis will not be adjusted upward for any subsequent increase in fair value. See Note "Digital Assets" for further information.

Non-operating income (expense), net

Non-operating income (expense), net includes gain (loss) on debt extinguishment, interest income, interest expense and other non-operating expenses, net.

Income tax expense (benefit)

Income tax expense (benefit) consists of U.S. federal, state and local income taxes, if any. For the nine months ended September 30, 2021 and 2020, our income tax benefit was \$0.7 million and \$0.0, respectively.

Our effective tax rate for the nine months ended September 30, 2021 was 5.0%. The estimated effective tax rate of 5.0% was lower than the US statutory of 21% primarily due to the valuation allowance on the Company's losses.

Our effective rate for the nine months ended September 30, 2020 was 0%. The estimated effective tax rate of 0% was lower than the US statutory of 21% primarily due to the full valuation allowance on the Company's losses.

As of September 30, 2021, our deferred tax assets were composed primarily of federal and state net operating loss carryforward, capital loss carryforward, and share based compensation.

Realization of deferred tax assets is dependent upon the generation of future taxable income, the timing and amount of which are uncertain. In determining the need for a valuation allowance, we evaluate both positive and negative evidence when concluding whether it is more likely than not that deferred tax assets are realizable.

In connection with the Blockcap Inc. acquisition on July 30, 2021, the Company did not record a deferred tax asset due to a full valuation allowance.

Results of Operations

The following table sets forth our selected consolidated statements of operations for each of the periods indicated. The discussion in this Results of Operations gives effect to the correction of certain immaterial errors in our Consolidated Financial Statements for the nine months ended September 30, 2021. Refer to Note 2—Revision of Previously-Issued Financial Statements for further information.

		Nine Months Ended September 30,		Ended ber 31,
	2021	usands)	2020 (in thou	2019
Total Revenue:	(in the	usanus)	(in thot	isanus)
Hosting revenue from customers	\$ 37,836	\$28,667	\$ 34,615	\$ 53,492
Hosting revenue from related parties	13,906	3,382	6,983	384
Equipment sales to customers	84,378	1,987	11,193	_
Equipment sales to related parties	29,057	285	1,402	_
Digital asset mining income	77,511	2,312	6,127	5,647
Total Revenue	242,688	36,633	60,320	59,523
Cost of revenue	145,193	31,906	50,928	48,996
Gross profit	97,495	4,727	9,392	10,527
(Loss) gain on legal settlements	(2,603)	5,814	5,814	
Gain from sales of digital currency assets	405	52	69	387
Impairment of digital currency assets	(12,552)	(4)	(4)	419
Operating expenses:				
Research and development	4,231	4,184	5,271	5,480
Sales and Marketing	2,186	1,401	1,771	2,833
General and administrative	46,992	10,797	14,556	14,707
Total operating expenses	53,409	16,382	21,598	23,020
Operating Income (Loss)	29,336	(5,793)	(6,327)	(11,687)
Non-operating income (expense), net:				
Loss on debt extinguishment	(8,016)	(1,333)	(1,333)	_
Interest expense, net	(26,550)	(2,683)	(4,436)	(235)
Other non-operating expenses, net	(8,661)	(88)	(110)	
Total non-operating income (expense), net .	(43,227)	(4,104)	(5,879)	(235)
Income (loss) before income taxes	(13,891)	(9,897)	(12,206)	(11,922)
Income tax expense	(697)			
Net Income (Loss)	\$(13,194)	\$(9,897)	\$(12,206)	\$(11,922)

The following table sets forth our selected consolidated statements of operations as a percentage of total revenue for each of the periods indicated.

	End	Nine Months Ended September 30,		Year Ended December 31.	
	2021	2020	2020	2019	
Total Revenue:			· 		
Hosting revenue from customers	16%	78%	57%	90%	
Hosting revenue from related parties	6	9	12	1	
Equipment sales to customers	35	5	19	_	
Equipment sales to related parties	12	1	2	_	
Digital asset mining income	32	6	10	9	
Total Revenue	100	100	100	100	
Cost of revenue	60	87	84	82 18	
Gross profit	40	13	_16	18	
(Loss) gain on legal settlements	(1)	16	10	_	
Gain from sales of digital currency assets	0	0	0	1	
Impairment of digital currency assets	(5)	(0)	0	1	
Operating expenses:					
Research and development	2	11	9	9	
Sales and Marketing	1	4	3	5	
General and administrative	19	29	24	25	
Total operating expenses	22	45	36	39	
Operating Income (Loss)	17	(16)	(10)	(20)	
Non-operating income (expense), net:					
Loss on debt extinguishment and other	(3)	(4)	(2)	_	
Interest expense, net	(11)	(7)	(7)	(0)	
Other non-operating expenses, net	_(4)	(0)	_(0)		
Total non-operating income (expense), net	(18)	(11)	(10)	(0)	
Income (loss) before income taxes	(6)	(27)	(20)	(21)	
Income tax expense	(0)		_		
Net Income (Loss)	(5)%	(27)%	(20)%	(21)%	

Comparison of the Nine months Ended September 30, 2021 and 2020

	Nine Months Ended September 30,		Period to Period Change	
	2021	2020	\$ Change	% Change
	(in thous		(in thous	
Total Revenue:	,		,	
Hosting revenue from customers	\$ 37,836	\$28,667	\$ 9,169	32%
Hosting revenue from related parties	13,906	3,382	10,524	311%
Equipment sales to customers	84,378	1,987	82,391	4147%
Equipment sales to related parties	29,057	285	28,772	n.m.
Digital asset mining income	77,511	2,312	75,199	3253%
Total Revenue	\$242,688	\$36,633	\$206,055	562%
Percentage of total revenue:				
Hosting revenue from customers	16%	78%		
Hosting revenue from related parties	6	9		
Equipment sales to customers	35	5		
Equipment sales to related parties	12	1		
Digital asset mining income	32	6		
Total Revenue	100	100		

Total revenue increased by \$206.1 million or 562%, to \$242.7 million for the nine months ended September 30, 2021 from \$36.6 million for the nine months ended September 30, 2020.

Total hosting revenue from customers increased by \$9.2 million or 32%, to \$37.8 million for the nine months ended September 30, 2021 from \$28.6 million for the nine months ended September 30, 2020.

Total hosting revenue from related parties increased by \$10.5 million, to \$13.9 million for the nine months ended September 30, 2021 from \$3.4 million for the nine months ended September 30, 2020.

The increase in hosting revenue from customers and related parties was driven by the recovery of the market, post-pandemic, and customer spending returning to prior levels.

Equipment sales to customers increased by \$82.4 million, to \$84.4 million for the nine months ended September 30, 2021 from \$2.0 million for the nine months ended September 30, 2020. Equipment sales to related parties increased by \$28.8 million, to \$29.1 million for the nine months ended September 30, 2021 from \$0.3 million for the nine months ended September 30, 2020. The increase was driven by higher demand for new generation mining equipment resulting from the Bitcoin halvening event.

Digital asset mining income increased by \$75.2 million, to \$77.5 million for the nine months ended September 30, 2021 from \$2.3 million for the nine months ended September 30, 2020. The year over year growth in mining income was driven by an increase in our self-mining hash rate and higher bitcoin prices. Our self-mining hash rate increased to 2.64 EH/s for the nine months ended September 30, 2021 from 0.12 EH/s for the nine months ended September 30, 2020. The total number of bitcoin awarded for the nine months ended September 30, 2021 was 1,525 compared to 104 for the nine months ended September 30, 2020, representing an increase of 1,365%. The average price of bitcoin for the nine months ended September 30, 2021 was \$44,591 compared to \$9,189 for the nine months ended September 30, 2020, an improvement of 385%.

Cost of revenue

		Nine Months Ended September 30,		Period ge
	2021	2020 ands)	\$ Change (in thous	% Change ands)
Cost of revenue	\$145,193	\$31,906	\$113,287	355%
Gross profit	97,495	4,727	92,768	1963%
Gross margin	40%	13%	_	27%

Cost of revenue increased by \$113.3 million or 355%, to \$145.2 million for the nine months ended September 30, 2021 from \$31.9 million for the nine months ended September 30, 2020. The increase was primarily attributable to an increase in the cost of equipment sold to customers and related parties and higher power consumption. As a percentage of total revenue, cost of revenue totaled 60% and 87% for the nine-months ended September 30, 2021 and 2020, respectively.

(Loss) gain on legal settlements

The Company recognized a (loss) on legal settlement of \$2.6 million during the nine months ended September 30, 2021 related to a settlement with a former customer. The Company recognized a gain on legal settlement of \$5.8 million for the nine months ended September 30, 2020 related to the resolution of legal actions for damages resulting from the early termination of agreements by former customers.

Gain (loss) from sales of digital currency assets

	Nine Mo	onths		
	Ended		Period to Period	
	September 30,		Change	
			\$	%
	2021	2020	Change	Change
	(in thous	ands)	(in thou	ısands)
Gain (loss) from sales of digital currency assets	\$405	\$52	\$ 353	679%
Percentage of total revenue	0%	0%		

Gain (loss) from sales of digital currency assets increased by \$0.3 million or 679%, to \$0.4 million for the nine months ended September 30, 2021 from \$0.1 million for nine months ended September 30, 2020. Gains (losses) are recorded when realized upon sale(s). In determining the gain (loss) to be recognized upon sale, we calculate the difference between the sales price and carrying value of the digital assets sold immediately prior to sale. For the nine months ended September 30, 2021, the carrying value of our digital assets sold was \$27.7 million and the sales price was \$28.1 million. For the nine months ended September 30, 2020 the amounts were nominal.

Impairment of digital currency assets

	Nine Months	Ended			
	Septembe	September 30,		Period to Period Change	
				%	
	2021	2020	\$ Change	Change	
	(in thousa	nds)	(in thous	ands)	
Impairment of digital currency assets	\$(12,552)	\$ (4)	\$ (12,548)	n.m.	
Percentage of total revenue	(5)%	0%			

Impairment of digital currency assets increased by \$(12.5) million, to \$(12.6) million for the nine months ended September 30, 2021 from \$4.0 thousand for nine months ended September 30, 2020. Impairment exists

when the carrying amount exceeds its fair value, which is measured using the quoted price of the digital currency asset at the time its fair value is being measured. We perform an analysis each quarter to identify whether events or changes in circumstances, principally decreases in the quoted prices on active exchanges, indicate that it is more likely than not that our digital assets are impaired. In determining if an impairment has occurred, we consider the lowest market price of one unit of digital asset quoted on the active exchange since acquiring the digital asset. If the then current carrying value of a digital asset exceeds the fair value so determined, an impairment loss has occurred with respect to those digital assets in the amount equal to the difference between their carrying values and the price determined.

The carrying value of our digital currency assets amounted to \$115.8 million and \$0.1 million as of September 30, 2021 and December 31, 2020 respectively, reflecting cumulative impairments of \$12.9 million and \$0.3 million, attributable to bitcoin trading price fluctuations.

Operating Expenses

Research and development

		Nine Months Ended September 30.		Period to Period Change		
				3	%	
	2021	2020	Cha	nge	Change	
	(in thou	ısands)	(in thousands)		ousands)	
Research and development	\$ 4,231	\$ 4,184	\$	47	1%	
Percentage of total revenue	2%	11%				

Research and development expenses were consistent year over year and equal to \$4.2 million for the nine months ended September 30, 2021 and September 30, 2020.

Sales and marketing

		nths Ended ober 30,	Period to Period Change	
				%
	2021	2020	Change	Change
	(in tho	usands)	(in thou	ısands)
Sales and marketing	\$ 2,186	\$ 1,401	\$ 785	56%
Percentage of total revenue	1%	4%		

Sales and marketing expenses increased by \$0.8 million or 56%, to \$2.2 million for the nine months ended September 30, 2021 from \$1.4 million for the nine months ended September 30, 2020. The increase was driven by higher marketing and advertising expenses of \$0.6 million, increased travel and entertainment expenses of \$0.1 million, and project related expenses of \$0.1 million.

General and administrative

	Nine Mont Septem		Period to Period Change	
		2020 sands)	\$ Change Change (in thousands)	
General and administrative	\$46,992	\$10,797	\$36,195	335%
Percentage of total revenue	19%	29%		

General and administrative expenses increased by \$36.2 million or 335%, to \$47.0 million for the nine months ended September 30, 2021 from \$10.8 million for the nine months ended September 30, 2020. The

increase of \$36.2 million was driven by \$28.6 million higher stock-based compensation, \$3.2 million of investments made to support public company readiness, \$2.1 million higher personnel and related costs, increased travel and entertainment expenses of \$0.6 million, increase in intangible asset amortization of \$0.5 million, higher insurance costs of \$0.2 million and increases in and other miscellaneous increases of \$1.0 million.

Non-operating income (expenses), net

		Nine Months Ended September 30,		Period ge
	2021 (in thous	2021 2020 (in thousands)		% Change ands)
Non-operating income (expense), net:	`	,	`	,
Loss on debt extinguishment	\$ (8,016)	\$(1,333)	\$ (6,683)	501%
Interest expense, net	(26,550)	(2,683)	(23,867)	890%
Other non-operating expenses, net	(8,661)	(88)	(8,573)	n.m.
Total non-operating income (expense), net	\$(43,227)	\$(4,104)	\$(39,123)	953%

Total non-operating income (expense), net increased by \$(39.1) million or 953%, to \$(43.2) million for the nine months ended September 30, 2021 from \$(4.1) million for the nine months ended September 30, 2020. The increase in non-operating income (expense), net of \$(39.1) million was driven by the loss on debt extinguishment of \$(6.7) million and higher interest expense, net of \$(23.9) million related to the execution of several funding agreements, including a senior secured credit facility and several equipment financing agreements, the proceeds of which were used to acquire self-mining equipment and to fund the build out of our hosting and self-mining facilities, and other non-operating expenses, net of \$(8.6) million to adjust our convertible note payable to fair value. See Note 2—Fair Value Measurements for further information.

Income tax expense (benefit)

Income tax expense (benefit) consists of U.S. federal, state and local income taxes, if any. For the nine months ended September 30, 2021 and 2020, our income tax benefit was \$0.7 million and \$0.0, respectively.

Our effective tax rate for the nine months ended September 30, 2021 was 5.0%. The estimated effective tax rate of 5.0% was lower than the US statutory of 21% primarily due to the valuation allowance on the Company's losses.

Our effective rate for the nine months ended September 30, 2020 was 0%. The estimated effective tax rate of 0% was lower than the US statutory of 21% primarily due to the full valuation allowance on the Company's losses.

As of September 30, 2021, our deferred tax assets were composed primarily of federal and state net operating loss carryforward, capital loss carryforward, and share based compensation.

Realization of deferred tax assets is dependent upon the generation of future taxable income, the timing and amount of which are uncertain. In determining the need for a valuation allowance, we evaluate both positive and negative evidence when concluding whether it is more likely than not that deferred tax assets are realizable.

In connection with the Blockcap Inc. acquisition on July 30, 2021, the Company did not record a deferred tax asset due to a full valuation allowance.

Segment Total Revenue and Gross Profit

The following table presents total revenue and gross profit by reportable segment for the periods presented:

		For the Nine Months Ended September 30,		Period ge
	2021	2020	\$ Change	% Change
Hosting and Equipment Sales Segment	(in thou	sands)	(in thous	ands)
Hosting revenue from customers	\$ 51,742	\$32,049	\$ 19,693	61%
Equipment sales to customers	113,435	2,272	111,163	n.m.
Total revenue	165,177	34,321	130,856	381%
Cost of revenue	131,284	30,308	100,976	333%
Gross profit	\$ 33,893	\$ 4,013	\$ 29,880	745%
Mining Segment				
Digital asset mining income	\$ 77,511	\$ 2,312	\$ 75,199	n.m.
Total revenue	\$ 77,511	2,312	75,199	n.m.
Cost of revenue	13,909	1,598	12,311	770%
Gross profit	\$ 63,602	\$ 714	\$ 62,888	n.m.
Consolidated total revenue	\$242,688	\$36,633	\$206,055	562%
Consolidated cost of revenue	\$145,193	\$31,906	\$113,287	355%
Consolidated gross profit	\$ 97,495	\$ 4,727	\$ 92,768	1963%

For the nine months ended September 30, 2021 and 2020, the top three customers accounted for approximately 73% and 64%, respectively, of the Equipment Sales and Hosting's segment total revenue. For the nine months ended September 30, 2021 and 2020, the concentration of customers comprising 10% or more of the Company's revenue was as follows:

	For the Nine Months Ended September 30,		
Customer	2021	2020	
A	34%	n/a	
В	21%	n/a	
C (related party)	18%	n/a	
D	n/a	42%	
E	n/a	16%	

A reconciliation of the reportable segment gross profit to income (loss) before income taxes included in our consolidated statements of operations for the years ended September 30, 2021 and 2020, is as follows:

	For the Nine Months Ended September 30,		Period to Period Change	
	2021 (in thou	2020 sands)	\$ Change (in thousa	Change ands)
Reportable segment gross profit	\$ 97,495	\$ 4,727	\$ 92,768	1963%
(Loss) gain on legal settlements	(2,603)	5,814	(8,417)	(145%)
Gain (loss) from sales of digital currency assets	405	52	353	679%
Impairment of digital currency assets	(12,552)	(4)	(12,548)	n.m.
Operating expense (income):				
Research and development	4,231	4,184	47	1%
Sales and marketing	2,186	1,401	785	56%
General and administrative	46,992	10,797	36,195	335%
Total operating expense (income)	53,409	16,382	37,027	226%
Operating Income (Loss)	29,336	(5,794)	35,130	(606%)
Other income (expense):				
Loss on debt extinguishment	(8,016)	(1,333)	(6,683)	501%
Interest expense, net	(26,550)	(2,683)	(23,867)	890%
Other non-operating expenses, net	(8,661)	(88)	(8,573)	n.m.
Total other income (expense)	(43,227)	(4,104)	(39,123)	953%
Income (loss) before income taxes	\$(13,891)	\$ (9,897)	\$ (3,993)	40%

Comparison of the Years Ended December 31, 2020, and 2019

	Year Ended December 31,		Period to I Chang	
	2020 (in thou	2019	\$ Change (in thousa	% Change ands)
Total Revenue:	((11 111 111	
Hosting revenue from customers	\$34,615	\$53,492	\$(18,877)	(35%)
Hosting revenue from related parties	6,983	384	6,599	n.m.
Equipment sales to customers	11,193	_	11,193	_
Equipment sales to related parties	1,402	_	1,402	_
Digital asset mining income	6,127	5,647	480	8%
Total Revenue	\$60,320	\$59,523	\$ 797	1%
Percentage of total revenue:				
Hosting revenue from customers	57%	90%		
Hosting revenue from related parties	12	1		
Equipment sales to customers	19	_		
Equipment sales to related parties	2	_		
Digital asset mining income	10	9		
Total Revenue	100	100		

Total revenue increased by \$0.8 million or 1%, to \$60.3 million for the year ended December 31, 2020 from \$59.5 million for the year ended December 31, 2019.

Total hosting revenue from customers decreased by \$18.9 million or 35%, to \$34.6 million for the year ended December 31, 2020 from \$53.5 million for the year ended December 31, 2019. The decrease was driven by a delay in the post-halvening refresh of mining equipment due to supply constraints for new miners as a result of COVID-19 and the early termination of hosting agreements by former customers due to change in mining economics post halvening. The next bitcoin halvening is expected to occur in the Spring of 2024.

Total hosting revenue from related parties increased by \$6.6 million, to \$7.0 million for the year ended December 31, 2020 from \$0.4 million for the year ended December 31, 2019. The increase in related party hosting contracts was driven by the recovery of the market, post-pandemic, and customer spending levels beginning to return to prior levels. See "Certain Relationships and Related Party Transactions."

Equipment sales to customers and related parties was \$11.2 million and \$1.4 million, respectively for the year ended December 31, 2020. The increase in sales was driven by higher demand for new generation mining equipment resulting from the Bitcoin halvening event.

Digital asset mining income increased by \$0.5 million or 8%, to \$6.1 million for the year ended December 31, 2020 from \$5.6 million for the year ended December 31, 2019. The year over year growth in mining income was driven by an increase in our self-mining hash rate and higher bitcoin prices, offset by a decrease in the number of bitcoin awarded. Our self-mining hash rate increased by 355%, to 0.35 EH/s for the year ended December 31, 2020 from 0.08 EH/s for the year ended December 31, 2019. The average price of bitcoin for the year ended December 31, 2020 was \$14,357 compared to \$6,880 for the year ended December 31, 2019, an improvement of 109%. The total number of bitcoin awarded for the year ended December 31, 2020 was 328 compared to 516 for the year ended December 31, 2019, representing a decrease of 36%.

Cost of Revenue

		Year Ended December 31,		Period nge
	2020	2020 2019		% Change
	(in thou	(in thousands)		sands)
Cost of revenue	\$50,928	\$48,996	\$ 1,932	4%
Gross profit	9,392	10,527	(1,135)	(11%)
Gross margin	16%	18%	_	(2%)

Cost of revenue increased by \$2.0 million or 4%, to \$51.0 million for the year ended December 31, 2020 from \$49.0 million for the year ended December 31, 2019. The increase was primarily attributable to an increase of \$11.0 million in cost of equipment sold to customers and related parties, offset by a decrease of \$8.4 million in overall power fees, driven by a delay in the post-halvening refresh of customer and related party mining equipment.

As a percentage of total revenue, cost of revenue totaled 84% and 82% for the year ended December 31, 2020 and 2019, respectively.

Gain on legal settlements

The gain on legal settlement of \$5.8 million for the year ended December 31, 2020, was driven by the resolution of legal actions for damages resulting from the early termination of agreements by former customers.

Gain (loss) from sales of digital currency assets

		Year Ended December 31,) Period
	Decem			nge
				%
	2020	2019	Change	Change
	(in thou	ısands)	(in thou	sands)
Gain (loss) on sales of digital currency assets	\$69	\$387	\$ (318)	(82%)
Percentage of total revenue	0%	1%		

Gain (loss) from sales of digital currency assets decreased by \$0.4 million or 82%, to \$0.07 million for the year ended December 31, 2020 from \$0.4 million for the year ended December 31, 2019.

Impairment of digital currency assets

	Year Ended December 31.		Period to Period Change	
	Decem	bei 51,	\$	%
	2020	2019	Change	Change
	(in thou	isands)	(in thou	sands)
Impairment of digital currency assets	\$ (4)	\$419	\$ (423)	(101%)
Percentage of total revenue	0%	1%		

Impairment of digital currency assets decreased by \$(0.4) million, to a loss of \$(4.0) thousand for the year ended December 31, 2020 from \$0.4 million for year ended December 31, 2019.

Operating Expenses

Research and development

		Year Ended December 31,		Period ige
				%
	2020	2019	Change	<u>Change</u>
	(in thou	(in thousands)		sands)
Research and development	\$5,271	\$5,480	\$ (209)	(4%)
Percentage of total revenue	9%	9%		

Total research and development expense decreased by \$0.2 million or 4%, to \$5.3 million for the year ended December 31, 2020 from \$5.5 million for the year ended December 31, 2019. The decrease was driven by a reduction in professional fees related to research projects and staff augmentation, lower travel and entertainment expenses due to the reduction in travel resulting from the COVID-19 pandemic and a non-recurring loss on fixed assets in 2019, offset by an increase in personnel and related costs due to replacing contract staff with full time employees.

Sales and marketing

	Year E Decemb		Period to Period Change	
	·	<u> </u>		%
	2020	2019	\$ Change	Change
	(in thou	sands)	(in thous	ands)
Sales and marketing	\$1,771	\$2,833	\$(1,062)	(37%)
Percentage of total revenue	3%	5%		

Total sales and marketing expense decreased by \$1.0 million or 37%, to \$1.8 million for the year ended December 31, 2020 from \$2.8 million for the year ended December 31, 2019. The decrease in sales and marketing expense was driven by a decrease in headcount and related personnel expenses of \$0.7 million and lower travel and conference expense of \$0.3 million as a result of COVID-19.

General and administrative

	Year E		Period to Period Change	
	<u></u>			%
		2019	Change	Change
	(in thous	sands)	(in thou	sands)
General and administrative	\$14,556	\$14,707	\$ (151)	(1%)
Percentage of total revenue	24%	25%		

Total general and administrative expense decreased by \$0.1 million or 1%, to \$14.6 million for the year ended December 31, 2020 from \$14.7 million for the year ended December 31, 2019. The decrease was primarily driven by a \$1.5 million reduction in gain (loss) and impairments of fixed assets, lower personnel costs for recruiting, relocation and paid time off of \$0.5 million, reduced travel and entertainment expenses of \$0.5 million due to less travel resulting from the COVID-19 pandemic, offset by an increase in intangible asset amortization of \$0.7 million, higher insurance costs of \$0.4 million, increases in stock-based compensation of \$0.2 million, higher indirect taxes of \$0.3 million, higher professional fees of \$0.1 million and other miscellaneous increases of \$0.7 million.

Total non-operating income (expense), net

	Year Ei Decemb		Period to Pe	
	2020 (in thous	2020 2019 (in thousands)		Change sands)
Non-operating income (expense), net:	,		,	
Loss on debt extinguishment	\$(1,333)	_	\$(1,333)	— %
Interest expense, net	(4,436)	(235)	(4,201)	1788%
Other non-operating expenses, net	(110)		(110)	
Total non-operating income (expense), net	\$(5,879)	\$(235)	\$(5,644)	2402%

Total non-operating income (expense), net increased by \$(5.6) million, to \$(5.9) million for the year ended December 31, 2020 from \$(0.2) million for the year ended December 31, 2019. The increase was primarily driven the interest expense related to the issuance of a \$21.0 million senior secured loan issued in May of 2020 and the loss on debt extinguishment of \$1.3 million associated with related party promissory notes in the amount of \$4.0 million which were refinanced in May 2020. The increase in interest expense, net was driven by the execution of several funding agreements, including a senior secured credit facility and several equipment financing agreements, the proceeds of which were used to acquire blockchain computing equipment and to fund the build out of our hosting and self-mining facilities.

Income tax expense (benefit)

Income tax expense (benefit) consists of U.S. federal, state and local income taxes, if any. For the years ended December 31, 2020 and 2019, our income tax expense was zero. A full valuation allowance is recorded against our net deferred tax assets, which are composed primarily of federal and state net operating loss carryforwards, stock-based compensation and non-goodwill intangibles. We evaluate our ability to recognize our deferred tax assets by considering all positive and negative evidence available as proscribed by the FASB under its general principles of ASC 740, Income Taxes.

Segment Total Revenue and Gross Profit

The following table presents total revenue and gross profit by reportable segment for the periods presented:

		For the Years Ended December 31,		Period ge
	2020	2019 usands)	\$ Change (in thousa	% Change
Equipment Sales and Hosting Segment	(iii tiioi	isanus)	(III tilousa	anus)
Hosting revenue from customers	\$41,598	\$53,876	\$(12,278)	(23%)
Equipment sales to customers	12,595	_	12,595	n.m.
Total revenue	54,193	53,876	317	1%
Cost of revenue	47,951	43,005	4,946	12%
Gross profit	\$ 6,242	\$10,871	\$ (4,629)	(43%)
Mining Segment	<u> </u>			
Digital asset mining income	\$ 6,127	\$ 5,647	\$ 480	8%
Total revenue	6,127	5,647	480	8%
Cost of revenue	2,977	5,991	(3,014)	(50%)
Gross profit	\$ 3,150	\$ (344)	\$ 3,494	n.m
Consolidated total revenue	\$60,320	\$59,523	\$ 797	1%
Consolidated cost of revenue	\$50,928	\$48,996	\$ 1,932	4%
Consolidated gross profit	\$ 9,392	\$10,527	\$ (1,135)	(11%)

For the years ended December 31, 2020 and 2019, one customer accounted for approximately 27% and 82% of the Hosting and Equipment Sales' segment total revenue, respectively. This customer exercised its right to terminate its hosting contracts with the Company effective April 2020.

A reconciliation of the reportable segment gross profit to income (loss) before income taxes included in our consolidated statements of operations for the years ended December 31, 2020 and 2019 is as follows:

	For the Years Ended December 31,		Period to Period Change	
	2020	2019	\$ Change	% Change
	(in thou		(in thousands)	
Reportable segment gross profit	\$ 9,392	\$ 10,527	\$(1,135)	(11%)
Gain on legal settlements	5,814	_	5,814	n.m.
Gain (loss) from sales of digital currency assets	69	387	(318)	(82%)
Impairment of digital currency assets	(4)	419	(423)	(101%)
Operating expense (income):				
Research and development	\$ 5,271	\$ 5,480	\$ (209)	(4%)
Sales and marketing	1,771	2,833	(1,062)	(37%)
General and administrative	14,556	14,707	(151)	(1%)
Total operating expense (income)	21,598	23,020	(1,422)	(6%)
Operating Income (loss)	(6,327)	(11,687)	5,360	(46%)
Other income (expense):				
Loss on debt extinguishment	(1,333)	_	(1,333)	n.m.
Interest expense, net	(4,436)	(235)	(4,201)	1788%
Other non-operating expenses, net	(110)		(110)	n.m.
Total other income (expense)	(5,879)	(235)	(5,644)	2402%
Income (loss) before income taxes	\$(12,206)	\$(11,922)	\$ (284)	2%

Liquidity and Capital Resources

Sources of liquidity

To date, we have financed our operations primarily through sales of equity securities, debt issuances, equipment financing arrangements and cash generated from operations. We continue to have access to several sources of liquidity to supplement cash flow from operations. We have access to private debt and equity capital markets, secured borrowing, equipment financing and digital asset-based financing. In the near term, we expect to continue to ramp up investing activities as we build out our facilities and grow our company. Some of our mining equipment purchase contracts include variable pricing provisions that offset some of the variability of cash flow from operations associated with fluctuations in the price of bitcoin. Proceeds from the SPAC transaction fortified our balance sheet and supported our continued investing activity. Gross cash generated by the SPAC transaction held in trust was \$345 million and net proceeds before any redemptions but after fees were \$310 million. In connection with the Business Combination, holders of 12,347,077 of the 34,500,000 then-outstanding shares of Class A common stock of XPDI exercised their right to redeem their shares for cash at a redemption price of approximately \$10.00 per share, for an aggregate redemption amount of \$123.5 million.

Cash, cash equivalents, restricted cash, cash requirements and cash flows

Cash and cash equivalents include all cash balances and highly liquid investments with original maturities of Nine months or less from the date of acquisition.

	Nine Montl			As of the Year riod to Period Ended Change December 31,		Period to Period Change		
	2021	2020 (in thou	\$ Change	% Change	2020	2019 (in thousands	\$ <u>Change</u>	% <u>Change</u>
Cash and cash equivalents	\$147,906	\$6,573	\$141,333	n.m.	\$8,671	\$6,657	\$2,014	30%
Restricted cash	12,101	125	11,976	n.m.	50	250	(200)	(80%)
Cash, cash equiv. and restricted cash	\$160,007	\$6,698			\$8,721	\$6,907		

As of September 30, 2021, cash and cash equivalents consisted of bank deposits and money market funds. As of September 30, 2020, cash and cash equivalents consisted solely of cash in bank deposits. As of September 30, 2021 and 2020, restricted cash of \$12.1 million and \$0.1 million, respectively, consisted of cash held in escrow to pay for construction and development activities.

As of December 31, 2020 and 2019, cash and cash equivalents consisted of cash in bank deposits. As of December 31, 2020 and 2019, restricted cash of \$0.05 million and \$0.25 million, respectively, consisted of cash deposited as collateral to secure our corporate card program.

The following table summarizes our cash and cash equivalents, restricted cash and cash flows for the periods indicated.

As of and for the Nine Months Ended September 30,		As of and for the Year Ended December 31,		
2021	2020	2020	2019	
(in thou	(in thousands)		(in thousands)	
\$ 160,007	\$ 6,698	\$ 8,721	\$ 6,907	
(166,474)	(18,462)	(23,765)	(832)	
(115,524)	(9,228)	(15,144)	(37,360)	
433,284	27,481	40,723	28,694	
8,721	6,907	6,907	16,405	
\$ 160,007	\$ 6,698	\$ 8,721	\$ 6,907	
	Months Ended 2021 (in thou \$ 160,007 (166,474) (115,524) 433,284 8,721	Months Ended September 30, 2021 2020	Months Ended September 30, Ended Dec 2021 2020 2020 (in thousands) (in thousands) \$ 160,007 \$ 6,698 \$ 8,721 (166,474) (18,462) (23,765) (115,524) (9,228) (15,144) 433,284 27,481 40,723 8,721 6,907 6,907	

Our principal uses of cash in recent periods have been funding our operations and investing in capital expenditures.

Operating Activities

Changes in net cash from operating activities results primarily from cash received from hosting customers and equipment sales and payments for power fees and equipment purchases. Other drivers of the changes in net cash from operating activities include research and development costs, sales and marketing costs and general and administrative expenses (including personnel expenses and fees for professional services) and interest payments on debt.

Net cash used by operating activities was \$166.5 million for the nine months ended September 30, 2021, compared to net cash used by operating activities of \$18.5 million for the nine months ended September 30, 2020.

The decrease in net cash from operating activities for the nine months ended September 30, 2021 compared to the nine months ended September 30, 2020 was primarily due to changes in working capital, which decreased cash from operating activities by \$146.0 million and was primarily due to a \$394.9 million increase in deposits for equipment and a \$34.3 million decrease in accounts payable, partially offset by a \$248.7 million increase in deferred revenue. The decrease in net cash from operating activities was increased by net income, excluding non-cash adjustments, of \$8.2 million for the nine months ended September 30, 2021 compared to the nine months ended September 30, 2020, primarily due to a \$7.4 million increase in cash generated from operating income driven by a \$10.3 million increase in cash inflows from gross profit on equipment sales from customers and related parties and a \$13.6 million increase in cash inflows from gross profit on hosting revenue from customers and related parties. Offsetting the increase in cash generated from operating income was a \$3.7 million increase in cash outflows for interest payments on debt (which included \$10.8 million of debt issuance costs paid for the convertible notes issued in April 2021) and \$8.4 million increase in cash outflows for operating expenses.

Net cash used in operating activities was \$23.8 million for the year ended December 31, 2020, compared to net cash used by operating activities of \$0.8 million for the year ended December 31, 2019. The decrease in net cash from operating activities for the year ended December 31, 2020 compared to the year ended December 31, 2019 was primarily due to changes in working capital, which decreased cash from operating activities by \$27.3 million and was primarily due to a \$54.7 million increase in deposits for equipment, partially offset by a \$30.9 million increase in deferred revenue. The decrease in net cash from operating activities due to changes in working capital was partially offset by an increase in net income, excluding non-cash adjustments, of \$4.4 million for the year ended December 31, 2020 compared to the year ended December 31, 2019, due to a \$7.4 million increase in cash generated from operating income, driven by a \$5.8 million increase in cash inflows from legal settlements and \$1.7 million increase in cash inflows from gross profit (primarily driven by a \$1.6 million increase in cash inflows from gross profit on equipment sales), partially offset by a \$2.9 million increase in cash outflows for general and administrative expenses.

Investing Activities

Our net cash used in investing activities consists of purchases of property, plant and equipment and acquisitions of intangible assets, net of proceeds from sales of property, plant and equipment.

Net cash used in investing activities for the nine months ended September 30, 2021 and the nine months ended September 30, 2020 was \$115.5 million and \$9.2 million, respectively, driven by \$116.1 million and \$8.0 million, respectively, used for the purchase of property, plant and equipment primarily related to the development of hosting facilities and the acquisition of equipment used for generating digital asset mining income.

For the year ended December 31, 2020, net cash used in investing activities was \$15.1 million which included \$13.7 million used for the purchase of property, plant and equipment, primarily related to the development of hosting facilities and the acquisition of equipment used for generating digital asset mining income. In addition, for the year ended December 31, 2020, we used \$1.6 million for the acquisition of intangible and other assets, primarily software intangible assets.

For the year ended December 31, 2019, net cash used in investing activities was \$37.4 million, driven by \$37.4 million used for purchases of property, plant and equipment, primarily related to the development of hosting facilities and the acquisition of equipment used for generating digital asset mining income.

Financing Activities

Net cash provided by financing activities consists of proceeds from stock issuances, issuances of debt, net of debt issuance costs and principal payments on debt, including notes payable and capital leases.

For the nine months ended September 30, 2021, net cash provided by financing activities was \$433.3 million, primarily related to \$475.3 million from the issuance of debt, including the issuance of \$420.9 million of convertible notes, \$10.0 million received January 2021 from a stockholder for the purchase of Bitcoin mining equipment, the issuance of a \$9.0 million tranche of senior secured notes (net of issuance costs) in February 2021, and \$3.8 million, \$13.4 million, \$25.6 million and \$1.0 million of additional loans under a master equipment finance agreement issued in March 2021, May 2021, July 2021 and August 2021, respectively. Offsetting this increase to net cash provided by financing activities for the nine months ended September 30, 2021 was \$42.5 million of principal payments on debt.

For the nine months ended September 30, 2020, net cash provided by financing activities was \$27.5 million, primarily driven by \$30.7 million from the issuance of debt, including \$4.0 million received in March and April 2020 for notes issued to our president and chief executive officer and a member of the Board of Directors, \$2.2 million received in April 2020 for a PPP loan through the SBA, \$19.8 million received in May 2020 for the issuance of a senior secured note and \$5.3 million received in the nine months ended September 30, 2020. For the nine months ended September 30, 2020, we also received \$1.5 million in February and March 2020 from the issuance of Series A Preferred Stock in private placement offerings. Offsetting these increases to net cash provided by financing activities for the nine months ended September 30, 2020 was \$5.8 million of principal payments on debt

For the year ended December 31, 2020, net cash provided by financing activities was \$40.7 million, primarily related to \$45.2 million from the issuance of debt, including \$4.0 million received in March and April 2020 for notes issued to our president and chief executive officer and a member of the Board of Directors, \$2.2 million received in April 2020 for a PPP loan through the SBA, \$19.8 million received in May 2020 for the issuance of a senior secured note, and the remaining \$19.9 million received primarily from equipment financing loans. For the year ended December 31, 2020, we also received \$2.6 million from the issuance of preferred stock, comprising \$1.5 million received in February and March 2020 from the issuance of Series A Preferred Stock in private placement offerings and \$1.1 million received in June 2020 from the issuance of Series B Preferred Stock in a private placement offering. Offsetting these increases to net cash provided by financing activities for the year ended December 31, 2020 was \$7.1 million of principal payments on debt, primarily related to the repayment in May 2020 of the \$4.0 million of notes that had been issued to officers and directors and payments for capital leases.

For the year ended December 31, 2019, net cash provided by financing activities was \$28.7 million, primarily related to \$29.5 million received for the issuance of Series A Preferred Stock in private placement offerings at various times during the year.

Operating and capital expenditure requirements

We believe our existing cash, cash equivalents, together with cash provided by operations and funding from debt or equity issuances, will be sufficient to meet our needs for at least the next 12 months. Our future capital requirements will depend on many factors including our revenue growth rate, the timing and extent of spending to support further sales and marketing and research and development efforts, the timing and extent of additional capital expenditures to invest in the expansion of existing facilities as well as new facilities. We may in the future enter into arrangements to acquire or invest in complementary businesses, services and technologies, including intellectual property rights. We may be required to seek additional equity or debt financing. If additional financing is required from outside sources, we may not be able to raise it on terms acceptable to us or at all. If we are unable to raise additional capital when desired, our business, results of operations and financial condition would be materially and adversely affected.

Commitments and Contractual Obligations

Legal Proceedings—The Company is subject to legal proceedings arising in the ordinary course of business. The Company accrues losses for a legal proceeding when it is probable that a loss has been incurred

and the amount of the loss can be reasonably estimated. However, the uncertainties inherent in legal proceedings make it difficult to reasonably estimate the costs and effects of resolving these matters. Accordingly, actual costs incurred may differ materially from amounts accrued and could materially adversely affect the Company's business, cash flows, results of operations, financial condition and prospects. Unless otherwise indicated, the Company is unable to estimate reasonably possible losses in excess of any amounts accrued. As of September 30, 2021, December 31, 2020 and 2019, there were no loss contingency accruals for legal matters.

Gain on legal settlements—The Company recognized a loss of \$2.6 million during the nine months ended September 30, 2021 related to a settlement with a former customer. The Company recognized a gain of \$5.8 million during the nine months ended September 30, 2020 with respect to the resolution of legal actions for damages resulting from the early termination of agreements by former customers.

Operating Leases—The Company leases its corporate headquarters under an operating lease agreement that expires in 2022. The Company leases other facilities under operating leases that have terms of less than one year. Certain lease agreements provide for rental payments on a graduated basis. The Company recognizes rent expense on a straight-line basis over the lease period. For the nine months ended September 30, 2021 and September 30, 2020, rent expense was \$0.4 and \$0.3 million respectively. Rent expense was \$0.3 million for the years ended December 31, 2020 and 2019, respectively.

The Company's minimum payments under noncancelable operating leases having initial terms in excess of one year are as follows at December 31, 2020:

Year ending December 31,	
2021	\$111
2022-2025	_
Total minimum lease payments	\$111

The lease agreements for certain of the Company's facilities require the Company to maintain cash deposits with the landlords of each facility as security for its payment obligations under the leases.

Capital Leases—The Company has entered into arrangements with various parties to finance the acquisition of computer and networking equipment, electrical infrastructure, and office equipment. These arrangements include options exercisable by the Company at the end of the initial terms to renew, purchase the equipment, or to terminate. These arrangements are classified as capital leases and as of September 30, 2021 and December 31, 2020, the related capital lease obligations were \$4.0 million and \$4.4 million, respectively.

As of December 31, 2020, the future minimum lease payments and present value of the net minimum lease payments under these capital leases are as follows:

Year ending December 31,	
2021	\$2,467
2022	1,990
2023	271
2024	102
2025	46
Total minimum lease payments	4,876
Less: interest	(467)
Present value of net minimum lease payments	\$4,409

The accumulated amortization of property, plant and equipment under capital leases for the years ended December 31, 2020 and 2019, was \$1.8 million and \$0.3 million, respectively.

Purchase obligations—During the nine months ended September 30, 2021, the Company entered into agreements to purchase blockchain mining equipment totaling approximately \$738.3 million (net of discounts of \$27.2 million) of which approximately \$418.5 million was paid for deliveries or as deposits for blockchain mining equipment scheduled to be delivered in 2021 and 2022. As of the date that the financial statements were available to be issued, the aggregate amount of the Company's purchase obligations totaled approximately \$319.9 million, substantially all of which are expected to be settled within one year of the date that the financial statements were available to be issued. The Company has entered into agreements to purchase blockchain mining equipment that are enforceable and legally binding and that specify all significant terms, including fixed or minimum quantities to be purchased, fixed, minimum or variable price provisions and the approximate timing of the transaction. As of December 31, 2020, the aggregate amount of the Company's purchase obligations totaled approximately \$88 million, substantially all of which are expected to be settled within one year of December 31, 2020.

Payment obligations—The Company has payment obligations under financing arrangements, including notes payable and lease obligations. As of December 31, 2020, the total obligations for notes payable principal payments and lease payments was \$43.7 million, of which \$18.6 million is due to be settled within one year of December 31, 2020.

Other Events

Blockcap Acquisition

On July 30, 2021, Legacy Core acquired Blockcap, one of its largest hosting customers. Consideration consisted of the issuance of approximately 72 million shares of Legacy Core common stock, approximately 4 million shares of Legacy Core restricted stock and approximately 5 million options to purchase shares of Legacy Core common stock. On July 1, 2021, prior to Legacy Core's acquisition of Blockcap, Blockcap purchased Radar Relay, Inc. ("RADAR"), an early stage company focused on technology enhancement and development in the digital asset industry. Consideration consisted of the issuance of 5,996,311 Blockcap common shares, subject to post-closing adjustments.

As of the effective time of the Core/Blockcap merger, Mr. Matthew Minnis, a member of Legacy Core's board of directors indirectly owned approximately 6.17% of Blockcap on a fully-diluted basis, and as such received consideration upon the consummation of the Core/Blockcap merger. Prior to the closing of the Core/Blockcap merger, Darin Feinstein, the Co-Chair of Legacy Core's board of directors, beneficially owned greater than 5% of Legacy Core's capital stock and held an approximately 12% equity ownership interest in Blockcap. Prior to the effective time of the Core/Blockcap merger, Mr. Feinstein was neither employed by, nor served on the board of directors of, Legacy Core. The terms of the merger agreement between Legacy Core and Blockcap were the result of arms-length negotiations between Legacy Core, Blockcap and their respective representatives.

The acquisition was accounted for as a business combination using the acquisition method whereby the net assets acquired, and the liabilities assumed were recorded at fair value. The difference between the aggregate consideration relative to the fair value of the identifiable net assets is recorded as goodwill. The Company is still in the process of completing its initial accounting for the business combination, including determining the fair value of assets acquired and liabilities assumed and the allocation of the purchase price to goodwill.

Financing activities

In April 2021, the Company issued secured convertible notes in an initial aggregate principal amount of \$215 million (the "Notes") in a private offering. The Notes bear interest at a rate of 10% per annum, 4% of which is payable in cash and 6% of which is payable-in-kind, and mature on April 19, 2025 (the "Maturity Date"), unless repurchased or converted. The Company used the net proceeds from the issuance of the Notes to fund the purchase of certain equipment and other capital leases, to repay in full all amounts due under the Company's existing credit facility and for general corporate purposes.

The Notes will be convertible into equity of the Company solely upon the first to occur of the following events: (i) a private placement or offering of equity securities of the Company in excess of \$50 million; (ii) an initial public offering of equity securities of the Company (including any transaction involving a special purpose acquisition corporation, a "SPAC"), (iii) the listing by the Company of its common stock on a national securities exchange or any other stock exchange (clauses (i) through (iii), each an "Offering") and (iv) a change of control of the Company (a "Change of Control" and, together with the Offerings, each a "Conversion Event"). From and after an Offering, the Notes may be converted, in whole or in part, into equity securities of the Company at a discount to (i) with respect to an Offering other than a SPAC, the cash price paid per share for such equity securities in such Offering and (ii) with respect to a SPAC, the cash and/or value of the property, rights or securities paid per share of common stock to the holders of common stock of the Company pursuant to such SPAC (the "Applicable Offering Conversion Price"). Upon a Change of Control, and unless otherwise redeemed, the Notes will be converted into common stock of the Company pursuant to such Change of Control (the "Applicable Change of Control Conversion Price"). For each conversion, the discount will be determined at the time of the Conversion Event and will be set initially at 20% for the first year following the issuance of the Notes, increasing by 5% in each subsequent year until the Maturity Date.

Prior to a Conversion Event, the Notes will not be redeemable by the Company except to the extent noteholders do not consent to increasing the amount of secured debt permitted to be incurred by the Company under the note purchase agreement, in which case the Company may elect to redeem the notes of non-consenting noteholders by repaying them in full and issuing a warrant entitling such non-consenting noteholders to acquire, upon a Conversion Event, the number of shares of equity securities of the Company equal to the (x) the principal amount of each Note being redeemed divided by (y) the Applicable Offering Conversion Price (if the Conversion Event is an Offering) or the Applicable Change of Control). The per share exercise price for such warrants shall be the Applicable Offering Conversion Price (if the Conversion Event is an Offering) or the Applicable Change of Control). Unless otherwise converted, (i) on or after an Offering, the Notes may be redeemed, in whole or in part, in an amount equal the outstanding principal amount of the notes (together with all accrued unpaid interest thereon) multiplied by 200% (the "Repayment Amount"), (ii) upon a Change of Control the Notes will be redeemed in full in an amount equal to the Repayment Amount.

The Notes are guaranteed by each wholly-owned domestic material subsidiary of the Company (the "Guarantors") and are secured by substantially all of the assets of the Company and the Guarantors (subject to certain exceptions including assets financed with purchase money financing and real property). The Company is subject to a financial reporting covenant and, prior to a Conversion Event, certain negative covenants restricting the Company's ability (i) to incur secured debt (other than debt to finance the acquisition of any fixed or capital assets, including real property assets) in excess of \$265 million, (ii) to make restricted payments (including dividends and repurchases of capital stock) and (iii) to dispose of material assets (with certain exceptions). The Notes include events of default that are customary for a transaction of this type and include, among others things, payment default, covenant default, bankruptcy default of the Company or any of its material subsidiaries and cross-acceleration on other material debt.

In May 2021, the Company received \$13 million of additional loans under the master equipment finance agreement with NYDIG to finance the Company's acquisition of blockchain computing equipment that bear an interest rate of 14.25% and have a term of 24 months from issuance.

In July 2021, the Company paid off the \$2.2 million PPP loan in full plus accrued interest.

In August through November 2021, the Company entered into unsecured convertible note agreements and issued \$300 million of additional convertible notes (including \$1 million of convertible notes issued to an entity

affiliated with XMS Capital Partners, LLC) subject to the same terms and conditions of the April 2021 notes, except that the August through November 2021 notes are unsecured until an initial public offering or SPAC merger and then they are secured pari passu with the Secured Convertible Notes.

In August 2021, the Company entered into equipment finance leases of \$1 million.

In 2021 through the date that the financial statements were available to be issued, the Company entered into arrangements with various parties to finance the acquisition of computer and networking equipment with lease terms between 2 and 4 years and fixed payments over the non-cancelable lease term of \$1 million.

Resignation of CEO

The Company's President and Chief Executive Officer, Kevin Turner, resigned in May 2021. Mr. Turner's employment agreement provides that he will be entitled to receive 12 months of base salary totaling \$300,000 and, if Mr. Turner elects COBRA coverage, for 12 months he will receive a monthly cash payment equal to the full premium for company-paid continuing health care benefits. As of Mr. Turner's separation date, 9,037,500 RSUs had satisfied the time vested condition but remain subject to the performance condition described in Note 11 of the Company's Consolidated Financial Statements as of and for the Years Ended December 31, 2020 and 2019 which must be satisfied within three years of Mr. Turner's separation date.

Related party transactions

In June through July 2021, the Company entered various promissory notes with Blockcap, a related party entity that was managed by individuals that are directors and executives of the Company. The Company had existing contracts for equipment sales and hosting services with Blockcap prior to the Company acquiring Blockcap on July 30, 2021 as described above. The promissory notes deferred \$32.7 million of amounts originally due in June through July 2021 from Blockcap contracts until August 2021, of which \$3 million was deferred subsequent to September 30, 2021. The promissory notes relate primarily to scheduled deposits for equipment to be deployed subsequent to September 30, 2021. For more information on related party transactions, see the section titled "Certain Relationships and Related Party Transactions" in this prospectus.

Employee RSU grants

In 2021 through the date that the financial statements were available to be issued, the Company granted 15.9 million RSUs to various employees of the Company, of which 13.7 million were granted subsequent to September 30, 2021. In October 2021, the Company granted 9.8 million RSUs and 13.0 million stock options to various employees and directors of the Company, including 5 million stock options granted to each of Mr. Levitt and Mr. Feinstein.

Letter of intent to develop hosting facility

In March 2021, the Company signed a letter of intent with Minnkota Power Cooperative to enter into leases and purchase property to develop a hosting facility in Grand Forks, North Dakota as well as enter into a power supply purchase agreement to purchase 100 megawatts of power supply once construction of the hosting facility is complete. Substantially all of the payments for the intended leases would be for a five-year to thirty-year term (comprising an initial five-year term with five five-year options to renew) with purchase options exercisable at any time for approximately \$6 million less any rent paid to date and subject to certain other adjustments. In addition, in 2021 through the date that the financial statements were available to be issued the Company made deposits of \$5.2 million towards the cost of construction and purchasing infrastructure for the development of the hosting facility that is refundable to the Company under certain conditions if the proposed transactions are not consummated on or before October 2021.

Purchase obligations

In 2021 through the date that the financial statements were available to be issued, the Company entered into agreements to purchase blockchain mining equipment totaling approximately \$738 million (net of discounts of \$27 million, all of which were entered into during the Nine months ended September 30, 2021) of which approximately \$419 million was paid for deliveries or as deposits for blockchain mining equipment scheduled to be delivered in 2021 and 2022. As of the date that the financial statements were available to be issued, the aggregate amount of the Company's purchase obligations totaled approximately \$320 million, substantially all of which are expected to be settled within one year of the date that the financial statements were available to be issued.

Foreign Currency and Exchange Risk

The vast majority of our cash generated from revenue are denominated in U.S. dollars, with a small amount denominated in foreign currencies.

Critical Accounting Policies and Estimates

Critical accounting policies and estimates are those accounting policies and estimates that are both the most important to the portrayal of our net assets and results of operations and require the most difficult, subjective or complex judgments, often as a result of the need to make estimates about the effect of matters that are inherently uncertain. These estimates are developed based on historical experience and various other assumptions that we believe to be reasonable under the circumstances. Critical accounting estimates are accounting estimates where the nature of the estimates are material due to the levels of subjectivity and judgment necessary to account for highly uncertain matters or the susceptibility of such matters to change and the impact of the estimates on financial condition or operating performance is material.

The critical accounting estimates, assumptions and judgments that we believe have the most significant impact on our consolidated financial statements are described below.

Revenue Recognition

Hosting and Equipment Sales

We may enter into contracts with more than one performance obligation. For example, we regularly enter into contracts that include both hosting services, for which revenue is recognized as services are performed on a consumption basis, and sales of computer equipment to those same customers, for which revenue is recognized at the point in time when control of the equipment is transferred to the customer (typically at the start of the contract period). For these contracts, revenue is recognized based on the relative standalone selling price of each performance obligation in the contract. The determination of the standalone selling price for contracts that involve more than one performance obligation can have a significant impact on the timing of revenue recognition and may require significant judgment, such as when the selling price of a good or service is not readily observable.

Stock-Based Compensation

The Black-Scholes assumptions used in evaluating our awards are as follows:

	Year Ended D	ecember 31,
	2020	2019
Expected term (years)	10.0	5.43-5.57
Expected volatility	36.26%	30.59%
Risk-free interest rate	0.70%	1.65%
Expected dividend yield	0.0%	0.0%

We will continue to use judgment in evaluating the assumptions related to our stock-based compensation on a prospective basis. As we continue to accumulate additional data related to our common stock, we may refine our estimation process, which could materially impact our future stock-based compensation expense.

In addition, for awards with performance conditions, primarily restricted stock unit awards, we recognize the estimated fair value of the awards based on management's judgment as to whether or not it is probable that the performance conditions will be achieved.

Common Stock Valuations

In valuing the fair value of our common stock, we use the most observable inputs available. We use the market approach, which estimates the value of our business by applying valuation multiples derived from the observed valuation multiples of comparable public companies to our expected financial results.

When observable inputs are not available, we may use the income approach. This approach typically projects cash flows for the forecast period and uses the perpetuity growth method to calculate terminal values. These cash flows and terminal values are then discounted using an appropriate discount rate. Projections of cash flows are based on management's earnings forecasts.

Applying these valuation and allocation approaches involves the use of estimates, judgments and assumptions that are highly complex and subjective, such as those regarding the Company's expected future revenue, expenses and cash flows, as well as discount rates, valuation multiples, the selection of comparable public companies and the probability of future events. Changes in any or all of these estimates and assumptions, or the relationships between these assumptions, impact the Company's valuation as of each valuation date and may have a material impact on the valuation of the Company's common stock and common stock warrants issued with the Company's debt and equity instruments.

Business Combinations and Goodwill

The total purchase price of any of our acquisitions is allocated to the tangible and intangible assets acquired and the liabilities assumed based on their estimated fair values as of the acquisition date. The excess of the purchase price over those fair values is recorded as goodwill. The fair value assigned to the tangible and intangible assets acquired and liabilities assumed are based on estimates and assumptions around the valuation of our common stock at the time of the acquisition.

We do not amortize goodwill, but tests it for impairment annually as of October 31, or more frequently if events or changes in circumstances indicate that the carrying value of goodwill may not be recoverable. We have the option to first assess qualitative factors to determine whether it is more likely than not that the fair value of its reporting units are less than their carrying amounts as a basis for determining whether it is necessary to perform the quantitative goodwill impairment test. If we determine that it is more likely than not that the fair value of a reporting unit is less than its carrying amount, or chooses not to perform a qualitative assessment, then the quantitative goodwill impairment test will be performed. The quantitative test compares the fair value of the reporting unit with its carrying amount. If the carrying amount exceeds its fair value, the excess of the carrying amount over the fair value is recognized as an impairment loss, and the resulting measurement of goodwill becomes its new cost basis. As of December 31, 2020 and 2019, the carrying amount of Goodwill was \$58.2 million. There were no accumulated impairment losses as of January 1, 2019, and no impairment losses were recorded for the years ended December 31, 2020 and 2019.

Long-Lived Assets

We test long-lived assets for recoverability whenever events or changes in circumstances have occurred that may affect the recoverability or the estimated useful lives of long-lived assets. Long-lived assets include

intangible assets subject to amortization. A long-lived asset may be impaired when the estimated future undiscounted cash flows are less than the carrying amount of the asset. If that comparison indicates that the asset's carrying value may not be recoverable, the impairment is measured based on the difference between the carrying amount and the estimated fair value of the asset. This evaluation is performed at the lowest level for which separately identifiable cash flows exist. Long-lived assets to be disposed of are reported at the lower of the carrying amount or estimated fair value less costs to sell.

In April 2019, management approved a plan to actively market and sell certain digital currency mining equipment. In April 2019, we recognized an impairment loss of \$0.9 million representing the excess of the equipment's carrying value over its estimated fair value less costs to sell, based on a quotation for the salvage value of the equipment. We completed the sale in October 2019. For the years ended December 31, 2020 and 2019, we recognized losses of \$2 thousand and \$0.6 million, respectively, on sales of property, plant and equipment, primarily digital currency mining equipment. We have included these amounts for impairments of long-lived assets and losses on disposals of property, plant and equipment within general and administrative expenses in our Consolidated Statements of Operations and Comprehensive Loss.

Digital Currency Assets

Our digital currency assets are accounted for as intangible assets with indefinite useful lives. We initially recognize digital currency assets that are received as digital asset mining income based on the fair value of the digital currency assets. Digital currency assets that are purchased in an exchange of one digital currency asset for another digital currency asset are recognized at the fair value of the asset received. The Company recognizes realized gains or losses when digital currency assets are sold in an exchange for other digital currency assets or for cash using a first-in first-out method of accounting. For the years ended December 31, 2020 and 2019, we recognized net gains of \$0.1 million and \$0.4 million, respectively, on sales of digital currency assets. We classified digital currency assets within current assets because we expected to sell the assets shortly after acquisition. We classify cash flows from digital currency assets within cash flows from operating activities.

Foreign Currency and Exchange Risk

Substantially all revenue and operating expenses are denominated in U.S. dollars.

Recent Accounting Pronouncements

For a discussion of new accounting standards relevant to our business, refer to Note 3 to the consolidated financial statements.

Emerging Growth Company and a Smaller Reporting Company Status

We are an "emerging growth company" as defined in the Jumpstart Our Business Startups Act of 2012 (the "JOBS Act"). We may take advantage of certain exemptions from various public company reporting requirements, including not being required to have our internal control over financial reporting audited by our independent registered public accounting firm under Section 404 of the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), reduced disclosure obligations regarding executive compensation in our periodic reports and proxy statements and exemptions from the requirements of holding a non-binding advisory vote on executive compensation and any golden parachute payments. We may take advantage of these exemptions for up to five years or until we are no longer an emerging growth company, whichever is earlier. In addition, the JOBS Act provides that an "emerging growth company" can delay adopting new or revised accounting standards until those standards apply to private companies. We have elected to use the extended transition period under the JOBS Act. Accordingly, our financial statements may not be comparable to the financial statements of public companies that comply with such new or revised accounting standards.

If we become a public company through this transaction, we expect to be an emerging growth company until the earliest of: (1) December 31, 2026; (2) the last day of the first fiscal year in which our annual gross revenue is \$1.07 billion or more; (3) the date on which we have, during the previous rolling three-year period, issued more than \$1.0 billion in non-convertible debt securities; and (4) the last day of the fiscal year in which the market value of our common stock held by non-affiliates exceeded \$700 million as of September 30 of such fiscal year.

We are also a "smaller reporting company" as defined in the Securities Exchange Act of 1934, as amended. We may continue to be a smaller reporting company even after we are no longer an emerging growth company. We may take advantage of certain of the scaled disclosures available to smaller reporting companies and will be able to take advantage of these scaled disclosures for so long as (i) the market value of our voting and non-voting common stock held by non-affiliates is less than \$250 million measured on the last business day of our second fiscal quarter or (ii) our annual revenue is less than \$100 million during the most recently completed fiscal year and the market value of our voting and non-voting common stock held by non-affiliates is less than \$700 million measured on the last business day of our second fiscal quarter. Specifically, as a smaller reporting company, we may choose to present only the two most recent fiscal years of audited financial statements in our Annual Report on Form 10-K and have reduced disclosure obligations regarding executive compensation, and, similar to emerging growth companies, if we are a smaller reporting company that qualifies as a non-accelerated filer, we would not be required to obtain an attestation report on internal control over financial reporting issued by our independent registered public accounting firm.

BUSINESS

Overview

Core is a best-in-class large-scale operator of dedicated, purpose-built facilities for digital asset mining and a premier provider of blockchain infrastructure, software solutions and services. We mine digital assets for our own account and provide hosting services for other large-scale miners. We are one of the largest blockchain infrastructure, hosting provider and digital asset mining companies in North America, with approximately 497MW of operating electric power as of January 31, 2022. We mine Bitcoin, Ethereum and other digital assets for third-party hosting customers and for our own account at our six fully operational data centers in North Carolina (2), Georgia (2), North Dakota (1) and Kentucky (1). In addition, in October 2021, we announced the entry of an agreement with the City of Denton, Texas and an affiliate of Tenaska Energy, Inc. to develop our seventh facility, a blockchain data center in Denton, Texas, which is expected to have 300MW of power capacity. In February 2022, the Muskogee City-County Port Authority announced an agreement with us to develop a 500MW data center at the Port of Muskogee John T. Griffin Industrial Park. We began digital asset mining in 2018 and in 2020 became one of the largest North American providers of hosting services for third-party mining customers. Historically, we derived almost all of our revenue from third-party hosting fees and the resale of digital asset mining machines.

In July 2021 we completed the acquisition of Blockcap, one of our largest hosting customers. Blockcap is a blockchain technology company with industrial scale digital asset mining operations. Blockcap's primary historical business was the mining of digital asset coins and tokens, primarily Bitcoin and, to a lesser extent, Siacoin and Ethereum. At the time of the Core/Blockcap merger, Blockcap claimed to be the largest independent cryptocurrency mining operator in North America. While Blockcap sold or exchanged the digital assets it mined to fund its growth strategies or for general corporate purposes from time to time, it generally retained its digital assets as investments in anticipation of continued adoption of digital assets as a "store of value" and a more accessible and efficient medium of exchange than traditional fiat currencies. Prior to its acquisition by Legacy Core, Blockcap purchased Radar Relay, Inc. ("RADAR") on July 1, 2021. An early stage company, RADAR is focused on technology enhancement and development in the digital asset industry. The estimated consideration for the acquisition \$65.0 million consisted of the issuance of 5,996,311 Blockcap common shares, subject to post-closing adjustments. The acquisition of Blockcap significantly expands our self-mining operations and increases the number of miners we own. We intend to utilize RADAR's business assets and the technical expertise of its principals in enhancing our existing blockchain mining technology and software and in further strengthening our leadership position and value creation potential through the development of decentralized finance (DeFi) products and services.

Our hosting business provides a full suite of services to digital asset mining customers. We provide deployment, monitoring, troubleshooting, optimization and maintenance of our customer's digital asset mining equipment and provide necessary electrical power and repair and other infrastructure services necessary to operate, maintain and efficiently mine digital assets.

Historically, we operated in two segments: "mining" consisting of digital asset mining for our own account, and "hosting, Artificial Intelligence and other" consisting of our blockchain infrastructure and third-party hosting business, and our AI software-as-a-service offerings and platforms. Going forward, we anticipate that our mining and hosting operations will comprise all or substantially all of our business activities, though we continue to evaluate additional infrastructure and blockchain service offerings, including decentralized finance (DeFi).

Our business strategy is to continue to grow our self-mining operations by significantly increasing the number of miners dedicated to producing digital assets for our own account, and to continue to develop and grow the infrastructure and facilities necessary to house and support our robust third-party hosting business as well as our self-mining operation. Following the Blockcap acquisition, we significantly expanded our self-mining operation and consequently reevaluated our digital asset investment policy. Moving forward, we intend to adopt

an investment policy pursuant to which an investment committee comprised of corporate officers use common risk management techniques to manage our assets in light of specified liquidity criteria. Liquidity will be maintained through management of a portfolio of money market instruments, obligations of the U.S. government, bank deposits, commercial paper, and certain digital asset currencies and digital asset instruments, each of which must satisfy certain risk criteria. The investment committee will retain the discretion to manage these approved investment instruments, including digital asset currencies and instruments, in accordance with the investment policy, which may involve opportunistic sales or conversions of digital asset currencies and instruments in light of market and other conditions. We may also explore adjacent lines of businesses that leverage our mining expertise and bitcoin assets.

Blockchain and Digital Assets

Blockchains are decentralized digital ledgers that record and enable secure peer-to-peer transactions without third-party intermediaries. Blockchains enable the existence of digital assets by allowing participants to confirm transactions without the need for a central certifying authority. When a participant requests a transaction, a peer-to-peer computer network consisting of nodes validates the transaction and the user's status using known algorithms. After the transaction is verified, it is combined with other transactions to create a new block of data for the ledger. The new block is added to the existing blockchain in a way that is permanent and unalterable, thereby completing the transaction. As each new block refers back to and "connects" with the immediately prior solved block associated with it, the addition of a new block adds to the blockchain in a manner similar to a new link being added to a chain.

Digital assets (also known as cryptocurrency) are a medium of exchange that uses encryption techniques to control the creation of units and to verify the transfer of funds. Consumers use digital assets because they offer lower cost and faster peer-to-peer payment options without the need to provide personal details. Every single transaction, and the ownership of every single digital asset in circulation, is recorded in the blockchain, which effectively contains a record of all account balances. Miners use powerful computers that tally digital asset transactions to operate the blockchain. These miners update stored records each time a transaction is made and ensure the authenticity of information. The miners receive a transaction fee for their service in the form of a portion of the new digital "coins" that are issued. Each account on the blockchain is identified solely by its unique public key, which renders it effectively anonymous, and is secured with its associated private key, which is kept secret, like a password. The combination of private and public cryptographic keys constitutes a secure digital identity in the form of a digital signature, providing strong control of ownership.

No single entity owns or operates the network. The infrastructure is collectively maintained by a decentralized public user base. Because the network operates in a decentralized manner, it does not rely on governmental authorities or financial institutions to create, transmit or determine the value of digital assets. Rather, value is determined by market factors, supply and demand for the units, with prices being set in transfers by mutual agreement or through barter among transacting parties, as well as by the number of merchants that may accept the digital asset. As digital asset transactions can be broadcast to the digital asset network by any user's blockchain and digital assets can be transferred without the involvement of intermediaries or third parties, there are currently little to no transaction costs in direct peer-to-peer transactions. Units of digital assets can be converted to fiat currencies, such as the U.S. dollar, at rates determined on various exchanges. Digital asset prices are quoted on various exchanges and demonstrate extreme volatility.

Digital assets have been viewed by the user community to offer several advantages over traditional (also known as 'fiat') currency, including:

- Acting as a fraud deterrent, as digital assets recorded on a blockchain are virtually impossible to counterfeit, reverse, or modify;
- Immediate settlement;
- Elimination of counterparty risk;

- No requirement for a trusted intermediary;
- Lower transaction fees;
- Identity theft prevention;
- Universal accessibility;
- Transaction verification and confirmation processes that prevent double spending;
- · Decentralized transaction processing at any time of day without any central authority (governments or financial institutions); and
- Universal value free from currency exchange rates.

The market for digital assets has been growing exponentially. In 2017, there were an estimated 2.7 million users of digital assets, whereas today estimated users are 100 million. Bitcoin's daily exchange volume has grown from \$92 million in January 2017 to more than \$50 billion in May 2021. The initial exchange rate recorded on October 5, 2009 was one Bitcoin equaled \$0.000764. Bitcoin remains the leading digital asset in terms of market capitalization, which, based on data sourced from coinmarketcap.com, exceeded \$823.6 billion as of September 30, 2021. As of November 1, 2021, based on data sourced from coinbase.com, the trading price of one Bitcoin was \$60,954.50.

Digital Asset Mining

Specialized computers, or "miners," power and secure blockchains by solving complex cryptographic algorithms to validate transactions on specific digital asset networks. In order to add blocks to the blockchain, a miner must map an input data set consisting of the existing blockchain, plus a block of the most recent digital asset transactions and an arbitrary number called a "nonce," to an output data set of a predetermined length using the SHA256 cryptographic hash algorithm. Solving these algorithms is also known as "solving or completing a block." Solving a block results in a reward of digital assets, such as bitcoin, in a process known as "mining." These rewards of digital assets currently can be sold profitably when the sale price of the digital asset exceeds the cost of "mining," which generally consists of the cost of mining hardware, the cost of the electrical power to operate the machine, and other facility costs to house and operate the equipment.

Mining processing power is generally referred to as "hashing power." A "hash" is the computation run by mining hardware in support of the blockchain. A miner's "hash rate" refers to the rate at which it is capable of solving such computations per second. Miners with higher rated hash rate when operating at maximum efficiency have a higher chance of completing a block in the blockchain and receiving a digital asset reward. Currently, the likelihood that an individual mining participant acting alone will solve a block and be awarded a digital asset is extremely low. As a result, to maximize the opportunities to receive a reward, most large-scale miners have joined with other miners in "mining pools" where the computing power of each pool participant is coordinated to complete the block on the blockchain and mining rewards are distributed to participants in accordance with the rules of the mining pool. Fees payable to the operator of the pool vary but are typically as much as 2% of the reward earned and are deducted from the amounts earned by each pool participant. Mining pools are subject to various risks including connection issues, outages and other disruptions which can impact the quantity of digital assets earned by participants.

Mathematically Controlled Supply

The method for creating new bitcoin is mathematically controlled in a manner such that the supply of bitcoin grows at a limited rate based on a pre-determined schedule. The number of bitcoin awarded for solving a new block is automatically halved every 210,000 blocks. This means every block up to and including block 210,000 produced a reward of 50 bitcoin, while blocks beginning with 210,001 produced a reward of 25 bitcoin. Since blocks are mined on average every 10 minutes, 144 blocks are mined per day on average. At 144 blocks

per day, 210,000 blocks take four years to mine on average. The current fixed reward for solving a new block is 6.25 bitcoin per block and it is anticipated that the reward will decrease by half to become 3.125 bitcoin per block in early 2024, according to estimates of the rate of block solution calculated by BitcoinClock.com. This deliberately controlled rate of bitcoin creation means that the number of bitcoin in existence will never exceed 21 million and that bitcoin cannot be devalued through excessive production unless the Bitcoin network's source code (and the underlying protocol for bitcoin issuance) is altered. We monitor the blockchain network and, as of August 1, 2021 based on the information we collected from our network access, approximately 18.77 million bitcoin have been mined.

Our Operations and Solutions

As a large-scale, vertically integrated provider of blockchain solutions, we believe that we are well positioned to serve customers in a rapidly expanding market for digital assets, blockchain solutions, and digital asset mining. We believe that the adoption and mainstream use of bitcoin and the blockchain technology on which it is based has accelerated the demand for bitcoin and other digital currencies.

As one of the largest blockchain hosting providers in North America, we focus on clients with large-scale deployments and provide power, racks, proprietary thermodynamic management (heat dissipation and airflow management), redundant connectivity, 24/7 security as well as our proprietary software platforms, MinderTM and MinderOSTM, which provide infrastructure management and custom firmware that are designed to increase performance and energy efficiency. Our blockchain business is one of the only large-scale, vertically integrated digital asset mining and blockchain infrastructure and hosting solutions business in North America. Our rapidly growing digital asset mining operation is focused on the generation of digital assets by solving complex cryptographic algorithms to validate transactions on specific digital asset network blockchains, which is commonly referred to as "mining."

Our proprietary data centers in North Carolina, Georgia, Kentucky and North Dakota are purpose-built facilities optimized for the unique requirements of high density blockchain computer servers. These facilities feature access to attractive long-term power contracts at stable rates and high emissions-free content. We currently have six fully operational data centers in North Carolina (2), Georgia (2), North Dakota (1) and Kentucky (1), with approximately 497MW of operating electric power as of January 31, 2022. In addition, we expect to add 300MW of power capacity from the development of our Denton, Texas facility. In February 2022, the Muskogee City-County Port Authority announced an agreement with us to develop a 500MW data center at the Port of Muskogee John T. Griffin Industrial Park. Our existing, completed facilities leverage our specialized construction proficiency by employing high-density, low-cost engineering and power designs. Our proprietary thermodynamic system manages heat and airflow to deliver a best-in-class uptime and ultimately increasing mining rewards to us and our customers. Our North Dakota facility is expected to benefit from our operational mining experience and techniques to maximize operational efficiency.

Since July 2018, we have hosted for ourselves and on behalf of our customers and related parties, miners of varying models, types and manufacturers. We have accumulated significant expertise in the installation, operation, optimization and repair of digital mining equipment. We have expanded our self-mining operation to take advantage of favorable market conditions and leverage our expertise for our own account. We believe that our self-mining activity as a percentage of our overall mining activity will continue to increase by significantly increasing our investment in miners. As of January 31, 2022, we have contracts to procure more than 135,000 miners for our self-mining operation and more than 85,000 miners for our hosting for customers and related parties. We anticipate delivery of these new miners over the next twelve months.

Our digital assets, primarily bitcoin, are mined to custodial wallets at Bittrex and Coinbase, each of which provides us protection through dual authentication security. Access is controlled by dual authentication to separate authentication from account access so that two authorized individuals are required to access our accounts. We monitor transaction and account balances through the Bittrex or Coinbase client portal, as

applicable. In addition, Bittrex sends an email notification to our treasury department whenever there is a successful login to our account, and bank account withdrawals will only be permitted if they are sent to our synced bank account. As of September 30, 2021, nearly all of the digital assets of Legacy Core and Blockcap were stored at Bittrex and Coinbase, respectively.

In July 2021, we acquired Blockcap, one of our largest hosting customers. As a result of the merger we acquired all of the digital asset mining machines owned by Blockcap and hosted by us in our facilities, and have dedicated them to self-mining. As a result, the existing hosting agreement between Legacy Core and Blockcap was terminated. Included in the Blockcap transaction was Blockcap's subsidiary, RADAR.

Suppliers

Mining Equipment

Digital asset mining is dependent on specialized digital asset mining hardware utilizing application-specific integrated circuit ("ASIC") chips to solve blocks on blockchains using the 256-bit secure hashing algorithm. Almost all of these miners are produced outside of the United States, mostly in China and Southeast Asia, by a few manufacturers, the largest of which is Bitmain Technologies, Ltd ("Bitmain").

We have entered into agreements with Bitmain to supply most of the miners we intend to acquire for our mining operations throughout 2021 and 2022. These agreements, like those of other miner manufacturers, generally require significant refundable deposits payable months in advance of delivery and additional advance payments in monthly installments thereafter. These agreements also contain other terms and conditions favorable to the manufacturer. As the market value of digital assets has increased, the demand for the newest, most efficient miners has also increased, leading to scarcity in the supply, and thereby a resulting increase in the price of miners. As a result of the terms of our agreements with Bitmain, we believe we are the only non-Bitmain repair center with Bitmain trained repair technicians in North America, allowing Core to service and repair its Bitmain miners in-house. Our mining business is highly dependent upon digital asset mining equipment suppliers such as Bitmain providing an adequate supply of new generation digital asset mining machines at economical prices to enable profitable mining by us and by third-party customers intending to purchase our hosting and other solutions.

Power Providers and Facility Development

We have an experienced in-house power and facility development team focused on sourcing, evaluating, developing and constructing the facilities where we self-mine digital assets and host for other third-party mining equipment. Historically, we have contracted with large electric utility providers to provide a sufficient supply of electricity to power the mining operations in our facilities. We have fixed and interruptible bi-lateral power supply agreements with electric power suppliers for of each of our facilities. These agreements provide for both firm and interruptible power supply through each provider's transmission system to dedicated substations owned by the power provider, the local utility or us. We own the dedicated substation at our Kentucky facility and our North Dakota facility. Where we own the substation, we contract with the local power utility for repair and maintenance services. We believe that our relationships with our power suppliers are good and that we have sufficient supply to conduct our business operations as presently contemplated.

In 2021, we spent approximately \$43.0 million on renewable and emission free non-renewable energy sources. As of September 30, 2021 over 50% of the power used in our operation was generated from non-carbon emitting sources by local power providers pursuant to long-term power contracts. We assess whether power is generated from non-emitting energy sources by analyzing dispatch reports produced by utilities on a per facility basis, or in the absence of such reports, grid generation mix reports received from our utility providers on aggregated basis. Based on this review of utility dispatch reports, we determine megawatts per hour ("MWH") generated from carbon emitting sources. We then determine total CO₂ emissions per MWH from such sources

(with coal generating an average of a ton of CO₂ per MWH generated and natural gas an average of half-a-ton of CO₂ per MWH generated) to calculate our total Green House Gas ("GHG") emission rate based on EPA guidelines. We leverage our GHG rate to determine the number of green e-certified renewable energy credits ("RECs") we are required to purchase to achieve net carbon neutral status.

By purchasing 530,000 RECs, we believe that we have purchased a number of RECs in excess of that required to achieve net carbon neutral status. In April 2021 we entered into an agreement with a power company that operates wind farms in North Dakota to supply 530,000 Green-e certified RECs for an aggregate purchase price of \$1.6 million. Under the REC program, a REC seller is required to disclose the quantity, type and geographic source of renewable energy underlying each certificate, which helps ensure that credits are not sold more than once. Based on a preliminary assessment, we believe our purchased volume of RECs was sufficient to meet our needs for 2021. Once our final energy consumption is known for 2021 upon receipt of the final GHG emission reports from our suppliers which are expected to be provided to us in the second quarter of 2022, we intend to re-assess our use of carbon-free energy by analyzing updated dispatch and generation mix reports, after which we may decide to purchase additional RECs.

We expect to continue to maintain its net carbon neutral status by increasing our overall use of renewable power and by purchasing RECs when necessary. We are currently actively engaged in negotiations with utilities, municipalities, alternative power brokers and other professionals to source and identify sources of green renewable power across the United States to increase the utilization of green renewable power in our operations. We expect that these activities and negotiations will result in a significant increase in the number of our mining facilities, megawatts under contract, mining capacity and utilization of green renewable energy.

Supplier Agreements

Industrial Power Contract and Related Arrangements with Murphy Electric Power Board

On December 15, 2017, our predecessor BCV 77, LLC ("BCV 77") entered into an Industrial Power Contract (as amended from time to time, the "Murphy IPC") with Murphy Electric Power Board ("Murphy") for the supply of electric power to its plant in Marble, North Carolina. BCV 77 subsequently assigned the Murphy IPC to, and the Murphy IPC was assumed by, Legacy Core on February 19, 2018, in connection with Legacy Core's acquisition of the account with Murphy formerly held by BCV 77 at its plant in Marble, North Carolina. BCV 77, together with its affiliates, is a major stockholder of Legacy Core prior to the merger and will remain a major stockholder following the merger. For more information regarding recent related party transactions among BCV 77 and its affiliated entities, on the one hand, and Legacy Core on the other hand, please see the section titled "Certain Relationships and Related Person Transactions."

Under the Murphy IPC, Murphy agreed to provide Legacy Core with an electric power capacity of up to 10,000 kilowatts (kW), for which Legacy Core agreed to pay Murphy in accordance with the applicable monthly rates and charges as modified or replaced from time to time by agreement between the parties, subject to a minimum monthly bill of not less than \$15,000. Legacy Core currently pays Murphy on average approximately \$450,000 per month for its provision of the power capacity, taking into effect any bill credits available to Legacy Core pursuant to the Murphy IP Product Agreement and the ICA, as defined and summarized below. The Murphy IPC provides for an initial term of five years and automatically renews for an additional five-year term following the initial term and for one-year terms thereafter, unless either party terminates the Murphy IPC by prior written notice no later than three months prior to the expiry of the then-existing term.

In connection with the Murphy IPC, Legacy Core entered into an Interruptible Power Product Agreement (the "Murphy IP Product Agreement") with Murphy, as distributor, and the Tennessee Valley Authority ("TVA"), as power provider, effective September 1, 2018. Under the Murphy IP Product Agreement, Legacy Core agreed to curtail its power demand upon notice from TVA, if TVA determines, in its sole judgment, it is necessary or appropriate to do so to ensure the operation and reliability of its system. In return, TVA provides

credits to Legacy Core's monthly power bill. Legacy Core assumes all risk of loss, injury or damage resulting from any interruption or curtailment of power. The Murphy IP Product Agreement provides for a term of five years, and may be terminated by any party by at least three years' written notice, or by Murphy or TVA upon at least 60 days' notice if Legacy Core does not meet certain eligibility requirements or benchmarks of performance under the Murphy IP Product Agreement. The Murphy IPC cannot be terminated sooner than the date on which the Murphy IP Product Agreement can be terminated using the methods listed above.

In addition, on October 10, 2018, Legacy Core entered into an Investment Credit Agreement (the "ICA") with Murphy and TVA. The ICA provides that TVA will reward Legacy Core in the form of monthly credits on Legacy Core's bill for firm power, if Legacy Core meets certain eligibility requirements, which include maintaining 500kW of power demand, annual capital investment benchmarks in Core's hosting sites, minimum workforce requirements, satisfying TVA's financial viability review and continued compliance with the Murphy IPC. The ICA has a term of five years, unless terminated earlier.

Copies of the Murphy IPC, the Murphy IP Product Agreement and the ICA, along with the amendments thereto, are attached as Exhibits 10.15, 10.16 and 10.17, respectively, to the registration statement of which this prospectus forms a part.

Master Services Agreement and Power Arrangements with Duke Energy Carolinas, LLC

On June 25, 2018, Legacy Core entered into a Master Services Agreement (the "MSA") with Duke Energy Carolinas, LLC ("Duke"). Pursuant to the MSA, Legacy Core engaged Duke to perform certain design, procurement, construction and project management activities and installation services related to the installation of overhead circuits to feed Legacy Core's transformers. Legacy Core paid approximately \$400,000 to Duke for the services. The MSA also governs the terms and conditions of future work orders to be entered into between Legacy Core and Duke. Either party may terminate the MSA upon 30 days prior written notice.

Legacy Core also entered into an Electric Service Agreement (the "ESA") with Duke, effective June 10, 2019, for the supply of electric power to Legacy Core's plant in Marble, North Carolina. The ESA provides for an electrical power capacity of 24,000kW, which can be adjusted based on Legacy Core's consumption, but will not exceed 48,000kW. Legacy Core agreed to pay to Duke in accordance with certain service rate schedules as modified or replaced from time to time by agreement between the parties. Legacy Core currently pays Duke on average approximately \$92,000 per month for its provision of the power capacity. The ESA provides for an initial term of 2.5 years, with a subsequent one-year automatic renewal provision until the ESA is terminated. Either Legacy Core or Duke may terminate the ESA by prior written notice no later than 60 days in advance of the termination of the then-existing term.

Copies of the MSA and the ESA are attached as Exhibits 10.18 and 10.19, respectively, to the registration statement of which this prospectus forms a part.

Electric Service Agreements with Dalton Utilities

On October 11, 2018, Legacy Core, through its wholly-owned subsidiary, American Property Acquisitions VII, LLC, entered into an Amended and Restated Electric Service Agreement with The Board of Water, Light and Sinking Fund Commissioners of the City of Dalton, Georgia (d/b/a Dalton Utilities, "Dalton") for the supply of electric power to each of its hosting sites located at Boring Drive, Dalton, Georgia (the "Boring Drive Site") and Industrial South, Dalton, Georgia (the "Industrial South Site"). The agreement for the Boring Drive Site provides for an electrical power capacity of up to 120,000kW, and the agreement for the Industrial South Site provides for an electrical power capacity of up to 50,000kW. Under each agreement, Legacy Core agreed to pay to Dalton \$0.0364 on a kW per hour basis as modified from time to time, but not to exceed \$0.042 prior to December 31, 2021. Each agreement has an indefinite term, which can be terminated by Legacy Core for convenience by providing 60 days written notice to Dalton.

Copies of the agreements for the Boring Drive Site and the Industrial South Site are attached as Exhibits 10.21 and 10.20, respectively, to the registration statement of which this prospectus forms a part.

Firm Power Contract and Related Arrangements with Tennessee Valley Authority

Effective as of May 1, 2019, Legacy Core entered into a Firm Power Contract, which was subsequently amended in February and April 2020 (as amended, the "FPC"), with TVA for the supply of electric power to Legacy Core's site near Calvert City, Kentucky. Subject to other terms and conditions, the FPC provides for an electrical power capacity of 125,000kW of firm power during "onpeak hours," and an electrical power capacity of 125,000kW of firm power during "offpeak hours." Legacy Core agreed to pay TVA for such supply in accordance with certain service rate schedules as modified or replaced from time to time by agreement between the parties. Legacy Core currently pays TVA on average approximately \$625,000 per month for its provision of the power capacity, taking into effect any bill credits available to Legacy Core pursuant to the TVA IP Product Agreement, as defined and summarized below. The FPC provides for an automatic one-year renewal beginning on May 1, 2021. The FPC may be terminated on or after April 29, 2025 by either Legacy Core or TVA upon at least five years' written notice, but in no event, shall the FPC be terminated sooner than the date on which the TVA IP Product Agreement (as defined below) can be terminated.

In connection with the FPC, Legacy Core entered into an Interruptible Power Product Agreement with TVA (the "TVA IP Product Agreement"), effective May 1, 2020. Under the TVA IP Product Agreement, Legacy Core agreed to curtail its power demand upon notice from TVA, if TVA determines, in its sole judgment, it is necessary or appropriate to do so to ensure the operation and reliability of its system. In return, TVA provides credits to Legacy Core's monthly power bill. Legacy Core assumes all risk of loss, injury or damage resulting from any interruption or curtailment of power. The TVA IP Product Agreement provides for a term of five years, and may be terminated by any party by at least three years' written notice, or by TVA upon at least 60 days' notice if Legacy Core does not meet certain eligibility requirements or benchmarks of performance under the TVA IP Product Agreement. The TVA IP Product Agreement continues in effect during the term of the FPC, unless it is sooner terminated using the methods listed above.

Copies of the FPC and the TVA IP Product Agreement, along with the amendments thereto, are attached as Exhibits 10.22 and 10.23, respectively, to the registration statement of which this prospectus forms a part.

Sales and Purchase Arrangements with Bitmain

Since 2018, Legacy Core has entered into multiple sales and purchase agreements with Bitmaintech Pte. Ltd. and/or its affiliates (collectively, "Bitmain") for the purchase of digital currency miners from Bitmain. Under the terms of such agreements, Bitmain has the right to discontinue the sale of its miners and/or to make changes thereto at any time without prior approval from or notice to Legacy Core. In addition, Bitmain (and/or its licensors, as applicable) retains the intellectual property rights to such miners. Under the terms of the agreements, Legacy Core agreed to indemnify Bitmain from all claims arising and any and all damages, suits, claims, judgments, liabilities, losses, fees, costs or expenses of any kind, including legal fees, whatsoever arising out of or incidental to Bitmain's products pursuant to the agreements. Each agreement will remain in effect for as long as one or more shipping orders are outstanding thereunder, and may be terminated upon either party's uncured material breach or upon insolvency proceedings against Legacy Core. A copy of a form of such agreements with Bitmain is attached as Exhibit 10.24 to the registration statement of which this prospectus forms a part.

Our Go-to-Market Model

We sell our services through direct sales teams with leads generated from technology partners, customer referrals, and through our corporate marketing efforts. Our go-to-market strategy is focused on acquiring new customers and driving increased use of our services for existing customers.

Competition

We operate in a highly competitive industry with an increasing number of participants.

Our self-mining operations compete with mining operations throughout the world to complete new blocks in the blockchain and earn the reward in the form of an established unit of a digital asset. We compete on the basis of our total number of miners, the degree of mining difficulty, the efficiency of our mining operations and the fiat value of the mining reward.

While miners of digital assets historically range from individual enthusiasts and entrepreneurs to large public company mining operations and large company mining hosting operations with dedicated data centers, the vast majority of mining is now undertaken and further trending towards large-scale, industrial mining farms. A mining pool is created when mining participants pool the processing power of their miners over a network and mine transactions together. Rewards are then distributed proportionately to the pool participants based on the work/hash power contributed to solving a block. Our self-mining operations also compete with non-digital asset operations for access to suitable real estate and access to affordable and dependable electric power. In addition to competing to solve new blocks, we compete to acquire new miners, to raise capital, to obtain access to facilities for location of mining operations, and to develop or acquire new technologies.

Our hosting activities compete with a large number of other hosting operations. Our success in our hosting operations depends on our ability to supply hosting space and power, our performance with respect to installation, operation and repair of customer equipment, our ability to obtain replacement parts, the value of our service offering to our customers and the availability of mining equipment. A significant percentage of mining equipment is manufactured by a single supplier and almost all mining equipment is provided by a small number of manufacturers. We believe that we compete favorably as a provider of hosting services because of the design, size, quality and geographic distribution of our existing facility operations, our proprietary technology and software, our power contracts, our commitment to net carbon neutral operations, our experience and expertise and our relationships with equipment manufacturers.

Several public companies (traded in the United States, Canada, and internationally), such as the following, may be considered competitors to the Company:

- Argo Blockchain PLC;
- Bit Digital, Inc.;
- Bitcoin Investment Trust;
- Bitfarms Technologies Ltd. (formerly Blockchain Mining Ltd);
- Blockchain Industries, Inc. (formerly Omni Global Technologies, Inc.);
- Cipher Mining Inc.;
- Coinbase, Inc.;
- Digihost International, Inc.;
- DMG Blockchain Solutions Inc.;
- DPW Holdings, Inc. (through its ownership of Digital Farms Inc.);
- Greenidge Generation Holdings Inc.;
- HashChain Technology, Inc.;
- Hive Blockchain Technologies Inc.;
- Hut 8 Mining Corp.;

- Layer1 Technologies, Inc.;
- Marathon Digital Holdings, Inc.;
- MGT Capital Investments, Inc.;
- Northern Data AG;
- Overstock.com Inc.; and
- · Riot Blockchain, Inc.

The digital assets industry is a highly competitive and evolving industry and new competitors and/or emerging technologies could enter the market and affect our competitiveness in the future. Other market participants in the digital assets industry include investors and speculators, retail users transacting in digital assets, and service companies that provide a variety of services including buying, selling, payment processing and storing of digital assets. To continue to be successful, we will require sufficient additional capital to build additional facilities and to acquire new available mining equipment and related infrastructure. Subject to raising additional capital, our digital asset initiatives will compete with other industry participants that focus on investing in and securing the blockchains of Bitcoin and other digital assets.

Intellectual Property

Intellectual property is an important aspect of our business, and we seek protection for our intellectual property as appropriate. To establish and protect our proprietary rights, we rely upon a combination of patent, copyright, trade secret and trademark laws and contractual restrictions such as confidentiality agreements, licenses and intellectual property assignment agreements.

As of February 4, 2022, we had over 70 filed patent applications and issued patents in technologies such as blockchain, data center management, infrastructure and cooling. We maintain a policy requiring our employees, contractors, consultants and other third parties to enter into confidentiality and proprietary rights agreements to control access to our proprietary information. These laws, procedures and restrictions provide only limited protection, and any of our intellectual property rights may be challenged, invalidated, circumvented, infringed or misappropriated. Furthermore, the laws of certain countries do not protect proprietary rights to the same extent as the laws of the United States, and we therefore may be unable to protect our proprietary technology in certain jurisdictions. Moreover, our platform incorporates software components licensed to the general public under open-source software licenses. We obtain many components from software developed and released by contributors to independent open-source components of our platform. Open-source licenses grant licensees broad permissions to use, copy, modify and redistribute those open-source components of our platform. As a result, open-source development and licensing practices can limit the value of our software copyright assets.

We continually review our development efforts to assess the existence and patentability of new intellectual property. We pursue the registration of our domain names, trademarks and service marks in the United States and in certain locations outside the United States. To protect our brand, we file trademark registrations in some jurisdictions.

Legal Proceedings

From time to time, we are involved in various legal proceedings arising from the normal course of business activities. We are not presently a party to any litigation the outcome of which, we believe, if determined adversely to us, would individually or taken together have a material adverse effect on our business, operating results, cash flows or financial condition. Defending such proceedings is costly and can impose a significant burden on management and employees. We may receive unfavorable preliminary or interim rulings in the course of litigation, and there can be no assurances that favorable final outcomes will be obtained.

Government Regulation

Due to the relatively short history of digital assets, and their emergence as a new asset class, government regulation of blockchain and digital assets is constantly evolving, with increased interest expressed by U.S. and internal regulators. For example, the Cyber-Digital Task Force of the U.S. Department of Justice published a report entitled "Cryptocurrency: An Enforcement Framework" in October 2020 that detailed the Department's view with respect to digital assets and the tools at the Department's disposal to deal with threats posed by digital assets. In March 2021, the nominee for Chair of the U.S. Securities and Exchange Commission expressed the need for investor protection along with promotion of innovation in the digital asset space. In February 2021, representatives of the government of Inner Mongolia, China announced plans to ban digital asset mining within the province due to the energy and rare Earth mineral demands of the industry.

Government regulation of blockchain and digital assets is under active consideration by the United States federal government via its agencies and regulatory bodies, as well as by similar entities in other countries and transnational organizations, such as the European Union. State and local regulations also may apply to our activities and other activities in which we may participate in the future. Other governmental or semi-governmental regulatory bodies have shown an interest in regulating or investigating companies engaged in blockchain or digital asset businesses. For instance, the SEC has taken an active role in regulating the use of public offerings of proprietary coins (so-called "initial coin offerings") and has made statements and official promulgations as to the status of certain digital assets as "securities" subject to regulation by the SEC.

The effect of any regulatory change, either by the Federal, state, local or foreign governments or any self-regulatory agencies on the Company is impossible to predict, but such change could be substantial and may have a material adverse effect on the Company's business, financial condition and results of operations. While we are unaware of significant adverse governmental or regulatory action adverse to Bitcoin or Ethereum mining in the United States, there is no guarantee that future regulation or adverse action will not take place and interpretation of existing regulations in a manner adverse to our business is possible.

In addition, various foreign jurisdictions either have adopted, or may adopt, laws, regulations or directives that affect digital assets, digital asset networks, and their users and participants. Such laws, regulations or directives may conflict with those of the United States, may negatively impact the acceptance of digital assets by users, merchants and service providers outside of the United States, and may therefore impede the growth of digital assets. A number of Eastern European and Asian countries currently have a more restrictive stance toward digital assets and, thereby, have reduced the rate of expansion of digital asset use, as well as digital asset transaction processing, in each of those countries. Presently, we do not believe any U.S. or State regulatory body has taken any action or position adverse to our main digital asset, Bitcoin, with respect to its production, sale, and use as a medium of exchange; however, future changes to existing regulations or entirely new regulations may affect our business in ways it is not presently possible for us to predict with any reasonable degree of reliability. As the regulatory and legal environment evolves, we may become subject to new laws, such as further regulation by the SEC and other agencies, which may affect our mining and other activities.

Our Facilities

Our corporate headquarters is located at 106 East 6th Street, Suite 900-145, Austin, Texas 78701. We also lease approximately 9,177 square feet of commercial office space in Bellevue, Washington pursuant to operating leases that expire June 30, 2022. In addition, we maintain operations in owned facilities in North Carolina, Georgia, North Dakota and Kentucky, with approximately 497MW of operating electric power as of January 31, 2022. We lease at a nominal amount with an option to purchase our facility in North Dakota. In October 2021, we announced the entry of an agreement with the City of Denton, Texas and an affiliate of Tenaska Energy, Inc. to develop a blockchain data center in Denton, Texas, which is expected to have 300MW of power capacity. In February 2022, the Muskogee City-County Port Authority announced an agreement with us to develop a 500MW data center at the Port of Muskogee John T. Griffin Industrial Park. We believe that our facilities are suitable to

meet our current and anticipated near term needs. We continually explore the acquisition and development of additional facilities to support our anticipated future growth.

Seasonality

The Company's business is not generally subject to seasonality. However, coin generation from the Company's mining operations may vary depending on the Company's total hash rate at a given point in time relative to the total hash rate of the Bitcoin network.

Our Employees

All aspects of our business require specialized knowledge and technical skill. Such knowledge and skills include the areas of blockchain technology, research and development, digital currency assets, digital currency market, digital asset operations, human resource management, data privacy, as well as legal compliance, finance and accounting. We believe that we have adequate personnel and resources with the specialized skills required to carry out our operations successfully. As of December 31, 2021, we had 205 full-time employees. All of these employees are located in the United States. None of our employees are represented by a labor union or covered by collective bargaining agreements, and we have not experienced any work stoppages. We believe our relationship with our employees is good.

Corporate Information

We were originally known as Power & Digital Infrastructure Acquisition Corp. On January 19, 2022, Legacy Core, XPDI, and Merger Sub consummated the transactions contemplated under the merger agreement, following the approval at the special meeting of the stockholders of XPDI held on January 19, 2022. In connection with the closing of the Business Combination, we changed our name from Power & Digital Infrastructure Acquisition Corp. to Core Scientific, Inc.

Our principal executive offices are located at 106 East 6th Street, Suite 900-145, Austin, Texas 78701, and our telephone number is (425) 998-5300. Our corporate website address is www.corescientific.com. Information contained on or accessible through our website is not a part of this prospectus, and the inclusion of our website address in this prospectus is an inactive textual reference only.

MANAGEMENT

Executive Officers and Directors

The following table sets forth certain information, as of February 4, 2022, concerning the persons who serve as executive officers and directors of Core:

Name	Age	Position
Executive Officers		
Michael Levitt	63	Chief Executive Officer and Co-Chair of the Board of Directors
Darin Feinstein	50	Chief Vision Officer and Co-Chair of the Board of Directors
Michael Trzupek	51	EVP and Chief Financial Officer
Todd M. DuChene	58	EVP, General Counsel, Chief Compliance Officer and Secretary
Non-Employee Directors		
Jarvis Hollingsworth(1)(3)	59	Director
Matt Minnis(2)(3)	57	Director
Stacie Olivares(1)(2)(3)	47	Director
Kneeland Youngblood(1)(2)(3)	66	Director

- Member of the Audit Committee.
- (2) Member of the Compensation Committee.
- (3) Member of the Nominating and Corporate Governance Committee.

Executive Officers

Michael Levitt. Upon the closing of the Business Combination, Mr. Levitt became the Chief Executive Officer and Co-Chair and a member of the board of directors of post-combination company. Mr. Levitt has served as the Chief Executive Officer of Legacy Core since May 2021, and as Chairman and a member of Legacy Core's board of directors since June 2018. From July 2016 to July 2021, Mr. Levitt was the Chief Executive Officer of Kayne Anderson Capital Advisors, L.P. Prior to joining Kayne, Mr. Levitt served as a Vice Chairman and Partner of Apollo Global Management, LLC (NYSE: APO) from April 2012 to May 2016. Mr. Levitt joined Apollo following Apollo's acquisition of Stone Tower Capital LLC, an investment firm, which he founded in 2001 and where he served as Chairman of the board of directors, Chief Executive Officer and Chief Investment Officer. Prior to that, Mr. Levitt worked as a Partner at Hicks, Muse, Tate & Furst Incorporated, a private equity firm. Earlier in his career, Mr. Levitt served as the Co-Head of the Investment Banking Division of Smith Barney Inc. Mr. Levitt began his investment banking career at Morgan Stanley & Co., Inc. Mr. Levitt has served on the board of directors of The Music Acquisition Corporation (NYSE: TMAC) since December 2020 and served on the board of directors of Kayne Anderson BDC, LLC from May 2018 to October 2021. Mr. Levitt holds a B.B.A. and J.D. from the University of Michigan. Mr. Levitt was appointed to serve on the board of directors of Core because of his significant financial expertise, his knowledge of the corporate credit and investment markets, his business acumen and his leadership as a sitting Chief Executive Officer.

Darin Feinstein. Upon the closing of the Business Combination, Mr. Feinstein became the Chief Vision Officer and Co-Chair of the board of directors of the post-combination company. Mr. Feinstein is an American entrepreneur who has been involved in many ventures within the digital assets space since 2012. Mr. Feinstein co-founded Legacy Core, the premier blockchain provider, delivering best-in-class infrastructure and software solutions in 2017 and founded Blockcap in 2020. In addition to these blockchain companies Mr. Feinstein owns and operates a portfolio of business ventures across a variety of industries including live entertainment, finance, food & beverage, hospitality and technology, among others. Mr. Feinstein is a philanthropist who co-founded The Feinstein Griffin foundation, with Eddie Griffin, and Mr. Feinstein and Mr. Griffin for the past decade have been hosting events and donating to various charities around the United States. Mr. Feinstein is a California Licensed attorney (currently non-practicing) and holds many privileged licenses including a gaming license,

liquor licenses and various financial licenses in multiple states across the United States. Mr. Feinstein was appointed to serve on the board of directors of Core because of his significant experience in the digital asset space and his institutional knowledge of our businesses as a co-founder of Legacy Core and Blockcap.

Michael Trzupek. Upon the closing of the Business Combination, Mr. Trzupek became the EVP and Chief Financial Officer of the post-combination company. Mr. Trzupek has served as the Chief Financial Officer of Legacy Core since September 2020. Prior to joining Core, Mr. Trzupek served as Chief Financial Officer of Premera Blue Cross, an independent licensee of Blue Cross Blue Shield for residents and businesses in the States of Alaska and Washington, from February 2019 to June 2020. Mr. Trzupek served as Interim Chief Financial Officer for the Central Puget Sound Regional Transit Authority, a public transit agency serving the metropolitan Seattle area, from February 2018 to August 2018. Prior to that, he served as the Group Vice President for Providence-St. Joseph Health and Services, one of the nation's largest non-profit healthcare systems, from November 2015 to February 2017. From April 2007 to October 2014, Mr. Trzupek served in various financial positions at Microsoft, Inc., including General Manager of Corporate Finance and General Manager of New Venture Integration. From June 1997 to March 2007, Mr. Trzupek served in various financial positions at Intel Corporation, including Senior Controller of System Materials and Manufacturing. Mr. Trzupek also serves as an advisor to a number of board of directors for private investment companies. Mr. Trzupek holds a B.S. in Chemical Engineering from the University of Illinois and an M.B.A. from the University of Chicago Booth School of Business.

Todd M. DuChene. Upon the closing of the Business Combination, Mr. DuChene became the EVP, General Counsel, Chief Compliance Officer and Secretary of the post-combination company. Mr. DuChene has served as the General Counsel and Secretary of Legacy Core since April 2019. Prior to joining Core, Mr. DuChene served as Senior Vice President, General Counsel and Secretary and Chief Compliance Officer for FLIR Systems, Inc., an industrial and military technology company, from September 2014 to April 2019. Prior to joining FLIR, Mr. DuChene served as Executive Vice President, General Counsel and Secretary of Nuance Communications, Inc., a leading provider of speech recognition and related technology to enterprise, healthcare and mobile and consumer customers, where he was responsible for the legal, intellectual property, corporate governance and regulatory activities of the company, from October 2011 to September 2014. Previously, Mr. DuChene served as Senior Vice President, General Counsel and Secretary of National Semiconductor Corporation from January 2008 to October 2011, prior to its acquisition by Texas Instruments Inc. In addition, Mr. DuChene has served as General Counsel to each of Solectron Corporation, Fisher Scientific International Inc. (now ThermoFisher Scientific) and OfficeMax, Inc. Mr. DuChene began his legal career as a corporate lawyer with BakerHostetler in Cleveland, Ohio in 1988. Mr. DuChene holds a B.A. in Political Science from The College of Wooster and a J.D. from the University of Michigan Law School.

Non-Employee Directors

Jarvis Hollingsworth. Upon the closing of the Business Combination, Mr. Hollingsworth became a member of the board of directors of the post-combination company. Mr. Hollingsworth has served as Chairman of the Board of Trustees of the Teacher Retirement System of Texas, a \$190 billion pension trust fund for retirees, teachers and employees of education-related institutions since November 2017. Since November 2019, Mr. Hollingsworth has also served on the board of directors of Laredo Petroleum, Inc. (NYSE:LPI), a diversified energy company, and on the finance committee of the Memorial Hermann Hospital System since May 2017. Mr. Hollingsworth served as General Counsel and a member and executive committee member of the board of directors for Kayne Anderson Capital Advisors, L.P., a leading alternatives investment management firm from May 2019 to July 2021. Mr. Hollingsworth served as a director of Emergent Technologies LP, a fintech company that provides payment services in emerging markets using a gold-backed digital currency from April 2016 to November 2019, and from May 2017 to July 2019 served as a director of Cullen Frost Bankers, Inc. (NYSE: CFR). Mr. Hollingsworth holds a Bachelor of Science from the United States Military Academy at West Point and served for several years on active and reserve duty in the United States Army. Mr. Hollingsworth holds a J.D. from the University of Houston Law Center. Mr. Hollingsworth was appointed to serve on the board of

directors of Core because of his knowledge of the digital assets industry and his understanding of public company governance and operations from his service on public company boards of directors.

Matt Minnis. Upon the closing of the Business Combination, Mr. Minnis became a member of the board of directors of the post-combination company. Mr. Minnis has served as a member of the Legacy Core board of directors since June 2018 and is a Co-Founder of Core. Mr. Minnis has also served as Managing Member of BEP 888 and BEP 999 digital asset mining companies, which were founded in 2020. Further, Mr. Minnis has served as Chairman of the board of directors of Pledgling Technologies, Inc., a company focused on charitable donations, which he co-founded in 2014. Mr. Minnis has also served as President and Chief Executive Officer of Minnis Investments LLC, a private investment firm focused on early-stage technology companies, real estate and venture capital investments since 2009. Mr. Minnis holds a B.B.A. in Marketing and M.B.A. in Finance from Texas Christian University. Mr. Minnis was appointed to serve on the board of directors of Core because of his knowledge of Legacy Core and the digital asset mining industry as well as his investment and venture capital expertise.

Stacie Olivares. Upon the closing of the Business Combination, Ms. Olivares became a member of the board of directors of the post-combination company. Appointed by the Governor in August 2019, Ms. Olivares serves as one of thirteen Trustees of the California Public Employees' Retirement System (CalPERS), the largest public pension in the United States overseeing approximately \$500 Billion. Ms. Olivares serves on four committees of the CalPERS Board of Administration: Governance, Finance & Administration, Investment, and Performance, Compensation & Talent Management. From March 2019 to April 2020, Ms. Olivares served as Chief Investment Officer of Lendistry, a minority-owned CDFI and fintech lender. Prior to Lendistry, from October 2011 to January 2018 Ms. Olivares was the Managing Director and Chief Investment Officer of the California Organized Investment Network (COIN), which guided insurance investments for financial, environmental, and social returns. During her tenure, the value of insurers' COIN holdings rose from \$6.6 billion in 2011 to approximately \$29 billion in 2018. Envisioning the potential of technology to create new markets and preserve the environment, Ms. Olivares developed early clean technology portfolios at Morgan Stanley that hedged 2009 financial crisis risk. Before Wall Street, she led the State of California's economic advisory board and advanced trade, housing, and job growth as the state's economy grew from the 6th to the 4th largest in the world. Ms. Olivares advises several early stage companies and serves on the board of the Hispanic Heritage Foundation. Ms. Olivares was appointed to serve on the board of directors of Core because of her financial expertise and her understanding of corporate and board governance early stage investment expertise and experience in matters related to ESG.

Kneeland Youngblood. Upon the closing of the Business Combination, Mr. Youngblood became a member of the board of directors of the post-combination company. Mr. Youngblood has served as a member of the board of directors of numerous TPG Pace SPAC transactions since 2015. Mr. Youngblood is a director nominee of TPG Pace Beneficial II Corp. Mr. Youngblood has served as a founding partner of Pharos Capital Group, LLC since 1998, a private equity firm that focuses on providing growth and expansion capital/buyouts in the health care service sector. Mr. Youngblood is a director of Mallinckrodt Pharmaceuticals, a director of Scientific Games Corporation and Chairman of the Finance Committee of the President's Advisory Board of the UT Southwestern Medical Center. Previously, Mr. Youngblood served as a member of the board of directors of Pace-I from September 2015 through its business combination with Playa in March 2017. Mr. Youngblood has also served as a member of the board of directors of Pace-II from June 2017 through its business combination with Accel. Mr. Youngblood is also a former director of Burger King Corporation, Starwood Hotels and Lodging, Gap Inc. and Energy Future Holdings (formerly TXXU). He also serves on several private company and not-for-profit boards. Mr. Youngblood graduated from Princeton University in 1978 with an A.B in Politics/Science in Human Affairs and earned an M.D. degree from the University of Texas, Southwestern Medical School. Mr. Youngblood was appointed to serve on the board of directors of Core because of his extensive leadership experience, expertise in growth investing and his service as a board member across a diverse set of industries.

Family Relationships

There are no family relationships among any of Core's directors or executive officers.

Composition of Our Board of Directors

Our business and affairs are organized under the direction of our board of directors. Our board of directors will meet on a regular basis and additionally as required.

In accordance with the terms of the Bylaws, our board of directors may establish the authorized number of directors from time to time by resolution. Our board of directors currently consists of six members. Each member of the board of directors will serve a one-year term expiring at our next annual meeting of stockholders, subject to his or her office being vacated sooner pursuant to the Bylaws.

Director Independence

Based on information provided by each director concerning his or her background, employment and affiliations, the board of directors of Core has determined that none of the Core directors, other than Mr. Levitt and Mr. Feinstein, has any relationships that would interfere with the exercise of independent judgment in carrying out the responsibilities of a director and that each of the directors is "independent" as that term is defined under the Nasdaq listing standards. In making these determinations, the board of directors of Core considered the current and prior relationships that each non-employee director has with Core and all other facts and circumstances the board of directors of Core deemed relevant in determining their independence, including the beneficial ownership of securities of Core by each non-employee director and the transactions described in the section titled "Certain Relationships and Related Party Transactions."

Board Leadership Structure

The board of directors of Core is co-chaired by Mr. Levitt and Mr. Feinstein. Mr. Levitt also serves as Core's Chief Executive Officer. Core believes that combining the positions of Chief Executive Officer and Chair of the board of directors will help to ensure that the board and management will act with a common purpose and provide a single, clear chain of command to execute Core's strategic initiatives and business plans. In addition, Core believes that a combined Chief Executive Officer/Co-Chair is better positioned to act as a bridge between management and the board of directors, facilitating the regular flow of information.

The Core board of directors appointed Jarvis Hollingsworth as lead independent director in order to help reinforce the independence of the board as a whole. The position of lead independent director has been structured to serve as an effective balance to Mr. Levitt's leadership as the combined Chief Executive Officer and Co-Chair. The lead independent director is empowered to, among other duties and responsibilities, work with the Chief Executive Officer to develop and approve an appropriate board meeting schedule; work with the Chief Executive Officer to develop and approve meeting agendas; provide the Chief Executive Officer feedback on the quality, quantity and timeliness of the information provided to the board; develop the agenda and moderate executive sessions of the independent members of the board of directors; preside over board meetings when the Chief Executive Officer is not present or when such person's performance or compensation is discussed; act as principal liaison between the independent members of the board of directors and the Chief Executive Officer; convene meetings of the independent directors as appropriate; and perform such other duties as may be established or delegated by the board of directors. As a result, Core believes that the lead independent director can help ensure the effective independent functioning of the board of directors in its oversight responsibilities. In addition, Core believes that the lead independent director serves as a conduit between the other independent directors and the Co-Chairs, for example, by facilitating the inclusion on meeting agendas of matters of concern to the independent directors.

Role of Core Board in Risk Oversight

One of the key functions of the board of directors is informed oversight of Core's risk management process. It is not anticipated that the board of directors of Core will have a standing risk management committee, but rather that this oversight function will be administered directly through the board of directors of Core as a whole, as well as through various standing committees of the board of directors of Core that address risks inherent in their respective areas of oversight. In particular, Core's board of directors will be responsible for monitoring and assessing strategic risk exposure and Core's audit committee will have the responsibility to consider and discuss Core's major financial risk exposures and the steps its management will take to monitor and control such exposures, including guidelines and policies to govern the process by which risk assessment and management is undertaken. The audit committee will also monitor compliance with legal and regulatory requirements. Core's compensation committee will also assess and monitor whether Core's compensation plans, policies and programs comply with applicable legal and regulatory requirements.

Board Committees

Upon the consummation of the Business Combination, the board of directors reconstituted its audit committee, compensation committee and nominating and corporate governance committee and adopted a new charter for each of these committees, which complies with the applicable requirements of current SEC and Nasdaq rules. Core is expected to comply with future requirements to the extent applicable. Copies of the charters for each committee will be available on the investor relations portion of Core's website.

Audit Committee

The audit committee of Core consists of Stacie Olivares, Kneeland Youngblood and Jarvis Hollingsworth, each of whom Core's board of directors has determined satisfies the independence requirements under Nasdaq listing standards and Rule 10A-3(b)(1) of the Exchange Act. The chair of the audit committee is Ms. Olivares, who Core's board of directors has determined is an "audit committee financial expert" within the meaning of SEC regulations. Each member of the audit committee can read and understand fundamental financial statements in accordance with applicable requirements. In arriving at these determinations, Core has examined each audit committee member's scope of experience and the nature of their employment in the corporate finance sector.

The primary purpose of the audit committee is to discharge the responsibilities of the board of directors with respect to Core's corporate accounting and financial reporting processes, systems of internal control and financial statement audits, and to oversee Core's independent registered public accounting firm. Specific responsibilities of the audit committee include:

- Helping the board of directors of Core oversee corporate accounting and financial reporting processes;
- Managing the selection, engagement, qualifications, independence and performance of a qualified firm to serve as the independent registered public accounting firm to audit the financial statements;
- Discussing the scope and results of the audit with the independent registered public accounting firm, and reviewing, with management and the independent accountants, the interim and year-end operating results;
- Developing procedures for employees to submit concerns anonymously about questionable accounting or audit matters;
- · Reviewing related person transactions;
- Obtaining and reviewing a report by the independent registered public accounting firm at least annually that describes internal quality control procedures, any material issues with such procedures and any steps taken to deal with such issues when required by applicable law;
 and
- Approving or, as permitted, pre-approving, audit and permissible non-audit services to be performed by the independent registered public
 accounting firm.

Compensation Committee

The compensation committee of Core consists of Kneeland Youngblood, Stacie Olivares and Matt Minnis. The chair of the compensation committee is Mr. Youngblood. Core's board of directors has determined that each member of the compensation committee is independent under the Nasdaq listing standards and a "non-employee director" as defined in Rule 16b-3 promulgated under the Exchange Act.

The primary purpose of the compensation committee is to discharge the responsibilities of the board of directors in overseeing the compensation policies, plans and programs and to review and determine the compensation to be paid to executive officers, directors and other senior management, as appropriate. Specific responsibilities of the compensation committee include:

- Reviewing and approving the compensation of the chief executive officer, other executive officers and senior management;
- Administering the equity incentive plans and other benefit programs;
- Reviewing, adopting, amending and terminating incentive compensation and equity plans, severance agreements, profit sharing plans, bonus plans, change-of-control protections and any other compensatory arrangements for the executive officers and other senior management; and
- Reviewing and establishing general policies relating to compensation and benefits of the employees.

Nominating and Corporate Governance Committee

The nominating and corporate governance committee of Core consists of Jarvis Hollingsworth, Stacie Olivares, Kneeland Youngblood and Matt Minnis. The chair of the nominating and corporate governance committee is Mr. Hollingsworth. Core's board of directors has determined that each member of the nominating and corporate governance committee is independent under the Nasdaq listing standards.

- Specific responsibilities of the nominating and corporate governance committee include:
- Identifying and evaluating candidates, including the nomination of incumbent directors for reelection and nominees recommended by stockholders, to serve on the board of directors of Core;
- Considering and making recommendations to the board of directors regarding the composition and chairmanship of the committees of the board of directors of Core;
- Developing and making recommendations to the board of directors regarding corporate governance guidelines and matters, including in relation to corporate social responsibility; and
- Overseeing periodic evaluations of the performance of the board of directors of Core, including its individual directors and committees.

Code of Business Conduct and Ethics

Core has adopted a code of conduct (the "Code of Conduct") applicable to all of Core's employees, directors and officers, including Core's principal executive officer, principal financial officer, principal accounting officer or controller, or persons performing similar functions. The Code of Conduct is available under the Governance section of Core's website at https://investors.corescientific.com. In addition, Core has posted on its website all disclosures that are required by law or Nasdaq listing standards concerning any amendments to, or waivers from, any provision of the Code of Conduct. The reference to Core's website address does not constitute incorporation by reference of the information contained at or available through such website, and you should not consider it to be a part of this prospectus.

Compensation Committee Interlocks and Insider Participation

None of the members of Core's compensation committee has ever been one of its executive officers or employees. None of Core's executive officers currently serves, or has served during the last completed fiscal

year, as a member of the board of directors or compensation committee of any other entity that has one or more executive officers that will serve as a member of Core's board of directors or compensation committee.

Limitation on Liability and Indemnification of Directors and Officers

The Charter, limits a directors' liability to the fullest extent permitted under the DGCL. The DGCL provides that directors of a corporation will not be personally liable for monetary damages for breach of their fiduciary duties as directors, except for liability:

- for any transaction from which the director derives an improper personal benefit;
- · for any act or omission not in good faith or that involves intentional misconduct or a knowing violation of law;
- · for any unlawful payment of dividends or redemption of shares; or
- for any breach of a director's duty of loyalty to the corporation or its stockholders.

If the DGCL is amended to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of the directors will be eliminated or limited to the fullest extent permitted by the DGCL, as so amended.

Delaware law and the Bylaws provide that Core will, in certain situations, indemnify its directors and officers and may indemnify other employees and other agents, to the fullest extent permitted by law. Any indemnified person is also entitled, subject to certain limitations, to advancement, direct payment, or reimbursement of reasonable expenses (including attorneys' fees and disbursements) in advance of the final disposition of the proceeding.

In addition, Core entered into separate indemnification agreements with its directors and officers. These agreements, among other things, require Core to indemnify its directors and officers for certain expenses, including attorneys' fees, judgments, fines, and settlement amounts incurred by a director or officer in any action or proceeding arising out of their services as one of its directors or officers or any other company or enterprise to which the person provides services at its request.

Core maintains a directors' and officers' insurance policy pursuant to which its directors and officers are insured against liability for actions taken in their capacities as directors and officers. We believe these provisions in the Charter and the Bylaws and these indemnification agreements are necessary to attract and retain qualified persons as directors and officers.

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers, or control persons, in the opinion of the SEC, such indemnification is against public policy as expressed in the Securities Act and is therefore unenforceable.

EXECUTIVE COMPENSATION

This discussion may contain forward-looking statements that are based on Core's current plans, considerations, expectations and determinations regarding future compensation programs. Actual compensation programs that Core adopts following the completion of the merger may differ materially from the currently planned programs summarized in this discussion. All share counts in this section are shown on a pre-merger basis.

For the fiscal year ended December 31, 2021, Core's named executive officers were:

- Michael Levitt, Chief Executive Officer;
- · Kevin Turner, former President and Chief Executive Officer;
- · Michael Trzupek, Executive Vice President and Chief Financial Officer; and
- Todd M. DuChene, Executive Vice President, General Counsel, Chief Compliance Officer and Secretary.

Summary Compensation Table

The following table provides information regarding total compensation awarded to, earned by, and paid to Core's named executive officers for services rendered to Legacy Core in all capacities for the fiscal years ended December 31, 2020 and 2021.

Name and Principal Position	Year	Salary(\$)(1)	Stock Awards(\$)(2)	All Other Compensation (\$)	Total(\$)
Michael Levitt(3)					
Chief Executive Officer	2021	30,824(4)	160,664,903	_	160,695,727
B. Kevin Turner(5)	2020	300,000	9,716,229	33,217(6)	10,049,446
Former President and Chief Executive Officer	2021	121,978(7)	_	153,679(8)	275,721
Michael Trzupek(9)	2020	75,000(10)	13,803,152	_	13,878,152
EVP and Chief Financial Officer	2021	300,824	9,210,000	_	9,510,824
Todd M. DuChene	2020	300,000	795,049	_	1,095,049
EVP, General Counsel, Chief Compliance Officer and					
Secretary	2021	300,824	12,292,500	498,691(11)	13,092,015

- (1) Salary amounts represent actual amounts earned during the fiscal year.
- (2) Amounts reported represent the aggregate grant date fair value of RSUs granted to the named executive officer during the fiscal year. The aggregate grant date fair value is based upon an estimate of Legacy Core Common Stock at the grant date. In accordance with the Financial Accounting Standard Board Accounting Standards Codification, Topic 718, or ASC Topic 718, recognition of compensation cost was deferred until consummation of the Business Combination. See Note 2 of the audited consolidated financial statements of Legacy Core included in the Prospectus for a discussion of the relevant assumptions used in calculating this amount. This amount does not reflect the actual economic value that may be realized by the named executive officer.
- (3) Mr. Levitt was not one of our named executive officers for the year ended December 31, 2020, and was appointed as our Chief Executive Officer in May 2021.
- (4) The salary reported for Mr. Levitt represents a pro-rata portion of his salary in 2021. His annualized base salary for 2021 was \$60,000.
- (5) Mr. Turner resigned as our President and Chief Executive Officer in May 2021.
- (6) Amount shown consists of medical benefits of \$32,325, and \$64 group life insurance premiums and \$828 out-of-pocket medical expenses incurred by Legacy Core for Mr. Turner.

- (7) The salary reported for Mr. Turner represents a pro-rata portion of his salary in 2021. His annualized base salary for 2021 was \$300,000.
- (8) Amount shown consists of \$17,217 COBRA payments and \$136,462 severance payment upon the termination of Mr. Turner's employment with
- (9) Mr. Trzupek was appointed as our Chief Financial Officer in September 2020.
- (10) The salary reported for Mr. Trzupek represents a pro-rata portion of his salary in 2020. His annualized base salary for 2020 was \$300,000.
- (11) Amount shown represents \$498,691 relocation expenses reimbursed by Legacy Core to Mr. DuChene.

Narrative Disclosure to Summary Compensation Table

Annual Base Salary

The compensation of Core's named executive officers is generally determined and approved by the compensation committee of its board of directors. The base salaries of each of the named executive officers for the fiscal years ended December 31, 2020 and 2021 are listed in the table below.

Name	Fiscal Year 2020 Base Salary(\$)	Fiscal Year 2021 Base Salary(\$)
Michael Levitt	- (1)	60,000
B. Kevin Turner	300,000	300,000
Michael Trzupek	300,000	300,000
Todd M. DuChene	300.000	300.000

Mr. Levitt was not one of our named executive officers for the year ended December 31, 2020, and was appointed as our Chief Executive Officer in May 2021.

Annual Performance-Based Bonus Opportunity

From time to time, Core's board of directors or compensation committee may approve bonuses for its executive officers based on individual performance, company performance or as otherwise determined appropriate. For the fiscal years ended December 31, 2020 and 2021, Core's executive officers were not eligible to receive performance-based cash bonuses.

Equity-Based Incentive Awards

Core's equity-based incentive awards are designed to align Core's interests and those of Core's stockholders with those of Core's employees and consultants, including its executive officers. The board of directors is responsible for approving equity grants.

To preserve cash and to incentivize exceptional performance from its executive officers, Core has historically used RSUs as an incentive for long-term compensation to its executive officers. Core's executives generally are awarded an initial grant in the form of RSUs in connection with their commencement of employment. Additional grants may occur periodically in order to specifically incentivize executives with respect to achieving certain corporate goals or to reward executives for exceptional performance. Core may grant equity awards at such times as its board of directors determines appropriate.

All RSUs have been granted pursuant to the Core Scientific, Inc. (f/k/a Mineco Holdings, Inc.) 2018 Omnibus Incentive Plan (as amended, the "2018 Plan"). For additional information regarding the equity awards held by Core's named executive officers as of December 31, 2021, please see the section below titled "-Outstanding Equity Awards at Fiscal Year-End of 2021."

Following the closing of the Business Combination, Core no longer grants equity incentive awards under the 2018 Plan.

Employment Agreements with Named Executive Officers

Michael Levitt. In October 2021, Legacy Core entered into an amended and restated employment agreement with Mr. Levitt, pursuant to which Mr. Levitt serves as our Chief Executive Officer. Mr. Levitt's agreement provides for a base salary of \$60,000 and an initial award of RSUs convertible into 8,400,000 shares of Core Common Stock, which was granted to Mr. Levitt in July 2021. In addition, pursuant to the terms of his agreement, Mr. Levitt was granted an award of 3,050,000 RSUs in connection with the Business Combination. Mr. Levitt also has the opportunity to earn an additional annual grant of up to 1,000,000 additional RSUs subject to achievement of certain performance metrics established by the board of directors of Core. If, for certain period during the term of his agreement, the positive total shareholder return ("TSR") of Core exceeds the TSR for the S&P 500 Index during the same period ("TSR Over Performance"), the Mr. Levitt is entitled to a bonus equal to the product of .0375 multiplied by the amount of the increase in Core equity value attributable to TSR Over Performance. The bonus amount is payable at the option and discretion of Core in cash, bitcoin or shares of Core Common Stock having a fair market value on the date of issue equal to the amount of the bonus. In the event Mr. Levitt's employment is terminated by Core without Cause (as defined therein), by Mr. Levitt for Good Reason (as defined therein), by either party after a Change in Control (as defined therein), or as a result of a Non-Renewal (as defined below) by the Company, Mr. Levitt is entitled to the following severance benefits: (i) accrued and unpaid base salary, accrued benefits, and reimbursement of expenses as of the date of termination, (ii) base salary continuation for twelve months, (iii) bonus, if any, for the year in which the termination occurs in an amount equal to the target bonus amount approved by the board of directors, and (iv) a cash payment under COBRA equal to the full premium for actively employed executives of Core with the same level of coverage for twelve months following the date of termination. Mr. Levitt's agreement continues until May 17, 2025, following which the agreement automatically renews for one additional year on each anniversary thereafter, unless, not less than ninety (90) days prior to the commencement of any such renewal term, either party has given written notice to the other that it does not wish to extend the agreement (a "Non-Renewal").

Kevin Turner. In June 2018, Legacy Core entered into an employment agreement with Mr. Turner. The agreement provided for Mr. Turner to serve as Core's President and Chief Executive Officer for an initial term of four years, commencing on July 1, 2018, which term automatically renewed for additional one-year terms unless either party notified the other party of its intention to terminate the agreement not less than ninety (90) days prior to the expiration of the current term. Mr. Turner resigned as Legacy Core's President and Chief Executive Officer in May 2021. Pursuant to the agreement, Mr. Turner was entitled to an initial annual base salary of \$300,000, which was reviewed and could have been adjusted by Core's board of directors on an annual basis. In addition, as contemplated by his agreement, Mr. Turner was granted 5,000,000 RSUs. Mr. Turner was entitled to certain severance benefits, the terms of which are described below under "-Potential Payments Upon Termination or Change in Control." Mr. Turner was also eligible for standard benefits such as paid time off, reimbursement of business expenses, and to participate in employee benefit plans and programs.

Michael Trzupek. In September 2020, Legacy Core entered into a letter agreement with Mr. Trzupek. Mr. Trzupek's agreement provides for a base salary of \$300,000 and an initial award of RSUs convertible into 2,000,000 shares of Legacy Core Common Stock, which was granted to Mr. Trzupek in October 2020. His agreement also provides for payment equal to three months base salary if Mr. Trzupek's employment is terminated by Core other than for Cause (as defined in his letter agreement). See also the section below titled "-Potential Payments upon Termination or Change in Control."

Todd M. DuChene. In December 2018, Legacy Core entered into a letter agreement with Mr. DuChene, effective upon his first day of employment on April 1, 2019. Mr. DuChene's agreement provides for a base salary of \$300,000 and an initial award of RSUs convertible into 1,000,000 shares of Legacy Core Common Stock. His agreement also provides for payment equal to three months base salary if Mr. DuChene's employment is terminated by Core other than for Cause (as defined in his letter agreement). See also the section below titled "-Potential Payments upon Termination or Change in Control."

Potential Payments upon Termination or Change in Control

Core has entered into certain agreements that will require Core to provide compensation to our named executive officers in the event of a termination of employment of Core.

In the event Mr. Levitt's employment is terminated as a result of death or "Disability" (as defined in his agreement), or by Core without "Cause" (as defined in his agreement) or for "Good Reason" (as defined in his agreement), then all of the equity awards granted to Mr. Levitt during the term of his employment will become fully vested and exercisable as of the date of his termination. See also the section above titled "-Employment Agreements with Named Executive Officers."

In the event of a termination of employment without "cause" (as defined in their respective offer letter), and subject to the delivery to Core of a general release of claims, each of Messrs. Trzupek and DuChene is entitled to three months of base salary. In addition, in the event Messrs. Trzupek and DuChene are terminated for any reason other than for "cause" (as defined in the 2018 Plan), all RSUs that have satisfied the time-based vesting condition as of the date of termination remain outstanding and eligible to fully vest for three years following the date of such termination, and, if the time-vested RSUs satisfy the transaction-based vesting condition prior to the third anniversary of the date of termination, the RSUs will be fully vested and will be settled through the issuance of common stock.

In May 2021, Mr. Turner executed a separation agreement, which provides for 12 months of base salary, and potential cash payments equal to the full premium for company-paid continuing health care benefits. As disclosed in the section above titled "-Summary Compensation Table," Mr. Turner received COBRA payments in the amount of \$17,217 and a severance payment of \$136,462 upon the termination of his employment in 2021. In addition, Mr. Turner's separation agreement provides that, as of Mr. Turner's separation date, 9,037,500 RSUs had satisfied the time-based vesting condition, but these RSUs remain subject to the transaction-based vesting condition described in Note 11 to Legacy Core's consolidated financial statements as of and for the years ended December 31, 2020 and 2019, which are included elsewhere in the registration statement of which this prospectus forms a part. The transaction-based vesting condition must be satisfied within three years of Mr. Turner's separation date. Further, under the terms of the 2018 Plan, the board of directors of Core may, in its sole discretion, provide for acceleration of the transaction-based vesting condition.

Outstanding Equity Awards at Fiscal Year-End of 2021

The following table presents the outstanding equity incentive plan awards held by each named executive officer as of December 31, 2021.

		Stock Awards(1)			
Name	Grant Date	Vesting Commencement Date	Number of Shares or Units of Stock that Have Not Vested (#)	Market Value of Shares or Units of Stock that Have Not Vested(\$)(2)	
Michael Levitt(3)	July 2, 2021	July 2, 2021	8,400,000(4)	146,664,000	
B. Kevin Turner(5)	September 21, 2018	July 1, 2018	3,645,864(5)	63,656,785	
	October 1, 2018	July 1, 2018	2,187,477(5)	38,193,348	
	October 9, 2018	July 1, 2018	729,159(5)	12,731,116	
	July 31, 2019	July 1, 2019	500,000(5)	8,730,000	
	April 29, 2020	April 30, 2020	1,725,000(5)	30,118,500	
	August 24, 2020	August 25, 2020	250,000(5)	4,365,000	
Michael Trzupek(6)	October 1, 2020	September 21, 2020	2,000,000(4)	34,920,000	
	July 9, 2021	June 9, 2021	500,000(4)	8,730,000	
Todd M. DuChene	April 1, 2019	April 1, 2019	1,000,000(7)	17,460,000	
	June 12, 2020	June 12, 2020	250,000(4)	4,365,000	
	February 2, 2021	January 1, 2021	250,000(4)	4,365,000	
	July 9, 2021	June 24, 2021	500,000(4)	8,730,000	

- (1) All stock awards listed in this table represent RSUs granted pursuant to the 2018 Plan, the terms of which are described below under "-Equity Plan"
- (2) This column represents the fair market value of a share of Core Common Stock of \$17.46 as of December 31, 2021 as determined by its board of directors, multiplied by the amount shown in the column "Stock Awards-Number of Shares or Units of Stock that Have Not Vested."
- (3) Mr. Levitt joined Core as Chief Executive Officer in May 2021.
- (4) One fourth of these RSUs vest on each of the first four anniversaries of the vesting commencement date, provided that the recipient remains in continuous service with us through each vesting date, and subject to the earlier to occur of (i) a change of control event, and (ii) an initial public offering of Legacy Core's equity securities.
- (5) Mr. Turner resigned as Legacy Core's President and Chief Executive Officer in May 2021. Mr. Turner's separation agreement provides that, as of Mr. Turner's separation date, 9,037,500 RSUs had satisfied the time-based vesting condition, but these RSUs remain subject to the transaction-based vesting condition described in Note 11 to Legacy Core's consolidated financial statements as of and for the years ended December 31, 2020 and 2019, which are included elsewhere in the registration statement of which this prospectus forms a part. The transaction-based vesting condition must be satisfied within three years of Mr. Turner's separation date.
- (6) Mr. Trzupek joined Core as Chief Financial Officer in September 2020.
- (7) One fourth of these RSUs vest on the one year anniversary of the vesting commencement date and 1/36 of the remaining RSUs vest monthly thereafter, provided that the recipient remains in continuous service with us through each vesting date, and subject to the earlier to occur of (i) a change of control event, and (ii) an initial public offering of Legacy Core's equity securities.

Other Compensation and Benefits

All of Core's named executive officers are eligible to participate in Core's employee benefit plans, including its medical, dental, vision, life and accidental death and dismemberment insurance plans, in each case on the same basis as all of Core's other employees. Mr. Turner, Core's former President and Chief Executive Officer, received reimbursements for certain medical (e.g., insurance premiums) and out-of-pocket expenses. Core generally does not provide perquisites or personal benefits to the named executive officers.

Core maintains a 401(k) plan that provides eligible U.S. employees with an opportunity to save for retirement on a tax advantaged basis. Eligible employees are able to defer eligible compensation up to certain Code limits, which are updated annually. Core has the ability to make matching and discretionary contributions to the 401(k) plan. Currently, Core does not make matching contributions or discretionary contributions to the 401(k) plan. The 401(k) plan is intended to be qualified under Section 401(a) of the Code, with the related trust intended to be tax exempt under Section 501(a) of the Code. As a tax-qualified retirement plan, contributions to the 401(k) plan are deductible by Core when made and contributions and earnings on those amounts are not generally taxable to the employees until withdrawn or distributed from the 401(k) plan.

Core's named executive officers did not participate in, or earn any benefits under, a nonqualified deferred compensation plan sponsored by Core during the fiscal years ended December 31, 2020 and 2021. Core's board of directors may elect to provide Core's officers and other employees with nonqualified defined contribution or other nonqualified deferred compensation benefits in the future if it determines that doing so is in Core's best interests

Emerging Growth Company

Core is an "emerging growth company," as defined in the JOBS Act. As an emerging growth company it is exempt from certain requirements related to executive compensation, including the requirements to hold a nonbinding advisory vote on executive compensation and to provide information relating to the ratio of total compensation of its chief executive officer to the median of the annual total compensation of all of its employees, each as required by the Investor Protection and Securities Reform Act of 2010, which is part of the Dodd-Frank Act.

Equity Plans

Core believes that its ability to grant equity-based awards is a valuable and necessary compensation tool that aligns the long-term financial interests of its employees, consultants and directors with the financial interests of its stockholders. In addition, Core believes that its ability to grant options, restricted stock units and other equity-based awards helps it to attract, retain and motivate employees, consultants, and directors, and encourages them to devote their best efforts to its business and financial success. The principal features of Core's existing equity incentive plan are summarized below. This summary is qualified in its entirety by reference to the actual text of the plans, which are attached as Exhibits 10.10 through 10.13 to the registration statement of which this prospectus forms a part.

2021 Equity Incentive Plan

Eligibility

Any individual who is an employee of Core or any of its affiliates, or any person who provides services to Core or its affiliates, including members of Core's board of directors, is eligible to receive awards under the 2021 Plan at the discretion of the plan administrator. All of Core's full-time employees, directors and consultants are eligible to receive awards under the 2021 Plan following the Business Combination.

Awards

The 2021 Plan provides for the grant of incentive stock options ("ISOs") within the meaning of Section 422 of the Code to employees, including employees of any parent or subsidiary, and for the grant of nonstatutory stock options ("NSOs"), stock appreciation rights, restricted stock awards, restricted stock unit awards, performance awards and other forms of awards to employees, directors and consultants, including employees and consultants of Core's affiliates.

Authorized Shares

The maximum number of shares of Core common stock that may be issued under the 2021 Plan will not exceed 45,000,000 shares of Core common stock. In addition, the number of shares of Core common stock reserved for issuance under the 2021 Plan will automatically increase on January 1 of each year, beginning on January 1, 2022 and continuing through and including January 1, 2031, in an amount equal to (1) 4% of the total number of shares of Core common stock outstanding on December 31 of the preceding year, or (2) a lesser number of shares of Core common stock determined by Core's board of directors prior to the date of the increase. The maximum number of shares of Core common stock that may be issued upon the exercise of ISOs under the 2021 Plan is 45,000,000 shares.

Shares subject to stock awards granted under the 2021 Plan that expire or terminate without being exercised or otherwise issued in full or that are paid out in cash rather than in shares do not reduce the number of shares available for issuance under the 2021 Plan. Shares withheld under a stock award to satisfy the exercise, strike or purchase price of a stock award or to satisfy a tax withholding obligation do not reduce the number of shares available for issuance under the 2021 Plan. If any shares of Core common stock issued pursuant to a stock award are forfeited back to or repurchased or reacquired by Core (1) because of the failure to vest, (2) to satisfy the exercise, strike or purchase price or (3) to satisfy a tax withholding obligation in connection with an award, the shares that are forfeited, repurchased or reacquired will revert to and again become available for issuance under the 2021 Plan.

Non-Employee Director Compensation Limit

The aggregate value of all compensation granted or paid to any non-employee director with respect to any calendar year, including awards granted and cash fees paid to such non-employee director, will not exceed (1) \$750,000 in total value or (2) if such non-employee director is first appointed or elected to Core's board of directors during such calendar year, \$1,000,000 in total value, in each case, calculating the value of any equity awards based on the grant date fair value of such equity awards for financial reporting purposes and excluding distributions from a deferred compensation program.

Plan Administration

Core's board of directors, or a duly authorized committee thereof, will administer the 2021 Plan and is referred to as the "plan administrator" herein. Core's board of directors may also delegate to one or more of Core's officers the authority to (1) designate employees (other than officers) to receive specified stock awards and (2) determine the number of shares subject to such stock awards. Under the 2021 Plan, Core's board of directors has the authority to determine award recipients, grant dates, the numbers and types of stock awards to be granted, the applicable fair market value, and the provisions of each stock award, including the period of exercisability and the vesting schedule applicable to a stock award.

Under the 2021 Plan, Core's board of directors also generally has the authority to effect, without the approval of stockholders but with the consent of any materially adversely affected participant, (1) the reduction of the exercise, purchase, or strike price of any outstanding option or stock appreciation right; (2) the cancellation of any outstanding option or stock appreciation right and the grant in substitution therefore of other awards, cash, or other consideration; or (3) any other action that is treated as a repricing under generally accepted accounting principles.

Stock Options

ISOs and NSOs are granted under stock option agreements adopted by the plan administrator. The plan administrator determines the exercise price for stock options, within the terms and conditions of the 2021 Plan, provided that the exercise price of a stock option generally cannot be less than 100% of the fair market value of a share of Core common stock on the date of grant. Options granted under the 2021 Plan vest at the rate specified in the stock option agreement as determined by the plan administrator.

The plan administrator determines the term of stock options granted under the 2021 Plan, up to a maximum of 10 years. Unless the terms of an optionholder's stock option agreement provide otherwise or as otherwise provided by the plan administrator, if an optionholder's service relationship with Core or any of Core's affiliates ceases for any reason other than disability, death, or cause, the optionholder may generally exercise any vested options for a period of three months following the cessation of service. This period may be extended in the event that exercise of the option is prohibited by applicable securities laws. Unless the terms of an optionholder's stock option agreement provide otherwise or as otherwise provided by the plan administrator, if an optionholder's service relationship with Core or any of Core's affiliates ceases due to death, or an optionholder dies within a certain period following cessation of service, the optionholder or a beneficiary may generally exercise any vested options for a period of 18 months following the date of death. Unless the terms of an optionholder's stock option agreement provide otherwise or as otherwise provided by the plan administrator, if an optionholder's service relationship with Core or any of Core's affiliates ceases due to disability, the optionholder may generally exercise any vested options for a period of 12 months following the cessation of service. In the event of a termination for cause, options generally terminate upon the termination date. In no event may an option be exercised beyond the expiration of its term.

Acceptable consideration for the purchase of Core common stock issued upon the exercise of a stock option will be determined by the plan administrator and may include (1) cash, check, bank draft or money order, (2) a broker-assisted cashless exercise, (3) the tender of shares of Core common stock previously owned by the optionholder, (4) a net exercise of the option if it is an NSO or (5) other legal consideration approved by the plan administrator.

Unless the plan administrator provides otherwise, options and stock appreciation rights generally are not transferable except by will or the laws of descent and distribution. Subject to approval of the plan administrator or a duly authorized officer, an option may be transferred pursuant to a domestic relations order.

Tax Limitations on ISOs

The aggregate fair market value, determined at the time of grant, of Core common stock with respect to ISOs that are exercisable for the first time by an award holder during any calendar year under all of Core's stock plans may not exceed \$100,000. Options or portions thereof that exceed such limit will generally be treated as NSOs. No ISO may be granted to any person who, at the time of the grant, owns or is deemed to own stock possessing more than 10% of Core's total combined voting power or that of any of Core's parent or subsidiary corporations unless (1) the option exercise price is at least 110% of the fair market value of the stock subject to the option on the date of grant and (2) the term of the ISO does not exceed five years from the date of grant.

Restricted Stock Unit Awards

Restricted stock unit awards are granted under restricted stock unit award agreements adopted by the plan administrator. Restricted stock unit awards may be granted in consideration for any form of legal consideration that may be acceptable to the plan administrator and permissible under applicable law. A restricted stock unit award may be settled by cash, delivery of shares of Core common stock, a combination of cash and shares of Core common stock as determined by the plan administrator, or in any other form of consideration set forth in the restricted stock unit award agreement. Additionally, dividend equivalents may be credited in respect of shares

covered by a restricted stock unit award. Except as otherwise provided in the applicable award agreement or by the plan administrator, restricted stock unit awards that have not vested will be forfeited once the participant's continuous service ends for any reason.

Restricted Stock Awards

Restricted stock awards are granted under restricted stock award agreements adopted by the plan administrator. A restricted stock award may be awarded in consideration for cash, check, bank draft or money order, services to us, or any other form of legal consideration that may be acceptable to the plan administrator and permissible under applicable law. The plan administrator determines the terms and conditions of restricted stock awards, including vesting and forfeiture terms. If a participant's service relationship with Core ends for any reason, Core may receive any or all of the shares of Core common stock held by the participant that have not vested as of the date the participant terminates service with Core through a forfeiture condition or a repurchase right.

Stock Appreciation Rights

Stock appreciation rights are granted under stock appreciation right agreements adopted by the plan administrator. The plan administrator determines the strike price for a stock appreciation right, which generally cannot be less than 100% of the fair market value of a share of Core common stock on the date of grant. A stock appreciation right granted under the 2021 Plan vests at the rate specified in the stock appreciation right agreement as determined by the plan administrator. Stock appreciation rights may be settled in cash or shares of Core common stock or in any other form of payment, as determined by the plan administrator and specified in the stock appreciation right agreement.

The plan administrator determines the term of stock appreciation rights granted under the 2021 Plan, up to a maximum of 10 years. Unless the terms of a participant's stock appreciation rights agreement provide otherwise or as otherwise provided by the plan administrator, if a participant's service relationship with Core or any of its affiliates ceases for any reason other than cause, disability, or death, the participant may generally exercise any vested stock appreciation right for a period of three months following the cessation of service. This period may be further extended in the event that exercise of the stock appreciation right following such a termination of service is prohibited by applicable securities laws. Unless the terms of a participant's stock appreciation rights agreement provide otherwise or as otherwise provided by the plan administrator, if a participant's service relationship with Core or any of its affiliates ceases due to disability or death, or a participant dies within a certain period following cessation of service, the participant or a beneficiary may generally exercise any vested stock appreciation right for a period of 12 months in the event of disability and 18 months in the event of death. In the event of a termination for cause, stock appreciation rights generally terminate immediately upon the occurrence of the event giving rise to the termination of the individual for cause. In no event may a stock appreciation right be exercised beyond the expiration of its term.

Performance Awards

The 2021 Plan permits the grant of performance awards that may be settled in stock, cash or other property. Performance awards may be structured so that the stock or cash will be issued or paid only following the achievement of certain pre-established performance goals during a designated performance period. Performance awards that are settled in cash or other property are not required to be valued in whole or in part by reference to, or otherwise based on, Core common stock.

The performance goals may be based on any measure of performance selected by the plan administrator. The performance goals may be based on company-wide performance or performance of one or more business units, divisions, affiliates or segments and may be either absolute or relative to the performance of one or more comparable companies or the performance of one or more relevant indices. Unless specified otherwise by the

plan administrator when the performance award is granted, the plan administrator will appropriately make adjustments in the method of calculating the attainment of performance goals as follows: (1) to exclude restructuring and/or other nonrecurring charges; (2) to exclude exchange rate effects; (3) to exclude the effects of changes to generally accepted accounting principles; (4) to exclude the effects of any statutory adjustments to corporate tax rates; (5) to exclude the effects of items that are "unusual" in nature or occur "infrequently" as determined under generally accepted accounting principles; (6) to exclude the dilutive effects of acquisitions or joint ventures; (7) to assume that any portion of Core's business which is divested achieved performance objectives at targeted levels during the balance of a performance period following such divestiture; (8) to exclude the effect of any change in the outstanding shares of Core common stock by reason of any stock dividend or split, stock repurchase, reorganization, recapitalization, merger, consolidation, spin-off, combination or exchange of shares or other similar corporate change, or any distributions to holders of Core common stock other than regular cash dividends; (9) to exclude the effects of stock based compensation and the award of bonuses under Core's bonus plans; (10) to exclude costs incurred in connection with potential acquisitions or divestitures that are required to be expensed under generally accepted accounting principles; and (11) to exclude the goodwill and intangible asset impairment charges that are required to be recorded under generally accepted accounting principles. In addition, Core's board of directors may establish or provide for other adjustment items in the award agreement at the time the award is granted or in such other document setting forth the performance goals at the time the performance goals are established.

Other Stock Awards

The plan administrator may grant other awards based in whole or in part by reference to Core common stock. The plan administrator will set the number of shares under the stock award (or cash equivalent) and all other terms and conditions of such awards.

Changes to Capital Structure

In the event there is a specified type of change in the capital structure of Core, such as a stock split, reverse stock split or recapitalization, appropriate adjustments will be made to (1) the class and maximum number of shares reserved for issuance under the 2021 Plan, (2) the class of shares used to determine the number of shares by which the share reserve may increase automatically each year, (3) the class and maximum number of shares that may be issued on the exercise of ISOs and (4) the class and number of shares and exercise price, strike price or purchase price, if applicable, of all outstanding stock awards.

Corporate Transactions

The following applies to stock awards under the 2021 Plan in the event of a corporate transaction (as defined in the 2021 Plan), unless otherwise provided in a participant's stock award agreement or other written agreement with Core or one of its affiliates or unless otherwise expressly provided by the plan administrator at the time of grant.

In the event of a corporate transaction, any stock awards outstanding under the 2021 Plan may be assumed, continued or substituted for by any surviving or acquiring corporation (or its parent company), and any reacquisition or repurchase rights held by Core with respect to the stock award may be assigned to Core's successor (or its parent company). If the surviving or acquiring corporation (or its parent company) does not assume, continue or substitute for such stock awards, then (i) with respect to any such stock awards that are held by participants whose continuous service has not terminated prior to the effective time of the corporate transaction, or current participants, the vesting (and exercisability, if applicable) of such stock awards will be accelerated in full (or, in the case of performance awards with multiple vesting levels depending on the level of performance, vesting will accelerate at 100% of the target level) to a date prior to the effective time of the corporate transaction (contingent upon the effectiveness of the corporate transaction), and such stock awards will terminate if not exercised (if applicable) at or prior to the effective time of the corporate transaction, and any

reacquisition or repurchase rights held by Core with respect to such stock awards will lapse (contingent upon the effectiveness of the corporate transaction), and (ii) any such stock awards that are held by persons other than current participants will terminate if not exercised (if applicable) prior to the effective time of the corporate transaction, except that any reacquisition or repurchase rights held by Core with respect to such stock awards will not terminate and may continue to be exercised notwithstanding the corporate transaction.

In the event a stock award will terminate if not exercised prior to the effective time of a corporate transaction, the plan administrator may provide, in its sole discretion, that the holder of such stock award may not exercise such stock award but instead will receive a payment equal in value to the excess (if any) of (i) the per share amount payable to holders of Core common stock in connection with the corporate transaction, over (ii) any per share exercise price payable by such holder, if applicable.

Plan Amendment or Termination

Core's board of directors has the authority to amend, suspend, or terminate the 2021 Plan at any time, provided that such action does not materially impair the existing rights of any participant without such participant's written consent. Certain material amendments also require approval of the Core stockholders. No ISOs may be granted after the tenth anniversary of the date XPDI's board of directors adopted the 2021 Plan. No stock awards may be granted under the 2021 Plan while it is suspended or after it is terminated.

2018 Omnibus Incentive Plan

Core's 2018 Plan was originally adopted by its board of directors and approved by its stockholders on May 18, 2018, and was amended on August 20, 2018; September 20, 2018; and June 20, 2020. The 2018 Plan provides for the grant of stock options, stock appreciation rights, restricted stock awards, restricted stock unit awards, performance awards, and other awards to eligible employees, non-employee directors and consultants. Awards granted under the 2018 Plan are generally subject to a minimum vesting period of at least one year commencing from the date of grant. The 2018 Plan was assumed by the post-combination company upon the closing of the Business Combination.

Authorized Shares. The maximum number of shares of Core common stock that may be issued under the 2018 Plan is 82,500,000 shares.

Plan Administration. Core's board of directors, or a duly authorized committee thereof, is granted the authority to administer the 2018 Plan and is referred to as the "plan administrator" herein. Core's board of directors may also delegate to one or more of Core's officers the authority to grant awards and/or execute agreements or other documents on behalf of the plan administrator. Under the 2018 Plan, Core's board of directors has the authority to:

- select the eligible individual to whom awards may be granted;
- determine whether and to what extent awards are to be granted to eligible individuals;
- · determine the number of shares of common stock to be covered by each award;
- determine the terms and conditions of awards (including exercise price, purchase price, vesting schedule or acceleration thereof, and forfeiture restrictions or waiver thereof):
- determine the amount of cash to be covered by each award;
- determine whether, to what extent and under what circumstances grants of awards under the 2018 Plan are to operate on a tandem basis and/or in conjunction with or apart from other awards made by the Company outside of the 2018 Plan;
- determine whether and under what circumstances, options may be settled in cash, common stock, and/or restricted stock;

- determine whether an option is an ISO or NSO;
- determine whether to require a participant not to sell or otherwise dispose of shares acquired pursuant to the exercise of an award for a period of time as determined by the plan administrator, in its sole discretion, following the date of the acquisition of such award;
- · modify, extend or renew an award; and
- determine whether, to what extent and under what circumstances to provide loans to participants in order to exercise options under the 2018 Plan.

Under the 2018 Plan, the plan administrator may modify, extend or renew options (provided that the rights of a participant are not reduced without the participant's consent) and accept the surrender of outstanding stock options and authorize the grant of substitute options.

Stock Options. ISOs and NSOs are granted under stock option agreements adopted by the plan administrator. The plan administrator determines the exercise price for stock options, within the terms and conditions of the 2018 Plan, provided that the exercise price of a stock option generally cannot be less than 100% of the fair market value of Core common stock on the date of grant. Options granted under the 2018 Plan vest at the rate specified in the stock option agreement as determined by the plan administrator.

The plan administrator determines the term of stock options granted under the 2018 Plan, up to a maximum of 10 years. Unless the terms of an option holder's stock option agreement provide otherwise, if an option holder's service relationship with Core or any of Core's affiliates ceases due to an involuntary termination without cause or resignation for good reason, the option holder may generally exercise any vested options for a period of ninety (90) days following the cessation of service. If the option holder's service relationship with Core or any of Core's affiliates ceases due to a voluntary resignation, the option holder may generally exercise any vested options for a period of thirty (30) days following the cessation of service. Exercise periods may be extended in the event that exercise of the option is prohibited by applicable securities laws. If an option holder's service relationship with Core or any of Core's affiliates ceases due to death or disability, the option holder or a beneficiary may generally exercise any vested options for a period of one (1) year following the date of death or disability. Provided, however, that in the event of a participant's termination by reason of disability, if the participant dies within such exercise period, all unexercised vested options may be exercised for a period of one (1) year from the date of such death. In the event of a termination for cause or a voluntary termination after the occurrence of an event that would be grounds for a termination for cause, options generally terminate upon the termination date. In no event may an option be exercised beyond the expiration of its term.

Acceptable consideration for the purchase of Core common stock issued upon the exercise of a stock option will be determined by the plan administrator and may include (1) cash, check, bank draft or money order, (2) a broker-assisted cashless exercise, (3) the tender of shares of Core common stock owned by the option holder, (4) a net exercise of the option if it is an NSO, or (5) other legal consideration approved by the plan administrator.

Unless the plan administrator provides otherwise, options generally are not transferable except by will or the laws of descent and distribution.

Tax Limitations on ISOs. The aggregate fair market value, determined at the time of grant, of Core common stock with respect to ISOs that are exercisable for the first time by an award holder during any calendar year under all of Core's stock plans may not exceed \$100,000. Options or portions thereof that exceed such limit will generally be treated as NSOs. No ISO may be granted to any person who, at the time of the grant, owns or is deemed to own stock possessing more than 10% of Core's total combined voting power or that of any of Core's affiliates unless (1) the option exercise price is at least 110% of the fair market value of the stock subject to the option on the date of grant, and (2) the term of the ISO does not exceed five years from the date of grant.

Restricted Stock Unit Awards. Restricted stock unit awards are granted under restricted stock unit award agreements adopted by the plan administrator. A restricted stock unit award may be subject to time-based and/or transaction-based vesting conditions as determined by the plan administrator. The plan administrator may, in its sole discretion, provide for acceleration of such vesting conditions. A restricted stock unit award may be settled as determined by the plan administrator, including by cash, shares, other awards, other property or any combination thereof. Additionally, dividend equivalents may be credited in respect of shares covered by a restricted stock unit award. Upon a participant's termination, except as otherwise provided in the applicable award agreement, restricted stock units still subject to restriction will be forfeited in accordance with the terms and conditions established by the plan administrator.

Restricted Stock Awards. Restricted stock awards are granted under restricted stock award agreements adopted by the plan administrator. The plan administrator determines the terms and conditions of restricted stock awards, including vesting and forfeiture terms. Upon a participant's termination, except as otherwise provided in the applicable award agreement, restricted stock still subject to restriction will be forfeited in accordance with the terms and conditions established by the plan administrator.

Stock Appreciation Rights. Stock appreciation rights are granted under stock appreciation right agreements adopted by the plan administrator. The plan administrator determines the purchase price or strike price for a stock appreciation right, which generally cannot be less than 100% of the fair market value of Core common stock on the date of grant. Stock appreciation rights may be settled in cash or shares of Core common stock.

The plan administrator determines the term of stock appreciation rights granted under the 2018 Plan, up to a maximum of 10 years. If a participant's service relationship with Core or any of Core's affiliates ceases due to an involuntary termination without cause or resignation for good reason, the participant may generally exercise any vested stock appreciation right for a period of ninety (90) days following the cessation of service. If the participant's service relationship with Core or any of Core's affiliates ceases due to a voluntary resignation, any vested stock appreciation right may be exercised for a period of thirty (30) days following the cessation of service. Exercise periods may be further extended in the event that exercise of the stock appreciation right following such a termination of service is prohibited by applicable securities laws. If a participant's service relationship with Core, or any of Core's affiliates, ceases due to disability or death, or a participant dies within a certain period following cessation of service, the participant or a beneficiary may generally exercise any vested stock appreciation right for a period of one (1) year from the date of death or disability. Provided, however, that in the event of a participant's termination by reason of disability, if the participant dies within such exercise period, all unexercised vested stock appreciation rights may be exercised for a period of one (1) year from the date of such death. In the event of a termination for cause or a voluntary termination after the occurrence of an event that would be grounds for a termination for cause, stock appreciation rights generally terminate immediately upon the occurrence of the event giving rise to the termination of the individual for cause. In no event may a stock appreciation right be exercised beyond the expiration of its term.

Performance Awards. The 2018 Plan permits the grant of performance awards that may be settled in stock or cash. Performance awards may be structured so that the stock or cash will be issued or paid only following the achievement of certain pre-established performance goals during a designated performance period. The performance goals may be based on one or more Company, divisional, individual or other performance conditions determined by the plan administrator in its sole discretion.

Other Awards. The plan administrator may grant other stock-based awards based in whole or in part by reference to Core common stock. The plan administrator will set the number of shares under the stock award and all other terms and conditions of such awards. The plan administrator may grant other cash-based awards to eligible individuals in such amounts, on such terms and conditions, and for such consideration, as it shall determine in its sole discretion.

Changes to Capital Structure. In the event there is a specified type of change in Core's capital structure, such as a stock split, recapitalization, then the respective exercise prices for outstanding awards that provide for a

participant elected exercise, the number of shares of Core common stock covered by outstanding awards and the number of share of common stock reserved for issuance under the 2018 Plan shall be appropriately adjusted by the plan administrator to prevent dilution or enlargement of the rights granted to, or available for, participants under the 2018 Plan.

In the event of a Reorganization (as defined in the 2018 Plan) subject to the provisions related to a "Change in Control," as described below, the aggregate number or kind of securities thereafter that may be issued under the 2018 Plan, the number or kind of securities or other property (including cash) to be issued pursuant to awards granted under the 2018 Plan (including as a result of the assumption of the 2018 Plan and the obligations hereunder by a successor entity, as applicable) or the purchase price thereof, shall be appropriately adjusted by the Committee to prevent dilution or enlargement of the rights granted to, or available for, participants under the 2018 Plan.

Change in Control. Awards granted under the 2018 Plan may be subject to acceleration of vesting and exercisability upon or after a change in control as may be provided in the applicable stock award agreement or in any other written agreement between Core or any affiliate and the participant, but in the absence of such provision, no such acceleration will automatically occur. Under the 2018 Plan, a change in control is generally (1) the sale or disposition of all or substantially all of the assets of the Company and its subsidiaries, or (2) if any person or group is or becomes the beneficial owner of more than 50% of the total voting power of the voting equity of the Company.

In the event of a Change in Control, except as otherwise provided in an award agreement, a participant's unvested awards will not vest automatically and will be treated in accordance with one or more of the following methods, as determined by the plan administrator:

- awards may be continued, assumed, or have new rights substituted therefore;
- the plan administrator may provide for the purchase of any awards by the Company or an Affiliate for cash;
- the plan administrator may, in its sole discretion, terminate all outstanding and unexercised options, stock appreciation rights, or any other stock-based award that provides for a Participant elected exercise; and/or
- the plan administrator may, in its sole discretion, provide for accelerated vesting or lapse of restrictions.

Plan Amendment or Termination. Core's board of directors has the authority to amend, suspend, or terminate the 2018 Plan, provided that such action does not materially impair the existing rights of any participant without such participant's written consent. Certain material amendments also require the approval of Core's stockholders.

2021 Employee Stock Purchase Plan

Purpose

The purpose of the ESPP is to provide a means by which eligible employees of Core and certain designated companies may be given an opportunity to purchase shares of Core common stock following the Closing, to assist Core in retaining the services of eligible employees, to secure and retain the services of new employees and to provide incentives for such persons to exert maximum efforts for Core's success.

The Plan includes two components: a 423 Component and a Non-423 Component. Core intends that the 423 Component will qualify as options issued under an "employee stock purchase plan" as that term is defined in Section 423(b) of the Code. Except as otherwise provided in the ESPP or determined by Core's board of directors, the Non-423 Component will operate and be administered in the same manner as the 423 Component.

Share Reserve

The maximum number of shares of Core common stock that may be issued under the ESPP is 1,000,000 shares. Additionally, the number of shares of Core common stock reserved for issuance under the ESPP will automatically increase on January 1 of each year, beginning on January 1, 2022 and continuing through and including January 1, 2031, by the lesser of (1) 1% of the total number of shares of Core common stock outstanding on December 31 of the preceding calendar year, (2) 1,000,000 shares of Core common stock, or (3) such lesser number of shares of Core common stock as determined by Core's board of directors. Shares subject to purchase rights granted under the ESPP that terminate without having been exercised in full will not reduce the number of shares available for issuance under the ESPP.

Administration

Core's board of directors, or a duly authorized committee thereof, will administer the ESPP.

Limitations

Core employees and the employees of any of its designated affiliates will be eligible to participate in the ESPP, provided they may have to satisfy one or more of the following service requirements before participating in the ESPP, as determined by the administrator: (1) customary employment with Core or one of its affiliates for more than 20 hours per week and for five or more months per calendar year or (2) continuous employment with Core or one of its affiliates for a minimum period of time, not to exceed two years, prior to the first date of an offering. In addition, Core's board of directors may also exclude from participation in the ESPP or any offering, employees who are "highly compensated employees" (within the meaning of Section 423(b)(4)(D) of the Code) or a subset of such highly compensated employees. All the 205 employees of Core (as of December 31, 2021) are be eligible to participate in the ESPP following the Closing. An employee may not be granted rights to purchase stock under the ESPP (a) if such employee immediately after the grant would own stock possessing 5% or more of the total combined voting power or value of all classes of Core stock or (b) to the extent that such rights would accrue at a rate that exceeds \$25,000 worth of Core stock for each calendar year that the rights remain outstanding.

The ESPP is intended to qualify as an employee stock purchase plan under Section 423 of the Code. The administrator may specify offerings with a duration of not more than 27 months, and may specify one or more shorter purchase periods within each offering. Each offering will have one or more purchase dates on which shares of Core common stock will be purchased for the employees who are participating in the offering. The administrator, in its discretion, will determine the terms of offerings under the ESPP. The administrator has the discretion to structure an offering so that if the fair market value of a share of Core common stock on any purchase date during the offering period is less than or equal to the fair market value of a share of Core common stock on the first day of the offering period, then that offering will terminate immediately, and the participants in such terminated offering will be automatically enrolled in a new offering that begins immediately after such purchase date.

A participant may not transfer purchase rights under the ESPP other than by will, the laws of descent and distribution, or as otherwise provided under the ESPP.

Payroll Deductions

The ESPP permits participants to purchase shares of Core common stock through payroll deductions of up to 15% of their earnings. Unless otherwise determined by the administrator, the purchase price per share will be 85% of the lower of the fair market value of a share of Core Common Stock on the first day of an offering or on the date of purchase. Participants may end their participation at any time during an offering and will be paid their accrued contributions that have not yet been used to purchase shares, without interest. Participation ends automatically upon termination of employment with Core and its related affiliates.

Withdrawal

Participants may withdraw from an offering by delivering a withdrawal form to Core and terminating their contributions. Such withdrawal may be elected at any time prior to the end of an offering, except as otherwise provided by the administrator. Upon such withdrawal, Core will distribute to the employee such employee's accumulated but unused contributions without interest, and such employee's right to participate in that offering will terminate. However, an employee's withdrawal from an offering does not affect such employee's eligibility to participate in any other offerings under the ESPP

Termination of Employment

A participant's rights under any offering under the ESPP will terminate immediately if the participant either (i) is no longer employed by Core or any of its parent or subsidiary companies (subject to any post-employment participation period required by law) or (ii) is otherwise no longer eligible to participate. In such event, Core will distribute to the participant such participant's accumulated but unused contributions, without interest.

Corporate Transactions

In the event of certain specified significant corporate transactions, such as a merger or change in control, a successor corporation may assume, continue, or substitute each outstanding purchase right. If the successor corporation does not assume, continue, or substitute for the outstanding purchase rights, the offering in progress will be shortened and a new purchase date will be set. The participants' purchase rights will be exercised on the new purchase date and such purchase rights will terminate immediately thereafter.

Amendment and Termination

Core's board of directors has the authority to amend, suspend, or terminate the ESPP, at any time and for any reason, provided certain types of amendments will require the approval of the Core stockholders. Any benefits privileges, entitlements and obligations under any outstanding purchase rights granted before an amendment, suspension or termination of the ESPP will not be materially impaired by any such amendment, suspension or termination except (i) with the consent of the person to whom such purchase rights were granted, (ii) as necessary to facilitate compliance with any laws, listing requirements, or governmental regulations or (iii) as necessary to obtain or maintain favorable tax, listing, or regulatory treatment. The ESPP will remain in effect until terminated by Core's board of directors in accordance with the terms of the ESPP.

Non-Employee Director Compensation

With respect to the years ended December 31, 2020 and 2021, except as set forth below, Core did not have formal arrangements under which its non-employee directors receive compensation for their service on Legacy Core's board of directors or its committees. Legacy Core's policy was to reimburse directors for reasonable and necessary out-of-pocket expenses incurred in connection with attending board and committee meetings or performing other services in their capacities as directors. Other than Matthew Bishop, none of the members of the board of directors of Core received any additional compensation for service as a director for the fiscal year ended December 31, 2020. Other than Matthew Bishop and Jarvis Hollingsworth, none of the members of the board of directors of Core received any additional compensation for service as a director for the fiscal year ended December 31, 2021. In connection with his appointment as Core's Co-Chair in July 2021, Darin Feinstein, who also serves as our Chief Vision Officer, received a retainer of \$60,000.

The following table sets forth information regarding the compensation earned by or paid to Mr. Bishop and Mr. Hollingsworth as directors during the fiscal years ended December 31, 2020 and 2021.

	Stock awards	
Name	(\$)(1)	Total(\$)
Matthew Bishop(2)	95,000(3)	95,000
	921,000(4)	921,000
Jarvis Hollingsworth(5)	4,060,000(6)	4,060,000

- (1) The amount reported represents the aggregate grant date fair value of the RSUs granted during the fiscal year ended December 31, 2020 under the 2018 Plan, computed in accordance with ASC Topic 718. The assumptions used in calculating the grant date fair value of the RSUs reported in this column are set forth in Note 11 to the notes to Legacy Core's consolidated financial statements included elsewhere in this prospectus. These amounts do not reflect the actual economic value that may be realized by the non-employee director. As of December 31, 2020, Mr. Bishop held 625,000 RSUs all of which are subject to forfeiture if the transaction vesting condition of the award is not met on or before March 31, 2024.
- (2) Mr. Bishop was appointed to the board of directors of Legacy Core in March 2020. Prior to that, Mr. Bishop served as Legacy Core's Chief Administrative Officer, in which capacity Mr. Bishop was paid \$133,077 during 2020.
- (3) During the fiscal year ended December 31, 2020, Mr. Bishop was granted 50,000 RSUs in connection with his service on the board of directors of Legacy Core. These RSUs have satisfied the time-based vesting condition on April 1, 2021, but remain subject to the earlier to occur of (i) a change of control event, and (ii) an initial public offering of Legacy Core's equity securities, provided that the recipient remains in continuous service with us through such vesting date.
- (4) During the fiscal year ended December 31, 2021, Mr. Bishop was granted 50,000 RSUs in connection with his service on the board of directors of Legacy Core. One fourth of these RSUs vest on each of the first four anniversaries of April 1, 2021, provided that the recipient remains in continuous service with us through each vesting date, and subject to the earlier to occur of (i) a change of control event, and (ii) an initial public offering of Legacy Core's equity securities.
- (5) Mr. Hollingsworth was appointed to the board of directors of Legacy Core in July 2021.
- (6) During the fiscal year ended December 31, 2021, Mr. Hollingsworth was granted 250,000 RSUs in connection with his service on the board of directors of Legacy Core. One fourth of these RSUs vest on each of the first four anniversaries of July 15, 2021, provided that the recipient remains in continuous service with us through each vesting date, and subject to the earlier to occur of (i) a change of control event, and (ii) an initial public offering of Legacy Core's equity securities.

The board of directors of Core approved the following non-employee director compensation program that became effective upon the Effective Time.

Each non-employee director is eligible to receive annual cash retainers for their service on our board of directors and committees as follows. An eligible director may make a timely election to receive all or a portion of his or her annual cash retainer in the form of shares of Core common stock. In addition, we reimburse reasonable expenses incurred by our non-employee directors in connection with attendance at Board or committee meetings.

Annual Board Service Retainer:

a. All Eligible Directors: \$150,000

b. Lead Director: \$200,000

Annual Committee Chair Service Retainer:

a. Chair of the Audit Committee: \$20,000

b. Chair of the Compensation Committee: \$20,000

- c. Chair of the Corporate Governance and Nominating Committee: \$20,000
- 3. Annual Committee Member Service Retainer (not applicable to Committee Chairs):

a. Member of the Audit Committee: \$10,000

b. Member of the Compensation Committee: \$10,000

c. Member of the Nominating and Corporate Governance Committee: \$10,000

For each non-employee director who is first elected or appointed to the board of directors following the Business Combination, on the date of such director's initial election or appointment to the Board (or, if such date is not a market trading day, the first market trading day thereafter), the director will be automatically, and without further action by the Board or the Compensation Committee of the Board, granted RSUs with respect to shares of Core common stock with an aggregate Fair Market Value (as defined in the Incentive Plan) as of the grant date equal to \$500,000 (the "Initial RSU Grant"). The Initial RSU Grant will vest over a four-year period, with one-fourth of the Initial RSU Grant vesting on each anniversary of the grant date, such that the Initial RSU Grant is fully vested on the fourth anniversary of the date of grant, subject to the director's Continuous Service (a defined in the Incentive Plan) through each such vesting date.

Further, on the date of each annual stockholder meeting of the Company held after the Business Combination, each non-employee director who continues to serve as a non-employee member of the Board following such annual meeting (excluding any director who is first appointed or elected by the Board at the annual meeting) will be automatically, and without further action by the Board or the Compensation Committee of the Board, granted RSUs with respect to shares of Core common stock with an aggregate Fair Market Value as of the grant date equal to \$150,000 (the "Annual RSU Grant"). The Annual RSU Grant will vest in full on the earlier of the date of the following year's annual meeting (or the date immediately prior to the next annual meeting if the non-employee director's service as a director ends at such annual meeting due to the director's failure to be re-elected or the director not standing for re-election), subject in all cases to such director's Continuous Service through such vesting date. No director may receive an Annual RSU Grant in the same fiscal year as the receipt of the Initial RSU Grant.

The foregoing summary is qualified in its entirety by reference to the text of the form of the Non-Employee Director Compensation Policy of Core Scientific, Inc., a copy of which is attached as Exhibit 10.14 to the registration statement of which this prospectus forms a part.

CERTAIN RELATIONSHIPS AND RELATED PARTY TRANSACTIONS

Other than compensation arrangements for our directors and executive officers, which are described elsewhere in this prospectus, below is a description of transactions since January 1, 2019 to which we, XPDI or Legacy Core were a party or will be a party, in which:

- the amounts involved exceeded or will exceed \$120,000; and
- any of our directors, executive officers or holders of more than 5% of our capital stock, or any member of the immediate family of, or person sharing the household with, the foregoing persons, had or will have a direct or indirect material interest.

Certain Relationships and Related Person Transactions — XPDI

Founder Shares

In December 2020, the Sponsor paid \$25,000 to cover for certain offering costs on behalf of XPDI in exchange for issuance of 7,187,500 shares of XPDI's Class B common stock, par value \$0.0001 per share (the "Founder Shares"). In January 2021, the Sponsor transferred 30,000 Founder Shares to each of Paul Dabbar, Paul J. Gaynor, Colleen Sullivan, and Scott Widham, XPDI's independent directors. In February 2021, XPDI effected a share capitalization of 1,437,500 shares of its Class B common stock, resulting in an aggregate of 8,625,000 shares of Class B common stock outstanding. All shares and associated amounts have been retroactively restated to reflect the share capitalization. Pursuant to previously agreed terms, in January 2022, the Sponsor sold an aggregate of 1,552,500 Founder Shares to the anchor investors in connection with the Business Combination for an aggregate purchase price of approximately \$4,500, or approximately \$0.003 per share. Immediately prior to the Effective Time of the Business Combination, each share of XPDI Class B common stock automatically converted into one share of Core common stock.

Pursuant to certain lock-up agreements entered into in connection with the Business Combination and the governance documents of Core, the holders of Founder Shares agreed, subject to limited exceptions, not to transfer, assign or sell any of the Founder Shares until the earlier to occur of: (A) one year after the closing of the Business Combination; and (B) subsequent to the Business Combination (x) if the last reported sale price of the Core common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20-trading days within any 30-trading day period commencing at least 150 days after the closing of the Business Combination or (y) the date on which Core completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of its stockholders having the right to exchange their shares of common stock for cash, securities or other property. Any permitted transferees would be subject to the same restrictions and other agreements of the current holders of the Founder Shares with respect to any Founder Shares.

Private Placement Warrants

The Sponsor and the anchor investors purchased an aggregate of 6,266,667 Private Placement Warrants at a price of \$1.50 per Private Placement Warrant, or approximately \$9.4 million in the aggregate in a private placement that occurred simultaneously with the closing of the IPO of XPDI. Each Private Placement Warrant is exercisable for one whole share of Core common stock at a price of \$11.50 per share.

The purchasers of the Private Placement Warrants agreed, subject to limited exceptions, not to transfer, assign or sell any of their Private Placement Warrants (except to permitted transferees) until 30 days after the closing of the Business Combination.

Related Party Loans

On December 31, 2020, the Sponsor agreed to loan XPDI an aggregate of up to \$300,000 to cover expenses related to the IPO of XPDI pursuant to a promissory note (the "Sponsor Note"). This loan is non-interest bearing

and payable on the earlier of June 30, 2021 or the completion of the IPO of XPDI. As of December 31, 2020, XPDI had not borrowed any amount under the Sponsor Note. As of February 10, 2021, XPDI had borrowed approximately \$90,000 under the Sponsor Note.

In addition, in order to finance transaction costs in connection with a business combination, the Sponsor or an affiliate of the Sponsor, or certain of XPDI's officers and directors may, but are not obligated to, loan XPDI funds as may be required ("Working Capital Loans"). If XPDI completes a business combination, XPDI would repay the Working Capital Loans out of the proceeds of the Trust Account released to XPDI. Otherwise, the Working Capital Loans would be repaid only out of funds held outside the Trust Account. In the event that a business combination does not close, XPDI may use a portion of proceeds held outside the Trust Account to repay the Working Capital Loans but no proceeds held in the Trust Account would be used to repay the Working Capital Loans. The Working Capital Loans would either be repaid upon consummation of a business combination or, at the lender's discretion, up to \$1.5 million of such Working Capital Loans may be convertible into warrants of the post business combination entity at a price of \$1.50 per warrant. The warrants would be identical to the Private Placement Warrants. Except for the foregoing, the terms of such Working Capital Loans, if any, have not been determined and no written agreements exist with respect to such loans. As of the closing of the Business Combination, XPDI had no borrowings under the Working Capital Loans.

Administrative Services Agreement

Commencing on the effective date of the registration statement for the IPO of XPDI through the closing date of the Business Combination, XPDI agreed to pay affiliates of the Sponsor a total of \$20,000 per month for office space and administrative and support services pursuant to an administrative services agreement.

Business Combination Payments

XPDI agreed to make a cash payment to XMS Capital Partners, LLC ("XMS Capital") or its affiliates for any financial advisory, placement agency or other similar investment banking services that XMS Capital or its affiliates provides to XPDI in connection with the Business Combination, and reimburse XMS Capital or its affiliates for any out-of-pocket expenses incurred by them in connection with the performance of such services.

Certain Relationships and Related Person Transactions — Core Scientific

Series A Preferred Stock Financing

From July 2019 to March 2020, Legacy Core issued and sold an aggregate of 4,649,443 shares of its Series A Preferred Stock at a purchase price of \$6.83 per share, for aggregate gross proceeds of \$31,755,696. The table below sets forth the number of shares of Series A Preferred Stock purchased directly or indirectly by Legacy Core related parties:

Stockholder	Series A Preferred Stock	Total Purchase Price
Kevin Turner (1)	146,412	\$ 999,994
Michael Levitt (2)	146,412	\$ 999,994
MPM Life, LLC (3)	124,088	\$ 847,521
The Aber Whitcomb Trust (4)	124,088	\$ 847,521
William & Marilyn Humes Charitable Lead Annuity Trust 2017 (5)	29,283	\$ 200,003
Darin Feinstein (6)	124,088	\$ 847,421

⁽¹⁾ Kevin Turner is Legacy Core's former President and Chief Executive Officer, and a former member of Core's board of directors.

- (2) Michael Levitt is Core's Chief Executive Officer and is the Co-Chair and a member of Core's board of directors. 14,641 shares of Mr. Levitt's Series A Preferred Stock holdings are held indirectly through HKM Investment LLC.
- (3) Matthew Minnis, a member of Core's board of directors, is the managing member of MPM Life, LLC.
- (4) Aber Whitcomb, a former member of Legacy Core's board of directors, is a trustee for The Aber Whitcomb Trust.
- (5) William Humes, Core's former Chief Financial Officer, is a trustee for the William & Marilyn Charitable Lead Annuity Trust 2017.
- (6) Darin Feinstein is an owner of greater than 5% of Core's capital stock through the BCV Entities (as defined below) and is the Co-Chair of Core's board of directors.

Bridge Financings

In March 2020, Legacy Core issued an unsecured promissory note to Kevin Turner (the "First Turner Note"), Legacy Core's former President and Chief Executive Officer and a former member of Legacy Core's board of directors, pursuant to which Mr. Turner agreed to loan Legacy Core \$1,500,000 as bridge financing. In April 2020, Legacy Core issued an unsecured promissory note to Mr. Turner (the "Second Turner Note," collectively with the First Turner Note, "The Turner Notes"), pursuant to which Mr. Turner agreed to loan Legacy Core \$500,000 as bridge financing. The Turner Notes were interest bearing at a 7% rate. In May 2020, upon the execution of a Credit and Guaranty Agreement with American Property Acquisitions, LLC, a wholly owned subsidiary of Legacy Core, the then outstanding balance of \$2,021,479.45 on the Turner Notes was repaid in full.

In addition, in March 2020, Legacy Core issued an unsecured promissory note to Michael Levitt (the "First Levitt Note"), Legacy Core's Chief Executive Officer, Co-Chair of its board of directors, and a member its board of directors, pursuant to which Mr. Levitt agreed to loan Legacy Core \$1,500,000 as bridge financing. In April 2020, Legacy Core issued an unsecured promissory note to Mr. Levitt (the "Second Levitt Note," collectively with the First Levitt Note, "The Levitt Notes"), pursuant to which Mr. Levitt agreed to loan Legacy Core \$500,000 as bridge financing. The Levitt Notes were interest bearing at a 7% rate. In May 2020, upon the execution of a Credit and Guaranty Agreement with American Property Acquisitions I, LLC, an indirect, wholly owned subsidiary of Legacy Core, the then outstanding balance of \$2,021,383.56 on the Levitt Notes was repaid in full.

Warrants

In March 2020, in connection with the bridge financings described above, Legacy Core issued certain warrants to Mr. Turner and Mr. Levitt, each exercisable for 2,000,000 shares of Core common stock at any time for two years following the issuance. The exercise price of Core common stock issuable under such warrants is \$1.34 per share. On January 18, 2022, Legacy Core's board of directors extended the exercise window for these warrants to the latter of January 20, 2023 or the expiration the applicable lock-up period.

Related Party Commercial Relationships

Executive Officers and Directors

In July 2019, Legacy Core entered into an ongoing commercial relationship with one of its customers, Gilley Enterprises, LLC ("Gilley"), where Gilley purchases hosting services from Legacy Core on an ongoing basis. Jeff Pratt, Core's Senior Vice President of Operations and Finance is a shareholder in Gilley. James Cleveland, Legacy Core's Chief Power Officer is a shareholder in Gilley. Pursuant to the arrangement, Legacy Core is paid on average approximately \$41,300 per month in hosting fees.

In October 2019, Legacy Core entered into an ongoing commercial relationship with one of its customers, UnionJack LLC ("UnionJack"), where UnionJack purchases hosting services from Legacy Core on an ongoing

basis. Kevin Turner, Legacy Core's former President and Chief Executive Officer, and a former member of Legacy Core's board of directors, is a shareholder in UnionJack. Matthew Bishop, a member of Legacy Core's board of directors, serves as director of UnionJack. Pursuant to the arrangement, Legacy Core is paid approximately \$45,000 per month in hosting fees.

In December 2020, Legacy Core entered into hosting services arrangements with a then-customer, Blockcap, before Legacy Core acquired Blockcap on July 30, 2021. Pursuant to the arrangements, Blockcap purchased hosting services from Legacy Core. Darin Feinstein, an owner of greater than 5% of Legacy Core's capital stock and the Co-Chair of Legacy Core's board of directors, held an approximately 12% equity ownership interest in Blockcap. Kevin Turner, Legacy Core's former President and Chief Executive Officer, and a former member of Legacy Core's board of directors, is a shareholder in Blockcap. Matthew Minnis, a member of Core's board of directors, is a shareholder in Blockcap. Michael Levitt, Core's Chief Executive Officer, Co-Chair of its board of directors, and a member of its board of directors, is a shareholder in Blockcap. James Cleveland, Core's Chief Power Officer, is an optionholder in Blockcap. Jeff Pratt, Core's Senior Vice President of Operations and Finance, is an optionholder in Blockcap. Sharon Orlopp, Core's Chief Human Resources Officer, is an optionholder in Blockcap. Russel Cann, Core's Chief Customer Success Officer, is an optionholder in Blockcap. Taras Kulyk, Core's Senior Vice President of Blockchain Business Development, is an optionholder in Blockcap. Pursuant to the arrangement, Legacy Core was paid on average approximately \$2,000,000 per month in hosting fees.

In May 2021, Legacy Core entered into an ongoing commercial relationship with one of its customers, GEM Mining 2 LLC ("GEM 2"), where GEM 2 purchases hosting services from Legacy Core on an ongoing basis. Russell Cann, Core's Chief Customer Success Officer, is the Managing Member of GEM 2. Pursuant to the arrangement, Legacy Core is paid on average approximately \$76,800 per month in hosting fees.

BCV Entities

Darin Feinstein, the Co-Chair of Core's board of directors, owns greater than 5% of Core's capital stock through BCV 55 LLC, BCV 66 LLC and BCV 77 LLC (collectively, the "BCV Entities"), in which Mr. Feinstein is the leading shareholder. Kevin Turner, Core's former President and Chief Executive Officer, and a former member of Legacy Core's board of directors, is a shareholder in BCV Entities. Michael Levitt, Core's Chief Executive Officer, Co-Chair of its board of directors, and a member its board of directors, is also a shareholder in BCV Entities through HKM Investment LLC and OYM LLC. Matthew Minnis, a member of Core's board of directors, is a shareholder in BCV Entities through MPM Life, LLC. Russel Cann, Core's Chief Customer Success Officer, is a shareholder in BCV entities. Aber Whitcomb, a former member of Legacy Core's board of directors, is a shareholder in BCV Entities. Entities owned, controlled or affiliated with the BCV Entities entered into the following commercial relationships with Legacy Core since January 1, 2019:

In February 2020, Legacy Core entered into a commercial relationship with one of its customers, RME Black 88, LLC ("RME Black 88"), where RME Black 88 purchases hosting services from Legacy Core on an ongoing basis. BCV Entities has an indirect material interest in RME Black 88, as Darin Feinstein, the Managing Member of RME Black 88 is the leading shareholder in BCV Entities. Prior to RME Black 88 entering into an assignment and assumption agreement with Blockcap, Legacy Core was paid on average approximately \$150,000 per month in hosting fees pursuant to the relationship.

In April 2020, Legacy Core entered into a commercial relationship with one of its customers, RME Black 100, LLC ("RME Black 100"), where RME Black 100 purchases hosting services from Legacy Core on an ongoing basis. BCV Entities has an indirect material interest in RME Black 100, as Darin Feinstein, the Managing Member of RME Black 100 is the leading shareholder in BCV Entities. Prior to RME Black 100 entering into an assignment and assumption agreement with Blockcap, Legacy Core was paid on average approximately \$440,000 per month in hosting fees pursuant to the arrangement.

In May 2020, Legacy Core entered into a commercial relationship with one of its customers, RME Black 200, LLC ("RME Black 200"), where RME Black 200 purchases hosting services from Legacy Core on an ongoing basis. BCV Entities has an indirect material interest in RME Black 200, as Darin Feinstein, the Managing Member of RME Black 100 is the leading shareholder in BCV Entities. Prior to RME Black 200 entering into an assignment and assumption agreement with Blockcap, Legacy Core was paid on average approximately \$440,000 per month in hosting fees pursuant to the arrangement.

In August 2020, Legacy Core entered into a commercial relationship with one of its customers, BEP 888, LLC ("BEP 888"), where BEP 888 purchases hosting services from Legacy Core on an ongoing basis. Matt Minnis, an indirect shareholder in BCV Entities, serves as the General Partner of BEP 888. Prior to BEP 888 entering into an assignment and assumption agreement with Blockcap, Legacy Core was paid on average approximately \$29,000 per month in hosting fees pursuant to the arrangement.

The BCV Entities were dissolved in January 2022 prior to the closing of the Business Combination, as a result of which the BCV Entities distributed their interests in Legacy Core to respective members of the BCV Entities, including Darin Feinstein, Chief Vision Officer and Co-Chair of the board of directors of Core, Michael Levitt, Chief Executive Officer and Co-Chair of the board of directors of Core, and Matt Minnis, a member of the board of directors of Core.

Core/Blockcap Merger

On July 15, 2021, Legacy Core entered into an Agreement and Plan of Merger (the "Core/Blockcap merger agreement") by and among (i) Legacy Core; (ii) Block Merger Sub, Inc., a Delaware corporation and direct, wholly owned subsidiary of Legacy Core ("Merger Sub"); (iii) Blockcap; and (iv) Harlin Dean, solely in his capacity as the representative of the Blockcap stockholders, pursuant to which, among other things, upon the terms and subject to the conditions set forth in the Core/Blockcap merger agreement, Merger Sub will merge with and into Blockcap (the "Core/Blockcap merger"), with Blockcap surviving the Core/Blockcap merger as a wholly owned subsidiary of Legacy Core. Assuming that the mergers occurs, following the mergers and as part of a single integrated transaction with the Core/Blockcap merger that is intended to qualify as a "reorganization" within the meaning of Section 368(a) of the Code, Blockcap, as the surviving corporation in the Core/Blockcap merger, will be merged into a limited liability company that is a direct wholly owned subsidiary of, and for U.S. federal income tax purposes is disregarded as an entity separate from, XPDI. The Core/Blockcap merger was consummated on July 30, 2021. Upon consummation of the Core/Blockcap merger, Legacy Core stockholders owned approximately 66.67% of the company formed by combining Legacy Core with Blockcap and former Blockcap stockholders own approximately 33.33% of such combined company.

In connection with the Core/Blockcap merger, Legacy Core issued an aggregate amount of 81,032,304 shares of Legacy Core common stock to Blockcap stockholders in accordance with the exchange ratio set forth in the Core/Blockcap merger agreement. At the effective time of the Core/Blockcap merger (other than any dissenters' shares, Blockcap treasury shares and Blockcap restricted shares) was automatically cancelled and extinguished and collectively converted into the right to receive a number of shares of Core common stock equal to the exchange ratio set forth in the Core/Blockcap merger agreement; (ii) each Blockcap restricted share was converted into a right to receive restricted shares of Core common stock on the same terms and conditions (including applicable vesting conditions) as were set forth in the applicable award agreement pursuant to which such Blockcap restricted share was granted and in effect as of immediately prior to the effective time of the Core/Blockcap merger, in a number equal to the exchange ratio set forth in the Core/Blockcap merger agreement; and (iii) each Blockcap option was assumed by Legacy Core and converted into an option to purchase shares of Legacy Core common stock on the same terms and conditions (including applicable vesting conditions) as were set forth in the applicable award agreement pursuant to which such Blockcap option was granted and in effect as of immediately prior to the effective time of the Core/Blockcap merger on terms set forth in the Core/Blockcap merger agreement.

As of the effective time of the Core/Blockcap merger, (i) Mr. Michael Levitt, Core's Chief Executive officer, Chairman of its board of directors, and a member of its board of directors, indirectly owned approximately 0.07% of Blockcap on a fully-diluted basis, and as such received consideration upon the consummation of the Core/Blockcap merger; and (ii) Mr. Matthew Minnis, a member of Core's board of directors, indirectly owned approximately 6.17% of Blockcap on a fully-diluted basis, and as such received consideration upon the consummation of the Core/Blockcap merger. Certain other officers of Core, Messrs. Cleveland, Pratt, Adams, Cann, Kulyk and Ms. Orlopp were granted options in Blockcap.

Blockcap Secured Demand Promissory Note

On June 30, 2021, Legacy Core issued a Secured Demand Promissory Note (the "Blockcap Note"), in an aggregate principal amount of \$16.2 million to Blockcap for outstanding amounts owed pursuant to certain equipment invoices. The Blockcap Note accrues interest at a rate of 10% compounded annually and was due on July 14, 2021. On July 16, 2021, the Blockcap Note was amended to extend the maturity date to July 23, 2021. The Blockcap Note was secured by the purchased equipment. Darin Feinstein, an owner of greater than 5% of Core's capital stock and the Co-Chair of Core's board of directors, held an approximately 12% equity ownership interest in Blockcap. Kevin Turner, Legacy Core's former President and Chief Executive Officer, and a former member of Legacy Core's board of directors, is a shareholder in Blockcap. Matthew Minnis, a member of Core's board of directors, is a shareholder in Blockcap. James Cleveland, Core's Chief Power Officer, is an optionholder in Blockcap. Jeff Pratt, Core's Senior Vice President of Operations and Finance, is an optionholder in Blockcap. Senior Vice President of Blockcap Revenue Officer and Managing Director of Business Development, is an optionholder in Blockcap. Taras Kulyk, Core's Senior Vice President of Blockchain Business Development, is an optionholder in Blockcap. As of July 30, 2021, the total principal outstanding was approximately \$32.7 million and the amount of accrued interest on the Blockcap Note was approximately \$260,000.

Other Related Party Transactions

Core has entered into employment agreements with its executive officers. For more information regarding employment agreements with Core's named executive officers, see the section titled "Executive Compensation—Employment Agreements with Named Executive Officers."

Core has also granted restricted stock units to its executive officers and directors. For a description of these equity awards, see the section titled "Executive Compensation."

Support Agreements

In connection with the entry into the merger agreement, XPDI, Legacy Core and certain Legacy Core stockholders, including holders affiliated with members of the Legacy Core board of directors and beneficial owners of greater than 5% of Legacy Core's capital stock, entered into certain support agreements, whereby such Legacy Core stockholders agreed to, among other things, vote all of their shares of Legacy Core's capital stock in favor of the approval and adoption of the transactions contemplated by the merger agreement.

Amended and Restated Registration Rights Agreement

In connection with the closing of the Business Combination, Core, the Sponsor, the former independent directors of XPDI and certain of our securityholders entered into an amended and restated registration rights agreement. Pursuant to the agreement, we agreed that we will file with the SEC a registration statement registering the resale of certain securities held by or issuable to such holders, and we will use reasonable best efforts to have such registration statement declared effective as soon as practicable after the filing thereof. In certain circumstances, certain holders can demand up to two underwritten offerings in any 12-month period, and certain holders are entitled to piggyback registration rights.

Lock-Up Restrictions

Pursuant to our Bylaws and certain lock-up agreements entered into prior to the consummation of the Business Combination by and among Core and the stockholders and employees signatories thereto, certain stockholders of Core, including former Legacy Core stockholders and the Sponsor, which held approximately 90.7% and 2.6%, respectively, of our outstanding common stock as of the closing of the Business Combination, have agreed that, with respect to our common stock (including securities convertible into our common stock) held by former Legacy Core stockholders, through the date that is 180 days after the closing of the Business Combination, with respect to the Private Placement Warrants and any of our common stock issuable upon the exercise of the Private Placement Warrants, through the date that is 30 days after the closing of the Business Combination, and, with respect to the Founder Shares, through the date that is one year after the closing of the Business Combination, subject to certain exceptions, to not, without the prior written consent of the our board of directors, among other things, sell, offer to sell, contract or agree to sell, hypothecate, pledge, grant any option to purchase or otherwise dispose of or agree to dispose of, directly or indirectly any shares of our common stock, the Private Placement Warrants, our common stock issuable upon the exercise of the Private Placement Warrants, as applicable, held by the respective parties. Our board of directors can terminate these restrictions at any time and may do so to increase the number of shares eligible for resale on the public market.

Indemnification Agreements

Our Charter contains provisions limiting the liability of executive officers and directors, and our Bylaws provide that we will indemnify each of our executive officers and directors to the fullest extent permitted under Delaware law.

We have entered into indemnification agreements with all of our directors and executive officers. The indemnification agreements provide that we will indemnify each of our directors, executive officers, and other key employees against any and all expenses incurred by such director, executive officer, or other key employee because of his or her status as one of our directors, executive officers, or other key employees, to the fullest extent permitted by Delaware law, the our certificate of incorporation and our bylaws. In addition, the indemnification agreements provide that, to the fullest extent permitted by Delaware law, we will advance all expenses incurred by its directors, executive officers, and other key employees in connection with a legal proceeding involving his or her status as a director, executive officer, or key employee.

Related Person Transactions Policy Following the Business Combination

We have adopted a written related person transactions policy that sets forth our policies and procedures regarding the identification, review, consideration and oversight of "related person transactions." For purposes of the policy only, a "related person transaction" is a transaction, arrangement or relationship (or any series of similar transactions, arrangements or relationships) in which we or any of our subsidiaries are participants involving an amount that exceeds \$120,000, in which any "related person" has a material interest.

Transactions involving compensation for services provided to us as an employee, consultant or director will not be considered related person transactions under this policy. A related person is any officer, director, nominee to become a director, employee or a holder of more than 5% of any class of our voting securities (including the common stock), including any of their immediate family members and affiliates, including entities owned or controlled by such persons.

Under the policy, the related person in question or, in the case of transactions with a holder of more than 5% of any class of our voting securities, an officer with knowledge of a proposed transaction, must present information regarding the proposed related person transaction to the audit committee (or, where review by the audit committee would be inappropriate, to another independent body of the board) for review. To identify related person transactions in advance, we will rely on information supplied by our officers, directors and certain

significant stockholders. In considering related person transactions, the audit committee will take into account the relevant available facts and circumstances, which may include, but are not limited to:

- the risks, costs, and benefits to us;
- the impact on a director's independence in the event the related person is a director, immediate family member of a director or an entity with which a director is affiliated;
- · the terms of the transaction;
- the availability of other sources for comparable services or products; and
- the terms available to or from, as the case may be, unrelated third parties.

The audit committee will approve only those transactions that it determines are fair to us and in our best interests. All of the transactions described above were entered into prior to the adoption of such policy.

PRINCIPAL STOCKHOLDERS

The following table sets forth information regarding the beneficial ownership of shares of our common stock as of January 19, 2022 by:

- each person known by us to be the beneficial owner of more than 5% of our common stock;
- each of our executive officers and directors; and
- all of our executive officers and directors as a group.

Beneficial ownership is determined according to the rules of the SEC, which generally provide that a person has beneficial ownership of a security if he, she or it possesses sole or shared voting or investment power over that security, including options and warrants that are currently exercisable or exercisable within 60 days.

This table is based upon information supplied by officers, directors and principal stockholders and Schedules 13G or 13D filed with the SEC. Unless otherwise indicated in the footnotes to this table and subject to community property laws where applicable, we believe that all persons named in the table have sole voting and investment power with respect to all shares of our common stock beneficially owned by them. Applicable percentages are based on 317,279,900 shares of common stock outstanding as of January 19, 2022, adjusted as required by rules promulgated by the SEC.

Name and Address of Beneficial Owner(1)	Number of Shares	Percentage of Ownership
Named Executive Officers and Directors		
Michael Levitt	22,055,357(2)	6.9%
Chief Executive Officer and Co-Chair of the Board of		
Directors		
B. Kevin Turner	8,281,826(3)	2.6%
Former President and Chief Executive Officer		
Darin Feinstein	40,362,498(4)	12.6%
Chief Vision Officer and Co-Chair of the Board of Directors		
Michael Trzupek	_	_
Executive Vice President and Chief Financial Officer		
Todd M. DuChene	_	_
Executive Vice President, General Counsel, Chief		
Compliance Officer and Secretary		
Jarvis Hollingsworth	_	_
Director		0.504
Matt Minnis	30,711,805(5)	9.6%
Director		
Stacie Olivares	_	_
Director		
Kneeland Youngblood	_	_
Director		
All current directors and executive officers as a group (8	02 120 ((0	20.50/
individuals)	93,129,660	28.5%

⁽¹⁾ Unless otherwise indicated, the business address of each of the directors, executive officers and five percent holders of Core is 106 East 6th Street, Suite 900-145, Austin, Texas 78701.

- (2) Consists of (i) 210,854 shares of Core common stock held of record by Mr. Levitt, (ii) 41,470 shares of Core common stock held of record by HKM Investment LLC ("HKM"), (iii) 869,110 shares of Core common stock held of record by The MJL 2012 Younger Children Trust, modified as of March 21, 2021, (iv) 869,110 shares of Core common stock held of record by The MJL 2012 Older Children Trust, modified as of March 21, 2021, (v) 3,835,368 shares of Core common stock held of record by The CS 1219 Trust, dated April 13, 2017, (vi) 10,629,668 shares of Core common stock held of record by The MJL Revocable Trust, modified as of June 18, 2021, (vii) 800,210 shares of Core common stock held of record by The NBL Revocable Trust, modified as of June 18, 2021, (viii) 3,199,414 shares of Core common stock held of record by MJL Blockchain LLC ("MJL Blockchain") and (ix) 1,600,153 shares of Core common stock issuable upon exercise of warrants within 60 days of the Closing Date held by Mr. Levitt. Mr. Levitt is the managing member of each of HKM and MJL Blockchain and a trustee of each of (i) The MJL 2012 Younger Children Trust, modified as of March 21, 2021, (iii) The MJL 2012 Older Children Trust, modified as of March 21, 2021, (iii) The CS 1219 Trust, dated April 13, 2017, (iv) The MJL Revocable Trust, modified as of June 18, 2021, and (v) The NBL Revocable Trust, modified as of June 18, 2021.
- (3) Consists of (i) 5,081,520 shares of Core common stock held of record by Mr. Turner and (ii) 3,200,306 shares of Core common stock issuable upon exercise of warrants within 60 days of the Closing Date held by Mr. Turner.
- (4) Consists of (i) 36,965,875 shares of Core common stock held of record by Mr. Feinstein, (ii) 811,918 shares of Core common stock held of record by Red Moon 88, LLC ("Red Moon") and (iii) 2,584,705 shares of Core common stock issuable upon exercise of options within 60 days of the Closing Date held by Mr. Feinstein. Mr. Feinstein is the managing member of Red Moon. The principal business address of Red Moon is 3753 Howard Hughes Pkwy, Suite 200, Las Vegas, NV 89169.
- (5) Consists of (i) 29,111,651 shares of Core common stock held of record by MPM Life, LLC ("MPM") and (ii) 1,600,153 shares of Core common stock issuable upon exercise of warrants within 60 days of the Closing Date held by Mr. Minnis through MPM. Mr. Minnis is the managing member of MPM.

SELLING SECURITYHOLDERS

The selling securityholders may offer and sell, from time to time, any or all of the shares of common stock or Private Placement Warrants being offered for resale by this prospectus, which consists of:

- up to 8,625,000 Founder Shares;
- up to 6,266,667 shares of common stock issuable upon exercise of the Private Placement Warrants;
- up to 91,052,355 shares of common stock (including shares issuable upon the exercise of convertible securities) held by certain affiliates
 of our company; and
- up to 80,228,401 shares of common stock issuable upon conversion of certain Convertible Notes.

Certain of the selling securityholders listed below entered into agreements that restrict the transfer of the shares of our common stock that otherwise may be sold from time to time pursuant to the registration statement of which this prospectus forms part. See the section titled "Certain Relationships and Related Party Transactions—Lock-up Restrictions" for further discussion.

As used in this prospectus, the term "selling securityholders" includes the selling securityholders listed in the table below, together with any additional selling securityholders listed in a subsequent amendment to this prospectus, and their donees, pledgees, assignees, transferees, distributees and successors-in-interest that receive shares in any non-sale transfer after the date of this prospectus.

The following table provides, as of the date of this prospectus, information regarding the beneficial ownership of our common stock of each selling securityholder, the number of shares of common stock that may be sold by each selling securityholder under this prospectus and that each selling securityholder will beneficially own assuming all securities that may be offered pursuant to this prospectus are sold. Because each selling securityholder may dispose of all, none or some portion of their securities, no estimate can be given as to the number of securities that will be beneficially owned by a selling securityholder upon termination of this offering. For purposes of the table below, however, we have assumed that after termination of this offering none of the securities covered by this prospectus will be beneficially owned by the selling securityholders and further assumed that the selling securityholders will not acquire beneficial ownership of any additional securities during the offering. In addition, the selling securityholders may have sold, transferred or otherwise disposed of, or may sell, transfer or otherwise dispose of, at any time and from time to time, our securities in transactions exempt from the registration requirements of the Securities Act after the date on which the information in the table is presented.

Except as set forth in the footnotes below, (i) the following table does not include up to 8,625,000 shares of common stock issuable upon exercise of the Public Warrants and (2) the address of each selling securityholder is 106 East 6th Street, Suite 900-145, Austin, Texas 78701.

Please see the section titled "Plan of Distribution" for further information regarding the stockholders' method of distributing these shares.

		Shares of Common Stock				Warrants to Purchase Common Stock					
<u>Name</u>	Number Beneficially Owned Prior to Offering	Number Registered for Sale Hereby	Number Beneficially Owned After Offering	Percent Owned After Offering	Number Beneficially Owned Prior to Offering	Number Registered For Sale Hereby	Number Beneficially Owner After Offering	Percent Owner After Offering			
Total				0/	/o			0	%		

DESCRIPTION OF CAPITAL STOCK

The following is a summary of the rights of our common stock and preferred stock. This summary is qualified by reference to the complete text of our amended and restated certificate of incorporation and bylaws filed as exhibits to the registration statement of which this prospectus forms a part.

General

Our second amended and restated certificate of incorporation authorizes us to issue up to 10,000,000,000 shares of common stock, \$0.0001 par value per share, and 2,000,000,000 shares of preferred stock, par value \$0.0001 per share. Upon the closing of the Business Combination, there were 317,279,900 shares of common stock issued and outstanding and no shares of preferred stock issued and outstanding.

Common Stock

Voting Power

Except as otherwise required by law or as otherwise provided in any certificate of designation for any series of preferred stock, the holders of Core common stock possess all voting power for the election of our directors and all other matters requiring stockholder action. Holders of Core common stock are entitled to one vote per share on matters to be voted on by stockholders.

Dividends

Subject to preferences that may apply to any shares of preferred stock outstanding at the time, the holders of Core common stock may be entitled to receive dividends out of funds legally available if the board of directors of Core, in its discretion, determines to issue dividends and then only at the times and in the amounts that the board of directors of Core may determine.

Liquidation, Dissolution and Winding Up

In the event of our voluntary or involuntary liquidation, dissolution, distribution of assets or winding-up, the holders of Core common stock will be entitled to receive an equal amount per share of all of our assets of whatever kind available for distribution to stockholders, after the rights of the holders of the preferred stock, if any, have been satisfied.

Preemptive or Other Rights

Our stockholders have no preemptive or other subscription rights and there are no sinking fund or redemption provisions applicable to Core common stock.

Preferred Stock

Under our second amended and restated certificate of incorporation, our board of directors may, without further action by our stockholders, fix the rights, preferences, privileges and restrictions of up to an aggregate of 2,000,000,000 shares of preferred stock in one or more series and authorize their issuance. These rights, preferences and privileges could include dividend rights, conversion rights, voting rights, terms of redemption, liquidation preferences and the number of shares constituting any series or the designation of such series, any or all of which may be greater than the rights of common stock. Any issuance of preferred stock could adversely affect the voting power of holders of common stock and the likelihood that such holders would receive dividend payments and payments on liquidation. In addition, the issuance of preferred stock could have the effect of delaying, deterring or preventing a change of control or other corporate action. We have no present plans to issue any shares of preferred stock.

Stock Awards

Upon the closing of the Business Combination, we assumed stock awards to purchase shares of common stock that were outstanding under Core's 2018 Omnibus Incentive Plan. As of the closing of the Business Combination, 45,000,000 shares of Core common stock were reserved for future issuance under our 2021 Equity Incentive Plan, which amount may be subject to increase from time to time, and no stock awards had been granted pursuant to such plan. For additional information regarding the terms of these plans, see "Executive Compensation—Equity Plans." We intend to file one or more registration statements on Form S-8 with respect to these plans after 60 days from the closing of the Business Combination.

Public Warrants

Unless otherwise indicated, references to "warrant" or "warrants" under this "-Public Warrants" heading are to the Public Warrants.

The warrants were originally issued in registered form under the Warrant Agreement between Continental Stock Transfer & Trust Company, as warrant agent, and XPDI. The Warrant Agreement provides that the terms of the warrants may be amended without the consent of any holder to cure any ambiguity or correct any defective provision, but requires the approval by the holders of at least 65% of the then-outstanding Public Warrants to make any change that adversely affects the interests of the registered holders of the Public Warrants.

The warrants may be exercised upon surrender of the warrant certificate on or prior to the expiration date at the offices of the warrant agent, with the exercise form on the reverse side of the warrant certificate completed and executed as indicated, accompanied by full payment of the exercise price (or on a cashless basis, if applicable), by certified or official bank check payable to Core, for the number of warrants being exercised. The warrantholders do not have the rights or privileges of holders of Core common stock or any voting rights until they exercise their warrants and receive shares of Core common stock. After the issuance of shares of Core common stock upon exercise of the warrants, each holder will be entitled to one (1) vote for each share held of record on all matters to be voted on by stockholders.

Each whole warrant entitles the registered holder to purchase one (1) whole share of Core common stock at a price of \$11.50 per share, subject to adjustment as discussed below, at any time after February 18, 2022, 30 days after the closing of the Business Combination, provided in each case that Core has an effective registration statement under the Securities Act covering the shares of Core common stock issuable upon exercise of the warrants and a current prospectus relating to them is available (or Core permits holders to exercise their warrants on a cashless basis under the circumstances specified in the Warrant Agreement) and such shares are registered, qualified or exempt from registration under the securities, or blue sky, laws of the state of residence of the holder. Pursuant to the Warrant Agreement, a warrantholder may exercise its warrants only for a whole number of shares of Core common stock.

Only whole warrants trade, and no fractional shares of Core common stock will be issued upon exercise of a warrant. This means that only a whole warrant may be exercised at any given time by a warrantholder. If, upon exercise, a holder would be entitled to receive a fractional interest in a share, Core will round down to the nearest whole number of the number of shares of Core common stock to be issued to the holder. Furthermore, no fractional warrants will be used upon the separation of a Core unit. Accordingly, unless you have purchased at least four (4) units issued in the IPO, you will not be able to receive, trade or exercise a Core warrant. The warrants will expire five (5) years after the completion of the merger, at 5:00 p.m., New York City time, or earlier upon redemption or liquidation.

Core is not be obligated to deliver any shares of Core common stock pursuant to the exercise of a warrant and will have no obligation to settle such warrant exercise unless a registration statement under the Securities Act with respect to the shares of Core common stock underlying the warrants is then-effective and a prospectus

relating thereto is current, subject to Core's satisfying Core's obligations described below with respect to registration, or a valid exemption from registration is available. No warrant is exercisable and Core will not be obligated to issue shares of Core common stock upon exercise of a warrant unless Core common stock issuable upon such warrant exercise has been registered, qualified or deemed to be exempt under the securities laws of the state of residence of the registered holder of the warrants. In the event that the conditions in the two immediately preceding sentences are not satisfied with respect to a warrant, the holder of such warrant will not be entitled to exercise such warrant and such warrant may have no value and expire worthless. In no event will Core be required to net cash settle any warrant. In the event that a registration statement is not effective for the exercised warrants, the purchaser of a unit containing such warrant will have paid the full purchase price for the unit solely for the share of Core common stock underlying such unit.

A holder of a warrant may notify Core in writing in the event it elects to be subject to a requirement that such holder will not have the right to exercise such warrant, to the extent that after giving effect to such exercise, such person (together with such person's affiliates), to the warrant agent's actual knowledge, would beneficially own in excess of 4.9% or 9.8% (as specified by the holder) of Core common stock issued and outstanding immediately after giving effect to such exercise.

Core has agreed that as soon as practicable, but in no event later than 20 business days after the closing of the merger, it will use its commercially reasonable efforts to (i) file with the SEC a registration statement for the registration, under the Securities Act, of the Core common stock issuable upon exercise of the warrants; (ii) cause such registration statement to become effective within 60 business days after the closing of the merger; and (iii) maintain the effectiveness of such registration statement, and a current prospectus relating thereto, until the expiration or redemption of the warrants in accordance with the provisions of the Warrant Agreement. If a registration statement covering the issuance of the Core common stock issuable upon exercise of the warrants is not effective by the 60th business day after the closing of the merger, warrantholders may, until such time as there is an effective registration statement and during any period when Core will have failed to maintain an effective registration statement, exercise warrants on a "cashless basis" in accordance with Section 3(a)(9) of the Securities Act, provided that such exemption is available, or another exemption. In addition, if Core's common stock are at the time of any exercise of a warrant not listed on a national securities exchange such that it satisfies the definition of a "covered security" under Section 18(b)(1) of the Securities Act, Core may, at its option, require holders of the Public Warrants who exercise their warrants to do so on a "cashless basis" in accordance with Section 3(a)(9) of the Securities Act and, in the event Core elects to do so, Core will not be required to file or maintain in effect a registration statement, but Core will use its commercially reasonable efforts to register or qualify the shares under applicable blue sky laws to the extent an exemption is not available. In such event, each holder would pay the exercise price by surrendering warrants in exchange for a number of shares of Core common stock equal to the lesser of (A) the quotient obtained by dividing (x) the product of (a) the number of shares of Core common stock underlying the warrants and (b) the excess of the "fair market value" over the exercise price of the warrants by (y) such fair market value and (B) the product of the number of warrants surrendered and 0.361, subject to adjustment. The "fair market value" shall mean the volume weighted average price of the shares of Core common stock for the 10 trading days ending on the trading day prior to the date on which the notice of exercise is received by the warrant agent.

Redemption of Warrants When the Price per Share of Core Common Stock Equals or Exceeds \$18.00.

Once the warrants become exercisable, Core may call the warrants for redemption:

- in whole and not in part;
- at a price of \$0.01 per warrant;
- upon not less than 30 days' prior written notice of redemption to each warrantholder; and
- if, and only if, the last reported sale price of the Core common stock for any 20 trading days within a 30-trading day period ending three (3) business days before Core sends the notice of redemption to the

warrantholders (the "Reference Value") equals or exceeds \$18.00 per share (as adjusted for adjustments to the number of shares issuable upon exercise or the exercise price of a warrant as described under the heading "—*Public Warrants*—*Anti-dilution Adjustments*").

If and when the warrants become redeemable, Core may exercise its redemption right even if it is unable to register or qualify the underlying securities for sale under all applicable state securities laws. However, Core will not redeem the warrants unless an effective registration statement under the Securities Act covering the shares of Core common stock issuable upon exercise of the warrants is effective and a current prospectus relating to those shares of Core common stock is available throughout the 30-day redemption period.

Core established the last of the redemption criteria discussed above to prevent a redemption call unless there is, at the time of the call, a significant premium to the warrant exercise price. If the foregoing conditions are satisfied and Core issues a notice of redemption of the warrants, each warrantholder will be entitled to exercise its warrant prior to the scheduled redemption date. Any such exercise would not be done on a "cashless" basis and would require the exercising warrantholder to pay the exercise price for each warrant being exercised. However, the price of the Core common stock may fall below the \$18.00 redemption trigger price (as adjusted for adjustments to the number of shares issuable upon exercise or the exercise price of a warrant as described under the heading "—Anti-dilution Adjustments") as well as the \$11.50 (for whole shares) warrant exercise price after the redemption notice is issued.

Redemption of Warrants When the Price per Share of Core Common Stock Equals or Exceeds \$10.00.

Once the warrants become exercisable, Core may redeem the outstanding warrants:

- in whole and not in part;
- at \$0.10 per warrant;
- upon a minimum of 30 days' prior written notice of redemption, provided that holders will be able to exercise their warrants on a cashless basis prior to redemption and receive that number of shares determined by reference to the table below, based on the redemption date and the "fair market value" of Core common stock except as otherwise described below;
- if, and only if, the Reference Value equals or exceeds \$10.00 per share (as adjusted for adjustments to the number of shares issuable upon exercise or the exercise price of a warrant as described under the heading "—Public Warrants—Anti-dilution Adjustments"); and
- if the Reference Value is less than \$18.00 per share (as adjusted for adjustments to the number of shares issuable upon exercise or the exercise price of a warrant as described under the heading "—Public Warrants—Anti-dilution Adjustments") the Private Placement Warrants must also be concurrently called for redemption on the same terms (except as described above with respect to a holder's ability to exercise its warrants on a "cashless" basis) as the outstanding Public Warrants, as described above.

Beginning on the date the notice of redemption is given until the warrants are redeemed or exercised, holders may elect to exercise their warrants on a cashless basis. The numbers in the table below represent the number of shares of Core common stock that a warrantholder will receive upon such cashless exercise in connection with a redemption by Core pursuant to this redemption feature, based on the "fair market value" of Core common stock on the corresponding redemption date (assuming holders elect to exercise their warrants and such warrants are not redeemed for \$0.10 per warrant), determined for these purposes based on volume-weighted average price of Core common stock as reported during the 10 trading days immediately following the date on which the notice of redemption is sent to the holders of warrants, and the number of months that the corresponding redemption date precedes the expiration date of the warrants, each as set forth in the table below. Core will provide the warrantholders with the final fair market value no later than one business day after the 10-trading day period described above ends.

The stock prices set forth in the column headings of the table below will be adjusted as of any date on which the number of shares issuable upon exercise of a warrant or the exercise price of a warrant is adjusted as set forth under the heading "—Anti-dilution Adjustments" below. If the number of shares issuable upon exercise of a warrant is adjusted, the adjusted share prices in the column headings will equal the share prices immediately prior to such adjustment, multiplied by a fraction, the numerator of which is the number of shares deliverable upon exercise of a warrant immediately prior to such adjustment and the denominator of which is the number of shares deliverable upon exercise of a warrant as so adjusted. The number of shares in the table below shall be adjusted in the same manner and at the same time as the number of shares issuable upon exercise of a warrant. If the exercise price of a warrant is adjusted (a) in the case of an adjustment pursuant to the fifth paragraph under the heading "—Anti-dilution Adjustments" below, the adjusted share prices in the column headings will equal the unadjusted share price multiplied by a fraction, the numerator of which is the higher of the Market Value and the Newly Issued Price (as defined below) as set forth under the heading "—Anti-dilution Adjustments" and the denominator of which is \$10.00 and (b) in the case of in the case of an adjustment pursuant to the second paragraph under the heading "—Anti-dilution Adjustments" below, the adjusted share prices in the column headings will equal the unadjusted share price less the decrease in the exercise price of a warrant pursuant to such exercise price adjustment.

Redemption Date (period to expiration	Fair Market Value of Core Common Stock								
of warrants)	≤\$10.00	\$11.00	\$12.00	\$13.00	\$14.00	\$15.00	\$16.00	\$17.00	≥\$18.00
60 months	0.261	0.281	0.297	0.311	0.324	0.337	0.348	0.358	0.361
57 months	0.257	0.277	0.294	0.310	0.324	0.337	0.348	0.358	0.361
54 months	0.252	0.272	0.291	0.307	0.322	0.335	0.347	0.357	0.361
51 months	0.246	0.268	0.287	0.304	0.320	0.333	0.346	0.357	0.361
48 months	0.241	0.263	0.283	0.301	0.317	0.332	0.344	0.356	0.361
45 months	0.235	0.258	0.279	0.298	0.315	0.330	0.343	0.356	0.361
42 months	0.228	0.252	0.274	0.294	0.312	0.328	0.342	0.355	0.361
39 months	0.221	0.246	0.269	0.290	0.309	0.325	0.340	0.354	0.361
36 months	0.213	0.239	0.263	0.285	0.305	0.323	0.339	0.353	0.361
33 months	0.205	0.232	0.257	0.280	0.301	0.320	0.337	0.352	0.361
30 months	0.196	0.224	0.250	0.274	0.297	0.316	0.335	0.351	0.361
27 months	0.185	0.214	0.242	0.268	0.291	0.313	0.332	0.350	0.361
24 months	0.173	0.204	0.233	0.260	0.285	0.308	0.329	0.348	0.361
21 months	0.161	0.193	0.223	0.252	0.279	0.304	0.326	0.347	0.361
18 months	0.146	0.179	0.211	0.242	0.271	0.298	0.322	0.345	0.361
15 months	0.130	0.164	0.197	0.230	0.262	0.291	0.317	0.342	0.361
12 months	0.111	0.146	0.181	0.216	0.250	0.282	0.312	0.339	0.361
9 months	0.090	0.125	0.162	0.199	0.237	0.272	0.305	0.336	0.361
6 months	0.065	0.099	0.137	0.178	0.219	0.259	0.296	0.331	0.361
3 months	0.034	0.065	0.104	0.150	0.197	0.243	0.286	0.326	0.361
0 months	_	_	0.042	0.115	0.179	0.233	0.281	0.323	0.361

The exact fair market value and redemption date may not be set forth in the table above, in which case, if the fair market value is between two values in the table or the redemption date is between two redemption dates in the table, the number of shares of Core common stock to be issued for each warrant exercised will be determined by a straight-line interpolation between the number of shares set forth for the higher and lower fair market values and the earlier and later redemption dates, as applicable, based on a 365 or 366-day year, as applicable. For example, if the volume-weighted average price of Core common stock as reported during the 10 trading days immediately following the date on which the notice of redemption is sent to the holders of the warrants is \$11.00 per share, and at such time there are 57 months until the expiration of the warrants, holders may choose to, in connection with this redemption feature, exercise their warrants for 0.277 shares of Core common stock for each whole warrant. For an example where the exact fair market value and redemption date are not as set forth in the

table above, if the volume-weighted average price of Core common stock as reported during the 10 trading days immediately following the date on which the notice of redemption is sent to the holders of the warrants is \$13.50 per share, and at such time there are 38 months until the expiration of the warrants, holders may choose to, in connection with this redemption feature, exercise their warrants for 0.298 shares of Core common stock for each whole warrant. In no event will the warrants be exercisable in connection with this redemption feature for more than 0.361 shares of Core common stock per warrant (subject to adjustment). Finally, as reflected in the table above, if the warrants are out of the money and about to expire, they cannot be exercised on a cashless basis in connection with a redemption by Core pursuant to this redemption feature, since they will not be exercisable for any shares of Core common stock.

This redemption feature is structured to allow for all of the outstanding warrants to be redeemed when the Core common stock is trading at or above \$10.00 per share, which may be at a time when the trading price of the Core common stock is below the exercise price of the warrants. This redemption feature has been established to provide Core with the flexibility to redeem the warrants without the warrants having to reach the \$18.00 per share threshold set forth above under "—*Redemption of Warrants When the Price per Share of Core Common Stock Equals or Exceeds \$18.00.*" Holders choosing to exercise their warrants in connection with a redemption pursuant to this feature will, in effect, receive a number of shares for their warrants based on an option pricing model with a fixed volatility input as of the date the prospectus forming a part of the registration statement filed in connection with XPDI's IPO. This redemption right provides Core with an additional mechanism by which to redeem all of the outstanding warrants, and therefore provides certainty as to Core's capital structure as the warrants would no longer be outstanding and would have been exercised or redeemed. Core will be required to pay the applicable redemption price to warrantholders if it chooses to exercise this redemption right, which will allow Core to quickly proceed with a redemption of the warrants if it determines doing so is in Core's best interest. As such, Core would redeem the warrants in this manner when it believe it is in Core's best interest to update its capital structure to remove the warrants and pay the redemption price to the warrantholders.

Anti-dilution Adjustments.

As stated above, Core can redeem the warrants when the Core common stock is trading at a price starting at \$10.00, which is below the exercise price of \$11.50, because it will provide certainty with respect to Core's capital structure and cash position while providing warrantholders with the opportunity to exercise their warrants on a cashless basis for the applicable number of shares. If Core chooses to redeem the warrants when the Core common stock is trading at a price below the exercise price of the warrants, this could result in the warrantholders receiving fewer shares of Core common stock than they would have received if they had chosen to wait to exercise their warrants for Core common stock if and when such shares of Core common stock were trading at a price higher than the exercise price of \$11.50.

A holder of a warrant may notify Core in writing in the event it elects to be subject to a requirement that such holder will not have the right to exercise such warrant, to the extent that, to the warrant agent's actual knowledge, after giving effect to such exercise, such person (together with such person's affiliates) would beneficially own in excess of 4.9% or 9.8% (or such other amount as a holder may specify) of the shares of Core common stock outstanding immediately after giving effect to such exercise.

If the number of outstanding shares of Core common stock is increased by a stock capitalization or stock dividend paid in shares of Core common stock to all or substantially all holders of Core common stock, or by a split-up of Core common stock or other similar event, then, on the effective date of such stock capitalization or stock dividend, split-up or similar event, the number of shares of Core common stock issuable on exercise of each warrant will be increased in proportion to such increase in the outstanding shares of Core common stock. A rights offering made to all or substantially all holders of Core common stock entitling holders to purchase Core common stock at a price less than the "historical fair market value" (as defined below) will be deemed a stock dividend of a number of shares of Core common stock equal to the product of (i) the number of shares of Core common stock actually sold in such rights offering (or issuable under any other equity securities sold in such

rights offering that are convertible into or exercisable for Core common stock) and (ii) one (1) minus the quotient of (x) the price per share of Core common stock paid in such rights offering and (y) the historical fair market value. For these purposes, (i) if the rights offering is for securities convertible into or exercisable for shares of Core common stock, in determining the price payable for Core common stock, there will be taken into account any consideration received for such rights, as well as any additional amount payable upon exercise or conversion and (ii) "historical fair market value" means the volume-weighted average price of shares of Core common stock as reported during the 10 trading day period ending on the trading day prior to the first date on which the shares of Core common stock trade on the applicable exchange or in the applicable market, regular way, without the right to receive such rights.

In addition, if Core, at any time while the warrants are outstanding and unexpired, pays to all or substantially all of the holders of common stock a dividend or makes a distribution in cash, securities or other assets to all or substantially all of the holders of common stock on account of such shares of common stock (or other securities into which the warrants are convertible), other than (a) as described above, or (b) any cash dividends or cash distributions which, when combined on a per-share basis with all other cash dividends and cash distributions paid on the shares of common stock during the 365-day period ending on the date of declaration of such dividend or distribution does not exceed \$0.50 (as adjusted to appropriately reflect any other adjustments and excluding cash dividends or cash distributions that resulted in an adjustment to the exercise price or to the number of shares of such common stock issuable on exercise of each warrant) but only with respect to the amount of the aggregate cash dividends or cash distributions equal to or less than \$0.50 per share, then the warrant exercise price will be decreased, effective immediately after the effective date of such event, by the amount of cash and/or the fair market value of any securities or other assets paid on each share of Core common stock in respect of such event.

If the number of outstanding shares of Core common stock is decreased by a consolidation, combination, reverse stock split or reclassification of shares of Core common stock or other similar event, then, on the effective date of such consolidation, combination, reverse stock split, reclassification or similar event, the number of shares of Core common stock issuable on exercise of each warrant will be decreased in proportion to such decrease in the number of outstanding shares of Core common stock.

Whenever the number of shares of Core common stock purchasable upon the exercise of the warrants is adjusted, as described above, the warrant exercise price will be adjusted by multiplying the warrant exercise price immediately prior to such adjustment by a fraction (x) the numerator of which will be the number of shares of Core common stock purchasable upon the exercise of the warrants immediately prior to such adjustment and (y) the denominator of which will be the number of shares of Core common stock so purchasable immediately thereafter.

In case of any reclassification or reorganization of the outstanding shares of Core common stock (other than those described above or that solely affects the par value of such shares of Core common stock), or in the case of any merger or consolidation of us with or into another corporation (other than a consolidation or merger in which we are the continuing corporation and that does not result in any reclassification or reorganization of the outstanding shares of Core common stock), or in the case of any sale or conveyance to another corporation or entity of the assets or other property of us as an entirety or substantially as an entirety in connection with which we are dissolved, the holders of the warrants will thereafter have the right to purchase and receive, upon the basis and upon the terms and conditions specified in the warrants and in lieu of the shares of Core common stock immediately theretofore purchasable and receivable upon the exercise of the rights represented thereby, the kind and amount of shares of Core common stock or other securities or property (including cash) receivable upon such reclassification, reorganization, merger or consolidation, or upon a dissolution following any such sale or transfer, that the holder of the warrants would have received if such holder had exercised their warrants immediately prior to such event. However, if such holders were entitled to exercise a right of election as to the kind or amount of securities, cash or other assets receivable upon such consolidation or merger, then the kind and amount of securities, cash or other assets for which each warrant will become exercisable will be deemed to be

the weighted average of the kind and amount received per share by such holders in such consolidation or merger that affirmatively make such election, and if a tender, exchange or redemption offer has been made to and accepted by such holders (other than a tender, exchange or redemption offer made by XPDI in connection with redemption rights held by stockholders of XPDI as provided for in the Existing Charter or as a result of the redemption of shares of Class A Common Stock by XPDI if a proposed initial business combination is presented to the stockholders of XPDI for approval) under circumstances in which, upon completion of such tender or exchange offer, the maker thereof, together with members of any group (within the meaning of Rule 13d-5(b)(1) under the Exchange Act) of which such maker is a part, and together with any affiliate or associate of such maker (within the meaning of Rule 12b-2 under the Exchange Act) and any members of any such group of which any such affiliate or associate is a part, own beneficially (within the meaning of Rule 13d-3 under the Exchange Act) more than 50% of the issued and outstanding shares of Class A Common Stock, the holder of a warrant will be entitled to receive the highest amount of cash, securities or other property to which such holder would actually have been entitled as a stockholder if such warrant holder had exercised the warrant prior to the expiration of such tender or exchange offer, accepted such offer and all of the shares of Class A common stock held by such holder had been purchased pursuant to such tender or exchange offer, subject to adjustment (from and after the consummation of such tender or exchange offer) as nearly equivalent as possible to the adjustments provided for in the Warrant Agreement. If less than 70% of the consideration receivable by the holders of Core common stock in such a transaction is payable in the form of our Core common stock in the successor entity that is listed for trading on a national securities exchange or is quoted in an established over-the-counter market, or is to be so listed for trading or quoted immediately following such event, and if the registered holder of the warrant properly exercises the warrant within 30 days following public disclosure of such transaction, the warrant exercise price will be reduced as specified in the Warrant Agreement based on the Black-Scholes value (as defined in the Warrant Agreement) of the warrant. The purpose of such exercise price reduction is to provide additional value to holders of the warrants when an extraordinary transaction occurs during the exercise period of the warrants pursuant to which the holders of the warrants otherwise do not receive the full potential value of the warrants.

Subject to applicable law, any action, proceeding or claim against Core arising out of or relating in any way to the Warrant Agreement will be brought and enforced in the courts of the State of New York or the United States District Court for the Southern District of New York, and Core irrevocably submits to such jurisdiction, which jurisdiction will be the exclusive forum for any such action, proceeding or claim. This provision applies to claims under the Securities Act but does not apply to claims under the Exchange Act or any claim for which the federal district courts of the United States of America are the sole and exclusive forum.

Private Placement Warrants

The Private Placement Warrants (including the shares of Core common stock issuable upon exercise of the Private Placement Warrants) are not transferable, assignable or salable until 30 days after the completion of the merger (except, pursuant to limited exceptions to Core's officers and directors and other persons or entities affiliated with the initial purchasers of the Private Placement Warrants) and they are not be redeemable by Core so long as they are held by the Sponsor, the anchor investors or their respective permitted transferees (except as otherwise set forth herein). Except as described below, the Private Placement Warrants have terms and provisions that are identical to those of the Public Warrants. If the Private Placement Warrants are held by holders other than the Sponsor, the anchor investors or their respective permitted transferees, the Private Placement Warrants are redeemable by Core in all redemption scenarios and exercisable by the holders on the same basis as the Public Warrants. The Sponsor, the direct anchor investors and their permitted transferees have certain registration rights related to the Private Placement Warrants (including the shares of Core common stock issuable upon exercise of the Private Placement Warrants), as described below.

The Sponsor, or its permitted transferees, have the option to exercise the Private Placement Warrants on a cashless basis. Except as described under "Description of Capital Stock—Public Warrants—Redemption of Warrants When the Price per Share of Core Common Stock Equals or Exceeds \$10.00," if holders of the Private Placement Warrants elect to exercise them on a cashless basis, such holders would pay the exercise price by

surrendering his, her or its warrants in exchange for a number of shares of Core common stock equal to the quotient obtained by dividing (x) the product of (a) the number of shares of Core common stock underlying the warrants, and (b) the excess of the "historical fair market value" (defined below) over the exercise price of the warrants by (y) such historical fair market value. For these purposes, the "historical fair market value" shall mean the average last reported sale price of the Core common stock for the 10 trading days ending on the third trading day prior to the date on which the notice of warrant exercise is sent to the warrant agent. The reason that it was agreed that these warrants will be exercisable on a cashless basis so long as they are held by the Sponsor, the anchor investors or their respective permitted transferees is because it was not known at the time of such agreement whether such holders would be affiliated with Core following the merger. If they remain affiliated with Core, the ability of such holders to sell Core's securities in the open market will be significantly limited. Core expects to have policies in place that restrict insiders from selling its securities except during specific periods of time. Even during such periods of time when insiders will be permitted to sell Core's securities, an insider will be prohibited from trading in Core's securities if he, she or it is in possession of material non-public information. Accordingly, unlike public stockholders who could exercise their warrants and sell the shares of Core common stock received upon such exercise freely in the open market in order to recoup the cost of such exercise, the insiders could be significantly restricted from selling such securities. As a result, we believe that allowing the holders to exercise such warrants on a cashless basis is appropriate.

Core Warrants

At the Effective Time of the Business Combination, warrants to acquire approximately 6,099,462 shares of Legacy Core common stock were assumed by Core and converted into warrants to purchase shares of Core common stock.

Anti-Takeover Provisions of Delaware Law

Special Meetings of Stockholders

The Charter and the Bylaws provide, that special meetings of stockholders may be called only by a majority vote of Core's board of directors, by the Chairman of the board of directors, or by the chief executive officer.

Advance Notice Requirements for Stockholder Proposals and Director Nominations

The Bylaws provide, that stockholders seeking to bring business before an annual meeting of stockholders, or to nominate candidates for election as directors at an annual meeting of stockholders, must provide timely notice of their intent in writing. To be considered timely, a stockholder's notice will need to be received by the company secretary at the principal executive offices not later than the close of business on the 90th day nor earlier than the open of business on the 120th day prior to the first anniversary of the preceding year's annual meeting. Pursuant to Rule 14a-8 of the Exchange Act, proposals seeking inclusion in Core's annual proxy statement must comply with the notice periods contained therein. The Bylaws specify, certain requirements as to the form and content of a stockholders' meeting. These provisions may preclude stockholders from bringing matters before an annual meeting of stockholders or from making nominations for directors at an annual meeting of stockholders.

Authorized but Unissued Shares

Core's authorized but unissued shares of common stock and preferred stock are available for future issuances without stockholder approval and could be utilized for a variety of corporate purposes, including future offerings to raise additional capital, acquisitions and employee benefit plans. The existence of authorized but unissued and unreserved shares of common stock and preferred stock could render more difficult or discourage an attempt to obtain control of Core by means of a proxy contest, tender offer, merger or otherwise.

Choice of Forum

The Charter provides that the Court of Chancery of the State of Delaware (or, if and only if, the Court of Chancery of the State of Delaware lacks subject matter jurisdiction, any state court located within the State of Delaware or, if and only if, all such state courts lack subject matter jurisdiction, the federal district court for the District of Delaware) and any appellate court therefrom shall be the sole and exclusive forum for the following claims or causes of action brought under Delaware statutory or common law: (1) any derivative claim or action brought on Core's behalf; (2) any claim or cause of action asserting a breach of fiduciary duty by any of Core's current or former director, officer or other employee; (3) any claim or cause of action asserting a claim against Core arising out of, or pursuant to, the DGCL, the Charter or the Bylaws; (4) any claim or cause of action seeking to interpret, apply, enforce or determine the validity of the Charter or the Bylaws (including any right, obligation, or remedy thereunder); (5) any claim or cause of action as to which the DGCL confers jurisdiction to the Court of Chancery of the State of Delaware; or (6) any claim or cause of action asserting a claim against Core or any of its directors, officers or other employees, that is governed by the internal affairs doctrine, in all cases to the fullest extent permitted by law and subject to the court having personal jurisdiction over the indispensable parties named as defendants. The aforementioned provision will not apply to claims or causes of action brought to enforce a duty or liability created by the Securities Act, the Exchange Act or any other claim for which the federal courts have exclusive jurisdiction. However, as Section 22 of the Securities Act creates concurrent jurisdiction for federal and state courts over all suits brought to enforce any duty or liability created by the Securities Act, and an investor cannot waive compliance with the federal securities laws and the rules and regulations thereunder, there is uncertainty as to whether a court would enforce such a provision. To prevent having to litigate claims in multiple jurisdictions and the threat of inconsistent or contrary rulings by different courts, among other considerations, the Charter provides that the U.S. federal district courts will be the exclusive forum for resolving any complaint asserting a cause or causes of action arising under the Securities Act, including all causes of action asserted against any defendant to such complaint.

While the Delaware courts have determined that such choice of forum provisions are facially valid, a stockholder may nevertheless seek to bring a claim in a venue other than those designated in the exclusive forum provisions. In such instance, Core would expect to vigorously assert the validity and enforceability of the exclusive forum provisions of the Charter. This may require significant additional costs associated with resolving such action in other jurisdictions and there can be no assurance that the provisions will be enforced by a court in those other jurisdictions.

These exclusive forum provisions may limit a stockholder's ability to bring a claim in a judicial forum that it finds favorable for disputes with Core or its directors, officers, or other employees, which may discourage lawsuits against Core or its directors, officers and other employees. If a court were to find either exclusive-forum provision in the Charter to be inapplicable or unenforceable in an action, Core may incur further significant additional costs associated with resolving the dispute in other jurisdictions, all of which could seriously harm Core's business.

Section 203 of the Delaware General Corporation Law

Core is subject to Section 203 of the DGCL, which prohibits a Delaware corporation from engaging in any business combination with any interested stockholder for a period of three years after the date that such stockholder became an interested stockholder, with the following exceptions:

- before such date, the board of directors of the corporation approved either the business combination or the transaction that resulted in the stockholder becoming an interested stockholder;
- upon completion of the transaction that resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least 85% of the voting stock of the corporation outstanding at the time the transaction began, excluding for purposes of determining the voting stock outstanding (but not the outstanding voting stock owned by the interested stockholder) those shares owned (i) by persons

- who are directors and also officers and (ii) employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer; or
- on or after such date, the business combination is approved by the board of directors and authorized at an annual or special meeting of the stockholders, and not by written consent, by the affirmative vote of at least 66 \(^2/_3\%\) of the outstanding voting stock that is not owned by the interested stockholder.

In general, Section 203 of the DGCL defines a "business combination" to include the following:

- · any merger or consolidation involving the corporation and the interested stockholder;
- any sale, transfer, pledge or other disposition of 10% or more of the assets of the corporation involving the interested stockholder;
- subject to certain exceptions, any transaction that results in the issuance or transfer by the corporation of any stock of the corporation to the interested stockholder;
- any transaction involving the corporation that has the effect of increasing the proportionate share of the stock or any class or series of the corporation beneficially owned by the interested stockholder; and
- the receipt by the interested stockholder of the benefit of any loans, advances, guarantees, pledges or other financial benefits by or through the corporation.

In general, Section 203 of the DGCL defines an "interested stockholder" as an entity or person who, together with the person's affiliates and associates, beneficially owns, or within three years prior to the time of determination of interested stockholder status did own, 15% or more of the outstanding voting stock of the corporation.

The statute could prohibit or delay mergers or other takeover or change in control attempts and, accordingly, may discourage attempts to acquire Core even though such a transaction may offer its stockholders the opportunity to sell their stock at a price above the prevailing market price.

A Delaware corporation may "opt out" of these provisions with an express provision in its certificate of incorporation. Core will not opt out of these provisions, which may as a result, discourage or prevent mergers or other takeover or change of control attempts of it.

Limitation of Liability and Indemnification

See "Management-Limitation on Liability and Indemnification of Directors and Officers."

Transfer Agent and Registrar

The transfer agent and registrar for our securities is Computershare Trust Company, N.A., with offices at 250 Royall Street, Canton, Massachusetts 02021.

MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES

The following discussion is a summary of certain material U.S. federal income tax consequences generally applicable to the ownership and disposition of our common stock and the exercise, disposition, and lapse of our Warrants. The common stock and the Warrants are referred to collectively herein as our securities. All prospective holders of our securities should consult their tax advisors with respect to the U.S. federal, state, local and non-U.S. tax consequences of the ownership and disposition of our securities.

This discussion is not a complete analysis of all potential U.S. federal income tax consequences relating to the ownership and disposition of our securities. This summary is based upon current provisions of the Code, existing U.S. Treasury Regulations promulgated thereunder, published administrative pronouncements and rulings of the IRS, and judicial decisions, all as in effect as of the date of this prospectus. These authorities are subject to change and differing interpretation, possibly with retroactive effect. Any change or differing interpretation could alter the tax consequences to holders of our securities described in this discussion. There can be no assurance that a court or the IRS will not challenge one or more of the tax consequences described herein, and we have not obtained, nor do we intend to obtain, a ruling with respect to the U.S. federal income tax consequences to a holder of our securities.

We assume in this discussion that a holder of our securities holds our securities as a "capital asset" within the meaning of Section 1221 of the Code (generally, property held for investment). This discussion does not address all aspects of U.S. federal income taxation that may be relevant to a particular holder of our securities in light of that holder's individual circumstances, nor does it address the special tax accounting rules under Section 451(b) of the Code, any alternative minimum, Medicare contribution, estate, or gift tax consequences, or any aspects of state, local, or non-U.S. taxes or any non-income tax laws. This discussion also does not address consequences relevant to holders of our securities subject to special tax rules, such as holders that own, or are deemed to own, more than 5% of our capital stock, corporations that accumulate earnings to avoid U.S. federal income tax, tax-exempt organizations (including private foundations), governmental or international organizations, banks, financial institutions, investment funds, insurance companies, brokers, dealers, or traders in securities, commodities, or currencies, persons that have elected to market securities to market, regulated investment companies or real estate investment trusts, persons that have a "functional currency" other than the U.S. dollar, tax-qualified retirement plans, partnerships, S corporations, or other entities or arrangements treated as pass-through entities for U.S. federal income tax purposes (and investors therein), holders who hold or receive our securities pursuant to the exercise of employee stock options or otherwise as compensation, holders holding our securities as part of a hedge, straddle, or other risk reduction strategy, conversion transaction, or other integrated investment, holders deemed to sell our securities under the constructive sale provisions of the Code, passive foreign investment companies, controlled foreign corporations, qualified foreign pension funds, and certain former U.S. citizens or long-te

If a partnership, including any entity or arrangement treated as a partnership for U.S. federal income tax purposes, holds our securities, the U.S. federal income tax treatment of a partner in such partnership generally will depend upon the status of the partner and the activities of the partnership. Such partners and partnerships should consult their tax advisors regarding the tax consequences of the ownership and disposition of our securities.

For purposes of this discussion, a "U.S. Holder" means a beneficial owner of our securities that is, for U.S. federal income tax purposes:

- an individual who is a citizen or resident of the United States;
- a corporation, or an entity treated as a corporation for U.S. federal income tax purposes, created or organized in the United States or under the laws of the United States or of any state thereof or the District of Columbia;

- an estate, the income of which is subject to U.S. federal income tax regardless of its source; or
- a trust if (a) a U.S. court can exercise primary supervision over the trust's administration and one or more U.S. persons have the authority to control all of the trust's substantial decisions or (b) the trust has a valid election in effect under applicable U.S. Treasury Regulations to be treated as a U.S. person.

For purposes of this discussion, a "**non-U.S. Holder**" is a beneficial owner of our securities that is neither a U.S. Holder nor a partnership or an entity or arrangement treated as a partnership for U.S. federal income tax purposes.

Tax Considerations Applicable to U.S. Holders

Taxation of Distributions

If we pay distributions or make constructive distributions (other than certain distributions of our stock or rights to acquire our stock) to U.S. Holders of shares of our common stock, such distributions generally will constitute dividends for U.S. federal income tax purposes to the extent paid or deemed paid from our current or accumulated earnings and profits, as determined under U.S. federal income tax principles. Distributions in excess of our current and accumulated earnings and profits will constitute a return of capital that will be applied against and reduce (but not below zero) the U.S. Holder's adjusted tax basis in our common stock. Any remaining excess will be treated as gain realized on the sale or other disposition of the common stock and will be treated as described under "—Tax Considerations Applicable to U.S. Holders—Gain or Loss on Sale, Taxable Exchange, or Other Taxable Disposition of Common Stock" below.

Dividends we pay to a U.S. Holder that is a taxable corporation generally will qualify for the dividends received deduction if the requisite holding period is satisfied. With certain exceptions (including dividends treated as investment income for purposes of investment interest deduction limitations), and provided certain holding period requirements are met, dividends we pay to a non-corporate U.S. Holder generally will constitute "qualified dividends" that under current law will be subject to tax at long-term capital gains rates. If the holding period requirements are not satisfied, a corporation may not be able to qualify for the dividends received deduction and would have taxable income equal to the entire dividend amount, and non-corporate holders may be subject to tax on such dividend at ordinary income tax rates instead of the preferential rates that apply to qualified dividend income.

Gain or Loss on Sale, Taxable Exchange, or Other Taxable Disposition of Common Stock

A U.S. Holder generally will recognize gain or loss on the sale, taxable exchange, or other taxable disposition of our common stock. Any such gain or loss will be capital gain or loss and will be long-term capital gain or loss if the U.S. Holder's holding period for the common stock disposed of exceeds one year. The amount of gain or loss recognized generally will be equal to the difference between (1) the sum of the amount of cash and the fair market value of any property received in such disposition and (2) the U.S. Holder's adjusted tax basis in its common stock disposed of. A U.S. Holder's adjusted tax basis in its common stock generally will equal the U.S. Holder's acquisition cost for such common stock (or, in the case of common stock received upon exercise of a Warrant, the U.S. Holder's initial basis for such common stock, as discussed under "—Tax Considerations Applicable to U.S. Holders—Exercise of a Warrant" below), less any prior distributions treated as a return of capital. Long-term capital gains recognized by non-corporate U.S. Holder generally are eligible under current law for reduced rates of tax. If the U.S. Holder's holding period for the common stock disposed of is one year or less, any gain on a sale, taxable exchange, or other taxable disposition of our common stock would be subject to short-term capital gain treatment and would be taxed at ordinary income tax rates. The deductibility of capital losses is subject to limitations.

Exercise of a Warrant

Except as discussed below with respect to the cashless exercise of a Warrant, a U.S. Holder generally will not recognize taxable gain or loss upon the exercise of a Warrant for cash. The U.S. Holder's initial tax basis in the share of our common stock received upon exercise of the Warrant generally will be an amount equal to the sum of the U.S. Holder's acquisition cost of the Warrant and the exercise price of such Warrant. It is unclear whether a U.S. Holder's holding period for the common stock received upon exercise of the Warrant would commence on the date of exercise of the Warrant or the day following the date of exercise of the Warrant; however, in either case, the holding period will not include the period during which the U.S. Holder held the Warrants.

In certain circumstances, the Warrants may be exercised on a cashless basis. The U.S. federal income tax treatment of an exercise of a warrant on a cashless basis is not clear and could differ from the consequences described above. It is possible that a cashless exercise could be a taxable event, a non-realization event, or a tax-free recapitalization. U.S. holders are urged to consult their tax advisors as to the consequences of an exercise of a Warrant on a cashless basis, including with respect to their holding period and tax basis in the common stock received upon exercise of the Warrant.

Sale, Exchange, Redemption, or Expiration of a Warrant

Upon a sale, exchange (other than by exercise), redemption, or expiration of a Warrant, a U.S. Holder will recognize taxable gain or loss in an amount equal to the difference between (1) the amount realized upon such disposition and (2) the U.S. Holder's adjusted tax basis in the Warrant. A U.S. Holder's adjusted tax basis in its Warrants generally will equal the U.S. Holder's acquisition cost of the Warrant, increased by the amount of any constructive distributions included in income by such U.S. Holder (as described below under "Tax Considerations Applicable to U.S. Holders—Possible Constructive Distributions"). Such gain or loss generally will be treated as long-term capital gain or loss if the Warrant is held by the U.S. Holder for more than one year at the time of such disposition or expiration.

If a Warrant is allowed to lapse unexercised, a U.S. Holder generally will recognize a capital loss equal to such holder's adjusted tax basis in the Warrant. Any such loss generally will be a capital loss and will be long-term capital loss if the Warrant is held for more than one year. The deductibility of capital losses is subject to certain limitations.

Possible Constructive Distributions

The terms of each Warrant provide for an adjustment to the number of shares of our common stock for which the Warrant may be exercised or to the exercise price of the Warrant in certain events, as discussed in the section of this prospectus captioned "Description of Capital Stock—Warrants." An adjustment which has the effect of preventing dilution generally should not be a taxable event. Nevertheless, a U.S. Holder of Warrants would be treated as receiving a constructive distribution from us if, for example, the adjustment increases the holder's proportionate interest in our assets or earnings and profits (e.g., through an increase in the number of shares of our common stock that would be obtained upon exercise or an adjustment to the exercise price of the Warrant) as a result of a distribution of cash to the holders of our common stock that is taxable to such holders as a distribution. Such constructive distribution would be subject to tax as described above under "Tax Considerations Applicable to U.S. Holders—Taxation of Distributions" in the same manner as if such U.S. Holder received a cash distribution from us on our common stock equal to the fair market value of such increased interest.

Information Reporting and Backup Withholding

In general, information reporting requirements may apply to distributions paid to a U.S. Holder and to the proceeds of a sale or other disposition of our securities, unless the U.S. Holder is an exempt recipient. Backup

withholding may apply to such payments if the U.S. Holder fails to provide a taxpayer identification number (or furnishes an incorrect taxpayer identification number) or a certification of exempt status, or has been notified by the IRS that it is subject to backup withholding (and such notification has not been withdrawn).

Backup withholding is not an additional tax. Any amounts withhold under the backup withholding rules will be allowed as a credit against a U.S. Holder's U.S. federal income tax liability and may entitle such holder to a refund, provided the required information is timely furnished to the IRS. U.S. Holders should consult their tax advisors regarding their qualification for an exemption from backup withholding and the procedures for obtaining such an exemption, as well as any information reporting requirements that may be applicable in their particular circumstances.

Tax Considerations Applicable to Non-U.S. Holders

Taxation of Distributions

In general, any distributions (including constructive distributions) we make to a non-U.S. Holder of our common stock, to the extent paid or deemed paid out of our current or accumulated earnings and profits (as determined under U.S. federal income tax principles), will constitute dividends for U.S. federal income tax purposes and, provided such dividends are not effectively connected with the non-U.S. Holder's conduct of a trade or business within the United States, we will be required to withhold tax from the gross amount of the dividend at a rate of 30%, unless such non-U.S. Holder is eligible for a reduced rate of withholding tax under an applicable income tax treaty and provides proper certification of its eligibility for such reduced rate (usually on an IRS Form W-8BEN or W-8BEN-E, as applicable). In the case of any constructive dividend (as described below under "—*Tax Considerations Applicable to Non-U.S. Holders—Possible Constructive Distributions*"), it is possible that this tax would be withheld from any amount owed to a non-U.S. Holder by the applicable withholding agent, including cash distributions or sale proceeds from Warrants or other property subsequently paid or credited to such holder. Any distribution to constituting a dividend will be treated first as reducing (but not below zero) the non-U.S. Holder's adjusted tax basis in our common stock and, to the extent such distribution exceeds the non-U.S. Holder's adjusted tax basis, as gain realized from the sale or other disposition of the common stock, which will be treated as described under "—*Tax Considerations Applicable to Non-U.S. Holders—Gain on Sale, Taxable Exchange, or Other Taxable Disposition of Common Stock and Warrants*" below. In addition, if we determine that we are likely to be classified as a "United States real property holding corporation" (see the section entitled "—*Tax Considerations Applicable to Non-U.S. Holders—Gain on Sale, Exchange, or Other Taxable Disposition of Common Stock and Warrants*" below), we will withhold

Dividends we pay to a non-U.S. Holder that are effectively connected with such non-U.S. Holder's conduct of a trade or business within the United States (or, if a tax treaty applies, are attributable to a U.S. permanent establishment or fixed place of business maintained by the non-U.S. Holder) generally will not be subject to U.S. withholding tax, provided such non-U.S. Holder complies with certain certification and disclosure requirements (generally by providing an IRS Form W-8ECI). Instead, such dividends generally will be subject to U.S. federal income tax, net of certain deductions, at the same individual or corporate rates applicable to U.S. Holders. If the non-U.S. Holder is a corporation, dividends that are effectively connected income may also be subject to a "branch profits tax" at a rate of 30% (or such lower rate as may be specified by an applicable income tax treaty).

Exercise of a Warrant

The U.S. federal income tax treatment of a non-U.S. Holder's exercise of a Warrant generally will correspond to the U.S. federal income tax treatment of the exercise of a Warrant by a U.S. Holder, as described under "—Tax Considerations Applicable to U.S. Holders—Exercise of a Warrant" above, although to the extent a cashless exercise results in a taxable exchange, the tax consequences to the non-U.S. Holder would be the same as those described below in "—Tax Considerations Applicable to Non-U.S. Holders—Gain on Sale, Exchange, or Other Taxable Disposition of Common Stock and Warrants."

Gain on Sale, Exchange, or Other Taxable Disposition of Common Stock and Warrants

A non-U.S. Holder generally will not be subject to U.S. federal income or withholding tax in respect of gain recognized on a sale, taxable exchange, or other taxable disposition of our common stock or Warrants or an expiration or redemption of our Warrants, unless:

- the gain is effectively connected with the conduct of a trade or business by the non-U.S. Holder within the United States (and, if an
 applicable tax treaty so requires, is attributable to a U.S. permanent establishment or fixed place of business maintained by the non-U.S.
 Holder);
- the non-U.S. Holder is an individual who is present in the United States for 183 days or more in the taxable year of disposition and certain other conditions are met; or
- we are or have been a "United States real property holding corporation" for U.S. federal income tax purposes at any time during the shorter of the five-year period ending on the date of disposition or the period that the non-U.S. Holder held our common stock or Warrants and, in the case where our common stock is regularly traded on an established securities market, (i) the non-U.S. Holder has owned, actually or constructively, more than 5% of our common stock at any time within the relevant period or (ii) provided that our Warrants are regularly traded on an established securities market, the non-U.S. Holder has owned, actually or constructively, more than 5% of our Warrants at any time within the within the relevant period. It is unclear how a non-U.S. Holder's ownership of Warrants will affect the determination of whether the non-U.S. Holder owns more than 5% of our common stock. In addition, special rules may apply in the case of a disposition of our Warrants if our common stock is considered to be regularly traded, but our Warrants are not considered to be regularly traded. There can be no assurance that our common stock or Warrants will or will not be treated as regularly traded on an established securities market for this purpose.

Gain described in the first bullet point above will be subject to tax at generally applicable U.S. federal income tax rates as if the non-U.S. Holder were a U.S. resident. Any gains described in the first bullet point above of a non-U.S. Holder that is a foreign corporation may also be subject to an additional "branch profits tax" at a 30% rate (or lower applicable treaty rate). Gain described in the second bullet point above generally will be subject to a flat 30% U.S. federal income tax. Non-U.S. Holders are urged to consult their tax advisors regarding possible eligibility for benefits under income tax treaties.

If the third bullet point above applies to a non-U.S. Holder and applicable exceptions are not available, gain recognized by such holder on the sale, exchange, or other disposition of our common stock or Warrants, as applicable, will be subject to tax at generally applicable U.S. federal income tax rates. In addition, a buyer of our common stock or Warrants may be required to withhold U.S. income tax at a rate of 15% of the amount realized upon such disposition. We will be classified as a United States real property holding corporation if the fair market value of our "United States real property interests" equals or exceeds 50% of the sum of the fair market value of our worldwide real property interests plus our other assets used or held for use in a trade or business, as determined for U.S. federal income tax purposes. We believe, but cannot provide any assurance, that we currently are not, nor do we expect to become, a United States real property holding corporation during the applicable period. Non-U.S. Holders are urged to consult their tax advisors regarding the application of these rules.

Possible Constructive Distributions

The terms of each Warrant provide for an adjustment to the number of shares of our common stock for which the Warrant may be exercised or to the exercise price of the Warrant in certain events, as discussed in the section of this prospectus captioned "Description of Capital Stock—Warrants." An adjustment that has the effect of preventing dilution generally should not be a taxable event. Nevertheless, a non-U.S. Holder of Warrants would be treated as receiving a constructive distribution from us if, for example, the adjustment increases the holder's proportionate interest in our assets or earnings and profits (e.g., through an increase in the

number of shares of our common stock that would be obtained upon exercise or an adjustment to the exercise price of the Warrant) as a result of a distribution of cash to the holders of our common stock that is taxable to such holders as a distribution. A non-U.S. Holder would be subject to U.S. federal income tax withholding as described above under "Tax Considerations Applicable to Non-U.S. Holders—Taxation of Distributions" in the same manner as if such non-U.S. Holder received a cash distribution from us on our common stock equal to the fair market value of such increased interest.

Foreign Account Tax Compliance Act

Sections 1471 through 1474 of the Code (commonly referred to as the "Foreign Account Tax Compliance Act" or "FATCA") and Treasury Regulations and administrative guidance promulgated thereunder impose a U.S. federal withholding tax of 30% on certain payments paid to a foreign financial institution (as specifically defined by applicable rules) unless such institution enters into an agreement with the U.S. government to withhold on certain payments and to collect and provide to the U.S. tax authorities substantial information regarding U.S. account holders of such institution (which includes certain equity holders of such institution, as well as certain account holders that are foreign entities with U.S. owners). FATCA also generally imposes a U.S. federal withholding tax of 30% on certain payments to a non-financial foreign entity unless such entity provides the withholding agent with either a certification that it does not have any substantial direct or indirect U.S. owners or provides information regarding substantial direct and indirect U.S. owners of the entity. An intergovernmental agreement between the United States and an applicable foreign country may modify these requirements. The withholding tax described above will not apply if the foreign financial institution or non-financial foreign entity otherwise qualifies for an exemption from the rules.

FATCA withholding currently applies to payments of dividends. The U.S. Treasury Department has released proposed regulations which, if finalized in their present form, would eliminate the U.S. federal withholding tax of 30% applicable to the gross proceeds of a disposition of our securities. In its preamble to such proposed regulations, the U.S. Treasury Department stated that taxpayers may generally rely on the proposed regulations until final regulations are issued. Non-U.S. Holders are encouraged to consult with their own tax advisors regarding the possible implications of FATCA on their investment in our securities.

Information Reporting and Backup Withholding.

Information returns will be filed with the IRS in connection with payments of distributions and the proceeds from a sale or other disposition of our securities. A non-U.S. Holder may have to comply with certification procedures to establish that it is not a United States person in order to avoid information reporting and backup withholding requirements. The certification procedures required to claim a reduced rate of withholding under a treaty generally will satisfy the certification requirements necessary to avoid the backup withholding as well. Backup withholding is not an additional tax. The amount of any backup withholding from a payment to a non-U.S. Holder will be allowed as a credit against such holder's U.S. federal income tax liability and may entitle such holder to a refund, provided that the required information is timely furnished to the IRS. Non-U.S. Holders should consult their tax advisors regarding their qualification for an exemption from backup withholding and the procedures for obtaining such an exemption, as well as any information reporting requirements that may be applicable in their particular circumstances.

This summary is for general informational purposes only and is not, and is not intended to be, tax advice. This summary also does not purport to be a complete analysis of the potential tax consequences of the ownership and disposition of our securities. All U.S. Holders and non-U.S. Holders should consult their own tax advisors with respect to all U.S. federal, state, and local and non-U.S. tax consequences of the ownership and disposition of our securities in their particular circumstances.

PLAN OF DISTRIBUTION

We are registering the issuance by us of (i) up to 14,8911,667 shares of common stock, including shares of common stock issuable upon exercise of the Warrants, consisting of (i) 6,266,667 shares of common stock that are issuable upon the exercise of Private Placement Warrants and (ii) 8,625,000 shares of common stock that are issuable upon the exercise of Public Warrants.

We are registering the resale by the selling securityholders named in this prospectus, or their permitted transferees, and aggregate of 186,172,423 shares of common stock, consisting of:

- up to 8,625,000 Founder Shares;
- up to 6,266,667 shares of common stock issuable upon exercise of the Private Placement Warrants;
- up to 91,052,355 shares of common stock (including shares issuable upon the exercise of convertible securities) held by certain affiliates
 of our company; and
- up to 80,228,401 shares of common stock issuable upon conversion of certain Convertible Notes.

We are required to pay all fees and expenses incident to the registration of the securities to be offered and sold pursuant to this prospectus. The selling securityholders will bear all commissions and discounts, if any, attributable to their sale of securities.

We will not receive any of the proceeds from the sale of the securities by the selling securityholders. We will receive proceeds from Warrants exercised in the event that such Warrants are exercised for cash. The aggregate proceeds to the selling securityholders will be the purchase price of the securities less any discounts and commissions borne by the selling securityholders.

The shares of common stock beneficially owned by the selling securityholders covered by this prospectus may be offered and sold from time to time by the selling securityholders. The term "selling securityholders" includes donees, pledgees, transferees or other successors in interest selling securities received after the date of this prospectus from a selling securityholder as a gift, pledge, partnership distribution or other transfer. The selling securityholders will act independently of us in making decisions with respect to the timing, manner and size of each sale. Such sales may be made on one or more exchanges or in the over-the-counter market or otherwise, at prices and under terms then prevailing or at prices related to the then current market price or in negotiated transactions. The selling securityholders may sell their securities by one or more of, or a combination of, the following methods:

- purchases by a broker-dealer as principal and resale by such broker-dealer for its own account pursuant to this prospectus;
- · ordinary brokerage transactions and transactions in which the broker solicits purchasers;
- block trades in which the broker-dealer so engaged will attempt to sell the shares as agent but may position and resell a portion of the block as principal to facilitate the transaction;
- an over-the-counter distribution in accordance with the rules of the Nasdaq Global Select Market;
- through trading plans entered into by a selling securityholder pursuant to Rule 10b5-1 under the Exchange Act, that are in place at the time of an offering pursuant to this prospectus and any applicable prospectus supplement hereto that provide for periodic sales of their securities on the basis of parameters described in such trading plans;
- short sales;
- distribution to employees, members, limited partners or stockholders of the selling securityholders; through the writing or settlement of
 options or other hedging transaction, whether through an options exchange or otherwise;

- by pledge to secured debts and other obligations;
- delayed delivery arrangements;
- to or through underwriters or broker-dealers;
- in "at the market" offerings, as defined in Rule 415 under the Securities Act, at negotiated prices, at prices prevailing at the time of sale or
 at prices related to such prevailing market prices, including sales made directly on a national securities exchange or sales made through a
 market maker other than on an exchange or other similar offerings through sales agents;
- · in privately negotiated transactions;
- in options transactions;
- through a combination of any of the above methods of sale; or
- any other method permitted pursuant to applicable law.

In addition, any securities that qualify for sale pursuant to Rule 144 may be sold under Rule 144 rather than pursuant to this prospectus.

To the extent required, this prospectus may be amended or supplemented from time to time to describe a specific plan of distribution. In connection with distributions of the securities or otherwise, the selling securityholders may enter into hedging transactions with broker-dealers or other financial institutions. In connection with such transactions, broker-dealers or other financial institutions may engage in short sales of the securities in the course of hedging the positions they assume with selling securityholders. The selling securityholders may also sell the securities short and redeliver the securities to close out such short positions. The selling securityholders may also enter into option or other transactions with broker-dealers or other financial institutions which require the delivery to such broker-dealer or other financial institution of securities offered by this prospectus, which securities such broker-dealer or other financial institution may resell pursuant to this prospectus (as supplemented or amended to reflect such transaction). The selling securityholders may also pledge securities to a broker-dealer or other financial institution, and, upon a default, such broker-dealer or other financial institution, may effect sales of the pledged securities pursuant to this prospectus (as supplemented or amended to reflect such transaction).

In effecting sales, broker-dealers or agents engaged by the selling securityholders may arrange for other broker-dealers to participate. Broker-dealers or agents may receive commissions, discounts or concessions from the selling securityholders in amounts to be negotiated immediately prior to the sale.

In offering the securities covered by this prospectus, the selling securityholders and any broker-dealers who execute sales for the selling securityholders may be deemed to be "underwriters" within the meaning of the Securities Act in connection with such sales. Any profits realized by the selling securityholders and the compensation of any broker-dealer may be deemed to be underwriting discounts and commissions.

In order to comply with the securities laws of certain states, if applicable, the securities must be sold in such jurisdictions only through registered or licensed brokers or dealers. In addition, in certain states the securities may not be sold unless they have been registered or qualified for sale in the applicable state or an exemption from the registration or qualification requirement is available and is complied with.

We have advised the selling securityholders that the anti-manipulation rules of Regulation M under the Exchange Act may apply to sales of securities in the market and to the activities of the selling securityholders and their affiliates. In addition, we will make copies of this prospectus available to the selling securityholders for the purpose of satisfying the prospectus delivery requirements of the Securities Act. The selling securityholders may indemnify any broker-dealer that participates in transactions involving the sale of the securities against certain liabilities, including liabilities arising under the Securities Act.

At the time a particular offer of securities is made, if required, a prospectus supplement will be distributed that will set forth the number of securities being offered and the terms of the offering, including the name of any underwriter, dealer or agent, the purchase price paid by any underwriter, any discount, commission and other item constituting compensation, any discount, commission or concession allowed or reallowed or paid to any dealer, and the proposed selling price to the public.

A holder of Warrants may exercise its Warrants in accordance with the Warrant Agreement on or before the expiration date set forth therein by surrendering, at the office of the Warrant Agent, Computershare Trust Company, N.A., the certificate evidencing such Warrant, with the form of election to purchase set forth thereon, properly completed and duly executed, accompanied by full payment of the exercise price and any and all applicable taxes due in connection with the exercise of the Warrant, subject to any applicable provisions relating to cashless exercises in accordance with the Warrant Agreement.

We have agreed to indemnify the selling securityholders against certain liabilities, including liabilities under the Securities Act and state securities laws, relating to the registration of the Warrants or shares offered by this prospectus.

We have agreed with certain of selling securityholders to keep the registration statement of which this prospectus constitutes a part effective until such time as all of the securities covered by this prospectus have been disposed of pursuant to and in accordance with the registration statement or such securities have been withdrawn.

LEGAL MATTERS

The validity of the securities offered hereby will be passed upon for us by Cooley LLP, New York, New York.

EXPERTS

The financial statements of Power & Digital Infrastructure Acquisition Corp. as of December 31, 2020 and for the period from December 29, 2020 (inception) through December 31, 2020 appearing in this prospectus, have been audited by Marcum LLP, independent registered public accounting firm, as set forth in its report thereon (which contains an explanatory paragraph relating to substantial doubt about the ability of Power & Digital Infrastructure Acquisition Corp. to continue as a going concern as described in Note 1 to such financial statements) appearing elsewhere in this prospectus, and are included in reliance on such report given on the authority of such firm as experts in auditing and accounting.

The consolidated financial statements of Core Scientific Holding Co. at December 31, 2020 and 2019 and for each of the two years in the period ended December 31, 2020, appearing in this Prospectus and Registration Statement have been audited by Ernst & Young LLP, independent registered public accounting firm, as set forth in their report thereon appearing elsewhere herein, and are included in reliance upon such report given on the authority of such firm as experts in accounting and auditing.

The audited financial statements of Blockcap, Inc. as of December 31, 2020 and 2019, for the year ended December 31, 2020 and for the period from February 19, 2019 (inception) through December 31, 2019, included in this prospectus have been so included in reliance on a report of Marcum LLP, an independent registered public accounting firm, appearing elsewhere herein and are included in reliance on such report given upon such firm as experts in auditing and accounting.

The audited financial statements of BEP 888, LLC as of November 30, 2020 and for the period from June 1, 2020 (inception) through November 30, 2020, included in this prospectus have been so included in reliance on a report of Marcum LLP, an independent registered public accounting firm, appearing elsewhere herein and are included in reliance on such report given upon such firm as experts in auditing and accounting.

The audited financial statements of BEP 999, LLC as of November 30, 2020 and for the period from November 5, 2020 (inception) through November 30, 2020, included in this prospectus have been so included in reliance on a report of Marcum LLP, an independent registered public accounting firm, appearing elsewhere herein and are included in reliance on such report given upon such firm as experts in auditing and accounting.

The audited financial statements of RME Black 100, LLC as of November 30, 2020 and for the period from April 16, 2020 (inception) through November 30, 2020, included in this prospectus have been so included in reliance on a report of Marcum LLP, an independent registered public accounting firm, appearing elsewhere herein and are included in reliance on such report given upon such firm as experts in auditing and accounting.

The audited financial statements of RME Black 200, LLC as of November 30, 2020 and for the period from April 27, 2020 (inception) through November 30, 2020, included in this prospectus have been so included in reliance on a report of Marcum LLP, an independent registered public accounting firm, appearing elsewhere herein and are included in reliance on such report given upon such firm as experts in auditing and accounting.

CHANGES IN INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

On January 21, 2022, the audit committee of our board of directors (the "Audit Committee") approved the dismissal of Marcum LLP ("Marcum") as our independent registered public accounting firm, effective upon completion of Marcum's audit of our consolidated financial statements as of and for the year ending December 31, 2021, and the issuance of their report thereon (the "Auditor Change Effective Date"). The management of our company communicated the Audit Committee's decision to Marcum on January 21, 2022.

Marcum's report of independent registered public accounting firm dated January 8, 2021, except for the subsequent events disclosed in Note 7, as to which the date is February 11, 2021, on the XPDI consolidated balance sheet as of December 31, 2020, the related consolidated statements of operations, changes in stockholders' equity and cash flows for the period from December 29, 2020 (inception) through December 31, 2020, and the related notes to the financial statements did not contain any adverse opinion or disclaimer of opinion, and were not qualified or modified as to uncertainties, audit scope or accounting principles, except for an explanatory paragraph in such report regarding substantial doubt about our ability to continue as a going concern.

During the period from December 29, 2020 (XPDI's inception) through December 31, 2020 and the subsequent interim period through January 21, 2022, there were no "disagreements" (as such term is defined in Item 304(a)(1)(iv) of Regulation S-K) with Marcum on any matter of accounting principles or practices, financial statement disclosure, or auditing scope or procedures, which disagreements, if not resolved to the satisfaction of Marcum, would have caused Marcum to make reference thereto in its reports on XPDI's financial statements for such periods. During the period from December 29, 2020 (XPDI's inception) through December 31, 2020 and the subsequent interim period through January 21, 2022, there have been no "reportable events" (as such term is defined in Item 304(a)(1)(v) of Regulation S-K), other than the material weakness in internal controls identified by management related to the accounting for a portion of the Class A Common Stock in permanent equity rather than temporary equity, which resulted in the restatement of XPDI's financial statements as set forth in XPDI's Forms 10-Q for the quarters ended March 31, 2021, June 30, 2021 and September 30, 2021 as filed with the SEC on May 25, 2021, August 23, 2021 and November 15, 2021, respectively.

We have provided Marcum with a copy of the foregoing disclosures and requested that Marcum furnish us with a letter addressed to the SEC stating whether it agrees with the above statements and, if not, stating the respects in which it does not agree. A copy of the letter from Marcum dated January 24, 2022 is filed as Exhibit 16.1 to the registration statement of which this prospectus forms a part.

On January 21, 2022, the Audit Committee approved the engagement of Ernst & Young LLP ("EY") as our independent registered public accounting firm, effective upon the Auditor Change Effective Date. EY previously served as the independent registered public accounting firm of Legacy Core prior to the Business Combination. During the period from December 29, 2020 (XPDI's inception) through December 31, 2020 and the subsequent interim period through January 21, 2022, neither we, nor anyone acting on our behalf, consulted with EY regarding the application of accounting principles to a specified transaction (either completed or proposed), the type of audit opinion that might be rendered on our financial statements, or any matter that was either the subject of a "disagreement," as defined in Item 304(a)(1)(iv) of Regulation S-K, or a "reportable event," as defined in Item 304(a)(1)(v) of Regulation S-K.

WHERE YOU CAN FIND MORE INFORMATION

We have filed with the SEC a registration statement on Form S-1 under the Securities Act, with respect to the securities being offered by this prospectus. This prospectus, which constitutes part of the registration statement, does not contain all of the information in the registration statement and its exhibits. For further information with respect to Core and the securities offered by this prospectus, we refer you to the registration

statement and its exhibits. Statements contained in this prospectus as to the contents of any contract or any other document referred to are not necessarily complete, and in each instance, we refer you to the copy of the contract or other document filed as an exhibit to the registration statement. Each of these statements is qualified in all respects by this reference. You can read our SEC filings, including the registration statement, over the internet at the SEC's website at www.sec.gov.

We are subject to the information reporting requirements of the Exchange Act, and we file reports, proxy statements and other information with the SEC. These reports, proxy statements and other information will be available for review at the SEC's website at www.sec.gov. We also maintain a website at www.corescientific.com, at which you may access these materials free of charge as soon as reasonably practicable after they are electronically filed with, or furnished to, the SEC. The information contained in, or that can be accessed through, our website is not part of this prospectus.

UNAUDITED HISTORICAL COMPARATIVE AND PRO FORMA COMBINED PER SHARE DATA OF XPDI AND LEGACY CORE

Presented below is the historical and pro forma per share data for Core, XPDI, Legacy Core and Blockcap for the year ended December 31, 2020 and the nine months ended September 30, 2021. The pro forma earnings per share data for the year ended December 31, 2020 and the nine months ended September 30, 2021 is presented as if the merger had been completed on January 1, 2020. The information provided below is unaudited.

This information should be read together with XPDI's, Legacy Core's and Blockcap's audited financial statements and related notes, the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations," in each case, included elsewhere in the registration statement of which this prospectus forms a part.

The historical per share information of XPDI, Legacy Core and Blockcap was derived from the audited consolidated financial statements of XPDI and Legacy Core respectively, and the audited financial statements of Blockcap, for the year ended December 31, 2020, and from the unaudited consolidated financial statements of XPDI and Legacy Core respectively, as of and for the nine months ended September 30, 2021 included elsewhere in the registration statement of which this prospectus forms a part.

The unaudited pro forma data of Legacy Core combines the historical results of Blockcap's financial statements for the period prior to the acquisition into Legacy Core's historical consolidated financial statements. The unaudited pro forma data then combines the derived pro forma results of Legacy Core with XPDI's consolidated financial statements.

The Core pro forma equivalent per share amount is calculated by dividing the pro forma net income by the Core equivalent weighted average basic and diluted shares. The Core equivalent shares are calculated by adding a) the Legacy Core pro forma combined shares multiplied by the exchange rate of 1.6001528688 and b) the XPDI pro forma combined shares after the redemption of 12,347,077 redeemable shares. See "Unaudited Pro Forma Condensed Combined Financial Information—Note 5".

The pro forma data is presented for illustrative purposes only and is not necessarily indicative of the results of operations or the financial condition that would have occurred if the merger had been completed as of the dates described above.

Unaudited Pro Forma Per Share Data For The Twelve Months Ended December 31, 2020

	December 31, 2020							
	Blockcap Historical	Legacy Legacy Core Core Pro Forma XPDI Historical Combined Historic			Pro Forma Combined Core	Pro Forma Equivalent Core		
Earnings (loss) per share for the twelve months ended December 31, 2020:								
Net income (loss) per share (Basic and Diluted) ⁽¹⁾	\$ (0.56)	\$ (0.23)	(0.60)	\$ 0.00	\$ (0.46)	\$ (0.30)		

⁽¹⁾ Potentially dilutive securities are not included in the calculation of diluted net loss per share because to do so would be anti-dilutive.

Unaudited Pro Forma Per Share Data For The Nine Months Ended September 30, 2021

	Blockcap					
	(Historical)					
	for					
	the period					
	January 1, 2021 through July 30, 2021	Legacy Core Historical	Legacy Core Pro Forma Combined	XPDI Historical	Pro Forma Combined Core	Pro Forma Equivalent Core
Earnings per share for the nine months ended September 30, 2021:						
Net income (loss) per share (Basic and Diluted) ⁽¹⁾	\$ (0.16)	\$ (0.11)	\$ (0.32)	\$ (0.51)	\$ (0.36)	\$ (0.24)

⁽¹⁾ Potentially dilutive securities are not included in the calculation of diluted net loss per share because to do so would be anti-dilutive.

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

The following unaudited pro forma condensed combined financial information is provided to aid you in your analysis of the financial aspects of the Transactions (as defined below). This information should be read together with XPDI's and Legacy Core's financial statements and related notes, the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations" and the other financial information, in each case, contained elsewhere in the registration statement of which this prospectus forms a part. Capitalized terms used but not defined herein have the respective meanings given to them in this prospectus.

The following unaudited pro forma condensed combined financial information presents the combination of the historical financial information of XPDI and Legacy Core, adjusted to give effect to the merger, other events contemplated by the merger agreement and certain recent transactions completed by Legacy Core (collectively, the "Transactions").

The Transactions include:

- The merger, which includes the merger of XPDI and the Merger Sub with Legacy Core;
- Legacy Core's acquisition of Blockcap (described below), including any elimination of the effect of transactions between Legacy Core and Blockcap, as required, is included in the Legacy Core historical financial statements as of and for the nine months ended September 30, 2021. The proforma impact to the Statement of Operations had the acquisition taken place on January 1, 2020 is included as an adjustment for the nine months ended September 30, 2021 and the twelve months ended December 31, 2020.
- The issuance by Legacy Core of secured convertible notes in April, 2021, net of the repayment of the existing Legacy Core loan (the "Silverpeak loan") and the issuance in August and September, 2021 of unsecured convertible notes is included in the Legacy Core historical financial statements as of and for the nine months ended September 30, 2021. The proforma impact to interest expense for the nine months ended September 30, 2021 and the twelve months ended December 31, 2020 is included as an adjustment. Also included as proforma adjustments is the impact of additional issuance of unsecured convertible notes during October and November of 2021.

The unaudited pro forma condensed combined balance sheet as of September 30, 2021 combines the historical condensed consolidated balance sheet of XPDI as of September 30, 2021, the historical balance sheet of Legacy Core as of September 30, 2021, on a pro forma basis as if the Transactions had been consummated on September 30, 2021. The unaudited pro forma condensed combined statement of operations for the year ended December 31, 2020 and the nine months ended September 30, 2021 combine the historical statements of operations of XPDI for the period from December 29, 2020 (inception) through December 31, 2020 and the nine months ended September 30, 2021, respectively, the historical statements of operations of Legacy Core for the year ended December 31, 2020 and the nine months ended September 30, 2021, respectively, and the historical statements of operations of Blockcap for the period from February 19, 2020 (inception) through December 31, 2020 and period from January 1, 2021 through July 30, 2021 (the date of acquisition), respectively, on a pro forma basis as if the Transactions had been consummated on January 1, 2020, the beginning of the earliest period presented.

The unaudited pro forma condensed combined financial information and accompanying notes have been derived from and should be read in conjunction with:

- the historical unaudited condensed consolidated financial statements of XPDI as of and for the nine months ended September 30, 2021 and the related notes, which are included elsewhere in the registration statement of which this prospectus forms a part;
- the historical audited financial statements of XPDI as of and for the period from December 29, 2020 (inception) through December 31, 2020 and the related notes, which are included elsewhere in the registration statement of which this prospectus forms a part;

- the historical unaudited financial statements of Legacy Core as of and for the nine months ended September 30, 2021 and the related notes, which are included elsewhere in the registration statement of which this prospectus forms a part;
- the historical audited financial statements of Legacy Core as of and for the year ended December 31, 2020 and the related notes, which are included elsewhere in the registration statement of which this prospectus forms a part;
- the historical unaudited financial statements of Blockcap for the period from January 1, 2021 through the acquisition date of July 30, 2021;
- the historical audited financial statements of Blockcap as of and for the year ended December 31, 2020 and the related notes, which are included elsewhere in the registration statement of which this prospectus forms a part;
- the historical audited financial statements of RME Black 88, LLC ("RME 88"), RME Black 100 LLC ("RME 100"), RME Black 200 ("RME 200") and BEP 888 LLC ("BEP 888") as of and for the periods from inception through December 31, 2020 and the related notes, which are included elsewhere in the registration statement of which this prospectus forms a part; and
- other information relating to XPDI and Legacy Core contained in the registration statement of which this prospectus forms a part.

Description of the Merger

On July 20, 2021, XPDI, the Merger Sub and Legacy Core entered into the merger agreement pursuant to which Merger Sub has merged with and into Legacy Core, with Legacy Core surviving the First Merger as a wholly owned subsidiary of XPDI, and promptly following the First Merger, Legacy Core merged with and into XPDI, with XPDI surviving the Second Merger. Upon the consummation of the merger, the name of the surviving entity is "Core Scientific, Inc."

Immediately prior to the Effective Time, each share of Legacy Core preferred stock issued and outstanding automatically converted into a number of shares of Legacy Core common stock in accordance with the Certificates of Designation for Legacy Core's Series A Preferred Stock and Series B Preferred Stock, in each case, upon receipt of the requisite vote of holders with respect to each respective class of preferred stock as set forth in each such Certificate of Designation.

At the Effective Time, each share of Legacy Core common stock (including, for the avoidance of doubt, common stock issued as a result of the conversion of Legacy Core preferred stock in connection with the merger and any shares of Legacy Core common stock that were issued as a result of any deemed exercise of an option to purchase Legacy Core common stock held by a former employee or service provider of Legacy Core, except as set forth in the merger agreement) that was issued and outstanding immediately prior to the Effective Time (other than shares held in treasury and dissenting shares) was cancelled and extinguished and collectively converted into the right to receive the applicable portion of the merger consideration, in accordance with the Allocation Schedule, consisting of a number of shares of Class A Common Stock equal to the Exchange Ratio. Each share of Legacy Core common stock held in treasury immediately prior to the Effective Time was cancelled and extinguished, and no consideration was paid or is payable with respect thereto. Each issued and outstanding share of common stock of Merger Sub was converted into and became one share of common stock of the surviving corporation of the First Merger. By virtue of the Second Merger, each share of common stock of the surviving corporation of the First Merger was converted into and became one common membership unit of the surviving entity of the Second Merger.

At the Effective Time, each warrant to purchase Legacy Core common stock held by a former employee or service provider that was issued, outstanding and unexercised immediately prior to the Effective Time was

deemed to have been exercised, on a net exercise basis with respect to the applicable exercise price and any required withholding or employment taxes thereon, immediately prior to the Closing and was settled in the applicable number of shares of Legacy Core common stock, rounded down to the nearest whole share. At the Effective Time, each warrant to purchase Legacy Core common stock that was issued, outstanding and unexercised immediately prior to the Effective Time (except as described above) was assumed by XPDI and converted into a warrant to purchase shares of Class A Common Stock on the same terms and conditions as were applicable to the warrants to purchase Legacy Core common stock immediately prior to the Effective Time, equal to the product (rounded down to the nearest whole number) of (i) the number of shares of Legacy Core common stock subject to such warrant immediately prior to the Effective Time and (ii) the Exchange Ratio, at an exercise price per share (rounded up to the nearest whole cent) equal to the quotient of (x) the exercise price per share applicable to such warrant immediately prior to the Effective Time divided by (y) the Exchange Ratio.

At the Effective Time, each of Legacy Core's RSUs were assumed by XPDI and converted into a restricted stock unit to be settled in shares of Class A Common Stock on the same terms and conditions as were applicable to such RSU immediately prior to the Effective Time, including applicable vesting conditions, equal to the product (rounded down to the nearest whole number) of (i) the number of shares of Legacy Core common stock underlying such RSU immediately prior to the Effective Time and (ii) the Exchange Ratio.

At the Effective Time, each option to purchase Legacy Core common stock held by a former employee or service provider of Legacy Core that was vested and outstanding immediately prior to the Effective Time, except as set forth in the merger agreement, was deemed to have been exercised, on a net exercise basis with respect to the applicable exercise price and any required withholding or employment taxes thereon, immediately prior to the Closing and settled in the applicable number of shares of Class A Common Stock, rounded down to the nearest whole share. At the Effective Time, each option to purchase Legacy Core common stock held by a former employee or service provider of Legacy Core that was unvested and outstanding immediately prior to the Effective Time, except as set forth in the merger agreement, was automatically cancelled at the Closing without the payment of consideration. From and after the Closing, except as set forth in the merger agreement and except with respect to the holder's right to receive Class A Common Stock, if any, each Legacy Core option held by a former employee or service provider of Legacy Core was cancelled and ceased to be outstanding and the holder ceased to have any rights with respect thereto. At the Effective Time, except as set forth in the merger agreement, each other option to purchase Legacy Core common stock issued and outstanding immediately prior to the Effective Time not held by a former employee or service provider of Legacy Core was assumed by XPDI and converted into an option to purchase shares of Class A Common Stock on the same terms and conditions as were applicable to such option immediately prior to the Effective Time, including applicable vesting conditions (an "Exchanged Option"), equal to the product (rounded down to the nearest whole number) of (i) the number of shares of Class A Common Stock subject to such option immediately prior to the Effective Time and (ii) the Exchange Ratio, at an exercise price per share (rounded up to the nearest whole cent) equal to (x) the exercise price per share of such option immediately prior to the Effective Time divided by (y) the Exchange Ratio; provided, however, that the exercise price and the number of shares of Class A Common Stock purchasable pursuant to such option shall be determined in a manner consistent with the requirements of Section 409A of the Internal Revenue Code of 1986, as amended (the "Code") (to the extent applicable to such option); provided further, that in the case of any such option to which Section 422 of the Code applies, the exercise price and the number of shares of Class A Common Stock purchasable pursuant to such option was determined in accordance with the foregoing, and the requirements of Section 424(a) of the Code. At the Effective Time, each option to purchase Legacy Core common stock issued and outstanding immediately prior to the Effective Time held by certain specified Legacy Core optionholders that is vested and outstanding immediately prior to the Effective Time was deemed to have been exercised, on a net exercise basis with respect to the applicable exercise price and any required withholding or employment taxes thereon, immediately prior to the Closing and settled in cash.

At the Effective Time, each secured convertible promissory note issued by Legacy Core that was outstanding immediately prior to the Effective Time was assumed by XPDI and remains outstanding (and Legacy Core or its successor by merger remains an obligor with respect to such notes) and is convertible into Class A

Common Stock (rather than equity securities of Legacy Core) in accordance with the terms of such convertible promissory note; provided, however, that with respect to any outstanding convertible promissory notes for which Legacy Core received a duly executed exercise of conversion in accordance with such convertible promissory note, exercising the right of such holder to convert such convertible promissory note subject to and conditioned upon the occurrence of the Effective Time, the outstanding principal amount and accrued interest as of the Effective Time with respect to such convertible promissory note was converted into shares of Class A Common Stock, equal to the product (rounded down to the nearest whole number) of (i) the number of shares of Legacy Core common stock issuable upon the conversion of such convertible promissory note in accordance with such convertible promissory note immediately prior to the Effective Time and (ii) the Exchange Ratio.

Prior to the Effective Time, XPDI appointed a commercial bank or trust company (the "Exchange Agent") for the purpose of distributing the merger consideration payable to each holder of Legacy Core common stock. At the Effective Time, XPDI deposited evidence of shares of Class A Common Stock in book-entry form (or certificates representing shares of Class A Common Stock, at XPDI's election) representing the aggregate merger consideration payable pursuant to the merger agreement (such equity deposited with the Exchange Agent referred to herein as the "Exchange Fund"). At the Effective Time, XPDI delivered irrevocable instructions to the Exchange Agent to deliver the merger consideration out of the Exchange Fund in accordance with the merger agreement. See the final prospectus dated January 3, 2022 filed with the SEC in connection with the Business Combination and other filings that the Company makes with the SEC from time to time.

Accounting for the Merger

The merger between XPDI and Legacy Core will be accounted for as a reverse recapitalization in accordance with U.S. GAAP. Under this method of accounting, XPDI will be treated as the "acquired" company for financial reporting purposes and the merger will be treated as the equivalent of Legacy Core issuing stock for the net assets of XPDI, accompanied by a recapitalization. The net assets of XPDI will be stated at historical cost, with no goodwill or other intangible assets recorded. Operations prior to the Transactions will be those of Legacy Core.

Description of Blockcap Acquisition

On July 30, 2021, Legacy Core acquired Blockcap, one of its largest hosting customers. Consideration consisted of the issuance of approximately 72 million shares of Legacy Core common stock, approximately 4 million shares of Legacy Core restricted stock and approximately 5 million options to purchase shares of Legacy Core common stock. The acquisition is expected to be accounted for as a business combination using the acquisition method of accounting, whereby the net assets acquired and the liabilities assumed were recorded at fair value. Legacy Core is still in the process of completing its initial accounting for the business combination, including determining the fair value of consideration transferred, assets acquired and liabilities assumed and the allocation of the purchase price to goodwill. Legacy Core and Blockcap had preexisting relationships which were settled on the acquisition date. Using the estimated purchase price for the transaction, Legacy Core has allocated the purchase price to identifiable assets and liabilities based upon preliminary fair value estimates. The difference between the aggregate consideration relative to the fair value of net assets is recorded as goodwill. Refer to Note 3 for further discussion.

On December 1, 2020, Blockcap entered into assignment and assumption agreements with RME 88, RME 100, RME 200, BEP 888 and BEP 999 LLC ("BEP 999") (collectively the "LLCs"). An assessment was made of the respective net assets of the LLCs, the operating history of the LLCs, the subsequent voting rights retained by the various LLC investors, the composition of the new board of directors, and the composition of the new executive management was completed immediately following the completion of the business combinations. Based on the results of the assessment of the above factors, it was determined that RME 88 was the accounting acquirer. As a result, Blockcap's financial statements present the results of RME 88 from the date of its inception, February 19, 2019, to the date of the transactions with Blockcap and the other LLCs. The assignment

and assumption agreements with RME 100, RME 200 and BEP 888 were recorded as business combinations and the assignment and assumption agreement with BEP 999 was recorded as a share-based equity-settled asset acquisition. Pro forma adjustments have been made to the Blockcap historical results for the year ending December 31, 2020 to reflect results as if the acquisitions had been in place for the full year. The acquisition of BEP 999 was not considered material for purposes of the combined pro forma financial statements and therefore, pro forma adjustments were not made to reflect such acquisition.

On July 1, 2021, prior to Legacy Core's acquisition of Blockcap, Blockcap purchased Radar Relay, Inc. ("RADAR"), an early stage company focused on technology enhancement and development in the digital asset industry. The acquisition was accounted for as a business combination using the acquisition method of accounting, whereby the net assets acquired, and the liabilities assumed were recorded at fair value. Using the estimated purchase price for the transaction, Blockcap has allocated the purchase price to identifiable assets and liabilities based upon preliminary fair value estimates. The difference between the aggregate consideration relative to the fair value of net assets is recorded as goodwill. Refer to Note 3 for further discussion

Description of the Issuance of Convertible Notes

In April 2021, Legacy Core entered into a Secured Convertible Note Purchase Agreement and issued \$215 million of senior secured convertible notes to new and existing lenders. The convertible notes have a maturity date of April 2025 and bear interest of 10% per annum, of which 4% is payable in cash and 6% is payable in kind. The convertible notes are convertible into common or preferred shares at the option of the holder upon the occurrence of a) an initial public offering or SPAC merger, b) a private placement of equity securities with gross proceeds to the Company of at least \$50 million or c) a change in control. The conversion price is equal to a) 65-80% of the price paid for equity securities in an initial public offering, depending on the timing of the initial public offering or b) 65-80% of the fair value per share paid in a change in control or SPAC merger, depending on the timing of the change in control or SPAC merger. At maturity, the Secured Convertible Notes not converted will be owed two times the face value of the note plus accrued interest. The proceeds from the convertible notes were used, in part, to repay the 15% per annum Silverpeak loan of which \$21 million was issued in May 2020 and an additional \$9 million was issued in February 2021.

In August through November of 2021, an additional \$300 million of senior unsecured convertible notes were issued under substantially the same terms and conditions as the original \$215 million of secured convertible notes issued in April 2021, except that such notes are unsecured until the occurrence of a) an initial public offering or SPAC merger, b) a private placement of equity securities with gross proceeds to the Company of at least \$50 million or c) a change in control, at which time they become secured on a pari passu basis with the Secured Convertible Notes.

Legacy Core has reflected the issuance of the April convertible note, the repayment of the Silverpeak loan and the issuance of the senior unsecured convertible notes in its historical financial statements as of and for the nine months ended September 30, 2021. The impact is only partially reflected for the entire pro forma period in the historical statements. An adjustment has been made to the pro forma statements of operations for the twelve and nine month periods ending December 31, 2020 and September 30, 2021 respectively, to reflect the impact for the full pro forma period.

Refer to Note 3 and Note 6 for further discussion.

Basis of Pro Forma Presentation

The unaudited pro forma condensed combined financial information has been presented for illustrative purposes only and is not necessarily indicative of the operating results and financial position that would have been achieved had the Transactions occurred on the dates indicated, and does not reflect adjustments for any anticipated synergies, operating efficiencies, tax savings or cost savings. Any cash proceeds remaining after the

consummation of the Transactions and the other related events contemplated by the merger agreement are expected to be used for general corporate purposes. The unaudited pro forma condensed combined financial information does not purport to project the future operating results or financial position of XPDI following the completion of the Transactions.

The unaudited pro forma adjustments are based on information currently available, and assumptions and estimates underlying the unaudited pro forma adjustments are described in the accompanying notes. Actual results may differ materially from the assumptions used to present the accompanying unaudited pro forma condensed combined financial information.

The following summarizes the unaudited pro forma New Core common stock issued and outstanding immediately after the Transactions:

	Share Ownership in New Core		
amounts in thousands	Shares	% of Total	
Legacy Core shareholders (1)	301,292	90.7%	
XPDI:			
XPDI Class A shares (4)	22,153	6.7%	
XPDI Class B shares converted to Class A (2)	8,625	2.6%	
XPDI shareholders	30,778	9.3%	
Closing Shares (3)	332,070	100.0%	

- (1) Includes equivalent Legacy Core common shares exchanged for New Core shares using the exchange rate of 1.6001528688 consisting of:
 - a. Legacy Core common shares of 158,212
 - b. Legacy Core common shares from convertible Series A and B preferred stock of 10,826
 - c. Legacy Core vested warrants and options of 10,556
 - d. Blockcap equivalent Legacy Core common shares of 115,508
 - e. Blockcap equivalent Legacy Core common shares from vested RSUs of 1,956
 - f. Blockcap equivalent Legacy Core vested options of 4,234
- (2) Includes 1,725 Class B common shares that will be converted to Class A unvested shares and will vest upon the date on which the volume-weighted average price of the Class A Common Stock is greater than \$12.50 per share for any 20 trading days within any 30 consecutive trading day period within five years of the Closing date. The unvested shares will be excluded from pro forma earnings per share calculations because the stock price threshold contingency has not yet been met.
- (3) Excludes Legacy Core and Blockcap equivalent Legacy Core common shares from unvested RSUs of 97,409 and 4,855, respectively, and 21,103 and 3,111 from unvested options, respectively, using the exchange rate of 1.6001528688 and all other potentially dilutive securities including Legacy Core options, warrants, restricted stock units and convertible notes as well as XPDI warrants. See Note 5—Earnings per Share.
- (4) Includes the impact of the redemption by XPDI shareholders of 12,347 shares of Class A common stock subject to redemption.

UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET AS OF SEPTEMBER 30, 2021

	As of Sept			As of September 30, 2021	
	Legacy Core				
in thousands	Pro Forma (Note 2)	XPDI (Historical)	Merger Adjustment	s	Pro Forma Combined
Assets					
Current Assets:					
Cash and cash equivalents	\$ 240,427	\$ 1,599	\$ (39,000) 221,556	A B	\$ 424,582 —
Restricted cash	12,101	_	_		12,101
Accounts receivable, net of allowance of \$620	602	_	_		602
Accounts receivable from related parties	261	_	_		261
Deposits for equipment	469,890				469,890
Digital currency assets	116,233		_		116,233
Other current assets	9,978	447	102.556		10,425
Total current assets	849,492	2,046	182,556		1,034,094
Property, plant and equipment, net	219,795				219,795
Goodwill	1,106,968		_		1,106,968
Intangible assets, net Investments held in Trust Account	8,709	345,027	(345,027)	В	8,709
Other noncurrent assets	14,110	343,027	(343,027)	ь	14.110
Total assets	\$ 2,199,074	\$ 347,073	\$(162,471)		\$ 2,383,676
	\$ 2,199,074	\$ 347,073	\$(102,471)		\$ 2,363,070
Liabilities, Redeemable Preferred Stock, Commitments and Contingencies and Stockholders' Equity Current Liabilities:					
Accounts payable	\$ 28,796	\$ 721	s —		\$ 29,517
Accrued expenses and other	35,074	3,226	_		38,300
Deferred revenue	206,139				206,139
Capital lease obligations, current portion	2,525	_	_		2,525
Notes payable, current portion Other current liabilities	25,202	148			25,202 148
	297,736				
Total current liabilities		4,095			301,831
Capital lease obligations, net of current portion	1,524	_	_		1,524 566,930
Notes payable, net of current portion Derivative warrant liabilities	566,930	36,932			36,932
Deferred underwriting commissions		12,075	(12,075)	A	30,932
Other noncurrent liabilities	1.994	12,073	(12,073)	_	1,994
Total liabilities	868,184	53,102	(12,075)		909,211
Contingently redeemable preferred stock	44,476	33,102		D	707,211
Commitments and Contingencies:	44,470	_	(44,470)	D	_
Class A common stock subject to redemption	_	345,000	(345,000)	E	_
Stockholders' Equity:		3.5,000	(3.5,000)	_	
Common stock; \$0.00001 par value	2	_	1	F	3
Class B common stock, \$0.0001 par value	_	1	(1)	F	_
Additional Paid in Capital	1,385,381	_	(51,030)	F	1,576,243
	_	_		A	
	_	_	44,476	D	
		_	345,000	E	
Assumption (deficie) / Families	(00.417)	(51.020)		C	(02 220)
Accumulated (deficit) / Earnings	(90,417)	(51,030)	51,030 (1,000)	F A	(93,229)
				A	
Accumulated other comprehensive loss	(8,552)	_	(1,012)	. 1	(8,552)
Total Stockholders' Equity	1,286,414	(51,029)	239,080		1,474,465
Total Liabilities, Redeemable Preferred Stock, Commitments and Contingencies and Stockholders' Equity		\$ 347.073	\$(162,471)		\$ 2.383.676
Tom Districts, redeciment referred stock, communicities and contingencies and stockholders. Equity	Ψ2,177,07 1	ψ J17,073	φ(102,1/1)		2,303,070

See accompanying notes to the unaudited pro forma combined financial statements

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE NINE MONTHS ENDED SEPTEMBER 30, 2021

	Septembe Legacy		S	eptember 30, 2021	
in thousands	Core Pro Forma (Note 2)	XPDI (Historical)	Merger Adjustments		Pro Forma Combined
Hosting revenue from customers	\$ 39,914	\$ —	\$ —	\$	39,914
Equipment sales to customers	95,741	_	_		95,741
Digital asset mining income	149,541			_	149,541
Total revenue	285,196	_			285,196
Costs of revenue	136,002	_	_		136,002
Gross profit	149,194		_		149,194
(Loss) gain on legal settlements	(2,603)	_	_		(2,603)
Gain (loss) from sales of digital currency assets	550	_	_		550
Impairment of digital currency assets	(30,024)	_	_		(30,024)
Operating expenses:					
Research and development	4,231	_	_		4,231
Sales and marketing	2,186	_	_		2,186
General and administrative	75,749	4,078	_		79,827
General and administrative-related party	_	160	_		160
Franchise tax expenses		148			148
Total operating expenses	82,166	4,386			86,552
Operating income	34,951	(4,386)			30,565
Other (expense) income:		` ` `			
(Loss) gain from sales of digital currency assets	(1,472)	_	_		(1,472)
(Loss) on debt from extinguishment	(8,016)	_	_		(8,016)
Change in fair value of derivative warrant liabilities	_	(13,904)	_		(13,904)
Offering costs associated with derivative warrant liabilities	_	(1,055)	_		(1,055)
Income from investments held in Trust Account	_	27	(27)	e	_
Other income (expense)	(8,661)	_	_		(8,661)
Interest expense, net	(65,767)		<u> </u>		(65,767)
Total other (expense) income	(83,916)	(14,932)	(27)		(98,875)
Income (loss) before income taxes	(48,965)	(19,318)	(27)	_	(68,310)
Income tax provision (benefit)	5,777		_		5,777
Net income (loss) and other comprehensive income (loss)	\$ (54,742)	\$ (19,318)	\$ (27)	\$	(74,087)
Net income (loss) attributable to common stockholders	\$ (54,742)	\$ (19,318)	\$ (27)	\$	(74,087)
Net income (loss) per share				_	
Common stock - Basic and Diluted	\$ (0.32)	\$ (0.51)		a \$	(0.24)
Weighted average shares outstanding:					
Basic and Diluted common stock	171,059	37,644		b	313,599

See accompanying notes to the unaudited pro forma combined financial statements

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE TWELVE MONTHS ENDED DECEMBER 31, 2020

	December	December 31, 2020		
in thousands	Legacy Core Pro Forma (Note 2)	XPDI (Historical)	Merger Adjustments	Pro Forma Combined
Hosting revenue from customers	\$ 39,470	\$ —	\$ —	\$ 39,470
Equipment sales to customers	11,842	_	_	11,842
Digital asset mining income	19,636			19,636
Total revenue	70,948	_	_	70,948
Costs of revenue	54,237			54,237
Gross profit	16,711	_	_	16,711
Gain on legal settlements	5,814	_	_	5,814
Gain on sale of digital assets	353	_	_	353
Operating expenses:				
Research and development	5,271	_	_	5,271
Sales and marketing	1,771	_	_	1,771
General and administrative	39,190	10	2,812 c	42,012
Total operating expenses	46,232	10	2,812	49,054
Operating income	(23,354)	(10)	(2,812)	(26,176)
Other (expense) income:				
(Loss) on debt from extinguishment	(1,333)	_	_	(1,333)
Other income (expense)	(10,113)	_	_	(10,113)
Interest expense, net	(56,988)			(56,988)
Total other (expense) income	(68,434)		<u> </u>	(68,434)
Income (loss) before income taxes	(91,788)	(10)	(2,812)	(94,610)
Income tax provision (benefit)	(134)		<u> </u>	(134)
Net income (loss) and other comprehensive income (loss)	\$ (91,653)	\$ (10)	\$(2,812)	\$ (94,475)
Deemed divided from common to preferred exchange	(10,478)			<u> </u>
Net income (loss) attributable to common stockholders	\$(102,131)	\$ (10)	\$ 7,666	\$ (94,475)
Net income (loss) per share				
Common stock – Basic and Diluted	\$ (0.60)	\$ —	a	\$ (0.30)
Weighted average shares outstanding:				
Common stock – Basic and Diluted	171,059	7,500	b	313,599

See accompanying notes to the unaudited pro forma combined financial statements

NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

1. Basis of Presentation

The unaudited pro forma condensed combined financial information has been prepared in accordance with Article 11 of Regulation S-X, as amended. The adjustments in the unaudited pro forma condensed combined financial information have been identified and presented to provide relevant information necessary for an illustrative understanding of New Core upon consummation of the Transactions. Assumptions and estimates underlying the unaudited pro forma adjustments set forth in the unaudited pro forma condensed combined financial information are described in the accompanying notes.

The unaudited pro forma condensed combined financial information has been prepared to illustrate the effect of the Transactions and has been prepared for informational purposes only.

The merger between XPDI and Legacy Core will be accounted for as a reverse recapitalization in accordance with GAAP. Under this method of accounting, XPDI will be treated as the "acquired" company for financial reporting purposes. Accordingly, the merger will be treated as the equivalent of Legacy Core issuing stock for the net assets of XPDI, accompanied by a recapitalization. The net assets of XPDI will be stated at historical cost, with no goodwill or other intangible assets recorded. Operations prior to the Transactions will be those of Legacy Core. The pro forma adjustments related to the merger are described in Note 3.

The acquisition of Blockcap by Legacy Core, as further summarized in Note 2, was accounted for as a business combination using the acquisition method of accounting in accordance with ASC 805, *Business Combinations*, with Legacy Core determined to be the accounting acquirer. Accordingly, the acquired assets and assumed liabilities of Blockcap have been adjusted to fair value, where appropriate. The difference between the aggregate consideration relative to the fair value of the identifiable net assets is recorded as goodwill. In addition, pro forma adjustments related to Legacy Core's issuance of convertible notes and the repayment of the Silverpeak loan are reflected in Note 2.

Management has made significant estimates and assumptions in its determination of the pro forma adjustments. As the unaudited pro forma condensed combined financial information has been prepared based on these preliminary estimates, the final amounts recorded may differ materially from the information presented.

The unaudited pro forma condensed combined financial information does not give effect to:

- any anticipated synergies, operating efficiencies, tax savings, cost savings or increased costs of a public company that may be associated with the Transactions, or
- the potential purchases of mining and hosting equipment.

The pro forma adjustments reflecting the consummation of the Transactions are based on certain currently available information and certain assumptions and methodologies that XPDI believes are reasonable under the circumstances. The unaudited condensed combined pro forma adjustments, which are described in the accompanying notes, may be revised as additional information becomes available and is evaluated. Therefore, it is likely that the actual adjustments will differ from the pro forma adjustments and it is possible the difference may be material. XPDI believes that its assumptions and methodologies provide a reasonable basis for presenting all of the significant effects of the Transactions based on information available to management at this time and that the pro forma adjustments give appropriate effect to those assumptions and are properly applied in the unaudited pro forma condensed combined financial information.

The unaudited pro forma condensed combined financial information is not necessarily indicative of what the actual results of operations and financial position would have been had the Transactions taken place on the dates indicated, nor are they indicative of the future consolidated results of operations or financial position of New Core. The unaudited pro forma condensed combined financial information should be read in conjunction with the historical financial statements and notes thereto of XPDI, Legacy Core and Blockcap.

2. Historical Legacy Core, Blockcap Acquisition and Convertible Note Issuance

The following unaudited pro forma adjustments reflect the impact of Transactions completed by Legacy Core prior to the anticipated merger, including (all share amounts are in thousands):

- The pro forma interest expense impact of the April 2021 Legacy Core issuance of \$215 million of convertible notes, net of \$5.6 million origination fees and the subsequent repayment of the \$30 million Silverpeak loan net of prepayment penalties, the interest expense impact of the August and September Convertible Notes issuance of \$205 million and the pro forma impact of the subsequent issuance in October and November 2021 of \$95 million of additional Convertible Notes. It is not anticipated that the holders of the convertible notes will exercise their conversion rights as a result of the merger; and
- The acquisition of Blockcap by Legacy Core on July 30, 2021, including any elimination of the effect of transactions between Legacy Core and Blockcap, as required, is included in the Legacy Core historical financial statements as of and for the nine months ended September 30, 2021. The pro forma impact to the Statement of Operations had the acquisition taken place on January 1, 2020 is included as an adjustment for the period from January 1, 2021 through July 30, 2021 and the twelve months ended December 31, 2020. Consideration paid consisted of 72,186 Legacy Core common shares, 657 common shares from vested options settlement, 4,256 Legacy Core Restricted Stock Units and 2,393 Legacy Core Options. The portion of the fair value of the replaced Blockcap share based payments attributable to pre-combination service as well as the impact of the effective settlement of preexisting hosting and equipment contracts between Legacy Core and Blockcap has been included in the aggregate purchase price of \$1,129.4 million.

The purchase price allocation is as follows:

PPA Allocation

Consideration (in thousands):	
72,186 common shares valued at \$16.18 per share (1) (2)	1,167,965
Fair value of replaced Blockcap share-based payments attributable to pre-combination service (3)	21,941
Settlement of pre-existing contracts (4)	(60,522)
Total Consideration:	\$ 1,129,384
Fair value of assets acquired, and liabilities assumed:	
Cash and cash equivalents	\$ 704
Digital assets-Bitcoin	73,304
Digital assets-Ethereum	365
Digital assets-Bitcoin cash	8
Digital assets-Siacoin	554
Digital asset-Other	3,329
Other current assets	633
Intangible assets, net	2,925
Property, plant and equipment, net	27,089
Other noncurrent assets	 1,293
Total assets acquired	\$ 110,204
Accounts payable	492
Accrued expenses and other	22,647
Other current liabilities	6,408
Total liabilities assumed	\$ 29,547
Total identifiable net assets	\$ 80,657
Goodwill on acquisition	\$ 1,048,727

- (1) 72,186 common shares represent the equivalent Legacy Core common shares issued to Blockcap shareholders as consideration for the purchase.
- (2) The price per share of our common shares was estimated to be \$16.18. As the Legacy Core common shares were not listed on a public marketplace, the calculation of the fair value of the common shares was subject to a greater degree of estimation. Given the absence of a public market, an estimate of the fair value of the common shares was required at the time of the Blockcap Acquisition. Objective and subjective factors were considered in determining the estimated fair value and because there is no active trading of the Legacy Core equity shares on an established securities market, an independent valuation specialist was engaged. The valuation was determined by weighting the outcomes of scenarios estimating share value based on both public company valuations and private company valuations. Both a market approach and common stock equivalency model were used to determine a range of outcomes, which were weighted based on probability to determine the result.
- (3) Reflects the estimated fair value of replaced Blockcap share-based payments allocated to purchase price based on the proportion of service related to the pre-combination period. The fair value of the stock-based awards was determined utilizing the Black-Scholes pricing model.
- (4) Blockcap had preexisting hosting and equipment with Legacy Core that were effectively settled by Legacy Core's acquisition of Blockcap. As a result, the consideration transferred to Blockcap has been adjusted by the deferred revenue balances that were settled at the time of acquisition.

This preliminary purchase price allocation has been used to prepare the unaudited pro forma adjustments. The final purchase price allocation will be determined when detailed valuations and necessary calculations have been completed. The final purchase price allocation could differ materially from the preliminary purchase price allocation and may include changes in value of consideration transferred (including the fair value of common shares on the acquisition date and the allocation of share-based compensation replacement awards to purchase consideration), changes in allocations to intangible assets such as developed technology, trade names and goodwill and other changes to assets and liabilities.

The goodwill of approximately \$1,047.8 million arising from the Blockcap acquisition preliminary purchase price allocation represents the excess of the purchase price over the aggregate fair value of the net identifiable assets acquired and liabilities assumed, including identifiable intangible assets. The goodwill is not deductible for U.S. federal income tax purposes. Management believes the acquisition of Blockcap strengthens its presence in the data mining market due to the scale of its operations. These factors are the basis for the excess purchase price paid over the value of the assets acquired and liabilities assumed, resulting in goodwill.

UNAUDITED PRO FORMA CORE SCIENTIFIC CONDENSED COMBINED BALANCE SHEET

	Septemb	As of September 30, 2021		
in thousands	Core Scientific (Historical) (A)	Convertible Note adjustments	Core Scientific Pro Forma Combined	
Assets				
Current Assets:				
Cash and cash equivalents	\$ 147,906	\$92,521 B	\$ 240,427	
Restricted cash	12,101	_	12,101	
Accounts receivable, net of allowance of \$620	602	_	602	
Accounts receivable from related parties	261	_	261	
Deposits for equipment	469,890	_	469,890	
Digital currency assets	116,233	_	116,233	
Other current assets	9,978	_	9,978	
Total current assets	756,971	92,521	849,492	
Property, plant and equipment, net	219,795		219,795	
Goodwill	1,106,968	_	1,106,968	
Intangible assets, net	8,709	_	8,709	
Other noncurrent assets	14,110	_	14,110	
Total assets	\$2,106,553	\$92,521	\$2,199,074	
Current Liabilities:				
Accounts payable	\$ 28,796	_	\$ 28,796	
Accrued expenses and other	35,074	_	35,074	
Deferred revenue	206,139	_	206,139	
Capital lease obligations, current portion	2,525	_	2,525	
Notes payable, current portion	25,202	_	25,202	
Total current liabilities	297,736		297,736	
Capital lease obligations, net of current portion	1,524		1,524	
Notes payable, net of current portion	471,930	95,000 B	566,930	
Other noncurrent liabilities	1,994	_	1,994	
Total liabilities	773,184	95,000	868,184	
Contingently redeemable preferred stock	44,476		44,476	
Stockholders' Equity:	,		,	
Common stock; \$0.00001 par value	2	_	2	
Additional Paid in Capital	1,385,381	_	1,385,381	
Accumulated (deficit) / Earnings	(87,938)	(2,479) B	(90,417)	
Accumulated other comprehensive loss	(8,552)	<u> </u>	(8,552)	
Total Stockholders' Equity	1,288,893	(2,479)	1,286,414	
Total Liabilities, Redeemable Preferred				
Stock, and Stockholders' Equity	\$2,106,553	\$92,521	\$2,199,074	

UNAUDITED PRO FORMA CONDENSED COMBINED CORE SCIENTIFIC

STATEMENT OF OPERATIONS FOR THE NINE MONTHS ENDED SEPTEMBER 30, 2021

					Transaction Accounting Adjustments				
<u>in thousands</u>	Core Scientific (Historical) September 30, 2021	Blockcap (Historical) for the period January 1, 2021 through July 30, 2021	Eliminatio Adjustmei	ıts	Blockcap Acquisitio		Convertibl Note	e	Core Scientific Pro Forma Combined
Hosting revenue from customers	\$ 51,742	\$ —	\$(11,828)	e1	\$ —		_		\$ 39,914
Equipment sales to customers	113,435		(17,694)	e2	_				95,741
Digital asset mining income	77,511	72,030							149,541
Total revenue	242,688	72,030	(29,522)		_				285,196
Costs of revenue	145,193	17,903	(11,828)	e1	(868)	a	_		136,002
			(14,398)	e2					
Gross profit	97,495	54,127	(3,296)		868		_		149,194
(Loss) / gain on legal settlements	(2,603)						_		(2,603)
Gain on sale of digital assets	405	145	_		_		_		550
Impairment of digital currency asset	(12,552)	(17,472)			_				(30,024)
Operating expenses:	4.001								4.001
Research and development	4,231								4,231
Sales and marketing	2,186	47.007	_		(10.240)		_		2,186
General and administrative	46,992	47,997			(19,240)	c			75,749
Total operating expenses	53,409	47,997			(19,240)				82,166
Operating income (loss)	29,336	(11,197)	(3,296)		20,108				34,951
Other (expense) income:									
(Loss) on sale of assets		(1,472)			_				(1,472)
(Loss) on debt from extinguishment	(8,016)		_		_		_		(8,016)
Other income (expense)	(8,661)						(27.500)		(8,661)
Interest (expense), net	(26,550)						(37,500)	d	(65,767)
Total other (expense) income	(43,227)						(37,500)		(83,916)
Income (loss) before income taxes	(13,891)		(3,296)		20,108		(37,500)		(48,965)
Income tax provision (benefit)	(697)	6,474							5,777
Net income (loss) and other comprehensive income (loss)	\$ (13,194)	\$ (20,860)	\$ (3,296)		\$ 20,108		\$(37,500)		\$ (54,742)
Net income (loss) attributable to common stockholders	\$ (13,194)	\$ (20,860)	\$ (3,296)		\$ 20,108		\$(37,500)		\$ (54,742)

UNAUDITED PRO FORMA CONDENSED COMBINED CORE SCIENTIFIC

STATEMENT OF OPERATIONS FOR THE TWELVE MONTHS ENDED DECEMBER 31, 2020

	December 31, 2020			Transaction Accounting Adjustments					
in thousands	Core Scientific (Historical)	Blockcap Pro Forma (Adjustment f)	Eliminati Adjustme		Blockcap Acquisition	•	Convertibl Note	e	Core Scientific Pro Forma Combined
Hosting revenue from customers	\$ 41,598	\$ —	\$(2,128)	e1	\$ —				\$ 39,470
Equipment sales to customers	12,595	_	(753)	e2	_		_		11,842
Digital asset mining income	6,127	13,509							19,636
Total revenue	60,320	13,509	(2,881)		_		_		70,948
Costs of revenue	50,928	7,655	(2,128)	e1	(1,488)	a	_		54,237
			(730)	e2					
Gross profit	9,392	5,854	(23)		1,488		_		16,711
Gain on legal settlements	5,814	_	_		_		_		5,814
Gain on sale of digital assets	65	288	_		_		_		353
Operating expenses:									
Research and development	5,271	_	_		_		_		5,271
Sales and marketing	1,771	_	_		_		_		1,771
General and administrative	14,556	7,073			17,561	b			39,190
Total operating expenses	21,598	7,073	_		17,561		_		46,232
Operating income	(6,327)	(931)	(23)		(16,073)		_		(23,354)
Other (expense) income:									
(Loss) on debt from extinguishment	(1,333)	_	_		_		_		(1,333)
Other income (expense)	(110)	(10,003)	_		_		_		(10,113)
Interest (expense), net	(4,436)	(1,198)	_		_		(2,479)	e	(56,988)
							(48,875)	d	
Total other (expense) income	(5,879)	(11,201)	_		_		(51,354)		(68,434)
Income (loss) before income taxes	(12,206)	(12,132)	(23)		(16,073)		(51,354)		(91,788)
Income tax provision (benefit)	_	(134)	_		_		_		(134)
Net income (loss) and other comprehensive income (loss)	\$ (12,206)	\$ (11,998)	\$ (23)		\$(16,073)		\$(51,354)		\$ (91,653)
Deemed divided from common to preferred exchange	(10,478)								(10,478)
Net income (loss) attributable to common stockholders	\$ (22,684)	\$ (11,998)	\$ (23)		\$(16,073)		\$(51,354)		\$(102,131)

Adjustments to Unaudited Pro Forma Condensed Combined Legacy Core and Blockcap Balance Sheet

The adjustments included in the unaudited pro forma condensed combined balance sheet as of September 30, 2021 are as follows (in thousands): Other adjustments:

- (A) The following purchase price adjustments related to the acquisition of Blockcap by Legacy Core have been included in the Legacy Core historical balance sheet as of September 30, 2021:
 - Changes to the Blockcap historical financials to represent the fair value changes associated with the acquisition. The
 resulting goodwill adjustment represents the excess of the purchase price over the aggregate fair value of the net identifiable
 assets acquired and liabilities assumed, including identifiable intangible assets. The goodwill is not deductible for U.S.
 federal income tax purposes.

The fair value of the digital assets acquired was determined using the average U.S. dollar spot rate of the related digital currency at the time of acquisition.

The following table presents the details of Blockcap's digital asset holdings:

	As of July 30, 2021					
dollars excluding Price in thousands	Quantity	Price	Fair value	Carrying value		
Bitcoin (BTC)	1,747	\$ 42,235.55	\$ 73,805	\$ 55,568		
Ethereum (ETH)	575	\$ 2,466.96	\$ 1,420	\$ 1,263		
Bitcoin Cash (BCH)	15	\$ 491	\$ 7	\$ 7		
Siacoin (SIA)	39,487,794	\$ 0.01	\$ 554	\$ 384		
Other	Various	Various	\$ 2,714	\$ 1,777		
TOTAL			\$ 78,500	\$ 58,999		

The fair value of the mining equipment was determined by applying the Cost Approach, which is based on estimating the Replacement Cost New ("RCN") and deducting for depreciation which considers physical deterioration, functional obsolescence and economic/external obsolescence. The primary depreciation applied to the mining equipment was physical depreciation as it was determined that functional and economic obsolescence was not present with the miners. The RCN of the miners was based on current pricing for such equipment, which fluctuates with the increase or decrease of digital asset prices.

- 2. The elimination of Blockcap additional paid in capital and accumulated earnings.
- **3.** The total consideration added to APIC as a result of the acquisition.
- 4. The transaction costs associated with the Blockcap acquisition that were expensed as an adjustment to cash and retained earnings.
- 5. The one-time expense Legacy Core has recorded as compensation expense in its financial statements for the period ending September 30, 2021 for the acceleration of certain equity awards of Blockcap's CEO and others.
- (B) During April, August and September 2021, Legacy Core entered into a Convertible Note Purchase Agreement with various Purchasers (Investors) and is continuing to enter into these agreements. As of the date of this filing, Legacy Core has issued notes in the aggregate amount of \$515 million with origination fees estimated at \$13,332. Legacy Core elected to measure the April, August and September issuance of convertible notes at fair value and accordingly recognized \$10,853 of debt issuance costs as incurred as interest expense. The pro forma adjustments reflect the same election for the additional \$95,000 issuance and recognize an additional \$2,479 of debt issuance cost as interest expense. The related adjustments for this transaction resulted in an increase to cash of \$92,521, an increase to notes payable of \$95,000 and an increase to accumulated deficit of \$2,479.

Adjustments to Unaudited Pro Forma Condensed Combined Legacy Core and Blockcap Statements of Operations

The pro forma adjustments included in the unaudited pro forma condensed combined statements of operations for the period January 1, 2021 through July 30, 2021 for Blockcap adjustments and for the nine months ended September 30, 2021 for the convertible debt adjustments, and for the twelve months ended December 31, 2020 are as follows (in thousands):

Eliminations of intercompany transactions:

- (e1) Reflects the elimination of Legacy Core hosting revenue from Blockcap and the corresponding Blockcap cost of revenue.
- (e2) Reflects the elimination of Legacy Core equipment sales revenue and cost of revenue sold from Blockcap.

Other adjustments:

- (a) Reflects the change in the cost of revenue attributable to incremental depreciation as a result of the fair value decrease to the basis of the Blockcap mining equipment assets as a result of the acquisition. The fair value adjustment decreases the historical cost of these assets by \$4.5 million on assets with an assumed useful life of three years resulting in a decrease in depreciation expense of \$0.9 million and \$1.5 million for the period January 1, 2021 through July 30, 2021 and for the twelve months ended December 31, 2020, respectively.
- (b) Reflects incremental stock compensation expense related to the replacement of unvested Blockcap share-based payments with awards issued by Legacy Core that will be recognized over the remaining service period using the acquisition date fair value, in excess of amounts recognized historically by Blockcap for the twelve months ended December 31, 2020. The associated incremental stock compensation expense for the nine months ended September 30, 2021 is included in the Legacy Core historical statement of operations.
- (c) Reflects the elimination of \$19,239 of expense recognized by Blockcap in July 2021 for the acceleration of certain equity awards of its CEO and others. Because this acceleration was deemed to be in contemplation of the Merger, Legacy Core has recorded \$23,294 of compensation expense for the acceleration in its financial statements for the period ending September 30, 2021, which was determined based on the fair value of the awards at the time of the Merger. This adjustment is necessary to avoid duplication of the expense attributable to the combined company related to the acceleration of the same awards.
- (d) Reflects the impact to interest expense derived from removing 15% per annum interest on the \$30,000 Silverpeak loan, originated in May 2020, and replacing it with 10% per annum interest on the \$515,000 million convertible notes (\$215,000 issued in April 2021 and the remaining \$300,000 issued in August through November 2021). This adjustment resulted in:
 - a. a reduction of \$2,625 in interest expense due to the elimination of the Silverpeak loan, an increase in interest expense of \$51,500 due to the issuance of the convertible notes r a net increase in interest expense of \$48,875 for the twelve months ending December 31, 2020.
 - b. a reduction of \$1,125 in interest expense due to the elimination of the Silverpeak loan, an increase in interest expense of \$38,625 due to the issuance of the convertible note for a net increase in interest expense of \$37,500 for the nine months ending September 30, 2021.
- (e) Reflects the non-recurring interest expense related to the assumption that Legacy Core will elect to measure the October through November issuance of convertible notes at fair value similarly to the April, August and September issuance of convertible notes and accordingly included \$2,479 of debt issuance costs as incurred as interest expense in the pro forma adjustments.

(f) Reflects the pro forma results for Blockcap for the year ending December 31, 2020 assuming the three significant LLCs acquired by Blockcap on December 1, 2020 had been in place for the full year as follows:

	For the year ended December 31, 2020	For the periods from inception through November 30, 2020 RME RME			ended December 31, For the peri 2020 through No			Decemb	er 31, 2020 Blockcap
in thousands	Blockcap Historical	Black 100	Black 200	BEP 888	Acquisition Adjustment				
Hosting revenue from customers	\$ —	\$ —	<u>\$</u> —	\$ —		\$ —			
Equipment sales to customers	_	_	_	_	_	_			
Digital asset mining income	5,972	4,010	3,374	153	_	13,509			
Total revenue	5,972	4,010	3,374	153		13,509			
Costs of revenue	2,781	2,623	2,034	48	169	i 7,655			
Gross profit	3,191	1,387	1,340	105	(169)	5,854			
Gain on legal settlements	_	_	_	_	_	_			
Gain on sale of digital assets	53	152	80	3	_	288			
Operating expenses:									
Research and development	_	_	_	_	_	_			
Sales and marketing	_	_	_	_	_	_			
General and administrative	2,221	1,559	1,343	1,950		7,073			
Total operating expenses	2,221	1,559	1,343	1,950		7,073			
Operating income (loss)	1,023	(20)	77	(1,842)	(169)	(931)			
Other (expense) income:									
(Loss) on debt from extinguishment	_	_	_	_	_				
Other income (expense)	(10,003)	_	_	_	_	(10,003)			
Interest expense, net	(419)	(414)	(365)	_	_	(1,198)			
Total other (expense) income	(10,422)	(414)	(365)			(11,201)			
Income (loss) before income taxes	(9,399)	(434)	(288)	(1,842)	(169)	(12,132)			
Income tax expense/(benefit)	538	(107)	(71)	(453)	(42)	ii <u>(134)</u>			
Net income (loss)	\$ (9,937)	\$ (327)	\$ (217)	\$(1,389)	\$ (127)	\$(11,998)			

- i. Reflects additional depreciation expense related to the acquisitions of the LLCs
- ii. The combined net income before tax of the Company includes net income before tax of \$ 406.5 attributable to multiple limited liability companies (of which \$409.3 is included in Blockcap Historical) that are classified as partnerships for Federal income tax purposes, and as such, any partnership income or loss, specifically allocated or otherwise are subjected to tax at the individual partner level.

3. Transaction Accounting Adjustments-Merger

Merger Adjustments to Unaudited Pro Forma Condensed Combined Balance Sheet (in thousands)

The adjustments included in the unaudited pro forma condensed combined balance sheet as of September 30, 2021 are as follows (in thousands):

(A) Reflects the estimated \$39,000 reduction in cash for estimated transaction related expenses. This includes \$12,075 that was previously recorded as deferred underwriting commissions and was therefore reversed and \$1,000 related to non-recurring expenses from third party consulting and other fees

- related to the merger that were not considered direct and incremental to the offering and are therefore reflected as an adjustment to the accumulated deficit. An additional \$1,812 of the \$39,000 is reflected as an adjustment to the accumulated deficit to properly reflect the direct and incremental costs allocated to the XPDI liability-classified instruments that are measured at fair value through earnings. The remaining \$24,113 is adjusted against additional paid in capital as direct and incremental costs of the offering.
- (B) Reflects the increase in cash from the release of the balance of funds equal to \$221,556 from the Trust Account after payment of \$123,471 for the redemption of 12,347 shares.
- C) Reflects the impact to equity related to \$123,471 of Trust Account funds paid to holders that redeemed 12,347 Class A redeemable shares at \$10 per share for a total of \$123,471.
- D) Reflects the conversion of 6,452 shares of Series A Preferred Stock and 314 shares of Series B Preferred Stock redeemable preferred stock to 6,766 shares of Legacy Core common stock just prior to the merger, eliminating the preferred stock balance. Subsequent to this conversion, these shares will be exchanged for New Core common shares using the exchange rate calculated per the merger agreement.
- (E) Reflects the elimination of 34,500 shares of Class A Common Stock subject to redemption. The \$345,000 subject to redemption is eliminated and added to additional paid in capital.
- (F) Represents the elimination of XPDI equity and accumulated earnings and the exchange of existing Legacy Core and XPDI outstanding equity instruments for New Core common stock as follows:
 - a. 171,059 existing Class A Legacy Core shares and 6,766 shares of Legacy Core Series A and Series B convertible preferred stock, were exchanged for New Core shares using an exchange rate of 1.6001528688 per the merger agreement resulting in a total of 284,546 New Core common shares outstanding with a par value of \$0.0001.
 - b. 22,153 existing Class A XPDI shares not redeemed and 8,625 existing XPDI Class B shares into 30,778 common shares of New Core. Immediately prior to the Closing, 1,725 Class B common shares will be unvested and will vest upon the date on which the volume-weighted average price of the Class A Common Stock is greater than \$12.50 per share for any 20 trading days within any 30 consecutive trading day period within five years of the Closing date. These unvested shares are excluded from earnings per share calculations until such time as conditions are met for vesting.
 - c. Removes XPDI accumulated earnings with the offsetting adjustment to paid in capital.

Merger Adjustments to Unaudited Pro Forma Condensed Combined Statements of Operations (in thousands)

The proforma adjustments included in the unaudited proforma condensed combined statements of operations for the nine months ended September 30, 2021 and twelve months ended December 31, 2020 are as follows:

- (a) Reflects the calculation of earnings per share, basic and diluted, of the combined companies after the conversion of equity shares into New Core shares using the adjusted net income of the combined companies. See Note 5—Earnings per Share.
- (b) Reflects the total New Core common shares outstanding at the completion of the transaction. See Note 5—Earnings per Share.
- (c) Reflects the sum of a) \$1,000 related to non-recurring expenses from third-party consulting and other fees related to the merger that were not considered direct and incremental to the offering and therefore reflected as an operating expense and b) \$1,812 related to the direct and incremental costs allocated to the XPDI liability-classified instruments that are measured at fair value through earnings, for a total adjustment of \$2,812.
- (d) Reflects the elimination of the dividend from common to preferred exchange. In February 2020, the Company completed an exchange of 1,096 shares of common stock that were originally issued in the

private placement offering from October 2018 to December 2019 for 1,802 newly issued shares of Series A Preferred Stock. The shares of common stock were retired upon reacquisition by the Company. The Company received no net proceeds from the exchange and recognized a deemed dividend of \$10,478 based on the incremental fair value of the preferred stock received by the stockholders compared to the fair value of the common stock exchanged. Upon consummation of the transactions, all preferred stock was exchanged for common stock under the terms of the merger agreement, eliminating the dividend.

(e) Reflects the elimination of income earned on investments in trust due to the release of trust funds to cash or the reduction in trust funds due to the redemption of redeemable stock.

4. Accounting Policies

Upon consummation of the Transactions, XPDI will perform a comprehensive review of all entities' accounting policies. As a result of the review, management may identify additional differences between the accounting policies of the entities which, when conformed, could have a material impact on the financial statements of New Core.

5. Earnings per Share

The Company computes earnings per share ("EPS") following Financial Accounting Standards Board ("FASB") Accounting Standards Codification ("ASC") Topic 260, *Earnings per share*. Basic EPS is measured as the income or loss available to common stockholders divided by the weighted average common shares outstanding for the period. Diluted EPS presents the dilutive effect on a per-share basis from the potential conversion of convertible securities or the exercise of options and or warrants; the dilutive impacts of potentially convertible securities are calculated using the as-if method; the potentially dilutive effect of options or warranties are computed using the treasury stock method. Securities that are potentially an anti-dilutive effect (i.e., those that increase income per share or decrease loss per share) are excluded from diluted EPS calculation.

The following adjustments have been made to calculate the pro forma EPS and weighted average shares both basic and diluted.

 171,059 existing Class A Legacy Core shares and 6,766 Legacy Core Series A and Series B convertible preferred stock were exchanged for New Core shares using an exchange rate of 1.6001528688 per the merger agreement resulting in a total of 284,546 New Core common shares outstanding with a par value of \$0.0001.

22,153 existing Class A XPDI shares that were not redeemed and 8,625 existing XPDI Class B shares into 30,778 common shares of New
Core. Immediately prior to the Closing, 1,725 Class B common shares will be unvested and will vest upon the date on which the volumeweighted average price of the Class A Common Stock is greater than \$12.50 per share for any 20 trading days within any 30 consecutive
trading day period within five years of the Closing date. These unvested shares are excluded from earnings per share calculations until
such time as conditions are met for vesting.

A fton

in thousands	Existing Shares	Shares just prior to exchange	exchange and redemption of 12,347 shares
Legacy Core common stock	171,059	177,825	284,546
XPDI Class A redeemable shares	22,153	22,153	22,153
XPDI Class B sponsor shares net of unvested shares	8,625	6,900	6,900
Total weighted average common shares—Basic and Diluted (1) and (2)	201,837	206,878	313,599
For the nine months ended September 30, 2021			
Pro forma net income (loss)			\$ (74,087)
Pro forma EPS—Basic and diluted (1)			\$ (0.24)
For the twelve months ended December, 2020			
Pro forma net income (loss)			\$ (94,475)
Pro forma net income (loss)—Basic and Diluted (1)			\$ (0.30)

⁽¹⁾ Potentially dilutive securities are not included in the calculation of diluted net loss per share because to do so would be anti-dilutive. Shares potentially dilutive with the conversion of the convertible debt were not included due to the add back of interest avoided upon conversion causing the conversion to be anti-dilutive.

The discount for the convertible notes will be determined at the time of the conversion event as follows:

Conversion Event	Discount
On or prior to April 19, 2022	20%
On or after April 19, 2022 but prior to April 19, 2023	25%
On or after April 19, 2023 but prior to April 19, 2024	30%
On or after April 19, 2024 but prior to April 19, 2025	35%

The Company currently anticipates a conversion price of 80% of the price paid in a SPAC merger because it currently expects a SPAC merger event to occur prior to April 19, 2022. Assuming a 20% discount to a price of \$10.00 per share at the time of the merger, the applicable conversion price would be expected to be \$8.00 per share.

As of September 30, 2021, the outstanding principal amount of the notes (including all accrued payment-in-kind ("PIK") interest not already added to the principal amount of the notes) was \$427,292. Issuance of the additional \$95 million of convertible notes results in a total of \$522,292 total convertible debt. Should some or all holders of these notes elect to convert such notes would result in potential dilutive issuance of up to 65,287 shares of New Core common stock based on an applicable conversion price of \$8.00 per share that was not included in the calculation of diluted net loss per share above.

- (2) Total Legacy Core outstanding options and warrants after exchanged is 32,195 and 6,808, respectively and are anti-dilutive so not included in the calculation of diluted EPS. 14,892 XPDI warrants are anti-dilutive so not included the calculation of diluted EPS.
- (3) Legacy Core outstanding RSUs are not included as these are contingently issuable shares for which the transaction condition has not yet been met. These RSUs are subject to a time-based vesting condition and a

transaction vesting condition. The transaction vesting condition is satisfied upon the earlier of a change in control or an initial public offering. The transaction vesting condition can be met in future years only with respect to a change in control or waiver of the condition by the Company's board of directors and is not expected to occur, if at all, prior to expiration of the applicable lock-up period. In the event the transaction-based condition were to be met, including by future action of the board of directors, the unrecognized compensation expense for which the requisite service had been provided that would have been recognized was \$526,064 as of September 30, 2021. Included in the total of Legacy Core's unvested RSUs are 9,038 RSUs held by a former employee. These units have met the timebased vesting condition but similarly have not met the transaction-based vesting condition. The transaction-based vesting condition must be satisfied within three years of the former employee's separation date or they will be forfeited. Had the transaction based vesting condition been met, unrecognized compensation expense would have been \$82,674 as of June 30, 2021. This does not account for 6,912 Blockcap RSUs granted to legacy Blockcap shareholders and subject only to time-vesting conditions because the inclusion of such RSUs would be anti-dilutive.

Unvested Legacy Core RSUs as of September 30, 2021	45,793
RSUs to be granted in October 2021	9,777
RSUs to be reserved for future issuances	5,304
Unvested RSUs of Legacy Core (excluding Blockcap)	60,874
Exchange Rate	1.6001528688
Univerted pro forma RSUs of Legacy Core (excluding Blockcap) after applying exchange rate	97 409

INDEX TO FINANCIAL STATEMENTS

P	
O	
V	
V	
E	
R	
į	
B.	
D	
ī	
G	
T	
T	
A	
Ĺ	
1	
П	
F	
R	
Δ	
١.	
ς'	
Г	
R	
I	
I	
T	
ľ	
I	
?	
E	
A	
۱	
C	
C	
I	
J	
ß	
3	
n	
Γ	
ľ	
)	
N	
J	
(
(
)	
R	
I	
)	

Audited Financial Statements of Power & Digital Infrastructure Acquisition Corp. as of December 31, 2020, for the Year Ended	
December 31, 2020 and the Period from December 29, 2020 (inception) Through December 31, 2020	Г.
Report of Independent Registered Public Accounting Firm Balance Sheet as of December 31, 2020	F-4 F-5
Statement of Operations for the Period from December 29, 2020 (inception) Through December 31, 2020	F-6
Statement of Changes in Stockholder's Equity for the Period from December 29, 2020 (inception) Through December 31, 2020	F-7
Statement of Cash Flows for the Period from December 29, 2020 (inception) Through December 31, 2020 Statement of Cash Flows for the Period from December 29, 2020 (inception) Through December 31, 2020	F-8
Notes to Financial Statements	F-9–F-20
Unaudited Condensed Consolidated Financial Statements of Power & Digital Infrastructure Acquisition Corp. as of and for the Nine	1 / 1 20
Months Ended September 30, 2021	
Condensed Consolidated Balance Sheets as of September 30, 2021 (Unaudited) and December 31, 2020	F-21
Unaudited Condensed Consolidated Statements of Operations for the Three and Nine Months Ended September 30, 2021	F-22
Unaudited Condensed Consolidated Statements of Changes in Stockholders' Equity for the Three and Nine Months Ended	
September 30, 2021	F-23
Unaudited Condensed Consolidated Statement of Cash Flows for the Nine Months Ended September 30, 2021	F-24
Notes to Unaudited Condensed Consolidated Financial Statements	F-25-F-45
Report of Independent Registered Public Accounting Firm	F-47
Balance Sheet as of February 12, 2021 (as Restated)	F-48
Notes to Financial Statements (as Restated)	F-49-F-62
CORE SCIENTIFIC HOLDING CO. AND SUBSIDIARIES	
Audited Consolidated Financial Statements of Core Scientific Holding Co. and Subsidiaries as of and for the Years Ended	
December 31, 2020 and 2019	
Report of Independent Registered Public Accounting Firm	F-63
Consolidated Balance Sheets	F-64
Consolidated Statements of Operations and Comprehensive Loss	F-65
Consolidated Statements of Changes in Contingently Redeemable Convertible Preferred Stock and Stockholders' Equity	F-66
Consolidated Statements of Cash Flows	F-67
Notes to Consolidated Financial Statements	F-68-F-98
Consolidated Financial Statements of Core Scientific Holding Co. and Subsidiaries as of September 30, 2021 (Unaudited) and	
December 31, 2020 and for the Nine Months Ended September 30, 2021 and 2020 (Unaudited)	
Consolidated Balance Sheets as of September 30, 2021 (Unaudited) and December 31, 2020	F-99
Consolidated Statements of Operations and Comprehensive Income (Loss) for nine months ended September 30, 2021 and 2020	
(Unaudited)	F-100
Consolidated Statements of Changes in Stockholders' Equity for the nine months ended September 30, 2021 and 2020	
(Unaudited)	F-102
Consolidated Statements of Cash Flows for the nine months ended September 30, 2021 and 2020 (Unaudited)	F-103
Notes to Unaudited Consolidated Financial Statements	F-104–F-127

BLOCKCAP, INC.	
Audited Financial Statements of Blockcap, Inc. as of December 31, 2020, for the Year Ended December 31, 2020 and the Period	
from February 19, 2019 (inception) Through December 31, 2020	
Report of Independent Registered Public Accounting Firm	F-128
Balance Sheets as of December 31, 2020 and 2019	F-129
Statements of Operations for the year ended December 31, 2020 and for the Period from February 19, 2019 (inception) to	
December 31, 2019	F-130
Statements of Stockholders' Equity (Deficit) For the year ended December 31, 2020 and for the period from February 19, 2019	
(inception) to December 31, 2019	F-131
Statements of Cash Flows for the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to	
<u>December 31, 2019</u>	F-132
Notes to Financial Statements	F-133-F-151
Condensed Financial Statements as of June 30, 2021 (Unaudited) and December 31, 2020 and for the Six Months Ended June 30,	
2021 and 2020 (Unaudited)	
Condensed Balance Sheets as of June 30, 2021 (Unaudited) and December 31, 2020	F-152
Condensed Statements of Operations for the six months ended June 30, 2021 and 2020 (Unaudited)	F-153
Condensed Statements of Changes in Stockholders' Equity (Deficit) for the six months ended June 30, 2021 and 2020 (Unaudited)	F-154
Condensed Statements of Cash Flows for the six months ended June 30, 2021 and 2020 (Unaudited)	F-155
Notes to Unaudited Condensed Financial Statements	F-156–F-165
BEP 888, LLC	
Audited Financial Statements for the period from June 1, 2020 (inception) to November 30, 2020	
Report of Independent Registered Public Accounting Firm	F-166
Balance Sheet as of November 30, 2020	F-167
Statement of Operations for the period from June 1, 2020 (inception) to November 30, 2020	F-168
Statement of Members' Equity for the period from June 1, 2020 (inception) to November 30, 2020	F-169
Statement of Cash Flows for the period from June 1, 2020 (inception) to November 30, 2020	F-170
Notes to Financial Statements For the Period from June 1, 2020 (inception) to November 30, 2020	F-171-F-178
BEP 999, LLC	
Audited Financial Statements for the period from November 5, 2020 (inception) to November 30, 2020	
Report of Independent Registered Public Accounting Firm	F-179
Balance Sheet as of November 30, 2020	F-180
Statement of Operations for the period from November 5, 2020 (inception) to November 30, 2020	F-181
Statement of Members' Equity for the period from November 5, 2020 (inception) to November 30, 2020	F-182
Statement of Cash Flows for the period from November 5, 2020 (inception) to November 30, 2020	F-183
Notes to Financial Statements For the Period from November 5, 2020 (inception) to November 30, 2020	F-184–F-186
RME BLACK 100, LLC	
Audited Financial Statements for the period from April 16, 2020 (inception) to November 30, 2020	T 405
Report of Independent Registered Public Accounting Firm	F-187

Balance Sheet as of November 30, 2020	F-188
Statement of Operations for the period from April 16, 2020 (inception) to November 30, 2020	F-189
Statement of Members' Equity for the period from April 16, 2020 (inception) to November 30, 2020	F-190
Statement of Cash Flows for the period from April 16, 2020 (inception) to November 30, 2020	F-191
Notes to Financial Statements For the Period from April 16, 2020 (inception) to November 30, 2020	F-192–F-199
RME BLACK 200, LLC Audited Financial Statements for the period from April 27, 2020 (inception) to November 30, 2020	
Report of Independent Registered Public Accounting Firm	F-200
Balance Sheet as of November 30, 2020	F-200 F-201
Statement of Operations for the period from April 27, 2020 (inception) to November 30, 2020	F-202
Statement of Members' Equity for the period from April 27, 2020 (inception) to November 30, 2020	F-203
Statement of Cash Flows for the period from April 27, 2020 (inception) to November 30, 2020	F-204
Notes to Financial Statements For the Period from April 27, 2020 (inception) to November 30, 2020	F-205-F-212

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Shareholders and Board of Directors of Power & Digital Infrastructure Acquisition Corp.

Opinion on the Financial Statements

We have audited the accompanying balance sheet of Power & Digital Infrastructure Acquisition Corp. (the "Company") as of December 31, 2020, the related statement of operations, stockholders' equity and cash flows for the period from December 29, 2020 (inception) through December 31, 2020, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2020, and the results of its operations and its cash flows for the period from December 29, 2020 (inception) through December 31, 2020, in conformity with accounting principles generally accepted in the United States of America.

Explanatory Paragraph — Going Concern

The accompanying financial statements have been prepared assuming that the Company will continue as a going concern. As more fully described in Note 1 to the financial statements, the Company's ability to execute its business plan is dependent upon its completion of the proposed initial public offering described in Note 3 to the financial statements. The Company has a working capital deficiency as of December 31, 2020 and lacks the financial resources it needs to sustain operations for a reasonable period of time, which is considered to be one year from the issuance date of the financial statements. These conditions raise substantial doubt about the Company's ability to continue as a going concern. Management's plans in regard to these matters are also described in Notes 1 and 3. The financial statements do not include any adjustments that might result from the outcome of this uncertainty.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audit. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audit we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audit included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audit also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audit provides a reasonable basis for our opinion.

/s/ Marcum LLP

Marcum LLP

We have served as the Company's auditor since 2021

New York, NY

January 8, 2021, except for Note 7 as to which the date is February 11, 2021

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. BALANCE SHEET DECEMBER 31, 2020

Assets:	
Deferred offering costs associated with proposed public offering	\$ 15,000
Total Assets	\$ 15,000
Liabilities and Stockholder's Equity:	
Current liabilities:	
Accrued expenses	\$ 400
Total current liabilities	400
Commitments and Contingencies	
Stockholder's Equity:	
Preferred stock, \$0.0001 par value; 1,000,000 shares authorized; none issued and outstanding	_
Class A common stock, \$0.0001 par value; 500,000,000 shares authorized; none issued and outstanding	_
Class B common stock, \$0.0001 par value; 50,000,000 shares authorized; 8,625,000 shares issued and outstanding ⁽¹⁾⁽²⁾	863
Additional paid-in capital	24,137
Accumulated deficit	(10,400)
Total stockholder's equity	14,600
Total Liabilities and Stockholder's Equity	\$ 15,000

⁽¹⁾ This number includes up to 1,125,000 shares of Class B common stock subject to forfeiture if the over-allotment option is not exercised in full or in part by the underwriters. (see Note 4)

⁽²⁾ On February 9, 2021, the Company effected a share capitalization of 1,437,500 shares of Class B common stock, resulting in an aggregate of 8,625,000 shares Class B common stock outstanding. All shares and associated amounts have been retroactively restated to reflect the share capitalization. (see Note 4, 6 and 7)

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. STATEMENT OF OPERATIONS

For the period from December 29, 2020 (inception) through December 31, 2020

General and administrative expenses	\$ 10,400
Net loss	\$ (10,400)
Weighted average shares outstanding, basic and diluted(1)(2)	7,500,000
Basic and diluted net loss per share	\$ (0.00)

⁽¹⁾ This number excludes an aggregate of up to 1,125,000 Class B common stock subject to forfeiture if the over-allotment option is not exercised in full or in part by the underwriters. (see Note 4)

⁽²⁾ On February 9, 2021, the Company effected a share capitalization of 1,437,500 shares of Class B common stock, resulting in an aggregate of 8,625,000 shares Class B common stock outstanding. All shares and associated amounts have been retroactively restated to reflect the share capitalization. (see Note 4, 6 and 7)

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. STATEMENT OF CHANGES IN STOCKHOLDER'S EQUITY

For the period from December 29, 2020 (inception) through December 31, 2020

	Common Stock			Additional		Total	
	Cla	ass A	Class	В	Paid-In	Accumulated	Stockholder's
	Shares	Amount	Shares	Amount	Capital	Deficit	Equity
Balance — December 29, 2020 (inception)	_	<u>s </u>	_	\$ —	<u>s</u> —	<u>s</u> —	\$ —
Issuance of Class B common stock to Sponsors ⁽¹⁾⁽²⁾	_	_	8,625,000	863	24,137	_	25,000
Net loss	_	_	_	_	_	(10,400)	(10,400)
Balance — December 31, 2020		<u>s </u>	8,625,000	\$ 863	\$ 24,137	\$ (10,400)	\$ 14,600

⁽¹⁾ This number includes up to 1,125,000 shares of Class B common stock subject to forfeiture if the over-allotment option is not exercised in full or in part by the underwriters. (see Note 4)

⁽²⁾ On February 9, 2021, the Company effected a share capitalization of 1,437,500 shares of Class B common stock, resulting in an aggregate of 8,625,000 shares Class B common stock outstanding. All shares and associated amounts have been retroactively restated to reflect the share capitalization. (see Note 4, 6 and 7)

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. STATEMENT OF CASH FLOWS

For the period from December 29, 2020 (inception) through December 31, 2020

Cash Flows from Operating Activities:	
Net loss	\$(10,400)
General and administrative expenses paid by related party	10,000
Changes in operating assets and liabilities:	
Accrued expenses	\$ 400
Net cash used in operating activities	
Net change in cash	_
Cash — beginning of the period	_
Cash — end of the period	<u> </u>
Supplemental disclosure of noncash activities:	<u></u> -
Deferred offering costs paid by Sponsor in exchange for issuance of Class B common stock	\$ 15,000

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 1 — Description of Organization, Business Operations, Going Concern and Basis of Presentation

Power & Digital Infrastructure Acquisition Corp. (the "Company") is a blank check company incorporated in Delaware on December 29, 2020. The Company was formed for the purpose of effecting a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses or entities (the "Business Combination"). The Company is an emerging growth company and, as such, the Company is subject to all of the risks associated with emerging growth companies.

As of December 31, 2020, the Company had not commenced any operations. All activity for the period from December 29, 2020 (inception) through December 31, 2020 relates to the Company's formation and the proposed initial public offering described below. The Company will not generate any operating revenues until after the completion of its initial Business Combination, at the earliest. The Company will generate non-operating income in the form of interest income on cash and cash equivalents from the proceeds derived from the Proposed Public Offering (as defined below). The Company has selected December 31 as its fiscal year end.

The Company's sponsor is XPDI Sponsor LLC, a Delaware limited liability company (the "Sponsor"). The Company's ability to commence operations is contingent upon obtaining adequate financial resources through a proposed public offering (the "Proposed Public Offering") of 30,000,000 units of the Company (each, a "Unit" and collectively, the "Units") at \$10.00 per Unit (or 34,500,000 Units if the underwriters' over-allotment option is exercised in full), which is discussed in Note 3, and the sale of an aggregate of 5,666,667 warrants of the Company (or 6,266,667 warrants if the underwriters' over-allotment option is exercised in full) (each, a "Private Placement Warrant" and collectively, the "Private Placement Warrants"), at a price of \$1.50 per Private Placement Warrant in a private placement to the Sponsor and certain funds and accounts managed by subsidiaries of BlackRock, Inc. (the "Anchor Investors"), that will close simultaneously with the Proposed Public Offering.

The Company's management has broad discretion with respect to the specific application of the net proceeds of the Proposed Public Offering and the sale of Private Placement Warrants, although substantially all of the net proceeds are intended to be applied generally toward consummating a Business Combination. There is no assurance that the Company will be able to complete a Business Combination successfully. The Company must complete one or more initial Business Combinations having an aggregate fair market value of at least 80% of the net assets held in the Trust Account (as defined below) (excluding the deferred underwriting commissions and taxes payable by us on the income earned on the trust account) at the time of the agreement to enter into the initial Business Combination. However, the Company only intends to complete a Business Combination if the post-transaction company owns or acquires 50% or more of the issued and outstanding voting securities of the target or otherwise acquires a controlling interest in the target sufficient for it not to be required to register as an investment company under the Investment Company Act 1940, as amended (the "Investment Company Act"). Upon the closing of the Proposed Public Offering, management has agreed that an amount equal to at least \$10.00 per Unit sold in the Proposed Public Offering, including the proceeds from the sale of the Private Placement Warrants, will be held in a trust account ("Trust Account") located in the United States with Continental Stock Transfer & Trust Company acting as trustee, and invested only in U.S. "government securities," within the meaning of Section 2(a)(16) of the Investment Company Act, having a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 promulgated under the Investment Company Act, which invest only in direct U.S. government treasury obligations, as determined by the Company, until the earlier of: (i) the completion of a Business Combination and (ii) the distribution of the Tru

The Company will provide the holders (the "Public Stockholders") of the Company's issued and outstanding shares of Class A common stock, par value \$0.0001 per share, sold in the Proposed Public Offering (the "Public Shares") with the opportunity to redeem all or a portion of their Public Shares upon the completion

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 1 — Description of Organization, Business Operations, Going Concern and Basis of Presentation (cont.)

of a Business Combination either (i) in connection with a stockholder meeting called to approve the Business Combination or (ii) by means of a tender offer. The decision as to whether the Company will seek stockholder approval of a Business Combination or conduct a tender offer will be made by the Company, solely in its discretion. The Public Stockholders will be entitled to redeem their Public Shares for a pro rata portion of the amount then held in the Trust Account (initially anticipated to be \$10.00 per Public Share). The per-share amount to be distributed to Public Stockholders who redeem their Public Shares will not be reduced by the deferred underwriting commissions the Company will pay to the underwriters (as discussed in Note 5). These Public Shares will be recorded at a redemption value and classified as temporary equity upon the completion of the Proposed Public Offering in accordance with the Financial Accounting Standards Board's ("FASB") Accounting Standards Codification ("ASC") Topic 480 "Distinguishing Liabilities from Equity." If the Company seeks stockholder approval, the Company will proceed with a Business Combination if a majority of the outstanding shares are voted by the stockholders at a stockholders meeting to approve the Business Combination, unless applicable law, the Company's corporate governing documents or applicable stock exchange rules require a different vote, in which case the Company will complete its Business Combination only if such requisite vote is received. The Company will not redeem the Public Shares in connection with a Business Combination in an amount that would cause its net tangible assets to be less than \$5,000,001. If a stockholder vote is not required by law and the Company does not decide to hold a stockholder vote for business or other legal reasons, the Company will, pursuant to its Amended and Restated Certificate of Incorporation (the "Certificate of Incorporation"), conduct the redemptions pursuant to the tender offer rules of the U.S. Securities and Exchange Commission ("SEC") and file tender offer documents with the SEC prior to completing a Business Combination. If, however, stockholder approval of the transaction is required by law, or the Company decides to obtain stockholder approval for business or legal reasons, the Company will offer to redeem the Public Shares in conjunction with a proxy solicitation pursuant to the proxy rules and not pursuant to the tender offer rules. Additionally, each Public Stockholder may elect to redeem their Public Shares irrespective of whether they vote for or against the proposed transaction. If the Company seeks stockholder approval in connection with a Business Combination, the initial stockholders (as defined below) will agree to vote their Founder Shares (as defined below in Note 4) and any Public Shares purchased during or after the Proposed Public Offering in favor of a Business Combination. In addition, the initial stockholders will agree to waive their redemption rights with respect to their Founder Shares and Public Shares in connection with the completion of a Business Combination.

The Certificate of Incorporation provides that a Public Stockholder, together with any affiliate of such stockholder or any other person with whom such stockholder is acting in concert or as a "group" (as defined under Section 13 of the Securities Exchange Act of 1934, as amended (the "Exchange Act")), will be restricted from redeeming its shares with respect to more than an aggregate of 15% of the Public Shares, without the prior consent of the Company.

The holders of the Founder Shares (the "initial stockholders") will agree not to propose an amendment to the Certificate of Incorporation (A) to modify the substance or timing of the Company's obligation to provide holders of shares of Class A common stock the right to have their shares redeemed in connection with a Business Combination or to redeem 100% of the Public Shares if the Company does not complete a Business Combination within the Combination Period (as defined below) or (B) with respect to any other provision relating to the rights of holders of Class A common stock, unless the Company provides the Public Stockholders with the opportunity to redeem their Public Shares in conjunction with any such amendment.

If the Company is unable to complete a Business Combination within 24 months from the closing of the Proposed Public Offering (the "Combination Period") and the Company's stockholders have not amended the

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 1 — Description of Organization, Business Operations, Going Concern and Basis of Presentation (cont.)

Certificate of Incorporation to extend such Combination Period, the Company will (1) cease all operations except for the purpose of winding up; (2) as promptly as reasonably possible but not more than 10 business days thereafter, redeem the Public Shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the Trust Account, including interest (less up to \$100,000 of interest to pay dissolution expenses and which interest shall be net of taxes payable by us), divided by the number of then issued and outstanding Public Shares, which redemption will completely extinguish Public Stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any); and (3) as promptly as reasonably possible following such redemption, subject to the approval of the remaining stockholders and the Company's board of directors, liquidate and dissolve, subject in each case to the Company's obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law.

The initial stockholders and Anchor Investors will agree to waive their rights to liquidating distributions from the Trust Account with respect to the Founder Shares if the Company fails to complete a Business Combination within the Combination Period. However, if the initial stockholders and Anchor Investors acquire Public Shares in or after the Proposed Public Offering, they will be entitled to liquidating distributions from the Trust Account with respect to such Public Shares if the Company fails to complete a Business Combination within the Combination Period. The underwriters will agree to waive their rights to the deferred underwriting commission (see Note 5) held in the Trust Account in the event the Company does not complete a Business Combination within in the Combination Period and, in such event, such amounts will be included with the other funds held in the Trust Account that will be available to fund the redemption of the Public Shares. In the event of such distribution, it is possible that the per share value of the residual assets remaining available for distribution (including Trust Account assets) will be only \$10.00. In order to protect the amounts held in the Trust Account, the Sponsor has agreed to be liable to the Company if and to the extent any claims by a third party (except for the Company's independent registered public accounting firm) for services rendered or products sold to the Company, or a prospective target business with which the Company has discussed entering into a transaction agreement (a "Target"), reduce the amount of funds in the Trust Account to below (i) \$10.00 per Public Share or (ii) the lesser amount per Public Share held in the Trust Account as of the date of the liquidation of the Trust Account due to reductions in the value of the trust assets, in each case net of interest which may be withdrawn to pay taxes, provided that such liability will not apply to any claims by a third party or Target that executed a waiver of any and all rights to seek access to the Trust Account nor will it apply to any claims under the Company's indemnity of the underwriters of the Proposed Public Offering against certain liabilities, including liabilities under the Securities Act of 1933, as amended (the "Securities Act"). In the event that an executed waiver is deemed to be unenforceable against a third party, our sponsor will not be responsible to the extent of any liability for such third-party claims. The Company will seek to reduce the possibility that the Sponsor will have to indemnify the Trust Account due to claims of creditors by endeavoring to have all vendors, service providers (other than the Company's independent registered public accounting firm), prospective target businesses or other entities with which the Company does business, execute agreements with the Company waiving any right, title, interest or claim of any kind in or to monies held in the Trust Account.

Going Concern Consideration

As of December 31, 2020, the Company had no cash and a working capital deficit of approximately \$400. Further, the Company has incurred and expects to continue to incur significant costs in pursuit of its financing and acquisition plans. The Company lacks the financial resources it needs to sustain operations for a reasonable period of time, which is considered to be one year from the issuance date of the financial statements. Management's plans to address this need for capital through the Proposed Public Offering. The Company cannot

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 1 — Description of Organization, Business Operations, Going Concern and Basis of Presentation (cont.)

assure that its plans to raise capital or to consummate an initial Business Combination will be successful. In addition, management is currently evaluating the impact of the COVID-19 pandemic on the industry and its effect on the Company's financial position, results of its operations and/or closing the initial public offering or search for a target company.

These factors, among others, raise substantial doubt about the Company's ability to continue as a going concern one year from the date these financial statements are issued. The financial statements do not include any adjustments that might result from its inability to consummate the Proposed Public Offering or its inability to continue as a going concern.

Basis of Presentation

The accompanying financial statements are presented in U.S. dollars in conformity with accounting principles generally accepted in the United States of America ("U.S. GAAP") and pursuant to the rules and regulations of the SEC.

Emerging Growth Company

The Company is an "emerging growth company," as defined in Section 2(a) of the Securities Act, as modified by the Jumpstart Our Business Startups Act of 2012 (the "JOBS Act"), and it may take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies including, but not limited to, not being required to comply with the independent registered public accounting firm attestation requirements of Section 404 of the Sarbanes-Oxley Act of 2002, reduced disclosure obligations regarding executive compensation in its periodic reports and proxy statements, and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved.

Further, Section 102(b)(1) of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies (that is, those that have not had a Securities Act registration statement declared effective or do not have a class of securities registered under the Exchange Act) are required to comply with the new or revised financial accounting standards. The JOBS Act provides that an emerging growth company can elect to opt out of the extended transition period and comply with the requirements that apply to non-emerging growth companies but any such an election to opt out is irrevocable. The Company has elected not to opt out of such extended transition period, which means that when a standard is issued or revised and it has different application dates for public or private companies, the Company, as an emerging growth company, can adopt the new or revised standard at the time private companies adopt the new or revised standard. This may make company of the Company's financial statements with another public company that is neither an emerging growth company nor an emerging growth company that has opted out of using the extended transition period difficult or impossible because of the potential differences in accounting standards used

Note 2 — Summary of Significant Accounting Policies

Use of Estimates

The preparation of financial statements in conformity with U.S. GAAP requires the Company's management to make estimates and assumptions that affect the reported amounts of assets and liabilities and

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 2 — Summary of Significant Accounting Policies (cont.)

disclosure of contingent assets and liabilities during the reporting period. Actual results could differ from those estimates.

Financial Instruments

The fair value of the Company's assets and liabilities, which qualify as financial instruments under FASB ASC 820, "Fair Value Measurements and Disclosures," approximates the carrying amounts represented in the balance sheet.

Deferred Offering Costs Associated with the Proposed Public Offering

Deferred offering costs consist of legal, accounting, and other costs incurred through the balance sheet date that are directly related to the Proposed Public Offering and that will be charged to stockholders' equity upon the completion of the Proposed Public Offering. Should the Proposed Public Offering prove to be unsuccessful, these deferred costs, as well as additional expenses to be incurred, will be charged to operations.

Net Loss Per Share of Common Stock

The Company complies with accounting and disclosure requirements of FASB ASC Topic 260, "Earnings Per Share." Net loss per share is computed by dividing net loss by the weighted average number of shares of common stock outstanding during the period excluding shares of common stock subject to forfeiture. Weighted average shares were reduced for the effect of an aggregate of 1,125,000 shares of Class B common stock that are subject to forfeiture if the over-allotment option is not exercised by the underwriters (Note 5). At December 31, 2020, the Company did not have any dilutive securities and other contracts that could, potentially, be exercised or converted into shares of common stock and then share in the earnings of the Company. As a result, diluted loss per share is the same as basic loss per share for the period presented.

Income Taxes

The Company follows the asset and liability method of accounting for income taxes under FASB ASC 740, "Income Taxes." Deferred tax assets and liabilities are recognized for the estimated future tax consequences attributable to differences between the financial statements carrying amounts of existing assets and liabilities and their respective tax bases. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled.

The effect on deferred tax assets and liabilities of a change in tax rates is recognized in income in the period that included the enactment date. Valuation allowances are established, when necessary, to reduce deferred tax assets to the amount expected to be realized. Deferred tax assets were deemed immaterial as of December 31, 2020.

FASB ASC 740 prescribes a recognition threshold and a measurement attribute for the financial statement recognition and measurement of tax positions taken or expected to be taken in a tax return. For those benefits to be recognized, a tax position must be more likely than not to be sustained upon examination by taxing authorities. There were no unrecognized tax benefits as of December 31, 2020. The Company recognizes accrued interest and penalties related to unrecognized tax benefits as income tax expense. No amounts were accrued for the payment of interest and penalties as of December 31, 2020. The Company is currently not aware of any issues under review that could result in significant payments, accruals or material deviation from its position. The Company is subject to income tax examinations by major taxing authorities since inception.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 2 — Summary of Significant Accounting Policies (cont.)

Recent Accounting Pronouncements

The Company's management does not believe that any recently issued, but not yet effective, accounting pronouncements, if currently adopted, would have a material effect on the accompanying financial statements.

Note 3 — Proposed Public Offering

Pursuant to the Proposed Public Offering, the Company offered for sale 30,000,000 Units at a price of \$10.00 per Unit. Each Unit consists of one share of Class A common stock, and one-fourth of one redeemable warrant (each, a "Public Warrant"). Each Public Warrant entitles the holder to purchase one share of Class A common stock at a price of \$11.50 per share, subject to adjustment (see Note 6).

The Company has granted the underwriters a 45-day option from the date of the final prospectus relating to the Proposed Public Offering to purchase up to 4,500,000 additional Units to cover over-allotments, if any, at the Proposed Public Offering price, less underwriting discounts and commissions.

Note 4 — Related Party Transactions

Founder Shares

On December 31, 2020, the Sponsor paid \$25,000 to cover for certain offering costs on behalf of the Company in exchange for issuance of 7,187,500 shares of the Company's Class B common stock, par value \$0.0001 per share, (the "Founder Shares"). On February 9, 2021, the Company effected a share capitalization of 1,437,500 shares of Class B common stock, resulting in an aggregate of 8,625,000 shares of Class B common stock outstanding (see Note 7). All shares and associated amounts have been retroactively restated to reflect the share capitalization. Up to 1,125,000 Founder Shares are subject to forfeiture to the extent that the over-allotment option is not exercised in full by the underwriters. The forfeiture will be adjusted to the extent that the over-allotment option is not exercised in full by the underwriters will represent 20.0% of the Company's issued and outstanding shares after the Proposed Public Offering.

In February 2021, the Sponsor agreed to sell to the Anchor Investors 1,350,000 Founder Shares (or 1,552,500 Founder Shares if the underwriters' over-allotment option is exercised in full) and the Anchor Investors agreed to purchase from the Sponsor on the date of the initial business combination an aggregate of 1,350,000 Founder Shares (or 1,552,500 Founder Shares if the underwriters' over-allotment option is exercised in full) for an aggregate purchase price of approximately \$3,913 (or \$4,500 if the underwriters' over-allotment option is exercised in full), or approximately \$0.003 per share.

The initial stockholders and the Anchor Investors will agree, subject to limited exceptions, not to transfer, assign or sell any of the Founder Shares until the earlier to occur of: (A) one year after the completion of the initial Business Combination; and (B) subsequent to the initial Business Combination (x) if the last reported sale price of the Class A common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20-trading days within any 30-trading day period commencing at least 150 days after the initial Business Combination or (y) the date on which the Company completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of the Company's stockholders having the right to exchange their shares of common stock for cash, securities or other property. Any permitted transferees would be subject to the same restrictions and other agreements of the initial stockholders and the Anchor Investors with respect to any Founder Shares.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 4 — Related Party Transactions (cont.)

Private Placement Warrants

The Sponsor and the Anchor Investors will agree to purchase an aggregate of 5,666,667 Private Placement Warrants (or 6,266,667 Private Placement Warrants if the underwriters' over-allotment option is exercised in full), at a price of \$1.50 per Private Placement Warrant, or \$8.5 million in the aggregate (or approximately \$9.4 million if the underwriters' over-allotment option is exercised in full) in a private placement that will occur simultaneously with the closing of the Proposed Public Offering.

Each Private Placement Warrant is exercisable for one whole share of Class A common stock at a price of \$11.50 per share. A portion of the proceeds from the sale of the Private Placement Warrants to the Sponsor will be added to the proceeds from the Proposed Public Offering to be held in the Trust Account. If the Company does not complete a Business Combination within the Combination Period, the Private Placement Warrants will expire worthless.

The purchasers of the Private Placement Warrants will agree, subject to limited exceptions, not to transfer, assign or sell any of their Private Placement Warrants (except to permitted transferees) until 30 days after the completion of the initial Business Combination.

Related Party Loans

On December 31, 2020, the Sponsor agreed to loan the Company an aggregate of up to \$300,000 to cover expenses related to the Proposed Public Offering pursuant to a promissory note (the "Note"). This loan is non-interest bearing and payable on the earlier of June 30, 2021 or the completion of the Proposed Public Offering. As of December 31, 2020, the Company had not borrowed any amount under the Note. As of February 10, 2021, the Company had borrowed approximately \$90,000 under the Note (see Note 7).

In addition, in order to finance transaction costs in connection with a Business Combination, the Sponsor or an affiliate of the Sponsor, or certain of the Company's officers and directors may, but are not obligated to, loan the Company funds as may be required ("Working Capital Loans"). If the Company completes a Business Combination, the Company would repay the Working Capital Loans out of the proceeds of the Trust Account released to the Company. Otherwise, the Working Capital Loans would be repaid only out of funds held outside the Trust Account. In the event that a Business Combination does not close, the Company may use a portion of proceeds held outside the Trust Account to repay the Working Capital Loans but no proceeds held in the Trust Account would be used to repay the Working Capital Loans. The Working Capital Loans would either be repaid upon consummation of a Business Combination or, at the lender's discretion, up to \$1.5 million of such Working Capital Loans may be convertible into warrants of the post Business Combination entity at a price of \$1.50 per warrant. The warrants would be identical to the Private Placement Warrants. Except for the foregoing, the terms of such Working Capital Loans, if any, have not been determined and no written agreements exist with respect to such loans. To date, the Company had no borrowings under the Working Capital Loans.

Administrative Services Agreement

The Company has entered into an agreement that provides that, commencing on the date that the Company's securities are first listed on the Nasdaq through the earlier of consummation of the initial Business Combination and the Company's liquidation, the Company will pay affiliates of the Sponsor a total of \$20,000 per month for office space, administrative and support services.

The Sponsor, officers and directors, or any of their respective affiliates, will be reimbursed for any out-of-pocket expenses incurred in connection with activities on the Company's behalf such as identifying

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 4 — Related Party Transactions (cont.)

potential target businesses and performing due diligence on suitable Business Combinations. The Company's audit committee will review on a quarterly basis all payments that were made by us to the Sponsor, directors, officers or the Company's or any of their affiliates.

Business Combination Payments

The Company may make a cash payment to XMS Capital or its affiliates for any financial advisory, placement agency or other similar investment banking services that XMS Capital or its affiliates may provide to the Company, in connection with its initial Business Combination, and reimburse XMS Capital or its affiliates for any out-of-pocket expenses incurred by them in connection with the performance of such services.

Note 5 — Commitments & Contingencies

Registration Rights

The holders of Founder Shares, Private Placement Warrants and warrants that may be issued upon conversion of Working Capital Loans, if any (and any shares of common stock issuable upon the exercise of the Private Placement Warrants or warrants issued upon conversion of the Working Capital Loans and upon conversion of the Founder Shares), will be entitled to registration rights pursuant to a registration rights agreement to be signed prior to the consummation of the Proposed Public Offering. These holders will be entitled to certain demand and "piggyback" registration rights. However, the registration rights agreement provides that we will not be required to effect or permit any registration or cause any registration statement to become effective until termination of the applicable lock-up period. The Company will bear the expenses incurred in connection with the filing of any such registration statements.

Underwriting Agreement

The underwriters are entitled to an underwriting discount of \$0.20 per Unit, or \$6.0 million in the aggregate (or \$6.9 million in the aggregate if the underwriters' over-allotment option is exercised in full), payable upon the closing of the Proposed Public Offering. An additional fee of \$0.35 per Unit, or \$10.5 million in the aggregate (or approximately \$12.1 million in the aggregate if the underwriters' over-allotment option is exercised in full) will be payable to the underwriters for deferred underwriting commissions. The deferred fee will become payable to the underwriters from the amounts held in the Trust Account solely in the event that the Company completes a Business Combination, subject to the terms of the underwriting agreement.

Risks and Uncertainties

Management is currently evaluating the impact of the COVID-19 pandemic on the industry and has concluded that while it is reasonably possible that the virus could have a negative effect on the Company's financial position, results of its operations and/or closing the initial public offering or search for a target company, the specific impact is not readily determinable as of the date of these financial statements. The financial statements do not include any adjustments that might result from the outcome of this uncertainty.

Note 6 — Stockholder's Equity

Class A Common Stock — The Company is authorized to issue 500,000,000 shares of Class A common stock with a par value of \$0.0001 per share. As of December 31, 2020, there were no shares of Class A common stock issued or outstanding.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 6 — Stockholder's Equity (cont.)

Class B Common Stock — The Company is authorized to issue 50,000,000 shares of Class B common stock with a par value of \$0.0001 per share. On December 31, 2020, the Company issued 7,187,500 shares of Class B common stock to the Company by the Sponsor for an aggregate price of \$25,000. On February 9, 2021, the Company effected a share capitalization of 1,437,500 shares of Class B common stock, resulting in an aggregate of 8,625,000 shares of Class B common stock outstanding. All shares and associated amounts have been retroactively restated to reflect the share capitalization. Up to an aggregate of 1,125,000 shares of Class B common stock are subject to forfeiture to the extent that the underwriter's overallotment option is not exercised in full or in part, so that the initial stockholders will collectively own 20% of the Company's issued and outstanding common stock after the Proposed Public Offering. All shares and associated amounts have been retroactively restated to reflect the stock split.

Common stockholders of record are entitled to one vote for each share held on all matters to be voted on by stockholders and vote together as a single class, except as required by law; provided, that, prior to the Company's initial Business Combination, holders of the Class B common stock will have the right to appoint all of the Company's directors and remove members of the board of directors for any reason, and holders of the Class A common stock will not be entitled to vote on the appointment of directors during such time.

The Class B common stock will automatically convert into Class A common stock at the time of the initial Business Combination, or earlier at the option of the holder, on a one-for-one basis, subject to adjustment for stock splits, stock dividends, reorganizations, recapitalizations and the like, and subject to further adjustment as provided herein. In the case that additional shares of Class A common stock, or equity-linked securities, are issued or deemed issued in excess of the amounts issued in the Proposed Public Offering and related to the closing of the initial Business Combination, the ratio at which the shares of Class B common stock will convert into shares of Class A common stock will be adjusted (unless the holders of a majority of the issued and outstanding shares of the Class B common stock agree to waive such anti-dilution adjustment with respect to any such issuance or deemed issuance) so that the number of shares of Class A common stock issuable upon conversion of all shares of Class B common stock will equal, in the aggregate, on an as-converted basis, 20% of the sum of all shares of common stock issued and outstanding upon the completion of the Proposed Public Offering plus all shares of Class A common stock and equity-linked securities issued or deemed issued in connection with the initial Business Combination, excluding any shares or equity-linked securities issued, or to be issued, to any seller in the initial Business Combination.

Preferred Stock — The Company is authorized to issue 1,000,000 shares of preferred stock, par value \$0.0001 per share, with such designations, voting and other rights and preferences as may be determined from time to time by the Company's board of directors. As of December 31, 2020, there were no shares of preferred stock issued or outstanding.

Warrants — Public Warrants may only be exercised for a whole number of shares. No fractional Public Warrants will be issued upon separation of the Units and only whole Public Warrants will trade. The Public Warrants will become exercisable on the later of (a) 30 days after the completion of a Business Combination or (b) 12 months from the closing of the Proposed Public Offering; provided in each case that the Company has an effective registration statement under the Securities Act covering the shares of Class A common stock issuable upon exercise of the Public Warrants and a current prospectus relating to them is available (or the Company permits holders to exercise their Public Warrants on a cashless basis and such cashless exercise is exempt from registration under the Securities Act). The Company has agreed that as soon as practicable, but in no event later than 20 business days after the closing of its initial Business Combination, the Company will use its commercially reasonable efforts to file with the SEC and have an effective registration statement covering the

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 6 — Stockholder's Equity (cont.)

shares of Class A common stock issuable upon exercise of the warrants and will use its commercially reasonable efforts to cause the same to become effective within 60 business days after the closing of the Company's initial Business Combination and to maintain a current prospectus relating to those shares of Class A common stock until the warrants expire or are redeemed. If the shares issuable upon exercise of the warrants are not registered under the Securities Act in accordance with the above requirements, the Company will be required to permit holders to exercise their warrants on a cashless basis. However, no warrant will be exercisable for cash or on a cashless basis, and the Company will not be obligated to issue any shares to holders seeking to exercise their warrants, unless the issuance of the shares upon such exercise is registered or qualified under the securities laws of the state of the exercising holder, or an exemption from registration is available. Notwithstanding the above, if the Company's shares of Class A common stock are at the time of any exercise of a warrant not listed on a national securities exchange such that they satisfy the definition of a "covered security" under Section 18(b)(1) of the Securities Act, the Company may, at its option, require holders of Public Warrants who exercise their warrants to do so on a "cashless basis" in accordance with Section 3(a)(9) of the Securities Act and, in the event the Company so elects, it will not be required to file or maintain in effect a registration statement, and in the event the Company does not so elect, it will use its commercially reasonable efforts to register or qualify the shares under applicable blue sky laws to the extent an exemption is not available.

The warrants have an exercise price of \$11.50 per share, subject to adjustments, and will expire five years after the completion of a Business Combination or earlier upon redemption or liquidation. In addition, if (x) the Company issues additional shares of Class A common stock or equitylinked securities for capital raising purposes in connection with the closing of the initial Business Combination at an issue price or effective issue price of less than \$9.20 per share of Class A common stock (with such issue price or effective issue price to be determined in good faith by the Company's board of directors and, in the case of any such issuance to the Sponsor or its affiliates, without taking into account any Founder Shares held by the Sponsor or such affiliates, as applicable, prior to such issuance) (the "Newly Issued Price"), (y) the aggregate gross proceeds from such issuances represent more than 60% of the total equity proceeds, and interest thereon, available for the funding of the initial Business Combination on the date of the consummation of the initial Business Combination (net of redemptions), and (z) the volume weighted average trading price of the Class A common stock during the 20 trading day period starting on the trading day prior to the day on which the Company consummates the initial Business Combination (such price, the "Market Value") is below \$9.20 per share, then the exercise price of the warrants will be adjusted (to the nearest cent) to be equal to 115% of the higher of the Market Value and the Newly Issued Price, the \$18.00 per share redemption trigger prices described below under "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$18.00" and "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$10.00" will be adjusted (to the nearest cent) to be equal to 180% of the higher of the Market Value and the Newly Issued Price, and the \$10.00 per share redemption trigger price described below under "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$18.00" will be adjusted (to the nearest cent) to be equal to the higher of the Market Value and the Newly Issued Price.

The Private Placement Warrants are identical to the Public Warrants, except that the Private Placement Warrants and the shares of Class A common stock issuable upon exercise of the Private Placement Warrants will not be transferable, assignable or salable until 30 days after the completion of a Business Combination, subject to certain limited exceptions. Additionally, the Private Placement Warrants will be non-redeemable (except as described below in "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$10.00") so long as they are held by the Sponsor or its permitted transferees. If the Private Placement Warrants are held by someone other than the Sponsor, the Anchor Investors or their respective permitted transferees, the Private Placement Warrants will be redeemable by the Company and exercisable by such holders on the same basis as the Public Warrants.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 6 — Stockholder's Equity (cont.)

Redemption of warrants when the price per share of Class A common stock equals or exceeds \$18.00: Once the warrants become exercisable, the Company may call the outstanding warrants for redemption (except as described herein with respect to the Private Placement Warrants):

- in whole and not in part;
- at a price of \$0.01 per warrant;
- upon a minimum of 30 days' prior written notice of redemption to each warrant holder; and
- if, and only if, the last reported sale price of Class A common stock for any 20 trading days within a 30-trading day period ending on the third trading day prior to the date on which the Company sends the notice of redemption to the warrant holders (the "Reference Value") equals or exceeds \$18.00 per share (as adjusted).

The Company will not redeem the warrants as described above unless a registration statement under the Securities Act covering the issuance of the shares of Class A common stock issuable upon exercise of the warrants is then effective and a current prospectus relating to those shares of Class A common stock is available throughout the 30-day redemption period. If and when the warrants become redeemable by the Company, the Company may exercise its redemption right even if it is unable to register or qualify the underlying securities for sale under all applicable state securities laws.

Redemption of warrants when the price per share of Class A common stock equals or exceeds \$10.00: Once the warrants become exercisable, the Company may redeem the outstanding warrants:

- in whole and not in part;
- at \$0.10 per warrant upon a minimum of 30 days' prior written notice of redemption provided that holders will be able to exercise their warrants on a cashless basis prior to redemption and receive that number of shares determined by reference to an agreed table based on the redemption date and the "fair market value" (as defined below) of Class A common stock;
- if, and only if, the Reference Value equals or exceeds \$10.00 per share (as adjusted for adjustments to the number of shares issuable upon exercise or the exercise price of a warrant as described herein under the heading "Description of Securities Warrants Public Stockholders' Warrants Anti-dilution Adjustments"); and
- if the Reference Value is less than \$18.00 per share (as adjusted for adjustments to the number of shares issuable upon exercise or the exercise price of a warrant as described herein under the heading "Description of Securities Warrants Public Stockholders' Warrants Anti-dilution Adjustments"), the Private Placement Warrants must also concurrently be called for redemption on the same terms as the outstanding Public Warrants, as described above.

The "fair market value" of Class A common stock shall mean the volume weighted average price of Class A common stock during the 10 trading days immediately following the date on which the notice of redemption is sent to the holders of warrants. In no event will the warrants be exercisable in connection with this redemption feature for more than 0.361 shares of Class A common stock per warrant (subject to adjustment).

In no event will the Company be required to net cash settle any warrant. If the Company is unable to complete a Business Combination within the Combination Period and the Company liquidates the funds held in the Trust Account, holders of warrants will not receive any of such funds with respect to their warrants, nor will they receive any distribution from the Company's assets held outside of the Trust Account with the respect to such warrants. Accordingly, the warrants may expire worthless.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENTS

Note 7 — Subsequent Events

On February 9, 2021, the Company effected a share capitalization of 1,437,500 shares of Class B common stock, resulting in an aggregate of 8,625,000 shares of Class B common stock outstanding. All shares and associated amounts have been retroactively restated to reflect the share capitalization.

As of February 10, 2021, the Company had borrowed approximately \$90,000 under the Note.

The Company has evaluated subsequent events after the balance sheet date through the date the financial statements were issued. Based on this review, the Company did not identify any subsequent events that would have required adjustment or disclosure in the financial statements.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. UNAUDITED CONDENSED CONSOLIDATED BALANCE SHEETS

	September 30, 2021 (Unaudited)	December 31, 2020
Assets:	(0.00000)	
Current assets:		
Cash	\$ 1,598,506	\$ —
Prepaid expenses	446,855	
Total current assets	2,045,361	_
Investments held in Trust Account	345,027,247	_
Deferred offering costs		15,000
Total Assets	\$347,072,608	\$ 15,000
Liabilities, Class A Common Stock Subject to Possible Redemption and Stockholders' Deficit:		
Current liabilities:		
Accounts payable	\$ 721,484	\$ —
Accrued expenses	3,225,538	400
Franchise tax payable	148,395	
Total current liabilities	4,095,417	400
Derivative warrant liabilities	36,931,330	_
Deferred underwriting commissions	12,075,000	
Total liabilities	53,101,747	400
Commitments and Contingencies		
Class A common stock subject to possible redemption, \$0.0001 par value; 34,500,000 shares at \$10.00 per share	345,000,000	_
Stockholders' Equity (Deficit):		
Preferred stock, \$0.0001 par value; 1,000,000 shares authorized; none issued and outstanding	_	_
Class A common stock, \$0.0001 par value; 500,000,000 shares authorized	_	_
Class B common stock, \$0.0001 par value; 50,000,000 shares authorized; 8,625,000 shares issued and		
outstanding as of September 30, 2021 and December 31, 2020	863	863
Additional paid-in capital		24,137
Accumulated deficit	(51,030,002)	(10,400
Total stockholders' equity (deficit)	(51,029,139)	14,600
Total Liabilities, Class A Common Stock Subject to Possible Redemption and Stockholders' Equity		
(Deficit)	\$347,072,608	\$ 15,000

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

	For the Three Months Ended September 30, 2021		For the Nine Months Ended September 30, 2021
General and administrative expenses	\$ 2,387,202	\$	4,077,585
General and administrative expenses — related party	60,000		160,000
Franchise tax expenses	49,863		147,995
Loss from operations	 (2,497,065)	_	(4,385,580)
Change in fair value of derivative warrant liabilities	(16,231,910)		(13,903,830)
Offering costs associated with derivative warrant liabilities	_		(1,055,577)
Income from investments held in Trust Account	 4,440		27,247
Net loss	\$ (18,724,535)	\$	(19,317,740)
Weighted average shares outstanding of Class A common stock, basic and diluted	34,500,000		29,192,308
Basic and diluted net loss per share, Class A common stock	\$ (0.43)	\$	(0.51)
Weighted average shares outstanding of Class B common stock, basic and diluted	8,625,000	_	8,451,923
Basic and diluted net loss per share, Class B common stock	\$ (0.43)	\$	(0.51)

 $\label{thm:companying} \textit{The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.}$

$POWER \& DIGITAL INFRASTRUCTURE ACQUISITION CORP. \\ CONDENSED CONSOLIDATED STATEMENTS OF CHANGES IN STOCKHOLDERS' EQUITY (DEFICIT)$

For The Three and Nine Months Ended September 30, 2021 (Unaudited)

		Com	mon Stock				Total
	Cl	ass A	Class	В	Additional Paid-In	Accumulated	Stockholders' Equity
	Shares	Amount	Shares	Amount	Capital	Deficit	(Deficit)
Balance — December 31, 2020	_	\$ —	8,625,000	\$ 863	\$ 24,137	\$ (10,400)	\$ 14,600
Accretion of Class A common stock subject to possible							
redemption amount	_	_	_	_	(24,137)	(31,701,862)	(31,725,999)
Net income						4,962,873	4,962,873
Balance — March 31, 2021 (restated)	_	s —	8,625,000	\$ 863	s —	\$(26,749,389)	\$(26,748,526)
Net loss	_	_	_	_	_	(5,556,078)	(5,556,078)
Balance — June 30, 2021 (restated)		<u>s </u>	8,625,000	\$ 863	<u>s — </u>	\$(32,305,467)	\$(32,304,604)
Net loss	_	_	_	_	_	(18,724,535)	(18,724,535)
Balance — September 30, 2021		\$ <u> </u>	8,625,000	\$ 863	s —	\$(51,030,002)	\$(51,029,139)

 $The\ accompanying\ notes\ are\ an\ integral\ part\ of\ these\ unaudited\ condensed\ consolidated\ financial\ statements.$

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. CONDENSED CONSOLIDATED STATEMENT OF CASH FLOWS

For The Nine Months Ended September 30, 2021 (Unaudited)

Cash Flows from Operating Activities:	_
Net loss	\$ (19,317,740)
Adjustments to reconcile net loss to net cash used in operating activities:	
Change in fair value of derivative warrant liabilities	13,903,830
Offering costs associated with derivative warrant liabilities	1,055,577
Income from investments held in Trust Account	(27,247)
General and administrative expenses paid by related party under promissory note	144
Changes in operating assets and liabilities:	
Prepaid expenses	(446,855)
Accrued expenses	2,718,988
Accounts payable	721,484
Franchise tax payable	147,995
Net cash used in operating activities	(1,243,824)
Cash Flows from Investing Activities	
Cash deposited in Trust Account	(345,000,000)
Net cash used in investing activities	(345,000,000)
Cash Flows from Financing Activities:	
Proceeds received from initial public offering, gross	345,000,000
Proceeds received from private placement	9,400,000
Repayment of note payable to related party	(90,035)
Offering costs paid	(6,467,635)
Net cash provided by financing activities	347,842,330
Net change in cash	1,598,506
Cash — beginning of the period	_
Cash — end of the period	\$ 1,598,506
Supplemental disclosure of noncash activities:	
Offering costs included in accrued expenses	\$ 506,550
Offering costs paid by related party under promissory note	\$ 89,891
Deferred underwriting commissions in connection with the initial public	\$ 12,075,000

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Note 1 — Description of Organization and Business Operations

Power & Digital Infrastructure Acquisition Corp. (the "Company") is a blank check company incorporated in Delaware on December 29, 2020. The Company was formed for the purpose of effecting a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses or entities (the "Business Combination"). The Company is an emerging growth company and, as such, the Company is subject to all of the risks associated with emerging growth companies.

The Company may pursue targets in any industry for purposes of consummating a Business Combination.

As of September 30, 2021, the Company had not commenced any operations. All activity for the period from December 29, 2020 (inception) through September 30, 2021 relates to the Company's formation and the initial public offering (the "Initial Public Offering") described below, and subsequent to the Initial Public Offering, identifying a target company for a Business Combination. The Company will not generate any operating revenues until after the completion of its initial Business Combination, at the earliest. The Company will generate non-operating income in the form of interest and other income on investments of the proceeds derived from the Initial Public Offering.

The Company has selected December 31 as its fiscal year end.

The Company's sponsor is XPDI Sponsor LLC, a Delaware limited liability company (the "Sponsor"). The registration statement for the Company's Initial Public Offering was declared effective on February 9, 2021. On February 12, 2021, the Company consummated its Initial Public Offering of 34,500,000 units (the "Units" and, with respect to the Class A common stock included in the Units being offered, the "Public Shares"), including the exercise of the underwriters' option to purchase 4,500,000 additional Units (the "Over-Allotment Units"), at \$10.00 per Unit, which generated gross proceeds of \$345.0 million, and incurring offering costs of approximately \$19.2 million, of which approximately \$12.1 million in deferred underwriting commissions (Note 5).

Simultaneously with the closing of the Initial Public Offering, the Company consummated the private placement ("Private Placement") of 6,266,667 warrants (each, a "Private Placement Warrant" and collectively, the "Private Placement Warrants"), at a price of \$1.50 per Private Placement Warrant, to the Sponsor and to certain qualified institutional buyers or institutional accredited investors, including certain funds and accounts managed by subsidiaries of BlackRock, Inc. (the "Anchor Investors"), generating proceeds of \$9.4 million (Note 4).

Upon the closing of the Initial Public Offering and the Private Placement, \$345.0 million (\$10.00 per Unit) of the net proceeds of the Initial Public Offering and certain of the proceeds of the Private Placement was placed in a trust account ("Trust Account") located in the United States with Continental Stock Transfer & Trust Company acting as trustee, and will be invested only in U.S. "government securities," within the meaning of Section 2(a)(16) of the Investment Company Act of 1940, as amended (the "Investment Company Act"), having a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 promulgated under the Investment Company Act, which invest only in direct U.S. government treasury obligations, as determined by the Company, until the earlier of: (i) the completion of a Business Combination and (ii) the distribution of the Trust Account as described below.

The Company's management has broad discretion with respect to the specific application of the net proceeds of the Initial Public Offering and the sale of Private Placement Warrants, although substantially all of the net proceeds are intended to be applied generally toward consummating a Business Combination. There is no assurance that the Company will be able to complete a Business Combination successfully. The Company must complete one or more initial Business Combinations having an aggregate fair market value of at least 80% of the

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

net assets held in the Trust Account (as defined below) (excluding the deferred underwriting commissions and taxes payable by us on the income earned on the trust account) at the time of the agreement to enter into the initial Business Combination. However, the Company only intends to complete a Business Combination if the post-transaction company owns or acquires 50% or more of the issued and outstanding voting securities of the target or otherwise acquires a controlling interest in the target sufficient for it not to be required to register as an investment company under the Investment Company Act.

The Company will provide the holders (the "Public Stockholders") of the Public Shares with the opportunity to redeem all or a portion of their Public Shares upon the completion of a Business Combination either (i) in connection with a stockholder meeting called to approve the Business Combination or (ii) by means of a tender offer. The decision as to whether the Company will seek stockholder approval of a Business Combination or conduct a tender offer will be made by the Company, solely in its discretion. The Public Stockholders will be entitled to redeem their Public Shares for a pro rata portion of the amount then held in the Trust Account (initially anticipated to be \$10.00 per Public Share). The per-share amount to be distributed to Public Stockholders who redeem their Public Shares will not be reduced by the deferred underwriting commissions the Company will pay to the underwriters (as discussed in Note 5). These Public Shares will be recorded at a redemption value and classified as temporary equity upon the completion of the Initial Public Offering in accordance with the Financial Accounting Standards Board's ("FASB") Accounting Standards Codification ("ASC") Topic 480 "Distinguishing Liabilities from Equity" ("ASC 480"). If the Company seeks stockholder approval, the Company will proceed with a Business Combination if a majority of the outstanding shares are voted by the stockholders at a stockholders' meeting to approve the Business Combination, unless applicable law, the Company's corporate governing documents or applicable stock exchange rules require a different vote, in which case the Company will complete its Business Combination only if such requisite vote is received. The Company will not redeem the Public Shares in connection with a Business Combination in an amount that would cause its net tangible assets to be less than \$5,000,001 upon the consummation of the Business Combination. If a stockholder vote is not required by law and the Company does not decide to hold a stockholder vote for business or other legal reasons, the Company will, pursuant to its Amended and Restated Certificate of Incorporation (the "Certificate of Incorporation"), conduct the redemptions pursuant to the tender offer rules of the U.S. Securities and Exchange Commission ("SEC") and file tender offer documents with the SEC prior to completing a Business Combination. If, however, stockholder approval of the transaction is required by law, or the Company decides to obtain stockholder approval for business or legal reasons, the Company will offer to redeem the Public Shares in conjunction with a proxy solicitation pursuant to the proxy rules and not pursuant to the tender offer rules. Additionally, each Public Stockholder may elect to redeem their Public Shares irrespective of whether they vote for or against the proposed transaction. If the Company seeks stockholder approval in connection with a Business Combination, the initial stockholders (as defined below) agreed to vote their Founder Shares (as defined below in Note 4) and any Public Shares purchased during or after the Initial Public Offering in favor of a Business Combination. In addition, the initial stockholders agreed to waive their redemption rights with respect to their Founder Shares and Public Shares in connection with the completion of a Business Combination.

The Certificate of Incorporation provides that a Public Stockholder, together with any affiliate of such stockholder or any other person with whom such stockholder is acting in concert or as a "group" (as defined under Section 13 of the Securities Exchange Act of 1934, as amended (the "Exchange Act")), will be restricted from redeeming its shares with respect to more than an aggregate of 15% of the Public Shares, without the prior consent of the Company.

The holders of the Founder Shares (the "initial stockholders") agreed not to propose an amendment to the Certificate of Incorporation (A) to modify the substance or timing of the Company's obligation to provide holders of shares of Class A common stock the right to have their shares redeemed in connection with a Business Combination or to redeem 100% of the Public Shares if the Company does not complete a Business Combination

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

within the Combination Period (as defined below) or (B) with respect to any other provision relating to the rights of holders of Class A common stock, unless the Company provides the Public Stockholders with the opportunity to redeem their Public Shares in conjunction with any such amendment.

If the Company is unable to complete a Business Combination within 24 months from the closing of the Initial Public Offering, or February 12, 2023 (the "Combination Period"), and the Company's stockholders have not amended the Certificate of Incorporation to extend such Combination Period, the Company will (1) cease all operations except for the purpose of winding up; (2) as promptly as reasonably possible but not more than 10 business days thereafter, redeem the Public Shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the Trust Account, including interest (less up to \$100,000 of interest to pay dissolution expenses and which interest shall be net of taxes payable by us), divided by the number of then issued and outstanding Public Shares, which redemption will completely extinguish Public Stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any); and (3) as promptly as reasonably possible following such redemption, subject to the approval of the remaining stockholders and the Company's board of directors, liquidate and dissolve, subject in each case to the Company's obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law.

The initial stockholders and Anchor Investors agreed to waive their rights to liquidating distributions from the Trust Account with respect to the Founder Shares if the Company fails to complete a Business Combination within the Combination Period. However, if the initial stockholders and Anchor Investors acquire Public Shares in or after the Initial Public Offering, they will be entitled to liquidating distributions from the Trust Account with respect to such Public Shares if the Company fails to complete a Business Combination within the Combination Period. The underwriters agreed to waive their rights to the deferred underwriting commission (see Note 5) held in the Trust Account in the event the Company does not complete a Business Combination within the Combination Period and, in such event, such amounts will be included with the other funds held in the Trust Account that will be available to fund the redemption of the Public Shares. In the event of such distribution, it is possible that the per share value of the residual assets remaining available for distribution (including Trust Account assets) will be only \$10.00. In order to protect the amounts held in the Trust Account, the Sponsor agreed to be liable to the Company if and to the extent any claims by a third party (except for the Company's independent registered public accounting firm) for services rendered or products sold to the Company, or a prospective target business with which the Company has discussed entering into a transaction agreement (a "Target"), reduce the amount of funds in the Trust Account to below (i) \$10.00 per Public Share or (ii) the lesser amount per Public Share held in the Trust Account as of the date of the liquidation of the Trust Account due to reductions in the value of the trust assets, in each case net of interest which may be withdrawn to pay taxes, provided that such liability will not apply to any claims by a third party or Target that executed a waiver of any and all rights to seek access to the Trust Account nor will it apply to any claims under the Company's indemnity of the underwriters of the Initial Public Offering against certain liabilities, including liabilities under the Securities Act of 1933, as amended (the "Securities Act"). In the event that an executed waiver is deemed to be unenforceable against a third party, the Sponsor will not be responsible to the extent of any liability for such third-party claims. The Company will seek to reduce the possibility that the Sponsor will have to indemnify the Trust Account due to claims of creditors by endeavoring to have all vendors, service providers (other than the Company's independent registered public accounting firm), prospective target businesses or other entities with which the Company does business, execute agreements with the Company waiving any right, title, interest or claim of any kind in or to monies held in the Trust Account.

Proposed Business Combination

On July 20, 2021, the Company entered into an Agreement and Plan of Merger and Reorganization (as amended by the First Amendment thereto, dated as of October 1, 2021, as further amended by the Second Amendment

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

thereto, dated as of December 29, 2021, and as it may be further amended and/or restated from time to time, the "Agreement"), with XPDI Merger Sub Inc., a wholly owned subsidiary of the Company ("Merger Sub" and, together with the Company, the "XPDI Parties"), and Core Scientific Holding Co. ("Core Scientific"). The Agreement and the transactions contemplated thereby (collectively, the "Proposed Business Combination") were unanimously approved by the boards of directors of each of XPDI and Core Scientific.

Pursuant to the Agreement, the Company will acquire Core Scientific through a series of transactions, including (x) the merger of Merger Sub with and into Core Scientific (the "First Merger"), with Core Scientific surviving the First Merger as a wholly owned subsidiary of the Company, and (y) following the closing of the First Merger, the merger of Core Scientific with and into the Company (the "Second Merger" and, together with the First Merger, the "Mergers"), with the Company surviving the Second Merger. As a result of the Mergers, among other things, each outstanding share of common stock of Core Scientific ("Core Scientific Common Stock") will be cancelled in exchange for the right to receive a number of shares of Class A common stock of the Company in an amount that is approximately equal to the quotient obtained by dividing (a) an amount equal to (x) \$4.0 billion, divided by (y) the number of shares of Core Scientific Common Stock on a fully-diluted basis, by (b) \$10.00.

Simultaneous with its entry into the Agreement, the Company also entered into a Sponsor Agreement (the "Sponsor Agreement"), by and among the Company, the Sponsor, the other holders of the Company's Class B common stock, (and, such holders, together with the Sponsor, the "Class B Holders"), and Core Scientific, whereby, among other things, (a) the Class B Holders agreed to vote their shares of Class A Common Stock and Class B Common Stock in favor of approving the Agreement and the Proposed Business Combination, (b) the Class B Holders agreed to waive any adjustment to the conversion ratio set forth in the Company's organizational documents or any other anti-dilution or similar protection with respect to the shares of Class B Common Stock and (c) the Class B Holders agreed to be bound by certain transfer restrictions with respect to their shares of Class A Common Stock and Class B Common Stock prior to the Closing.

Additionally, pursuant to the terms of the Sponsor Agreement, 20% of the shares of Class B Common Stock held by the Class B Holders (the "SPAC Vesting Shares") will be unvested at the Closing and will vest (and shall not be subject to forfeiture) upon the date on which the volume-weighted average price of the Class A Common Stock is greater than \$12.50 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30 consecutive trading day period within five years of the Closing (the "Vesting Period"). Any SPAC Vesting Shares that have not vested by the end of the Vesting Period will be deemed to be transferred by the forfeiting holder to the Company without any consideration and shall be cancelled by the Company and cease to exist.

Simultaneous with its entry into the Agreement, the Company entered into a Company Support Agreement (the "Support Agreement"), by and among the Company, Core Scientific and certain stockholders of Core Scientific (the "Core Scientific Stockholders"). Under the Support Agreement, the Core Scientific Stockholders have agreed to vote or cause to be voted or to execute and deliver a written consent with respect to the Core Scientific equity securities held by the Core Scientific Stockholders adopting the Agreement and approving the Proposed Business Combination. The Core Scientific equity securities that are owned by the Core Scientific Stockholders and subject to the Support Agreement represent more than a majority of the outstanding voting power of Core Scientific shares (on a fully-diluted, as-converted basis).

Going Concern

As of September 30, 2021, the Company had approximately \$1.6 million in its operating bank account and working capital deficit of approximately \$1.9 million.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

The Company's liquidity needs to date have been satisfied through a payment of \$25,000 from the Sponsor to cover for certain offering costs in exchange for issuance of the Founder Shares (as defined in Note 4), the loan under the Note (as defined in Note 4) of approximately \$90,000 and the net proceeds from the consummation of the Private Placement not held in the Trust Account. The Company fully repaid the Note on February 15, 2021. In addition, in order to finance transaction costs in connection with an Initial Business Combination, the Company's officers, directors and initial stockholders may, but are not obligated to, provide the Company Working Capital Loans (see Note 5). As of September 30, 2021, there were no amounts outstanding under any Working Capital Loans.

Until the consummation of a Business Combination, the Company will be using the funds not held in the Trust Account for identifying and evaluating prospective acquisition candidates, performing due diligence on prospective target businesses, paying for travel expenditures, selecting the target business to acquire, and structuring, negotiating and consummating the Business Combination. The Company will need to raise additional capital through loans or additional investments from its Sponsor, shareholders, officers, directors, or third parties. The Company's officers, directors and Sponsor may, but are not obligated to, loan the Company funds from time to time or at any time, in whatever amount they deem reasonable in their sole discretion, to meet the Company's working capital needs. Accordingly, the Company may not be able to obtain additional financing. If the Company is unable to raise additional capital, it may be required to take additional measures to conserve liquidity, which could include, but not necessarily be limited to, curtailing operations, suspending the pursuit of a potential transaction, and reducing overhead expenses.

The Company cannot provide any assurance that new financing will be available to it on commercially acceptable terms, if at all. These conditions raise substantial doubt about the Company's ability to continue as a going concern through one year from the issuance date of these condensed consolidated financial statements.

Risks and Uncertainties

Management is currently evaluating the impact of the COVID-19 pandemic, and the emergence of new variant strains of COVID-19, on the industry and has concluded that while it is reasonably possible that the virus could have a negative effect on the Company's financial position, results of its operations or search for a target company, the specific impact is not readily determinable as of the date of these condensed consolidated financial statements. The condensed consolidated financial statements does not include any adjustments that might result from the outcome of this uncertainty.

Note 2 — Basis of Presentation and Summary of Significant Accounting Policies

Basis of Presentation

The accompanying unaudited condensed consolidated financial statements are presented in U.S. dollars in conformity with accounting principles generally accepted in the United States of America ("GAAP") for financial information and pursuant to the rules and regulations of the SEC. Accordingly, they do not include all of the information and footnotes required by GAAP. In the opinion of management, the unaudited condensed consolidated financial statements reflect all adjustments, which include only normal recurring adjustments necessary for the fair statement of the balances and results for the periods presented. Operating results for the period for the three and nine months ended September 30, 2021 are not necessarily indicative of the results that may be expected through December 31, 2021.

The accompanying unaudited condensed consolidated financial statements should be read in conjunction with the audited financial statements and notes thereto included in the Current Report on Form 8-K and the final prospectus filed by the Company with the SEC on February 19, 2021 and February 11, 2021, respectively.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Principles of Consolidation

The condensed consolidated financial statements of the Company include its wholly-owned subsidiaries in connection with the Proposed Business Combination. All inter-company accounts and transactions are eliminated in consolidation.

Restatement to Previously Reported Financial Statements

In preparation of the Company's unaudited condensed consolidated financial statements as of and for quarterly period ended September 30, 2021, the Company concluded it should restate its previously filed financial statements to classify all Class A common stock subject to possible redemption in temporary equity. In accordance with the SEC and its staff's guidance on redeemable equity instruments in ASC 480, paragraph 10-S99, redemption provisions not solely within the control of the Company require common stock subject to redemption to be classified outside of permanent equity. The Company had previously classified a portion of its Class A common stock in permanent equity, or total stockholders' equity. Although the Company did not specify a maximum redemption threshold, its charter currently provides that, the Company will not redeem its public shares in an amount that would cause its net tangible assets to be less than \$5,000,001. Previously, the Company did not consider redeemable stock classified as temporary equity as part of net tangible assets. Effective with these condensed consolidated financial statements, the Company revised this interpretation to include temporary equity in net tangible assets. As a result, the Company restated its previously filed financial statements to present all redeemable Class A common stock as temporary equity and to recognize accretion from the initial book value to redemption value at the time of its Initial Public Offering. The Company's previously filed financial statements that contained the error were reported in the Company's Form 8-K filed with the SEC on February 19, 2021 (the "Post-IPO Balance Sheet") and the Company's Form 10-Qs for the quarterly periods ended March 31, 2021, and June 30, 2021 (the "Affected Quarterly Periods").

Impact of the Restatement

The change in the carrying value of the redeemable shares of Class A common stock at the Initial Public Offering resulted in a decrease of approximately \$5.8 million in additional paid-in capital and an increase of approximately \$32.0 million to accumulated deficit, as well as a reclassification of 3,780,366 shares of Class A common stock from permanent equity to temporary equity.

As of February 12, 2021	As Reported, As Restated	Adjustment	As Restated
Total assets	\$ 348,644,865		\$ 348,644,865
Total liabilities	\$ 36,448,520		\$ 36,448,520
Class A common stock subject to possible redemption	307,196,340	37,803,660	345,000,000
Preferred stock	_	_	_
Class A common stock	378	(378)	_
Class B common stock	863	_	863
Additional paid-in capital	5,809,812	(5,809,812)	_
Retained earnings (accumulated deficit)	(811,048)	(31,993,470)	(32,804,518)
Total stockholders' equity (deficit)	\$ 5,000,005	\$ (37,803,660)	\$ (32,803,655)
Total Liabilities, Class A Common Stock Subject to Possible Redemption and			
Stockholders' Equity (Deficit)	\$ 348,644,865	s —	\$ 348,644,865

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

The impact of the restatement on the financial statements for the Affected Quarterly Periods is presented below.

The table below presents the effect of the financial statement adjustments related to the restatement discussed above of the Company's previously reported balance sheet as of March 31, 2021:

As of March 31, 2021	As Reported	Adjustment	As Restated
Total assets	\$ 347,742,624	4	\$ 347,742,624
Total liabilities	\$ 29,769,280	6	\$ 29,769,286
Class A common stock subject to possible redemption	312,973,330	32,026,670	345,000,000
Preferred stock	_	_	_
Class A common stock	320	0 (320)	_
Class B common stock	86.	-	863
Additional paid-in capital	46,352	2 (46,352)	_
Retained earnings (accumulated deficit)	4,952,47	3 (31,979,998)	(27,027,525)
Total stockholders' equity (deficit)	\$ 5,000,000	8 \$ (32,026,670)	\$ (27,026,662)
Total Liabilities, Class A Common Stock Subject to Possible Redemption and			
Stockholders' Equity (Deficit)	\$ 347,742,624	4 \$ —	\$ 347,742,624

The table below presents the effect of the financial statement adjustments related to the restatement discussed above of the Company's previously reported statement of cash flows for the three months ended March 31, 2021:

Form 10-Q: Three Months Ended March 31, 2021

As Reported Adjustment (709 200)

	As Reported	Adjustment	As Restated
Cash Flow used in Operating Activities	\$ (798,200)	\$ 	\$ (798,200)
Cash Flows used in Investing Activities	\$ (345,000,000)	\$ _	\$ (345,000,000)
Cash Flows provided by Financing Activities	\$ 347,842,331	\$ _	\$ 347,842,331
Supplemental Disclosure of Noncash Financing Activities:			
Offering costs included in accrued expenses	\$ 506,550	\$ _	\$ 506,550
Offering costs paid by related party under promissory note	\$ 89,891	\$ _	\$ 89,891
Deferred underwriting commissions in connection with the initial public offering	\$ 12,075,000	\$ _	\$ 12,075,000
Initial value of Class A common stock subject to possible redemption	\$ 307,196,340	\$ (307,196,340)	\$ _
Change in value of Class A common stock subject to possible redemption	\$ 499,050	\$ (499,050)	\$ _

The table below presents the effect of the financial statement adjustments related to the restatement discussed above of the Company's previously reported balance sheet as of June 30, 2021:

As of June 30, 2021	As Reported	Adjustment	As Restated
Total assets	\$ 347,606,096		\$ 347,606,096
Total liabilities	\$ 34,910,699		\$ 34,910,699
Class A common stock subject to possible redemption	307,695,390	37,304,610	345,000,000
Preferred stock	_	_	_
Class A common stock	373	(373)	_
Class B common stock	863	_	863
Additional paid-in capital	5,602,376	(5,602,376)	_
Retained earnings (accumulated deficit)	(603,605)	(31,701,861)	(32,305,466)
Total stockholders' equity (deficit)	\$ 5,000,007	\$ (37,304,610)	\$ (32,304,603)
Total Liabilities, Class A Common Stock Subject to Possible Redemption and			
Stockholders' Equity (Deficit)	\$ 347,606,096	\$ —	\$ 347,606,096

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

The table below presents the effect of the financial statement adjustments related to the restatement discussed above of the Company's previously reported statement of cash flows for the six months ended June 30, 2021:

Form 10-Q: Six Months Ended June 30, 2021

	As Reported	Adjustment	As Restated
Cash Flow used in Operating Activities	\$ (730,082)	\$ 	\$ (730,082)
Cash Flows used in Investing Activities	\$ (345,000,000)	\$ _	\$ (345,000,000)
Cash Flows provided by Financing Activities	\$ 347,784,164	\$ _	\$ 347,784,164
Supplemental Disclosure of Noncash Financing Activities:			
Offering costs included in accrued expenses	\$ 448,383	\$ _	\$ 448,383
Offering costs paid by related party under promissory note	\$ 89,891	\$ _	\$ 89,891
Deferred underwriting commissions in connection with the initial public offering	\$ 12,075,000	\$ _	\$ 12,075,000
Initial value of Class A common stock subject to possible redemption	\$ 307,196,340	\$ (307,196,340)	\$ _
Change in value of Class A common stock subject to possible redemption	\$ 5,776,990	\$ (5,776,990)	\$ _

In connection with the change in presentation for the Class A common stock subject to possible redemption, the Company also revised its earnings per share calculation to allocate income and losses shared pro rata between the two classes of shares. This presentation contemplates a Business Combination as the most likely outcome, in which case, both classes of shares share pro rata in the income and losses of the Company. The impact to the reported amounts of weighted average shares outstanding and basic and diluted earnings per common share is presented below for the Affected Quarterly Periods:

	EPS for Cla	EPS for Class A common stock (redeemable)				
	As Reported	Adjustment	As Adjusted			
Form 10-Q (March 31, 2021) — three months ended March 31, 2021						
Net income	\$ 4,962,873	\$ —	\$ 4,962,873			
Weighted average shares outstanding	30,731,669	3,768,331	34,500,000			
Basic and diluted earnings per share	\$ —	\$ 0.12	\$ 0.12			
Form 10-Q (June 30, 2021) — three months ended June 30, 2021						
Net loss	\$ (5,556,078)	\$ —	\$ (5,556,078)			
Weighted average shares outstanding	31,291,533	3,208,467	34,500,000			
Basic and diluted earnings per share	\$ —	\$ (0.13)	\$ (0.13)			
Form 10-Q (June 30, 2021) — six months ended June 30, 2021						
Net loss	\$ (593,205)	\$ —	\$ (593,205)			
Weighted average shares outstanding	31,098,199	3,401,801	34,500,000			
Basic and diluted earnings per share	\$ —	\$ (0.01)	\$ (0.01)			

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

	EPS for Class	EPS for Class B common stock (non-redeemable)				
	As Reported	Adjustment	As Adjusted			
Form 10-Q (March 31, 2021) — three months ended March 31, 2021						
Net income	\$ 4,962,873	\$ —	\$ 4,962,873			
Weighted average shares outstanding	10,109,776	(2,009,776)	8,100,000			
Basic and diluted earnings per share	\$ 0.49	\$ (0.37)	\$ 0.12			
Form 10-Q (June 30, 2021) — three months ended June 30, 2021						
Net loss	\$ (5,556,078)	\$ —	\$(5,556,078)			
Weighted average shares outstanding	11,833,467	(3,208,467)	8,625,000			
Basic and diluted earnings per share	\$ (0.47)	\$ 0.34	\$ (0.13)			
Form 10-Q (June 30, 2021) — six months ended June 30, 2021						
Net loss	\$ (593,205)	\$ —	\$ (593,205)			
Weighted average shares outstanding	10,976,383	(2,612,433)	8,363,950			
Basic and diluted earnings per share	\$ (0.05)	\$ 0.04	\$ (0.01)			

Emerging Growth Company

The Company is an "emerging growth company," as defined in Section 2(a) of the Securities Act, as modified by the Jumpstart Our Business Startups Act of 2012 (the "JOBS Act"), and it may take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies including, but not limited to, not being required to comply with the independent registered public accounting firm attestation requirements of Section 404 of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in its periodic reports and proxy statements, and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved.

Further, Section 102(b)(1) of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies (that is, those that have not had a Securities Act registration statement declared effective or do not have a class of securities registered under the Exchange Act) are required to comply with the new or revised financial accounting standards. The JOBS Act provides that an emerging growth company can elect to opt out of the extended transition period and comply with the requirements that apply to non-emerging growth companies but any such an election to opt out is irrevocable. The Company has elected not to opt out of such extended transition period, which means that when a standard is issued or revised and it has different application dates for public or private companies, the Company, as an emerging growth company, can adopt the new or revised standard at the time private companies adopt the new or revised standard. This may make comparison of the Company's unaudited condensed consolidated financial statements with another public company that is neither an emerging growth company nor an emerging growth company that has opted out of using the extended transition period difficult or impossible because of the potential differences in accounting standards used.

Use of Estimates

The preparation of financial statements in conformity with GAAP requires the Company's management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements. Actual results could differ from those estimates.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Investments Held in Trust Account

The Company's portfolio of investments is comprised of U.S. government securities, within the meaning set forth in Section 2(a)(16) of the Investment Company Act, with a maturity of 185 days or less, or investments in money market funds that invest in U.S. government securities and generally have a readily determinable fair value, or a combination thereof. When the Company's investments held in the Trust Account are comprised of U.S. government securities, the investments are classified as trading securities. When the Company's investments held in the Trust Account are comprised of money market funds, the investments are recognized at fair value. Trading securities and investments in money market funds are presented on the condensed consolidated balance sheets at fair value at the end of each reporting period. Gains and losses resulting from the change in fair value of these securities is included in income from investments held in Trust Account in the accompanying unaudited condensed consolidated statements of operations. The estimated fair values of investments held in the Trust Account are determined using available market information.

Concentration of Credit Risk

Financial instruments that potentially subject the Company to concentrations of credit risk consist of cash accounts in a financial institution, which, at times, may exceed the Federal Depository Insurance Coverage limit of \$250,000 and investments held in Trust Account. As of September 30, 2021, the Company has not experienced losses on these accounts and management believes the Company is not exposed to significant risks on such accounts.

Financial Instruments

The fair value of the Company's assets and liabilities, which qualify as financial instruments under the FASB ASC Topic 825, "Financial Instruments," approximates the carrying amounts represented in the condensed consolidated balance sheets.

Fair Value Measurements

Fair value is defined as the price that would be received for sale of an asset or paid for transfer of a liability, in an orderly transaction between market participants at the measurement date. GAAP establishes a three-tier fair value hierarchy, which prioritizes the inputs used in measuring fair value. The hierarchy gives the highest priority to unadjusted quoted prices in active markets for identical assets or liabilities (Level 1 measurements) and the lowest priority to unobservable inputs (Level 3 measurements). These tiers include:

- · Level 1, defined as observable inputs such as quoted prices (unadjusted) for identical instruments in active markets;
- Level 2, defined as inputs other than quoted prices in active markets that are either directly or indirectly observable such as quoted prices
 for similar instruments in active markets or quoted prices for identical or similar instruments in markets that are not active; and
- Level 3, defined as unobservable inputs in which little or no market data exists, therefore requiring an entity to develop its own
 assumptions, such as valuations derived from valuation techniques in which one or more significant inputs or significant value drivers are
 unobservable.

In some circumstances, the inputs used to measure fair value might be categorized within different levels of the fair value hierarchy. In those instances, the fair value measurement is categorized in its entirety in the fair value hierarchy based on the lowest level input that is significant to the fair value measurement

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Derivative Warrant Liabilities

The Company does not use derivative instruments to hedge exposures to cash flow, market, or foreign currency risks. The Company evaluates all of its financial instruments, including issued stock purchase warrants, to determine if such instruments are derivatives or contain features that qualify as embedded derivatives, pursuant to ASC 480 and ASC 815. The classification of derivative instruments, including whether such instruments should be classified as liabilities or as equity, is re-assessed at the end of each reporting period.

The Company accounts for its warrants issued in connection with its Initial Public Offering (the "Public Warrants") and the Private Placement Warrants are recognized as derivative liabilities in accordance with ASC 815. Accordingly, the Company recognizes the warrant instruments as liabilities at fair value and adjusts the instruments to fair value at each reporting period. The liabilities are subject to re-measurement at each balance sheet date until exercised, and any change in fair value is recognized in the Company's condensed consolidated statements of operations. The initial estimated fair value of the Public Warrants was measured using a Monte Carlo simulation. The initial and subsequent fair value estimates of the Private Placement Warrants is measured using a Black-Scholes option pricing model. Beginning in April 2021, the estimated fair value of the Public Warrants is based on the listed price in an active market for such warrants.

Offering Costs Associated with the Initial Public Offering

Offering costs consisted of legal, accounting, underwriting fees and other costs incurred through the Initial Public Offering that were directly related to the Initial Public Offering. Offering costs were allocated to the separable financial instruments issued in the Initial Public Offering based on a relative fair value basis, compared to total proceeds received. Offering costs associated with derivative warrant liabilities were expensed as incurred, presented as non-operating expenses in the condensed consolidated statements of operations. Offering costs associated with the Class A common stock were charged to the carrying value of Class A common stock subject to possible redemption upon the completion of the Initial Public Offering. Deferred underwriting commissions are classified as non-current liabilities as their liquidation is not reasonably expected to require the use of current assets or require the creation of current liabilities.

Class A Common Stock Subject to Possible Redemption

The Company accounts for its Class A common stock subject to possible redemption in accordance with the guidance in ASC Topic 480 "Distinguishing Liabilities from Equity." Class A common stock subject to mandatory redemption (if any) is classified as liability instruments and are measured at fair value. Conditionally redeemable Class A common stock (including Class A common stock that features redemption rights that are either within the control of the holder or subject to redemption upon the occurrence of uncertain events not solely within the Company's control) are classified as temporary equity. At all other times, Class A common stock is classified as stockholders' equity. The Company's Class A common stock feature certain redemption rights that are considered to be outside of the Company's control and subject to the occurrence of uncertain future events. Accordingly, 34,500,000 shares of Class A common stock subject to possible redemption are presented at redemption value as temporary equity, outside of the stockholders' equity section of the Company's condensed consolidated balance sheets. There was no Class A common stock issued or outstanding as of December 31, 2020.

Effective with the closing of the Initial Public Offering, the Company recognized the accretion from initial book value to redemption amount, which resulted in charges against additional paid-in capital (to the extent available) and accumulated deficit.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Income Taxes

The Company follows the asset and liability method of accounting for income taxes under FASB ASC Topic 740, "Income Taxes" ("ASC 740"). Deferred tax assets and liabilities are recognized for the estimated future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax bases. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred tax assets and liabilities of a change in tax rates is recognized in income in the period that included the enactment date. Valuation allowances are established, when necessary, to reduce deferred tax assets to the amount expected to be realized.

ASC 740 prescribes a recognition threshold and a measurement attribute for the financial statement recognition and measurement of tax positions taken or expected to be taken in a tax return. For those benefits to be recognized, a tax position must be more likely than not to be sustained upon examination by taxing authorities. There were no unrecognized tax benefits as of September 30, 2021. The Company recognizes accrued interest and penalties related to unrecognized tax benefits as income tax expense. No amounts were accrued for the payment of interest and penalties as of September 30, 2021. The Company is currently not aware of any issues under review that could result in significant payments, accruals or material deviation from its position. The Company is subject to income tax examinations by major taxing authorities since inception.

Net income per common shares

The Company complies with accounting and disclosure requirements of FASB ASC Topic 260, "Earnings Per Share." The Company has two classes of shares, which are referred to as Class A common stock and Class B common stock. Income and losses are shared pro rata between the two classes of shares. Net income (loss) per common share is calculated by dividing the net income (loss) by the weighted average shares of common stock outstanding for the respective period.

The calculation of diluted net income (loss) does not consider the effect of the Public Warrants and the Private Placement Warrants to purchase an aggregate of 14,891,667 Class A common stock in the calculation of diluted income (loss) per share, because their exercise is contingent upon future events and their inclusion would be anti-dilutive under the treasury stock method. As a result, diluted net income (loss) per share is the same as basic net income (loss) per share for the three and nine months ended September 30, 2021. Accretion associated with the redeemable Class A common stock is excluded from earnings per share as the redemption value approximates fair value.

The table below presents a reconciliation of the numerator and denominator used to compute basic and diluted net loss per share for each class of common stock:

		Three Months Ended tember 30, 2021		e Months Ended ber 30, 2021
	Class A	Class A Class B		Class B
Basic and diluted net loss per common stock:				
Numerator:				
Allocation of net loss	\$ (14,979,	,628) \$ (3,744,907)	(14,980,500)	0) \$ (4,337,240)
Denominator:				
Basic and diluted weighted average common stock outstanding	34,500,	,000 8,625,000	29,192,30	8,451,923
Basic and diluted net loss per common stock	\$ (0	0.43) \$ (0.43)) \$ (0.5	1) \$ (0.51)

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Recent Accounting Pronouncements

In August 2020, the FASB issued ASU No. 2020-06, *Debt-Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging-Contracts in Entity's Own Equity (Subtopic 815-40): Accounting for Convertible Instruments and Contracts in an Entity's Own Equity ("ASU 2020-06"), which simplifies accounting for convertible instruments by removing major separation models required under current GAAP. The ASU also removes certain settlement conditions that are required for equity-linked contracts to qualify for the derivative scope exception, and it simplifies the diluted earnings per share calculation in certain areas. The Company adopted ASU 2020-06 on January 1, 2021. Adoption of the ASU did not impact the Company's financial position, results of operations or cash flows.*

Management does not believe that any other recently issued, but not yet effective, accounting standards if currently adopted would have a material effect on the accompanying condensed consolidated financial statements.

Note 3 — Initial Public Offering

On February 12, 2021, the Company consummated its Initial Public Offering of 34,500,000 Units, including the exercise of the underwriters' option to purchase 4,500,000 Over-Allotment Units, at \$10.00 per Unit, generating gross proceeds of \$345.0 million, and incurring offering costs of approximately \$19.2 million, of which approximately \$12.1 million in deferred underwriting commissions. Of the 34,500,000 Units sold, an aggregate of 2,405,700 Units were purchased by the Anchor Investors.

Each Unit consists of one share of Class A common stock, and one-fourth of one redeemable warrant (each, a "Public Warrant"). Each Public Warrant entitles the holder to purchase one share of Class A common stock at a price of \$11.50 per share, subject to adjustment (see Note 8).

Note 4 — Related Party Transactions

Founder Shares

On December 31, 2020, the Sponsor paid \$25,000 to cover for certain offering costs on behalf of the Company in exchange for issuance of 7,187,500 shares of the Company's Class B common stock, par value \$0.0001 per share, (the "Founder Shares"). On February 9, 2021, the Company effected a share capitalization of 1,437,500 shares of Class B common stock, resulting in an aggregate of 8,625,000 shares of Class B common stock outstanding. Up to 1,125,000 Founder Shares were subject to forfeiture to the extent that the over-allotment option was not exercised in full by the underwriters, so that the Founder Shares would represent 20.0% of the Company's issued and outstanding shares after the Initial Public Offering. On February 12, 2021, the underwriter fully exercised its option to purchase additional; thus, these 1,125,000 Founder Shares were no longer subject to forfeiture.

In February 2021, the Sponsor agreed to sell to the Anchor Investors 1,552,500 Founder Shares and the Anchor Investors agreed to purchase from the Sponsor on the date of the initial business combination an aggregate of 1,552,500 Founder Shares for an aggregate purchase price of approximately \$4,500, or approximately \$0.003 per share. The Company estimated the aggregate fair value of the Sponsor's agreement to sell Founder Shares to the Anchor Investors to be approximately \$7.0 million using a Monte Carlo simulation. The fair value of the agreement to sell Founder Shares was determined to be an offering cost of the Company in accordance with Staff Accounting Bulletin Topic 5A. Accordingly, the offering cost was allocated to the separable financial instruments issued in the Initial Public Offering based on a relative fair value basis, compared to total proceeds received. Offering costs related to the agreement to sell Founder Shares amounted to approximately \$7.0 million,

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

of which approximately \$6.8 million was charged to stockholder's equity and approximately \$278,000 was expensed to the unaudited condensed consolidated statements of operations as offering costs associated with derivative warrant liabilities.

The initial stockholders and the Anchor Investors agreed, subject to limited exceptions, not to transfer, assign or sell any of the Founder Shares until the earlier to occur of: (A) one year after the completion of the initial Business Combination; and (B) subsequent to the initial Business Combination (x) if the last reported sale price of the Class A common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20-trading days within any 30-trading day period commencing at least 150 days after the initial Business Combination or (y) the date on which the Company completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of the Company's stockholders having the right to exchange their shares of common stock for cash, securities or other property. Any permitted transferees would be subject to the same restrictions and other agreements of the initial stockholders and the Anchor Investors with respect to any Founder Shares.

Private Placement Warrants

Simultaneously with the closing of the Initial Public Offering, the Company consummated the Private Placement of 6,266,667 Private Placement Warrants, at a price of \$1.50 per Private Placement Warrant, to the Sponsor and the Anchor Investors, generating proceeds of \$9.4 million.

Each Private Placement Warrant is exercisable for one whole share of Class A common stock at a price of \$11.50 per share. A portion of the proceeds from the sale of the Private Placement Warrants was added to the proceeds from the Initial Public Offering held in the Trust Account. If the Company does not complete a Business Combination within the Combination Period, the Private Placement Warrants will expire worthless.

The purchasers of the Private Placement Warrants agreed, subject to limited exceptions, not to transfer, assign or sell any of their Private Placement Warrants (except to permitted transferees) until 30 days after the completion of the initial Business Combination.

Related Party Loans

On December 31, 2020, the Sponsor agreed to loan the Company an aggregate of up to \$300,000 to cover expenses related to the Initial Public Offering pursuant to a promissory note (the "Note"). This loan was non-interest bearing and payable upon the completion of the Initial Public Offering. As of February 12, 2021, the Company borrowed a total of approximately \$90,000 under the Note. On February 15, 2021, the Company repaid the Note in full

In addition, in order to finance transaction costs in connection with a Business Combination, the Sponsor or an affiliate of the Sponsor, or certain of the Company's officers and directors may, but are not obligated to, loan the Company funds as may be required ("Working Capital Loans"). If the Company completes a Business Combination, the Company would repay the Working Capital Loans out of the proceeds of the Trust Account released to the Company. Otherwise, the Working Capital Loans would be repaid only out of funds held outside the Trust Account. In the event that a Business Combination does not close, the Company may use a portion of proceeds held outside the Trust Account to repay the Working Capital Loans but no proceeds held in the Trust Account would be used to repay the Working Capital Loans. The Working Capital Loans would either be repaid upon consummation of a Business Combination or, at the lenders' discretion, up to \$1.5 million of such Working Capital Loans may be convertible into warrants of the post Business Combination entity at a price of \$1.50 per warrant. The warrants would be identical to the Private Placement Warrants. Except for the foregoing, the terms

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

of such Working Capital Loans, if any, have not been determined and no written agreements exist with respect to such loans. As of September 30, 2021, the Company had no borrowings under the Working Capital Loans.

Administrative Services Agreement

Commencing on the date that the Company's securities were first listed on the Nasdaq through the earlier of consummation of the initial Business Combination and the Company's liquidation, the Company agreed to pay affiliates of the Sponsor a total of \$20,000 per month for office space, administrative and support services. During the three and nine months ended September 30, 2021 the Company incurred \$60,000 and \$160,000 of such fees, respectively, which are recognized in general and administrative expenses—related party, in the accompanying unaudited condensed consolidated statements of operations. As of September 30, 2021, the Company had \$160,000 payable in connection with such agreement, included as accrued expenses in the accompanying unaudited condensed consolidated balance sheets.

Payments to Insiders

The Sponsor, officers and directors, or any of their respective affiliates, will be reimbursed for any out-of-pocket expenses incurred in connection with activities on the Company's behalf such as identifying potential target businesses and performing due diligence on suitable Business Combinations. The Company's audit committee will review on a quarterly basis all payments that were made by us to the Sponsor, directors, officers or the Company's or any of their affiliates.

Business Combination Payments

The Company may make a cash payment to XMS Capital Partners, LLC ("XMS Capital") or its affiliates for any financial advisory, placement agency or other similar investment banking services that XMS Capital or its affiliates may provide to the Company, in connection with its initial Business Combination, and reimburse XMS Capital or its affiliates for any out-of-pocket expenses incurred by them in connection with the performance of such services.

Advisory Services Agreements

In September and October of 2021, the Company entered into advisory services agreements with four financial advisors, pursuant to which payments of up to \$2.05 million in the aggregate will become due and payable upon the closing of the merger.

Note 5 — Commitments and Contingencies

Advisory Agreement

In September 2021, the Company entered into advisory services agreements with two financial advisors, pursuant to which payments of \$1.15 million in the aggregate will become due and payable upon the closing of the merger.

Registration Rights

The holders of Founder Shares, Private Placement Warrants and warrants that may be issued upon conversion of Working Capital Loans, if any (and any shares of common stock issuable upon the exercise of the Private Placement Warrants or warrants issued upon conversion of the Working Capital Loans and upon conversion of

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

the Founder Shares), were entitled to registration rights pursuant to a registration rights agreement signed upon the consummation of the Initial Public Offering. These holders were entitled to certain demand and "piggyback" registration rights. However, the registration rights agreement provides that the Company will not be required to effect or permit any registration or cause any registration statement to become effective until termination of the applicable lock-up period. The Company will bear the expenses incurred in connection with the filing of any such registration statements.

Underwriting Agreement

The Company granted the underwriters a 45-day option from the date of the final prospectus relating to the Initial Public Offering to purchase up to 4,500,000 additional Units to cover over-allotments, if any, at the Initial Public Offering price, less underwriting discounts and commissions. On February 12, 2021, the underwriter fully exercised its option to purchase additional Units.

The underwriters did not earn any upfront underwriting commission in connection with 2,760,000 Units, including the 2,405,700 Units sold to the Anchor Investors. Except for those Units, the underwriters were entitled to an underwriting discount of \$0.20 per Unit on 31,740,000 Units, or approximately \$6.3 million, paid upon the closing of the Initial Public Offering. An additional fee of \$0.35 per Unit, or approximately \$12.1 million in the aggregate will be payable to the underwriters for deferred underwriting commissions. The deferred fee will become payable to the underwriters from the amounts held in the Trust Account solely in the event that the Company completes a Business Combination, subject to the terms of the underwriting agreement.

Risks and Uncertainties

Management continues to evaluate the impact of the COVID-19 pandemic, and the emergence of new variant strains of COVID-19, on the industry and has concluded that while it is reasonably possible that the virus could have a negative effect on the Company's financial position, results of its operations and/or search for a target company, the specific impact is not readily determinable as of the date of these condensed consolidated financial statements. The condensed consolidated financial statements do not include any adjustments that might result from the outcome of this uncertainty.

Note 6 — Class A Common Stock Subject to Possible Redemption

The Company's Class A common stock feature certain redemption rights that are considered to be outside of the Company's control and subject to the occurrence of future events. The Company is authorized to issue 500,000,000 shares of Class A common stock with a par value of \$0.0001 per share. As of September 30, 2021, there were 34,500,000 shares of Class A common stock outstanding, all of which were subject to possible redemption and are therefore classified outside of permanent equity in the condensed consolidated balance sheets.

The Class A common stock subject to possible redemption reflected on the condensed consolidated balance sheets is reconciled in the following table:

Gross proceeds	\$ 345,000,000
Less:	
Fair value of Public Warrants at issuance	(13,627,500)
Offering costs allocated to Class A common stock subject to possible redemption	(18,098,499)
Plus:	
Accretion on Class A common stock subject to possible redemption amount	31,725,999
Class A common stock subject to possible redemption	\$ 345,000,000

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Note 7 — Stockholders' Equity

Preferred Stock — The Company is authorized to issue 1,000,000 shares of preferred stock, par value \$0.0001 per share, with such designations, voting and other rights and preferences as may be determined from time to time by the Company's board of directors. As of September 30, 2021 and December 31, 2020, there were no shares of preferred stock issued or outstanding.

Class A Common Stock — The Company is authorized to issue 500,000,000 shares of Class A common stock with a par value of \$0.0001 per share. As of September 30, 2021, there were 34,500,000 shares of Class A common stock issued and outstanding, all of which are subject to possible redemption and have been classified as temporary equity (see Note 6). As of December 31, 2020, there were no shares of Class A common stock issued or outstanding.

Class B Common Stock — The Company is authorized to issue 50,000,000 shares of Class B common stock with a par value of \$0.0001 per share. On December 31, 2020, the Company issued 7,187,500 shares of Class B common stock to the Sponsor for an aggregate price of \$25,000. On February 9, 2021, the Company effected a share capitalization of 1,437,500 shares of Class B common stock, resulting in an aggregate of 8,625,000 shares of Class B common stock outstanding. Up to an aggregate of 1,125,000 shares of Class B common stock were subject to forfeiture to the extent that the underwriters' over-allotment option was not exercised in full or in part, so that the initial stockholders would collectively own 20% of the Company's issued and outstanding common stock after the Initial Public Offering. On February 12, 2021, the underwriter fully exercised its option to purchase additional; thus, these 1,125,000 shares of Class B common stock were no longer subject to forfeiture.

In January 2021, our Sponsor transferred 30,000 shares of our Class B common stock to each of our independent directors.

Common stockholders of record are entitled to one vote for each share held on all matters to be voted on by stockholders and vote together as a single class, except as required by law; provided, that, prior to the Company's initial Business Combination, holders of the Class B common stock will have the right to appoint all of the Company's directors and remove members of the board of directors for any reason, and holders of the Class A common stock will not be entitled to vote on the appointment of directors during such time.

The Class B common stock will automatically convert into Class A common stock at the time of the initial Business Combination, or earlier at the option of the holder, on a one-for-one basis, subject to adjustment for stock splits, stock dividends, reorganizations, recapitalizations and the like, and subject to further adjustment as provided herein. In the case that additional shares of Class A common stock, or equity-linked securities, are issued or deemed issued in excess of the amounts issued in the Initial Public Offering and related to the closing of the initial Business Combination, the ratio at which the shares of Class B common stock will convert into shares of Class A common stock will be adjusted (unless the holders of a majority of the issued and outstanding shares of the Class B common stock agree to waive such anti-dilution adjustment with respect to any such issuance or deemed issuance) so that the number of shares of Class A common stock issuable upon conversion of all shares of Class B common stock will equal, in the aggregate, on an as-converted basis, 20% of the sum of all shares of common stock issued and outstanding upon the completion of the Initial Public Offering plus all shares of Class A common stock and equity-linked securities issued or deemed issued in connection with the initial Business Combination, excluding any shares or equity-linked securities issued, or to be issued, to any seller in the initial Business Combination.

Note 8 — Derivative Warrant Liabilities

As of September 30, 2021, the Company had 8,625,000 Public Warrants and the 6,266,667 Private Placement Warrants outstanding.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Public Warrants may only be exercised for a whole number of shares. No fractional Public Warrants will be issued upon separation of the Units and only whole Public Warrants will trade. The Public Warrants will become exercisable on the later of (a) 30 days after the completion of a Business Combination or (b) 12 months from the closing of the Initial Public Offering; provided in each case that the Company has an effective registration statement under the Securities Act covering the shares of Class A common stock issuable upon exercise of the Public Warrants and a current prospectus relating to them is available (or the Company permits holders to exercise their Public Warrants on a cashless basis and such cashless exercise is exempt from registration under the Securities Act). The Company agreed that as soon as practicable, but in no event later than 20 business days after the closing of its initial Business Combination, the Company will use its commercially reasonable efforts to file with the SEC and have an effective registration statement covering the shares of Class A common stock issuable upon exercise of the warrants and will use its commercially reasonable efforts to cause the same to become effective within 60 business days after the closing of the Company's initial Business Combination and to maintain a current prospectus relating to those shares of Class A common stock until the warrants expire or are redeemed. If the shares issuable upon exercise of the warrants are not registered under the Securities Act in accordance with the above requirements, the Company will be required to permit holders to exercise their warrants on a cashless basis. However, no warrant will be exercisable for cash or on a cashless basis, and the Company will not be obligated to issue any shares to holders seeking to exercise their warrants, unless the issuance of the shares upon such exercise is registered or qualified under the securities laws of the state of the exercising holder, or an exemption from registration is available. Notwithstanding the above, if the Company's shares of Class A common stock are at the time of any exercise of a warrant not listed on a national securities exchange such that they satisfy the definition of a "covered security" under Section 18(b)(1) of the Securities Act, the Company may, at its option, require holders of Public Warrants who exercise their warrants to do so on a "cashless basis" in accordance with Section 3(a)(9) of the Securities Act and, in the event the Company so elects, it will not be required to file or maintain in effect a registration statement, and in the event the Company does not so elect, it will use its commercially reasonable efforts to register or qualify the shares under applicable blue sky laws to the extent an exemption is not available.

The warrants have an exercise price of \$11.50 per share, subject to adjustments, and will expire five years after the completion of a Business Combination or earlier upon redemption or liquidation. In addition, if (x) the Company issues additional shares of Class A common stock or equitylinked securities for capital raising purposes in connection with the closing of the initial Business Combination at an issue price or effective issue price of less than \$9.20 per share of Class A common stock (with such issue price or effective issue price to be determined in good faith by the Company's board of directors and, in the case of any such issuance to the Sponsor or its affiliates, without taking into account any Founder Shares held by the Sponsor or such affiliates, as applicable, prior to such issuance) (the "Newly Issued Price"), (y) the aggregate gross proceeds from such issuances represent more than 60% of the total equity proceeds, and interest thereon, available for the funding of the initial Business Combination on the date of the consummation of the initial Business Combination (net of redemptions), and (z) the volume weighted average trading price of the Class A common stock during the 20 trading day period starting on the trading day prior to the day on which the Company consummates the initial Business Combination (such price, the "Market Value") is below \$9.20 per share, then the exercise price of the warrants will be adjusted (to the nearest cent) to be equal to 115% of the higher of the Market Value and the Newly Issued Price, the \$18.00 per share redemption trigger prices described below under "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$18.00" and "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$10.00" will be adjusted (to the nearest cent) to be equal to 180% of the higher of the Market Value and the Newly Issued Price, and the \$10.00 per share redemption trigger price described below under "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$18.00" will be adjusted (to the nearest cent) to be equal to the higher of the Market Value and the Newly Issued Price.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

The Private Placement Warrants are identical to the Public Warrants, except that the Private Placement Warrants and the shares of Class A common stock issuable upon exercise of the Private Placement Warrants will not be transferable, assignable or salable until 30 days after the completion of a Business Combination, subject to certain limited exceptions. Additionally, the Private Placement Warrants will be non-redeemable (except as described below in "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$10.00") so long as they are held by the Sponsor or its permitted transferees. If the Private Placement Warrants are held by someone other than the Sponsor, the Anchor Investors or their respective permitted transferees, the Private Placement Warrants will be redeemable by the Company and exercisable by such holders on the same basis as the Public Warrants.

Redemption of warrants when the price per share of Class A common stock equals or exceeds \$18.00:

Once the warrants become exercisable, the Company may call the outstanding warrants for redemption (except as described herein with respect to the Private Placement Warrants):

- in whole and not in part;
- at a price of \$0.01 per warrant;
- upon a minimum of 30 days' prior written notice of redemption to each warrant holder; and
- if, and only if, the last reported sale price of Class A common stock for any 20 trading days within a 30-trading day period ending on the third trading day prior to the date on which the Company sends the notice of redemption to the warrant holders (the "Reference Value") equals or exceeds \$18.00 per share (as adjusted).

The Company will not redeem the warrants as described above unless a registration statement under the Securities Act covering the issuance of the shares of Class A common stock issuable upon exercise of the warrants is then effective and a current prospectus relating to those shares of Class A common stock is available throughout the 30-day redemption period. If and when the warrants become redeemable by the Company, the Company may exercise its redemption right even if it is unable to register or qualify the underlying securities for sale under all applicable state securities laws.

Redemption of warrants when the price per share of Class A common stock equals or exceeds \$10.00:

Once the warrants become exercisable, the Company may redeem the outstanding warrants:

- in whole and not in part;
- at \$0.10 per warrant upon a minimum of 30 days' prior written notice of redemption provided that holders will be able to exercise their warrants on a cashless basis prior to redemption and receive that number of shares determined by reference to an agreed table based on the redemption date and the "fair market value" (as defined below) of Class A common stock;
- if, and only if, the Reference Value equals or exceeds \$10.00 per share (as adjusted for adjustments to the number of shares issuable upon exercise or the exercise price of a warrant as described herein under the heading "Description of Securities Warrants Public Stockholders' Warrants Anti-dilution Adjustments"); and
- if the Reference Value is less than \$18.00 per share (as adjusted for adjustments to the number of shares issuable upon exercise or the exercise price of a warrant as described herein under the heading "Description of Securities Warrants Public Stockholders' Warrants Anti-dilution Adjustments"), the Private Placement Warrants must also concurrently be called for redemption on the same terms as the outstanding Public Warrants, as described above.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

The "fair market value" of Class A common stock shall mean the volume weighted average price of Class A common stock during the 10 trading days immediately following the date on which the notice of redemption is sent to the holders of warrants. In no event will the warrants be exercisable in connection with this redemption feature for more than 0.361 shares of Class A common stock per warrant (subject to adjustment).

If the Company is unable to complete a Business Combination within the Combination Period and the Company liquidates the funds held in the Trust Account, holders of warrants will not receive any of such funds with respect to their warrants, nor will they receive any distribution from the Company's assets held outside of the Trust Account with the respect to such warrants. Accordingly, the warrants may expire worthless.

Note 9 — Fair Value Measurements

The following table presents information about the Company's assets and liabilities that are measured at fair value on a recurring basis as of September 30, 2021 and indicates the fair value hierarchy of the valuation techniques that the Company utilized to determine such fair value.

<u>Description</u>	Quoted Prices in Active Markets (Level 1)	Significant Other Observable Inputs (Level 2)	Significant Other Unobservable Inputs (Level 3)
Assets:			
Investments held in Trust Account — Money market fund	\$345,027,247	\$ —	\$ —
Liabilities:			
Derivative warrant liabilities — Public warrants	\$ 21,390,000	\$ —	\$ —
Derivative warrant liabilities — Private placement warrants	\$ —	\$ —	\$15,541,330

Transfers to/from Levels 1, 2, and 3 are recognized at the beginning of the reporting period. The estimated fair value of the Public Warrants was transferred from a Level 3 measurement to a Level 1 measurement in April 2021, when the Public Warrants were separately listed and traded in an active market

The initial estimated fair value of the Public Warrants was measured using a Monte Carlo simulation. The initial and subsequent fair value estimates of the Private Placement Warrants is measured using a Black-Scholes option pricing model.

For the three and nine months ended September 30, 2021, the Company recognized a gain resulting from changes in the fair value of derivative warrant liabilities of approximately \$16.2 million and \$13.9 million, respectively, which is presented in the accompanying unaudited condensed consolidated statements of operations.

The initial estimated fair value of the Public Warrants, using a Monte Carlo simulation, and the Private Placement Warrants, Black-Scholes model, was determined using Level 3 inputs. Inherent in a Monte Carlo simulation and a Black-Scholes model are assumptions related to expected stock-price volatility, expected term, risk-free interest rate and dividend yield. The Company estimates the volatility of its warrants based on historical and implied volatility of select peer companies. The risk-free interest rate is based on the U.S. Treasury zero-coupon yield curve on the valuation date for a maturity commensurate with the expected remaining life of the warrants. The expected term of the warrants is assumed to be equivalent to their remaining contractual term. The dividend rate is based on the historical rate, which the Company anticipates remaining at zero.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

The following table provides quantitative information regarding Level 3 fair value measurements inputs at their measurement dates:

	February 12, 2021	September 30, 2021	
Exercise price	\$ 11.50	\$	11.50
Stock price	\$ 10.87	\$	10.11
Volatility	20.0%		31.3%
Term	5.0		5.0
Risk-free rate	0.50%		0.98%

The change in the fair value of derivative liabilities, measured using Level 3 inputs, for the period ended September 30, 2021 is summarized as follows:

Derivative warrant liabilities at December 31, 2020	\$ —
Issuance of Public and Private Warrants	23,027,500
Change in fair value of derivative warrant liabilities	(6,160,830)
Derivative warrant liabilities at March 31, 2021	\$ 16,866,670
Transfer of Public Warrants to Level 1	(9,660,000)
Change in fair value of derivative warrant liabilities	1,504,000
Derivative warrant liabilities at June 30, 2021	\$ 8,710,670
Change in fair value of derivative warrant liabilities Derivative warrant liabilities at September 30, 2021	6,830,660
Derivative warrant liabilities at September 30, 2021	\$ 15,541,330

Note 10 — Subsequent Events

Management has evaluated subsequent events and transactions that occurred after the balance sheet date through the date the condensed consolidated financial statements are available for issuance. Based upon this review, except as noted above, the Company did not identify any subsequent events that would have required adjustment or disclosure in the condensed consolidated financial statements.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP.

Report of Independent Registered Public Accounting Firm	F-47
Balance Sheet as of February 12, 2021 (as Restated)	F-48
Notes to Financial Statement (as Restated)	F-49

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Shareholders and Board of Directors of

Power & Digital Infrastructure Acquisition Corp.

Opinion on the Financial Statement

We have audited the accompanying balance sheet of Power & Digital Infrastructure Acquisition Corp. (the "Company") as of February 12, 2021, and the related notes (collectively referred to as the "financial statement"). In our opinion, the financial statement presents fairly, in all material respects, the financial position of the Company as of February 12, 2021, in conformity with accounting principles generally accepted in the United States of America.

Restatement of Financial Statement

As discussed in Note 2 to the financial statement, the accompanying balance sheet has been restated.

Explanatory Paragraph - Going Concern

The accompanying financial statement has been prepared assuming that the Company will continue as a going concern. As more fully described in Note 1 to the financial statement, the Company's ability to execute its business plan is dependent upon its completion of the proposed initial public offering described in Note 1 to the financial statement. The Company has a working capital deficiency as of February 12, 2021 and lacks the financial resources it needs to sustain operations for a reasonable period of time, which is considered to be one year from the issuance date of the financial statement. These conditions raise substantial doubt about the Company's ability to continue as a going concern. Management's plans in regard to these matters are also described in Note 1. The financial statement does not include any adjustments that might result from the outcome of this uncertainty.

Basis for Opinion

This financial statement is the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statement based on our audit. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statement is free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audit, we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audit included performing procedures to assess the risks of material misstatement of the financial statement, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statement. Our audit also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statement. We believe that our audit provides a reasonable basis for our opinion.

/s/ Marcum LLP

Marcum LLP

We have served as the Company's auditor since 2021.

New York, NY

February 19, 2021 except for Note 1,2,3,7 and 8, which are dated December 10, 2021

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. BALANCE SHEET

February 12, 2021 (as Restated – See Note 2)

Assets:	
Current assets:	
Cash	\$ 2,968,065
Prepaid expenses	676,800
Total current assets	3,644,865
Cash held in Trust Account	345,000,000
Total Assets	\$ 348,644,865
Liabilities and Stockholders' Equity:	
Current liabilities:	
Accounts payable	\$ 687,500
Accrued expenses	545,021
Franchise tax payable	23,464
Note payable - related party	90,035
Total current liabilities	1,346,020
Deferred underwriting commissions	12,075,000
Derivative warrant liabilities	23,027,500
Total liabilities	36,448,520
Commitments and Contingencies	
Class A common stock, \$0.0001 par value; 34,500,000 shares subject to possible redemption at \$10.00 per share	345,000,000
Stockholders' Equity:	
Preferred stock, \$0.0001 par value; 1,000,000 shares authorized; none issued and outstanding	_
Class A common stock, \$0.0001 par value; 500,000,000 shares authorized;	_
Class B common stock, \$0.0001 par value; 50,000,000 shares authorized; 8,625,000 shares issued and outstanding	863
Additional paid-in capital	_
Accumulated deficit	(32,804,518)
Total stockholders' equity	(32,803,655)
Total Liabilities and Stockholders' Equity	\$ 348,644,865

The accompanying notes are an integral part of this financial statement.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

Note 1 — Description of Organization, Business Operations and Going Concern

Power & Digital Infrastructure Acquisition Corp. (the "Company") is a blank check company incorporated in Delaware on December 29, 2020. The Company was formed for the purpose of effecting a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses or entities (the "Business Combination"). The Company is an emerging growth company and, as such, the Company is subject to all of the risks associated with emerging growth companies.

As of February 12, 2021, the Company had not commenced any operations. All activity for the period from December 29, 2020 (inception) through February 12, 2021 relates to the Company's formation and the initial public offering (the "Initial Public Offering") described below. The Company will not generate any operating revenues until after the completion of its initial Business Combination, at the earliest. The Company will generate non-operating income in the form of interest income on cash and cash equivalents from the proceeds derived from the Initial Public Offering. The Company has selected December 31 as its fiscal year end.

The Company's sponsor is XPDI Sponsor LLC, a Delaware limited liability company (the "Sponsor"). The registration statement for the Company's Initial Public Offering was declared effective on February 9, 2021. On February 12, 2021, the Company consummated its Initial Public Offering of 34,500,000 units (the "Units" and, with respect to the Class A common stock included in the Units being offered, the "Public Shares"), including the exercise of the underwriters' option to purchase 4,500,000 additional Units (the "Over-Allotment Units"), at \$10.00 per Unit, generating gross proceeds of \$345.0 million, and incurring offering costs of approximately \$19.2 million, of which approximately \$12.1 million in deferred underwriting commissions (Note 6).

Simultaneously with the closing of the Initial Public Offering, the Company consummated the private placement ("Private Placement") of 6,266,667 warrants (each, a "Private Placement Warrant" and collectively, the "Private Placement Warrants"), at a price of \$1.50 per Private Placement Warrant, to the Sponsor and to certain qualified institutional buyers or institutional accredited investors, including certain funds and accounts managed by subsidiaries of BlackRock, Inc. (the "Anchor Investors"), generating proceeds of \$9.4 million (Note 5).

Upon the closing of the Initial Public Offering and the Private Placement, \$345.0 million (\$10.00 per Unit) of the net proceeds of the Initial Public Offering and certain of the proceeds of the Private Placement was placed in a trust account ("Trust Account") located in the United States with Continental Stock Transfer & Trust Company acting as trustee, and will be invested only in U.S. "government securities," within the meaning of Section 2(a)(16) of the Investment Company Act of 1940, as amended (the "Investment Company Act"), having a maturity of 185 days or less or in money market funds meeting certain conditions under Rule 2a-7 promulgated under the Investment Company Act, which invest only in direct U.S. government treasury obligations, as determined by the Company, until the earlier of: (i) the completion of a Business Combination and (ii) the distribution of the Trust Account as described below.

The Company's management has broad discretion with respect to the specific application of the net proceeds of the Initial Public Offering and the sale of Private Placement Warrants, although substantially all of the net proceeds are intended to be applied generally toward consummating a Business Combination. There is no assurance that the Company will be able to complete a Business Combination successfully. The Company must complete one or more initial Business Combinations having an aggregate fair market value of at least 80% of the net assets held in the Trust Account (as defined below) (excluding the deferred underwriting commissions and taxes payable by us on the income earned on the trust account) at the time of the agreement to enter into the initial Business Combination. However, the Company only intends to complete a Business Combination if the post-transaction company owns or acquires 50% or more of the issued and outstanding voting securities of the target or otherwise acquires a controlling interest in the target sufficient for it not to be required to register as an investment company under the Investment Company Act.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

The Company will provide the holders (the "Public Stockholders") of the Public Shares with the opportunity to redeem all or a portion of their Public Shares upon the completion of a Business Combination either (i) in connection with a stockholder meeting called to approve the Business Combination or (ii) by means of a tender offer. The decision as to whether the Company will seek stockholder approval of a Business Combination or conduct a tender offer will be made by the Company, solely in its discretion. The Public Stockholders will be entitled to redeem their Public Shares for a pro rata portion of the amount then held in the Trust Account (initially anticipated to be \$10.00 per Public Share). The per-share amount to be distributed to Public Stockholders who redeem their Public Shares will not be reduced by the deferred underwriting commissions the Company will pay to the underwriters (as discussed in Note 6). These Public Shares will be recorded at a redemption value and classified as temporary equity upon the completion of the Initial Public Offering in accordance with the Financial Accounting Standards Board's ("FASB") Accounting Standards Codification ("ASC") Topic 480 "Distinguishing Liabilities from Equity," ("ASC 480"). If the Company seeks stockholder approval, the Company will proceed with a Business Combination if a majority of the outstanding shares are voted by the stockholders at a stockholders meeting to approve the Business Combination, unless applicable law, the Company's corporate governing documents or applicable stock exchange rules require a different vote, in which case the Company will complete its Business Combination only if such requisite vote is received. The Company will not redeem the Public Shares in connection with a Business Combination in an amount that would cause its net tangible assets to be less than \$5,000,001. If a stockholder vote is not required by law and the Company does not decide to hold a stockholder vote for business or other legal reasons, the Company will, pursuant to its Amended and Restated Certificate of Incorporation (the "Certificate of Incorporation"), conduct the redemptions pursuant to the tender offer rules of the U.S. Securities and Exchange Commission ("SEC") and file tender offer documents with the SEC prior to completing a Business Combination. If, however, stockholder approval of the transaction is required by law, or the Company decides to obtain stockholder approval for business or legal reasons, the Company will offer to redeem the Public Shares in conjunction with a proxy solicitation pursuant to the proxy rules and not pursuant to the tender offer rules. Additionally, each Public Stockholder may elect to redeem their Public Shares irrespective of whether they vote for or against the proposed transaction. If the Company seeks stockholder approval in connection with a Business Combination, the initial stockholders (as defined below) agreed to vote their Founder Shares (as defined below in Note 5) and any Public Shares purchased during or after the Initial Public Offering in favor of a Business Combination. In addition, the initial stockholders agreed to waive their redemption rights with respect to their Founder Shares and Public Shares in connection with the completion of a Business Combination.

The Certificate of Incorporation provides that a Public Stockholder, together with any affiliate of such stockholder or any other person with whom such stockholder is acting in concert or as a "group" (as defined under Section 13 of the Securities Exchange Act of 1934, as amended (the "Exchange Act")), will be restricted from redeeming its shares with respect to more than an aggregate of 15% of the Public Shares, without the prior consent of the Company.

The holders of the Founder Shares (the "initial stockholders") agreed not to propose an amendment to the Certificate of Incorporation (A) to modify the substance or timing of the Company's obligation to provide holders of shares of Class A common stock the right to have their shares redeemed in connection with a Business Combination or to redeem 100% of the Public Shares if the Company does not complete a Business Combination within the Combination Period (as defined below) or (B) with respect to any other provision relating to the rights of holders of Class A common stock, unless the Company provides the Public Stockholders with the opportunity to redeem their Public Shares in conjunction with any such amendment.

If the Company is unable to complete a Business Combination within 24 months from the closing of the Initial Public Offering, or February 12, 2023, (the "Combination Period") and the Company's stockholders have not amended the Certificate of Incorporation to extend such Combination Period, the Company will (1) cease all

Table of Contents		

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

operations except for the purpose of winding up; (2) as promptly as reasonably possible but not more than 10 business days thereafter, redeem the Public Shares, at a per-share price, payable in cash, equal to the aggregate amount then on deposit in the Trust Account, including interest (less up to \$100,000 of interest to pay dissolution expenses and which interest shall be net of taxes payable by us), divided by the number of then issued and outstanding Public Shares, which redemption will completely extinguish Public Stockholders' rights as stockholders (including the right to receive further liquidating distributions, if any); and (3) as promptly as reasonably possible following such redemption, subject to the approval of the remaining stockholders and the Company's board of directors, liquidate and dissolve, subject in each case to the Company's obligations under Delaware law to provide for claims of creditors and the requirements of other applicable law.

The initial stockholders and Anchor Investors agreed to waive their rights to liquidating distributions from the Trust Account with respect to the Founder Shares if the Company fails to complete a Business Combination within the Combination Period. However, if the initial stockholders and Anchor Investors acquire Public Shares in or after the Initial Public Offering, they will be entitled to liquidating distributions from the Trust Account with respect to such Public Shares if the Company fails to complete a Business Combination within the Combination Period. The underwriters agreed to waive their rights to the deferred underwriting commission (see Note 6) held in the Trust Account in the event the Company does not complete a Business Combination within in the Combination Period and, in such event, such amounts will be included with the other funds held in the Trust Account that will be available to fund the redemption of the Public Shares. In the event of such distribution, it is possible that the per share value of the residual assets remaining available for distribution (including Trust Account assets) will be only \$10.00. In order to protect the amounts held in the Trust Account, the Sponsor agreed to be liable to the Company if and to the extent any claims by a third party (except for the Company's independent registered public accounting firm) for services rendered or products sold to the Company, or a prospective target business with which the Company has discussed entering into a transaction agreement (a "Target"), reduce the amount of funds in the Trust Account to below (i) \$10.00 per Public Share or (ii) the lesser amount per Public Share held in the Trust Account as of the date of the liquidation of the Trust Account due to reductions in the value of the trust assets, in each case net of interest which may be withdrawn to pay taxes, provided that such liability will not apply to any claims by a third party or Target that executed a waiver of any and all rights to seek access to the Trust Account nor will it apply to any claims under the Company's indemnity of the underwriters of the Initial Public Offering against certain liabilities, including liabilities under the Securities Act of 1933, as amended (the "Securities Act"). In the event that an executed waiver is deemed to be unenforceable against a third party, the Sponsor will not be responsible to the extent of any liability for such third-party claims. The Company will seek to reduce the possibility that the Sponsor will have to indemnify the Trust Account due to claims of creditors by endeavoring to have all vendors, service providers (other than the Company's independent registered public accounting firm), prospective target businesses or other entities with which the Company does business, execute agreements with the Company waiving any right, title, interest or claim of any kind in or to monies held in the Trust Account.

Liquidity and Capital Resources

As of February 12, 2021, the Company had approximately \$3.0 million in its operating bank account, and working capital of approximately \$2.3 million.

The Company's liquidity needs to date have been satisfied through a payment of \$25,000 from the Sponsor to cover for certain offering costs in exchange for issuance of the Founder Shares (as defined in Note 5), the loan under the Note of approximately \$90,000 (as defined in Note 5), and the net proceeds from the consummation of the Private Placement not held in the Trust Account. The Company fully repaid the Note on February 15, 2021. In addition, in order to finance transaction costs in connection with an Initial Business Combination, the Company's officers, directors and initial stockholders may, but are not obligated to, provide the Company Working Capital Loans (see Note 6). As of February 12, 2021, there were no amounts outstanding under any Working Capital Loans.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

Until the consummation of a Business Combination, the Company will be using the funds not held in the Trust Account for identifying and evaluating prospective acquisition candidates, performing due diligence on prospective target businesses, paying for travel expenditures, selecting the target business to acquire, and structuring, negotiating and consummating the Business Combination. The Company will need to raise additional capital through loans or additional investments from its Sponsor, shareholders, officers, directors, or third parties. The Company's officers, directors and Sponsor may, but are not obligated to, loan the Company funds from time to time or at any time, in whatever amount they deem reasonable in their sole discretion, to meet the Company's working capital needs. Accordingly, the Company may not be able to obtain additional financing. If the Company is unable to raise additional capital, it may be required to take additional measures to conserve liquidity, which could include, but not necessarily be limited to, curtailing operations, suspending the pursuit of a potential transaction, and reducing overhead expenses. The Company cannot provide any assurance that new financing will be available to it on commercially acceptable terms, if at all. These conditions raise substantial doubt about the Company's ability to continue as a going concern through one year from the issuance date of these condensed consolidated financial statements.

Risks and Uncertainties

Management is currently evaluating the impact of the COVID-19 pandemic on the industry and has concluded that while it is reasonably possible that the virus could have a negative effect on the Company's financial position, results of its operations or search for a target company, the specific impact is not readily determinable as of the date of these financial statement. The financial statement does not include any adjustments that might result from the outcome of this uncertainty.

Note 2 - Restatement of Previously Filed Balance Sheet

The Company concluded it should restate its previously issued financial statements to classify all Class A ordinary shares subject to redemption in temporary equity and to classify its outstanding warrants as liabilities.

In accordance with ASC 480-10-S99, redemption provisions not solely within the control of the Company require shares subject to redemption to be classified outside of permanent equity. The Company had previously classified a portion of its Class A ordinary shares in permanent equity. Although the Company did not specify a maximum redemption threshold, its charter currently provides that the Company will not redeem its Public Shares in an amount that would cause its net tangible assets to be less than \$5,000,001. Previously, the Company did not consider redeemable shares classified as temporary equity as part of net tangible assets. Effective with this balance sheet, the Company revised this interpretation to include temporary equity in net tangible assets.

Additionally, the Company reevaluated the accounting treatment of (i) the 8,625,000 redeemable warrants (the "Public Warrants") that were included in the units issued by the Company in it's the Initial Public Offering and (ii) the 6,266,667 Private Placement Warrants that were issued to the Company's sponsor in a private placement that closed concurrently with the closing of the Initial Public Offering (together with the Public Warrants, the "Warrants"). The Company previously classified the Warrants in stockholders' equity. In further consideration of the guidance in FASB ASC Topic 815, "Derivatives and Hedging" ("ASC 815"), the Company concluded that a provision in the warrant agreement related to certain tender or exchange offers precludes the Warrants from being accounted for as components of equity. As the Warrants meet the definition of a derivative as contemplated in ASC 815, the Warrants should be recorded as derivative liabilities on the balance sheet and measured at fair value at inception (on the date of the Initial Public Offering) and at each subsequent reporting date, with changes in fair value recognized in earnings and losses.

In accordance with FASB ASC Topic 340, "Other Assets and Deferred Costs," as a result of the classification of the Warrants as derivative liabilities, the Company expensed a portion of the offering costs originally recorded as

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

a reduction in equity. The portion of offering costs that was expensed was determined based on the relative fair value of the Public Warrants and shares of Class A common stock included in the Units.

In accordance with SEC Staff Accounting Bulletin No. 99, "Materiality," and SEC Staff Accounting Bulletin No. 108, "Considering the Effects of Prior Year Misstatements when Quantifying Misstatements in Current Year Financial Statements," the Company evaluated the corrections and has determined that the related impact was material to the previously filed balance sheet that contained the error, reported in the Company's Form 8-K filed with the SEC on February 19, 2021 (the "Post-IPO Balance Sheet"). Therefore, the Company, in consultation with its Audit Committee, concluded that the Post-IPO Balance Sheet should be restated to present all outstanding shares of Class A common stock subject to possible redemption as temporary equity, to recognize accretion from the initial book value to redemption value at the time of its Initial Public Offering, and to classify all outstanding Warrants as liabilities. As such, the Company is reporting these restatements to those periods in this Form S-4. The previously presented Post-IPO Balance Sheet and should no longer be relied upon.

The following tables summarize the effect of the revision on each financial statement line item as of the date indicated:

<u>As of February 12, 2021</u>	As Reported	Adjustment	As Restated
Total assets	\$ 348,644,865		\$ 348,644,865
Total current liabilities	\$ 1,346,020	_	\$ 1,346,020
Deferred underwriting commissions	12,075,000	_	12,075,000
Derivative warrant liabilities	_	23,027,500	23,027,500
Total liabilities	\$ 13,421,020		\$ 36,448,520
Class A common stock subject to possible redemption	330,223,840	14,776,160	345,000,000
Preferred stock	_	_	_
Class A common stock	148	(148)	_
Class B common stock	863	_	863
Additional paid-in capital	5,032,602	(5,032,602)	_
Retained earnings (accumulated deficit)	(33,608)	(32,770,910)	(32,804,518)
Total stockholders' equity (deficit)	\$ 5,000,005	\$ (37,803,660)	\$ (32,803,655)
Total Liabilities, Class A Common Stock Subject to Possible Redemption and			
Stockholders' Equity (Deficit)	\$ 348,644,865	\$ _	\$ 348,644,865

Note 3 — Summary of Significant Accounting Policies

Basis of Presentation

The accompanying financial statement is presented in U.S. dollars in conformity with accounting principles generally accepted in the United States of America ("U.S. GAAP") and pursuant to the rules and regulations of the SEC.

Emerging Growth Company

The Company is an "emerging growth company," as defined in Section 2(a) of the Securities Act, as modified by the Jumpstart Our Business Startups Act of 2012 (the "JOBS Act"), and it may take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies including, but not limited to, not being required to comply with the independent registered public accounting firm attestation requirements of Section 404 of the Sarbanes-Oxley Act of 2002, reduced disclosure obligations regarding executive compensation in its periodic reports and proxy statements, and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

Further, Section 102(b)(1) of the JOBS Act exempts emerging growth companies from being required to comply with new or revised financial accounting standards until private companies (that is, those that have not had a Securities Act registration statement declared effective or do not have a class of securities registered under the Exchange Act) are required to comply with the new or revised financial accounting standards. The JOBS Act provides that an emerging growth company can elect to opt out of the extended transition period and comply with the requirements that apply to non-emerging growth companies but any such an election to opt out is irrevocable. The Company has elected not to opt out of such extended transition period, which means that when a standard is issued or revised and it has different application dates for public or private companies, the Company, as an emerging growth company, can adopt the new or revised standard at the time private companies adopt the new or revised standard. This may make comparison of the Company's financial statement with another public company that is neither an emerging growth company nor an emerging growth company that has opted out of using the extended transition period difficult or impossible because of the potential differences in accounting standards used.

Concentration of Credit Risk

Financial instruments that potentially subject the Company to concentrations of credit risk consist of cash accounts in a financial institution, which, at times, may exceed the Federal Depository Insurance Coverage limit of \$250,000. As of February 12, 2021, the Company has not experienced losses on these accounts and management believes the Company is not exposed to significant risks on such account.

Cash and Cash Equivalents

The Company considers all short-term investments with an original maturity of three months or less when purchased to be cash equivalents. The Company had no cash equivalents as of February 12, 2021.

Cash Held in Trust Account

At February 12, 2021, the Company had \$345.0 million in cash held in the Trust Account.

Use of Estimates

The preparation of the financial statement in conformity with GAAP requires the Company's management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statement. Actual results could differ from those estimates.

Fair Value of Financial Instruments

The fair value of the Company's assets and liabilities, which qualify as financial instruments under FASB ASC 820, "Fair Value Measurements and Disclosures," approximates the carrying amounts represented in the balance sheet, primarily due to their short-term nature.

Derivative Warrant Liabilities

The Company does not use derivative instruments to hedge exposures to cash flow, market, or foreign currency risks. The Company evaluates all of its financial instruments, including issued stock purchase warrants, to determine if such instruments are derivatives or contain features that qualify as embedded derivatives, pursuant to ASC 480 and ASC 815. The classification of derivative instruments, including whether such instruments should be classified as liabilities or as equity, is re-assessed at the end of each reporting period.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

The Public Warrants and the Private Placement Warrants are recognized as derivative liabilities in accordance with ASC 815. Accordingly, the Company recognizes the warrant instruments as liabilities at fair value and adjusts the instruments to fair value at each reporting period. The liabilities are subject to re-measurement at each balance sheet date until exercised, with any change in fair value recognized in earnings and losses. The estimated fair value of the Public Warrants is measured at fair value using a Monte Carlo simulation. The estimated fair value of the Private Placement Warrants is measured at fair value using a Black-Scholes option pricing model.

Offering Costs Associated with the Initial Public Offering

Offering costs consisted of legal, accounting, underwriting fees and other costs incurred through the Initial Public Offering that were directly related to the Initial Public Offering. Offering costs were allocated to the separable financial instruments issued in the Initial Public Offering based on a relative fair value basis, compared to total proceeds received. Offering costs associated with derivative warrant liabilities were expensed as incurred. Offering costs associated with the Class A common stock were charged to the carrying value of Class A common stock subject to possible redemption upon the completion of the Initial Public Offering. Deferred underwriting commissions are classified as non-current liabilities as their liquidation is not reasonably expected to require the use of current assets or require the creation of current liabilities.

Class A Common Stock Subject to Possible Redemption

The Company accounts for its Class A common stock subject to possible redemption in accordance with the guidance in ASC Topic 480 "Distinguishing Liabilities from Equity." Class A common stock subject to mandatory redemption (if any) are classified as liability instruments and are measured at fair value. Conditionally redeemable Class A common stock (including shares of Class A common stock that feature redemption rights that are either within the control of the holder or subject to redemption upon the occurrence of uncertain events not solely within the Company's control) are classified as temporary equity. At all other times, Class A common stock are classified as stockholders' equity. The Company's Class A common stock feature certain redemption rights that are considered to be outside of the Company's control and subject to occurrence of uncertain future events. Accordingly, at February 12, 2021, 34,500,000 shares of Class A common stock subject to possible redemption at the redemption amount were presented at redemption value as temporary equity, outside of the stockholders' equity section of the Company's balance sheet.

Effective with the closing of the Initial Public Offering, the Company recognized the accretion from initial book value to redemption amount, which resulted in charges against additional paid-in capital (to the extent available) and accumulated deficit.

Income Taxes

The Company follows the asset and liability method of accounting for income taxes under FASB ASC 740, "Income Taxes." Deferred tax assets and liabilities are recognized for the estimated future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax bases. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred tax assets and liabilities of a change in tax rates is recognized in income in the period that included the enactment date. Valuation allowances are established, when necessary, to reduce deferred tax assets to the amount expected to be realized. Deferred tax assets were deemed immaterial as of February 12, 2021.

FASB ASC 740 prescribes a recognition threshold and a measurement attribute for the financial statement recognition and measurement of tax positions taken or expected to be taken in a tax return. For those benefits to be

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

recognized, a tax position must be more likely than not to be sustained upon examination by taxing authorities. There were no unrecognized tax benefits as of February 12, 2021. The Company recognizes accrued interest and penalties related to unrecognized tax benefits as income tax expense. No amounts were accrued for the payment of interest and penalties as of February 12, 2021. The Company is currently not aware of any issues under review that could result in significant payments, accruals or material deviation from its position. The Company is subject to income tax examinations by major taxing authorities since inception.

Recent Accounting Standards

The Company's management does not believe that any recently issued, but not yet effective, accounting standards, if currently adopted, would have a material effect on the accompanying financial statement.

Note 4 — Initial Public Offering

On February 12, 2021, the Company consummated its Initial Public Offering of 34,500,000 Units, including the exercise of the underwriters' option to purchase 4,500,000 Over-Allotment Units, at \$10.00 per Unit, generating gross proceeds of \$345.0 million, and incurring offering costs of approximately \$19.2 million, of which approximately \$12.1 million in deferred underwriting commissions. Of the 34,500,000 Units sold, an aggregate of 2,405,700 Units were purchased by the Anchor Investors.

Each Unit consists of one share of Class A common stock, and one-fourth of one redeemable warrant (each, a "Public Warrant"). Each Public Warrant entitles the holder to purchase one share of Class A common stock at a price of \$11.50 per share, subject to adjustment (see Note 8).

Note 5 — Related Party Transactions

Founder Shares

On December 31, 2020, the Sponsor paid \$25,000 to cover for certain offering costs on behalf of the Company in exchange for issuance of 7,187,500 shares of the Company's Class B common stock, par value \$0.0001 per share, (the "Founder Shares"). On February 9, 2021, the Company effected a share capitalization of 1,437,500 shares of Class B common stock, resulting in an aggregate of 8,625,000 shares of Class B common stock outstanding. Up to 1,125,000 Founder Shares were subject to forfeiture to the extent that the over-allotment option was not exercised in full by the underwriters, so that the Founder Shares would represent 20.0% of the Company's issued and outstanding shares after the Initial Public Offering. On February 12, 2021, the underwriter fully exercised its option to purchase additional; thus, these 1,125,000 Founder Shares were no longer subject to forfeiture.

In February 2021, the Sponsor agreed to sell to the Anchor Investors 1,552,500 Founder Shares and the Anchor Investors agreed to purchase from the Sponsor on the date of the initial business combination an aggregate of 1,552,500 Founder Shares for an aggregate purchase price of approximately \$4,500, or approximately \$0.003 per share.

The initial stockholders and the Anchor Investors agreed, subject to limited exceptions, not to transfer, assign or sell any of the Founder Shares until the earlier to occur of: (A) one year after the completion of the initial Business Combination; and (B) subsequent to the initial Business Combination (x) if the last reported sale price of the Class A common stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20-trading days within any 30-trading day period commencing at least 150 days after the initial Business Combination or (y) the date on which the Company completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

results in all of the Company's stockholders having the right to exchange their shares of common stock for cash, securities or other property. Any permitted transferees would be subject to the same restrictions and other agreements of the initial stockholders and the Anchor Investors with respect to any Founder Shares.

Private Placement Warrants

Simultaneously with the closing of the Initial Public Offering, the Company consummated the Private Placement of 6,266,667 Private Placement Warrants, at a price of \$1.50 per Private Placement Warrant, to the Sponsor and the Anchor Investors, generating proceeds of \$9.4 million.

Each Private Placement Warrant is exercisable for one whole share of Class A common stock at a price of \$11.50 per share. A portion of the proceeds from the sale of the Private Placement Warrants was added to the proceeds from the Initial Public Offering held in the Trust Account. If the Company does not complete a Business Combination within the Combination Period, the Private Placement Warrants will expire worthless.

The purchasers of the Private Placement Warrants agreed, subject to limited exceptions, not to transfer, assign or sell any of their Private Placement Warrants (except to permitted transferees) until 30 days after the completion of the initial Business Combination.

Related Party Loans

On December 31, 2020, the Sponsor agreed to loan the Company an aggregate of up to \$300,000 to cover expenses related to the Initial Public Offering pursuant to a promissory note (the "Note"). This loan was non-interest bearing and payable upon the completion of the Initial Public Offering. As of February 12, 2021, the Company borrowed approximately \$90,000 under the Note. On February 15, 2021, the Company repaid the Note in full.

In addition, in order to finance transaction costs in connection with a Business Combination, the Sponsor or an affiliate of the Sponsor, or certain of the Company's officers and directors may, but are not obligated to, loan the Company funds as may be required ("Working Capital Loans"). If the Company completes a Business Combination, the Company would repay the Working Capital Loans out of the proceeds of the Trust Account released to the Company. Otherwise, the Working Capital Loans would be repaid only out of funds held outside the Trust Account. In the event that a Business Combination does not close, the Company may use a portion of proceeds held outside the Trust Account to repay the Working Capital Loans but no proceeds held in the Trust Account would be used to repay the Working Capital Loans. The Working Capital Loans would either be repaid upon consummation of a Business Combination or, at the lenders' discretion, up to \$1.5 million of such Working Capital Loans may be convertible into warrants of the post Business Combination entity at a price of \$1.50 per warrant. The warrants would be identical to the Private Placement Warrants. Except for the foregoing, the terms of such Working Capital Loans, if any, have not been determined and no written agreements exist with respect to such loans. As of February 12, 2021, the Company had no borrowings under the Working Capital Loans.

Administrative Services Agreement

Commencing on the date that the Company's securities were first listed on the Nasdaq through the earlier of consummation of the initial Business Combination and the Company's liquidation, the Company agreed to pay affiliates of the Sponsor a total of \$20,000 per month for office space, administrative and support services.

The Sponsor, officers and directors, or any of their respective affiliates, will be reimbursed for any out-of-pocket expenses incurred in connection with activities on the Company's behalf such as identifying potential

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

target businesses and performing due diligence on suitable Business Combinations. The Company's audit committee will review on a quarterly basis all payments that were made by us to the Sponsor, directors, officers or the Company's or any of their affiliates.

Business Combination Payments

The Company may make a cash payment to XMS Capital or its affiliates for any financial advisory, placement agency or other similar investment banking services that XMS Capital or its affiliates may provide to the Company, in connection with its initial Business Combination, and reimburse XMS Capital or its affiliates for any out-of-pocket expenses incurred by them in connection with the performance of such services.

Note 6 — Commitments & Contingencies

Registration Rights

The holders of Founder Shares, Private Placement Warrants and warrants that may be issued upon conversion of Working Capital Loans, if any (and any shares of common stock issuable upon the exercise of the Private Placement Warrants or warrants issued upon conversion of the Working Capital Loans and upon conversion of the Founder Shares), were entitled to registration rights pursuant to a registration rights agreement signed upon the consummation of the Initial Public Offering. These holders were entitled to certain demand and "piggyback" registration rights. However, the registration rights agreement provides that the Company will not be required to effect or permit any registration or cause any registration statement to become effective until termination of the applicable lock-up period. The Company will bear the expenses incurred in connection with the filing of any such registration statements.

Underwriting Agreement

The Company granted the underwriters a 45-day option from the date of the final prospectus relating to the Initial Public Offering to purchase up to 4,500,000 additional Units to cover over-allotments, if any, at the Initial Public Offering price, less underwriting discounts and commissions. On February 12, 2021, the underwriter fully exercised its option to purchase additional Units.

The underwriters did not earn any upfront underwriting commission in connection with 2,760,000 Units, including the 2,405,700 Units sold to the Anchor Investors. Except for those Units, the underwriters were entitled to an underwriting discount of \$0.20 per Unit on 31,740,000 Units, or approximately \$6.3 million, paid upon the closing of the Initial Public Offering. An additional fee of \$0.35 per Unit, or approximately \$12.1 million in the aggregate will be payable to the underwriters for deferred underwriting commissions. The deferred fee will become payable to the underwriters from the amounts held in the Trust Account solely in the event that the Company completes a Business Combination, subject to the terms of the underwriting agreement.

Note 7 — Stockholders' Equity

Preferred Stock — The Company is authorized to issue 1,000,000 shares of preferred stock, par value \$0.0001 per share, with such designations, voting and other rights and preferences as may be determined from time to time by the Company's board of directors. As of February 12, 2021, there were no shares of preferred stock issued or outstanding.

Class A Common Stock — The Company is authorized to issue 500,000,000 shares of Class A common stock with a par value of \$0.0001 per share. As of February 12, 2021, there were 34,500,000 shares of Class A common stock issued and outstanding, all of which are subject to possible redemption and have been classified as temporary equity.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

Class B Common Stock — The Company is authorized to issue 50,000,000 shares of Class B common stock with a par value of \$0.0001 per share. On December 31, 2020, the Company issued 7,187,500 shares of Class B common stock to the Company by the Sponsor for an aggregate price of \$25,000. On February 9, 2021, the Company effected a share capitalization of 1,437,500 shares of Class B common stock, resulting in an aggregate of 8,625,000 shares of Class B common stock outstanding. Up to an aggregate of 1,125,000 shares of Class B common stock were subject to forfeiture to the extent that the underwriters' over-allotment option was not exercised in full or in part, so that the initial stockholders would collectively own 20% of the Company's issued and outstanding common stock after the Initial Public Offering. On February 12, 2021, the underwriter fully exercised its option to purchase additional; thus, these 1,125,000 shares of Class B common stock were no longer subject to forfeiture.

Common stockholders of record are entitled to one vote for each share held on all matters to be voted on by stockholders and vote together as a single class, except as required by law; provided, that, prior to the Company's initial Business Combination, holders of the Class B common stock will have the right to appoint all of the Company's directors and remove members of the board of directors for any reason, and holders of the Class A common stock will not be entitled to vote on the appointment of directors during such time.

The Class B common stock will automatically convert into Class A common stock at the time of the initial Business Combination, or earlier at the option of the holder, on a one-for-one basis, subject to adjustment for stock splits, stock dividends, reorganizations, recapitalizations and the like, and subject to further adjustment as provided herein. In the case that additional shares of Class A common stock, or equity-linked securities, are issued or deemed issued in excess of the amounts issued in the Initial Public Offering and related to the closing of the initial Business Combination, the ratio at which the shares of Class B common stock will convert into shares of Class A common stock will be adjusted (unless the holders of a majority of the issued and outstanding shares of the Class B common stock agree to waive such anti-dilution adjustment with respect to any such issuance or deemed issuance) so that the number of shares of Class A common stock issuable upon conversion of all shares of Class B common stock will equal, in the aggregate, on an as-converted basis, 20% of the sum of all shares of common stock issued and outstanding upon the completion of the Initial Public Offering plus all shares of Class A common stock and equity-linked securities issued or deemed issued in connection with the initial Business Combination, excluding any shares or equity-linked securities issued, or to be issued, to any seller in the initial Business Combination.

Note 8 — Warrant Liabilities

Public Warrants may only be exercised for a whole number of shares. No fractional Public Warrants will be issued upon separation of the Units and only whole Public Warrants will trade. The Public Warrants will become exercisable on the later of (a) 30 days after the completion of a Business Combination or (b) 12 months from the closing of the Initial Public Offering; provided in each case that the Company has an effective registration statement under the Securities Act covering the shares of Class A common stock issuable upon exercise of the Public Warrants and a current prospectus relating to them is available (or the Company permits holders to exercise their Public Warrants on a cashless basis and such cashless exercise is exempt from registration under the Securities Act). The Company agreed that as soon as practicable, but in no event later than 20 business days after the closing of its initial Business Combination, the Company will use its commercially reasonable efforts to file with the SEC and have an effective registration statement covering the shares of Class A common stock issuable upon exercise of the warrants and will use its commercially reasonable efforts to cause the same to become effective within 60 business days after the closing of the Company's initial Business Combination and to maintain a current prospectus relating to those shares of Class A common stock until the warrants expire or are redeemed. If the shares issuable upon exercise of the warrants are not registered under the Securities Act in accordance with the above requirements, the Company will be required to permit holders to exercise their

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

warrants on a cashless basis. However, no warrant will be exercisable for cash or on a cashless basis, and the Company will not be obligated to issue any shares to holders seeking to exercise their warrants, unless the issuance of the shares upon such exercise is registered or qualified under the securities laws of the state of the exercising holder, or an exemption from registration is available. Notwithstanding the above, if the Company's shares of Class A common stock are at the time of any exercise of a warrant not listed on a national securities exchange such that they satisfy the definition of a "covered security" under Section 18(b)(1) of the Securities Act, the Company may, at its option, require holders of Public Warrants who exercise their warrants to do so on a "cashless basis" in accordance with Section 3(a)(9) of the Securities Act and, in the event the Company so elects, it will not be required to file or maintain in effect a registration statement, and in the event the Company does not so elect, it will use its commercially reasonable efforts to register or qualify the shares under applicable blue sky laws to the extent an exemption is not available.

The warrants have an exercise price of \$11.50 per share, subject to adjustments, and will expire five years after the completion of a Business Combination or earlier upon redemption or liquidation. In addition, if (x) the Company issues additional shares of Class A common stock or equitylinked securities for capital raising purposes in connection with the closing of the initial Business Combination at an issue price or effective issue price of less than \$9.20 per share of Class A common stock (with such issue price or effective issue price to be determined in good faith by the Company's board of directors and, in the case of any such issuance to the Sponsor or its affiliates, without taking into account any Founder Shares held by the Sponsor or such affiliates, as applicable, prior to such issuance) (the "Newly Issued Price"), (y) the aggregate gross proceeds from such issuances represent more than 60% of the total equity proceeds, and interest thereon, available for the funding of the initial Business Combination on the date of the consummation of the initial Business Combination (net of redemptions), and (z) the volume weighted average trading price of the Class A common stock during the 20 trading day period starting on the trading day prior to the day on which the Company consummates the initial Business Combination (such price, the "Market Value") is below \$9.20 per share, then the exercise price of the warrants will be adjusted (to the nearest cent) to be equal to 115% of the higher of the Market Value and the Newly Issued Price, the \$18.00 per share redemption trigger prices described below under "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$18.00" and "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$10.00" will be adjusted (to the nearest cent) to be equal to 180% of the higher of the Market Value and the Newly Issued Price, and the \$10.00 per share redemption trigger price described below under "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$18.00" will be adjusted (to the nearest cent) to be equal to the higher of the Market Value and the Newly Issued Price.

The Private Placement Warrants are identical to the Public Warrants, except that the Private Placement Warrants and the shares of Class A common stock issuable upon exercise of the Private Placement Warrants will not be transferable, assignable or salable until 30 days after the completion of a Business Combination, subject to certain limited exceptions. Additionally, the Private Placement Warrants will be non-redeemable (except as described below in "Redemption of warrants when the price per share of Class A common stock equals or exceeds \$10.00") so long as they are held by the Sponsor or its permitted transferees. If the Private Placement Warrants are held by someone other than the Sponsor, the Anchor Investors or their respective permitted transferees, the Private Placement Warrants will be redeemable by the Company and exercisable by such holders on the same basis as the Public Warrants.

Redemption of warrants when the price per share of Class A common stock equals or exceeds \$18.00: Once the warrants become exercisable, the Company may call the outstanding warrants for redemption (except as described herein with respect to the Private Placement Warrants):

• in whole and not in part;

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

- at a price of \$0.01 per warrant;
- upon a minimum of 30 days' prior written notice of redemption to each warrant holder; and
- if, and only if, the last reported sale price of Class A common stock for any 20 trading days within a 30-trading day period ending on the third trading day prior to the date on which the Company sends the notice of redemption to the warrant holders (the "Reference Value") equals or exceeds \$18.00 per share (as adjusted).

The Company will not redeem the warrants as described above unless a registration statement under the Securities Act covering the issuance of the shares of Class A common stock issuable upon exercise of the warrants is then effective and a current prospectus relating to those shares of Class A common stock is available throughout the 30-day redemption period. If and when the warrants become redeemable by the Company, the Company may exercise its redemption right even if it is unable to register or qualify the underlying securities for sale under all applicable state securities laws.

Redemption of warrants when the price per share of Class A common stock equals or exceeds \$10.00: Once the warrants become exercisable, the Company may redeem the outstanding warrants:

- in whole and not in part;
- at \$0.10 per warrant upon a minimum of 30 days' prior written notice of redemption provided that holders will be able to exercise their warrants on a cashless basis prior to redemption and receive that number of shares determined by reference to an agreed table based on the redemption date and the "fair market value" (as defined below) of Class A common stock;
- if, and only if, the Reference Value equals or exceeds \$10.00 per share (as adjusted for adjustments to the number of shares issuable upon exercise or the exercise price of a warrant as described herein under the heading "Description of Securities Warrants Public Stockholders' Warrants Anti-dilution Adjustments"); and
- if the Reference Value is less than \$18.00 per share (as adjusted for adjustments to the number of shares issuable upon exercise or the exercise price of a warrant as described herein under the heading "Description of Securities Warrants Public Stockholders' Warrants Anti-dilution Adjustments"), the Private Placement Warrants must also concurrently be called for redemption on the same terms as the outstanding Public Warrants, as described above.

The "fair market value" of Class A common stock shall mean the volume weighted average price of Class A common stock during the 10 trading days immediately following the date on which the notice of redemption is sent to the holders of warrants. In no event will the warrants be exercisable in connection with this redemption feature for more than 0.361 shares of Class A common stock per warrant (subject to adjustment).

In no event will the Company be required to net cash settle any warrant. If the Company is unable to complete a Business Combination within the Combination Period and the Company liquidates the funds held in the Trust Account, holders of warrants will not receive any of such funds with respect to their warrants, nor will they receive any distribution from the Company's assets held outside of the Trust Account with the respect to such warrants. Accordingly, the warrants may expire worthless.

POWER & DIGITAL INFRASTRUCTURE ACQUISITION CORP. NOTES TO FINANCIAL STATEMENT

Note 9 — Subsequent Events

The Company fully repaid the Note payable - related party of approximately \$90,000 on February 15, 2021.

Management has evaluated subsequent events and transactions that occurred after the balance sheet date up to the date that the financial statement was available to be issued. Based upon this review, the Company did not identify any subsequent events that would have required adjustment or disclosure in the financial statement other than noted above.

Report of Independent Registered Public Accounting Firm

To the Shareholders and the Board of Directors of Core Scientific Holding Co.

Opinion on the Financial Statements

We have audited the accompanying consolidated balance sheets of Core Scientific Holding Co. and subsidiaries (the Company) as of December 31, 2020 and 2019, the related consolidated statements of operations and comprehensive loss, contingently redeemable convertible preferred stock and stockholders' equity and cash flows for each of the two years in the period ended December 31, 2020, and the related notes (collectively referred to as the "consolidated financial statements"). In our opinion, the consolidated financial statements present fairly, in all material respects, the financial position of the Company at December 31, 2020 and 2019, and the results of its operations and its cash flows for each of the two years in the period ended December 31, 2020, in conformity with U.S. generally accepted accounting principles.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audits. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) (PCAOB) and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB and in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audits we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

/s/ Ernst & Young LLP

We have served as the Company's auditor since 2018.

Seattle, Washington August 11, 2021

Core Scientific Holding Co. Consolidated Balance Sheets December 31, 2020 and 2019 (amounts in thousands, except par value)

		ber 31,
Assets	2020	2019
Current Assets:		
Cash and cash equivalents	\$ 8,671	\$ 6,657
Restricted cash	50	250
Accounts receivable, net of allowance of \$620 and \$76, respectively	792	105
Accounts receivable from related parties	315	72
Deposits for equipment	54,818	_
Other current assets	6,273	3,969
Total Current Assets	70,919	11,053
Property, plant and equipment, net	85,244	81,296
Goodwill	58,241	58,241
Intangible assets, net	6,674	3,977
Other noncurrent assets	4,499	7,447
Total Assets	\$225,577	\$162,014
Liabilities, Redeemable Preferred Stock and Stockholders' Equity		
Current Liabilities:		
Accounts payable	\$ 3,057	\$ 7,503
Accrued expenses and other	3,585	1,916
Deferred revenue	44,843	7,834
Capital lease obligations, current portion	2,146	1,571
Notes payable, current portion	16,016	648
Total Current Liabilities	69,647	19,472
Capital lease obligations, net of current portion	2,263	3,183
Notes payable, net of current portion	19,864	2,092
Other noncurrent liabilities	103	1,412
Total Liabilities	91,877	26,159
Contingently redeemable convertible preferred stock; \$0.00001 par value; 50,000 shares authorized; 6,766 and 4,421 shares	<u> </u>	
issued and outstanding at December 31, 2020 and 2019; \$45,164 and \$30,195 total liquidation preference at December 31,		
2020 and 2019	44,476	29,526
Commitments and contingencies		
Stockholders' Equity:		
Common stock; \$0.00001 par value; 200,000 shares authorized; 98,607 and 99,141 shares issued and outstanding at December 31, 2020 and 2019	1	1
Additional paid-in capital	163,967	168,866
Accumulated deficit	(74,744)	(62,538)
Total Stockholders' Equity	89,224	106,329
Total Liabilities, Contingently Redeemable Convertible Preferred Stock and Stockholders' Equity	\$225,577	\$162,014

Core Scientific Holding Co. Consolidated Statements of Operations and Comprehensive Loss Years Ended December 31, 2020 and 2019 (amounts in thousands, except per share amounts)

		Ended December 31,
Hasting rayonya from austamans	2020	2019
Hosting revenue from customers	\$ 34,615	\$ 53,492
Hosting revenue from related parties Equipment sales to customers	6,983 11,193	384
Equipment sales to customers Equipment sales to related parties	1,402	_
Digital asset mining income	6,127	5,647
Total revenue	60,320	59,523
Costs of revenue	50,928	48,996
Gross profit	9,392	10,527
Gain on legal settlements	5,814	— 806
Gain from sales of digital currency assets Operating expense:	65	800
Research and development	5,271	5,480
Sales and marketing	1,771	2,833
General and administrative	14,556	14,707
Total operating expense	21,598	23,020
Operating loss	(6,327)	(11,687)
Non-operating income (expense), net:	(1.222)	
Loss on debt extinguishment	(1,333)	(22.5)
Interest expense, net	(4,436)	(235)
Other non-operating expenses, net	(110)	
Total non-operating income (expense), net	(5,879)	(235)
Loss before income taxes	(12,206)	(11,922)
Income tax expense		
Net loss and comprehensive loss	\$ (12,206)	\$ (11,922)
Deemed dividend from common to preferred exchange	(10,478)	
Net loss attributable to common stockholders	\$ (22,684)	\$ (11,922)
Net loss attributable to common stockholders per share (Note 13):		
Basic and diluted	\$ (0.23)	\$ (0.12)
Weighted average shares outstanding:		
Basic and diluted	98,492	98,684

Core Scientific Holding Co. Consolidated Statements of Changes in Contingently Redeemable Convertible Preferred Stock and Stockholders' Equity Years Ended December 31, 2020 and 2019 (amounts in thousands)

	Contingent Converti	ble Pr Stock		Commo Shares	 ck nount	Pai	Additional d-In Capital Amount	_	cumulated Deficit Amount	 Total ckholders' Equity Amount
Balance at December 31, 2018		\$	_	98,531	\$ 1	\$	161,365	\$	(50,616)	\$ 110,750
Net loss									(11,922)	(11,922)
Stock-based compensation				50	_		3,100			3,100
Issuances of Series A contingently redeemable										
convertible preferred stock	4,421		29,526							
Issuances of common stock				560			4,401			 4,401
Balance at December 31, 2019	4,421	\$	29,526	99,141	\$ 1	\$	168,866	\$	(62,538)	\$ 106,329
Net loss									(12,206)	(12,206)
Stock-based compensation							3,037			3,037
Exchange of common stock for Series A										
contingently redeemable convertible preferred										
stock	1,802		12,308	(1,096)	—		(12,308)			(12,308)
Issuances of common stock- asset acquisition				562	—		1,967			1,967
Issuances of Series A contingently redeemable										
convertible preferred stock	229		1,545							
Issuances of Series B contingently redeemable										
convertible preferred stock	314		1,097							
Issuances of common stock warrants and options							2,405			2,405
Balance at December 31, 2020	6,766	\$	44,476	98,607	\$ 1	\$	163,967	\$	(74,744)	\$ 89,224

Core Scientific Holding Co. Consolidated Statements of Cash Flows Years Ended December 31, 2020 and 2019 (amounts in thousands)

		For the Years Ended December 31,					
Cook flows from Onewating Activities		2020		2019			
Cash flows from Operating Activities: Net loss	\$	(12,206)	\$	(11,922			
Adjustments to reconcile net loss to net cash used in operating activities:	J.	(12,200)	φ	(11,922			
Depreciation and amortization		9.403		6,118			
Stock-based compensation		3,037		2,880			
Digital asset mining income		(6,127)		(5,647			
Amortization of debt discount and debt issuance costs		1,300		(3,047			
Impairments of long-lived assets		1,500		932			
Losses on disposals of property, plant and equipment		2		621			
Loss on debt extinguishment		1,333		- 021			
Provision for doubtful accounts		616		14			
Changes in working capital:		010					
Accounts receivable, net		(1,303)		(11			
Accounts receivable from related parties		(243)		(72			
Digital currency assets		6,094		6,497			
Deposits for equipment		(54,736)		0,477			
Other current assets		(2,353)		(1,603			
Accounts payable		(1,770)		1,398			
Accrued expenses and other		1,625		266			
Deferred revenue		30,009		(911			
Other noncurrent assets and liabilities, net		1,554		608			
Net cash used in operating activities		(23,765)	_	(832			
Cash flows from Investing Activities:		(23,703)	_	(032			
Purchases of property, plant and equipment		(13,668)		(37,412			
Proceeds from sales of property, plant and equipment		92		160			
Acquisition of intangible assets and other		(1,568)		(108			
*		(15,144)					
Net cash used in investing activities		(13,144)		(37,360			
Cash flows from Financing Activities:		2 (12		20.526			
Proceeds from issuances of preferred stock		2,642		29,526			
Issuances of debt		45,178		(022			
Principal payments on debt		(7,097)	_	(832			
Net cash provided by financing activities		40,723		28,694			
Increase (decrease) in cash, cash equivalents, and restricted cash		1,814		(9,498			
Cash, cash equivalents and restricted cash - beginning of year		6,907		16,405			
Cash, cash equivalents and restricted cash - end of year	\$	8,721	\$	6,907			
Supplemental disclosure of other cash flow information:			_				
Cash paid for interest	\$	2,903	\$	336			
Supplemental disclosure of noncash investing and financing activities:	*	_,,	*				
Acquisition:							
Fair value of assets acquired	\$	3,359	\$	_			
Fair value of common stock issued as consideration	Ψ	1,966	*	_			
Accrued capital expenditures	\$	2,544	\$	3,341			
Increase in notes payable for acquisition of property, plant and equipment	\$	19,882	\$				
Decrease in notes payable in exchange for equipment	\$	7,000	\$	_			
Property, plant and equipment acquired under capital leases	\$	1,486	\$	4,957			
Common stock issuances for acquisition of long-lived assets	\$	_	S	3,825			

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

1. ORGANIZATION AND DESCRIPTION OF BUSINESS

MineCo Holdings, Inc. was incorporated on December 13, 2017 in the State of Delaware and changed its name to Core Scientific, Inc. pursuant to an amendment to its Certificate of Incorporation dated June 12, 2018. On August 17, 2020 the Company engaged in a holdco restructuring to facilitate a borrowing arrangement by the Company pursuant to which Core Scientific Inc. was merged with and into a wholly owned subsidiary of the Company and became a wholly owned subsidiary of the Company and the stockholders of Core Scientific Inc. became the shareholders of Core Scientific Holding Co. ("Core Scientific" or the "Company"). The Company currently operates under a holding company structure with Core Scientific Holding Co. as parent and Core Scientific, Inc. as its operating subsidiary. The Company, headquartered in Bellevue, Washington, is an infrastructure, technology and services company that conducts the following business activities:

- Owning and operating datacenter facilities in the U.S. to provide colocation and hosting services for artificial intelligence ("AI") and distributed ledger technology, also commonly known as blockchain;
- Owning and operating computer equipment used to process transactions conducted on one or more blockchain networks in exchange for transaction processing fees rewarded in digital currency assets, commonly referred to as mining;
- Developing AI and blockchain-based platforms and applications, including infrastructure management, security technologies, mining optimization, and record keeping;

On January 31, 2018, as amended on August 20, 2018, Core Scientific entered into an assignment and assumption agreement with BCV 77, LLC ("BCV 77"), pursuant to which the Company issued 41,445 shares of common stock in exchange for all of the assets and liabilities of BCV 77. Concurrently, Core Scientific entered into assignment and assumption agreements with each of BCV 55 ("BCV 55"), and BCV 66 ("BCV 66"), pursuant to which the Company issued 36,197 shares of common stock in exchange for all of the assets and liabilities of BCV 55, and 10,859 shares of common stock in exchange for all of the assets and liabilities of BCV 66. BCV 55, BCV 66, and BCV 77 are collectively referred to as the "BCV Entities" and these transactions are collectively referred to as the "BCV Acquisition." As of the date of the BCV Acquisition, the shares issued to the BCV Entities represented all of the outstanding common stock of Core Scientific. Please see Note 2, under "Basis of Presentation.

Impact of the COVID-19 Pandemic

On January 30, 2020, the World Health Organization ("WHO") announced a global health emergency because of a new strain of coronavirus originating in Wuhan, China (the "COVID-19 outbreak") and the risks to the international community as the virus spreads globally beyond its point of origin. In March 2020, the WHO classified the COVID-19 outbreak as a pandemic, based on the rapid increase in exposure globally.

On March 27, 2020, President Trump signed into law the "Coronavirus Aid, Relief, and Economic Security ("CARES") Act." The CARES Act, among other things, includes provisions relating to refundable payroll tax credits, deferment of employer side social security payments, net operating loss carryback periods, alternative minimum tax credit refunds, modifications to the net interest deduction limitations, increased limitations on qualified charitable contributions, and technical corrections to tax depreciation methods for qualified improvement property.

It also allocated funds for the U.S. Small Business Administration's ("SBA") Paycheck Protection Program ("PPP") loans that are forgivable in certain situations to promote continued employment, as well as Economic Injury Disaster Loans to provide liquidity to small businesses harmed by COVID-19. In April 2020, the

Core Scientific Holding Co.
Notes to Consolidated Financial Statements
(Amounts in thousands, except per share amounts)

Company received a loan of \$2,154 from the PPP through the SBA. The loan was unsecured and bore interest at a rate per annum of 1% and monthly payments of principal were to begin in July 2021. The loan was due in full in April 2022. Under the PPP, all or a portion of the loan may be forgiven if the Company uses the proceeds of the loan for its payroll costs and other expenses in accordance with the requirements of the PPP. On July 13, 2021, the Company repaid the loan in full.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

The accompanying consolidated financial statements reflect the application of certain significant accounting policies as described below and elsewhere in these notes to the consolidated financial statements.

Basis of Presentation

The consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP"). All intercompany balances and transactions have been eliminated in consolidation.

Use of Estimates

The preparation of the Company's consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of income and expenses during the reporting period. Some of the more significant estimates include the valuation of the Company's common shares, the determination of the grant date fair value of stock-based compensation awards, the valuation of goodwill and intangibles, and income taxes. These estimates are based on information available as of the date of the financial statements; therefore, actual results could differ from management's estimates.

Valuation of Common Stock

The Company determined the fair value of its common stock using the most observable inputs available, including the recent sales of its Series A and Series B Contingently Redeemable Convertible Preferred Stock, discussed in Note 9. The Company also uses the market approach, which estimates the value of the Company's business by applying valuation multiples derived from the observed valuation multiples of comparable public companies to the Company's expected financial results.

When observable inputs are not available the Company may use the income approach. This approach typically projects cash flows for the forecast period and uses the perpetuity growth method to calculate terminal values. These cash flows and terminal values are then discounted using an appropriate discount rate. Projections of cash flows are based on management's earnings forecasts.

Applying these valuation and allocation approaches involves the use of estimates, judgments and assumptions that are highly complex and subjective, such as those regarding the Company's expected future revenue, expenses and cash flows, as well as discount rates, valuation multiples, the selection of comparable public companies and the probability of future events. Changes in any or all of these estimates and assumptions, or the relationships between these assumptions, impact the Company's valuation as of each valuation date and may have a material impact on the valuation of the Company's common stock and common stock warrants issued with the Company's debt and equity instruments.

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

Fair Value Measurements

The Company measures certain assets at fair value on a non-recurring basis in certain circumstances. Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. To increase the comparability of fair value measures, the following hierarchy prioritizes the inputs to valuation methodologies used to measure fair value:

- Level 1 Valuations based on quoted prices for identical assets and liabilities in active markets.
- Level 2 Valuations based on observable inputs other than quoted prices included in Level 1, such as quoted prices for similar assets and liabilities in active markets, quoted prices for identical or similar assets and liabilities in markets that are not active, or other inputs that are observable or can be corroborated by observable market data.
- Level 3 Valuations based on unobservable inputs reflecting the Company's own assumptions, consistent with reasonably available assumptions made by other market participants. These valuations require significant judgment.

The Company uses observable market data when determining fair value whenever possible and relies on unobservable inputs only when observable market data is not available. The Company has no material assets that are measured at fair value on a recurring basis.

The Company's non-financial assets, including property, plant and equipment, goodwill, and intangible assets are measured at estimated fair value on a nonrecurring basis. These assets are adjusted to fair value only when an impairment is recognized, or the underlying asset is held for sale. No non-financial assets were classified as Level 2 or Level 3 as of December 31, 2020 and 2019.

The Company's financial instruments include cash and cash equivalents, restricted cash, accounts receivable, net, accounts payable, notes payable and certain accrued expenses and other liabilities. The carrying amount of these financial instruments, other than notes payable discussed below, approximate fair value due to the short-term nature of these instruments.

The fair value of the Company's notes payable was determined based on a discounted cash flow approach using market interest rates of instruments with similar terms and maturities and an estimate for our standalone credit risk. We classified the notes payable as Level 3 financial instruments due to the considerable judgment required to develop assumptions of the Company's standalone credit risk and the significance of those assumptions to the fair value measurement. The estimated fair value of the Company's notes payable, including both the current and noncurrent portion, was \$39,005 and \$2,810 at December 31, 2020 and 2019, respectively. The carrying values of the notes payable, including both the current and noncurrent portion, was \$35,880 and \$2,740 at December 31, 2020 and 2019, respectively.

Cash, Cash Equivalents, and Restricted Cash

Cash and cash equivalents include all cash balances and highly liquid investments with original maturities of three months or less from the date of acquisition. As of December 31, 2020 and 2019, cash and cash equivalents consisted solely of cash in bank deposits. As of December 31, 2020 and 2019, restricted cash of \$50 and \$250, respectively, consisted of cash collateral deposited with a financial institution as credit support for financing provided to the Company. The following table provides a reconciliation of the amount of cash, cash

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

equivalents, and restricted cash reported on the consolidated balance sheets to the total of the same amount shown in the consolidated statements of cash flows:

	Dece	mber 31,
	2020	2019
Cash and cash equivalents	\$8,671	\$6,657
Restricted Cash	50	250
Total cash, cash equivalents and restricted cash	\$8,721	\$6,907

Accounts Receivable and Allowance for Doubtful Accounts

The Company records accounts receivable at the amount billed to the customer based on the contractual terms, net of the allowance for doubtful accounts.

The Company records an allowance for doubtful accounts based on an estimate of amounts that are not collectible. The Company's credit risk is mitigated by certain customer prepayments, and for transactions that are not prepaid, the relatively short collection period. The Company does not require collateral for accounts receivable, however, the Company's hosting customer agreements allow the Company to use customer equipment for processing transactions on digital asset networks until the Company has recovered the past due receivables.

The Company records adjustments to the allowance when new information becomes available that indicates they are required. The Company writes off accounts receivable in the period in which it deems the receivable to be uncollectible. The Company records recoveries of accounts receivable previously written off when it is known that they will be received. The Company's allowance for doubtful accounts was \$620 and \$76 as of December 31, 2020 and December 31, 2019.

Concentrations of Credit Risk

Financial instruments that potentially subject the Company to concentration of credit risk consist primarily of cash and cash equivalents and accounts receivable. The Company places its cash and cash equivalents with major financial institutions, which management assesses to be of high credit quality, in order to limit the exposure to credit risk.

Credit risk with respect to accounts receivable is concentrated with a small number of customers. As of December 31, 2020 and 2019, the concentration of customers comprising 10% or more of accounts receivable, net was as follows:

	For the Years December	
	2020	2019
Customer		
A	26%	N/A
В	N/A	92%

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

For the years ended December 31, 2020 and 2019, the concentration of customers comprising 10% or more of the Company's revenue was as follows:

	For the Years December	
	2020	2019
<u>Customer</u>		
В	24%	74%
C	13%	N/A

The customer with a balance as of December 31, 2019 exercised its right to terminate its hosting contracts with the Company effective April 2020.

Digital Currency Assets

The Company's digital currency assets consist solely of cryptocurrency assets and are accounted for as intangible assets with indefinite useful lives. The Company initially recognizes digital currency assets that are received as transaction processing fee income based on the fair value of the digital currency assets. Digital currency assets that are purchased in an exchange of one digital currency asset for another digital currency asset are recognized at the fair value of the asset received. The Company recognizes realized gains or losses when digital currency assets are sold in an exchange for other digital currency assets or for cash using a first-in first-out method of accounting. For the years ended December 31, 2020 and 2019, the Company recognized net gains of \$65 and \$806, respectively, on sales of digital currency assets. The Company classifies digital currency assets within current assets because it expects to sell the assets shortly after acquisition. The Company classifies cash flows from digital currency assets within cash flows from operating activities.

Property, Plant and Equipment, Net

Property, plant and equipment includes land, buildings and improvements for datacenter facilities and leasehold improvements for the Company's corporate headquarters. Property and equipment consists of computer, mining, network, electrical and other equipment, including property and equipment under capital leases. Property, plant and equipment, net is stated at cost less accumulated depreciation and amortization. Depreciation and amortization is computed using the straight-line method over the estimated useful lives of the assets, which are approximately 1 to 39 years. Leasehold improvements are capitalized at cost and amortized over the shorter of their estimated useful lives or the lease term. Future obligations related to capital leases are presented as Capital lease obligations, current portion and Capital lease obligations, net of current portion in the Company's Consolidated Balance Sheets. Depreciation expense, including amortization of assets held under capital leases, is primarily included in Cost of Revenue in the Company's Consolidated Statements of Operations and Comprehensive Loss.

Business Combinations and Goodwill

The total purchase price of any of the Company's acquisitions is allocated to the tangible and intangible assets acquired and the liabilities assumed based on their estimated fair values as of the acquisition date. The excess of the purchase price over those fair values is recorded as goodwill. The fair value assigned to the tangible and intangible assets acquired and liabilities assumed are based on estimates and assumptions around the valuation of the Company's common stock at the time of the acquisition.

Core Scientific Holding Co.
Notes to Consolidated Financial Statements
(Amounts in thousands, except per share amounts)

The Company does not amortize goodwill, but tests it for impairment annually as of October 31, or more frequently if events or changes in circumstances indicate that the carrying value of goodwill may not be recoverable. The Company has the option to first assess qualitative factors to determine whether it is more likely than not that the fair value of its reporting units are less than their carrying amounts as a basis for determining whether it is necessary to perform the quantitative goodwill impairment test. If the Company determines that it is more likely than not that the fair value of a reporting unit is less than its carrying amount, or chooses not to perform a qualitative assessment, then the quantitative goodwill impairment test will be performed. The quantitative test compares the fair value of the reporting unit with its carrying amount. If the carrying amount exceeds its fair value, the excess of the carrying amount over the fair value is recognized as an impairment loss, and the resulting measurement of goodwill becomes its new cost basis. As of December 31, 2020 and 2019, the carrying amount of Goodwill was \$58,241. There were no accumulated impairment losses as of January 1, 2019, and no impairment losses were recorded for the years ended December 31, 2020 and 2019.

Long-Lived Assets

The Company tests long-lived assets for recoverability whenever events or changes in circumstances have occurred that may affect the recoverability or the estimated useful lives of long-lived assets. Long-lived assets include intangible assets subject to amortization. A long-lived asset may be impaired when the estimated future undiscounted cash flows are less than the carrying amount of the asset. If that comparison indicates that the asset's carrying value may not be recoverable, the impairment is measured based on the difference between the carrying amount and the estimated fair value of the asset. This evaluation is performed at the lowest level for which separately identifiable cash flows exist. Long-lived assets to be disposed of are reported at the lower of the carrying amount or estimated fair value less costs to sell.

Debt Issuance Costs

Debt issuance costs are capitalized and amortized over the term of the associated debt. Debt issuance costs are presented in the balance sheet as a direct deduction from the carrying amount of the debt liability consistent with the debt discount.

Revenue From Contracts With Customers

The Company primarily generates revenue from contracts with customers from hosting services and to a lesser extent, sales of computer equipment. The Company generally recognizes revenue when the promised service is performed, or control of the promised equipment is transferred to customers. Revenue excludes any amounts collected on behalf of third parties, including sales and indirect taxes.

Performance Obligations

The Company's performance obligations primarily relate to hosting services and equipment sales, which are described below. The Company has performance obligations associated with commitments in customer hosting contracts for future services and commitments to acquire and deploy customer equipment that have not yet been recognized in the financial statements. For contracts with original terms that exceed one year (typically ranging from 18 to 36 months), those commitments not yet recognized as of December 31, 2020 were \$303,693. As of December 31, 2019, substantially all of the Company's remaining performance obligations had an original expected duration of one year or less.

Core Scientific Holding Co.
Notes to Consolidated Financial Statements
(Amounts in thousands, except per share amounts)

Hosting Services

The Company performs hosting services that enable customers to run AI, blockchain and other high-performance computing operations. The Company's performance obligation related to these services is satisfied over time. The Company recognizes revenue for services that are performed on a consumption basis, such as the amount of electricity used in a period, based on the customer's use of such resources. The Company recognizes variable usage at its estimate of the stand-alone selling price of its hosting services. Terms included in certain of the Company's contracts for hosting services cause the prices for hosting services to increase or decrease as the market prices of digital currency assets increase or decrease. As a result, the Company's hosting revenue is sensitive to changes in the market prices of digital currency assets. As the market prices of digital currency assets are volatile, the Company cannot reasonably estimate the future price of its hosting services for such contracts over the contract term such that a significant revenue reversal is not probable. Therefore, the Company's best estimate of the transaction price in these contracts is based on the current market price of the digital currency asset. The Company constrains recognition of revenue from contracts with hosting customers with variable pricing to the contractual price determined by the market price of the digital currency asset in the current period.

The Company generally bills its customers in advance based on estimated consumption or the fixed quantity for the period under the contract. The Company recognizes revenue based on actual consumption in the period and invoices adjustments in subsequent periods or retains credits toward future consumption. The term between invoicing and when payment is due typically does not exceed 30 days.

Equipment Sales

The Company recognizes revenue from sales of computer equipment to customers at the point in time when control of the equipment is transferred to the customer, which generally occurs upon deployment of the equipment.

Customer Contract Assets

The Company records customer contract assets in Other Current Assets and Other Noncurrent Assets on the consolidated balance sheets when revenue is recognized for performance obligations satisfied in advance of cash payments. The Company had no customer contract assets as of December 31, 2020. The Company's total balance of customer contract assets as of December 31, 2019 was \$428.

Deferred Revenue

The Company records contract liabilities in Deferred Revenue on the consolidated balance sheets when cash payments are received in advance of performance and recognizes them as revenue when the performance obligations are satisfied. The Company's deferred revenue balance as of December 31, 2020 and 2019 was \$44,843 and \$7,834, respectively, all from advance payments received during the years then ended.

In the year ended December 31, 2020 and 2019, the Company recognized \$7,834 and \$8,745, respectively, of revenue that was included in the deferred revenue balance as of the beginning of the year. The change in deferred revenue for the year ended December 31, 2020 and 2019 was primarily due to the performance of hosting services during the year for which advanced payments had been received from customers in the prior year. Advanced payments for hosting services are typically recognized in the following month and advanced payments for equipment sales are generally recognized within one year.

Core Scientific Holding Co.
Notes to Consolidated Financial Statements
(Amounts in thousands, except per share amounts)

Digital Asset Mining Income

The Company derives its digital asset mining income from operating its owned computer equipment as part of a pool of users, facilitated by a pool operator, that processes transactions conducted on one or more blockchain networks. The contracts with pool operators are terminable at any time by either party. In exchange for providing computing power to the pool, the Company is entitled to a fractional share of the cryptocurrency award the mining pool operator receives from the blockchain network upon successfully adding a block to the blockchain. The Company's fractional share is based on the proportion of computing power the Company contributed to the mining pool operator to the total computing power contributed by all mining pool participants in processing the block.

Providing computing power in digital asset transaction verification services is an output of the Company's ordinary activities. The provision of providing such computing power is the only performance obligation in the Company's contracts with mining pool operators. The transaction consideration the Company receives, if any, is variable noncash consideration, which the Company measures at fair value on the date received, which is not materially different than the fair value at contract inception or the time the Company has earned the award from the pools. Because the Company does not have a reliable means to estimate its relative share of the rewards until they are paid to it, the variable consideration is constrained until the Company receives the consideration, at which time revenue is recognized. The Company's digital asset mining income is sensitive to changes in the market prices of digital currency assets that may be significant.

Deposits for Equipment

The Company has entered into agreements with vendors to supply equipment for its customers and for the Company's own digital asset mining operations. These agreements generally require significant refundable deposits payable months in advance of delivery and additional advance payments in monthly installments thereafter.

Costs of Revenue

The Company's costs of revenue primarily consist of electricity costs, salaries, depreciation of property, plant and equipment used to perform hosting services and mining operations and other related costs, and costs of computer equipment sold to customers.

Research and Development

The Company's research and development expenses primarily include personnel costs associated with technology and product development and data science research. Research and development costs are expensed as incurred.

Stock-Based Compensation

The Company recognizes the cost of services received in exchange for awards of equity instruments based upon the fair value of those awards on the grant date. For the years ended December 31, 2020 and 2019, the Company's consolidated operating results included \$387 and \$295 of stock-based compensation expense related to restricted stock units issued to former consultants, respectively, and \$2,572 and \$2,585 of stock-based compensation expense related to stock options issued to employees and consultants, respectively. In addition, for the year ended December 31, 2020 the Company's consolidated operating results included \$78 of stock-based compensation expense related to stock issued to vendors. Stock-based compensation expense for the years ended December 31, 2020 and 2019 relates to consulting services and general corporate operations and is included within general and administrative expenses in the Company's Consolidated Statements of Operations and Comprehensive Loss.

Core Scientific Holding Co.
Notes to Consolidated Financial Statements
(Amounts in thousands, except per share amounts)

Stock-based compensation expense is measured at the grant date based on the value of the equity award. The fair value of stock option awards is estimated on the date of grant using the Black-Scholes option-pricing model. The fair value of restricted stock unit awards is estimated on the date of grant using the estimated fair value of the Company's common stock on the date of grant.

For awards with only service conditions, primarily stock options and certain restricted stock units, the estimated fair value of the equity awards are recognized as expense on a straight-line basis, less actual forfeitures as they occur, over the requisite service period for the entire award, which is generally the vesting period.

For awards with service and performance conditions, primarily restricted stock unit awards, the compensation expense is recognized separately for each tranche of each award as if it were a separate award with its own vesting date (i.e., on an accelerated attribution basis) and the estimated fair value of the equity awards are recognized as expense when it is probable that the performance conditions will be achieved. If the performance conditions become probable of being achieved before the end of the requisite service period, the unrecognized compensation costs for which the requisite service have been provided is recognized in the period in which achievement becomes probable and the remaining unrecognized compensation costs for which requisite service has not been provided is recognized as expense prospectively on an accelerated attribution basis over the remaining requisite service period for the entire award, less actual forfeitures as they occur. See Note 11 for more information about the service and performance conditions associated with the Company's equity awards.

Employee Benefit Plan

The Company currently maintains a defined contribution retirement and savings plan for the benefit of our employees who satisfy certain eligibility requirements (the "401(k) Plan"). The U.S. Internal Revenue Code allows eligible employees to defer a portion of their compensation, within prescribed limits, on a pre-tax basis through contributions to the 401(k) Plan. Currently, the Company does not match contributions made by participants in the 401(k) Plan.

Earnings Per Share

The Company computes earnings per share ("EPS") following Financial Accounting Standards Board ("FASB") Accounting Standards Codification ("ASC") Topic 260, *Earnings per share*. Basic EPS is measured as the income or loss available to common stockholders divided by the weighted average common shares outstanding for the period. Diluted EPS presents the dilutive effect on a per-share basis from the potential conversion of convertible securities or the exercise of options and or warrants; the dilutive impacts of potentially convertible securities are calculated using the as-if method; the potentially dilutive effect of options or warranties are computed using the treasury stock method. Securities that are potentially an anti-dilutive effect (i.e., those that increase income per share or decrease loss per share) are excluded from diluted EPS calculation.

Income Taxes

The Company is subject to income taxes mainly in the jurisdictions in which it provides various infrastructure, technology and colocation and hosting services. The Company's tax position requires significant judgment in order to properly evaluate and quantify tax positions and to determine the provision for income taxes.

The Company uses the assets and liabilities method to account for income taxes, which requires that deferred tax assets and deferred tax liabilities be determined based on the differences between the financial

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

statement and tax basis of assets and liabilities, using enacted tax rates in effect for the years in which the differences are expected to be reversed. The Company estimates its actual current tax expense, including permanent charges and benefits, and the temporary differences resulting from differing treatment of items, for tax and financial accounting purposes.

The Company assesses whether it is more likely than not that its deferred tax assets will be realized by considering both positive and negative evidence. If the Company believes that recovery of these deferred tax assets is not more likely than not, the Company establishes a valuation allowance. Significant judgment is required in determining any valuation allowance recorded against deferred tax assets. In assessing the need for a valuation allowance, the Company considered all available evidence, including recent operating results, projections of future taxable income, the reversal of taxable temporary differences, and the feasibility of tax planning strategies.

GAAP sets forth a two-step approach to recognizing and measuring uncertain tax positions. The first step is to evaluate the tax position for recognition by determining if the weight of available evidence indicates that it is more likely than not that the position will be sustained upon examination, including resolution of related appeals or litigation processes, if any. The second step is to measure the tax benefit as the largest amount that is more than 50% likely of being realized upon settlement. Interest and penalties related to unrecognized tax benefits are included within Income Tax Expense. Accrued interest and penalties are included in the related tax liability line in the consolidated balance sheets. No penalties or interest have been recognized or accrued for unrecognized tax benefits for the years ended December 31, 2020 and 2019.

The Company adjusts its reserves for tax positions in light of changing facts and circumstances, such as the closing of a tax audit, the refinement of an estimate or changes in tax laws. To the extent that the final tax outcome of these matters is different than the amounts recorded, the differences are recorded as adjustments to the provision for income taxes in the period in which such determination is made. The provision for income taxes includes the impact of reserve provisions and changes to reserves that are considered appropriate.

The Company's future effective tax rates could be adversely affected by changes in the valuation of the Company's deferred tax assets or liabilities, or changes in tax laws, regulations, accounting principles or interpretations thereof. In addition, the Company is subject to examination of income tax returns by various tax authorities. The Company regularly assesses the likelihood of adverse outcomes resulting from these examinations to determine the adequacy of its provisions for income taxes.

Recently Adopted Accounting Standards

In June 2018, the FASB issued Accounting Standards Update ("ASU") No. 2018-07, Stock Compensation (Topic 718): Improvements to Nonemployee Share-based Payment Accounting ("ASU 2018-07"). ASU 2018-07 expands the guidance in Topic 718 to include share-based payments for goods and services to non-employees and generally aligns it with the guidance for share-based payments to employees. This ASU is effective for the Company in fiscal years beginning after December 15, 2019, and interim periods within fiscal years beginning after December 15, 2020. The Company elected to adopt ASU 2018-07 on January 1, 2019. The adoption of this ASU did not have a material impact on the Company's consolidated financial statements or related footnote disclosures.

Accounting Standards not yet adopted

In June 2016, the FASB issued ASU 2016-13, Financial Instruments—Measurement of Credit Losses on Financial Instruments, which will require an entity to measure credit losses for certain financial instruments and

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

financial assets, including trade receivables. Under this update, on initial recognition and at each reporting period, an entity will be required to recognize an allowance that reflects the entity's current estimate of credit losses expected to be incurred over the life of the financial instrument. This update will be effective for the Company with the annual reporting period beginning *January 1, 2023,* including interim periods within that reporting period. Early adoption is permitted. The Company is currently evaluating the impacts the adoption of this standard will have on the consolidated financial statements.

In December 2019, the FASB issued ASU 2019-12, Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes, which simplifies the accounting for income taxes by removing the exceptions to the incremental approach for intra-period tax allocation in certain situations, the requirement to recognize a deferred tax liability for a change in the status of a foreign investment, and the general methodology for computing income taxes in an interim period when year-to date loss exceeds the anticipated loss for the year. The amendments also simplify the accounting for income taxes with regard to franchise tax, the evaluation of step up in the tax basis goodwill in certain business combinations, allocating current and deferred tax expense to legal entities that are not subject to tax and enacted change in tax laws or rates. The standard is effective for the Company beginning January 1, 2022. The Company is currently evaluating the impacts the adoption of this standard will have on the consolidated financial statements.

In February 2016, the FASB issued ASU No. 2016-02, *Leases-(Topic 842)*. Under this new guidance, lessees will be required to recognize for all leases (with the exception of short-term leases): 1) a lease liability equal to the lessee's obligation to make lease payments arising from a lease, measured on a discounted basis and 2) a right-of-use asset which will represent the lessee's right to use, or control the use of, a specified asset for the lease term. Topic 842 is effective for the Company's annual and interim reporting periods beginning January 1, 2022. The adoption of Topic 842 will require the Company to recognize non-current assets and liabilities for right-of-use assets and operating lease liabilities on its consolidated balance sheet, but it is not expected to have a material effect on the Company's results of operations or cash flows. Topic 842 will also require additional footnote disclosures to the Company's consolidated financial statements.

In August 2020, the FASB issued ASU No. 2020-06 (ASU 2020-06) Accounting for Convertible Instruments and Contracts in an Entity's Own Equity, which simplifies the accounting for certain financial instruments with characteristics of liabilities and equity, including convertible instruments and contracts on an entity's own equity. The standard is effective for the Company beginning January 1, 2024, with early adoption permitted as of the beginning of an annual fiscal year but no earlier than January 1, 2021. The Company is currently evaluating the impacts the adoption of this standard will have on the consolidated financial statements.

There are no other new accounting pronouncements that are expected to have a significant impact on the Company's consolidated financial statements.

3. ASSET ACQUISITIONS

In June 2020, Core Scientific entered into an Asset Purchase Agreement with Atrio Inc. ("Atrio") to acquire certain assets of Atrio in exchange for \$1,235 cash consideration and 293 shares of the Company's common stock. The Company and Atrio contemplated a valuation for the transaction of approximately \$2,419 based on an estimate of the fair value of the Company's common stock of \$3.50 per share which was allocated to cost of the acquired software intangible assets.

In a separate transaction in June 2020, the Company entered into an agreement with RStor, Inc. ("RStor") to obtain a non-exclusive license to three specific patents held by RStor for consideration of 269 shares of the

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

Company's common stock. The Company and RStor contemplated a valuation for the transaction of approximately \$940 based on an estimate of the fair value of the Company's common stock of \$3.50 per share which was allocated to the cost of the acquired patent intangible assets.

The software acquired from Atrio and the acquired patents from RStor are not businesses under ASC 805, *Business Combinations*, because substantially all the fair value of the acquired assets is concentrated in a single intellectual property asset. Accordingly, the asset purchases are accounted for as asset acquisitions where the cost of the acquisition, measured as the fair value of the cash consideration transferred and the common stock issued by the Company, is allocated to the assets acquired.

In September 2019, the Company acquired certain assets and liabilities of Stax Digital LLC in exchange for 560 shares of the Company's common stock valued at \$3,825. The set of assets acquired by the Company did not meet the definition of a business and the transaction was accounted for as an asset acquisition. The Company allocated \$3,959 of the cost of the asset acquisition to acquired software and recognized assumed liabilities comprising accounts payable of \$134.

4. OTHER ASSETS

Other current assets as of December 31, 2020 and 2019 consist of the following:

	Decembe	er 31,
	2020	2019
Prepaid expenses \$	1,212	\$2,702
Security deposits	2,230	972
Customer contract asset	_	110
Other	2,831	185
Total other current assets \$	6,273	\$3,969

Prepaid expenses include prepayments of insurance premiums, rent expense, licenses, and subscriptions. Security deposits represent payments by the Company primarily associated with utilities and leases.

Other noncurrent assets as of December 31, 2020 and 2019 consist of the following:

Decembe	er 31,
2020	2019
81,150	\$1,246
3,000	5,348
_	535
349	318
64,499	\$7,447
	2020 61,150 3,000

Security deposits represents payments by the Company primarily associated with utilities and leases. Utility construction contributions include amounts contributed to utilities for the construction of assets that allow the Company to obtain utility services, primarily electricity.

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

5. PROPERTY, PLANT AND EQUIPMENT, NET

Property, plant and equipment, net as of December 31, 2020 and 2019 consist of the following:

	Decemb		
	2020	2019	Estimated Useful Lives
Land and improvements	\$ 5,458	\$ 5,258	20 years(1)
Building and improvements	46,811	43,407	12 to 39 years
Computer, mining and network equipment ⁽²⁾	20,270	19,247	1 to 5 years
Electrical equipment ⁽³⁾	24,681	24,259	10 years
Other property, plant and equipment ⁽⁴⁾	1,243	1,499	5 to 7 years
Total	98,463	93,670	
Less accumulated depreciation and amortization ⁽⁵⁾	(13,219)	(12,374)	
Property, plant and equipment, net	\$ 85,244	\$ 81,296	

- (1) Estimated useful life of improvements. Land is not depreciated.
- (2) Includes capital lease assets of \$3,277 and \$2,951 at December 31, 2020 and 2019, respectively.
- (3) Includes capital lease assets of \$2,588 and \$2,588 at December 31, 2020 and 2019, respectively.
- (4) Includes capital lease assets of \$432 and \$0 at December 31, 2020 and 2019, respectively.
- (5) Includes accumulated amortization for assets under capital leases of \$1,762 and \$290 at December 31, 2020 and 2019, respectively.

Depreciation expense, including amortization of capital lease assets, for the years ended December 31, 2020 and 2019 was \$8,566 and \$5,990, respectively. Depreciation for the year ended December 31, 2020 and 2019 allocated to costs of revenue was \$8,509 and was \$5,923, respectively.

In April 2019, the Company's management approved a plan to actively market and sell certain digital currency mining equipment. In April 2019, the Company recognized an impairment loss of \$932 representing the excess of the equipment's carrying value over its estimated fair value less costs to sell, based on a quotation for the salvage value of the equipment. The Company completed the sale in October 2019. For the years ended December 31, 2020 and 2019, the Company recognized losses of \$2 and \$621, respectively, on sales of property, plant and equipment, primarily digital currency mining equipment. The Company has included these amounts for impairments of long-lived assets and losses on disposals of property, plant and equipment within general and administrative expenses in the Company's Consolidated Statements of Operations and Comprehensive Loss.

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

6. INTANGIBLE ASSETS, NET

Intangible assets, net as of December 31, 2020 and 2019 consist of the following:

		December 31, 2020				
	Gross	Accumulated Amortization	Net Carrying Amount	Estimated Useful Lives		
Acquired software	\$7,318	\$ (954)	\$ 6,364	5-8 years		
Patents	260	(4)	256	20 years		
Trademarks	59	(5)	54	8 years		
Total intangible assets, net	\$7,637	\$ (963)	\$ 6,674			

		December 31, 2019			
	Gross		ımulated rtization	Net Carrying Amount	Estimated Useful Lives
Acquired software	\$3,959	\$	(124)	\$ 3,835	8 years
Patents	111		(2)	109	20 years
Trademarks	34		(1)	33	8 years
Total intangible assets, net	\$4,104	\$	(127)	\$ 3,977	

The Company amortizes intangible assets subject to amortization over their estimated useful lives. Amortization of intangibles assets is included within general and administrative expenses in the Company's Consolidated Statements of Operations and Comprehensive Loss. Amortization expense for intangible assets was \$837 and \$127 for the years ended December 31, 2020 and 2019, respectively. The Company estimates the amortization expense for the next five years to be \$772 per year.

7. ACCRUED EXPENSES AND OTHER

Accrued expenses and other as of December 31, 2020 and 2019 consist of the following:

	Decei	mber 31,
	2020	2019
Accrued expenses and other	\$ 936	\$1,058
Accrued taxes	1,645	847
Other current liabilities	1,004	11
Total accrued expenses and other	\$3,585	\$1,916

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

8. NOTES PAYABLE

Notes payable as of December 31, 2020 and 2019 consist of the following:

	Decemb	
	2020	2019
Georgia note	\$ 581	\$ 773
Kentucky note	1,511	1,967
PPP note	2,154	_
Silverpeak note	22,260	_
Genesis note	4,648	_
NYDIG note	718	_
Celsius note	6,842	_
Total	38,714	2,740
Unamortized discount and debt issuance costs	(2,834)	_
Total notes payable, net	\$35,880	\$2,740

Georgia Note - In September 2018, the Company entered into a five-year secured purchase note payable agreement for \$1,000 in connection with the acquisition of property in Georgia for datacenter development ("Georgia note"). The note bears interest at a rate per annum of 5% and the Company is required to make monthly payments of principal and interest. Interest expense on the notes has been recognized based on an effective interest rate of 5%

Kentucky Note - In December 2018, the Company entered into a five-year secured promissory note agreement for \$2,400 in connection with the acquisition of property in Kentucky for datacenter development ("Kentucky note"). The note bears interest at a rate per annum of 5% and the Company is required to make monthly payments of principal and interest. Interest expense on the notes has been recognized based on an effective interest rate of 5%

Executive Notes – In March and April 2020, the Company issued an aggregate of \$4,000 of notes to the Company's president and chief executive officer and a member of the Board of Directors. The notes are unsecured and bear interest at 7% per annum, with all principal and interest accrued due at the earlier of 12 months from issuance or the occurrence of a financing of at least \$20,000 of gross proceeds. The Company granted common stock warrants to the noteholders for the purchase of 4,000 shares of common stock at an exercise price of \$1.00 per share. The exercise price was subsequently amended to \$1.34 per share. The Company allocated proceeds of \$2,151 to the notes and \$1,636 to the warrants on a relative fair value basis and recognized a derivative liability of \$212 equal to the fair value of the contingent repayment feature. Interest expense on the notes was recognized based on an effective interest rate of 98%. The notes were repaid in full in May 2020 from the proceeds of the Silverpeak Loan described below. This resulted in an extinguishment loss of \$1,333 based on the carrying amount of the related party notes and the contingent repayment derivative.

PPP Loan - In April 2020, the Company received a loan of \$2,154 from the PPP through the SBA ("PPP note"). The loan is unsecured and bears interest at a rate per annum of 1% and is paid monthly. Payments of principal are scheduled to begin in July 2021 and to be paid monthly until April 2022 when it would be fully paid off. Interest expense on the note has been recognized based on an effective interest rate of 1%. On July 13, 2021, the Company repaid the PPP note in full.

Core Scientific Holding Co.
Notes to Consolidated Financial Statements
(Amounts in thousands, except per share amounts)

Silverpeak Loan - In May 2020, the Company issued a \$21,000 senior secured loan to Silverpeak Credit Partners LP. The loan is secured by substantially all of the assets of the Company. The loan bears interest at a rate per annum of 15%, interest payments due monthly. Principal payments are due on a quarterly basis from June 2021 through June 2022 in the amount equal to the lesser of \$1,000 or 30% of a measure of excess cash flow defined in the loan agreement. From September 2022 through March 2023, principal payments are due on a quarterly basis in the amount equal to the lesser of \$1,000 or 40% of the measure of excess cash flow. All remaining outstanding principal is due in May 2023. The loan requires the Company to maintain unrestricted cash on hand of at least \$3,750 throughout the term of the loan. The Company was in compliance with this requirement as of December 31, 2020. Beginning in September 2021, the Company may not exceed specified limits of the Company's ratio of indebtedness to a measure of the Company's earnings before interest, taxes, and depreciation and amortization ("EBITDA") and the Company's ratio of EBITDA to fixed charges for debt service, maintenance of the Company's facilities, and income taxes. The Company received proceeds from the loan issuance of \$19,837, which are net of an original issuance discount and issuance costs of \$1,163. In addition, the issuance costs also included a mon-refundable exit fee of \$1,260 due upon maturity or early payment which was included as part of the debt discount. Interest expense on the loan has been recognized based on an effective interest rate of 21%, which includes the amortization of the discount and debt issuance costs. In May 2020, the Company used \$4,043 of the proceeds from the loan issuance to repay the outstanding principal and interest on the promissory notes to the Company's president and chief executive officer and a member of the Board of Directors, as discussed above.

Genesis Loan - In July 2020, the Company entered into a credit facility with Genesis Global Capital, LLC that provides capacity of up to \$13,028 to finance the Company's acquisition of blockchain computing equipment ("Genesis Loan"). The Company borrowed \$5,266 in three installments and the borrowing capacity of the facility was reduced via an amendment in September 2020 to equal the actual amounts borrowed. The loans under the credit facility are secured by the blockchain computing equipment and the Company is required to comply with an approved mining strategy and other restrictions on use of the collateral. Loans under the credit facility have terms of 20 months, bear interest at a rate per annum of 16% plus a fixed risk premium, and require monthly payments. Interest expense on the loans have been recognized based on an effective interest rate of 28%, which includes the amortization of a debt discount.

NYDIG Loan - In October 2020, the Company entered into a master equipment finance agreement with NYDIG and received a loan of \$774 to finance the Company's acquisition of blockchain computing equipment. The loan bears an interest rate of 15% and has a term of 24 months from issuance. Interest expense on the loan has been recognized based on an effective interest rate of 15%.

Celsius Loan – In November 2020 the Company entered into an equipment lending agreement with Celsius Networks Lending, LLC to borrow up to \$27,684 in five tranches through May 20, 2021 for the purchase of blockchain mining equipment. The Company borrowed the first two tranches totaling \$13,842 in November and December 2020. The loans bear interest at 13.5% annually and are due at the earlier of the date of sale of the underlying mining equipment or 24 months from issuance. Interest expense on the loans has been recognized based on an effective interest rate of 13.5%. In December 2020, the Company repaid \$7,000 of the loan and in January 2021, the Company repaid the remaining balance of \$6,842. No gain or loss was recognized upon loan extinguishment.

As of December 31, 2020, the Company is not in violation of any of its covenants.

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

Maturities of notes payable as of December 31, 2020 are as follows:

Year ending December 31,	
2021	\$ 16,016
2022	6,743
2023	15,955
Total notes payable	\$ 38,714

9. CONTINGENTLY REDEEMABLE CONVERTIBLE PREFERRED STOCK

The Company is authorized to issue 50,000 shares of preferred stock, \$0.00001 par value. As of December 31, 2020 and 2019, 6,766 and 4,421 shares of preferred stock were issued and outstanding.

]	December 31, 202	20	
	Shares Authorized	Shares Issued and Outstanding	Issuance Price per Share	Net Proceeds	Liquidation preference
Contingently Redeemable Convertible Preferred Stock:			<u> </u>		
Series A	14,641	6,452	\$ 6.83	\$ 31,070	\$ 44,064
Series B	14,327	314	3.50	1,097	1,100
Undesignated	21,032			·	
Total contingently redeemable convertible preferred stock	50,000	6,766		\$ 32,167	\$ 45,164

			December 31, 2019		
	Shares Authorized	Shares Issued and Outstanding	Issuance Price per Share	Net Proceeds	Liquidation preference
Contingently Redeemable Convertible Preferred Stock:					
Series A	14,641	4,421	\$ 6.83	\$ 29,526	\$ 30,195
Undesignated	35,359				
Total contingently redeemable convertible preferred stock	50,000	4,421		\$ 29,526	\$ 30,195

At various dates during the year ended December 31, 2019, the Company issued a total of 4,421 shares of \$0.00001 par value Series A Contingently Redeemable Convertible Preferred Stock ("Series A Preferred Stock") at \$6.83 per share in private placement offerings. The aggregate proceeds from the offerings were \$29,526, which were net of offering costs of \$670.

In February and March 2020, the Company issued a total of 229 shares of Series A Preferred Stock at \$6.83 per share in private placement offerings. The aggregate proceeds from the offerings were \$1,545 which are net of offering costs of \$16.

In February 2020, the Company completed an exchange of 1,096 shares of common stock that were originally issued in the private placement offering from October 2018 to December 2019 for 1,802 newly issued

Core Scientific Holding Co.
Notes to Consolidated Financial Statements
(Amounts in thousands, except per share amounts)

shares of Series A Preferred Stock. The shares of common stock were retired upon reacquisition by the Company. The Company received no net proceeds from the exchange and recognized a deemed dividend of \$10,478 based on the incremental fair value of the preferred stock received by the stockholders compared to the fair value of the common stock exchanged. The difference between the fair value of the Preferred Stock issued and common stock retired is accounted for as a deemed dividend and is reflected as an increase to the amount of the net loss attributable to common shareholders.

In June 2020, the Company issued 314 shares of \$0.00001 par value Series B Contingently Redeemable Convertible Preferred Stock ("Series B Preferred Stock") at \$3.50 per share in a private placement offering. The aggregate proceeds from the offering was \$1,097, which was net of offering costs of \$3.

Each share of Series A and Series B Preferred Stock has a liquidation value of \$6.83 and \$3.50, respectively. Holders may elect at any time to convert each share of preferred stock into one share of the Company's common stock. Unless earlier converted, each share of preferred stock will automatically convert to shares of the Company's common stock upon the earlier of an initial public offering of the Company's common stock for aggregate proceeds of not less than \$100 million, or an agreement to convert by the holders of a majority of the outstanding shares of preferred stock. In the event of an initial public offering, the number of shares of the Company's common stock issuable on conversion of each share of preferred stock will be either (i) the amount determined by dividing the liquidation value by the public offering price if such offering price is less than the liquidation value per share, or (ii) one share. In the event of an agreement to convert by the holders of a majority of the outstanding shares of preferred stock, one share of the Company's common stock is issuable on conversion of each share of preferred stock.

In addition to the conversion rights, the Series A and B Preferred Stock is redeemable upon the occurrence of a deemed liquidation event, the definition of which includes a transaction that results in a change in control of the Company or a sale or transfer by the Company of substantially all its assets. In the event of a deemed liquidation, the holders of shares of Preferred Stock are entitled to a cash distribution equal to the liquidation value per share. As of December 31, 2020 and 2019 the Series A and B preferred stock was not redeemable because no deemed liquidation event had occurred. Furthermore, the Company has not made any adjustments to the carrying value of the Series A and B Preferred Stock to reflect the liquidation value of the shares because the Company has determined that a deemed liquidation event is not probable of occurring.

All outstanding shares of Series A and B Preferred Stock have one vote per share. Dividends on Series A and Series B Preferred Stock are payable when, as, and if declared by the Company's Board of Directors. Series A and B Preferred Stock rank in parity to each other and rank senior to the Company's common stock, including capital stock established in the future, unless the terms of such capital stock expressly provide otherwise, and junior to existing and future indebtedness and other liabilities with respect to dividend rights and distribution rights upon liquidation, winding-up, and dissolution.

The Series A and B Preferred Stock is classified as contingently redeemable because it is redeemable upon the occurrence of a deemed liquidation event, and such a redemption triggering event is not solely within the control of the Company. The Series A and B Preferred Stock is therefore presented separate from and outside of stockholders' equity in a manner consistent with temporary equity.

10. COMMITMENTS AND CONTINGENCIES

Legal Proceedings—The Company is subject to legal proceedings arising in the ordinary course of business. The Company accrues losses for a legal proceeding when it is probable that a loss has been incurred

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

and the amount of the loss can be reasonably estimated. However, the uncertainties inherent in legal proceedings make it difficult to reasonably estimate the costs and effects of resolving these matters. Accordingly, actual costs incurred may differ materially from amounts accrued and could materially adversely affect the Company's business, cash flows, results of operations, financial condition and prospects. Unless otherwise indicated, the Company is unable to estimate reasonably possible losses in excess of any amounts accrued. As of December 31, 2020 and 2019, there were no loss contingency accruals for legal matters.

Gain on legal settlements—The Company recognized a gain of \$5,814 during the year ended December 31, 2020 with respect to the resolution of legal actions for damages resulting from the early termination of agreements by former customers.

Operating Leases—The Company leases its corporate headquarters under an operating lease agreement that expires in 2022. The Company leases other facilities under operating leases that have terms of less than one year. Certain lease agreements provide for rental payments on a graduated basis. The Company recognizes rent expense on a straight-line basis over the lease period. Rent expense was \$325 and \$584 for the years ended December 31, 2020 and 2019, respectively.

The Company's minimum payments under noncancelable operating leases having initial terms in excess of one year are as follows at December 31, 2020:

Year ending December 31,	
2021	\$111
2022 - 2025	<u>—</u>
Total minimum lease payments	\$111

The lease agreements for certain of the Company's facilities require the Company to maintain cash deposits with the landlords of each facility as security for its payment obligations under the leases. As of December 31, 2020 and 2019, the Company had deposits totaling \$2 and \$24, respectively, related to such leases.

Capital Leases—The Company has entered into arrangements with various parties to finance the acquisition of computer and networking equipment, electrical infrastructure, and office equipment. These arrangements include options exercisable by the Company at the end of the initial terms to renew, purchase the equipment, or to terminate. These arrangements are classified as capital leases and as of December 31, 2020 and 2019, the related capital lease obligations were \$4,409 and \$4,754, respectively.

As of December 31, 2020, the future minimum lease payments and present value of the net minimum lease payments under these capital leases are as follows:

Year ending December 31,	
2021	\$2,467
2022	1,990
2023	271
2024	102
2025	46
Total minimum lease payments	4,876
Less: interest	(467)
Present value of net minimum lease payments	\$4,409

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

The accumulated amortization of property, plant and equipment under capital leases for the years ended December 31, 2020 and 2019, was \$1,762 and \$290, respectively.

Purchase obligations— The Company has entered into agreements to purchase blockchain mining equipment that are enforceable and legally binding and that specifies all significant terms, including fixed or minimum quantities to be purchased, fixed, minimum or variable price provisions and the approximate timing of the transaction. As of December 31, 2020, the aggregate amount of the Company's purchase obligations totaled approximately \$88 million, substantially all of which are expected to be settled within one year of December 31, 2020.

11. STOCKHOLDERS' EQUITY

Authorized Capital—As of December 31, 2020, the Company was authorized to issue 200,000 shares of common stock, \$0.00001 par value. The holders of the Company's common stock are entitled to one vote per share.

As discussed in Note 3, in September 2019, the Company acquired certain assets and liabilities of Stax Digital LLC in exchange for 560 shares of the Company's common stock valued at \$3,825.

In September 2019, the Company issued 50 shares of common stock to a consulting firm to settle a payment obligation for services rendered during the year ended December 31, 2018.

As discussed in Note 9, In February 2020, the Company completed an exchange of 1,096 shares of common stock that were originally issued in the private placement offering from October 2018 to December 2019 for 1,802 newly issued shares of Preferred Stock.

As discussed in Note 3, in June 2020, the Company issued 562 shares of the Company's common stock valued at \$3.50 per share, or \$1,967 as part of the consideration for certain assets acquired from Atrio and RStor.

In March 2020, in connection with the Executive Notes described in Note 8, the Company issued warrants to the Company's president and chief executive officer and a member of the Board of Directors to purchase up to 4,000 shares of the Company's common stock at an exercise price of \$1.00 per share. The exercise price was subsequently amended to \$1.34 per share. The warrants expire in March 2022 and are unexercised as of December 31, 2020.

In March 2020, the Company issued warrants to service providers in exchange for services provided related to the issuance of Series A Convertible Preferred Stock. The warrants are for an aggregate of 135,000 shares at an exercise price of \$6.83. The warrants expire in March 2022 and are unexercised as of December 31, 2020.

Equity Incentive Plans

The Company has outstanding awards under the 2018 Omnibus Incentive Plan (the "2018 Plan"), which has a ten-year life for granting up to 40,500 shares of common stock for awards. In June 2020, the Company's Board of Directors approved an increase in the number of common shares that may be issued for awards under the 2018 Plan to 52,500 shares. Awards granted under the 2018 Plan may be incentive stock options (must meet all statutory requirements), non-qualified stock options, stock appreciation rights, restricted stock and stock units, performance awards and other cash-based or stock-based awards. Awards granted under the 2018 Plan are subject to a minimum vesting period of at least one year commencing from the date of grant. Additionally,

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

options granted under the plan must expire within ten years of the grant date and must be granted with exercise prices of no less than the fair value of the common stock on the grant date, as determined by the Company's Board of Directors.

As of December 31, 2020, the Company had reserved shares of common stock for future issuances under the 2018 Plan as follows:

Options outstanding	2,530
Restricted stock units and awards outstanding	37,946
Available for future stock option and restricted stock units and grants	12,024
Total outstanding and reserved for future issuance	52,500

Stock Options

Stock options granted under the 2018 Plan are granted at a price per share not less than the fair value at date of grant. Options granted to date generally vest over 4 years and are exercisable for up to 10 years. Determining the fair value of stock options at the grant date requires judgment, including estimating the expected term, expected volatility, risk-free interest rate, and expected dividends.

Expected Term—The Company's expected term is determined using the simplified method and represents the midpoint between the vesting period and the contractual term of the awards.

Expected Volatility—The Company's volatility factor is estimated using comparable public company volatility for similar terms.

Risk-Free Interest Rate—The Company bases the risk-free interest rate used in the Black-Scholes option-pricing model on the implied yield currently available on US Treasury zero coupon issues with an equivalent remaining term. Where the expected term of the Company's stock-based awards does not correspond with the term for which an interest rate is quoted, the Company performs a straight-line interpolation to determine the rate from the available term maturities.

Expected Dividends—The Company has no history of paying cash dividends and has no present intention to pay common stock cash dividends in the future; as a result, the expected dividend yield is 0% as of December 31, 2020 and 2019.

The weighted-average assumptions for options granted for the years ended December 31, 2020 and 2019, are as follows:

	For the Years Ended	December 31,
	2020	2019
Dividend yield	0.00%	0.00%
Expected volatility	36.26%	30.59%
Risk-free interest rate	0.70%	1.65%
Expected life (years)	10.00	5.51

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

A summary of stock option activity for the years ended December 31, 2020 and 2019 is as follows:

	Number of Shares	Weighted- Average Exercise Price	Weighted-Average Remaining Contractual Term (in years)	Aggregate Intrinsic Value
Options outstanding - December 31, 2018	1,880	\$ 11.23		
Granted	350	6.83		
Exercised	_	_		
Forfeited	_	_		
Options outstanding - December 31, 2019	2,230	10.54		
Granted	300	1		
Exercised	_	_		
Forfeited	_	_		
Options outstanding - December 31, 2020	2,530	\$ 9.41	8.08	\$ 3,959
Options vested and expected to vest as of December 31, 2020	2,530	\$ 9.41	8.08	\$ 3,959
Options vested and exercisable as of December 31, 2020	1,159	\$ 7.54	8.43	\$ 3,704

The weighted-average grant date fair value of options granted for the year ended December 31, 2020 and 2019 was \$2.56 and \$2.14, respectively. The total fair value of stock options vested during the year ended December 31, 2020 and 2019 was \$2,241 and \$1,572, respectively.

As of December 31, 2020, total unrecognized stock-based compensation expense related to unvested stock options was approximately \$2,988, which is expected to be recognized over a weighted-average time period of 1.47 years.

Restricted Stock Units—Restricted stock units ("RSUs") granted in 2018 required that the holder elect before the date of grant whether the RSUs will vest either

- Over a 4-year service period, or
- Over a 4-year service period and upon either i) completion of an initial public offering of the Company's common stock, or ii) upon consummation of a transaction resulting in a change in control of the Company.

RSUs granted in 2020 and 2019 vest over a 4-year service period and upon either i) completion of an initial public offering of the Company's common stock, or ii) upon consummation of a transaction resulting in a change in control of the Company.

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

A summary of RSU activity for the years ended December 31, 2020 and 2019 is as follows:

	Number of Shares	Gran	ted-Average t Date Fair Value
Unvested - December 31, 2018	28,412	\$	11.23
Granted	13,065		6.83
Vested	_		_
Forfeited	(6,430)		10.42
Unvested - December 31, 2019	35,047	\$	9.39
Granted	8,056		4.46
Vested	(802)		11.23
Forfeited	(5,157)		6.83
Unvested - December 31, 2020	37,144	\$	8.55

As of December 31, 2020, the Company had approximately \$320,969 of unrecognized stock-based compensation expense related to RSUs that were fully vested under the service condition but had performance conditions that had not yet been achieved. For RSUs subject to both the service and performance conditions, the unrecognized compensation expense will be recognized as expense when it is probable that the performance conditions will be achieved. If the performance conditions become probable of being achieved before the end of the requisite service period, the unrecognized compensation expense for which requisite service has not been provided will be recognized as expense prospectively on an accelerated attribution basis over the remaining requisite service period.

In September and October 2019, the Company modified the RSU grants to two executives that separated from the Company. As a result of the modifications, 1,500 RSUs that otherwise would have been forfeited were amended to satisfy the service condition at the end of a 3-year service period ending in October 2022, and 183 RSUs that otherwise would have been forfeited received accelerated satisfaction of the service condition in October 2019. Vesting of the total of 1,683 modified RSUs are subject to the performance condition. As the performance condition was not probable of being achieved as of the modification dates, the Company recognized no compensation expense associated with the modifications.

12. INCOME TAXES

Current income tax expense represents the amount expected to be reported on the Company's income tax returns, and deferred tax expense or benefit represents the change in net deferred tax assets and liabilities. Deferred tax assets and liabilities are determined based on the difference between the financial statement and tax basis of assets and liabilities as measured by the enacted tax rates that will be in effect when these differences reverse. Valuation allowances are recorded as appropriate to reduce deferred tax assets to the amount considered likely to be realized. The Company had no income tax expense for the years ended December 31, 2020 and 2019.

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

The reconciliation between the U.S. statutory tax rate and the Company's effective tax rate is presented as follows:

	For the Years Ended December 31,	
	2020	2019
U.S. federal statutory income tax benefit applied to loss before income taxes	\$(2,563)	\$(2,525)
State income taxes, net of federal benefit	(410)	(429)
Valuation allowance	1,106	2,835
Deferred tax adjustments	1,827	_
Other permanent items	40	119
Total income tax expense	\$ —	\$ —

The Company's deferred tax assets and liabilities are detailed as follows:

2019
8,438
177
19
104
2,943
499
141
—
3,069
5,390
4,855)
535
(535)
(535)
_

The changes in the Company's valuation allowance were as follows:

	Decemb	December 31,	
	2020	2019	
Beginning Balance	\$14,855	\$11,689	
Change related to current net operating losses	2,238	2,835	
Net change related to generation of tax attributes	695	331	
Change related to deferred tax adjustments	(1,827)		
Ending Balance	\$15,961	\$14,855	

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

As of December 31, 2020, the Company has state and federal net operating loss carryforwards in the amount of \$32,765 and \$46,643, respectively. The federal net operating loss can be carried forward indefinitely, however the utilization of the federal net operating loss for a tax year is equal to the lesser of (1) the aggregate of the net operating loss carryovers to such year, plus the net operating carrybacks to such tax year, or (2) 80% of taxable income determined without regard to the deduction. The Company's state net operating loss carryforwards range in expiration from 2033 to indefinite.

In addition, the Company's net operating loss may be subject to utilization limitations due to changes of control, as defined by tax law under Internal Revenue Code Sections 382.

The Company had no unrecognized income tax benefits for the years ended December 31, 2020 and 2019. To date, no interest and penalties have been recognized related to the underpayment of income taxes. The Company continues to believe its positions are supportable; however, due to uncertainties in any tax audit outcome, the Company's estimates of the ultimate settlement of uncertain tax positions may change and the actual tax benefits may differ from the estimates.

The Company files income tax returns in the U.S. federal and various state jurisdictions. The Company's 2017 through 2019 tax years are subject to U.S. federal and state examination.

13. NET LOSS PER SHARE ATTRIBUTABLE TO COMMON STOCKHOLDERS

Basic EPS is measured as the income or loss available to common stockholders divided by the weighted average common shares outstanding for the period. Diluted EPS presents the dilutive effect on a per-share basis from the potential conversion of convertible securities or the exercise of options and or warrants; the dilutive impacts of potentially convertible securities are calculated using the if-converted method; the potentially dilutive effect of options or warrants are computed using the treasury stock method. Securities that are potentially an anti-dilutive effect (i.e., those that increase income per share or decrease loss per share) are excluded from diluted EPS calculation.

	For the Years Ended December 31		ber 31,	
		2020		2019
Net loss	\$	(12,206)	\$	(11,922)
Deemed dividend		(10,478)		
Net loss attributable to common stockholders (numerator)	\$	(22,684)	\$	(11,922)
Weighted average common shares outstanding - basic and diluted				
(denominator)		98,492		98,684
Net loss per share – basic and diluted	\$	(0.23)	\$	(0.12)

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

Potentially dilutive securities not included in the calculation of diluted net loss per share because to do so would be anti-dilutive are as follows (in common stock equivalent shares):

	December 31,	
	2020	2019
Stock options	2,530	2,230
Preferred stock	6,766	4,421
Warrants	4,135	150
Restricted stock	37,946	35,047
Total potentially anti-dilutive shares	51,377	41,848

14. SEGMENT REPORTING

The Company's operating segments are aggregated into reportable segments only if they exhibit similar economic characteristics and have similar business activities.

The Company has two reportable segments: "Equipment Sales and Hosting" which aggregates the results of its Blockchain Hosting and Equipment Sales operating segments and consists primarily of its blockchain infrastructure and third-party hosting business and equipment sales, and "Mining" consisting of digital asset mining for its own account. The blockchain hosting business generates revenue through the sale of consumption-based contracts for its hosting services which are recurring in nature. Equipment sales revenue is derived from its ability to leverage its partnership with leading equipment manufacturers to secure equipment in advance, which is then sold to its customers. The digital asset mining operation segment generates revenue from operating owned computer equipment as part of a pool of users that process transactions conducted on one or more blockchain networks. In exchange for these services, the Company receives digital currency assets.

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

The primary financial measures used by the CODM to evaluate performance and allocate resources are revenue and gross profit. The CODM does not evaluate performance or allocate resources based on segment asset or liability information; accordingly, the Company has not presented a measure of assets by segment. The segments' accounting policies are the same as those described in the summary of significant accounting policies. The Company excludes certain operating expenses and other income (expense) from the allocations to operating segments. The following table presents revenue and gross profit by reportable segment for the periods presented:

		For the Years Ended December 31,	
	2020	2019	
Equipment Sales and Hosting Segment			
Hosting revenue	\$41,598	\$53,876	
Equipment sales	12,595	_	
Total revenue	54,193	53,876	
Cost of revenue	47,951	43,005	
Gross profit	\$ 6,242	\$10,871	
Mining Segment			
Digital asset mining income	\$ 6,127	\$ 5,647	
Total revenue	6,127	5,647	
Cost of revenue	2,977	5,991	
Gross profit	\$ 3,150	\$ (344)	
Consolidated total revenue	\$60,320	\$59,523	
Consolidated cost of revenue	\$50,928	\$48,996	
Consolidated gross profit	\$ 9,392	\$10,527	

For the years ended December 31, 2020 and 2019, cost of revenue included depreciation expense of \$7,418 and \$3,765, respectively for the Equipment Sales and Hosting segment and \$1,091 and \$2,158, respectively for the Mining segment.

For the years ended December 31, 2020 and 2019, one customer accounted for approximately 27% and 82% of the Equipment Sales and Hosting's segment total revenue, respectively. This customer exercised its right to terminate its hosting contracts with the Company effective April 2020.

A reconciliation of the reportable segment gross profit to loss before income taxes included in the Company's consolidated statements of operations for the years ended December 31, 2020 and 2019 is as follows:

	For the Years Ended	December 31,
	2020	2019
Reportable segment gross profit	9,392	10,527
Gain on legal settlement	5,814	_
Gain from sales of digital currency assets	65	806
Operating expense (income):		
Research and development	5,271	5,480
Sales and marketing	1,771	2,833
General and administrative	14,556	14,707
Total operating expense (income)	21,598	23,020
Operating loss	(6,327)	(11,687)

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

	For the Years Ended	For the Years Ended December 31,		
	2020	2019		
Non-operating income (expense), net:				
Loss on debt extinguishment and other	(1,333)	_		
Interest expense, net	(4,436)	(235)		
Other non-operating expenses, net	(110)	_		
Total non-operating income (expense), net	(5,879)	(235)		
Loss before income taxes	(12,206)	(11,922)		

15. RELATED-PARTY TRANSACTIONS

In the ordinary course of business, the Company enters into various transactions with related parties.

In March 2018, the Company agreed in principle to acquire the assets of two companies principally engaged in digital currency mining from an individual who was subsequently hired by the Company as an executive. In connection with the agreed-upon transaction, the Company entered into an operating lease in March 2018 with a lessor entity controlled by the executive for a facility in South Carolina. In November 2018, the Company exercised its option to terminate the lease in May 2019. For the year ended December 31, 2019, the Company incurred rent expense of \$167 related to the lease.

In June 2018, the Company entered into an agreement for hosting services with an entity that is managed by an individual who was subsequently hired by the Company as an executive. Under the agreement, the Company performed hosting services for a third-party entity that was a customer of the entity. For the year ended December 31, 2018 the Company recognized hosting revenue from the contract with the entity of \$11,406. As of December 31, 2018, no amount was receivable from the entity and the Company had \$8,745 of deferred revenue associated with the agreement. In February 2019, the agreement with the entity was terminated and the Company became the direct seller of hosting services to the third-party entity. Prior to the termination, the Company recognized hosting revenue from the contract with the entity of \$4,662 for the year ended December 31, 2019.

In October 2018, the Company entered into an operating lease with a lessor entity controlled by an executive of the Company for an office facility in Ohio. The Company may terminate the lease at any time with 30 days' notice to the landlord. The lease was terminated in April 2020. For the year ended December 31, 2020 and 2019, the Company incurred rent expense of \$4 and \$14, respectively, related to the lease.

During the year ended December 31, 2019, the Company entered into agreements to provide hosting services to various entities that are managed by individuals that are directors and executives of the Company. For the years ended December 31, 2020 and 2019, the Company recognized hosting revenue from the contracts with these entities of \$6,983 and \$384, respectively. In addition, for the year ended December 31, 2020 there were equipment sales revenue recognized of \$1,402 to these same various entities. As of December 31, 2020 and 2019, the Company had accounts receivable of \$315 and \$72, respectively, from these entities.

As discussed in Note 5, in October 2019, the Company sold certain computer mining equipment to an entity that is managed by an executive of the Company. The Company received cash proceeds from the sale of \$122 and recognized a loss from the transaction of \$917.

The Company reimburses its president and chief executive officer for use of a personal aircraft for flights taken on Company business. For the years ended December 31, 2020 and 2019, the Company incurred reimbursements of \$68 and \$325, respectively. No amount was payable at December 31, 2020 and 2019.

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

16. SUBSEQUENT EVENTS

The Company has evaluated subsequent events through August 11, 2021 the date that the financial statements were available to be issued.

Merger Agreement

On July 20, 2021, the Company entered into an Agreement and Plan of Merger and Reorganization (as the same may be amended, supplemented or otherwise modified from time to time, the "Agreement") with Power & Digital Infrastructure Acquisition Corp., a Delaware corporation ("XPDI"), with XPDI Merger Sub Inc., a Delaware corporation and wholly owned subsidiary of XPDI ("First Merger Sub"), and XPDI Merger Sub 2, LLC, a Delaware limited liability company and wholly owned subsidiary of XPDI ("Second Merger Sub" and, together with First Merger Sub, the "Merger Subs" and, together with XPDI, the "XPDI Parties"). The Agreement and the transactions contemplated thereby were unanimously approved by the boards of directors of each of Core Scientific and XPDI.

Pursuant to the Agreement, and subject to the terms and conditions set forth therein, XPDI will acquire the Company through a series of transactions, including (x) the merger of Merger Sub with and into Core Scientific (the "First Merger"), with Core Scientific surviving the First Merger as a wholly owned subsidiary of XPDI, and (y) the merger of Core Scientific with and into Second Merger Sub (the "Second Merger" and, together with the First Merger, the "Mergers"), with Second Merger Sub surviving the Second Merger as a wholly owned subsidiary of XPDI. As a result of the Mergers, among other things, each outstanding share of common stock, \$0.00001 per share, of Core Scientific ("Core Scientific Common Stock") will be cancelled in exchange for the right to receive a number of shares of Class A common stock, par value \$0.0001 per share, of XPDI in an amount that is approximately equal to the quotient obtained by dividing (a) an amount equal to (x) \$4.0 billion, divided by (y) the number of shares of Core Scientific Common Stock on a fully-diluted basis, by (b) \$10.00.

Blockcap Acquisition

Core Scientific acquired Blockcap, Inc. ("Blockcap") on July 30, 2021. Consideration given consisted of the issuance of approximately 72 million shares of Core Scientific common stock, approximately 4 million shares of Core Scientific restricted stock and approximately 5 million options to purchase shares of Core Scientific common stock. The acquisition is expected to be accounted for as a business combination using the acquisition method whereby the net assets acquired, and the liabilities assumed are recorded at fair value. The difference between the aggregate consideration relative to the fair value of the identifiable net assets will be recorded as goodwill. The Company is still in the process of completing its initial accounting for the business combination, including determining the fair value of consideration transferred, assets acquired and liabilities assumed and the allocation of the purchase price to goodwill.

Financing activities

In January 2021, the Company borrowed \$10 million from a stockholder for the purchase of bitcoin mining equipment. The loan bears interest at 10% per annum over a two-year term. The loan was issued with a warrant to purchase 120 shares of common stock at an exercise price of \$6.74 per share. The warrant has a two-year term.

In February 2021, the Company entered in an amendment of the Silverpeak loan that included the issuance of an additional \$9,000 tranche of senior secured loans. This loan, including the additional tranche issued in February 2021, was paid off in full in April 2021. See Note 8 for further discussion on the Silverpeak loan.

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

In March 2021, the Company received \$4 million of additional loans under the master equipment finance agreement with NYDIG to finance the Company's acquisition of blockchain computing equipment that bear an interest rate of 15% and have a term of 24 months from issuance. In April 2021, the Company entered into a Secured Convertible Note Purchase Agreement and issued \$215 million of convertible notes to new and existing lenders. The convertible notes have a maturity date of April 2025 and bear interest of 10% per annum, of which 4% is payable in cash and 6% is payable in kind. The convertible notes are convertible into common or preferred shares at the option of the holder upon the occurrence of a) an IPO or SPAC merger with gross proceeds to the Company of at least \$50 million or b) a change in control. The conversion price is equal to a) 65-80% of the price paid for equity securities in an IPO, depending on the timing of the IPO or b) 65-80% of the fair value per share paid in a change in control or SPAC merger, depending on the timing of the change in control or SPAC merger. The proceeds from the convertible notes were used, in part, to repay the Silverpeak loan described in Note 8 in full. In May 2021, the Company received \$13 million of additional loans under the master equipment finance agreement with NYDIG to finance the Company's acquisition of blockchain computing equipment that bear an interest rate of 14.25% and have a term of 24 months from issuance.

In 2021 through the date that the financial statements were available to be issued, the Company entered into arrangements with various parties to finance the acquisition of computer and networking equipment with lease terms between 2 and 4 years and fixed payments over the non-cancelable lease term of \$1 million.

Related party transactions

In May through July 2021, the Company entered various promissory notes with Blockcap, a related party entity that was managed by individuals that are directors and executives of the Company. The Company provided hosting services to Blockcap prior to the Company acquiring Blockcap on July 30, 2021 as described above. The promissory notes deferred \$19 million of amounts originally due in June and July 2021 from Blockcap under the hosting arrangement until August 2021.

Resignation of CEO

The Company's President and Chief Executive Officer, Kevin Turner, resigned in May 2021. Mr. Turner's employment agreement provides that he will be entitled to receive 12 months of base salary totaling \$300 and, if Mr. Turner elects COBRA coverage, for 12 months he will receive a monthly cash payment equal to the full premium for company-paid continuing health care benefits. As of Mr. Turner's separation date, 9,038 RSUs had satisfied the time vested condition but remain subject to performance condition described in Note 11.

Employee RSU grants

In 2021 through the date that the financial statements were available to be issued, the Company granted 16 million RSUs to various employees of the Company.

Letter of intent to develop hosting facility

In March 2021 the Company signed a letter of intent with Minnkota Power Cooperative to enter into leases and purchase property to develop a hosting facility in Grand Forks, North Dakota as well as enter into a power supply purchase agreement to purchase 100 megawatts of power supply once construction of the hosting facility is complete. Substantially all of the payments for the intended leases would be for a five-year to thirty-year term (comprising an initial five-year term with five five-year options to renew) with purchase options exercisable at any time for approximately \$6 million less any rent paid to date and subject to certain other adjustments. In

Core Scientific Holding Co. Notes to Consolidated Financial Statements (Amounts in thousands, except per share amounts)

addition, in 2021 through the date that the financial statements were available to be issued the Company made deposits of \$5.2 million towards the cost of construction and purchasing infrastructure for the development of the hosting facility that is refundable to the Company under certain conditions if the proposed transactions are not consummated on or before October 2021.

Purchase obligations

In 2021 through the date that the financial statements were available to be issued, the Company entered into agreements to purchase blockchain mining equipment totaling approximately \$760 million of which approximately \$310 million was paid for deliveries or as deposits for blockchain mining equipment scheduled to be delivered in 2021 and 2022. As of the date that the financial statements were available to be issued, the aggregate amount of the Company's purchase obligations totaled approximately \$477 million, substantially all of which are expected to be settled within one year of the date that the financial statements were available to be issued.

Core Scientific Holding Co. Consolidated Balance Sheets (in thousands, except par value)

	September 30, 2021 Restated (Unaudited)	December 31, 2020
Assets	(Chauditeu)	
Current Assets:		
Cash and cash equivalents	\$ 147,906	\$ 8,671
Restricted cash	12,101	50
Accounts receivable	602	792
Accounts receivable from related parties	261	315
Deposits for equipment	469,890	54,818
Digital currency assets	116,233	63
Other current assets	9,978	6,210
Total Current Assets	756,971	70,919
Property, plant and equipment, net	219,795	85,244
Goodwill	1,106,968	58,241
Intangible assets, net	8,709	6,674
Other noncurrent assets	14,110	4,499
Total Assets	\$ 2,106,553	\$ 225,577
Liabilities, Redeemable Preferred Stock and Stockholders' Equity		
Current Liabilities:		
Accounts payable	\$ 28,796	\$ 3,057
Accrued expenses and other	35,074	3,585
Deferred revenue	74,855	38,113
Deferred revenue from related parties	131,284	6,730
Capital lease obligations, current portion	2,525	2,146
Notes payable, current portion	25,202	16,016
Total Current Liabilities	297,736	69,647
Capital lease obligations, net of current portion	1,524	2,263
Notes payable, net of current portion (includes \$442,834 and \$- at fair value)	471,930	19,864
Other noncurrent liabilities	1,994	103
Total Liabilities	773,184	91,877
Contingently redeemable preferred stock; \$0.00001 par value; 50,000 shares authorized; 6,766 shares issued and		
outstanding at September 30, 2021 and December 31, 2020	44,476	44,476
Commitments and contingencies (Note 6)	,	,
Stockholders' Equity:		
Common stock; \$0.00001 par value; 200,000 shares authorized; 170,818 and 98,607 shares issued and		
outstanding at September 30, 2021 and December 31, 2020	2	1
Additional paid-in capital	1,385,381	163,967
Accumulated deficit	(87,938)	(74,744)
Accumulated other comprehensive loss	(8,552)	
Total Stockholders' Equity	1,288,893	89,224
Total Liabilities, Redeemable Preferred Stock and Stockholders' Equity	\$ 2,106,553	\$ 225,577

Core Scientific Holding Co. Consolidated Statements of Operations (in thousands, except per share amounts) (Unaudited)

	Nine Months Ende	d September 30,
	2021	2020
Total evenue:	Restated	
Hosting revenue from customers	\$ 37,836	\$ 28,667
Hosting revenue from related parties	13,906	3,382
Equipment sales to customers	84,378	1,987
Equipment sales to related parties	29,057	285
Digital asset mining income	77,511	2,312
Total revenue	242,688	36,633
Costs of revenue:		
Cost of hosting services	48,956	28,185
Cost of equipment sales	82,328	2,123
Cost of digital asset mining	13,909	1,598
Total costs of revenue	145,193	31,906
Gross profit	97,495	4,727
(Loss) gain on legal settlements	(2,603)	5,814
Gain from sales of digital currency assets	405	52
Impairment of digital currency assets	(12,552)	(5)
Operating expenses:		
Research and development	4,231	4,183
Sales and marketing	2,186	1,401
General and administrative	46,992	10,797
Total operating expenses	53,409	16,381
Operating income (loss)	29,336	(5,793)
Non-operating expenses, net:		
Loss on debt from extinguishment	8,016	1,333
Interest expense, net	26,550	2,683
Other non-operating expense, net	8,661	88
Total non-operating expense, net	43,227	4,104
Loss before income taxes	(13,891)	(9,897)
Income tax expense	(697)	
Net loss	(13,194)	(9,897)
Deemed dividend from common to preferred exchange		(10,478)
Net loss attributable to common stockholders	\$ (13,194)	\$ (20,375)
Net loss per share (Note 8):	<u> </u>	
Basic	\$ (0.11)	\$ (0.21)
Diluted	\$ (0.11)	\$ (0.21)
Weighted average shares outstanding:		
Basic	115,482	98,453
Diluted	115,482	98,453
Diluitu	113,462	70,433

Core Scientific Holding Co. Consolidated Statements of Comprehensive Loss (in thousands, except per share amounts) (Unaudited)

	Nine Montl Septemb	
	2021 Restated	2020
Net loss	\$(13,194)	\$(9,897)
Other comprehensive loss, net of income taxes:		
Change in fair value attributable to instrument-specific credit risk of convertible notes measured at fair value under the fair		
value option, net of tax effect of \$— and \$—	(8,552)	
Total other comprehensive loss, net of income taxes	(8,552)	_
Comprehensive loss	\$(21,746)	\$(9,897)

Core Scientific Holding Co. Consolidated Statements of Changes in Contingently Redeemable Convertible Preferred Stock and Stockholders' Equity (in thousands) (Unaudited)

	Contingently Rec		Commo	on Stock	Additional Paid-In	Accumulated	Accumulated Other Comprehensive	Total Stockholders'
	Shares	Amount	Shares	Amount	Capital	Deficit	Loss	Equity
Balance at December 31, 2020	6,766	\$ 44,476	98,607	\$ 1	\$ 163,967	\$ (74,744)	\$ —	\$ 89,224
Net loss (restated)	_	_	_	_	_	(13,194)	_	(13,194)
Stock-based compensation	_	_	_	_	31,012	_	_	31,012
Accumulated other comprehensive loss (restated)	_	_	_	_	_	_	(8,552)	(8,552)
Issuances of common stock- business combination	_	_	72,186	1	1,189,906	_	_	1,189,907
Issuances of common stock warrants and options			25		496			496
Balance at September 30, 2021 (restated)	6,766	\$ 44,476	170,818	\$ 2	\$1,385,381	\$ (87,938)	\$ (8,552)	\$ 1,288,893
•								
Balance at December 31, 2019	4,421	\$ 29,526	99,141	\$ 1	\$ 168,866	\$ (62,538)	\$ —	\$ 106,329
Net loss	_	_	_	_	_	(9,897)	_	(9,897)
Stock-based compensation	_	_	_	_	2,446	_	_	2,446
Exchange of common stock for Series A contingently redeemable convertible preferred stock	1,802	12,308	(1,096)	_	(12,308)	_	_	(12,308)
Issuances of Series A contingently redeemable convertible preferred stock	229	1,545	_	_	_	_	_	_
Issuances of Series B contingently redeemable convertible preferred stock	314	1,097	_	_	_	_	_	_
Issuances of common stock- asset acquisition	_	_	562	_	2,405	_	_	2,405
Issuances of common stock warrants and options	_	_	_	_	1,967	_	_	1,967
Balance at September 30, 2020	6,766	\$ 44,476	98,607	\$ 1	\$ 163,376	\$ (72,435)	<u> </u>	\$ 90,942

Core Scientific Holding Co. Consolidated Statements of Cash Flows (in thousands) (Unaudited)

	Nine Montl Septemb	
	2021 Restated	2020
Cash flows from Operating Activities:		
Net loss	\$ (13,194)	\$ (9,897)
Adjustments to reconcile net loss to net cash provided by (used in) operating activities:		
Depreciation and amortization	12,886	6,613
Stock-based compensation	31,012	2,446
Digital asset mining income	(77,511)	(2,312)
Deferred income taxes	3,604	_
Loss on legal settlements	2,603	_
Loss on debt extinguishment	8,016	1,333
Fair value adjustment on convertible notes	15,937	_
Amortization of debt discount and debt issuance costs	1,024	858
Losses on disposals of property, plant and equipment	17	(92)
Impairments of digital currency assets	12,552	5
Provision for doubtful accounts	_	(4)
Changes in working capital components:		
Accounts receivable, net	(6,641)	(3,415)
Accounts receivable from related parties	55	(1,903)
Digital currency assets	27,316	2,314
Deposits for equipment	(414,771)	(19,846)
Other current assets	970	(360)
Accounts payable	(35,132)	(790)
Accrued expenses and other	17,945	1,330
Deferred revenue	254,530	5,878
Other noncurrent assets and liabilities, net	(7,692)	(620)
Net cash used in operating activities	(166,474)	(18,462)
Cash flows from Investing Activities:		
Purchases of property, plant and equipment	(116,074)	(7,970)
Cash acquired in business combination	704	_
Proceeds from sales of property, plant and equipment	_	92
Other	(154)	(1,350)
Net cash used in investing activities	(115,524)	(9,228)
Cash flows from Financing Activities:	' <u></u>	
Proceeds from issuances of common stock options and warrants	496	2,642
Issuances of debt	475,301	30,658
Principal payments on debt	(42,513)	(5,819)
Net cash provided by financing activities	433,284	27,481
Increase (decrease) in cash, cash equivalents, and restricted cash	151,286	(209)
Cash, cash equivalents and restricted cash—beginning of period	8,721	6,907
Cash, cash equivalents and restricted cash—end of period	\$ 160,007	\$ 6,698
Cash, Cash equivalents and restricted cash—end of period	\$ 100,007	\$ 0,098

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

1. ORGANIZATION AND DESCRIPTION OF BUSINESS

MineCo Holdings, Inc. was incorporated on December 13, 2017 in the State of Delaware and changed its name to Core Scientific, Inc. pursuant to an amendment to its Certificate of Incorporation dated June 12, 2018. On August 17, 2020 the Company engaged in a holdco restructuring to facilitate a borrowing arrangement by the Company pursuant to which Core Scientific Inc. was merged with and into a wholly owned subsidiary of the Company and became a wholly owned subsidiary of the Company and the stockholders of Core Scientific Inc. became the shareholders of Core Scientific Holding Co. ("Core Scientific" or the "Company"). The Company currently operates under a holding company structure with Core Scientific Holding Co. as parent and Core Scientific, Inc. as its operating subsidiary. The Company, headquartered in Austin, Texas, is an infrastructure, technology and services company that conducts, or plans to conduct, the following business activities:

- Owning and operating datacenter facilities in the U.S. to provide colocation and hosting services for distributed ledger technology, also commonly known as blockchain;
- Owning and operating computer equipment used to process transactions conducted on one or more blockchain networks in exchange for transaction processing fees rewarded in digital currency assets, commonly referred to as mining;
- Developing blockchain-based platforms and applications, including infrastructure management, security technologies, mining optimization, and recordkeeping;

Merger Agreement

On July 20, 2021, the Company entered into an Agreement and Plan of Merger and Reorganization (as the same may be amended, supplemented or otherwise modified from time to time, the "Agreement") with Power & Digital Infrastructure Acquisition Corp., a Delaware corporation ("XPDI"), with XPDI Merger Sub Inc., a

Delaware corporation and wholly owned subsidiary of XPDI ("First Merger Sub"), and XPDI Merger Sub 2,

LLC, a Delaware limited liability company and wholly owned subsidiary of XPDI ("Second Merger Sub" and,

together with First Merger Sub, the "Merger Subs" and, together with XPDI, the "XPDI Parties"). The

Agreement and the transactions contemplated thereby were unanimously approved by the boards of directors of each of Core Scientific and XPDI.

Pursuant to the Agreement, and subject to the terms and conditions set forth therein, XPDI will acquire the Company through a series of transactions, including (x) the merger of Merger Sub with and into Core Scientific (the "First Merger"), with Core Scientific surviving the First Merger as a wholly owned subsidiary of XPDI, and (y) the merger of Core Scientific with and into Second Merger Sub (the "Second Merger" and, together with the First Merger, the "Mergers"), with Second Merger Sub surviving the Second Merger as a wholly owned subsidiary of XPDI. As a result of the Mergers, among other things, each outstanding share of common stock, \$0.00001 per share, of Core Scientific ("Core Scientific Common Stock") will be cancelled in exchange for the right to receive a number of shares of Class A common stock, par value \$0.0001 per share, of XPDI in an amount that is approximately equal to the quotient obtained by dividing (a) an amount equal to (x) \$4.0 billion, divided by (y) the number of shares of Core Scientific Common Stock on a fully-diluted basis, by (b) \$10.00.

2. RESTATEMENT OF PREVIOUSLY-ISSUED FINANCIAL STATEMENTS

Subsequent to the filing of its unaudited consolidated financial statements for the nine months ended September 30, 2021, the Company determined the fair value measurement of its convertible notes as of September 30, 2021 did not appropriately consider that the minimum payoff at maturity of certain of its convertible notes is two times the face value of the notes plus accrued interest. In addition, the Company had not presented separately in other comprehensive income the portion of the total change in the fair value of the convertible notes that resulted from a change in the instrument-specific credit risk as required by U.S. GAAP for

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

financial liabilities for which the fair value option is elected. The Company has concluded that the errors identified were material to its consolidated statement of operations for the nine months ended September 30, 2021. Accordingly, previously reported amounts have been revised to reflect the correction of these as well as other previously identified immaterial errors related to the accounting for the acquisition of Blockcap described in Note 4 impacting digital currency assets, goodwill, other noncurrent assets, accounts payable and other noncurrent liabilities.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

The following tables present the revised results for the nine months ended September 30, 2021, the adjustments made and the previously reported amounts to summarize the effect of the restatement on the previously-issued consolidated financial statements for the periods impacted (in thousands, unaudited).

Consolidated Balance Sheets	_	Reported S		nber 30, 2021 djustment		Restated
Assets:		перопец	210	уизинени		resilieu
Digital currency assets	\$	115,856	\$	377	\$	116,233
Total Current Assets	\$	756,594	\$	377	\$	756,971
Goodwill	\$	1,106,015	\$	953	\$1	,106,968
Total Assets	\$2	2,105,223	\$	1,330	\$2	,106,553
Liabilities:						
Accounts payable	\$	28,689	\$	107	\$	28,796
Accrued expenses and other	\$	33,849	\$	1,225	\$	35,074
Total Current Liabilities	\$	296,404	\$	1,332	\$	297,736
Notes payable, net of current portion	\$	467,662	\$	4,268	\$	471,930
Total Liabilities	\$	767,584	\$	5,600	\$	773,184
Stockholders' Equity:						
Accumulated deficit	\$	(92,220)	\$	4,282	\$	(87,938)
Accumulated other comprehensive loss	\$	_	\$	(8,552)	\$	(8,552)
Total Stockholders' Equity	\$	1,293,163	\$	(4,270)	\$1	,288,893
Notes payable, net of current portion at fair value	\$	438,566	\$	4,268	\$	442,834
Consolidated Statements of Operations		Nine Mo Reported	nths	Ended Septeml Adjustment		, 2021 Restated
Interest expense, net		\$ 26,806		\$ (256)		26,550
Other non-operating expense, net		\$ 12,945		\$ (4,284)		8,661
Total non-operating expense, net		\$ 47,767		\$ (4,540)		43,227
Loss before income taxes		\$(18,431)		\$ 4,540		(13,891)
Income tax (benefit)		\$ (955)		\$ 258		(697)
Net loss		\$(17,476)		\$ 4,282		(13,194)
1000		\$(17,470)		Ψ 4,202	Ψ	(13,174)
Net loss per share (Note 9):						
Basic		\$ (0.15)		\$ 0.04	\$	(0.11)
Diluted		\$ (0.15)		\$ 0.04	\$	(0.11)
Consolidated Statements of Other Comprehensive Loss			nths	Ended Septeml		
Net loss		Reported \$(17,476)		Adjustment \$ 4,282		Restated (13,194)
Other comprehensive loss, net of income taxes:		\$(17,470)		\$ 4,202	Ψ	(13,174)
Change in fair value attributable to instrument-specific credit risk of						
convertible notes measured at fair value under the fair value option,						
net of tax effect of \$— and \$—				(8,552)		(8,552)
		0(17.470)			<u></u>	
Comprehensive loss		<u>\$(17,476)</u>		\$ (4,270)	\$	(21,746)

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

Consolidated Statements of Cash Flows	Nine Months Ended September 30, 2021				
	Reported	Adjustment	Restated		
Cash flows from Operating Activities:					
Net loss	\$(17,476)	\$ 4,282	\$(13,194)		
Adjustments to reconcile net loss to net cash provided by (used in) operating activities:					
Fair value adjustment on convertible notes	\$ 20,221	\$ (4,284)	\$ 15,937		
Changes in working capital components:					
Accrued expenses and other	\$ 17,943	\$ 2	\$ 17,945		

3. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

The accompanying unaudited consolidated financial statements reflect the application of certain significant accounting policies as described below and elsewhere in these notes to the financial statements.

Basis of Presentation

The consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP"). All intercompany balances and transactions have been eliminated in consolidation.

Unaudited Interim Financial Information

The accompanying interim consolidated balance sheet as of September 30, 2021, the consolidated statements of operations and comprehensive loss, consolidated statements of changes in contingently redeemable convertible preferred stock and stockholders' equity, and cash flows for the nine months ended September 30, 2021 and 2020 and the related footnote disclosures are unaudited. The unaudited interim consolidated financial statements have been prepared on the same basis as the audited consolidated financial statements and, in the opinion of management, include all adjustments, which include only normal recurring adjustments, necessary for the fair statement of these interim financial statements. The results for the nine months ended September 30, 2021 are not necessarily indicative of the results expected for the full fiscal year or any other future annual or interim period.

Reported	Adjustment	Restated
Keporteu	Aujustinent	Restateu

Use of Estimates

The preparation of the Company's unaudited consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of income and expenses during the reporting period. Some of the more significant estimates include the valuation of the Company's common and preferred shares, the determination of the grant date fair value of stock-based compensation awards, the valuation of goodwill, the fair value of convertible debt, acquisition purchase price accounting, and income taxes. These estimates are based on information available as of the date of the financial statements; therefore, actual results could differ from management's estimates.

Cash and Cash Equivalents

As of September 30, 2021, cash equivalents included \$144.0 million of highly liquid money market funds, which are classified as Level 1 within the fair value hierarchy. There were no cash equivalents as of December 31, 2020.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

Restricted Cash

As of September 30, 2021 and December 31, 2020, restricted cash included \$12.1 million and \$0.1 million respectively, primarily related to cash held in escrow to pay for construction and development activities.

Fair Value Measurements

The Company measures certain assets and liabilities at fair value on a recurring or non-recurring basis in certain circumstances. Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. To increase the comparability of fair value measures, the following hierarchy prioritizes the inputs to valuation methodologies used to measure fair value:

- Level 1 Valuations based on quoted prices for identical assets and liabilities in active markets.
- Level 2 Valuations based on observable inputs other than quoted prices included in Level 1, such as quoted prices for similar assets and liabilities in active markets, quoted prices for identical or similar assets and liabilities in markets that are not active, or other inputs that are observable or can be corroborated by observable market data.
- Level 3 Valuations based on unobservable inputs reflecting the Company's own assumptions, consistent with reasonably available assumptions made by other market participants. These valuations require significant judgment.

The Company uses observable market data when determining fair value whenever possible and relies on unobservable inputs only when observable market data is not available.

Recurring fair value measurements

As discussed in Note 5 below, in April 2021, the Company entered into a Secured Convertible Note Purchase Agreement (the "Secured Convertible Notes") and issued \$215.0 million of Secured Convertible Notes to new and existing lenders. In the third quarter of 2021, the Company issued an additional \$203.3 million of convertible notes in August and September 2021 under substantially the same terms and conditions as the original April 2021 notes except that the August and September 2021 notes have a minimum payoff based on the face value plus accrued interest rather than two times the outstanding face amount plus accrued interest. In addition, the August and September 2021 notes are unsecured until an IPO or SPAC merger and then they are secured pari passu with the Secured Convertible Notes (the "Unsecured Convertible Notes"). Payment in Kind (PIK) interest is recorded quarterly by increasing the outstanding principal amount of the notes. As a result principal balance of the convertible notes increased by \$2.6 million during the nine months ended September 30, 2021. The Company has elected to measure its convertible notes at fair value on a recurring basis because the Company believes it better reflects the underlying economics of the convertible notes, which contain multiple embedded derivative features. The Company presents changes in fair value of the convertible notes during the period as follows: (1) the 10% contractual rate of interest on the convertible notes (consisting of 4% cash interest and 6% PIK interest) is presented as interest expense, net on the Consolidated Statements of Operations, (2) changes in fair value attributable to the Company's own credit risk are presented within accumulated other comprehensive loss on the Consolidated Balance Sheets and as a component of other comprehensive loss on the Consolidated Statements of Comprehensive Loss; and (3) other fair value changes are presented within other non-operating expense, net on the Consolidated Statements of Operations.

The Company's convertible notes had a fair value of \$442.8 million compared to a principal amount of \$420.9 million at September 30, 2021.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

The principal amount of the Secured Convertible Notes and Unsecured Convertible Notes as of September 30, 2021 reflects the proceeds received plus any PIK interest added to the principal balance of the notes. Upon a successful financing prior to April 30, 2022 (e.g. a SPAC transaction or qualified financing), the conversion price for both the Secured and Unsecured Convertible Notes will become fixed at 80% of the financing price. The holders then have the right to convert at any time until maturity. At maturity, any Secured Convertible Notes not converted will be owed two times the original face value plus accrued interest; any Unsecured Convertible Notes not converted will be owed the original face value plus accrued interest. In addition, at any time on or after the successful financing, the Company has the right to prepay the Secured and Unsecured Notes at the minimum payoff of two times the outstanding face value plus accrued interest for the Secured Convertible Notes and the outstanding face value plus accrued interest for the Unsecured Convertible Notes. The total amount that would be owed on the Secured Notes plus Unsecured Notes if prepaid as of September 30, 2021 was \$638.5 million.

The following presents the levels of the fair value hierarchy for the Company's convertible notes by issuance date measured at fair value on a recurring basis as of September 30, 2021 (in thousands):

		Fair value hierarchy			
	Principal	Level 1	Level 2	Level 3	Fair value
Convertible notes:					
April 19, 2021 ¹	\$ 96,068	\$ —	\$ —	\$ 97,618	\$ 97,618
April 21, 2021 ¹	5,060	_	_	5,480	5,480
April 23, 2021 ¹	45,540	_	_	49,314	49,314
April 26, 2021 ¹	76,912	_	_	83,215	83,215
August 20, 2021 notes ²	50,250	_	_	49,508	49,508
September 10, 2021 ²	16,000	_	_	15,910	15,910
September 23, 2021 ²	75,530	_	75,530	_	75,530
September 24, 2021 ²	59,604	_	59,604	_	59,604
September 27, 2021 ²	1,961	_	1,961	_	1,961
Accrued PIK interest ^{1,2,3}	_	_	4,694	_	4,694
Total convertible notes	\$420,925	\$ —	\$141,789	\$301,045	\$442,834

Secured convertible notes (includes principal balance at issuance and PIK interest) which considers the minimum payoff at maturity of two times the face value of the note plus accrued interest.

The following presents a roll forward of the activity for the Company's convertible notes measured at fair value on a recurring basis as of September 30, 2021 (in thousands):

	Level 1	Level 2	Level 3	Total
Balance at December 31, 2020	<u>\$</u>	<u> </u>	\$ <u> </u>	\$ —
Convertible notes issued (including PIK principal recorded)	_	420,925	_	420,925
Transfers to level 3	_	(283,830)	283,830	_
Change in fair value from inception	_	4,694	17,215	21,909
Balance at September 30, 2021	\$ <u> </u>	\$ 141,789	\$ 301,045	\$ 442,834

Unsecured convertible notes which considers the minimum payoff at maturity of one times the face value of the note plus accrued interest.

Represents PIK interest accrued as of September 30, 2021 which will be recorded as additional principal for each respective convertible note on October 1, 2021.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

The fair value of the Company's convertible notes payable was determined using a market approach based on observable market prices for similar securities when available. When observable market data is not available, the Company uses an as-converted value plus risk put option model that includes certain unobservable inputs that may be significant to the fair value measurement such as probability of a financing event occurring (e.g., a SPAC merger or qualified financing), expected term, volatility and the negotiation discount. The fair value of the Secured Convertible Notes considers the minimum payoff at maturity of two times the face value of the note plus accrued interest, as well as the opportunity for upside appreciation if the value of the Company's stock increases 60% or more relative to the pricing at the financing event (since the conversion price is set at 80% of the stock price at the financing event, a stock price appreciation of 60% would match the minimum payoff of two times the face value plus accrued interest). The fair value of the Unsecured Convertible Notes considers the minimum payoff at maturity of one times the face value of the note plus accrued interest, as well as the opportunity for upside appreciation if the value of the Company's stock falls no more than 20% relative to the pricing at the financing event (since the conversion price is set at 80% of the stock price at the financing event, a stock price decline of 20% would match the minimum payoff of one times the face value plus accrued interest).

Securities are transferred from Level 2 to Level 3 when observable market prices for similar securities are no longer available and unobservable inputs becomes significant to the fair value measurement. All transfers into and out of level 3 are assumed to occur at the beginning of the quarterly reporting period in which they occur. As of September 30, 2021, Level 3 financial instruments included all the Secured Convertible Notes issued in April 2021 and certain Unsecured Convertible Notes issued in August and early September 2021 as the Company effect of unobservable inputs became significant to the fair value measurement due to the time lapse between the issuance of the notes and the reporting date. The Unsecured Convertible Notes issued in late September 2021 are considered to be Level 2 financial instruments as of September 30, 2021 based on the availability of observable market transactions for similar securities that occurred close to the valuation date. If unobservable inputs to the fair value measurement of the Unsecured Convertible Notes issued in late September 2021 becomes significant, the Notes will be transferred from Level 2 to Level 3.

The following presents significant Level 3 unobservable inputs used to measure fair value of certain convertible notes September 30, 2021 (dollars in thousands):

	Unobservable				Weighted
	Fair value	Input	Low	High	Average1
Convertible notes (Level 3)	\$301,045	Probability of financing event	100.0%	100.0%	100.0%
		Expected term (years)	0.25	0.75	0.28
		Volatility	39.8%	40.2%	39.8%
		Negotiation discount	44.6%	54.0%	52.0%

Weighted average based on the fair value of convertible notes. In addition, expected term and volatility are also weighted based on 95% probability of a SPAC merger occurring and 5% probability of a qualified financing occurring.

The probability of a financing event occurring refers to the probably of a SPAC merger or qualified financing occurring that would give the holders of the convertible notes the option to convert the instrument into shares of the Company's common stock. An increase in the probability of a conversion event occurring would, in isolation, result in an increase in the fair value measurement of the convertible notes. As of September 30, 2021, the Company's valuation of the convertible notes assumed a 95% probability of a SPAC merger occurring and a 5% probability of a qualified financing occurring.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

Expected term is an input into the risk put option model that measures the length of time the instrument is expected to be outstanding before it is exercised or terminated. An increase in expected term, in isolation, would generally result in an increase in the fair value measurement of the convertible notes.

Volatility is an input into the risk put option model that measures the variability in possible returns for the convertible notes based on how much the price of underlying shares change in value over time. An increase in volatility, in isolation, would generally result in an increase in the fair value measurement of the convertible notes.

The negotiation discount is a calibration adjustment which reflects the illiquidity of the instruments and the Company's negotiating position. Since the transaction was an orderly transaction, the Company deemed that the fair value equaled the transaction price at initial recognition. A decrease in the negotiation discount, in isolation, would generally result in an increase in the fair value measurement of the convertible notes.

The increase or decrease in the fair value of the convertible notes resulting from changes to the probability of a financing event, expected term or volatility assumptions would be largely mitigated by interrelated changes to the negotiation discount. However, other changes, such as a decrease to the negotiation discount resulting from the occurrence of a financing event, are not interrelated to other inputs and could result in a significant increase in the fair value of the convertible notes.

The Company presents separately in other comprehensive income (loss) the portion of the total change in the fair value of the convertible notes that resulted from a change in the instrument-specific credit risk on the convertible notes. The amount of change in the fair value attributable to instrument-specific credit risk is determined by comparing the amount of the total change in fair value to the amount of change in fair value that would have occurred if the Company's credit risk had not changed during the period as reflected in the discount rates applied to the debt and risk put option.

The following summarizes the fair value adjustments and debt issuance costs recognized on the convertible notes:

(in thousands)	Financial statement line item	nonths ended aber 30, 2021
Cash interest payments	Interest expense, net	\$ 4,850
Payment-in-kind (PIK) interest	Interest expense, net	7,274
Instrument specific credit risk		
	Accumulated other comprehensive loss	8,552
Other fair value adjustments	Other non-operating expense, net	8,663
Total fair value adjustments		\$ 29,339
Debt issuance costs	Interest expense, net	\$ 10,664

Nonrecurring fair value measurements

The Company's non-financial assets, including digital currency assets, property, plant and equipment, goodwill, and intangible assets are measured at estimated fair value on a nonrecurring basis. These assets are adjusted to fair value only when an impairment is recognized, or the underlying asset is held for sale. Refer to the discussion of digital assets below for more information regarding fair value considerations when measuring the impairment of digital assets held.

No non-financial assets were classified as Level 3 as of September 30, 2021 or December 31, 2020.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

Fair value of financial instruments

The Company's financial instruments include cash and cash equivalents, restricted cash, accounts receivable, net, accounts payable, notes payable and certain accrued expenses and other liabilities. The carrying amount of these financial instruments, other than notes payable discussed below, approximates fair value due to the short-term nature of these instruments.

The fair value of the Company's notes payable (excluding the Convertible Notes carried at fair value described above), which are carried at amortized cost, was determined based on a discounted cash flow approach using market interest rates of instruments with similar terms and maturities and an estimate for our standalone credit risk. We classified the other notes payable as Level 3 financial instruments due to the considerable judgment required to develop assumptions of the Company's standalone credit risk and the significance of those assumptions to the fair value measurement. The estimated fair value of the Company's other notes payable, including both the current and noncurrent portion, was \$54.9 million at September 30, 2021 and \$39.0 million at December 31, 2020. The carrying values of the notes payable, including both the current and noncurrent portion, was \$54.3 million and \$35.9 million at September 30, 2021 and December 31, 2020, respectively.

Revenue From Contracts With Customers

The Company primarily generates revenue from contracts with customers from hosting services and, sales of computer equipment. The Company generally recognizes revenue when the promised service is performed, or control of the promised equipment is transferred to customers. Revenue excludes any amounts collected on behalf of third parties, including sales and indirect taxes.

Performance Obligations

The Company's performance obligations primarily relate to hosting services and equipment sales, which are described in Note 2 of the Company's Consolidated Financial Statements as of and for the nine months ended September 30, 2021 and 2020. The Company has performance obligations associated with commitments in customer hosting contracts for future services and commitments to acquire and deploy customer equipment that have not yet been recognized in the financial statements. For contracts with original terms that exceed one year (typically ranging from 18 to 36 months), the transaction price allocated to those remaining performance obligation as of September 30, 2021 were \$535.4 million. For contracts with original terms that exceed one year (typically ranging from 18 to 36 months), those remaining performance obligations not yet recognized as of September 30, 2020 were \$13.2 million.

Deferred Revenue

The Company records contract liabilities in Deferred Revenue on the consolidated balance sheets when cash payments are received in advance of performance and recognizes them as revenue when the performance obligations are satisfied. The Company's deferred revenue balance as of September 30, 2021 and December 31, 2020 was \$206.1 million and \$44.8 million, respectively, all from advance payments received during the periods then ended

In the nine months ended September 30, 2021, the Company recognized \$44.2 million of revenue that was included in the deferred revenue balance as of the beginning of the year, primarily due to the deployment of customer equipment for which advanced payment had been received from customers prior to January 1, 2021. In the nine months ended September 30, 2020, the Company recognized \$7.8 million of revenue that was included in the deferred revenue balance as of the beginning of the year, primarily due to the performance of hosting

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

services for which advance payments had been received from customers prior to January 1, 2020. Advanced payments for hosting services are typically recognized in the following month and advanced payments for equipment sales are generally recognized within one year.

Digital Asset Mining Income

The Company derives its digital asset mining income from operating its owned computer equipment as part of a pool of users, facilitated by a pool operator, that processes transactions conducted on one or more blockchain networks. The arrangements with pool operators are terminable at any time by either party. In exchange for providing computing power to the pool, the Company is entitled to a fractional share of the cryptocurrency award the mining pool operator receives from the blockchain network upon successfully adding a block to the blockchain. The Company's fractional share is based on the proportion of computing power the Company contributed to the mining pool operator to the total computing power contributed by all mining pool participants in processing the block.

Providing computing power in digital asset transaction verification services is an output of the Company's ordinary activities. The provision of providing such computing power is the only performance obligation in the Company's arrangements with mining pool operators. The transaction consideration the Company receives, if any, is variable noncash consideration. In certain arrangements, the consideration becomes fixed to the extent that the Company has provided computing power to the mining pool. The Company measures the fair value of fixed consideration on the date earned, which is not materially different than the fair value at inception of the arrangement or the date received. In certain arrangements, the Company does not have a reliable means to estimate its relative share of the rewards until they are paid to it and the variable consideration is constrained until the Company receives the consideration, at which time revenue is recognized. The Company measures variable consideration at fair value on the date received, which is not materially different than the fair value at inception of the arrangement or the time the Company has earned the award from the pools. The Company's digital asset mining income is sensitive to changes in the market prices of digital currency assets which may be significant.

Digital Assets

The Company's digital asset policy, including for the periods presented in the consolidated financial statements through June 30, 2021, included selling all digital assets and converting them into fiat currency shortly after they are mined, typically within 1-3 days, in order to fund the growth of the Company's operations. Following the Blockcap acquisition, the Company significantly expanded its self-mining operation and consequently reevaluated its digital asset investment policy and began holding a more significant portion of its digital assets mined on its balance sheet.

The Company classifies digital currency assets primarily as Level 1. The Company's digital currency assets are accounted for as intangible assets with indefinite useful lives. The Company initially recognizes digital currency assets that are received as digital asset mining income based on the fair value of the digital currency assets. Digital currency assets that are purchased in an exchange of one digital currency asset for another digital currency asset are recognized at the fair value of the asset surrendered. Impairment exists when the carrying amount exceeds its fair value, which is measured using the quoted price of the digital currency asset at the time its fair value is being measured, which is measured on a daily basis. To the extent that an impairment loss is recognized, the loss establishes the new costs basis of the digital asset. In the nine months ended September 30, 2021 the Company recognized impairments of digital currency assets of \$12.6 million compared to a nominal amount for the nine months ended September 30, 2020. The Company's digital currency assets amounted to

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

\$116.2 million and \$0.1 million as of September 30, 2021 and December 31, 2020. The Company does not have any off-balance sheet holdings of digital currency assets, the details of which are presented below.

(in thousands)	September 30 	nber 31 020
Bitcoin (BTC)	\$ 109,871	\$ 51
Ethereum (ETH)	1,834	_
Siacoin (SC)	945	_
Other	3,583	 12
Total digital currencies	\$ 116,233	\$ 63

Recently Adopted Accounting Standard

Convertible Debt Instruments. In August 2020, the FASB issued ASU No. 2020-06, Debt—Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging—Contracts in Entity's Own Equity (Subtopic 815-40): Accounting for Convertible Instruments and Contracts in an Entity's Own Equity, which amends the accounting for certain financial instruments with characteristics of liabilities and equity, including convertible instruments and contracts in an entity's own equity. This standard can be adopted using either a modified retrospective or a fully retrospective method of transition. This standard is effective for us beginning January 1, 2022, with early adoption permitted. We elected to adopt the standard effective January 1, 2021. Adoption of the standard had no impact on the Company's prior period financial statements.

Accounting Pronouncements Not Yet Adopted

Lease Accounting. In February 2016, the FASB issued ASU No. 2016-02, Leases-(Topic 842). Under this new guidance, lessees will be required to recognize for all leases (with the exception of short-term leases): 1) a lease liability equal to the lessee's obligation to make lease payments arising from a lease, measured on a discounted basis and 2) a right-of-use asset which will represent the lessee's right to use, or control the use of, a specified asset for the lease term. Topic 842 is effective for the Company's annual and interim reporting periods beginning January 1, 2022. The adoption of Topic 842 will require the Company to recognize non-current assets and liabilities for right-of-use assets and operating lease liabilities on its consolidated balance sheet, but it is not expected to have a material effect on the Company's results of operations or cash flows. Topic 842 will also require additional footnote disclosures to the Company's consolidated financial statements.

4. ACQUISITIONS

In June 2020, the Company entered into an Asset Purchase Agreement with Atrio Inc. ("Atrio") to acquire certain assets of Atrio in exchange for \$1.2 million cash consideration and \$0.3 million shares of the Company's common stock. The Company and Atrio contemplated a valuation for the transaction of approximately \$2.4 million based on an estimate of the fair value of the Company's common stock of \$3.50 per share which was allocated to the cost of the acquired software intangible assets.

In a separate transaction in June 2020, the Company entered into an agreement with RStor, Inc. ("RStor") to obtain a non-exclusive license to three specific patents held by RStor for consideration of \$0.3 million shares of the Company's common stock. The Company and RStor contemplated a valuation for the transaction of approximately \$0.9 million based on an estimate of the fair value of the Company's common stock of \$3.50 per share which was allocated to the cost of the acquired patent intangible assets.

The software acquired from Atrio and the acquired patents from RStor are not businesses under ASC 805, Business Combinations, because substantially all the fair value of the acquired assets is concentrated in a

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

single intellectual property asset. Accordingly, the asset purchases are accounted for as asset acquisitions where the cost of the acquisition, measured as the fair value of the cash consideration transferred and the common stock issued by the Company, is allocated to the assets acquired.

In September 2019, the Company acquired certain assets and liabilities of Stax Digital LLC in exchange for 0.6 million shares of the Company's common stock valued at \$3.8 million. The set of assets acquired by the Company did not meet the definition of a business and the transaction was accounted for as an asset acquisition. The Company allocated \$4.0 million of the cost of the asset acquisition to acquired software and recognized assumed liabilities comprising accounts payable of \$0.1 million.

Description of Blockcap Acquisition

On July 30, 2021, the Company acquired 100% of the equity interest in Blockcap, one of its largest hosting customers. Consideration consisted of the issuance of 72.2 million shares of the Company's common stock, approximately 4.3 million shares of the Company's restricted stock and approximately 4.6 million options to purchase shares of the Company's common stock. The acquisition has been accounted for as a business combination using the acquisition method of accounting, whereby the net assets acquired and the liabilities assumed were recorded at fair value. The Company is still in the process of completing its initial accounting for the business combination, including determining the fair value of consideration transferred, assets acquired and liabilities assumed and the allocation of the purchase price to goodwill. The Company and Blockcap had preexisting relationships which were settled on the acquisition date. Using the estimated purchase price for the transaction, the Company has allocated the purchase price to identifiable assets and liabilities based upon preliminary fair value estimates. The excess of the purchase price over the fair value of the net identifiable assets acquired was allocated to as goodwill.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

The following table summarizes the preliminary fair values for each major class of assets acquired and liabilities assumed at the acquisition date. The Company retained the services of certified valuation specialists to assist with assigning estimated values to certain acquired assets and assumed liabilities. The Company is in the process of finalizing the valuation of the assets acquired and liabilities assumed including income tax related amounts. Therefore, the preliminary fair values set forth below are subject to adjustment as additional information is obtained and the valuations are completed.

PPA Allocation

Consideration (in thousands):		
72.2 million common shares valued at \$16.18 per share ^{1,2}	\$ 1,	167,965
Fair value of replaced Blockcap share-based payments attributable to pre-combination service ³		21,941
Settlement of preexisting services contracts ⁴		(60,522)
Total Consideration:	\$ 1,	129,384
Fair value of assets acquired, and liabilities assumed:		
Cash and cash equivalents	\$	704
Digital assets-Bitcoin		73,304
Digital assets-Ethereum		365
Digital assets-Bitcoin cash		8
Digital assets-Siacoin		554
Digital assets-Other		3,329
Other current assets		633
Intangible assets, net		2,925
Property, plant and equipment, net		27,089
Other noncurrent assets		1,293
Total assets acquired		110,204
Accounts payable		492
Accrued expenses and other		22,647
Other current liabilities		6,408
Deferred tax liability		_
Total liabilities assumed	\$	29,547
Total identifiable net assets	\$	80,657
Goodwill on acquisition	\$ 1,0	048,727

^{72.2} million common shares represent the equivalent Core Scientific common shares issued to Blockcap shareholders as consideration for the purchase.

The price per share of our common shares was estimated to be \$16.18. As the Core Scientific common shares were not listed on a public marketplace, the calculation of the fair value of the common shares was subject to a greater degree of estimation. Given the absence of a public market, an estimate of the fair value of the common shares was required at the time of the Blockcap Acquisition. the Company's and subjective factors were considered in determining the estimated fair value and because there is no active trading of the Core Scientific equity shares on an established securities market, an independent valuation specialist was engaged. The valuation was determined by weighting the outcomes of scenarios estimating share value based on both public company valuations and private company valuations. Both a market approach and common stock equivalency model were used to determine a range of outcomes, which were weighted based on probability to determine the result.

³ Reflects the estimated fair value of replaced Blockcap share-based payments allocated to purchase price based on the proportion of service related to the pre-combination services.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

4 Blockcap had preexisting hosting and equipment contracts with the Company that were effectively settled by the Company's acquisition of Blockcap. As a result, the consideration transferred to Blockcap has been adjusted by the deferred revenue balances that were settled at the time of acquisition.

Intangible Assets and Liabilities

Goodwill with a provisionally assigned value of \$1.05 billion represents the excess of the consideration transferred over the estimated fair values of assets acquired and liabilities assumed. The preliminary goodwill recognized includes the assembled workforce of Blockcap and intangible assets that do not qualify for separate recognition. None of the goodwill resulting from the acquisition is deductible for tax purposes. All of the goodwill acquired is allocated to the Mining segment. Management believes the acquisition of Blockcap strengthens its presence in the data mining market due to the scale of its operations. These factors are the basis for the excess purchase price paid over the value of the assets acquired and liabilities assumed, resulting in goodwill.

The following table provides the reconciliation of the carrying amount of goodwill at the beginning and end of the reporting period:

(in thousands)	(Goodwill
Balance as of December 31, 2020	\$	58,241
Acquisitions		1,048,727
Balance as of September 30, 2021	\$	1,106,968

Other intangible assets include \$2.8 million developed technology intangibles and 0.1 million of customer relationships with a weighted-average useful life of 3 years.

Transaction Costs

The Company recognized transaction costs of \$1.1 million for the nine months ended September 30, 2021, respectively. These costs were associated with legal and professional services and were recognized as General and administrative expenses in the Company's Consolidated Statements of Operations and Comprehensive Loss.

Pro Forma Information

The following unaudited pro forma financial information gives effect to the Transactions as if they had been completed on January 1, 2020. The unaudited pro forma information was prepared in accordance with the requirements of ASC 805, *Business Combinations*, which is a different basis than pro forma information prepared under Article 11 of Regulation S-X ("Article 11"). As such, they are not directly comparable with historical results for stand-alone Core Scientific prior to July 30, 2021, historical results for Core Scientific from July 30, 2021 that reflect the acquisition and are inclusive of the results and operations of Blockcap, nor our previously provided pro forma financials prepared in accordance with Article 11. The pro forma results for the nine months ended September 30, 2021 and 2020 include the impact of several significant nonrecurring pro forma adjustments to previously reported operating results. The pro forma adjustments are based on historically reported transactions by the respective companies. The pro forma results do not include any anticipated synergies or other expected benefits of the acquisition.

		September 30,	
(in thousands)	2021	2020	
Total revenues	\$ 285,196	\$ 38,645	
Operating income (loss)	\$ 34,951	\$ (4,024)	

Nine Months Ended

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

Significant nonrecurring pro forma adjustments include:

- Transaction costs of \$1.9 million are assumed to have occurred on the pro forma close date of January 1, 2020, and are recognized as if
 incurred in the first quarter of 2020;
- Tangible and intangible assets are assumed to be recorded at their estimated fair values as of January 1, 2020 and are depreciated or amortized over their estimated useful lives; and
- Accounting policies of Blockcap are conformed to those of Core Scientific including depreciation for mining equipment.
- Share-based compensation awards of Blockcap for which the performance condition of the award is assumed to be probable of being met
 as of January 1, 2020 and expensed as they are earned based on the service condition.
- The elimination of \$19,239 of expense recognized by Blockcap in July 2021 for the acceleration of certain equity awards of its CEO and others. Because this acceleration was deemed to be in contemplation of the Merger, Core Scientific has recorded \$23,294 of compensation expense for the acceleration in its financial statements for the period ending September 30, 2021, which was determined based on the fair value of the awards at the time of the Merger. This adjustment is necessary to avoid duplication of the expense attributable to the combined company related to the acceleration of the same awards.

The selected unaudited pro forma condensed combined financial information is provided for illustrative purposes only and does not purport to represent what the actual consolidated results of operations would have been had the acquisition actually occurred on January 1, 2020, nor do they purport to project the future consolidated results of operations.

For the periods subsequent to the acquisition, Blockcap contributed total revenues of \$10.8 million to the nine months ended September 30, 2021 and operating loss of (\$0.3) million for the nine months ended September 30, 2021 that were included in our Consolidated Statements of Operations and Comprehensive Income (Loss).

5. NOTES PAYABLE

(in thousands)	September 30 2021	December 31 2020
Georgia note	\$ 430	\$ 581
Kentucky note	1,154	1,511
PPP loan	-	2,154
Silverpeak loan	_	22,260
Stockholder loan	10,000	_
Genesis loan	2,015	4,648
Celsius loan	_	6,842
NYDIG loan	40,510	718
Trinity loan	1,000	_
Convertible notes	420,925	_
Other	304	_
Total	476,338	38,714
Unamortized discount and debt issuance costs	(1,115)	(2,834)
Fair value adjustments to convertible notes	21,909	` <u> </u>
Total notes payable, net	\$ 497,132	\$ 35,880

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

Georgia Note—In September 2018, the Company entered into a five-year secured purchase note payable agreement for \$1.0 million in connection with the acquisition of property in Georgia for datacenter development ("Georgia note"). The note bears interest at a rate per annum of 5% and the Company is required to make monthly payments of principal and interest. Interest expense on the notes has been recognized based on an effective interest rate of 5%.

Kentucky Note—In December 2018, the Company entered into a five-year secured promissory note agreement for \$2.4 million in connection with the acquisition of property in Kentucky for datacenter development ("Kentucky note"). The note bears interest at a rate per annum of 5% and the Company is required to make monthly payments of principal and interest. Interest expense on the notes has been recognized based on an effective interest rate of 5%.

Executive Notes—In March and April 2020, the Company issued an aggregate of \$4.0 million of notes to the Company's president and chief executive officer and a member of the Board of Directors. The notes are unsecured and bear interest at 7% per annum, with all principal and interest accrued due at the earlier of 12 months from issuance or the occurrence of a financing of at least \$20.0 million of gross proceeds. The Company granted common stock warrants to the noteholders for the purchase of 4,000,000 shares of common stock at an exercise price of \$1.00 per share. The exercise price was subsequently amended to \$1.34 per share. The Company allocated proceeds of \$2.2 million to the notes and \$1.6 million to the warrants on a relative fair value basis and recognized a derivative liability of \$0.2 million equal to the fair value of the contingent repayment feature. Interest expense on the notes was recognized based on an effective interest rate of 98%. The notes were repaid in full in May 2020 from the proceeds of the Silverpeak Loan described below.

PPP Loan—In April 2020, the Company received a loan of \$2.2 million from the PPP through the SBA ("PPP note"). The loan is unsecured and bears interest at a rate per annum of 1% and is paid monthly. Payments of principal are scheduled to begin in July 2021 and to be paid monthly until April 2022 when it would be fully paid off. Interest expense on the note has been recognized based on an effective interest rate of 1%. In July 2021, the Company paid off the \$2.1 million PPP note in full plus accrued interest.

Silverpeak Loan—In May 2020, the Company issued a \$21.0 million senior secured loan to Silverpeak Credit Partners LP. The loan is secured by substantially all of the assets of the Company. The loan bears interest at a rate per annum of 15%, interest payments due monthly. Principal payments are due on a quarterly basis from June 2021 through June 2022 in the amount equal to the lesser of \$1.0 million or 30% of a measure of excess cash flow defined in the loan agreement. From September 2022 through March 2023, principal payments are due on a quarterly basis in the amount equal to the lesser of \$1.0 million or 40% of the measure of excess cash flow. All remaining outstanding principal is due in May 2023. The loan requires the Company to maintain unrestricted cash on hand of at least \$3.8 million throughout the term of the loan. The Company was in compliance with this requirement as of September 30, 2021 and December 31, 2020. Beginning in September 2021, the Company may not exceed specified limits of the Company's ratio of indebtedness to a measure of the Company's earnings before interest, taxes, and depreciation and amortization ("EBITDA") and the Company's ratio of EBITDA to fixed charges for debt service, maintenance of the Company's facilities, and income taxes. The Company received proceeds from the loan issuance of \$19.8 million, which are net of an original issuance discount and issuance costs of \$1.2 million. In addition, the issuance costs also included common stock options issued to advisors with an exercise price of \$1.00 per share and a fair value of \$0.8 million on the issuance date. The loan also included a non-refundable exit fee of \$1.3 million due upon maturity or early payment which was included as part of the debt discount. Interest expense on the loan has been recognized based on an effective interest rate of 21%, which includes the amortization of the discount and debt issuance costs. In May 2020, the Company used \$4.0 million of the proceeds from the loan issuance to repay the outstanding principal and interest on the promissory notes to the Company's president and chief executive officer and a member of the Board of Directors, as discussed above. In February 2021, the Company entered in an amendment of the Silverpeak loan

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

that included the issuance of an additional \$9.0 million tranche of senior secured loans. This loan, including the additional tranche issued in February 2021, was paid off in full in April 2021.

Genesis Loan—In July 2020, the Company entered into a credit facility with Genesis Global Capital, LLC that provides capacity of up to \$13.0 million to finance the Company's acquisition of blockchain computing equipment ("Genesis Loan"). The Company borrowed \$5.3 million in three installments and the borrowing capacity of the facility was reduced via an amendment in September 2020 to equal the actual amounts borrowed. The loans under the credit facility are secured by the blockchain computing equipment and the Company is required to comply with an approved mining strategy and other restrictions on use of the collateral. Loans under the credit facility have terms of 20 months, bear interest at a rate per annum of 16% plus a fixed risk premium, and require monthly payments. Interest expense on the loans have been recognized based on an effective interest rate of 28%, which includes the amortization of a debt discount.

NYDIG Loans—In October 2020, the Company entered into a master equipment finance agreement with NYDIG and received a loan of \$0.8 million to finance the Company's acquisition of blockchain computing equipment. In March 2021, the Company received \$3.8 million of additional loans under the master equipment finance agreement with NYDIG to finance the Company's acquisition of blockchain computing equipment. The loans bear an interest rate of 15% and have a term of 24 months from issuance. Interest expense on the loans has been recognized based on an effective interest rate of 15%

In May 2021, the Company received \$13.4 million of additional loans under the master equipment finance agreement with NYDIG to finance the Company's acquisition of blockchain computing equipment that bear an interest rate of 14.25% and have a term of 24 months from issuance. Interest expense on the loans issued in May 2021 has been recognized based on an effective interest rate of 16%.

In July 2021, the Company received blockchain computing equipment from NYDIG in exchange for \$25.6 million of additional loans under the master equipment finance agreement with NYDIG that bear an interest rate of 14.25% and have a term of 24 months from issuance. Interest expense on the loans issued in July 2021 has been recognized based on an effective interest rate of 16%.

Trinity Loans—In August 2021, the Company entered into a \$30.0 million master equipment finance facility agreement with Trinity Capital Inc. ("Trinity") to finance the Company's acquisition of blockchain computing equipment and received a loan of \$1.0 million at close. The loan has a term of 36 months from issuance. Interest expense on the loan has been recognized based on an effective interest rate of 10.325%.

The remaining balance of \$29.0 million can be drawn, at the Company's option, no later than twelve months after the initial close. An additional \$14.0 million of this facility may be drawn prior to raising at least \$300.0 million in qualified funding transactions and the remaining \$15.0 million may be drawn after raising at least \$300.0 million in qualified funding transactions.

Celsius Loan—In November 2020 the Company entered into an equipment lending agreement with Celsius Networks Lending, LLC to borrow up to \$27.7 million in five tranches through May 20, 2021 for the purchase of blockchain mining equipment. The Company borrowed the first two tranches totaling \$13.8 million in November and December 2020. The loans bear interest at 13.5% annually and are due at the earlier of the date of sale of the underlying mining equipment or 24 months from issuance. Interest expense on the loans has been recognized based on an effective interest rate of 13.5%. In December 2020, the Company repaid \$7.0 million of the loan and in January 2021, the Company repaid the remaining balance of \$6.8 million. The Company recognized a loss of \$42 upon loan extinguishment related to unamortized debt issuance costs.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

Stockholder loan—In January 2021, the Company borrowed \$10.0 million from a stockholder for the purchase of bitcoin mining equipment. The loan bears interest at 10% per annum over a two-year term. The loan was issued with a warrant to purchase 120,000 shares of common stock at an exercise price of \$6.74 per share. The warrant has a two-year term. The Company allocated proceeds of \$9.5 million to the notes and \$0.5 million to the warrants on a relative fair value basis. Interest expense on the loan has been recognized based on an effective interest rate of 13%.

Convertible Notes—In April 2021, the Company entered into a Secured Convertible Note Purchase Agreement and issued \$215.0 million of Secured Convertible Notes to new and existing lenders. In August and September 2021, the Company issued an additional \$203.3 million of Unsecured Convertible Notes under substantially the same terms and conditions as the original April 2021 notes except that the August and September 2021 notes are unsecured until an IPO or SPAC merger and then they are secured pari passu with the Secured Convertible Notes. Payment in Kind (PIK) interest is accrued quarterly by increasing the outstanding principal amount of the notes. As a result, \$2.6 million during the nine months ended September 30, 2021. The convertible notes have a maturity date of April 2025 and bear interest at a rate of 10% per annum, of which 4% is payable in cash and 6% is payable in kind. The convertible notes are convertible into common or preferred shares at the option of the holder upon the occurrence of a) an IPO or SPAC merger with gross proceeds to the Company of at least \$50.0 million or b) a change in control. The conversion price is equal to a) 65-80% of the price paid for equity securities in an IPO, depending on the timing of the IPO or b) 65-80% of the fair value per share paid in a change in control or SPAC merger, depending on the timing of the change in control or SPAC merger. At maturity, the Secured Convertible Notes not converted will be owed two times the face value of the note plus accrued interest. The proceeds from the convertible notes were used, in part, to repay the Silverpeak loan described above. As discussed in Note 2, the Company has elected to measure its convertible notes at fair value and accordingly during the nine months ended September 30, 2021 recognized \$10.7 million of debt issuance costs as incurred within Interest Expense, Net in the Company's Consolidated Statements of Operations and Comprehensive loss. See Note 3 for further information on fair value measurement of the convertibl

As of September 30, 2021 and December 31, 2020, the Company is not in violation of any of its debt covenants.

6. CONTINGENTLY REDEEMABLE CONVERTIBLE PREFERRED STOCK

At various dates during the year ended December 31, 2019, the Company issued a total of 4,421,000 shares of \$0.00001 par value Series A Contingently Redeemable Convertible Preferred Stock ("Series A Preferred Stock") at \$6.83 per share in private placement offerings. The aggregate proceeds from the offerings were \$29.5 million, which were net of offering costs of \$0.7 million.

In February and March 2020, the Company issued a total of 229,000 shares of Series A Preferred Stock at \$6.83 per share in private placement offerings. The aggregate net proceeds from the offerings were \$1.5 million.

In February 2020, the Company completed an exchange of 1.1 million shares of common stock that were originally issued in the private placement offering from October 2018 to December 2019 for 1,802,000 newly issued shares of Series A Preferred Stock. The shares of common stock were retired upon reacquisition by the Company. The Company received no net proceeds from the exchange and recognized a deemed dividend of \$10.5 million based on the incremental fair value of the preferred stock received by the stockholders compared to the fair value of the common stock exchanged. The difference between the fair value of the Preferred Stock issued and common stock retired is accounted for as a deemed dividend and is reflected as an increase to the amount of the net loss attributable to common shareholders.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

In July 2020, the Company issued 314,000 shares of \$0.00001 par value Series B Contingently Redeemable Convertible Preferred Stock ("Series B Preferred Stock") at \$3.50 per share in a private placement offering. The aggregate net proceeds from the offering was \$1.1 million.

Each share of Series A and Series B Preferred Stock has a liquidation value of \$6.83 and \$3.50, respectively. Holders may elect at any time to convert each share of preferred stock into one share of the Company's common stock. Unless earlier converted, each share of preferred stock will automatically convert to shares of the Company's common stock upon the earlier of an initial public offering of the Company's common stock for aggregate proceeds of not less than \$100.0 million, or an agreement to convert by the holders of a majority of the outstanding shares of preferred stock. In the event of an initial public offering, the number of shares of the Company's common stock issuable on conversion of each share of preferred stock will be either (i) the amount determined by dividing the liquidation value by the public offering price if such offering price is less than the liquidation value per share, or (ii) one share. In the event of an agreement to convert by the holders of a majority of the outstanding shares of preferred stock, one share of the Company's common stock is issuable on conversion of each share of preferred stock.

In addition to the conversion rights, the Series A and B Preferred Stock is redeemable upon the occurrence of a deemed liquidation event, the definition of which includes a transaction that results in a change in control of the Company or a sale or transfer by the Company of substantially all its assets. In the event of a deemed liquidation, the holders of shares of Preferred Stock are entitled to a cash distribution equal to the liquidation value per share. As of September 30, 2021 and December 31, 2020 the Series A and B preferred stock was not redeemable because no deemed liquidation event had occurred. Furthermore, the Company has not made any adjustments to the carrying value of the Series A and B Preferred Stock to reflect the liquidation value of the shares because the Company has determined that a deemed liquidation event is not probable of occurring.

All outstanding shares of Series A and B Preferred Stock have one vote per share. Dividends on Series A and Series B Preferred Stock are payable when, as, and if declared by the Company's Board of Directors. Series A and B Preferred Stock rank in parity to each other and rank senior to the Company's common stock, including capital stock established in the future, unless the terms of such capital stock expressly provide otherwise, and junior to existing and future indebtedness and other liabilities with respect to dividend rights and distribution rights upon liquidation, winding-up, and dissolution.

The Series A and B Preferred Stock is classified as contingently redeemable because it is redeemable upon the occurrence of a deemed liquidation event, and such a redemption triggering event is not solely within the control of the Company. The Series A and B Preferred Stock is therefore presented separate from and outside of stockholders' equity in a manner consistent with temporary equity.

7. COMMITMENTS AND CONTINGENCIES

Legal Proceedings—The Company is subject to legal proceedings arising in the ordinary course of business. The Company accrues losses for a legal proceeding when it is probable that a loss has been incurred and the amount of the loss can be reasonably estimated. However, the uncertainties inherent in legal proceedings make it difficult to reasonably estimate the costs and effects of resolving these matters. Accordingly, actual costs incurred may differ materially from amounts accrued and could materially adversely affect the Company's business, cash flows, results of operations, financial condition and prospects. Unless otherwise indicated, the Company is unable to estimate reasonably possible losses in excess of any amounts accrued. As of September 30, 2021 and December 31, 2020, there were no loss contingency accruals for legal matters.

Operating Leases—The Company leases its corporate headquarters under an operating lease agreement that expires in 2022. The Company leases other facilities under operating leases that have terms of less than one

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

year. Certain lease agreements provide for rental payments on a graduated basis. The Company recognizes rent expense on a straight-line basis over the lease period. For the nine months ended September 30, 2021, rent expense was \$0.4 million. For the nine months ended September 30, 2020, rent expense was \$0.3 million.

The lease agreements for certain of the Company's facilities require the Company to maintain cash deposits with the landlords of each facility as security for its payment obligations under the leases. As of September 30, 2021 and December 31, 2020, the Company had nominal deposits related to such leases.

Capital Leases—The Company has entered into arrangements with various parties to finance the acquisition of computer and networking equipment, electrical infrastructure, and office equipment. These arrangements include options exercisable by the Company at the end of the initial terms to renew, to purchase the equipment, or to terminate. These arrangements are capital leases and as of September 30, 2021 and December 31, 2020, the related capital lease obligations were \$4.0 million and \$4.4 million, respectively.

The depreciation of property, plant and equipment under capital leases for the nine months ended September 30, 2021 was \$1.5 million. The depreciation of property, plant and equipment under capital leases for the nine months ended September 30, 2020 was \$1.1 million.

Purchase obligations—During the nine months ended September 30, 2021, the Company entered into agreements to purchase blockchain mining equipment totaling approximately \$738.3 million (net of discounts of \$27.2 million) of which approximately \$418.5 million was paid for deliveries or as deposits for blockchain mining equipment scheduled to be delivered in 2021 and 2022. As of the date that the financial statements were available to be issued, the aggregate amount of the Company's purchase obligations totaled approximately \$319.9 million, substantially all of which are expected to be settled within one year of the date that the financial statements were available to be issued.

(Loss) gain on legal settlements—The Company recognized a loss of \$2.6 million during the nine months ended September 30, 2021 related to a settlement with a former customer. The Company recognized a gain of \$5.8 million during the nine months ended September 30, 2020 with respect to the resolution of legal actions for damages resulting from the early termination of agreements by former customers.

8. INCOME TAXES

Current income tax expense represents the amount expected to be reported on the Company's income tax returns, and deferred tax expense or benefit represents the change in net deferred tax assets and liabilities. Deferred tax assets and liabilities are determined based on the difference between the financial statement and tax basis of assets and liabilities as measured by the enacted tax rates that will be in effect when these differences reverse. Valuation allowances are recorded as appropriate to reduce deferred tax assets to the amount considered likely to be realized.

For the nine months ended September 30, 2021, the Company had an income tax benefit of \$0.7 million resulting in an effective income tax rate of 5.0%. The Company's estimated annual effective income tax rate of 5.0% is lower than the US federal statutory rate of 21.0%, primarily due to the valuation allowance on the current year loss. There was no income tax impact for the nine months ended September 30, 2020.

9. NET LOSS ATTRIBUTABLE TO COMMON STOCKHOLDERS

Basic EPS is measured as the income or loss available to common stockholders divided by the weighted average common shares outstanding for the period. Diluted EPS presents the dilutive effect on a per-share basis

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

from the potential conversion of convertible securities or the exercise of options and or warrants; the dilutive impacts of potentially convertible securities are calculated using the if-converted method; the potentially dilutive effect of options or warranties are computed using the treasury stock method. Securities that are potentially anti-dilutive (i.e., those that increase income per share or decrease loss per share) are excluded from diluted EPS calculation.

		Nine Months Ended September 30,	
	2021	2020	
Net loss	\$(13,194)	(9,897)	
Deemed dividend from common to preferred exchange		(10,478)	
Net loss attributable to common stockholders	\$(13,194)	5(20,375)	
Weighted average shares outstanding - basic	115,482	98,453	
Add: Dilutive share-based compensation awards	<u> </u>		
Weighted average shares outstanding - diluted	115,482	98,453	
Net income (loss) per share - basic	\$ (0.11) \$	(0.21)	
Net income (loss) per share - diluted	\$ (0.11) \$	(0.21)	

Potentially dilutive securities that are not included in the calculation of diluted net loss per share for the nine months ended September 30, 2021 and 2020 because to do so would be anti-dilutive or for certain share-based compensation awards, because the shares are subject to performance conditions that had not been satisfied by the end of the period are as follows (in common stock equivalent shares):

		Nine Months Ended September 30,	
	2021	2020	
Stock options	7,320	7,320	
Preferred stock	6,766	6,766	
Warrants	4,255	4,255	
Restricted stock	50,901	37,512	
Total potentially dilutive shares	69,242	55,853	

2021 Employee Grants

During the nine months ended September 30, 2021, the Company granted 16.0 million restricted stock units to employees with a weighted-average grant-date fair value of \$17.93 per share.

Resignation of CEO

The Company's President and Chief Executive Officer Mr. Kevin Turner resigned in May 2021. Mr. Turner's employment agreement provides that he will be entitled to receive 12 months of base salary totaling \$300,000 and, if Mr. Turner elects COBRA coverage, for 12 months he will receive a monthly cash payment equal to the full premium for company-paid continuing health care benefits. As of Mr. Turner's separation date, 9,037,500 RSUs had satisfied the time vested condition but remain subject to the performance condition described in Note 11 of the Company's Consolidated Financial Statements as of and for the years ended December 31, 2020 and 2019.

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

10. SEGMENT REPORTING

The Company's operating segments are aggregated into reportable segments only if they exhibit similar economic characteristics and have similar business activities.

The Company has two reportable segments: "Equipment Sales and Hosting" which aggregates the results of its Blockchain Hosting and Equipment Sales operating segments and consists primarily of its blockchain infrastructure and third-party hosting business and equipment sales, and "Mining" consisting of digital asset mining for its own account. The blockchain hosting business generates revenue through the sale of consumption-based contracts for its hosting services which are recurring in nature. Equipment sales revenue is derived from its ability to leverage its partnership with leading equipment manufacturers to secure equipment in advance, which is then sold to its customers. The digital asset mining operation segment generates revenue from operating owned computer equipment as part of a pool of users that process transactions conducted on one or more blockchain networks. In exchange for these services, the Company receives digital currency assets.

The primary financial measures used by the CODM to evaluate performance and allocate resources are revenue and gross profit. The CODM does not evaluate performance or allocate resources based on segment asset or liability information; accordingly, the Company has not presented a measure of assets by segment. The segments' accounting policies are the same as those described in the summary of significant accounting policies. The Company excludes certain operating expenses and other (income) expense from the allocations to operating segments. The following table presents revenue and gross profit by reportable segment for the periods presented:

		Nine Months Ended September 30,	
	2021	2020	
Equipment Sales and Hosting Segment			
Hosting revenue from customers	\$ 51,742	\$ 32,049	
Equipment sales to customers	113,435	2,272	
Total revenue	165,177	34,321	
Cost of revenue	131,284	30,308	
Gross profit	\$ 33,893	\$ 4,013	
Mining Segment			
Digital asset mining income	\$ 77,511	\$ 2,312	
Total revenue	77,511	2,312	
Cost of revenue	13,909	1,598	
Gross profit	\$ 63,602	\$ 714	
Consolidated total revenue	\$ 242,688	\$ 36,633	
Consolidated cost of revenue	\$ 145,193	\$31,906	
Consolidated gross profit	\$ 97,495	\$ 4,727	

For the nine months ended September 30, 2021, cost of revenue included depreciation expense of \$5.2 million for the Equipment Sales and Hosting segment and \$6.2 million for the Mining segment. For the nine months ended September 30, 2020, cost of revenue included depreciation expense of \$5.6 million for the Equipment Sales and Hosting segment \$0.4 million for the Mining segment.

For the nine months ended September 30, 2021 and 2020, the top three customers accounted for approximately 73% and 64%, respectively, of the Equipment Sales and Hosting's segment total revenue. For the

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

nine months ended September 30, 2021 and 2020, the concentration of customers comprising 10% or more of the Company's revenue was as follows:

		Nine Months Ended September 30,	
	2021	2020	
Customer			
A	344	% N/A	
В	219	% N/A	
Blockcap	189	% N/A	
C	N/A	42%	
D	N/A	16%	

A reconciliation of the reportable segment gross profit to loss before income taxes included in our consolidated statements of operations for the nine months ended September 30, 2021 and 2020 is as follows:

		Nine Months Ended September 30,	
	2021	2020	
Gross profit	97,495	4,727	
(Loss) gain on legal settlements	(2,603)	5,814	
Gain from sales of digital currency assets	405	52	
Impairment of digital currency assets	(12,552)	(5)	
Operating expenses:			
Research and development	4,231	4,183	
Sales and marketing	2,186	1,401	
General and administrative	46,992	10,797	
Total operating expenses	53,409	16,381	
Operating income (loss)	29,336	(5,793)	
Non-operating expenses, net:			
Loss on debt from extinguishment	8,016	1,333	
Interest expense, net	26,550	2,683	
Other non-operating (income) expense, net	8,661	88	
Total non-operating expense, net	43,227	4,104	
Income (loss) before income taxes	\$(13.891)	(9.897)	

11. RELATED-PARTY TRANSACTIONS

In the ordinary course of business, the Company enters into various transactions with related parties.

In October 2018, the Company entered into an operating lease with a lessor entity controlled by an executive of the Company for an office facility in Ohio. The Company may terminate the lease at any time with 30 days' notice to the landlord. The lease was terminated in April 2020. There was no rent expense for the nine months ended September 30, 2021. For the nine months ended September 30, 2020, the Company incurred rent expense of \$4 thousand related to the lease.

During the year ended December 31, 2019, the Company entered into agreements to provide hosting services to various entities that are managed by or invested in individuals that are directors and executives of the

Core Scientific Holding Co. Notes to Unaudited Consolidated Financial Statements

Company. For the nine months ended September 30, 2021 and 2020, the Company recognized hosting revenue from the contracts with these entities of and \$13.9 million and \$3.4 million, respectively. In addition, the Company recognized Equipment Sales revenue of \$29.1 million and \$0.3 million for the nine months ended September 30, 2021 and 2020, respectively. As of September 30, 2021 and December 31, 2020, the Company had accounts receivable of \$0.3 million from these entities.

During the nine months ended September 30, 2021, the Company entered various promissory notes with Blockcap, a related party entity that was managed by individuals that are directors and executives of the Company. The Company had existing contracts for equipment sales and hosting services with Blockcap prior to the Company acquiring Blockcap on July 30, 2021 as described above. The promissory notes deferred \$32.7 million of amounts originally due in June through July 2021 from Blockcap contracts until August 2021.

The Company reimburses its president and chief executive officer for use of a personal aircraft for flights taken on Company business. For the nine months ended September 30, 2021, the Company incurred reimbursements of \$0.6 million, respectively. For the nine months ended September 30, 2020, the Company incurred reimbursements of \$0.1 million. There was \$0.1 million payable at September 30, 2021 and no amount was payable December 31, 2020.

12. SUBSEQUENT EVENTS

The Company has evaluated subsequent events through December 29, 2021 the date that the financial statements were available to be issued.

Financing activities

In October and November 2021, the Company issued an additional \$96.5 million of convertible notes subject to the same terms and conditions of the Unsecured Convertible Notes described in Note 3 and Note 5 above.

In October 2021, the Company entered into an arrangement to finance the acquisition of computer and networking equipment with a lease term of 3 years and fixed payments over the non-cancelable lease term of \$0.5 million.

In October 2021, the Company entered into a financing arrangement that includes a \$7.0 million loan for the purchase of hosting and mining equipment and a \$9.0 million loan for the purchase of equipment and to finance improvement of a server/data center and surrounding energy infrastructure facility.

Employee RSU grants

In October 2021, the Company granted 9.8 million RSUs and 13.0 million stock options to various employees and directors of the Company.

Amendment to Merger Agreement

In December 2021, the Company entered into an Amendment No. 2 to the Agreement with the XPDI Parties to remove the Second Merger Sub (as defined therein) as a party and provide instead that the Company, having survived the First Merger as a wholly owned subsidiary of XPDI, will merge with and into XPDI as soon as possible on the day after the date on which the First Merger is consummated, with XPDI continuing as the surviving entity.

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Shareholders and Board of Directors of Blockcap, Inc.

Opinion on the Financial Statements

We have audited the accompanying balance sheets of Blockcap, Inc. (the "Company") as of December 31, 2020 and 2019, the related statements of operations, stockholders' equity (deficit) and cash flows for the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2020 and 2019, and the results of its operations and its cash flows for the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019, in conformity with accounting principles generally accepted in the United States of America.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audits. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audits we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

/s/ Marcum LLP

Marcum LLP

We served as the Company's auditor from 2020 to 2021.

Los Angeles, CA July 30, 2021

Blockcap, Inc. Balance Sheets As of December 31,

	2020	2019
Assets		
Current assets		
Cash	\$ 8,756,697	_
Digital assets	2,404,047	_
Digital assets receivable	50,979	_
Prepaid expenses	1,664,186	
Total current assets	12,875,909	<u></u> - -
Deposits on mining equipment assets	5,087,685	_
Mining equipment assets, net	16,715,171	_
Goodwill	21,171,370	_
Total assets	\$55,850,135	<u>_</u>
Liabilities and stockholders' equity		
Current liabilities		
Accounts payable and accrued liabilities	\$ 60,556	\$ 609
Loans payable	7,992,455	_
Due to related parties	41,884	_
Common shares issuable	6,204,202	_
Taxes payable	115,564	
Total current liabilities	14,414,661	609
Long term liabilities		
Deferred tax liability	1,125,495	_
Total liabilities	15,540,156	609
Commitments and contingencies (Note 15)		
Stockholders' equity (deficit)		
Common stock, \$0.00001 par value; 400,000,000 shares authorized; 121,909,000 and nil shares issued and outstanding		
as of December 31, 2020 and 2019, respectively	1,219	_
Additional paid-in capital	50,246,144	_
Accumulated deficit	(9,937,384)	(609)
Total stockholders' equity (deficit)	40,309,979	(609)
Total liabilities and stockholders' equity	\$55,850,135	

The accompanying notes are an integral part of these financial statements.

Blockcap, Inc. Statements of Operations For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

	2020	2019
Revenue		
Digital assets mined	\$ 5,972,142	_
Cost of Revenue		
Hosting costs	2,128,211	_
Depreciation	652,765	
Total cost of revenue	2,780,976	
Gross profit	3,191,166	_
Operating expenses		
Share-based compensation	1,992,000	_
Management and professional fees	104,162	_
Office and administration	125,028	609
Total operating expenses	2,221,190	609
Operating income (loss)	969,976	(609)
Other expenses		
Share-based merger expense	10,002,974	_
Interest expense	419,121	_
Gain on sale of digital assets	(53,518)	
Total other expenses, net	10,368,577	_
Net loss before tax	(9,398,601)	(609)
Income tax expense	\$ 538,174	_
Net loss	\$ (9,936,775)	\$(609)
Basic and diluted net loss per share	(\$0.46)	\$ nil
Weighted average number of shares outstanding:	<u> </u>	
Basic and diluted	21,587,741	_

The accompanying notes are an integral part of these financial statements.

Blockcap, Inc. Statement of Stockholders' Equity (Deficit) For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

	Number of shares	Comm		Additional paid in capital	Accumulated deficit	Total
February 19, 2019	_	_	_	_	_	_
Net loss	_	_	_	_	\$ (609)	\$ (609)
December 31, 2019	_	-	_	_	\$ (609)	\$ (609)
Common shares issued in connection with reverse merger (note 3)	43,688,944	\$ 4	37 \$	12,718,245		12,718,682
Common shares issued in connection with Business Combinations						
(note 4)	71,037,695	7	10	34,097,596	_	34,098,306
Common shares issued in connection with asset acquisition (note 5)	7,182,361		72	3,430,303	_	3,430,375
Net loss	_	_	_		(9,936,775)	(9,936,775)
December 31, 2020	121,909,000	\$ 1,2	19 \$	50,246,144	\$ (9,937,384)	\$ 40,309,979

The accompanying notes are an integral part of these financial statements.

Blockcap, Inc. Statements of Cash Flows For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

	2020	2019
Operating activities Net loss	\$ (9,936,775)	\$ (609)
Adjustments to reconcile net loss to net cash used in operating activities Digital assets mined	(5,972,142)	\$ (009) —
Gain on sale of digital assets	(53,518)	_
Depreciation	652,765	_
Income taxes	538,174	
Share-based compensation	1,992,000	
Share-based merger expense	10,002,974	
Digital assets receivable	(28,579)	
Changes in operating assets and liabilities	(20,377)	
Prepaid expenses	(246,332)	_
Accounts payable and accrued liabilities	(217,325)	609
Net cash used in operating activities	(3,268,758)	
	(3,208,738)	_
Investing activities	(2.0(4.216)	
Purchase of mining equipment assets	(3,064,216)	_
Deposits made on mining equipment assets	(2,364,135)	
Sales of digital assets	4,615,466	
Advances from related parties	436,823	
Advances to related parties	(498,613)	_
Cash received on acquisition of RME Black 100 (Note 4(a))	266,490	_
Cash received on acquisition of RME Black 200 (Note 4(b))	152,264	_
Cash received on acquisition of BEP 888 (Note 4(c))	794,445	_
Cash received on acquisition of BEP 999 (Note 5)	1,150,000	_
Proceeds from BEP 999 subscriptions receivable	2,450,000	
Net cash provided by investing activities	3,938,524	_
Financing activities		
Founders' initial contributions	723,921	_
Cash received for common shares issuable	6,204,202	
Repayments of loan payable	(1,338,721)	_
Proceeds from loan payable	2,497,529	_
Net cash provided by financing activities	8,086,931	
Net increase in cash	8,756,697	
Cash, beginning of year	_	_
Cash, end of year	\$ 8,756,697	<u> </u>
Supplemental disclosure of cash flow information		
Cash paid for interest		\$ 415,686
Cash paid for taxes		\$ —
Cush paid for taxes		Ψ

The accompanying notes are an integral part of these financial statements

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

1. Organization and description of business

Blockcap, Inc. (the "Company" or "Blockcap") was incorporated under the laws of Nevada on November 25, 2020. On December 1, 2020, the Company entered into assignment and assumption agreements with RME Black 88, LLC ("RME 88"), RME Black 100 LLC ("RME 100"), and RME Black 200 ("RME 200"), BEP 888 LLC ("BEP 888"), and BEP 999 LLC ("BEP 999") (collectively the "LLCs").

Financial Accounting Standards Board ("FASB") Accounting Standards Codification ("ASC") 805, *Business Combinations*, states that if a new entity is formed to issue equity to affect a business combination one of the combining entities that existed before the business combination shall be identified as the acquirer.

An assessment was made of the respective net assets of the LLCs, the operating history of the LLCs, the subsequent voting rights retained by the various LLC investors, the composition of the new board of directors, and the composition of the new executive management was completed immediately following the completion of the business combinations. Based on the results of the above factors it was determined that RME 88 was the accounting acquirer.

Through the agreement (the "Agreement") between Blockcap and RME 88, Blockcap issued 43,688,944 common shares to the shareholders of RME 88, resulting in a reverse merger with Blockcap (the "Transaction") (see note 3). As a result, the Company's financial statements present the results of RME 88 from the date of its inception, February 19, 2019, to the date of the Transaction.

The assignment and assumption agreements with RME 100, RME 200, BEP 888, were recorded as business combinations (see note 4) and the assignment and assumption agreement with BEP 999 was recorded as a share-based equity-settled asset acquisition (see note 5).

Blockcap is in the business of utilizing specialized "mining" equipment to solve complex computational problems to validate transactions on the bitcoin blockchain. It receives bitcoin in return for successful service. Blockcap holds the rewards it received in its e-wallet and converts them to fiat money as cash is needed for working capital needs. All mining equipment purchased to date has been co-located in specially designed "hosting" facilities owned and operated by Core Scientific, Inc ("Core Scientific"). The registered office of the Company is located at 101 Convention Center Drive, Suite 700 Las Vegas, Nevada, 89109 and the headquarters is located at 200 E 6th street, Austin, Texas, 78701.

2. Summary of significant accounting policies

a) Basis of presentation

The accompanying financial statements and notes of the Company have been prepared in accordance with generally accepted accounting principles in the United States of America ("US GAAP").

The financial statements include the accounts of Blockcap, RME 88, RME 100, RME 200, BEP 888, and BEP 999. All significant intercompany accounts and transactions have been eliminated in consolidation.

b) COVID-19

The COVID-19 global pandemic has been unpredictable and unprecedented and is likely to continue to result in significant national and global economic disruption, which may adversely affect our business.

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

Based on the Company's current assessment, however, the Company does not expect any material impact on its long-term development, its operations, or its liquidity due to the worldwide spread of COVID-19. However, the Company is actively monitoring this situation and the possible effects on its financial condition, liquidity, operations, suppliers, and industry.

c) Use of estimates

The preparation of the Company's financial statements in conformity with US GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of income and expenses during the reporting period. Some of the more significant estimates include the valuation of the Company's common shares, the valuation of goodwill, the valuation of digital assets, the useful life of the mining equipment assets, and income taxes. These estimates are based on information available as of the date of the financial statements; therefore, actual results could differ from management's estimates.

d) Fair Value Measurements

The Company measures certain assets at fair value on a non-recurring basis in certain circumstances. Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. To increase the comparability of fair value measures, the following hierarchy prioritizes the inputs to valuation methodologies used to measure fair value:

- Level 1 Valuations based on quoted prices for identical assets and liabilities in active markets.
- Level 2 Valuations based on observable inputs other than quoted prices included in Level 1, such as quoted prices for similar assets and liabilities in active markets, quoted prices for identical or similar assets and liabilities in markets that are not active, or other inputs that are observable or can be corroborated by observable market data.
- Level 3 Valuations based on unobservable inputs reflecting the Company's own assumptions, consistent with reasonably available
 assumptions made by other market participants. These valuations require significant judgment.

The Company uses observable market data when determining fair value whenever possible and relies on unobservable inputs only when observable market data is not available. The Company has no material assets that are measured at fair value on a recurring basis.

The Company's non-financial assets, including mining equipment assets, goodwill, and digital assets are measured at estimated fair value on a nonrecurring basis. These assets are adjusted to fair value only when an impairment is recognized, or the underlying asset is held for sale. No non-financial assets were classified as Level 2 or Level 3 as of December 31, 2020 and 2019.

e) Cash

For the purpose of presentation in these financial statements, cash includes cash on hand. The Company holds all cash balances at a Federal Deposit Insurance Corporation ("FDIC") insured bank. Credit risk arises from the balances above the FDIC insured amount of \$250,000 held at the bank. Management constantly assesses credit risk exposure and makes appropriate adjustments when possible.

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

f) Digital assets

Digital assets are composed of crypto currencies, primarily Bitcoin. Digital assets are accounted for as indefinite-lived intangible assets. Such assets are not amortized but are assessed for impairment annually, or more frequently, when events or changes in circumstances occur indicating that it is more likely than not that the indefinite-lived asset is impaired. Impairment exists when the carrying value of a digital asset exceeds its fair value, which is measured using the quoted price at the time its fair value is being assessed. To the extent that an impairment loss is recognized, the loss establishes the new costs basis of the digital asset. The Company has concluded that no impairment or impairment events have occurred during the year ended December 31, 2020. Where digital assets are recognized as revenue, the fair value of the bitcoin received is considered to be the cost of the digital assets. The unit of account for the digital assets is defined as each individual digital asset held.

Blockcap intends to receive the economic benefits of the digital assets within one year to support operations and growth thus digital assets are classified as current on the balance sheets. Each digital asset is individually recorded at the fair value on the day it was received (see revenue recognition below) and is liquidated using the first-in, first-out (FIFO) methodology.

g) Digital assets receivable

The Company receives remuneration for the provision of computer hash power every 24 hours, at a set time, from the mining pools with which the Company has contracted. An estimate of the digital assets earned from the time the Company receives payment for the previous 24-hour period until midnight on the date of the balance sheets is recorded as digital assets receivable.

h) Mining equipment assets

Mining equipment assets are comprised of Application Specific Integrated Circuit ("ASIC") computers customised to solve blockchain algorithms. The assets are recorded at historical cost, less accumulated depreciation and impairment charges, if any. Repairs and maintenance on the servers is charged to profit and loss during the reporting period in which they occurred.

Depreciation is calculated using the straight-line method to expense the cost over their estimated useful lives of four years.

The continued value of each asset, and their expected remaining useful lives, are reviewed, and adjusted if appropriate, for impairment at the end of each reporting period.

i) Business combinations and asset acquisitions

The Company assesses whether an acquisition should be accounted for as an asset acquisition or a business combination under FASB ASC 805, *Business Combinations*. This assessment requires management to make judgements on whether the assets acquired, and liabilities assumed, constitute a business as defined in ASC 805 and if the integrated set of activities, including inputs and processes acquired, is capable of being conducted and managed for the purpose of providing a return.

FASB ASC 805, *Business Combinations*, states that if a new entity is formed to issue equity to affect a business combination one of the combining entities that existed before the business combination shall be identified as the acquirer. The identification of an acquirer requires the Company to make judgements based on the facts available to determine the most appropriate entity.

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

The acquired assets and assumed liabilities are recognized at fair value on the date the Company effectively obtains control. The measurement of each business combination is based on the information available on the acquisition date. The estimate of fair value of the acquired intangible assets (including goodwill), mining equipment assets, other assets and the liabilities assumed are based on assumptions.

j) Revenue from contracts with customers

In accordance with FASB ASC 606, Revenue from Contracts with Customers, the Company determines revenue recognition through the following steps:

- Identification of the contract with a customer
- Identification of the performance obligations in the contract
- Determination of the transaction price
- Allocation of the transaction price to the performance obligations in the contract
- Recognition of revenue when, or as, the performance obligations are satisfied

The Company accounts for a contract when it has approval and commitment from both parties, the rights of the parties are identified, payment terms are identified, the contract has commercial substance and collectability of consideration is probable.

The Company has entered into digital asset mining pools by executing contracts, as amended from time to time, with the mining pool operators to provide computing power to the mining pool. The contracts are terminable at any time by either party and the Company's enforceable right to compensation only begins when the Company provides computing power to the mining pool operator. In exchange for providing computing power, the Company is entitled to a fractional share of the fixed cryptocurrency award the mining pool operator receives (less digital asset transaction fees to the mining pool operator which are recorded as a component of cost of revenues), for successfully adding a block to the blockchain. The Company's fractional share is based on the proportion of computing power the Company contributed to the mining pool operator to the total computing power contributed by all mining pool participants in solving the current algorithm.

Providing computing power in digital asset transaction verification services is an output of the Company's ordinary activities. The provision of providing such computing power is the only performance obligation in the Company's contracts with mining pool operators. The transaction consideration the Company receives, if any, is noncash consideration, which the Company measures at fair value on the date received, which is not materially different than the fair value at the time the Company has earned the award from the pools. The consideration is all variable. There is no significant financing component in these transactions.

Fair value of the cryptocurrency award received is determined using the spot price of the coin on the date of receipt. There is currently no specific definitive guidance under US GAAP or alternative accounting framework for the accounting for cryptocurrencies recognized as revenue or held, and management has exercised significant judgment in determining the appropriate accounting treatment. In the event authoritative guidance is enacted by the FASB, the Company may be required to change its policies, which could have an effect on the Company's financial position and results from operations.

The mining pools provide the digital assets earned by the Company every 24 hours. At the close of business on any given day, the Company's total credit risk from customers is between 25% and 33% of

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

earnings from that single day. On December 31, 2020, the Company's total exposure to credit risk from customers was \$50,979 which was paid in full on January 1, 2021.

k) Cost of Revenue

The Company's cost of revenue consists primarily of direct costs of earning digital assets related to mining operations including electric power costs, hosting costs, and repairs and maintenance and the indirect costs incurred from the depreciation of the Company's mining equipment assets

l) Depreciation

Depreciation of the Company's mining equipment assets in the hosted facilities is based on an estimate of the mining assets' expected life. In order to determine the useful life of the mining assets, assumptions are required about a range of computing industry market and economic factors, including global hashrates dedicated to proof of work mining, network difficulty, technological changes, release and availability of newer and more efficient hardware and other inputs, and production costs. Based on the data that management has reviewed, management has determined to use the straight-line method of depreciation over four years, to best reflect the current expected useful life of the mining equipment assets. Management will review its estimates and assumptions at each reporting date and will revise its assumptions if new information supports a change.

m) Deferred taxes

Judgements are made by management to determine the likelihood of whether deferred tax assets at the end of the reporting period will be realized from future taxable earnings. To the extent that assumptions regarding future profitability change, there can be an increase or decrease in the amounts recognized in respect of deferred tax assets as well as the amounts recognized in profit and loss in the period in which the change occurs

n) Stock compensation

The Company recognizes the cost of services received in exchange for awards of equity instruments based upon the fair value of those awards on the grant date. For the year ended December 31, 2020, the Company's operating results included \$1,992,000 of stock-based compensation expense related to stock options units issued to consultants. Stock-based compensation expense is included within operating expenses in the Company's Statements of Operations.

Stock-based compensation expense is measured at the grant date based on the value of the equity award. The fair value of stock option awards is estimated on the date of grant using a binomial option-pricing model. The fair value of restricted stock unit awards is estimated on the date of grant using the estimated fair value of the Company's common stock on the date of grant, less a discount for lack of marketability. The value of shares issued is estimated on the date of grant using the estimated fair value of the Company's common stock on the date of the grant.

The estimated fair value of the stock-based compensation was recognized immediately as the shares issued vested on issuance.

Blockcap, Inc. Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

o) Income Taxes

Tax expense comprises current and deferred tax. Current tax and deferred tax are recognized in the statement of operations except to the extent that the tax relates to a business combination, or items recognized directly in equity.

ASC Topic 740, Income Taxes, ("ASC 740"), also clarifies the accounting for uncertainty in income taxes recognized in an enterprise's financial statements and prescribes a recognition threshold and measurement process for financial statement recognition and measurement of a tax position taken or expected to be taken in a tax return. For those benefits to be recognized, a tax position must be more-likely-than-not to be sustained upon examination by taxing authorities. ASC 740 also provides guidance on derecognition, classification, interest and penalties, accounting in interim period, disclosure and transition. Based on the Company's evaluation, it has been concluded that there are no significant uncertain tax positions requiring recognition in the Company's financial statements. Interest and penalties related to unrecognized tax liabilities are included within income tax expense. Accrued interest and penalties are included in the related tax liability line in the balance sheets. No penalties or interest have been recognized or accrued for unrecognized tax benefits for the period from inception to December 31, 2019 or the year ended December 31, 2020.

The Company believes that its income tax positions and deductions would be sustained on audit and does not anticipate any adjustments that would result in material changes to its financial position.

A deferred tax asset is recognized for unused tax losses, tax credits, and deductible temporary differences, to the extent that it is more likely than not that future taxable profits will be available against which they can be utilized.

Deferred tax assets are reviewed each reporting date and are reduced to the extent that it is more likely than not that the tax benefit will be realized.

Deferred tax is recognized in respect of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for taxation purposes. Deferred tax is not recognized for the temporary differences arising from the initial recognition of assets or liabilities in a transaction that is not a business combination and that affects neither accounting, nor taxable income or loss.

Deferred tax is measured at the tax rates that are expected to be in effect when temporary differences reverse, based on the laws that have been enacted or substantively enacted at the reporting date. Deferred tax assets and liabilities are offset if there is a legally enforceable right to offset the tax liabilities and assets, and they relate to income taxes levied by the same tax authority.

p) Loss per common share

Basic loss for common share is calculated by dividing the loss for the period by the weighted average number of common shares outstanding.

Diluted loss per common share is calculated by adjusting the figures used in the determination of basic loss per common share for the weighted average number of additional common shares that would have been outstanding assuming the conversion of all dilutive potential common shares.

q) Segment reporting

Operating segments are defined as components of an enterprise about which separate financial information is available that is evaluated regularly by the chief operating decision maker, or decision-

Blockcap, Inc. Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

making group in deciding how to allocate resources an in assessing performance. Our chief operating decision making group is composed of the executive team. We currently operate in one segment surrounding our cryptocurrency mining operation.

r) Recently issued and adopted accounting pronouncements

Accounting Standards Update ("ASU") 2016-02, *Leases*. In 2016, the FASB issued new guidance related to accounting for leases. The new guidance requires the recognition of right of use ("ROU") assets and lease liabilities for those leases classified as operating leases under previous guidance. In 2018, the FASB also approved an amendment that would permit the option to adopt the new standard prospectively as of the effective date, without adjusting comparative periods presented. On August 15, 2019, the FASB proposed a one-year delay and the effective date was deferred until fiscal years beginning after December 15, 2020. The Company is evaluating the effect of adopting ASU 2016-02.

In December 2019, the FASB issued ASU No. 2019-12, *Income Taxes* (Topic 740): *Simplifying the Accounting for Income Taxes*. The standard includes multiple key provisions, including removal of certain exceptions to FASB ASC 740, Income Taxes, and simplification in several other areas such as accounting for a franchise tax that is partially based on income. ASU 2019-12 is effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years. The Company is currently assessing the impact of adopting this standard but does not expect the adoption of this guidance to have a material impact on its financial statements.

In October 2020, the FASB issued ASU No. 2020-10 *Codification Improvements*. The new accounting rules improve the consistency of the Codification by including all disclosure guidance in the appropriate Disclosure Section (Section 50) that had only been included in the Other Presentation Matters Section (Section 45) of the Codification. Additionally, the new rules also clarify guidance across various topics including defined benefit plans, foreign currency transactions, and interest expense. The standard is effective for the Company in the first quarter of 2021. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

3. Reverse merger

On December 1, 2020, Blockcap entered into the Agreement with RME 88, under which Blockcap acquired all of the assets, liabilities, and commitments of RME 88 in exchange for the issuance of 43,688,944 shares of common stock (the "Transaction").

In accordance with FASB ASC 805, *Business Combinations*, the substance of the transaction was a reverse merger of a non-operating Company by RME 88. The accounting information and results of operations of the legal parent, Blockcap, are included in the financial statements from the date of the reverse merger and the historical financial position and results from operations are those of RME 88.

In connection with the reverse merger, the Company issued shares of common stock to individuals who were not existing owners in either Blockcap or RME 88. As a result, in connection with the reverse merger, the Company recorded \$1,992,000 of compensation expense through the issuance of 4,150,000 shares. Under FASB ASC 718, Compensation — Stock Compensation the shares were deemed to contain a service component which presumes an associated compensation expense.

Also, in connection with the reverse merger, the Company recorded \$10,002,974 of non-compensation costs through the issuance of 20,837,154 shares to other individuals. The shares were issued for nominal consideration

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

to individuals for which there was no service to Blockcap. The Company included this expense as a merger related charge in other expense in the accompanying Statement of Operations.

4. Business combinations

a) RME Black 100

On December 1, 2020, Blockcap entered into an assignment and assumption agreement with RME 100, under which Blockcap acquired all the assets, liabilities, and commitments constituting the business of RME 100 (the "RME 100 Acquisition").

Consideration given consisted of the issuance of 30,708,776 common shares to investors and founders. The RME 100 Acquisition was accounted for as a business combination using the acquisition method whereby the net assets acquired, and the liabilities assumed were recorded at fair value. The difference between the aggregate consideration relative to the fair value of the identifiable net assets is recorded as Goodwill. The Company assessed the fair values of the net assets acquired based on management's best estimate of the fair value, which takes into consideration the condition of the assets acquired and current industry conditions. The fair value of the shares issued is based on the transaction price of the recent shares issued to arms length parties.

The allocation of the purchase consideration to the fair value of the net assets acquired is presented below:

Consideration:	
30,708,776 common shares valued at \$0.48 per share	\$ 14,740,213
Fair value of assets acquired and liabilities assumed:	
Cash	\$ 266,490
Digital assets	450,412
Digital assets receivable	10,557
Prepaid expenses	675,323
Due from related parties	26,628
Mining equipment assets	6,852,352
Accrued liabilities	(7,655)
Loan payable	(3,292,973)
Deferred tax liability	(\$ 336,725)
Total identifiable net assets	\$ 4,644,409
Goodwill on acquisition, not deductible for income tax purposes	\$ 10,095,804

From the date of the RME 100 Acquisition to December 31, 2020, RME 100 contributed revenue of \$1,428,517 and net income of \$668,318. Had RME 100 been acquired on January 1, 2020, it would have contributed additional revenue of approximately \$4,010,167 and additional net income of \$873,189.

Upon completion of the transaction, the due from related parties was eliminated in consolidation upon the acquisition of RME 200.

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

b) RME Black 200

On December 1, 2020, Blockcap entered into an assignment and assumption agreement with RME 200, under which Blockcap acquired all the assets, liabilities, and commitments that constituted the business of RME 200 (the "RME 200 Acquisition").

Consideration given consisted of the issuance of 26,827,7301 common shares. The RME 200 Acquisition was accounted for as a business combination using the acquisition method whereby the net assets acquired, and the liabilities assumed were recorded at fair value. The difference between the aggregate consideration relative to the fair value of the identifiable net assets is recorded as Goodwill. The Company assessed the fair values of the net assets acquired based on management's best estimate of the fair value, which takes into consideration the condition of the assets acquired, and current industry conditions. The fair value of the shares issued is based on the transaction price of the recent shares issued to arms length parties.

The allocation of the purchase consideration to the fair value of the net assets acquired is presented as follows:

Consideration:	
26,827,730 common shares valued at \$0.48 per share	\$ 12,877,310
Fair value of assets acquired and liabilities assumed:	
Cash	\$ 152,264
Digital assets	461,788
Digital assets receivable	10,789
Prepaid expenses	674,256
Due from related parties	16,958
Mining equipment assets	6,841,524
Accrued liabilities	(5,770)
Investor draws payable	(18,519)
Due to related parties	(27,626)
Loan payable	(3,540,673)
Deferred tax liability	(\$ 336,193)
Total identifiable net assets	\$ 4,228,798
Goodwill on acquisition, not deductible for income tax purposes	\$ 8,648,512

From the date of the RME 200 Acquisition to December 31, 2020, RME 200 contributed revenue of \$1,463,854 and net income of \$742,977. Had RME 200 been acquired on January 1, 2020, it would have contributed additional revenue of approximately \$3,373,890 and additional net income of \$869,130.

Upon completion of the transaction, the due from (to) related parties were eliminated upon the acquisition of RME 200.

c) BEP 888

On December 1, 2020, Blockcap entered into an assignment and assumption agreement with BEP 888, under which Blockcap acquired all the assets, liabilities, and commitments that constituted the business of BEP 888 (the "BEP 888 Acquisition").

Consideration given consisted of the issuance of 13,501,189 common shares. The BEP 888 Acquisition was accounted for as a business combination using the acquisition method whereby the net assets acquired,

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

and the liabilities assumed were recorded at fair value. The difference between the aggregate consideration relative to the fair value of the identifiable net assets is recorded as Goodwill. The Company assessed the fair values of the net assets acquired based on management's best estimate of the fair value, which takes into consideration the condition of the assets acquired, and current industry conditions. The fair value of the shares issued is based on the transaction price of the recent shares issued to arms length parties.

The allocation of the purchase consideration to the fair value of the net assets acquired is presented as follows:

Consideration:	
13,501,189 common shares valued at \$0.48 per share	<u>\$ 6,480,571</u>
Fair value of assets acquired and liabilities assumed:	
Cash	\$ 794,445
Digital assets	81,653
Digital assets receivable	1,052
Prepaid expenses	68,277
Deposits on mining equipment assets	2,723,550
Mining equipment assets	609,842
Due to related parties	(195,336)
Deferred tax liability	(\$ 29,967)
Total identifiable net assets	\$ 4,053,516
Goodwill on acquisition, not deductible for income tax purposes	\$ 2,427,055

From the date of the BEP 888 Acquisition to December 31, 2020, BEP 888 contributed revenue of \$187,766 and a net loss of \$159,407. Had BEP 888 been acquired on January 1, 2020, it would have contributed additional revenue of approximately \$152,672 and additional net loss of (\$87,521).

5. Asset acquisition

On December 1, 2020, Blockcap entered into an assignment and assumption agreement with BEP 999, under which Blockcap acquired all the assets, liabilities, and commitments of BEP 999 (the "BEP 999 Acquisition").

On closing of the BEP 999 Acquisition, BEP 999's assets consisted primarily of working capital. As BEP 999 did not have processes capable of generating outputs, BEP 999 did not meet the definition of a business in accordance with FASB ASC 805, *Business Combinations*, and as a result the BEP 999 Acquisition has been accounted for as an equity-settled share-based transaction under FASB ASC 718, *Compensation*— *Stock Compensation*. The fair value of the consideration paid was measured at the fair values of the assets acquired and liabilities assumed.

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

Subsequent to the Transaction on December 1, 2020, the full amount of subscriptions receivable totaling \$2,450,000 were received into the bank accounts of the Company.

Purchase price:	
7,182,361 common shares	\$ 3,430,375
Fair value of assets acquired and liabilities assumed:	
Cash	\$ 1,150,000
Subscription receivable from shareholders	2,450,000
Due to related parties	(169,625)
Total identifiable net assets	\$ 3,430,375

Digital assets

Bitcoin	
Opening balance as of January 1, 2020	_
Revenue from Bitcoin mined	\$ 5,972,142
Bitcoin acquired through the business combinations	993,853
Bitcoin sold	(4,561,948)
End balance as of December 31, 2020	\$ 2,404,047

Digital assets receivable

The Company receives digital assets in exchange for the hash power provided by their mining equipment assets. The Company receives the respective reward daily and accrues any unremitted portions of digital assets earned. As of December 31, 2020 and 2019, the undelivered digital assets earned is \$50,979 and \$nil, respectively.

8. **Prepaid expenses**

The Company had the following prepaid expenses as of December 31, 2020 and 2019:

	December 31, 	December 31, 2019
Prepaid hosting fees (see note 13)	\$1,622,349	_
Other prepaid expenses	41,837	
Total	\$1,664,186	_

Blockcap, Inc. Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

9. Mining equipment assets

For the period ended December 31, 2019 and the year ended December 31, 2020, the Company's mining equipment assets are comprised wholly of ASIC miners.

Cost	
Balance, January 1, 2019	_
Additions	
Balance December 1, 2019	_
Additions	\$ 3,064,216
Mining equipment assets acquired through the business combinations	14,303,720
Balance December 31, 2020	\$ 17,367,936
	·
Accumulated depreciation	
Balance, January 1, 2019	_
Additions	
Balance December 31, 2019	
Additions	\$ 652,765
Balance December 31, 2020	\$ 652,765
Net Book value	
Cost	\$ 17,367,936
Accumulated depreciation	(652,765)
Mining equipment assets, net December 31, 2020	\$ 16,715,171

10. Accounts payable and accrued liabilities

	December 31, 2020
Accounts payable	\$ 10,167
Accrued liabilities	50,389
Total	\$ 60,556

The Company has contracted Core Scientific to host its bitcoin mining machines. Although the units are spread out among four different locations, all are hosted and maintained by the single vendor.

11. Loans payable

On April 28, 2020, RME 100 entered into an equipment finance and security agreement in the amount of \$4,559,498, which bore interest at a rate of 16% per annum which is repayable monthly ending January 31, 2022.

On June 24, 2020, RME 200 entered into an equipment finance and security agreement in the amount of \$4,552,294, which bore interest at a rate of 16% per annum which is repayable monthly ending January 23, 2022.

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

As part of the business combinations on December 1, 2020, the Company assumed the two loans noted above. The aggregate outstanding balance as of December 31, 2020 was \$3,039,665 and \$3,287,768, respectively. Both of the loans above were fully paid on March 30, 2021 (note 16).

On April 9, 2020, RME 88 entered into an equipment finance and security agreement in the amount of \$2,497,529, which bore interest at 16% per annum repayable monthly ending October 2021. The outstanding balance as of December 31, 2020 was \$1,665,019 (2019 - \$nil). The loan balance was fully paid on March 30, 2021 (note 16).

The Company incurred \$419,121 of interest during the year ended December 31, 2020 (2019 - \$nil) of which \$16,871 was accrued as of December 31, 2020 (2019 - \$nil).

The loans are secured by the mining equipment assets acquired using the borrowed funds.

12. Stockholders' equity

a) Authorized

400,000,000 common shares with a par value of \$0.00001.

b) Issued and outstanding

During the year ended December 31, 2020, the Company completed the following transactions:

- i. The Company received \$723,458 in connection with its initial formation. As a result of the Transaction, the issuance of the 43,688,944 shares of common stock has been presented on the statement of stockholders' equity to show the retroactive effect of that issuance. The Company also incurred charges of \$11,995,224 in connection with a portion of the share issuance being made to individuals with no prior ownership in the entities. (see note 3)
- ii. The Company issued 71,037,695 shares through the business combinations (see note 4).
- iii. The Company entered into subscription agreements for the issuance of 7,182,361 shares through the asset acquisition (see note 5).

As of December 31, 2020, the Company had received \$6,204,665 in relation to the private placement to be completed subsequent to year-end at January 19, 2021 as described in Note 16.

13. Related party transactions

Transfers from related parties

The managing member of RME 88 and RME 200 became the executive chairman of the Company, a key member of management, subsequent to the Agreement.

During the period from January 1, 2020 to the business combination on December 1, 2020 (see Note 4), the Company sold a total of \$195,248 of digital assets to RME 200. No receivable balance remained as of December 31, 2020 (December 31, 2019 -\$nil) as they were eliminated in connection with the business combinations.

Blockcap, Inc. Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

Transactions with key management

Key management personnel include those persons having authority and responsibility for planning, directing, and controlling the activities of the Company as a whole. The Company has determined that key management personnel consist of the Company's executive officers and directors.

During the year ended December 31, 2020, the Company was charged \$53,562 (2019 - \$nil) of management and professional fees from key management of which \$0 remained due and payable at December 31, 2020 (2019 - \$nil).

Through the Transaction and the BEP 999 Acquisition, the Company assumed the net liability of \$247,506 of balances owed to key management against which \$262,816 of payments were made creating a receivable balance of \$15,192 related to these amounts as of December 31, 2020 (2019 - \$nil)

During the year ended December 31, 2020, the Company obtained \$347,614 (2019 - \$nil) of short-term financing from key management to support the Company's operations. Of the amounts borrowed, \$4,764 remained due and payable as of December 31, 2020.

During the year ended December 31, 2020, the Company closed a bank account and \$36,778 of funds were forwarded to the accounts of an entity controlled by a member of key management. The funds will be repaid to the Company and as of December 31, 2020, \$36,778 was owed to the Company.

Of the 46,298,463 shares issued to Founders on the inception of the Company, 11,781,500 shares were issued to key management.

Transactions with affiliates

The executive chairman of the Company is a significant shareholder of Core Scientific. During the year ended December 31, 2020, the Company purchased \$3,064,216 of mining equipment assets from Core Scientific of which \$89,209 remained unpaid as of December 31, 2020. The entire balance of the deposits on mining equipment assets, \$5,087,685 total, are all amounts held at Core Scientific for the future delivery of ASIC servers.

The Company also made payments to Core Scientific totaling \$2,243,495 for hosting the mining equipment delivered and activated during the year of which \$1,622,349 was included as prepaid hosting deposits on the balance sheets. No amounts of the hosting costs remained unpaid as of December 31, 2020.

14. Income taxes

Current income tax expense represents the amount expected to be reported on the Company's income tax returns, and deferred tax expense or benefit represents the change in net deferred tax assets and liabilities. Deferred tax assets and liabilities are determined based on the difference between the financial statement and tax basis of assets and liabilities as measured by the enacted tax rates that will be in effect when these differences reverse. Valuation allowances are recorded, as appropriate, to reduce deferred tax assets to the amount considered likely to be realized.

Blockcap, Inc. Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

The expense for income taxes is summarized as follows:

Current	2020	2019
Federal	\$ 98,626	
State	16,938	
Total current	\$115,564	_
Deferred	2020	2019
Beierreu	2020	2017
Federal	\$360,668	
Federal	\$360,668	

Differences between actual provisions for income tax and provisions for income taxes at the US federal statutory rate (21%) were as follows:

	2020	2019
		(609
US federal statutory income tax benefit applied to loss before income taxes	(1,973,706)	_
State income taxes, net of federal benefit	(338,552)	_
RME 88 pre-merger income	(100,793)	_
Transaction expense	2,951,225	
Income tax provision §	538,174	

The Company's net deferred tax liability consists of the following:

	2020	2019
Computer servers	(\$ 742,391)	_
Prepaid expenses	(383,104)	
Total deferred tax liabilities	<u>(</u> \$1,125,495)	

The Company had no unrecognized income tax benefits for the year ended December 31, 2020 and period from inception to December 31, 2019. To date, no interest and penalties have been recognized related to the underpayment of income taxes. The Company continues to believe its positions are supportable; however, due to uncertainties in any tax audit outcome, the Company's estimates of the ultimate settlement of uncertain tax positions may change and the actual tax benefits may differ from the estimates.

The Company files income tax returns in the U.S. federal and various state jurisdictions. The Company's 2019 tax year is subject to U.S. federal and state examination.

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

15. Commitments

The Company has various commitments for the acquisition and hosting of mining equipment assets from Core Scientific for which there are minimum future required payments as follows:

	Year ended	Year ended
	December 31, 2020	December 31, 2019
Within one year	\$ 20,251,566	_
Later than one year but not later than five years	\$ 66,727,201	
Total	\$ 86,978,767	

The contracts with Core Scientific for hosting are at a fixed rate per miner based on power consumption and have a three-year term which automatically renews.

16. Subsequent events

The Company has evaluated subsequent events through July 30, 2021, the date the financial statements were issued, and no other matters required recognition or disclosure, other than the below, in the financial statements.

Acquisition of Subsidiary

On June 4, 2021, the Company entered into an agreement and plan of merger with Radar Relay, Inc. ("RADAR") for all the issued and outstanding equity interests of RADAR. The merger closed on July 1, 2021 and as consideration, subject to post-closing adjustments, the Company issued a total of 5,996,311 Blockcap common shares. As of July 30, 2021 the complete accounting for the merger has not been completed since the parties agreed to a maximum sixty day period within which the post-closing adjustments must be submitted and final post-closing statement would be agreed upon within 31 days of the receipt of the post-closing adjustments.

An early stage company, RADAR is focused on technology enhancement and development in the digital asset industry.

Merger with related party

On July 15, 2021 the Company entered into an agreement and plan of merger with the parent company of Core Scientific, Core Scientific Holding Co. ("Holding"), where Holding will issue up to 81,034,177 for all the outstanding shares of Blockcap. The agreement is subject to approval by the shareholders of both Blockcap and Holding and is expected to close before the end of July 2021.

Core Scientific, a wholly owned subsidiary of Holding, is an infrastructure, technology, and services company based out of Bellevue, Washington, which has three primary business activities: owning and operating datacenter facilities, owning and operating ASIC mining equipment assets, and developing AI and blockchain-based applications.

Change in management

On May 21, 2021, Clark Swanson resigned from his positions as the chief executive officer and vice chairman of the board of the Company. In connection with the resignation, Mr. Swanson forfeited 400,000 stock

Blockcap, Inc.

Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

options granted to him. Subsequent to the resignation, Darin Feinstein, executive chairman of the Company, was appointed as the chief executive officer. Mr. Feinstein was granted 1,200,000 restricted shares, vesting over two years, as compensation for his increased duties. See "Stock-based compensation" below.

Issuance of stock

Subsequent to year-end, the Company completed four private placements and issued a total of 15,840,512 shares for an aggregate amount of \$73,718,386. Of this amount, \$6,204,202 had been received in 2020 and was included as common shares issuable on the balance sheets.

Debt repayment

As of December 31, 2020, the Company had three equipment finance agreements in place originating from prior to the Transaction and business combinations as seen in Note 3 and 4.

On February 15, 2021, the Company entered into an agreement to repay their outstanding loan balance. The outstanding principal balance on February 15, 2021 was \$8,885,495 of which \$2,000,001 was settled through common stock of Blockcap at the price of \$3.15 per share. Blockcap continued to make payments in ordinary course until March 30, 2021 at which time the Company made the final payment of \$5,245,689.

Financing

On April 23, 2021, the Company entered into the Master Repurchase Agreement pursuant to which (i) Blockcap transferred 757 bitcoins (the "Purchased Securities") to Nydig Funding LLC (NYDIG) at a purchase price of up to \$30 million (the "NYDIG Purchase Price"); and (ii) Blockcap will repurchase the Purchased Securities on or before July 23, 2021 (the "Repurchase Date") by repaying the then outstanding balance of the NYDIG Purchase Price (the "Repurchase Price"). The Company had obtained \$25.6 million of funds through the agreement as of July 23, 2021.

During the period prior to the Repurchase Date, the Company is required to make monthly price differential payments relating to the outstanding Repurchase Price calculated at a price differential rate of 10% per annum. The Company may repurchase all or a portion of the Purchased Securities prior to the Repurchase Date without penalty.

On July 23, 2021, the Company entered into a bill of sale with Arctos Credit, LLC, an entity acquired by NYDIG, for 14,175 ASIC miners with an aggregate purchase price of \$25.6M to fully settle the outstanding balance. No outstanding balance as of July 30, 2021.

On May 28, 2021, the Company entered into a promissory note with a related party for \$7,749,042, bearing an interest rate equal to 10 percent compounded annually with a payment date on or before June 15th, 2021. The note was later increased \$18,864,579 and extended to August 5, 2021.

Investing

On July 23, 2021, the Company entered into two separate loan transactions with NYDIG, of 470 bitcoin each, where NYDIG borrows the bitcoins from July 23, 2021, until August 6th, 2021, and August 13, 2021, respectively. The loans carry an interest rate of 2.5% APY, payable on termination in United States dollars.

Blockcap, Inc. Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

Stock-based compensation

Subsequent to year-end, the Company issued stock-based compensation awards in the form of options to acquire shares of the Company, the details of the transactions are as follows:

Options granted	Exer	cise Price	Vesting period	Expiration Date	Quantity
January 2021	\$	2.25	Immediate	Jan-28	15,000
January 2021	\$	2.25	4 years	Jan-28	4,295,000
February 2021	\$	2.25	Immediate	Feb-28	20,000
January 2021	\$	3.15	4 years	Jan-28	1,360,000
March 2021	\$	6.00	4 years	Mar-28	565,000
March 2021	\$	10.95	4 years	Mar-28	1,940,000
April 2021	\$	10.95	4 years	Apr-28	250,000
May 2021	\$	10.95	4 years	May-28	170,000
June 2021	\$	10.95	4 years	Jun-28	250,000
July 2021	\$	10.95	Immediate	Jul-28	196,466
					9,061,466
				Forfeited	(650,000)
				Total	8,411,466

Subsequent to the year end, the Company issued restricted stock to key employees, directors, and consultants. The shares are issuable immediately and vest annually over four years.

Shares granted	Vesting period	Quantity
May 2021	4 years	100,000
June 2021	4 years	1,200,000
June 2021	4 years	1,775,000
June 2021	4 years	25,000
June 2021	4 years	20,000
July 2021	4 years	4,985,000
July 2021	Immediate	30,000
July 2021	Immediate	253,534
		8,388,534

Leases

Subsequent to the year-end, the Company entered into two new office leases. The details of the transactions are as follows:

- a) On February 11, 2021, the Company entered into an office lease in Dallas, TX effective as of April 1, 2021 for a term of 37 months. Expected rent and operating costs were expected to be approximately \$268,000 per year.
- b) On March 16, 2021, the Company entered into an office lease in Austin, TX effective as of May 1, 2021 for a term of 6 months at a cost of \$38,000 per month.

On June 30, 2021 the Company paid a termination fee of \$125,000 and terminated the Dallas lease.

Blockcap, Inc. Notes to the Financial Statements

For the year ended December 31, 2020 and for the period from February 19, 2019 (inception) to December 31, 2019

Fixed Asset Purchases

Subsequent to the year end, the Company entered into transactions with Core Scientific to acquire and host 37,000 additional Bitcoin ASIC miners, to be received throughout 2021 and 2022, for the amount of \$196,207,755 of which \$130,630,738 remains payable over the next 24 months.

The Company also entered into transactions with Core Scientific to acquire and host 750 Sia blockchain ASIC mining machines for the amount of \$1,776,750 of which \$198,750 remains due in 2021. 500 units are active as of July 30, 2021, and the remaining units will be delivered over the summer of 2021.

The Company signed a memorandum of understanding with a total value of \$12,000,000 to have 1600 Ethereum ASIC miners developed and built for which \$300,000 has been committed. The units are expected to be received in the Fall of 2021 and \$11,700,000 will be paid in fiat currency over the summer of 2021.

Also in April and May 2021, the Company purchased 475 Innosilicon A10 Pro mining computers for \$5,462,500 which connect to the Ethereum blockchain. The units are hosted in Canada and are fully paid as of July 30, 2021.

On May 5, 2021, Blockcap acquired 2,106 Bitmain ASIC miners in exchange for \$21,312,720 of the Company's common shares, the number of which will be determined upon the completion of the Company's contemplated initial public offering. The units are hosted at Core Scientific and are actively contributing to the Company's total hashrate.

Blockcap, Inc. Condensed Balance Sheets (amounts in thousands, except par value)

	Α	s of
	June 30,	December 31,
	2021 (Unaudited)	2020
Assets	(chaudica)	
Current Assets:		
Cash	\$ 552	\$ 8,757
Digital assets	43,884	2,404
Digital assets receivable	74	51
Taxes receivable	115	_
Prepaid expenses	4,039	1,664
Total Current Assets	48,664	12,876
Deposits on mining equipment assets	78,071	5,087
Property, plant, and equipment, net	55,318	16,715
Goodwill	21,172	21,172
Total Assets	\$ 203,225	\$ 55,850
Liabilities and Stockholders' Equity	· <u></u>	
Current Liabilities:		
Accounts payable and accrued expenses	\$ 220	\$ 61
Equipment purchase payable	21,313	_
Lease termination payable	102	_
Accrued interest	190	_
Accrued interest, related party	83	_
Notes payable	24.455	_
Notes payable, related party	16,146	7,992
Due to related parties	97	42
Common shares issuable		6,204
Taxes payable	_	116
Total Current Liabilities	62,606	14,415
Deferred tax liability	9,903	1,125
Total Liabilities	72,509	15,540
Commitments and contingencies (Note 6)		
Stockholders' Equity:		
Common stock; \$0.00001 par value; 400,000 shares authorized; 138,078 and 121,909 shares issued and		
outstanding at June 30, 2021 and December 31, 2020, respectively	1	1
Additional paid-in capital	131,681	50,246
Accumulated deficit	(966)	(9,937)
Total Stockholders' Equity	130,716	40,310
Total Liabilities and Stockholders' Equity	\$ 203,225	\$ 55,850
Total Diagnities and Stockholders Equity	Ψ 203,223	Φ 55,650

Blockcap, Inc. Condensed Statements of Operations (amounts in thousands, except per share amounts) (Unaudited)

	For the Six Mor June 3	
	2021	2020
Digital asset mining	\$ 59,390	208
Total revenue	59,390	208
Costs of revenue		
Hosting costs	9,468	117
Depreciation	4,340	22
Total costs of revenue	13,808	139
Gross profit	45,582	69
Operating expenses:		
Wage, benefits, and contractor costs	7,218	2
Amortization	60	_
General and administrative	1,006	_
Loss on impairment of assets	17,607	
Total operating expenses	25,891	2
Operating profit	19,691	67
Non-operating income (expense), net:		
Gain on sale of digital assets	145	10
Interest expense	(2,204)	_
Loss on termination of lease	(81)	_
Miscellaneous income	135	
Total non-operating income (expense), net	(2,005)	10
Income before income taxes	17,686	77
Income tax expense	8,715	-
Net income	8,971	77
Net income per share:		
Basic	\$ 0.07	\$ —
Diluted	\$ 0.07	\$ —
Weighted average shares outstanding:		
Basic	133,826	_
Diluted	133,855	
2 nativa	155,055	

Blockcap, Inc. Condensed Statements of Changes in Stockholders' Equity (Deficit) For the Six Months Ended June 30, 2021 and 2020 (amounts in thousands) (Unaudited)

	Commo Shares	on Stock Amount	Additional Paid-In Capital	Accumulated Deficit	Total Stockholders' Equity
Balance at January 1, 2021	121,909	\$ 1	\$ 50,246	\$ (9,937)	\$ 40,310
Net income	_	_	_	8,971	8,971
Issuance of common stock for cash and digital assets	15,840	_	73,719	_	73,719
Repurchase of shares	(3,426)	_	_	_	_
Conversion of note payable into common shares	635	_	2,000	_	2,000
Stock-based compensation from issuance of restricted shares	3,120	_	1,153		1,153
Stock-based compensation from issuance of stock options	_	_	4,563	_	4,563
Balance at June 30, 2021	138,078	\$ 1	\$ 131,681	(\$ 966)	\$ 130,716
	Common Shares	Stock Amount	Additional Paid-In Capital	Retained Earnings (Accumulated Deficit)	Total Stockholders' Equity
Balance at January 1, 2020			\$ —	\$ (1)	\$ (1)
Net income	_	_	_	77	77
Balance at June 30, 2020				76	\$ 76

Blockcap, Inc. Condensed Statements of Cash Flows (amounts in thousands) (Unaudited)

	For the Six Months Ended Jun		
Cash flows from Operating Activities:		2021	2020
Net income	\$	8,971	77
Adjustments to reconcile net income to net cash provided by (used in) operating activities:	Ψ	0,771	, ,
Depreciation and amortization		4,400	22
Stock-based compensation		5,716	
Digital asset mining		(59,390)	(208
Deferred income taxes		8,778	_
Gain on sale of digital assets		(145)	(10
Loss on impairment of digital assets		17,607	(10
Loss on termination of lease		81	
Changes in working capital components:		01	
Prepaid expenses		(2,398)	
Lease liability		(2,398)	
Income taxes payable		(231)	
Net charges paid through related parties		55	
Accrued interest		190	2
Accrued interest Accrued interest, related party		83	
		159	
Accounts payable and accrued expenses			19
Net cash used in operating activities		(16,115)	(98
Cash flows from Investing Activities:			
Deposits on mining equipment assets		(88,303)	_
Purchase of property, plant, and equipment		(6,335)	(3,064
Proceeds from sale of digital assets		871	213
Net cash used in investing activities		(93,767)	(2,851
Cash flows from Financing Activities:			
Proceeds from issuances of common stock, net		67,068	_
Advances from related parties		_	723
Proceeds from notes payable		24,455	2,498
Proceeds from notes payable, related party		16,146	_
Amounts due from related parties			(63
Repayment of notes payable		(5,992)	_
Net cash provided by financing activities		101,677	3,158
(Decrease) increase in cash		(8,205)	209
Cash - beginning of period		8,757	
	<u>c</u>	552	209
Cash - end of period	\$	332	209
Cash paid for interest	\$	1,931	_
Cash paid for taxes	\$	168	_
Supplemental disclosure of noncash investing and financing activities:	Ψ	100	
Digital assets received for common shares issued	\$	445	_
Conversion of notes payable to common shares	\$	2,000	_
Issuance of common shares to settle shares issuable	\$	6,204	_
Reclassification of miner deposits to fixed assets	\$	15,319	_
Equipment purchase payable	\$	21,313	
	\$ \$	439	
Recognition of right of use asset and liability	\$	439	_

Blockcap, Inc. Notes to Unaudited Condensed Financial Statements (Amounts in thousands, except share, per share, bitcoin, and miners ordered information)

1. ORGANIZATION AND DESCRIPTION OF BUSINESS

Blockcap, Inc. (the "Company" or "Blockcap") was incorporated under the laws of Nevada on November 25, 2020. The Company is in the business of utilizing specialized "mining" equipment to solve complex computational problems to validate transactions on blockchains, primarily bitcoin. It receives digital assets, primarily bitcoin, in return for successful service. Blockcap holds the rewards it received in its e-wallets and converts them to fiat money as cash is needed for working capital needs. All mining equipment purchased to date has been co-located in specially designed "hosting" facilities owned and operated by Core Scientific, Inc ("Core Scientific"). The registered office of the Company is located at 101 Convention Center Drive, Suite 700 Las Vegas, Nevada, 89109 and the headquarters is located at 200 E 6th Street, Austin, Texas, 78701.

On July 15, 2021, the Company entered an agreement and plan of merger with Core Scientific Holding Co. ("Holding") which later closed on July 30, 2021 (note 10).

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Basis of Presentation

The accompanying unaudited condensed interim financial statements have been prepared in accordance with the accounting principles generally accepted in the United States of America ("GAAP") for interim financial information. In the opinion of management, the accompanying unaudited condensed interim financial statements reflect all adjustments, consisting of normal recurring adjustments, considered necessary for a fair presentation of such interim results.

COVID-19

The COVID-19 global pandemic has been unpredictable and unprecedented and is likely to continue to result in significant national and global economic disruption, which may adversely affect our business. Based on the Company's current assessment, however, the Company does not expect any material impact on its long-term development, its operations, or its liquidity due to the worldwide spread of COVID-19. However, the Company is actively monitoring this situation and the possible effects on its financial condition, liquidity, operations, suppliers, and industry.

Use of Estimates

The preparation of the Company's financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of income and expenses during the reporting period. Some of the more significant estimates include the valuation of the Company's common shares, the valuation of goodwill, the valuation of digital assets, the valuation of stock based compensation, the useful life of the mining equipment assets, and income taxes. These estimates are based on information available as of the date of the financial statements; therefore, actual results could differ from management's estimates.

Fair Value Measurements

The Company measures certain assets at fair value on a non-recurring basis in certain circumstances. Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly

Blockcap, Inc. Notes to Unaudited Condensed Financial Statements (Amounts in thousands, except share, per share, bitcoin, and miners ordered information)

transaction between market participants at the measurement date. To increase the comparability of fair value measures, the following hierarchy prioritizes the inputs to valuation methodologies used to measure fair value:

- Level 1 Valuations based on quoted prices for identical assets and liabilities in active markets.
- Level 2 Valuations based on observable inputs other than quoted prices included in Level 1, such as quoted prices for similar assets and liabilities in active markets, quoted prices for identical or similar assets and liabilities in markets that are not active, or other inputs that are observable or can be corroborated by observable market data.
- Level 3 Valuations based on unobservable inputs reflecting the Company's own assumptions, consistent with reasonably available assumptions made by other market participants. These valuations require significant judgment.

The Company uses observable market data when determining fair value whenever possible and relies on unobservable inputs only when observable market data is not available. The Company has no material assets that are measured at fair value on a recurring basis.

The Company's non-financial assets, including mining equipment assets, goodwill, and digital assets are measured at estimated fair value on a nonrecurring basis. These assets are adjusted to fair value only when an impairment is recognized, or the underlying asset is held for sale. No non-financial assets were classified as Level 2 or Level 3 as of June 30, 2021 or December 31, 2020.

Revenue from contracts with customers

In accordance with FASB ASC 606, Revenue from Contracts with Customers, the Company determines revenue recognition through the following steps:

- Identification of the contract with a customer
- Identification of the performance obligations in the contract
- Determination of the transaction price
- Allocation of the transaction price to the performance obligations in the contract
- Recognition of revenue when, or as, the performance obligations are satisfied

The Company accounts for a contract when it has approval and commitment from both parties, the rights of the parties are identified, payment terms are identified, the contract has commercial substance and collectability of consideration is probable.

The Company has entered into digital asset mining pools by executing contracts, as amended from time to time, with the mining pool operators to provide computing power to the mining pool. The contracts are terminable at any time by either party and the Company's enforceable right to compensation only begins when the Company provides computing power to the mining pool operator. In exchange for providing computing power, the Company is entitled to a fractional share of the fixed cryptocurrency award the mining pool operator receives (less digital asset transaction fees to the mining pool operator which are recorded as a component of cost of revenues), for successfully adding a block to the blockchain. The Company's fractional share is based on the proportion of computing power the Company contributed to the mining pool operator to the total computing power contributed by all mining pool participants in solving the current algorithm. The pool operator may charge fees for the services provided.

Blockcap, Inc. Notes to Unaudited Condensed Financial Statements (Amounts in thousands, except share, per share, bitcoin, and miners ordered information)

Providing computing power in digital asset transaction verification services is an output of the Company's ordinary activities. The provision of providing such computing power is the only performance obligation in the Company's contracts with mining pool operators. The transaction consideration the Company receives, if any, is noncash consideration, which the Company measures at fair value on the date received, which is not materially different than the fair value at the time the Company has earned the award from the pools. The consideration is all variable. There is no significant financing component in these transactions.

Fair value of the cryptocurrency award received is determined using the spot price of the coin on the date of receipt. There is currently no specific definitive guidance under GAAP or alternative accounting framework for the accounting for cryptocurrencies recognized as revenue or held, and management has exercised significant judgment in determining the appropriate accounting treatment. In the event authoritative guidance is enacted by the FASB, the Company may be required to change its policies, which could have an effect on the Company's financial position and results from operations.

Stock Compensation

The Company recognizes the cost of services received in exchange for awards of equity instruments based upon the fair value of those awards on the grant date. For the six month period ended June 30, 2021, the Company's operating results included \$5,716, respectively, (2020 – nil) of stock-based compensation expense related to stock options and restricted stock units issued to employees, directors and consultants. Stock-based compensation expense is included within operating expenses in the Company's Statements of Operations.

Stock-based compensation expense is measured at the grant date based on the value of the equity award. The fair value of stock option awards is estimated on the date of grant using the Black Scholes option pricing model. The fair value of restricted stock unit awards is estimated on the date of grant using the estimated fair value of the Company's common stock on the date of grant.

The awards granted in the six month period ended June 30, 2021 contained only service conditions. The estimated fair value of the stock options is recognized as expense on a straight-line basis, less actual forfeitures, over the requisite service period for the entire award, which is generally the vesting period.

Leases

The Company accounts for its leases under ASC 842, *Leases* ("ASC 842"). Under this guidance, arrangements meeting the definition of a lease are classified as operating or financing leases and are recorded on the balance sheet as both a right of use asset and lease liability, calculated by discounting fixed lease payments over the lease term at the rate implicit in the lease or the Company's incremental borrowing rate. Lease liabilities are increased by interest and reduced by payments each period, and the right of use asset is amortized over the lease term. For operating leases, interest on the lease liability and the amortization of the right of use asset result in straight-line rent expense over the lease term.

In calculating the right of use asset and lease liability, the Company chose to not elect to combine lease and non-lease components as permitted under ASC 842. The Company excludes short-term leases having initial terms of 12 months or less from the new guidance as an accounting policy election and recognizes rent expense on a straight-line basis over the lease term. Additionally, for all asset classes, the Company chose not to separate non-lease components from lease components and instead account for the combined lease and non-lease components associated with that lease component as a single lease component.

Blockcap, Inc.

Notes to Unaudited Condensed Financial Statements (Amounts in thousands, except share, per share, bitcoin, and miners ordered information)

3. DIGITAL ASSETS

	As	of
	June 30 2021	December 31
Opening balance	\$ 2,404	\$ 2020
Revenue from digital asset mining	59,390	5,972
Digital assets acquired through business combinations	_	994
Digital assets acquired through issuance of shares	445	_
Net digital asset receivable change in revenue	(23)	_
Digital assets sold	(726)	(4,562)
Impairment of digital assets	(17,606)	_
Ending balance	\$ 43,884	\$ 2,404

Management reviewed the digital assets prices during the six month period ended June 30, 2021 and impaired the value of all digital assets held by \$17,606, the unit of account for the impairment being each individual digital asset held. The impaired value of the digital assets reflects the new carrying value.

4. PROPERTY, PLANT AND EQUIPMENT

As of June 30, 2021, the Company's mining equipment assets are comprised of Application Specific Integrated Circuit ("ASIC") miners.

Cost	
Balance January 1, 2021	\$17,368
Additions: mining equipment assets	42,889
Additions: Office equipment	79
Disposal of furniture	(25)
Balance June 30, 2021	\$60,311
Accumulated depreciation	
Balance January 1, 2021	\$ 653
Additions: mining equipment assets	4,340
Balance June 30, 2021	\$ 4,993
Net Book Value	
Cost	\$60,311
Accumulated depreciation	(4,993)
Balance June 30, 2021	\$55,318

During the six months ended June 30, 2021, the Company signed a memorandum of understanding with a total value of \$12,000 to have 1,600 Ethereum ASIC miners developed and built for which \$300 has been committed. Subsequent to the balance sheet date, the production of the units was halted, and the memorandum was terminated. T Company will apply the \$300 deposit against future orders.

Blockcap, Inc.

Notes to Unaudited Condensed Financial Statements (Amounts in thousands, except share, per share, bitcoin, and miners ordered information)

In April and May 2021, the Company purchased 475 Innosilicon A10 Pro mining computers for \$5,463 which connect to the Ethereum blockchain. The units are hosted in Canada.

On May 5, 2021, Blockcap acquired 2,106 Bitmain ASIC miners in exchange for \$21,313 to be paid at the option of the Company in either cash, debt, or equivalent value of the Company's common shares. As of the date of the financial statements were available to be delivered, the Company reserved 1,946,367 common shares within the exchange ratio calculation for converting Blockcap shares into shares of Core Scientific in connection with the merger of Blockcap with Core Scientific (see note 10).

In the six months ended June 30, 2021, the Company recorded \$4,340 of depreciation (2020-\$22). Depreciation for mining equipment assets is calculated on a straight-line basis for the period the assets are in service.

5. NOTES PAYABLE

	As	of
	June 30	December 31
	2021	2020
Foundry	\$ —	\$ 7,992
NYDIG repurchase agreement	24,455	
Core Scientific promissory note	16,146	_
Total notes payable, net	\$40,601	\$ 7,992

Foundry notes payable - On December 31, 2020, the Company had three equipment finance agreements with Foundry Digital, LLC. On February 15, 2021, the Company entered into an agreement to repay their outstanding loan balance. The outstanding principal balance on February 15, 2021 was \$8,885 of which \$2,000 was settled through common stock of Blockcap at the price of \$3.15 per share. Blockcap continued to make payments in ordinary course until March 30, 2021 at which time the Company made the final payment of \$5,246. The Company incurred \$1,726 of interest expense during the six month period ended June 30, 2021.

NYDIG repurchase agreement - On April 23, 2021, the Company entered into a Master Repurchase Agreement as a de-facto loan agreement in which the Company could borrow up to \$30,000 until July 23, 2021, using 757 bitcoin as collateral and at an annualized interest rate of 10%. The Company had obtained \$24,454 million of funds through the agreement by June 30, 2021. The Company incurred \$402 of interest expense during the six month period ended June 30, 2021, of which \$190 was payable as of June 30, 2021.

On July 23, 2021, the Company entered into a bill of sale with Arctos Credit, LLC, an entity acquired by NYDIG, for 14,175 ASIC miners with an aggregate purchase price of \$25.6 million to fully settle the outstanding balance.

Core Scientific promissory note - On May 28, 2021, the Company entered into a promissory note with a related party for \$7,749, bearing an interest rate equal to 10 percent compounded annually with a payment date on or before June 15, 2021. The note was later extended to July 14, 2021 and held a balance of \$16,229 as of June 30, 2021 of which \$16,146 was for equipment purchases and \$83 was accrued interest. Subsequent to June 30, 2021, the note was increased to \$32,712 and extended to October 29, 2021.

Blockcap, Inc.

Notes to Unaudited Condensed Financial Statements (Amounts in thousands, except share, per share, bitcoin, and miners ordered information)

6. COMMITMENTS AND CONTINGENCIES

Hosting

The Company has various commitments for the acquisition and hosting of mining equipment assets from Core Scientific for which there are minimum future required payments as follows:

	As	of,
	June 30	June 30
	2021	2020
Within one year	\$ 39,377	\$2,646
Later than one year but not later than five years	163,076	7,095
Total	\$202,453	\$9,741

The contracts with Core Scientific for hosting are at a fixed rate per miner based on power consumption and have a between one and three-year terms which automatically renew.

Leases

The Company leases its corporate headquarters under an operating lease agreement that expires in 2021. The lease has a six month term and thus the Company recognizes rent expense on a straight-line basis over the lease period. Rent expense from the lease was \$76 for the six month period ended June 30, 2021 (2020-nil).

In March 2021, the Company had entered into an arrangement to lease office space in Dallas, Texas which had a term of 37 months. The arrangement included options exercisable by the Company at the end of the initial terms to renew or to terminate. The company recognized \$59 of amortization from the leased assets and \$9 of interest.

On June 30, 2021, the Company paid a termination fee of \$125 and terminated the Dallas lease. As part of the termination, the lessor acquired from Blockcap the furniture located within the office space. The Company recognized a loss on termination of the lease of \$81 which represented the net balance remaining between the lease assets and liabilities as well as the required termination payment.

7. STOCKHOLDERS' EQUITY

Common stock

During the six months ended June 30, 2021, the Company completed four private placements and issued 15,840,512 common shares in exchange for a net amount of \$67,068 of cash and \$445 of digital assets.

During the six months ended June 30, 2021, the Company issued 634,921 common shares valued at \$2,000 for the settlement of the notes payable with Foundry Digital, LLC.

On June 24, 2021, the Company repurchased 3,425,702 fully vested shares at par value of \$0.00001. The shares were record by the Company as authorized but unissued shares.

Stock-based compensation

The Company's stock-based compensation expense recognized during the six months ended June 30, 2021 was attributable to Wage, benefits, and contractor costs, which are included in the operating expenses of the accompanying condensed Statement of Operations.

Blockcap, Inc. Notes to Unaudited Condensed Financial Statements (Amounts in thousands, except share, per share, bitcoin, and miners ordered information)

The Company recognized stock-based compensation expense during the six month period ended June 30, 2021 totaling \$5,716 granted and under the Company's equity incentive plan, through the issuance of 10,615,000 options to purchase common stock and 3,120,000 restricted shares granted to employees, directors, and consultants.

Stock option awards

10,580,000 of the options vest over four years with one quarter vesting each year on the anniversary date of the respective grant. 35,000 of the options vested immediately upon grant. During the six months ended June 30, 2021, the fair market value of the awards granted totaled \$48,044 and as of June 30, 2021, there was approximately \$38,371 of total unrecognized compensation cost related to unvested options, which is expected to be recognized over a remaining weighted-average vesting period of 2.15 years.

During the six month period ended June 30, 2021, a total of 1,900,000 options to acquire common shares were forfeited and 250,000 options to acquire common shares were cancelled.

A summary of the Company's unvested stock options awards activity in the six month period ended June 30, 2021 is presented here:

	Number of options	Weighted average exercise price
Unvested at January 1, 2021	-	
Granted	10,615,000	4.73
Vested	(35,000)	2.25
Forfeited	(1,900,000)	2.25
Cancelled	(250,000)	3.15
Unvested at June 30, 2021	8,430,000	5.33

The stock options had a weighted average remaining contractual term of 6.62 years

Restricted stock unit awards

During the six months ended June 30, 2021, the Company awarded 3.120,000 restricted shares of common stock to directors, employees, and advisors. 1,895,000 of the restricted shares vest over four years with one quarter vesting each year on the anniversary date of the respective grant, 1,200,000 vest over two years with the same conditions and 25,000 of the restricted shares vested upon the grant. The value of the restricted common stock grants is measured based on their fair market value on the date of the grant and amortized over their respective vesting periods. During the six months ended June 30, 2021, the fair market value of awards granted totaled \$34,328 and as of June 30, 2021, there was approximately \$33,175 of total unrecognized compensation cost related to unvested restricted common stock rights, which is expected to be recognized over a remaining weighted-average vesting period of approximately 2.3 years. The restricted shares have a weighted average remaining contractual term of 6.92 years.

Net income per share attributable to common stockholders

Basic Earnings Per Share ("EPS") is measured as the income available to common stockholders divided by the weighted average common shares outstanding for the period. Diluted EPS presents the dilutive effect on a

Blockcap, Inc. Notes to Unaudited Condensed Financial Statements

(Amounts in thousands, except share, per share, bitcoin, and miners ordered information)

per-share basis from the potential exercise of stock options. The potentially dilutive effect of options are computed using the treasury stock method. Under the treasury stock method, the amount of compensation cost for future service that the Company has not yet recognized, if any, is assumed to be used to repurchase shares.

	For the six mended June	
	2021	2020
Net income attributable to common stockholders	\$ 8,971	\$ 77
Weighted average common shares outstanding - basic	133,826	_
Net income per share – basic	\$ 0.07	\$ —
Weighted average common shares outstanding – diluted	133,855	_
Net income per share – diluted	\$ 0.07	\$ —

8. INCOME TAXES

Current income tax expense represents the amount expected to be reported on the Company's income tax returns, and deferred tax expense or benefit represents the change in net deferred tax assets and liabilities. Deferred tax assets and liabilities are determined based on the difference between the financial statement and tax basis of assets and liabilities as measured by the enacted tax rates that will be in effect when these differences reverse. Valuation allowances are recorded as appropriate to reduce deferred tax assets to the amount considered likely to be realized. The Company recorded \$8,715 of income tax expense for the six months ended June 30, 2021 (2020 - \$nil).

9. RELATED-PARTY TRANSACTIONS

In the ordinary course of business, the Company enters into various transactions with related parties.

During the six month period ended June 30, 2021, the Company was charged \$120 (2020 -\$2) for management and professional fees from an affiliated company of which \$0 remained due and payable at June 30, 2021 (2020 -\$2).

During the six month period ended June 30, 2021, the Company incurred \$311 of expenses through the affiliate for which a balance of \$97 remained unpaid as of June 30, 2021. All the expense items were charged by third parties to the affiliated entity which was then reimbursed by the Company.

Transactions with affiliates

The executive chairman of the Company is a significant shareholder of Core Scientific. During the six month period ended June 30, 2021, the Company was invoiced \$88,364 for mining equipment assets from Core Scientific of which \$16,146 remained payable through a promissory note as of June 30, 2021 (see note 5). The entire balance of the deposits on mining equipment assets, \$78,071 total, are all amounts held at Core Scientific for the future delivery of ASIC servers.

The Company was also invoiced \$11,382 for costs associated with hosting the mining equipment of which \$1,575 of prepaid hosting and \$2,222 of deposits were included within prepaid expenses on the accompanying condensed balance sheets.

Blockcap, Inc.

Notes to Unaudited Condensed Financial Statements (Amounts in thousands, except share, per share, bitcoin, and miners ordered information)

10. SUBSEQUENT EVENTS

The Company has evaluated subsequent events through October 4, 2021, the date the financial statements were available to be issued, and no other matters required recognition or disclosure, other than the below, in the financial statements.

Acquisition of Subsidiary

On June 4, 2021, the Company entered into an agreement and plan of merger with Radar Relay, Inc. ("RADAR") for all the issued and outstanding equity interests of RADAR for an estimated consideration of \$65,000, subject to closing adjustments. The merger closed on July 1, 2021 and as consideration, subject to post-closing adjustments, the Company issued a total of 5,996,310 Blockcap common shares. As of the date these financial statements were available to be issued, the complete accounting for the merger has not been completed.

An early stage company, RADAR is focused on technology enhancement and development in the digital asset industry.

Merger with related party

On July 15, 2021, the Company entered into an agreement and plan of merger with Holding, where Holding will issue up to 81,034,177 common shares in exchange for all the outstanding shares of Blockcap. The agreement closed on July 30, 2021.

Core Scientific, a wholly owned subsidiary of Holding, is an infrastructure, technology, and services company based out of Bellevue, Washington, which has three primary business activities: owning and operating datacenter facilities, owning and operating ASIC mining equipment assets, and developing AI and blockchain-based applications.

In August 2021 the Company was added as a guarantor to certain convertible notes held by Holding.

Stock-Based compensation

Subsequent to June 30, 2021, the Company assumed outstanding incentive stock options with its acquisition of RADAR. The stock-based compensation awards, in the form of options, are as follows:

Options Granted	Exercise Price	Vesting Period	Expiration Date	Quantity
July 2021		_	December	•
	\$0.74	Immediate	2028	287,637
July 2021			September	
•	\$0.74	Immediate	2029	6,403
July 2021	\$1.16	Immediate	July 2030	341,111
Total				635,151

Subsequent to June 30, 2021, the Company issued restricted stock units to key employees, directors, and consultants. The units are issuable immediately and vest as follows:

Restricted Stock Units Granted	Vesting Period	Vesting Notes	Expiration Date	Quantity
July 2021	4	_	N/A	4,985,000
July 2021	Immediate		N/A	283,534
Total				5,268,534

Blockcap, Inc. Notes to Unaudited Condensed Financial Statements (Amounts in thousands, except share, per share, bitcoin, and miners ordered information)

Subsequent to June 30, 2021, the Company issued stock-based compensation awards in the form of options to acquire shares of the Company. The details of the transactions are as follows:

Options Granted	Exercise Price	Vesting Period	Expiration Date	Quantity
July 2021			July	
	\$10.95	Immediate	2028	196,466
Total				196,466

Subsequent to June 30, 2021, options to acquire shares of the Company were forfeited as a result of terminations and departures. The details of the forfeitures are as follows:

Options Granted	Exercise Price	Vesting Period	Forfeiture Date	Quantity
April 2021	\$10.95	4	July 2021	250,000
Total				250,000

Investing

In July 2021, the Company entered into two separate loan transactions with NYDIG, of 470 bitcoin each, where NYDIG borrows the bitcoins from July 23, 2021, until August 6, 2021, and August 13, 2021, respectively. The loans carry an interest rate of 2.5% APY based on an estimated nominal dollar value of \$15,028 for each loan, payable on termination in United States dollars. As of the date these financial statements were made available to be issued, the Company has received the total bitcoin loaned and the associated interest payments.

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Members of BEP 888, LLC

Opinion on the Financial Statements

We have audited the accompanying balance sheet of BEP 888, LLC (the "Company") as of November 30, 2020, the related statements of operations, members' equity and cash flows for the period from June 1, 2020 (inception) to November 30, 2020, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Company as of November 30, 2020, and the results of its operations and its cash flows for the period from June 1, 2020 (inception) to November 30, 2020, in conformity with accounting principles generally accepted in the United States of America.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audit. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audit we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audit included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audit also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audit provide a reasonable basis for our opinion.

/s/ Marcum LLP

Marcum LLP

We served as the Company's auditor from 2020 to 2021.

Los Angeles, CA July 30, 2021

BEP 888, LLC Balance Sheet As of November 30,

	2020
Assets	
Current assets	
Cash	\$ 794,445
Digital assets	74,790
Digital assets receivable	1,052
Prepaid expenses	68,277
Total current assets	938,564
Deposits on mining equipment assets	2,723,550
Mining equipment assets	538,701
Total assets	\$ 4,200,815
Liabilities and members' equity	
Current liabilities	
Due to related parties	\$ 195,336
Total liabilities	195,336
Commitments and Contingencies (Note 9)	
Members' equity	
Ownership interests	5,847,143
Accumulated deficit	(1,841,664)
Total members' equity	4,005,479
Total liabilities and members' equity	\$ 4,200,815

BEP 888, LLC Statement of Operations For the period from June 1, 2020 (inception) to November 30, 2020

Revenue	
	\$ 152.672
Digital assets mined	\$ 152,672
Cost of revenue	
Hosting costs	40,023
Depreciation	7,653
Total cost of revenue	47,676
Gross profit	104,996
Operating expenses	
Management and start-up fees	193,960
Compensation expense from ownership interests	1,754,143
Administrative expenses	1,630
Total operating expenses	1,949,733
Operating loss	\$ (1,844,737)
Other income	
Gain on sale of digital assets	3,073
Net loss	\$ (1,841,664)

BEP 888, LLC Statement of Members' Equity For the period from June 1, 2020 (inception) to November 30, 2020

	Ownership units	Ownership interests	Accumulated deficit	Total
June 1, 2020	_	<u> </u>	<u>s</u> —	<u>s</u> —
Issuance of ownership units to managing member	30	1,754,143	_	1,754,143
Issuance of ownership units for cash	70	4,093,000	_	4,093,000
Net loss	_	_	(1,841,664)	(1,841,664)
November 30, 2020	100	\$5,847,143	\$(1,841,664)	\$ 4,005,479

BEP 888, LLC Statement of Cash Flows For the period from June 1, 2020 (inception) to November 30, 2020

	2020
Operating activities	
Net loss	\$ (1,841,664)
Adjustments to reconcile net loss to net cash used in operating activities	
Digital assets mined	(152,672)
Compensation expense from ownership interests	1,754,143
Gain on sale of digital assets	(3,073)
Digital assets receivable	(1,052)
Depreciation	7,653
Net changes in working capital	
Prepaid expenses	(68,276)
Due to related parties	193,960
Net cash used in operating activities	(110,981)
Investing activities	
Sales of digital assets	80,955
Deposits on mining equipment assets	(2,723,550)
Purchase of mining equipment assets	(546,354)
Net cash used in investing activities	(3,188,949)
Financing activities	
Proceeds from private placements	4,093,000
Advances from related parties	1,375
Net cash provided by financing activities	4,094,375
Net increase in cash	794,445
Cash, beginning of period	_
Cash, end of period	\$ 794,445
Supplemental disclosure of cash flow information	
Cash paid for interest	<u>\$</u>
Cash paid for taxes	<u>\$</u>

BEP 888 LLC

Notes to the Financial Statements For the period from June 1, 2020 (inception) to November 30, 2020

1. Nature of operations

BEP 888, LLC (the "Company" or "BEP 888") was incorporated under the laws of Nevada, USA on June 1, 2020.

BEP 888 is in the business of utilizing specialized "mining" equipment to solve complex computational problems to validate transactions on the bitcoin blockchain. It receives bitcoin in return for successful service. All mining equipment purchased to date has been co-located in specially designed "hosting" facilities owned and operated by Core Scientific, Inc. ("Core Scientific").

On December 1, 2020, the net assets of the Company were acquired by Blockcap Inc in exchange for shares (note 10).

The registered office of the Company is located at 101 Convention Center Drive, Suite 700 Las Vegas, Nevada, 89109.

2. Summary of significant accounting policies

a) Basis of presentation

The accompanying financial statements and notes of the Company have been prepared in accordance with generally accepted accounting principles in the United States of America ("US GAAP").

b) COVID-19

The COVID-19 global pandemic has been unpredictable and unprecedented and is likely to continue to result in significant national and global economic disruption, which may adversely affect our business. Based on the Company's current assessment, however, the Company does not expect any material impact on its long-term development, its operations, or its liquidity due to the worldwide spread of COVID-19. However, the Company is actively monitoring this situation and the possible effects on its financial condition, liquidity, operations, suppliers, and industry.

c) Recently issued and adopted accounting pronouncements

Accounting Standards Update ("ASU") 2016-02, Leases. In 2016, the FASB issued new guidance related to accounting for leases. The new guidance requires the recognition of right of use ("ROU") assets and lease liabilities for those leases classified as operating leases under previous guidance. In 2018, the FASB also approved an amendment that would permit the option to adopt the new standard prospectively as of the effective date, without adjusting comparative periods presented. On August 15, 2019, the FASB proposed a one-year delay and the effective date was deferred until fiscal years beginning after December 15, 2020. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

In December 2019, the FASB issued ASU No. 2019-12, Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes. The standard includes multiple key provisions, including removal of certain exceptions to FASB ASC 740, Income Taxes, and simplification in several other areas such as accounting for a franchise tax that is partially based on income. ASU 2019-12 is effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

BEP 888, LLC

Notes to the Financial Statements For the period from June 1, 2020 (inception) to November 30, 2020

In October 2020, the FASB issued ASU No. 2020-10 Codification Improvements. The new accounting rules improve the consistency of the Codification by including all disclosure guidance in the appropriate Disclosure Section (Section 50) that had only been included in the Other Presentation Matters Section (Section 45) of the Codification. Additionally, the new rules also clarify guidance across various topics including defined benefit plans, foreign currency transactions, and interest expense. The standard is effective for the Company in the first quarter of 2021. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

d) Use of estimates

The preparation of the Company's financial statements in conformity with US GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of income and expenses during the reporting period. The only significant estimate is the valuation of the useful life of the mining equipment assets. This estimate is based on information available as of the date of the financial statements; therefore, actual results could differ from management's estimates.

e) Fair value measurements

The Company measures certain assets at fair value on a non-recurring basis in certain circumstances. Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. To increase the comparability of fair value measures, the following hierarchy prioritizes the inputs to valuation methodologies used to measure fair value:

- Level 1 Valuations based on quoted prices for identical assets and liabilities in active markets.
- Level 2 Valuations based on observable inputs, other than quoted prices included in Level 1, such as quoted prices for similar assets and
 liabilities in active markets, quoted prices for identical or similar assets and liabilities in markets that are not active, or other inputs that are
 observable or can be corroborated by observable market data.
- Level 3 Valuations based on unobservable inputs reflecting the Company's own assumptions, consistent with reasonably available assumptions made by other market participants. These valuations require significant judgment.

The Company uses observable market data when determining fair value whenever possible and relies on unobservable inputs only when observable market data is not available. The Company has no material assets that are measured at fair value on a recurring basis.

The Company's non-financial assets, including mining equipment assets and digital assets are measured at estimated fair value on a nonrecurring basis. These assets are adjusted to fair value only when an impairment is recognized, or the underlying asset is held for sale. No non-financial assets were classified as Level 2 or Level 3 as of November 30, 2020.

f) Cash

For the purpose of presentation in these financial statements, cash includes cash on hand. The Company holds all cash balances at a Federal Deposit Insurance Corporation ("FDIC") insured bank. Credit risk arises from the balances above the FDIC insured amount of \$250,000 held at the bank. Management constantly assesses credit risk exposure and makes appropriate adjustments when possible.

BEP 888, LLC

Notes to the Financial Statements For the period from June 1, 2020 (inception) to November 30, 2020

g) Digital assets

Digital assets are composed of crypto currencies, primarily Bitcoin. Digital assets are accounted for as indefinite-lived intangible assets. Such assets are not amortized but are assessed for impairment annually, or more frequently, when events or changes in circumstances occur indicating that it is more likely than not that the indefinite-lived asset is impaired. Impairment exists when the carrying value of a digital asset exceeds its fair value, which is measured using the quoted price at the time its fair value is being assessed. To the extent that an impairment loss is recognized, the loss establishes the new cost basis of the digital asset. The Company has concluded that no impairment or impairment events have occurred during the year ended November 30, 2020. Where digital assets are recognized as revenue, the fair value of the bitcoin received is considered to be the cost of the digital assets.

Each digital asset is individually recorded at the fair value on the day it was received (see revenue recognition below) and is liquidated using the first-in, first-out (FIFO) methodology.

h) Digital assets receivable

The Company receives remuneration for the provision of computer hash power every 24 hours, at a set time, from the mining pools with which the Company has contracted. An estimate of the digital assets earned from the time the Company receives payment for the previous 24-hour period until midnight on the date of the balance sheet is recorded as digital assets receivable.

Mining equipment assets

Mining equipment assets are comprised of Application Specific Integrated Circuit ("ASIC") computers customized to solve blockchain algorithms. The assets are recorded at historical cost, less accumulated depreciation and impairment charges, if any. Repairs and maintenance on the servers is charged to profit and loss during the reporting period in which they occurred.

Depreciation is calculated using the straight-line method to expense the cost over their estimated useful lives of four years.

The continued value of each asset, and their expected remaining useful lives, are reviewed, and adjusted if appropriate, for impairment at the end of each reporting period.

j) Revenue from contracts with customers

In accordance with FASB ASC 606, *Revenue from Contracts with Customers*, the Company determines revenue recognition through the following steps:

- Identification of the contract with a customer
- Identification of the performance obligations in the contract
- Determination of the transaction price
- Allocation of the transaction price to the performance obligations in the contract
- · Recognition of revenue when, or as, the performance obligations are satisfied

The Company accounts for a contract when it has approval and commitment from both parties, the rights of the parties are identified, payment terms are identified, the contract has commercial substance and collectability of consideration is probable.

BEP 888, LLC

Notes to the Financial Statements For the period from June 1, 2020 (inception) to November 30, 2020

The Company has entered into digital asset mining pools by executing contracts, as amended from time to time, with the mining pool operators to provide computing power to the mining pool. The contracts are terminable at any time by either party and the Company's enforceable right to compensation only begins when the Company provides computing power to the mining pool operator. In exchange for providing computing power, the Company is entitled to a fractional share of the fixed cryptocurrency award the mining pool operator receives (less digital asset transaction fees to the mining pool operator which are recorded as a component of cost of revenues), for successfully adding a block to the blockchain. The Company's fractional share is based on the proportion of computing power the Company contributed to the mining pool operator to the total computing power contributed by all mining pool participants in solving the current algorithm.

Providing computing power in digital asset transaction verification services is an output of the Company's ordinary activities. The provision of providing such computing power is the only performance obligation in the Company's contracts with mining pool operators. The transaction consideration the Company receives, if any, is noncash consideration, which the Company measures at fair value on the date received, which is not materially different than the fair value at the time the Company has earned the award from the pools. The consideration is all variable. There is no significant financing component in these transactions.

Fair value of the cryptocurrency award received is determined using the spot price of the coin on the date of receipt. There is currently no specific definitive guidance under US GAAP or alternative accounting framework for the accounting for cryptocurrencies recognized as revenue or held, and management has exercised significant judgment in determining the appropriate accounting treatment. In the event authoritative guidance is enacted by the FASB, the Company may be required to change its policies, which could have an effect on the Company's financial position and results from operations.

The Company earns revenue through mining pools which provide the digital assets earned every 24 hours.

k) Cost of Revenue

The Company's cost of revenue consists primarily of direct costs of earning digital assets related to mining operations including electric power costs, hosting costs, and repairs and maintenance and the indirect costs incurred from the depreciation of the Company's mining equipment assets.

l) Depreciation

Depreciation of the Company's mining equipment assets in the hosted facilities is based on an estimate of the mining assets' expected life. In order to determine the useful life of the mining assets, assumptions are required about a range of computing industry market and economic factors, including global hashrates dedicated to proof of work mining, network difficulty, technological changes, release and availability of newer and more efficient hardware and other inputs, and production costs. Based on the data that management has reviewed, management has determined to use the straight-line method of depreciation over four years, to best reflect the current expected useful life of the mining equipment assets. Management will review its estimates and assumptions at each reporting date and will revise its assumptions if new information supports a change.

BEP 888 LLC

Notes to the Financial Statements For the period from June 1, 2020 (inception) to November 30, 2020

m) Segment reporting

Operating segments are defined as components of an enterprise about which separate financial information is available that is evaluated regularly by the chief operating decision maker, or decision-making group in deciding how to allocate resources an in assessing performance. Our chief operating decision making group is composed of the executive team. We currently operate in one segment surrounding our cryptocurrency mining operation.

n) Stock Compensation

The Company recognizes the awards of equity instruments issued based upon the fair value of those awards on the grant date. Stock-based compensation expense is included within operating expenses in the Company's Statements of Operations.

Stock-based compensation expense is measured at the grant date based on the value of the equity award which was determined through the valuation of the amounts raised as ownership interests during the period from June 1, 2020 (inception) and November 30, 2020.

The estimated fair value of the stock-based compensation was recognized immediately as the ownership interests granted vested on issuance.

o) Income Taxes

The Company and its members are legally organized as a limited liability company ("LLC") formed in accordance with the laws of Nevada and classified as a partnership for U.S. federal income tax purposes. Under this election, the members of the Company are solely liable for federal and state income taxes arising from the net income of the LLC. The Company has provided for and is liable for certain LLC-level state taxes based on gross revenue, and well as any partnership level withholding taxes.

The Company applies the provisions of Accounting for Uncertainty in Income Taxes under ASC 740. Under ASC 740, the Company utilizes a two-step approach to recognizing and measuring uncertain tax positions (tax contingencies). The first step is to evaluate the tax position for recognition by determining if the weight of available evidence indicates it is more likely than not that the position will be sustained on audit, including resolution of related appeals, or litigation processes. The second step is to measure the tax benefit as the largest amount, which is more than 50% likely of being realized upon ultimate settlement. There were no liabilities recorded for unrecognized tax benefits related to uncertain tax positions taken during the period from inception to November 30, 2020. The Company considers many factors when evaluating and estimating its tax positions and tax benefits, which may require periodic adjustments and which may not accurately forecast actual outcomes.

In accordance with ASC 740, the Company has elected to include interest and penalties related to its tax contingencies in income tax expense. There were no accruals for interest and penalties recognized for uncertain tax positions during the period from inception to November 30, 2020.

The Company files a U.S. federal Form 1065 tax return and a Nevada state tax return. The Company is not currently under examination by any taxing authority nor has it been notified of a pending examination.

BEP 888, LLC Notes to the Financial Statements For the period from June 1, 2020 (inception) to November 30, 2020

3. Digital assets

<u>Bitcoin</u>	2020
Opening balance at June 1, 2020	\$ —
Revenue from Bitcoin mined	152,672
Bitcoin sold	(77,882)
Ending balance as of November 30, 2020	\$ 74,790

4. Digital assets receivable

The Company receives digital assets in exchange for the hash power provided by their mining equipment assets. The Company receives the respective reward daily and accrues any unremitted portions of digital assets earned. As of the balance sheet date, the undelivered digital assets earned is \$1,052.

5. Prepaid expenses

The Company had the following prepaid expenses as of November 30, 2020:

	November 30, 2020
Prepaid hosting fees (see note 7)	\$ 68,277
Total	\$ 68,277

6. Mining equipment assets

As at November 30, 2020, the Company's mining equipment assets are comprised wholly of ASIC miners.

Cost	
Additions	\$ 546,354
Balance November 30, 2020	\$ 546,354
Accumulated depreciation	
Additions	\$ 7,653
Balance November 30, 2020	\$ 7,653
Net Book value	
Cost	\$ 546,354
Accumulated depreciation	(\$ 7,653)
Mining equipment assets, net November 30, 2020	\$ 538,701

During the period ended November 30, 2020, the Company paid \$2,723,550 of deposits on mining equipment assets to be delivered after November 30, 2020.

BEP 888, LLC

Notes to the Financial Statements For the period from June 1, 2020 (inception) to November 30, 2020

7. Due to related parties

Transactions with related parties

BEP 888, LLC, BEP 999, LLC, RME Black 100, LLC, RME Black 88, LLC, and BEP 888 (The "Related Companies") have common minority ownership.

During November 30, 2020, the Company sold \$79,907 of digital assets to a Related Company. No balance due remained unpaid as of November 30, 2020.

Entities with control, joint control, or significant influence

During the period from June 1, 2020 to November 30, 2020, the Company was charged \$193,960 of management and start-up fees by the managing member of which \$193,960 remained due and payable at November 30, 2020.

To support the operations of the Company, short term loans totaling \$1,375 were received from the affiliate during the period from June 1, 2020 to November 30, 2020. As of November 30, 2020, \$1,375 of the funds remained due and payable to the affiliate.

The managing member has a seat on the board of directors of Core Scientific. During the period ended November 30, 2020, the Company sent to Core Scientific payments totaling \$3,269,904 for the purchase of mining equipment assets and payments totaling \$108,300 for hosting the mining equipment delivered and activated during the year. No balance remained unpaid as of November 30, 2020.

8. Members' equity

During the period ended November 30, 2020, the Company completed the following transactions:

- i. On June 1, 2020, the Company issued 30 membership units to the managing member to which a charge of \$1,754,143 was recorded against compensation expense.
- ii. Between August 26, 2020 and November 9, 2020, the Company raised a total of \$4,093,000 in exchange for a total of 70 ownership units.

9. Commitments

The Company has various commitments for the acquisition and hosting of mining equipment assets from Core Scientific for which there are minimum future required payments as follows:

	As of November 30, 2020
Within one year	\$ 3,233,516
Later than one year but not later than five years	\$ 9,548,122
Total	\$ 12,781,638

The contracts with Core Scientific for hosting are at a fixed rate per miner based on power consumption and have a three-year term which automatically renews.

BEP 888, LLC Notes to the Financial Statements

For the period from June 1, 2020 (inception) to November 30, 2020

10. Subsequent events

The Company has evaluated subsequent events through July 30, 2021, the date the financial statements were issued, and no other matters required recognition or disclosure, other than the below, in the financial statements.

On December 1, 2020, Blockcap Inc. ("Blockcap") entered into an amendment and exchange agreement with the investors of BEP 888 (the "Investors") in which the Investors agreed to exchange all their investor's ownership interests in BEP 888 for a total of 9,450,832 Blockcap shares. As part of the agreement, the managing member voluntarily resigned and withdrew from BEP 888.

On December 1, 2020, BEP 888 entered into an assignment and assumption agreement with Blockcap under which Blockcap acquired all of the assets, liabilities, and commitments of BEP 888. The financial statements include the operations of BEP 888 up until December 1, 2020 at which time the assets and liabilities were acquired.

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Members of BEP 999, LLC

Opinionon the Financial Statements

We have audited the accompanying balance sheet of BEP 999, LLC (the "Company") as of November 30, 2020, the related statements of operations, members' equity and cash flows for the period from November 5, 2020 (inception) to November 30, 2020, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Company as of November 30, 2020, and the results of its operations and its cash flows for the period from November 5, 2020 (inception) to November 30, 2020, in conformity with accounting principles generally accepted in the United States of America.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audit. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audit we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audit included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audit also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audit provide a reasonable basis for our opinion.

/s/ Marcum LLP

Marcum LLP

We served as the Company's auditor from 2020 to 2021.

Los Angeles, CA July 30, 2021

BEP 999, LLC Balance Sheet As of November 30,

	2020
Assets	
Current assets	
Cash	\$ 1,150,000
Total assets	\$ 1,150,000
Liabilities and members' equity	
Current liabilities	
Due to related parties	\$ 169,625
Total liabilities	169,625
Members' equity	
Ownership interests	3,600,000
Investors' equity receivable	(2,450,000)
Accumulated deficit	(169,625)
Total members' equity	980,375
Total liabilities and members' equity	\$ 1,150,000

BEP 999, LLC Statement of Operations For the period from November 5, 2020 (inception) to November 30, 2020

Expenses	
Management and filing fees	169,625
Operating loss	(169,625)
Net loss	\$ (169,625)

BEP 999, LLC Statement of Members' Equity For the period from November 5, 2020 (inception) to November 30, 2020

	Ownership units	Ownership interests	Accumulated deficit	Total
November 5, 2020		\$ —	\$ —	\$
Issuance of ownership units to managing member	30	_	_	_
Issuance of ownership units for cash	70	3,600,000	_	3,600,000
Investors equity receivable	_	(2,450,000)	_	(2,450,000)
Net loss			(169,625)	(169,625)
November 30, 2020	100	\$ 2,692,857	\$ (169,625)	\$ 980,375

BEP 999, LLC Statement of Cash Flows For the period from November 5, 2020 (inception) to November 30, 2020

	2020
Operating activities	
Net loss	\$ (169,625)
Due to related parties	169,200
Net cash used in operating activities	(425)
Financing activities	
Proceeds from private placements	1,150,000
Advances from related parties	425
Net cash provided by financing activities	1,150,425
Net increase in cash	1,150,000
Cash, beginning of period	
Cash, end of period	\$ 1,150,000
Supplemental disclosure of cash flow information	
Cash paid for interest	\$ —
Cash paid for taxes	\$ —

BEP 999 LLC

Notes to the Financial Statements For the period from November 5, 2020 (inception) to November 30, 2020

1. Nature of operations

BEP 999, LLC. (the "Company" or "BEP 999") was incorporated under the laws of Nevada, USA on November 5, 2020.

BEP 999 was formed to utilize specialized "mining" equipment to solve complex computational problems to validate transactions on the bitcoin blockchain, receiving bitcoin in return for successful service. As of November 30, 2020, the Company had no material operations as the Company had not yet made any acquisitions of mining equipment nor derived any revenue. On December 1, 2020, the net assets of the Company were acquired by Blockcap Inc in exchange for shares (note 5).

The registered office of the Company is located at 101 Convention Center Drive, Suite 700 Las Vegas, Nevada, 89109.

2. Summary of significant accounting policies

a) Basis of presentation

The accompanying financial statements and notes of the Company have been prepared in accordance with generally accepted accounting principles in the United States of America ("US GAAP").

b) COVID-19

The COVID-19 global pandemic has been unpredictable and unprecedented and is likely to continue to result in significant national and global economic disruption, which may adversely affect our business. Based on the Company's current assessment, however, the Company does not expect any material impact on its long-term development, its operations, or its liquidity due to the worldwide spread of COVID-19. However, the Company is actively monitoring this situation and the possible effects on its financial condition, liquidity, operations, suppliers, and industry.

c) Recently issued and adopted accounting pronouncements

Accounting Standards Update ("ASU") 2016-02, Leases. In 2016, the FASB issued new guidance related to accounting for leases. The new guidance requires the recognition of right of use ("ROU") assets and lease liabilities for those leases classified as operating leases under previous guidance. In 2018, the FASB also approved an amendment that would permit the option to adopt the new standard prospectively as of the effective date, without adjusting comparative periods presented. On August 15, 2019, the FASB proposed a one-year delay and the effective date was deferred until fiscal years beginning after December 15, 2020. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

In December 2019, the FASB issued ASU No. 2019-12, Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes. The standard includes multiple key provisions, including removal of certain exceptions to FASB ASC 740, Income Taxes, and simplification in several other areas such as accounting for a franchise tax that is partially based on income. ASU 2019-12 is effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

In October 2020, the FASB issued ASU No. 2020-10 Codification Improvements. The new accounting rules improve the consistency of the Codification by including all disclosure guidance in the appropriate

BEP 999, LLC

Notes to the Financial Statements For the period from November 5, 2020 (inception) to November 30, 2020

Disclosure Section (Section 50) that had only been included in the Other Presentation Matters Section (Section 45) of the Codification.

Additionally, the new rules also clarify guidance across various topics including defined benefit plans, foreign currency transactions, and interest expense. The standard is effective for the Company in the first quarter of 2021. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

d) Use of estimates

The preparation of the Company's financial statements in conformity with US GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of income and expenses during the reporting period. No significant estimates were noted during the period ended November 30, 2020. Segment reporting Operating segments are defined as components of an enterprise about which separate financial information is available that is evaluated regularly by the chief operating decision maker, or decision-making group in deciding how to allocate resources an in assessing performance. Our chief operating decision making group is composed of the executive team.

e) Cash

For the purpose of presentation in these financial statements, cash includes cash on hand. The Company holds all cash balances at a Federal Deposit Insurance Corporation ("FDIC") insured bank. Credit risk arises from the balances above the FDIC insured amount of \$250,000 held at the bank. Management constantly assesses credit risk exposure and makes appropriate adjustments when possible.

f) Income Tax

The Company and its members are legally organized as a limited liability company ("LLC") formed in accordance with the laws of Nevada and classified as a partnership for U.S. federal income tax purposes. Under this election, the members of the Company are solely liable for federal and state income taxes arising from the net income of the LLC. The Company has provided for and is liable for certain LLC-level state taxes based on gross revenue, and well as any partnership level withholding taxes.

The Company applies the provisions of Accounting for Uncertainty in Income Taxes under ASC 740. Under ASC 740, the Company utilizes a two-step approach to recognizing and measuring uncertain tax positions (tax contingencies). The first step is to evaluate the tax position for recognition by determining if the weight of available evidence indicates it is more likely than not that the position will be sustained on audit, including resolution of related appeals, or litigation processes. The second step is to measure the tax benefit as the largest amount, which is more than 50% likely of being realized upon ultimate settlement. There were no liabilities recorded for unrecognized tax benefits related to uncertain tax positions taken during the period from inception to November 30, 2020. The Company considers many factors when evaluating and estimating its tax positions and tax benefits, which may require periodic adjustments and which may not accurately forecast actual outcomes.

In accordance with ASC 740, the Company has elected to include interest and penalties related to its tax contingencies in income tax expense. There were no accruals for interest and penalties recognized for uncertain tax positions during the period from inception to November 30, 2020

The Company files a U.S. federal Form 1065 tax return and a Nevada state tax return. The Company is not currently under examination by any taxing authority nor has it been notified of a pending examination.

BEP 999, LLC

Notes to the Financial Statements For the period from November 5, 2020 (inception) to November 30, 2020

3. Due to related parties

Entities with control, joint control, or significant influence

During the period from November 5, 2020 to November 30, 2020, the Company was charged \$169,200 of management fees by the managing member and the associate of which \$169,200 remained due and payable as of November 30, 2020.

To support the operations of the Company, short term loans totaling \$425 were received from an affiliate during the period from November 5, 2020 to November 30, 2020. As of November 30, 2020, \$425 of the funds remained due and payable to the affiliate.

4. Members' equity

During the period ended November 30, 2020, the Company completed the following transactions:

- i. In connection with the initial formation and capital raising efforts, the Company issued 30 ownership units to the managing member for no consideration. On December 1, 2020, in connection with the amendment and exchange agreement with Blockcap, Inc., the managing member forfeited these units.
- ii. Between November 22, 2020 and November 30, 2020, the Company issued 70 ownership units for \$3,600,000 of which \$2,450,000 was outstanding as of November 30, 2020. The outstanding amount was assigned to, and collected by, Blockcap, Inc. in December 2020 (note 5).

5. Subsequent events

The Company has evaluated subsequent events through July 30, 2021, the date the financial statements were issued, and no other matters required recognition or disclosure, other than the below, in the financial statements.

On December 1, 2020, Blockcap Inc. ("Blockcap") entered into an amendment and exchange agreement with the investors of BEP 999 (the "Investors") in which the Investors agreed to exchange all their investor's ownership interests in BEP 999 for a total of 7,182,361 Blockcap shares. As part of the agreement, the managing member voluntarily resigned and withdrew from BEP 999.

On December 1, 2020, BEP 999 entered into an assignment and assumption agreement with Blockcap under which Blockcap acquired all of the assets, liabilities, and commitments of BEP 999. The financial statements include the operations of BEP 999 up until December 1, 2020 at which time the assets and liabilities were acquired.

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Members of RME Black 100, LLC

Opinion on the Financial Statements

We have audited the accompanying balance sheet of RME Black 100, LLC (the "Company") as of November 30, 2020, the related statements of operations, members' equity and cash flows for the period from April 16, 2020 (inception) to November 30, 2020, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Company as of November 30, 2020, and the results of its operations and its cash flows for the period from April 16, 2020 (inception) to November 30, 2020, in conformity with accounting principles generally accepted in the United States of America.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audit. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB. We conducted our audit in accordance with the standards of the PCAOB and in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audit we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting, we express no such opinion.

Our audit included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audit also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audit provide a reasonable basis for our opinion.

/s/ Marcum LLP

Marcum LLP

We served as the Company's auditor from 2020 to 2021.

Los Angeles, CA July 30, 2021

RME Black 100, LLC Balance Sheet As of November 30,

	2020
Assets	
Current assets	
Cash	\$ 266,490
Digital assets	436,370
Digital assets receivable	10,557
Due from related parties	26,628
Prepaid expenses	675,323
Total current assets	1,415,368
Mining equipment assets	5,708,446
Total assets	\$ 7,123,814
Liabilities and members' equity	
Current liabilities	
Accrued interest	\$ 7,655
Loan payable	3,292,973
Total liabilities	3,300,628
Commitments and contingencies (Note 10)	
Members' equity	
Ownership interests	4,257,140
Accumulated deficit	(433,954)
Total members' equity	3,823,186
Total liabilities and members' equity	\$ 7,123,814

RME Black 100, LLC Statement of Operations For the period from April 16, 2020 (inception) to November 30, 2020

	2020
Revenue	
Digital assets mined	\$4,010,167
Cost of Revenue	
Hosting costs	1,959,327
Depreciation	663,773
Total cost of revenue	2,623,100
Gross profit	1,387,067
Operating Expenses	
Management and professional fees	64,389
Compensation expense from ownership interests	1,307,143
Management and start-up fees	143,350
Office and administration fees	44,263
Total operating expenses	1,559,145
Operating loss	(\$ 172,078)
Other expenses	
Interest expense	414,070
Gain on sale of digital assets	(152,194)
Total other expenses, net	\$ 261,876
Net loss	(\$ 433,954)

RME Black 100, LLC Statement of Members' Equity For the period from April 16, 2020 (inception) to November 30, 2020

	Ownership units	Ownership interests	Accumulated deficit	Total
April 16, 2020		\$ —	\$ —	\$ —
Issuance of ownership units to managing member	30	1,307,143	_	1,307,143
Issuance of ownership units for cash	70	3,050,000	_	3,050,000
Members draws	_	(100,003)	_	(100,003)
Net loss			(433,954)	(433,954)
November 30, 2020	100	\$4,257,140	(\$ 433,954)	\$3,823,186

RME Black 100, LLC Statement of Cash Flows For the period from April 16, 2020 (inception) to November 30, 2020

	2020
Operating activities	Ф. (422.05A)
Net loss	\$ (433,954)
Adjustments to reconcile net loss to net cash used in operating activities	(4.010.167)
Digital asset mined	(4,010,167)
Compensation expense from ownership interests	1,307,143
Gain on sale of digital assets	(152,194)
Digital assets receivable	(10,557)
Depreciation	663,773
Net changes in working capital	(675 222)
Prepaid expenses Accrued liabilities	(675,323) 7.655
Net cash used in operating activities	(3,303,624)
Investing activities	
Sales of digital assets	3,725,991
Advances to related parties	(313,967)
Purchase of mining equipment assets	(6,372,219)
Net cash used in investing activities	(2,960,195)
Financing activities	
Proceeds from private placements	3,050,000
Members draws	(100,003)
Payments received from related parties	287,339
Repayments of loan payable	(1,266,528)
Finance draw on loan payable	4,559,501
Net cash provided by financing activities	6,530,309
Net increase in cash	266,490
Cash, beginning of period	_
Cash, end of period	\$ 266,490
Supplemental disclosure of cash flow information	
Cash paid for interest	\$ 406,414
Cash paid for taxes	<u>\$</u>

RME Black 100, LLC Notes to the Financial Statements For the period from April 16, 2020 (inception) to November 30, 2020

1. Nature of operations

RME Black 100, LLC. (the "Company" or "RME 100") was incorporated under the laws of Nevada, USA on April 16, 2020.

RME 100 is in the business of utilizing specialized "mining" equipment to solve complex computational problems to validate transactions on the bitcoin blockchain. It receives bitcoin in return for successful service. The Company holds the rewards it received in its e-wallet and converts them to fiat money as cash is needed for working capital needs. All mining equipment purchased to date has been co-located in specially designed "hosting" facilities owned and operated by Core Scientific, Inc ("Core Scientific").

On December 1, 2020, the net assets of the Company were acquired by Blockcap Inc in exchange for shares (note 11).

The registered office of the Company is located at 101 Convention Center Drive, Suite 700 Las Vegas, Nevada, 89109.

2. Summary of significant accounting policies

a) Basis of presentation

The accompanying financial statements and notes of the Company have been prepared in accordance with generally accepted accounting principles in the United States of America ("US GAAP").

b) COVID-19

The COVID-19 global pandemic has been unpredictable and unprecedented and is likely to continue to result in significant national and global economic disruption, which may adversely affect our business. Based on the Company's current assessment, however, the Company does not expect any material impact on its long-term development, its operations, or its liquidity due to the worldwide spread of COVID-19. However, the Company is actively monitoring this situation and the possible effects on its financial condition, liquidity, operations, suppliers, and industry.

c) Recently issued and adopted accounting pronouncements

Accounting Standards Update ("ASU") 2016-02, Leases. In 2016, the FASB issued new guidance related to accounting for leases. The new guidance requires the recognition of right of use ("ROU") assets and lease liabilities for those leases classified as operating leases under previous guidance. In 2018, the FASB also approved an amendment that would permit the option to adopt the new standard prospectively as of the effective date, without adjusting comparative periods presented. On August 15, 2019, the FASB proposed a one-year delay and the effective date was deferred until fiscal years beginning after December 15, 2020. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

In December 2019, the FASB issued ASU No. 2019-12, Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes. The standard includes multiple key provisions, including removal of certain exceptions to FASB ASC 740, Income Taxes, and simplification in several other areas such as accounting for a franchise tax that is partially based on income. ASU 2019-12 is effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

RME Black 100, LLC Notes to the Financial Statements For the period from April 16, 2020 (inception) to November 30, 2020

In October 2020, the FASB issued ASU No. 2020-10 Codification Improvements. The new accounting rules improve the consistency of the Codification by including all disclosure guidance in the appropriate Disclosure Section (Section 50) that had only been included in the Other Presentation Matters Section (Section 45) of the Codification. Additionally, the new rules also clarify guidance across various topics including defined benefit plans, foreign currency transactions, and interest expense. The standard is effective for the Company in the first quarter of 2021. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

d) Use of estimates

The preparation of the Company's financial statements in conformity with US GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of income and expenses during the reporting period. The only significant estimate is the valuation of the useful life of the mining equipment assets. This estimate is based on information available as of the date of the financial statements; therefore, actual results could differ from management's estimates.

e) Fair value measurements

The Company measures certain assets at fair value on a non-recurring basis in certain circumstances. Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. To increase the comparability of fair value measures, the following hierarchy prioritizes the inputs to valuation methodologies used to measure fair value:

- Level 1—Valuations based on quoted prices for identical assets and liabilities in active markets.
- Level 2—Valuations based on observable inputs, other than quoted prices included in Level 1, such as quoted prices for similar assets and liabilities in active markets, quoted prices for identical or similar assets and liabilities in markets that are not active, or other inputs that are observable or can be corroborated by observable market data.
- Level 3—Valuations based on unobservable inputs reflecting the Company's own assumptions, consistent with reasonably available
 assumptions made by other market participants. These valuations require significant judgment.

The Company uses observable market data when determining fair value whenever possible and relies on unobservable inputs only when observable market data is not available. The Company has no material assets that are measured at fair value on a recurring basis.

The Company's non-financial assets, including mining equipment assets and digital assets are measured at estimated fair value on a nonrecurring basis. These assets are adjusted to fair value only when an impairment is recognized, or the underlying asset is held for sale. No non-financial assets were classified as Level 2 or Level 3 as of November 30, 2020.

f) Cash

For the purpose of presentation in these financial statements, cash includes cash on hand. The Company holds all cash balances at a Federal Deposit Insurance Corporation ("FDIC") insured bank. Credit risk arises from the balances above the FDIC insured amount of \$250,000 held at the bank. Management constantly assesses credit risk exposure and makes appropriate adjustments when possible.

RME Black 100, LLC Notes to the Financial Statements For the period from April 16, 2020 (inception) to November 30, 2020

g) Digital assets

Digital assets are composed of crypto currencies, primarily Bitcoin. Digital assets are accounted for as indefinite-lived intangible assets. Such assets are not amortized but are assessed for impairment annually, or more frequently, when events or changes in circumstances occur indicating that it is more likely than not that the indefinite-lived asset is impaired. Impairment exists when the carrying value of a digital asset exceeds its fair value, which is measured using the quoted price at the time its fair value is being assessed. To the extent that an impairment loss is recognized, the loss establishes the new cost basis of the digital asset. The Company has concluded that no impairment or impairment events have occurred during the period ended November 30, 2020. Where digital assets are recognized as revenue, the fair value of the bitcoin received is considered to be the cost of the digital assets.

Each digital asset is individually recorded at the fair value on the day it was received (see revenue recognition below) and is liquidated using the first-in, first-out (FIFO) methodology.

h) Digital assets receivable

The Company receives remuneration for the provision of computer hash power every 24 hours, at a set time, from the mining pools with which the Company has contracted. An estimate of the digital assets earned from the time the Company receives payment for the previous 24-hour period until midnight on the date of the balance sheet is recorded as digital assets receivable.

i) Mining equipment assets

Mining equipment assets are comprised of Application Specific Integrated Circuit ("ASIC") computers customized to solve blockchain algorithms. The assets are recorded at historical cost, less accumulated depreciation and impairment charges, if any. Repairs and maintenance on the servers is charged to profit and loss during the reporting period in which they occurred.

Depreciation is calculated using the straight-line method to expense the cost over their estimated useful lives of four years.

The continued value of each asset, and their expected remaining useful lives, are reviewed, and adjusted if appropriate, for impairment at the end of each reporting period.

j) Revenue from contracts with customers

In accordance with FASB ASC 606, Revenue from Contracts with Customers, the Company determines revenue recognition through the following steps:

- Identification of the contract with a customer
- Identification of the performance obligations in the contract
- Determination of the transaction price
- Allocation of the transaction price to the performance obligations in the contract
- · Recognition of revenue when, or as, the performance obligations are satisfied

The Company accounts for a contract when it has approval and commitment from both parties, the rights of the parties are identified, payment terms are identified, the contract has commercial substance and collectability of consideration is probable.

RME Black 100, LLC Notes to the Financial Statements For the period from April 16, 2020 (inception) to November 30, 2020

The Company has entered into digital asset mining pools by executing contracts, as amended from time to time, with the mining pool operators to provide computing power to the mining pool. The contracts are terminable at any time by either party and the Company's enforceable right to compensation only begins when the Company provides computing power to the mining pool operator. In exchange for providing computing power, the Company is entitled to a fractional share of the fixed cryptocurrency award the mining pool operator receives (less digital asset transaction fees to the mining pool operator which are recorded as a component of cost of revenues), for successfully adding a block to the blockchain. The Company's fractional share is based on the proportion of computing power the Company contributed to the mining pool operator to the total computing power contributed by all mining pool participants in solving the current algorithm.

Providing computing power in digital asset transaction verification services is an output of the Company's ordinary activities. The provision of providing such computing power is the only performance obligation in the Company's contracts with mining pool operators. The transaction consideration the Company receives, if any, is noncash consideration, which the Company measures at fair value on the date received, which is not materially different than the fair value at the time the Company has earned the award from the pools. The consideration is all variable. There is no significant financing component in these transactions.

Fair value of the cryptocurrency award received is determined using the spot price of the coin on the date of receipt. There is currently no specific definitive guidance under US GAAP or alternative accounting framework for the accounting for cryptocurrencies recognized as revenue or held, and management has exercised significant judgment in determining the appropriate accounting treatment. In the event authoritative guidance is enacted by the FASB, the Company may be required to change its policies, which could have an effect on the Company's financial position and results from operations.

The Company earns revenue through mining pools which provide the digital assets earned every 24 hours.

k) Cost of Revenue

The Company's cost of revenue consists primarily of direct costs of earning digital assets related to mining operations including electric power costs, hosting costs, and repairs and maintenance and the indirect costs incurred from the depreciation of the Company's mining equipment assets.

l) Depreciation

Depreciation of the Company's mining equipment assets in the hosted facilities is based on an estimate of the mining assets' expected life. In order to determine the useful life of the mining assets, assumptions are required about a range of computing industry market and economic factors, including global hashrates dedicated to proof of work mining, network difficulty, technological changes, release and availability of newer and more efficient hardware and other inputs, and production costs. Based on the data that management has reviewed, management has determined to use the straight-line method of depreciation over four years, to best reflect the current expected useful life of the mining equipment assets.

Management will review its estimates and assumptions at each reporting date and will revise its assumptions if new information supports a change.

RME Black 100, LLC Notes to the Financial Statements For the period from April 16, 2020 (inception) to November 30, 2020

m) Segment reporting

Operating segments are defined as components of an enterprise about which separate financial information is available that is evaluated regularly by the chief operating decision maker, or decision-making group in deciding how to allocate resources an in assessing performance. Our chief operating decision making group is composed of the executive team. We currently operate in one segment surrounding our cryptocurrency mining operation.

n) Stock compensation

The Company recognizes the awards of equity instruments issued based upon the fair value of those awards on the grant date. Stock-based compensation expense is included within operating expenses in the Company's Statements of Operations.

Stock-based compensation expense is measured at the grant date based on the value of the equity award which was determined through the valuation of the amounts raised as ownership interests during the period from April 16, 2020 (inception) and November 30, 2020.

The estimated fair value of the stock-based compensation was recognized immediately as the ownership interests granted vested on issuance.

o) Income taxes

The Company and its members are legally organized as a limited liability company ("LLC") formed in accordance with the laws of Nevada and classified as a partnership for U.S. federal income tax purposes. Under this election, the members of the Company are solely liable for federal and state income taxes arising from the net income of the LLC. The Company has provided for and is liable for certain LLC-level state taxes based on gross revenue, and well as any partnership level withholding taxes.

The Company applies the provisions of Accounting for Uncertainty in Income Taxes under ASC 740. Under ASC 740, the Company utilizes a two-step approach to recognizing and measuring uncertain tax positions (tax contingencies). The first step is to evaluate the tax position for recognition by determining if the weight of available evidence indicates it is more likely than not that the position will be sustained on audit, including resolution of related appeals, or litigation processes. The second step is to measure the tax benefit as the largest amount, which is more than 50% likely of being realized upon ultimate settlement. There were no liabilities recorded for unrecognized tax benefits related to uncertain tax positions taken during the period from inception to December 31, 2020. The Company considers many factors when evaluating and estimating its tax positions and tax benefits, which may require periodic adjustments and which may not accurately forecast actual outcomes.

In accordance with ASC 740, the Company has elected to include interest and penalties related to its tax contingencies in income tax expense. There were no accruals for interest and penalties recognized for uncertain tax positions during the period from inception to December 31, 2020.

The Company files a U.S. federal Form 1065 tax return and a Nevada state tax return. The Company is not currently under examination by any taxing authority nor has it been notified of a pending examination.

RME Black 100, LLC Notes to the Financial Statements For the period from April 16, 2020 (inception) to November 30, 2020

3. Digital assets

Bitcoin	2020
Opening balance at April 16, 2020	<u> </u>
Revenue from Bitcoin mined	4,010,167
Bitcoin sold	(3,573,797)
Ending balance as of November 30, 2020	\$ 436,370

4. Digital assets receivable

Blockcap receives digital assets in exchange for hash power provided by the mining equipment assets. The Company receives the respective reward daily and accrues unremitted portions of digital assets earned. As of the balance sheet date, the undelivered digital assets earned is \$10,557.

5. Prepaid expenses

The Company had the following prepaid expenses as of November 30, 2020:

	November 30, 2020
Prepaid hosting fees (see note 9)	\$ 675,323
Total	\$ 675.323

6. Mining equipment assets

As of November 30, 2020, the Company's mining equipment assets are comprised wholly of ASIC miners.

Cost	
Additions	\$ 6,372,219
Balance November 30, 2020	\$ 6,372,219
Accumulated depreciation	
Additions	\$ 663,773
Balance November 30, 2020	\$ 663,773
Net Book value	
Cost	\$ 6,372,219
Accumulated depreciation	(663,773)
Net mining equipment assets, November 30, 2020	\$ 5,708,446

7. Loan payable

On April 28, 2020, RME 100 entered into an equipment finance and security agreement in the amount of \$4,559,498, which bore interest at a rate of 16% per annum which is repayable monthly ending January 31, 2022. \$3,292,973 was due and payable as of November 30, 2020. The Company intends to pay down the balance of the loan within 12 months of November 30, 2020.

RME Black 100, LLC Notes to the Financial Statements For the period from April 16, 2020 (inception) to November 30, 2020

The Company incurred \$414,070 of interest during the period from April 16, 2020 (inception) to November 30, 2020 of which \$7,656 was accrued as of November 30, 2020.

The loan was secured by all of the mining equipment assets acquired by the Company.

8. Members' equity

During the period ended November 30, 2020, the Company completed the following transactions:

- On April 16, 2020, the Company issued 30 ownership units to the managing to which a charge of \$1,307,143 was recorded against compensation expense.
- ii. Between April 20, 2020 and May 8, 2020, the Company raised a total of \$3,050,000 in exchange for a total of 70 ownership units.
- iii. During the period ended November 30, 2020, members withdrew total funds amounting to \$100,003 of their initial investment.

9. Related party transactions

Transactions with related parties

BEP 888, LLC, BEP 999, LLC, RME Black 200, LLC, RME Black 88, LLC, and RME 100 (The "Related Companies") have common minority ownership.

During the period from April 16, 2020 to November 30, 2020, the Company sold \$275,892 of digital assets to a Related Company. No balance remained unpaid as of November 30, 2020.

During the period from April 16, 2020 to November 30, 2020, the Company purchased \$276,024 of digital assets from Related Companies. None of the purchases remained unpaid as of November 30, 2020.

During the period ended November 30, 2020, the Company loaned \$135,070 to a Related Company to support the operations of the other entity. \$26,628 remained outstanding as of November 30, 2020.

Entities with control, joint control, or significant influence

During the period ended November 30, 2020, the Company was charged \$187,239 of management, professional, and start-up fees from the managing member of which no amount remained due and payable at November 30, 2020.

To support the operations of the Company, short term loans totaling \$100,100 were received from the managing member during the period from April 16, 2020 to November 30, 2020. As of November 30, 2020, no amount remained payable.

The managing member is a shareholder of Core Scientific. During the period ended November 30, 2020, the Company purchased \$6,372,219 of mining equipment assets from Core Scientific and made payments totaling \$2,634,650 for hosting the mining equipment delivered and activated during the period. No balance remained unpaid as of November 30, 2020.

RME Black 100, LLC Notes to the Financial Statements For the period from April 16, 2020 (inception) to November 30, 2020

10. Commitments

The Company has various commitments for the acquisition and hosting of mining equipment assets from Core Scientific for which there are minimum future required payments as follows:

	As o	of November 30, 2020
Within one year	\$	5,361,399
Later than one year but not later than five years	\$	13,850,281
Total	\$	19,211,680

The contracts with Core Scientific for hosting are at a fixed rate per miner based on power consumption and have a three-year term which automatically renews.

11. Subsequent events

The Company has evaluated subsequent events through July 30, 2021, the date the financial statements were issued, and no other matters required recognition or disclosure, other than the below, in the financial statements.

On December 1, 2020, Blockcap Inc. ("Blockcap") entered into an amendment and exchange agreement with the investors of RME 100 (the "Investors") in which the Investors agreed to exchange all their investor's ownership interests in RME 100 for a total of 21,496,143 Blockcap shares. As part of the agreement, the managing member voluntarily resigned and withdrew from RME 100.

On December 1, 2020, RME 100 entered into an assignment and assumption agreement with Blockcap under which Blockcap acquired all of the assets, liabilities, and commitments of RME 100. The financial statements include the operations of RME 100 up until December 1, 2020 at which time the assets and liabilities were acquired.

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Members of RME Black 200, LLC

Opinion on the Financial Statements

We have audited the accompanying balance sheet of RME Black 200, LLC (the "Company") as of November 30, 2020, the related statements of operations, members' equity and cash flows for the period from April 27, 2020 (inception) to November 30, 2020, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Company as of November 30, 2020, and the results of its operations and its cash flows for the period from April 27, 2020 (inception) to November 30, 2020, in conformity with accounting principles generally accepted in the United States of America.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audit. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audit we are required to obtain an understanding of internal control over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audit included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audit also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audit provide a reasonable basis for our opinion.

/s/ Marcum LLP

Marcum LLP

We served as the Company's auditor from 2020 to 2021.

Los Angeles, CA July 30, 2021

RME Black 200, LLC Balance Sheet As of November 30,

	2020
Assets	
Current assets	
Cash	\$ 152,264
Digital assets	447,397
Digital assets receivable	10,789
Prepaid expenses	674,256
Due from related parties	16,958
Total current assets	1,301,664
Mining equipment assets	5,760,051
Total assets	\$ 7,061,715
Liabilities and equity	
Current liabilities	
Accrued liabilities	\$ 5,770
Members' draws payable	18,519
Due to related parties	27,626
Loan payable	3,540,673
Total liabilities	3,592,588
Commitments and contingencies (note 10)	
Members' equity	
Ownership interests	3,757,140
Accumulated deficit	(288,013)
Total members' equity	3,469,127
Total liabilities and members' equity	\$ 7,061,715

RME Black 200, LLC Statement of Operations For the period from April 27, 2020 (inception) to November 30, 2020

	2020
Revenue	
Digital assets mined	\$ 3,373,890
Cost of Revenue	
Hosting costs	1,510,439
Depreciation	523,641
Total cost of revenue	2,034,080
Gross Profit	1,339,810
Operating expenses	
Professional fees	5,000
Compensation expense from ownership interests	1,157,143
Management and start-up fees	165,908
Office and administration fees	14,914
Total operating expenses	1,342,965
Operating loss	(3,155)
Other expenses	
Interest expense	365,457
Gain on sale of digital assets	(80,599)
Total other expenses, net	284,858
Net loss	\$ (288,013)

RME Black 200, LLC Statement of Members' equity For the period from April 27, 2020 (inception) to November 30, 2020

	Ownership units	Ownership interests	Accumulated deficit	Total
April 27, 2020	_	\$ —	\$ —	\$ —
Issuance of ownership units to managing member	30	1,157,143	_	1,157,143
Issuance of ownership units for cash	70	2,700,000	_	2,700,000
Members draws	_	(100,003)	_	(100,003)
Net loss	_	_	(288,013)	(288,013)
November 30, 2020	100	\$ 3,757,140	\$ (288,013)	\$ 3,469,127

RME Black 200, LLC Statement of Cash Flows For the period from April 27, 2020 (inception) to November 30, 2020

Net loss \$ (288,013) Adjustments to reconcile net loss to net cash used in operating activities 3,373,890 Compensation expense from ownership interests 1,157,143 Gain on sale of digital assets (80,599) Digital assets receivable (10,789) Depreciation 523,641 Net canges in working capital 8,159 Perpendic expenses (674,256) Members draws payable 18,519 Accrued liabilities 5,770 Net cash used in operating activities (2,722,474) Investing activities (463,911) Purchase of mining equipment assets (6,283,692) Sale of digital assets 3,007,092 Net cash used in investing activities 3,007,092 Net cash used in investing activities 2,700,000 Members draw used in investing activities 2,700,000 Members draw used in investing activities 2,700,000 Proceeds from private placements 2,700,000 Members draw (10,003) Advances from related parties 474,579 Repayments of loan payable (10,116,621) Finance draw on loan payable 4,552,294		2020
Adjustments to reconcile net loss to net cash used in operating activities (3,373,890) Digital asset mined (3,373,890) Compensation expense from ownership interests 1,157,143 Gain on sale of digital assets (80,599) Digital assets receivable (10,789) Depreciation 523,641 Net changes in working capital (674,256) Prepaid expenses (674,256) Members draws payable 18,519 Accrued liabilities 5,770 Net cash used in operating activities (2,722,474) Inversing activities (463,911) Purchase of mining equipment assets (6,283,692) Sale of digital assets 3,007,092 Net cash used in investing activities (3,740,511) Financing activities Proceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable (1,011,621) Finance draw on loan payable (5,615,249) Net incre	Operating activities	
Digital asset mined (3,373,890) Compensation expense from ownership interests 1,157,143 Gain on sale of digital assets (80,599) Digital assets receivable (10,789) Depreciation 523,641 Net changes in working capital 87,770 Prepaid expenses (674,256) Members draws payable 18,519 Accrued liabilities 5,770 Net cash used in operating activities (2,722,474) Investing activities (463,911) Purchase of mining equipment assets (6,283,692) Sale of digital assets of digital assets 3,007,092 Net cash used in investing activities (37,40,511) Financing activities 2,700,000 Members draw used in investing activities (3,740,511) Finance grain investing activities (100,003) Advances from related parties (474,579) Repayments of loan payable (1,011,621) Finance draw on loan payable (1,011,621) Finance draw on loan payable (4,552,294) Net cash provided by financing activities 6,615,249		\$ (288,013)
Compensation expense from ownership interests 1,157,143 Gain on sale of digital assets (80,599) Digital assets receivable (10,789) Depreciation 523,641 Net changes in working capital (674,256) Members draws payable 18,519 Accrued liabilities 5,770 Net cash used in operating activities (463,911) Investing activities (463,911) Purchase of mining equipment assets (6,283,692) Sale of digital assets 3,007,092 Net cash used in investing activities (3,740,511) Financing activities (3,740,511) Froceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable (1,011,621) Finance draw on loan payable (5,252,64) Net increase in cash 5,552,94 Net increase in cash 152,264 Cash, beginning of period 5,552,94 Cash, end of period 5,552,94 </td <td></td> <td></td>		
Gain on sale of digital assets (80,599) Digital assets receivable (10,789) Depreciation 523,641 Net changes in working capital **** Prepaid expenses (674,256) Members draws payable 18,519 Accrued liabilities 5,770 Net cash used in operating activities (2,722,474) Investing activities *** Advances to related parties (6,283,692) Sale of digital assets (5,070,002) Sale of digital assets 3,007,092 Net cash used in investing activities (3,740,511) Financing activities 2,700,000 Members draws (100,003) Advances from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, end of period 5,152,264 Supplemental disclosure of cash flow information 3,395,687		
Digital assets receivable (10,789) Depreciation 523,641 Net changes in working capital (674,256) Prepaid expenses (674,256) Members draws payable 18,519 Accrued liabilities 5,770 Net cash used in operating activities (2,722,474) Investing activities (463,911) Purchase of mining equipment assets (6,283,692) Sale of digital assets (3,070,902) Net cash used in investing activities (3,740,511) Financing activities 2,700,000 Members draws (100,003) Advances from private placements 2,700,000 Members draws (100,003) Advances from private placements 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,618,249 Net increase in cash 5,2264 Cash, beginning of period Cash, end of period 5,522,64 Supplemental disclosure of cash flow information 5,359,687		
Depreciation 523,641 Net changes in working capital (674,256) Prepaid expenses (674,256) Members draws payable 18,519 Accrued liabilities 5,770 Net cash used in operating activities (2,722,474) Investing activities 4(463,911) Purchase of mining equipment assets (6,283,692) Sale of digital assets 3,007,092 Net cash used in investing activities (3,740,511) Financing activities 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (101,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 6,615,249 Net increase in cash 5,2264 Supplemental disclosure of cash flow information 5,359,687 Cash, peginning of period 5,359,687 Cash paid for interest 8,359,687 Cash paid for interest 8,359,687 Cash paid for interest 8,259,687		
Net changes in working capital 674,256 Prepaid expenses (674,256) Members draws payable 18,519 Accrued liabilities 5,770 Net cash used in operating activities (2,722,474) Investing activities 8 Advances to related parties (463,911) Purchase of mining equipment assets 3,007,092 Sale of digital assets 3,007,092 Net cash used in investing activities 3,007,092 Proceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net cash provided by financing activities 5,52,624 Supplemental disclosure of cash flow information 5 Cash, beginning of period 5 Cash paid for interest \$359,687 Cash paid for interest \$359,687 Cash paid for interest	E Company of the Comp	
Prepaid expenses (674,256) Members draws payable 18,519 Accrued liabilities 5,770 Net cash used in operating activities (2,722,474) Investing activities 463,911 Purchase of mining equipment assets (6,283,692) Sale of digital assets 3,007,092 Net cash used in investing activities 3,007,092 Proceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, end of period — Cash, end of period 5,152,264 Supplemental disclosure of cash flow information \$359,687 Cash paid for interest \$359,687 Cash paid for interest \$359,687 Cash paid for necesh investing activities \$359,687		523,641
Members draws payable 18,519 Accrued liabilities 5,770 Net cash used in operating activities (2,722,474) Investing activities 4463,911 Purchase of mining equipment assets (6,283,692) Sale of digital assets 3,007,092 Net cash used in investing activities 3,007,092 Proceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period — Cash, end of period 5 Supplemental disclosure of cash flow information \$359,687 Cash paid for interest \$359,687 Cash paid for taxes \$		
Accrued liabilities 5,770 Net cash used in operating activities (2,722,474) Investing activities 8 Advances to related parties (463,911) Purchase of mining equipment assets 3,007,092 Sale of digital assets 3,007,092 Net cash used in investing activities 2,700,000 Financing activities 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period — Cash, end of period \$152,264 Supplemental disclosure of cash flow information \$359,687 Cash paid for interest \$359,687 Cash paid for taxes \$359,687		
Net cash used in operating activities (2,722,474) Investing activities (463,911) Advances to related parties (6,283,692) Sale of digital assets 3,007,092 Net cash used in investing activities (3,740,511) Financing activities 2,700,000 Proceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period — Cash, end of period 5 Cash paid for interest \$ 359,687 Cash paid for interest \$ 359,687 Cash paid for taxes \$ - Supplemental disclosure of noncash investing activities \$ -		
Investing activities Advances to related parties (463,911) Purchase of mining equipment assets (6,283,692) Sale of digital assets 3,007,092 Net cash used in investing activities (3,740,511) Financing activities 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, end of period 5 Cash, end of period 5 Cash paid for interest \$ 359,687 Cash paid for interest \$ 359,687 Cash paid for taxes \$ - Supplemental disclosure of noncash investing activities \$ -	Accrued liabilities	5,770
Advances to related parties (463,911) Purchase of mining equipment assets (6,283,692) Sale of digital assets 3,007,092 Net cash used in investing activities (3,740,511) Financing activities 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period — Cash, end of period \$ 152,264 Supplemental disclosure of cash flow information \$ 152,264 Cash paid for interest \$ 359,687 Cash paid for taxes \$ — Supplemental disclosure of noncash investing activities \$ —	Net cash used in operating activities	(2,722,474)
Purchase of mining equipment assets (6,283,692) Sale of digital assets 3,007,092 Net cash used in investing activities (3,740,511) Financing activities Proceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period — Cash, end of period \$ 152,264 Supplemental disclosure of cash flow information \$ 359,687 Cash paid for interest \$ 359,687 Cash paid for taxes \$ —	Investing activities	
Sale of digital assets 3,007,092 Net cash used in investing activities (3,740,511) Financing activities Proceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period — Cash, end of period \$ 152,264 Supplemental disclosure of cash flow information \$ 359,687 Cash paid for interest \$ 359,687 Cash paid for taxes \$ —	Advances to related parties	(463,911)
Net cash used in investing activities (3,740,511) Financing activities 2,700,000 Proceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period — Cash, end of period \$ 152,264 Supplemental disclosure of cash flow information \$ 359,687 Cash paid for interest \$ 359,687 Cash paid for taxes \$ —	Purchase of mining equipment assets	(6,283,692)
Financing activities Proceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period - Cash, end of period \$ 152,264 Supplemental disclosure of cash flow information \$ 359,687 Cash paid for interest \$ 359,687 Cash paid for taxes \$ - Supplemental disclosure of noncash investing activities \$ -	Sale of digital assets	3,007,092
Proceeds from private placements 2,700,000 Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period — Cash, end of period \$ 152,264 Supplemental disclosure of cash flow information \$ 359,687 Cash paid for interest \$ 359,687 Cash paid for taxes \$ —	Net cash used in investing activities	(3,740,511)
Members draws (100,003) Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period — Cash, end of period \$ 152,264 Supplemental disclosure of cash flow information \$ 152,264 Cash paid for interest \$ 359,687 Cash paid for taxes \$ — Supplemental disclosure of noncash investing activities \$ —	Financing activities	
Advances from related parties 474,579 Repayments of loan payable (1,011,621) Finance draw on loan payable 4,552,294 Net cash provided by financing activities 6,615,249 Net increase in cash 152,264 Cash, beginning of period — Cash, end of period \$ 152,264 Supplemental disclosure of cash flow information \$ 359,687 Cash paid for interest \$ 359,687 Cash paid for taxes \$ — Supplemental disclosure of noncash investing activities \$ —	Proceeds from private placements	2,700,000
Repayments of loan payable Finance draw on loan payable A,552,294 Net cash provided by financing activities Net increase in cash Cash, beginning of period Cash, end of period Supplemental disclosure of cash flow information Cash paid for interest Cash paid for taxes Supplemental disclosure of noncash investing activities (1,011,621) 4,552,294 6,615,249 152,264 Supplemental disclosure of cash flow information Cash paid for interest Supplemental disclosure of noncash investing activities	Members draws	(100,003)
Finance draw on loan payable Net cash provided by financing activities Net increase in cash Cash, beginning of period Cash, beginning of period Cash, end of period Supplemental disclosure of cash flow information Cash paid for interest Cash paid for taxes Supplemental disclosure of noncash investing activities		474,579
Net cash provided by financing activities Net increase in cash Cash, beginning of period Cash, end of period Supplemental disclosure of cash flow information Cash paid for interest Cash paid for taxes Supplemental disclosure of noncash investing activities	Repayments of loan payable	(1,011,621)
Net increase in cash Cash, beginning of period Cash, end of period Supplemental disclosure of cash flow information Cash paid for interest Cash paid for taxes Supplemental disclosure of noncash investing activities	Finance draw on loan payable	4,552,294
Net increase in cash Cash, beginning of period Cash, end of period Supplemental disclosure of cash flow information Cash paid for interest Cash paid for taxes Supplemental disclosure of noncash investing activities	Net cash provided by financing activities	6,615,249
Cash, beginning of period Cash, end of period Supplemental disclosure of cash flow information Cash paid for interest Cash paid for taxes Supplemental disclosure of noncash investing activities		152.264
Cash, end of period Supplemental disclosure of cash flow information Cash paid for interest Cash paid for taxes Supplemental disclosure of noncash investing activities	Cash, beginning of period	_
Cash paid for interest Cash paid for taxes Supplemental disclosure of noncash investing activities \$ 359,687 \$ —	, 6 6 1	\$ 152,264
Cash paid for taxes Supplemental disclosure of noncash investing activities	Supplemental disclosure of cash flow information	
Supplemental disclosure of noncash investing activities	Cash paid for interest	\$ 359,687
	Cash paid for taxes	<u>\$</u> —
Digital assets used to pay finance and management fees \$ 233,487	Supplemental disclosure of noncash investing activities	
	Digital assets used to pay finance and management fees	\$ 233,487

RME Black 200, LLC Notes to the Financial Statements For the period from April 27, 2020 (inception) to November 30, 2020

1. Nature of operations

RME Black 200, LLC. (the "Company" or "RME 200") was incorporated under the laws of Nevada, USA on April 27, 2020.

RME 200 is in the business of utilizing specialized "mining" equipment to solve complex computational problems to validate transactions on the bitcoin blockchain. It receives bitcoin in return for successful service. The Company holds the rewards it received in its e-wallet and converts them to fiat money as cash is needed for working capital needs. All mining equipment purchased to date has been co-located in specially designed "hosting" facilities owned and operated by Core Scientific, Inc ("Core Scientific").

On December 1, 2020, the net assets of the Company were acquired by Blockcap Inc in exchange for shares (note 11).

The registered office of the Company is located at 101 Convention Center Drive, Suite 700 Las Vegas, Nevada, 89109.

2. Summary of significant accounting policies

a) Basis of presentation

The accompanying financial statements and notes of the Company have been prepared in accordance with generally accepted accounting principles in the United States of America ("US GAAP").

b) COVID-19

The COVID-19 global pandemic has been unpredictable and unprecedented and is likely to continue to result in significant national and global economic disruption, which may adversely affect our business. Based on the Company's current assessment, however, the Company does not expect any material impact on its long-term development, its operations, or its liquidity due to the worldwide spread of COVID-19. However, the Company is actively monitoring this situation and the possible effects on its financial condition, liquidity, operations, suppliers, and industry.

c) Recently issued and adopted accounting pronouncements

Accounting Standards Update ("ASU") 2016-02, Leases. In 2016, the FASB issued new guidance related to accounting for leases. The new guidance requires the recognition of right of use ("ROU") assets and lease liabilities for those leases classified as operating leases under previous guidance. In 2018, the FASB also approved an amendment that would permit the option to adopt the new standard prospectively as of the effective date, without adjusting comparative periods presented. On August 15, 2019, the FASB proposed a one-year delay and the effective date was deferred until fiscal years beginning after December 15, 2020. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

In December 2019, the FASB issued ASU No. 2019-12, Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes. The standard includes multiple key provisions, including removal of certain exceptions to FASB ASC 740, Income Taxes, and simplification in several other areas such as accounting for a franchise tax that is partially based on income. ASU 2019-12 is effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

RME Black 200, LLC Notes to the Financial Statements For the period from April 27, 2020 (inception) to November 30, 2020

In October 2020, the FASB issued ASU No. 2020-10 Codification Improvements. The new accounting rules improve the consistency of the Codification by including all disclosure guidance in the appropriate Disclosure Section (Section 50) that had only been included in the Other Presentation Matters Section (Section 45) of the Codification. Additionally, the new rules also clarify guidance across various topics including defined benefit plans, foreign currency transactions, and interest expense. The standard is effective for the Company in the first quarter of 2021. The Company does not expect the adoption of the new accounting rules to have a material impact on its financial statements.

d) Use of estimates

The preparation of the Company's financial statements in conformity with US GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of income and expenses during the reporting period. The only significant estimate is the valuation of the useful life of the mining equipment assets. This estimate is based on information available as of the date of the financial statements; therefore, actual results could differ from management's estimates.

e) Fair value measurements

The Company measures certain assets at fair value on a non-recurring basis in certain circumstances. Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. To increase the comparability of fair value measures, the following hierarchy prioritizes the inputs to valuation methodologies used to measure fair value:

- Level 1 Valuations based on quoted prices for identical assets and liabilities in active markets.
- Level 2 Valuations based on observable inputs, other than quoted prices included in Level 1, such as quoted prices for similar assets and
 liabilities in active markets, quoted prices for identical or similar assets and liabilities in markets that are not active, or other inputs that are
 observable or can be corroborated by observable market data.
- Level 3 Valuations based on unobservable inputs reflecting the Company's own assumptions, consistent with reasonably available assumptions made by other market participants. These valuations require significant judgment.

The Company uses observable market data when determining fair value whenever possible and relies on unobservable inputs only when observable market data is not available. The Company has no material assets that are measured at fair value on a recurring basis.

The Company's non-financial assets, including mining equipment assets, and digital assets are measured at estimated fair value on a nonrecurring basis. These assets are adjusted to fair value only when an impairment is recognized, or the underlying asset is held for sale. No non-financial assets were classified as Level 2 or Level 3 as of November 30, 2020.

f) Cash

For the purpose of presentation in these financial statements, cash includes cash on hand. The Company holds all cash balances at a Federal Deposit Insurance Corporation ("FDIC") insured bank. Credit risk arises from the balances above the FDIC insured amount of \$250,000 held at the bank. Management constantly assesses credit risk exposure and makes appropriate adjustments when possible.

RME Black 200, LLC Notes to the Financial Statements For the period from April 27, 2020 (inception) to November 30, 2020

g) Digital assets

Digital assets are composed of crypto currencies, primarily Bitcoin. Digital assets are accounted for as indefinite-lived intangible assets. Such assets are not amortized but are assessed for impairment annually, or more frequently, when events or changes in circumstances occur indicating that it is more likely than not that the indefinite-lived asset is impaired. Impairment exists when the carrying value of a digital asset exceeds its fair value, which is measured using the quoted price at the time its fair value is being assessed. To the extent that an impairment loss is recognized, the loss establishes the new cost basis of the digital asset. The Company has concluded that no impairment or impairment events have occurred during the period ended November 30, 2020. Where digital assets are recognized as revenue, the fair value of the bitcoin received is considered to be the cost of the digital assets.

Each digital asset is individually recorded at the fair value on the day it was received (see revenue recognition below) and is liquidated using the first-in, first-out (FIFO) methodology.

h) Digital assets receivable

The Company receives remuneration for the provision of computer hash power every 24 hours, at a set time, from the mining pools with which the Company has contracted. An estimate of the digital assets earned from the time the Company receives payment for the previous 24-hour period until midnight on the date of the balance sheet is recorded as digital assets receivable.

i) Mining equipment assets

Mining equipment assets are comprised of Application Specific Integrated Circuit ("ASIC") computers customized to solve blockchain algorithms. The assets are recorded at historical cost, less accumulated depreciation and impairment charges, if any. Repairs and maintenance on the servers is charged to profit and loss during the reporting period in which they occurred.

Depreciation is calculated using the straight-line method to expense the cost over their estimated useful lives of four years.

The continued value of each asset, and their expected remaining useful lives, are reviewed, and adjusted if appropriate, for impairment at the end of each reporting period.

j) Revenue from contracts with customers

In accordance with FASB ASC 606, *Revenue from Contracts with Customers*, the Company determines revenue recognition through the following steps:

- Identification of the contract with a customer
- Identification of the performance obligations in the contract
- Determination of the transaction price
- Allocation of the transaction price to the performance obligations in the contract
- · Recognition of revenue when, or as, the performance obligations are satisfied

The Company accounts for a contract when it has approval and commitment from both parties, the rights of the parties are identified, payment terms are identified, the contract has commercial substance and collectability of consideration is probable.

RME Black 200, LLC Notes to the Financial Statements For the period from April 27, 2020 (inception) to November 30, 2020

The Company has entered into digital asset mining pools by executing contracts, as amended from time to time, with the mining pool operators to provide computing power to the mining pool. The contracts are terminable at any time by either party and the Company's enforceable right to compensation only begins when the Company provides computing power to the mining pool operator. In exchange for providing computing power, the Company is entitled to a fractional share of the fixed cryptocurrency award the mining pool operator receives (less digital asset transaction fees to the mining pool operator which are recorded as a component of cost of revenues), for successfully adding a block to the blockchain. The Company's fractional share is based on the proportion of computing power the Company contributed to the mining pool operator to the total computing power contributed by all mining pool participants in solving the current algorithm.

Providing computing power in digital asset transaction verification services is an output of the Company's ordinary activities. The provision of providing such computing power is the only performance obligation in the Company's contracts with mining pool operators. The transaction consideration the Company receives, if any, is noncash consideration, which the Company measures at fair value on the date received, which is not materially different than the fair value at the time the Company has earned the award from the pools. The consideration is all variable. There is no significant financing component in these transactions.

Fair value of the cryptocurrency award received is determined using the spot price of the coin on the date of receipt. There is currently no specific definitive guidance under US GAAP or alternative accounting framework for the accounting for cryptocurrencies recognized as revenue or held, and management has exercised significant judgment in determining the appropriate accounting treatment. In the event authoritative guidance is enacted by the FASB, the Company may be required to change its policies, which could have an effect on the Company's financial position and results from operations.

The Company earns revenue through mining pools which provide the digital assets earned every 24 hours.

k) Cost of Revenue

The Company's cost of revenue consists primarily of direct costs of earning digital assets related to mining operations including electric power costs, hosting costs, and repairs and maintenance and the indirect costs incurred from the depreciation of the Company's mining equipment assets.

l) Depreciation

Depreciation of the Company's mining equipment assets in the hosted facilities is based on an estimate of the mining assets' expected life. In order to determine the useful life of the mining assets, assumptions are required about a range of computing industry market and economic factors, including global hashrates dedicated to proof of work mining, network difficulty, technological changes, release and availability of newer and more efficient hardware and other inputs, and production costs. Based on the data that management has reviewed, management has determined to use the straight-line method of depreciation over four years, to best reflect the current expected useful life of the mining equipment assets. Management will review its estimates and assumptions at each reporting date and will revise its assumptions if new information supports a change.

RME Black 200, LLC Notes to the Financial Statements For the period from April 27, 2020 (inception) to November 30, 2020

m) Segment reporting

Operating segments are defined as components of an enterprise about which separate financial information is available that is evaluated regularly by the chief operating decision maker, or decision-making group in deciding how to allocate resources an in assessing performance. Our chief operating decision making group is composed of the executive team. We currently operate in one segment surrounding our cryptocurrency mining operation.

n) Stock compensation

The Company recognizes the awards of equity instruments issued based upon the fair value of those awards on the grant date. Stock-based compensation expense is included within operating expenses in the Company's Statements of Operations.

Stock-based compensation expense is measured at the grant date based on the value of the equity award which was determined through the valuation of the amounts raised as ownership interests during the period from April 27, 2020 (inception) and November 30, 2020.

The estimated fair value of the stock-based compensation was recognized immediately as the ownership interests granted vested on issuance.

o) Income Tax

The Company and its members are legally organized as a limited liability company ("LLC") formed in accordance with the laws of Nevada and classified as a partnership for U.S. federal income tax purposes. Under this election, the members of the Company are solely liable for federal and state income taxes arising from the net income of the LLC. The Company has provided for and is liable for certain LLC-level state taxes based on gross revenue, and well as any partnership level withholding taxes.

The Company applies the provisions of Accounting for Uncertainty in Income Taxes under ASC 740. Under ASC 740, the Company utilizes a two-step approach to recognizing and measuring uncertain tax positions (tax contingencies). The first step is to evaluate the tax position for recognition by determining if the weight of available evidence indicates it is more likely than not that the position will be sustained on audit, including resolution of related appeals, or litigation processes. The second step is to measure the tax benefit as the largest amount, which is more than 50% likely of being realized upon ultimate settlement. There were no liabilities recorded for unrecognized tax benefits related to uncertain tax positions taken during the period from inception to November 30, 2020. The Company considers many factors when evaluating and estimating its tax positions and tax benefits, which may require periodic adjustments and which may not accurately forecast actual outcomes.

In accordance with ASC 740, the Company has elected to include interest and penalties related to its tax contingencies in income tax expense. There were no accruals for interest and penalties recognized for uncertain tax positions during the period from inception to November 30, 2020.

The Company files a U.S. federal Form 1065 tax return and a Nevada state tax return. The Company is not currently under examination by any taxing authority nor has it been notified of a pending examination.

RME Black 200, LLC Notes to the Financial Statements For the period from April 27, 2020 (inception) to November 30, 2020

3. Digital assets

Bitcoin	2020
Opening balance at April 27, 2020	<u>s — </u>
Revenue from Bitcoin mined	3,373,890
Bitcoin sold	(2,926,493)
Ending balance as of November 30, 2020	\$ 447,397

4. Digital assets receivable

The Company receives digital assets in exchange for the hash power provided by their mining equipment assets. The Company receives the respective reward daily and accrues any unremitted portions of digital assets earned. As of the balance sheet date, the undelivered digital assets earned is \$10,789.

5. Prepaid expenses

The Company had the following prepaid expenses as of November 30, 2020:

	Nov	vember 30, 2020
Prepaid hosting fees (see note 9)	\$	674,256
Total	\$	674,256

6. Mining equipment assets

As of November 30, 2020, the Company's mining equipment assets is comprised wholly of ASIC miners.

Cost	
Additions	\$ 6,283,692
Balance November 30, 2020	\$ 6,283,692
Accumulated depreciation	
Additions	\$ 523,641
Balance November 30, 2020	\$ 523,641
Net Book value	
Cost	\$ 6,283,692
Accumulated depreciation	(\$ 523,641)
Net mining equipment assets, November 30, 2020	\$ 5,760,051

7. Loan payable

On April 28, 2020, RME 200 entered into an equipment finance and security agreement in the amount of \$4,552,294, which bore interest at a rate of 16% per annum which is repayable monthly ending January 23, 2022. \$3,540,673 was due and payable as at November 30, 2020. The Company intends to pay down the balance of the loan within 12 months of November 30, 2020.

RME Black 200, LLC Notes to the Financial Statements For the period from April 27, 2020 (inception) to November 30, 2020

The Company incurred \$365,457 of interest during the period from April 27, 2020 (inception) to November 30, 2020 of which \$5,770 was accrued as of November 30, 2020.

The loan was secured by the mining equipment assets acquired by the Company.

8. Members' equity

During the period ended November 30, 2020, the Company completed the following transactions:

- On April 27, 2020, the Company issued 30 ownership units to the managing member to which a charge of \$1,157,143 was recorded against compensation expense.
- ii. Between April 27, 2020 and May 15, 2020, the Company raised a total of \$2,700,000 in exchange for a total of 70 ownership units.
- iii. On November 25, 2020 members withdrew \$100,003 of their initial investment of which \$18,519 remained payable as of November 30,

9. Related party transactions

Transactions with related parties

BEP 888, LLC, BEP 999, LLC, RME Black 100, LLC, RME Black 88, LLC, and RME 200 (The "Related Companies") have common minority ownership.

During the period from April 27, 2020 to November 30, 2020, the Company bought \$428,810 of digital assets from Related Companies. No balance remains unpaid as of November 30, 2020.

During the period from April 27, 2020 to November 30, 2020, the Company sold \$153,787 of digital assets from Related Companies. None of the purchased remained unpaid as of November 30, 2020.

RME 200 borrowed \$136,070 from Related Companies during the period to support the operations of the Company. \$27,626 remained outstanding as of November 30, 2020.

Entities with control, joint control, or significant influence

During the period ended November 30, 2020, the Company was charged \$165,908 of management and start-up fees from the managing member of which \$nil remained due and payable as at November 30, 2020.

To support the operations of the Company, short term loans totaling \$130,000 were received from the managing member during the period from April 27, 2020 to November 30, 2020. No balance remained payable as of November 30, 2020.

The Company made a payment of digital assets to the managing member during the period ended November 30, 2020. The fair value of the payment made on September 14, 2020 via transfer of digital asset was determined to be \$233,487. Of the total amount, \$108,442 represents a payment on behalf of RME 100, a Connected Company. The assessed amount was applied against the balance due to the managing member for finance and management fees leaving a receivable from the managing member of \$16,958.

The managing member is a shareholder of Core Scientific. During the period ended November 30, 2020, the Company purchased \$6,283,692 of mining equipment assets from Core Scientific and made payments totaling \$2,187,712 for hosting the mining equipment delivered and activated during the period. No balance remained unpaid as of November 30, 2020.

RME Black 200, LLC Notes to the Financial Statements For the period from April 27, 2020 (inception) to November 30, 2020

10. Commitments

The Company has various commitments for the acquisition and hosting of mining equipment assets from Core Scientific for which there are minimum future required payments as follows:

	As of November 30, 2020
Within one year	\$ 4,898,297
Later than one year but not later than five years	\$ 12,653,934
Total	\$ 17,552,231

The contracts with Core Scientific for hosting are at a fixed rate per miner based on power consumption and have a three-year term which automatically renews.

11. Subsequent events

The Company has evaluated subsequent events through July 30, 2021, the date the financial statements were issued, and no other matters required recognition or disclosure, other than the below, in the financial statements.

On December 1, 2020, Blockcap Inc. ("Blockcap") entered into an amendment and exchange agreement with the investors of RME 200 (the "Investors") in which the Investors agreed to exchange all their investor's ownership interests in RME 200 for a total of 18,779,411 Blockcap shares. As part of the agreement, the managing member voluntarily resigned and withdrew from RME 200.

On December 1, 2020, RME 200 entered into an assignment and assumption agreement with Blockcap under which Blockcap acquired all of the assets, liabilities, and commitments of RME 200. The financial statements include the operations of RME 200 up until December 1, 2020 at which time the assets and liabilities were acquired.



PART II

INFORMATION NOT REQUIRED IN PROSPECTUS

Item 13. Other Expenses of Issuance and Distribution.

The following table sets forth all costs and expenses, other than underwriting discounts and commissions, payable by us in connection with the sale of the securities being registered. All amounts shown are estimates except for the SEC registration fee.

	Amount
SEC registration fee	\$ 167,747.78
Accountants' fees and expenses	100,000.00
Legal fees and expenses	200,000.00
Printing fees	90,000.00
Miscellaneous	50,000.00
Total expenses	\$ 607,747.78

Discounts, concessions, commissions and similar selling expenses attributable to the sale of shares of common stock covered by this prospectus will be borne by the selling securityholders. We will pay all expenses (other than discounts, concessions, commissions and similar selling expenses) relating to the registration of the securities with the SEC, as estimated in the table above.

Item 14. Indemnification of Directors and Officers.

Section 145 of the Delaware General Corporation Law authorizes a court to award, or a corporation's board of directors to grant, indemnity to directors and officers in terms sufficiently broad to permit such indemnification under certain circumstances for liabilities, including reimbursement for expenses incurred, arising under the Securities Act of 1933, as amended.

Our amended and restated certificate of incorporation provides for indemnification of our directors, officers, employees and other agents to the maximum extent permitted by the Delaware General Corporation Law, and our amended and restated bylaws provide for indemnification of our directors, officers, employees and other agents to the maximum extent permitted by the Delaware General Corporation Law.

In addition, we have entered into indemnification agreements with our directors and executive officers containing provisions which are in some respects broader than the specific indemnification provisions contained in the Delaware General Corporation Law. The indemnification agreements require us, among other things, to indemnify our directors against certain liabilities that may arise by reason of their status or service as directors and to advance their expenses incurred as a result of any proceeding against them as to which they could be indemnified.

Item 15. Recent Sales of Unregistered Securities.

The following list sets forth information regarding all unregistered securities sold by XPDI and Legacy Core since XPDI's incorporation on December 29, 2020:

(1) In December 2020, XPDI Sponsor LLC, a Delaware limited liability company (the "Sponsor"), purchased 7,187,500 shares of Class B common stock of XPDI (the "Founder Shares") for an aggregate purchase price of \$25,000, or approximately \$0.003 per share. In January 2021, the Sponsor transferred 30,000 Founder Shares to each of Paul Dabbar, Paul J. Gaynor, Colleen Sullivan, and Scott Widham, XPDI's independent directors. XPDI effected a stock split of its Class B common stock on February 9, 2021, resulting in the Sponsor and the independent directors holding 8,625,000 Founder Shares. Immediately prior to the consummation of the Business Combination, the Sponsor sold 1,552,500 Founder Shares to the anchor investors for an aggregate

purchase price of \$4,500, or approximately \$0.003 per share. Upon the closing of the Business Combination, each share of XPDI Class B common stock automatically converted into a share of common stock of the post-combination company in accordance with the company's amended and restated certificate of incorporation, and all shares of Class B common stock were renamed common stock.

- (2) In February 2021, the Sponsor and the anchor investors purchased an aggregate of 8,625,000 Private Placement Warrants at a price of \$1.50 per warrant for an aggregate purchase price of \$12,937,500. Each whole Private Placement Warrant entitles the holder thereof to purchase one share of common stock of the post-combination company at a price of \$11.50 per share, subject to certain adjustments.
- (3) In April 2021, Legacy Core issued \$215.0 million of secured convertible notes entered into pursuant to a note purchase agreement dated as of April 19, 2021 to new and existing lenders. In August and September 2021, Legacy Core issued an additional \$203.3 million of unsecured convertible notes pursuant to a note purchase agreement dated as of August 20, 2021.
- (4) In June 2021, Legacy Core issued a secured demand promissory note in an aggregate principal amount of \$16.2 million to Blockcap for outstanding amounts owed pursuant to certain equipment invoices.
- (5) In July 2021, in connection with the Core/Blockcap merger, Legacy Core issued an aggregate amount of 81,032,304 shares of Legacy Core common stock to Blockcap stockholders in accordance with the exchange ratio set forth in the Core/Blockcap merger agreement.
- (6) Since December 29, 2020, Legacy Core has granted to certain employees, directors and service providers of Legacy Core and its subsidiaries (i) restricted stock units ("RSUs"), net of forfeitures and cancellations, representing shares of Core common stock, and (ii) stock options, net of forfeitures and cancellations, representing shares of Core common stock, under Core's 2018 Omnibus Incentive Plan. Upon the closing of the Business Combination, RSUs and stock options were deemed to have been exercised on a net exercise basis. The remaining outstanding RSUs were automatically assumed by Core and converted into RSUs to purchase an aggregate of shares of Core common stock, and the remaining outstanding stock options were automatically assumed by Core and converted into options to purchase an aggregate of shares of Core common stock.

None of the foregoing transactions involved any underwriters, underwriting discounts or commissions, or any public offering. We believe each of these transactions was exempt from registration under the Securities Act in reliance on Section 4(a)(2) of the Securities Act (and Regulation D promulgated thereunder) as transactions by an issuer not involving any public offering or Rule 701 promulgated under Section 3(b) of the Securities Act as transactions by an issuer under benefit plans and contracts relating to compensation as provided under Rule 701. The recipients of the securities in each of these transactions represented their intentions to acquire the securities for investment only and not with a view to or for sale in connection with any distribution thereof, and appropriate legends were placed on the share certificates issued in these transactions. All recipients had adequate access, through their relationships with us, to information about us. The sales of these securities were made without any general solicitation or advertising.

Item 16. Exhibits and Financial Statement Schedules.

(a) Exhibits.

The exhibits listed below are filed as part of this registration statement.

		Incorporated by Reference			
Exhibit	<u>Description</u>	Schedule/ Form	File Number	Exhibits	Filing Date
2.1†	Agreement and Plan of Merger and Reorganization, dated as of July 20, 2021, by and among Power & Digital Infrastructure Acquisition Corp., XPDI Merger Sub Inc., XPDI Merger Sub 2, LLC, and Core Scientific Holding Co.	S-4	333-258720	2.1	August 11, 2021
2.2†	First Amendment to Agreement and Plan of Merger and Reorganization, dated as of October 1, 2021, by and among Power & Digital Infrastructure Acquisition Corp., XPDI Merger Sub Inc., XPDI Merger Sub 2, LLC, and Core Scientific Holding Co.	S-4/A	333-258720	2.2	October 4, 2021
2.3†	Second Amendment to Agreement and Plan of Merger and Reorganization, dated as of December 29, 2021, by and among Power & Digital Infrastructure Acquisition Corp., XPDI Merger Sub Inc., and Core Scientific Holding Co.	S-4/A	333-258720	2.3	December 30, 2021
3.1	Second Amended and Restated Certificate of Incorporation of Core Scientific, Inc.	8-K	001-40046	3.1	January 24, 2022
3.2	Second Amended and Restated Bylaws of Core Scientific, Inc.	8-K	001-40046	3.2	January 24, 2022
4.1**	Specimen Common Stock Certificate.				
4.2**	Specimen Warrant Certificate.				
4.3	Warrant Agreement, dated as of February 9, 2021, by and between Power & Digital Infrastructure Acquisition Corp. and Continental Stock Transfer & Trust Company.	8-K	001-40046	4.1	February 12, 2021
4.4	Assignment, Assumption and Amendment Agreement, dated as of January 18, 2022, by and among Power & Digital Infrastructure Acquisition Corp., Core Scientific Holding Co., Continental Stock Transfer & Trust Company, Computershare Inc. and Computershare Trust Company, N.A.	8-K	001-40046	4.2	January 24, 2022

		Incorporated by Reference			ence
Exhibit	<u>Description</u>	Schedule/ Form	File Number	Exhibits	Filing Date
4.5*†	Secured Convertible Note Purchase Agreement, dated as of April 19, 2021, by and among Core Scientific Holding Co., the Guarantors thereto, the Purchasers thereto and U.S. Bank National Association as note agent and collateral agent.				
4.6*	First Amendment to Secured Convertible Note Purchase Agreement, dated as of April 22, 2021, by and among Core Scientific Holding Co., the Guarantors thereto, the Purchasers thereto and U.S. Bank National Association as note agent and collateral agent.				
4.7*	Form of Secured Convertible Promissory Note (included in Exhibit 4.5).				
4.8†	Convertible Note Purchase Agreement, dated as of August 20, 2021, by and among Core Scientific Holding Co., the Guarantors thereto, the Purchasers thereto and U.S. Bank National Association as note agent and collateral agent.	S-4/A	333-258720	4.7	November 19, 2021
4.9	First Amendment to Convertible Note Purchase Agreement, dated as of September 23, 2021, by and among Core Scientific Holding Co., the Guarantors thereto, the Purchasers thereto and U.S. Bank National Association as note agent and collateral agent.	S-4/A	333-258720	4.8	November 19, 2021
4.10	Form of Convertible Promissory Note (included in Exhibit 4.8).	S-4/A	333-258720	4.9	November 19, 2021
5.1**	Opinion of Cooley LLP.				
10.1	Amended and Restated Registration Rights Agreement, dated as of January 19, 2022, by and among Power & Digital Infrastructure Acquisition Corp., XPDI Sponsor LLC, Core Scientific Holding Co., and other parties thereto.	8-K	001-40046	10.2	January 24, 2022
10.2	Form of Lock-Up Agreement of Core Scientific, Inc.	S-4/A	333-258720	10.23	December 30, 2021
10.3	Sponsor Agreement, dated as of July 20, 2021, among Power & Digital Infrastructure Acquisition Corp., XPDI Sponsor LLC and the other parties thereto.	S-4	333-258720	10.1	August 11, 2021
10.4	Form of Company Support Agreement.	S-4	333-258720	10.2	August 11, 2021

			Incorporated by Reference		
Exhibit	<u>Description</u>	Schedule/ Form	File Number	Exhibits	Filing Date
10.5#	Amended and Restated Employment Agreement, by and between Core Scientific Holding Co. and Michael J. Levitt, dated October 10, 2021.	S-4/A	333-258720	10.28	November 19, 2021
10.6#	Employment Agreement by and between Darin Feinstein and Core Scientific Holding Co., dated October 10, 2021.	S-4/A	333-258720	10.29	November 19, 2021
10.7#	Letter Agreement, by and between Core Scientific, Inc. and Michael Trzupek, dated September 14, 2020.	S-4	333-258720	10.15	August 11, 2021
10.8#	Letter Agreement, by and between Core Scientific, Inc. and Todd DuChene, dated December 15, 2018.	S-4	333-258720	10.16	August 11, 2021
10.9#	Form of Indemnification Agreement of Core Scientific, Inc.	S-4/A	333-258720	10.30	November 19, 2021
10.10#	Core Scientific, Inc. 2021 Equity Incentive Plan.	8-K	001-40046	10.8	January 24, 2022
10.11#	Form of Restricted Stock Unit Award Agreement underlying the Core Scientific, Inc. 2021 Equity Incentive Plan.	S-4/A	333-258720	10.9	October 4, 2021
10.12#	Form of Stock Option Agreement underlying the Core Scientific, Inc. 2021 Equity Incentive Plan.	S-4/A	333-258720	10.27	October 4, 2021
10.13#	Core Scientific, Inc. 2021 Employee Stock Purchase Plan.	8-K	001-40046	10.11	January 24, 2022
10.14#	Non-Employee Director Compensation Policy of Core Scientific, Inc.	S-4/A	333-258720	10.14	November 19, 2021
10.15+	Industrial Power Contract by and between Murphy Electric Power Board and BCV 77, LLC, dated December 15, 2017, as assigned and assumed on February 19, 2018.	S-4	333-258720	10.3	August 11, 2021
10.16†	Interruptible Power Product Agreement by and between Murphy Electric Power Board and Core Scientific Holding Co., dated August 30, 2018.	S-4	333-258720	10.4	August 11, 2021
10.17+	Investment Credit Agreement by and among Core Scientific Holding Co., Murphy Electric Power Board and the Tennessee Valley Authority, dated October 10, 2018.	S-4	333-258720	10.5	August 11, 2021
10.18+	Master Services Agreement by and between Core Scientific Holding Co. and Duke Energy Carolinas, LLC, dated June 25, 2018.	S-4	333-258720	10.6	August 11, 2021

		Incorporated by Reference			ce
Exhibit	<u>Description</u>	Schedule/ Form	File Number	Exhibits	Filing Date
10.19†+	Electric Service Agreement by and between Core Scientific Holding Co. and Duke Energy Carolinas, LLC, dated June 10, 2019.	S-4	333-258720	10.7	August 11, 2021
10.20†+	American Property Acquisitions VII, LLC and The Board of Water, Light and Sinking Fund Commissioners of the City of Dalton, Georgia, dated October 11, 2018 (Industrial South Premises).	S-4	333-258720	10.8	August 11, 2021
10.21†+	Amended and Restated Electric Service Agreement by and between American Property Acquisitions VII, LLC and The Board of Water, Light and Sinking Fund Commissioners of the City of Dalton, Georgia, dated October 11, 2018 (Boring Drive Property).	S-4	333-258720	10.9	August 11, 2021
10.22†+	Firm Power Contract by and between Core Scientific Holding Co. and the Tennessee Valley Authority, dated March 12, 2019, as amended on April 30, 2020 and February 25, 2021.	S-4	333-258720	10.10	August 11, 2021
10.23	Interruptible Power Product Agreement by and between Core Scientific Holding Co. and the Tennessee Valley Authority, dated April 28, 2020.	S-4	333-258720	10.11	August 11, 2021
10.24	Form of Sales and Purchase Agreement by and between Core Scientific and Bitmain Technologies Limited and affiliates.	S-4	333-258720	10.12	August 11, 2021
16.1	Letter from Marcum LLP.	8-K	001-40046	16.1	January 24, 2022
21.1	<u>List of Subsidiaries.</u>	8-K	001-40046	21.1	January 24, 2022
23.1*	Consent of Marcum LLP, independent registered public accounting firm.				
23.2*	Consent of Ernst & Young LLP, independent registered public accounting firm.				
23.3*	Consent of independent registered public accounting firm of Blockcap, Inc.				
23.4*	Consent of independent registered public accounting firm of BEP 888, LLC.				
23.5*	Consent of independent registered public accounting firm of BEP 999, LLC.				
23.6*	Consent of independent registered public accounting firm of RME Black 100, LLC.				

		Incorporated by Reference			
Exhibit	Description	Schedule/ Form	File <u>Number</u>	Exhibits	Filing Date
23.7*	Consent of independent registered public accounting firm of RME Black 200, LLC.				
23.8**	Consent of Cooley LLP (included in Exhibit 5.1).				
24.1*	Power of Attorney (included on signature page).				
101.INS	Inline XBRL Instance Document.				
101.CAL	Inline XBRL Taxonomy Extension Calculation Linkbase Document.				
101.SCH	Inline XBRL Taxonomy Extension Schema Document.				
101.DEF	Inline XBRL Taxonomy Extension Definition Linkbase Document.				
101.LAB	Inline XBRL Taxonomy Extension Labels Linkbase Document.				
101.PRE	Inline XBRL Taxonomy Extension Presentation Linkbase Document.				
104	Cover Page Interactive Data File (embedded within the Inline XBRL document).				
107*	<u>Filing Fee Table</u>				

 ^{*} Filed herewith.

Item 17. Undertakings.

- (a) The undersigned registrant hereby undertakes as follows:
 - (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
 - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act of 1933;
 - (ii) To reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent posteffective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price

^{**} To be filed by amendment.

[#] Indicates a management contract or compensatory plan, contract or arrangement.

[†] Certain of the exhibits and schedules to these exhibits have been omitted in accordance with Regulation S-K Item 601(a)(5). The registrant agrees to furnish a copy of all omitted exhibits and schedules to the SEC upon its request.

⁺ Portions of this Exhibit (indicated with [***]) have been omitted as the Registrant has determined that (i) the omitted information is not material and (ii) the omitted information would likely cause competitive harm to the Registrant if publicly disclosed.

- represent no more than 20 percent change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement;
- (iii) To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement.
- (2) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (4) That, for the purpose of determining liability under the Securities Act of 1933 to any purchaser, each prospectus filed pursuant to Rule 424(b) as part of a registration statement relating to an offering, other than registration statements relying on Rule 430B or other than prospectuses filed in reliance on Rule 430A, shall be deemed to be part of and included in the registration statement as of the date it is first used after effectiveness. Provided, however, that no statement made in a registration statement or prospectus that is part of the registration statement or made in a document incorporated or deemed incorporated by reference into the registration statement or prospectus that is part of the registration statement will, as to a purchaser with a time of contract of sale prior to such first use, supersede or modify any statement that was made in the registration statement or prospectus that was part of the registration statement or made in any such document immediately prior to such date of first use.
- (5) That, for the purpose of determining any liability under the Securities Act of 1933 to any purchaser in the initial distribution of the securities, the undersigned registrant undertakes that in a primary offering of securities of the undersigned registrant pursuant to this registration statement, regardless of the underwriting method used to sell the securities to the purchaser, if the securities are offered or sold to such purchaser by means of any of the following communications, the undersigned registrant will be a seller to the purchaser and will be considered to offer or sell such securities to such purchaser:
 - (i) Any preliminary prospectus or prospectus of the undersigned registrant relating to the offering required to be filed pursuant to Rule 424:
 - (ii) Any free writing prospectus relating to the offering prepared by or on behalf of the undersigned registrant or used or referred to by the undersigned registrant;
 - (iii) The portion of any other free writing prospectus relating to the offering containing material information about the undersigned registrant or our securities provided by or on behalf of the undersigned registrant; and
 - (iv) Any other communication that is an offer in the offering made by the undersigned registrant to the purchaser.
- (b) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the undersigned pursuant to the foregoing provisions, or otherwise, the undersigned has been advised that in the opinion of the SEC such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the undersigned of expenses incurred or paid by a director, officer or controlling person of the undersigned in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the undersigned will, unless in the opinion of our counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the registrant has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized in the City of Austin, State of Texas, on this 8th day of February, 2022.

CORE SCIENTIFIC, INC.

By: /s/ Michael Levitt

Michael Levitt

Chief Executive Officer and Co-Chair of the Board of

Directors

POWER OF ATTORNEY

KNOW ALL BY THESE PRESENTS, that each person whose signature appears below hereby constitutes and appoints Michael Levitt, Michael Trzupek and Todd DuChene and each of them, as his or her true and lawful agents, proxies and attorneys-in-fact, with full power of substitution and resubstitution, for him or her and in his or her name, place and stead, in any and all capacities, to (i) act on, sign and file with the Securities and Exchange Commission any and all amendments (including post-effective amendments) to this registration statement together with all schedules and exhibits thereto and any subsequent registration statement filed pursuant to Rule 462(b) under the Securities Act of 1933, as amended, together with all schedules and exhibits thereto, (ii) act on, sign and file such certificates, instruments, agreements and other documents as may be necessary or appropriate in connection therewith, (iii) act on and file any supplement to any prospectus included in this registration statement or any such amendment or any subsequent registration statement filed pursuant to Rule 462(b) under the Securities Act of 1933, as amended, and (iv) take any and all actions which may be necessary or appropriate to be done, as fully for all intents and purposes as he might or could do in person, hereby approving, ratifying and confirming all that such agent, proxy and attorney-in-fact or any of his substitutes may lawfully do or cause to be done by virtue thereof.

Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following persons in the capacities and on the dates indicated.

Signature	Title	Date
/s/ Michael Levitt	Chief Executive Officer and	February 8, 2022
Michael Levitt	Co-Chair of the Board	
	(Principal Executive Officer)	
/s/ Michael Trzupek	Executive Vice President and	February 8, 2022
Michael Trzupek	Chief Financial Officer	
	(Principal Financial Officer)	
/s/ Brian Neville	Chief Accounting Officer	February 8, 2022
Brian Neville	(Principal Accounting Officer)	
/s/ Darin Feinstein	Chief Vision Officer and Co-Chair of the	February 8, 2022
Darin Feinstein	Board	
/s/ Jarvis Hollingsworth	Director	February 8, 2022
Jarvis Hollingsworth		

Signature		Title	Date
/s/ Matt Minnis Matt Minnis	_ Director		February 8, 2022
/s/ Stacie Olivares Stacie Olivares	Director		February 8, 2022
/s/ Kneeland Youngblood Kneeland Youngblood	Director		February 8, 2022

SECURED CONVERTIBLE NOTE PURCHASE AGREEMENT

THIS SECURED CONVERTIBLE NOTE PURCHASE AGREEMENT (this "Agreement") is made as of April 19, 2021, by and among Core Scientific Holding Co., a Delaware corporation (the "Company"), the Guarantors from time to time party hereto, the persons and entities named on the Schedule of Purchasers under the header 'Initial Purchasers' attached hereto as Schedule 2 (individually, an "Initial Purchaser" and collectively, the "Initial Purchasers"), each Additional Purchaser from time to time party hereto and U.S. Bank National Association, as note agent (in such capacity, the "Note Agent") and as collateral agent for the Secured Parties (in such capacity, the "Collateral Agent" and, together with the Note Agent, the "Agents").

RECITALS

WHEREAS, the Company desires (i) to issue and to sell to the Initial Purchasers, and the Initial Purchasers have agreed to purchase from the Company, on the Initial Closing Date, senior secured convertible promissory notes substantially in the form attached hereto as Exhibit A (each, an "Initial Note" and collectively, the "Initial Notes") and (ii) to issue and to sell to Additional Purchasers on Additional Closing Dates senior secured convertible promissory notes substantially in the form attached hereto as Exhibit A (each, an "Additional Note" and collectively, the "Additional Notes" and the Initial Notes, together with the Initial Notes, the "Notes"), in an aggregate principal amount for all such Notes of up to \$215,000,000 on the terms, and subject to the conditions, specified herein; and

WHEREAS, to induce the Purchasers to purchase the Notes, the Note Parties have agreed to execute and deliver, contemporaneously with the issuance, sale and purchase of the Initial Notes on the Initial Closing Date, (i) a Guaranty, pursuant to which each Guarantor will guarantee the Obligations of the Company under the Notes and (ii) a Security Agreement, pursuant to which the Note Parties will grant to the Collateral Agent, for the ratable benefit of the Secured Parties, a security interest in the Note Parties' assets to secure the Obligations or their respective Guaranteed Obligations (as applicable).

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing, and the representations, warranties, covenants and conditions set forth below, which represent integral components of the transactions contemplated hereby and shall be fully enforceable by the parties hereto, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Note Parties, the Collateral Agent and each Purchaser, intending to be legally bound, hereby agree as follows:

1. DEFINITIONS.

As used herein, the following terms shall have the meanings set forth below.

(a) "Act" means the Securities Exchange Act of 1934, as amended.

- (b) "Additional Closing Date" has the meaning ascribed thereto in Section 4.2.
- (c) "Additional Note" has the meaning ascribed thereto in Section 2.1.
- (d) "Additional Purchaser" has the meaning ascribed thereto in Section 4.2.
- (e) "Administrative Questionnaire" means an Administrative Questionnaire substantially in the form attached hereto as

Exhibit G.

- (f) "Agent-Related Person" means the Note Agent and the Collateral Agent, together with their affiliates, officers, directors, employees, agents and attorneys-in-fact of the Agents and their affiliates.
 - (g) "Agents" has the meaning ascribed to such term in the preamble to this Agreement.
- (h) "Asset Disposition" means any sale, lease, license, transfer, assignment or other disposition (whether by a single transaction or through series of transactions, and including by means of a merger, consolidation, amalgamation or similar transaction) by the Company or any of its Subsidiaries of any asset or property.
- (i) "Business Day" means a day other than a Saturday, Sunday or other day on which commercial banks in the City of New York or the Principal Office are authorized or required to close.
- (j) "Capital Lease" means any lease of any property by such Person as lessee which would, in accordance with GAAP, be required to be accounted for as a capital lease on the balance sheet of such Person.
 - (k) "Capital Lease Obligations" means, with respect to any Person, all obligations of such Person under Capital Leases.
- (l) "Capital Stock" means (a) any capital stock, partnership, membership, limited liability company, joint venture or other ownership or equity interest or other equivalent, participation or securities (whether voting or non-voting, whether preferred, common or otherwise, whether certificated or uncertificated, and however designated), and (b) any option, warrant, security, appreciation right, profits interests or other right directly or indirectly convertible into or exercisable or exchangeable for, or otherwise to acquire directly or indirectly, any capital stock, partnership, membership, limited liability company, joint venture or other ownership or equity interest, participation or security described in clause (a) above (excluding the Notes and any other Indebtedness convertible or exchangeable into any such capital stock, partnership, membership, limited liability company, joint venture or other ownership or equity interest, participation or security described in clause (a) above unless and to the extent converted or exchangead).
- (m) "Charter Documents" means (i) the articles or certificate of incorporation or formation (as applicable), (ii) the by-laws, operating agreement or limited liability company agreement (as applicable) and (iii) other similar organizational and governing documents of any Person, as amended, restated, supplemented or otherwise modified from time to time.

- (n) "Closing Date" means the Initial Closing Date or each Additional Closing Date, as the context may require.
- (o) "Collateral" means the "Collateral" as defined in the Security Agreement and.
- (p) "Collateral Agent" has the meaning ascribed to such term in the preamble to this Agreement.
- (q) "Collateral Documents" means, collectively, the Security Agreement, the Intellectual Property Security Agreements and each other agreement or writing pursuant to which the Note Parties purport to pledge or grant a security interest in any property or assets securing the Obligations or the Guaranteed Obligations, in each case, as amended, restated, supplemented or otherwise modified from time to time.
- (r) "Company Covered Persons" means those persons specified in Rule 506(d)(1) promulgated under the Act; <u>provided</u>, <u>however</u>, that Company Covered Persons do not include (a) any Holder or (b) any person or entity that is deemed to be an affiliated issuer of the Company solely as a result of the relationship between the Company and any Holder.
- (s) "Contractual Obligations" means as to any Person, any provision of any security issued by such Person or of any agreement, undertaking, contract, indenture, mortgage, deed of trust or other instrument or arrangement (whether in writing or otherwise) to which such Person is a party or by which it or any of such Person's property is bound.
 - (t) "Conversion Event" has the meaning ascribed to such term in the Notes.
 - (u) "Conversion Securities" has the meaning ascribed to such term in Section 5.3.
- (v) "Default" means any condition or event which with the giving of notice or lapse of time or both would, unless cured or waived, become an Event of Default.
- (w) "Disqualified Capital Stock" means any Capital Stock issued by any Person that (a) matures or is mandatorily redeemable pursuant to a sinking fund obligation or otherwise, (b) is or may become redeemable or repurchaseable by such Person at the option of the holder thereof, in whole or in part (other than in connection with an asset sale, change of control or similar event) or (c) is convertible or exchangeable at the option of the holder thereof for Indebtedness or Capital Stock described in this definition, on or prior to, in the case of clause (a), (b) or (c), ninety (90) days after the Maturity Date (as defined in the Notes).
- (x) "Environmental Laws" means any and all federal, state, local and foreign statutes, laws, judicial decisions, regulations, ordinances, rules, judgments, orders, decrees, plans, injunctions, Licenses, concessions, grants, franchises, agreements and other governmental restrictions relating to (i) the protection of the environment, (ii) the effect of the environment on

human health, (iii) emissions, discharges or releases of pollutants, contaminants, hazardous substances or wastes into surface water, ground water, air or land, or (iv) the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of pollutants, contaminants, hazardous substances or wastes or the clean-up or other remediation thereof, including, without limitation, the Clean Air Act, 42 U.S.C. § 7401 et seq., the Clean Water Act, 33 U.S.C. § 1251 et seq. ("*CWA*"), the Solid Waste Disposal Act (as amended by the Resource Conservation and Recovery Act), 42 U.S.C. § 6901 et seq. ("*RCRA*") and the Comprehensive Environmental Response, Compensation, and Liability Act, 42 U.S.C. § 9601 et seq. ("*CERCLA*").

- (y) "ERISA" means the Employee Retirement Income Security Act of 1974.
- (z) "ERISA Affiliate" means, any person that for purposes of Title I or Title IV of ERISA or Section 412 of the Internal Revenue Code would be deemed at any relevant time to be a "single employer" or otherwise aggregated with the Company or any of its Subsidiaries under Section 414(b), 414(c), 414(m) or 414(o) of the Internal Revenue Code or Section 4001 of ERISA.
 - (aa) "Event of Default" has the meaning ascribed thereto in the Notes.
- (bb) "Excluded Subsidiary" means (i) any Subsidiary that is not a wholly-owned Subsidiary, (ii) any Subsidiary that is not a Material Subsidiary and (iii) any Subsidiary that is (a) a Foreign Subsidiary (as defined in the Security Agreement), (b) a CFC Holdco (as defined in the Security Agreement) or (c) a Subsidiary of a CFC or a CFC Holdco (as defined in the Security Agreement); provided that notwithstanding the foregoing clauses (i) through (iii), the Company may in its sole discretion designate any Excluded Subsidiary as a Guarantor.
- (cc) "Existing Credit Agreement" means that certain Credit and Guaranty Agreement, dated as of May 19, 2020, by and among the Company, Core Scientific, Inc., as borrower, the Subsidiaries (as defined therein) of Core Scientific, Inc. from time to time party thereto, the lenders from time to time party thereto and Silverpeak Credit Partners LP, as administrative agent and as collateral agent.
- (dd) "GAAP" means the United States generally accepted accounting principles in effect as of the date of determination thereof. GAAP will be deemed to treat operating leases and Capital Leases in a manner consistent with the treatment under GAAP as in effect prior to the issuance by the Financial Accounting Standards Board on February 25, 2016 of Accounting Standards Update No. 2016-02.
- (ee) "Governmental Authority" means the government of the United States of America or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supranational bodies such as the European Union or the European Central Bank).
- (ff) "Guarantee" of or by any Person (the "guarantor") means any obligation, contingent or otherwise, of the guarantor guaranteeing or having the economic effect of

guaranteeing any Indebtedness or other obligation of any other Person (the "primary obligor") in any manner, whether directly or indirectly and including any obligation, direct or indirect, of the guarantor (a) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation or to purchase (or to advance or supply funds for the purchase of) any security for the payment thereof, (b) to purchase or lease property, securities or services for the purpose of assuring the owner of such Indebtedness or other obligation of the payment thereof, (c) to maintain working capital, equity capital or any other financial statement condition or liquidity of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other obligation or (d) as an account party in respect of any letter of credit or letter of guaranty issued in support of such Indebtedness or obligation; provided that the term "Guarantee" shall not include endorsements for collection or deposit in the ordinary course of business. The amount of any Guarantee shall be deemed to be an amount equal to the stated or determinable amount of the primary obligation in respect of which such Guarantee is made or, if not so stated or determinable, the maximum reasonably anticipated liability in respect thereof (assuming such Person is required to perform thereunder) as determined by such Person in good faith.

- (gg) "Guaranteed Obligations" has the meaning ascribed thereto in the Guaranty.
- (hh) "Guarantor" means each Subsidiary of the Company party to the Guaranty and each other Subsidiary of a Note Party that becomes a Guarantor under the Note Documents pursuant to Section 6.10.
- (ii) "Guaranty" means the Guaranty, dated as the date hereof, made by the Guarantors in favor of the Collateral Agent, substantially in the form attached hereto as Exhibit C. (as amended, restated, supplemented and/or otherwise modified from time to time).
- (jj) "Hazardous Materials" means (i) any "hazardous substance", as defined by CERCLA, (ii) any "hazardous waste", as defined by RCRA, (iii) any petroleum product, (iv) any "pollutant," as defined by the CWA, or (v) contaminant or hazardous, dangerous or toxic chemical, material or substance within the meaning of any other Environmental Law.
 - (kk) "Holders" means the holders of the Notes, including the Purchasers. "Holder" means any of the Holders, individually.
- (II) "Indebtedness" as applied to any Person, means, without duplication, (a) all obligations of such Person for borrowed money, (b) all obligations of such Person evidenced by bonds, debentures, notes or other similar instruments, (c) all obligations of such Person in respect of the deferred purchase price of property or services (other than trade payables and accounts payable incurred in the ordinary course of business and any deferred compensation, earn-out, purchase price adjustment or other deferred purchase price obligation which is contingently payable based on the achievement of future financial performance); (d) all obligations of such Person under any conditional sale or other title retention agreement(s) relating to property acquired by such Person, (e) all obligations in respect of Capital Leases of such Person, (f) all obligations, contingent or otherwise, of such Person in respect of letters of credit, banker's acceptances or similar extensions of credit, (g) all Guarantees of such Person of

Indebtedness of other Persons that would count as a liability on the balance sheet of such person in accordance with the GAAP, (h) all Indebtedness of a third party secured by any Lien on any property or asset owned or held by such Person, whether or not such Indebtedness has been assumed by such Person, (i) all obligations of such Person, contingent or otherwise, to purchase, redeem, retire or otherwise acquire for value any Disqualified Capital Stock of such Person and (j) all monetary obligations under any receivables factoring, receivables sales, receivable securitization or similar transactions or other Off-Balance Sheet Liabilities. The Indebtedness of any Person shall include the Indebtedness of any partnership or joint venture in which such Person is a general partner or a joint venturer, except to the extent that the terms of such Indebtedness provide that such Person is not liable therefor.

- (mm) "Initial Closing Date" means April 19, 2021.
- (nn) "Initial Note" has the meaning ascribed thereto in Section 2.1.
- (oo) "Initial Purchaser" has the meaning ascribed thereto in the preamble to this Agreement.
- (pp) "Intellectual Property" means all intellectual and similar property of a Person, including inventions, designs, patents, copyrights, trademarks, service marks, trade names, trade secrets, confidential or proprietary information, customer lists, know-how, software and databases.
 - (qq) "Intellectual Property License" has the meaning assigned thereto in the Security Agreement.
- (rr) "Intellectual Property Security Agreement" means the Intellectual Property Security Agreement, dated as the date hereof, made by Core Scientific, Inc. in favor of the Collateral Agent, substantially in the form attached hereto as Exhibit E (as amended, restated, supplemented and/or otherwise modified from time to time).
 - (ss) "Intercreditor Agreement" has the meaning ascribed thereto in Section 7.1(a).
 - (tt) "Ledger" shall have the meaning ascribed thereto in the Notes.
- (uu) "Licenses" means all licenses, permits, authorizations, determinations, and registrations issued by any Governmental Authority to the Company or any Subsidiary in connection with the conduct of its business.
- (vv) "Lien" means any lien, mortgage, pledge, assignment, security interest, charge or encumbrance of any kind (including any agreement to give any of the foregoing, any conditional sale or other title retention agreement, and any lease in the nature thereof) and any option, trust or other preferential arrangement having the practical effect of any of the foregoing.
- (ww) "Material Adverse Effect" means, individually or in the aggregate, (a) a material adverse change in, or a material adverse effect upon, the operations, business, properties or financial condition of the Company and its Subsidiaries taken as a whole; (b) a material

impairment of the rights and remedies of the Collateral Agent or the Purchasers under the Note Documents, or of the ability of the Note Parties to perform its obligations under the Note Documents; (c) a material adverse effect upon the legality, validity, binding effect or enforceability against the Note Parties of any Note Document; or (d) a material adverse effect on the Liens in favor of the Collateral Agent (for the benefit of the Secured Parties) on the Collateral or the priority of such Liens.

- (xx) "*Material Subsidiary*" mean each Subsidiary which, as of the most recent fiscal quarter of the Company, for the period of four consecutive fiscal quarters of the Company then last ended (taken as one accounting period) in respect of which financial statements were (or were required to be) delivered pursuant to <u>Sections 6.1(a)</u> or <u>(b)</u>, as applicable (a "*Test Period*"), had Total Assets as of the last day of such Test Period that were in excess of 10% of the Total Assets of the Company and its Subsidiaries as of such date.
- (yy) "Multiemployer Plan" means any "multiemployer plan" as defined in Section 4001(a)(3) of ERISA, which is contributed to by (or to which there is or may be an obligation to contribute of) the Company, any of its Subsidiaries or an ERISA Affiliate, and each such plan for the five-year period immediately following the latest date on which the Company, any of its Subsidiaries or an ERISA Affiliate contributed to or had an obligation to contribute to such plan.
 - (zz) "Note" or "Notes" has the meaning ascribed thereto in the recital to this Agreement.
- (aaa) "Note Documents" means, collectively, this Agreement, each Note, the Guaranty, each Collateral Document, the fee schedule delivered from the Agents to the Company in connection with this Agreement, any Intercreditor Agreement, and any other document, agreement or instrument which has or will be executed by or for the benefit of the Secured Parties in connection with such agreements and the transactions described therein, each as may be amended, restated, supplemented/and or otherwise modified from time to time.
 - (bbb) "Note Party" or "Note Parties" means, individually and collectively, the Company and the Guarantors.
- (ccc) "Obligations" means all advances to, and debts, liabilities (including without limitation (x) prepayment premiums (if any) and other premiums payable under the Notes (any), including all or any portion of the Covenant Amendment Repayment Amount (as defined in the Notes) or the Repayment Amount (as defined in the Notes), if applicable, (y) accrued and unpaid interest and (z) capitalized interest), obligations, fees, expenses, indemnities, covenants and duties of, the Note Parties arising under any Note Document or otherwise with respect to any Note, whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against the Note Parties of any proceeding under any debtor relief laws naming any Note Party as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims or allowable in such proceeding.

- (ddd) "OFAC" means the U.S. Department of the Treasury's Office of Foreign Assets Control.
- (eee) "PBGC" mean the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA (or any successor thereto).
- (fff) "Plan" means any "employee benefit plan" as defined in Section 3 of ERISA (other than a Multiemployer Plan) maintained or contributed to by any Note Party or any ERISA Affiliate or to which a Note Party or any ERISA Affiliate has or may have an obligation to contribute, and each such plan that is subject to Title IV of ERISA for the five-year period immediately following the latest date on which a Note Party or any ERISA Affiliate maintained, contributed to or had an obligation to contribute to (or is deemed under Section 4069 of ERISA to have maintained or contributed to or to have had an obligation to contribute to, or otherwise to have liability with respect to) such plan.

(ggg) "Permitted Lien" means:

- (i) Liens in favor of Collateral Agent for the benefit of Secured Parties granted pursuant to the Note Documents;
- (ii) Liens set forth on Schedule 5.17 existing as of the Initial Closing Date;
- (iii) Liens for taxes if obligations with respect to such taxes either (i) are not due and payable, (ii) are being contested in good faith by appropriate proceedings promptly instituted and diligently conducted and adequate reserves have been made in accordance with GAAP or (iii) could not reasonably be excepted to have a Material Adverse Effect;
- (iv) statutory Liens of landlords, banks (and rights of set-off), of carriers, warehousemen, mechanics, repairmen, workmen and materialmen, and other Liens imposed by Law (other than any such Lien imposed pursuant to Section 401 (a)(29) or 412(n) of the Internal Revenue Code or by the Employee Retirement Income Security Act of 1974), in each case incurred in the ordinary course of business (i) for amounts not yet overdue, or (ii) for amounts that are overdue and that (in the case of any such amounts overdue for a period in excess of five (5) days) are being contested in good faith by appropriate proceedings, so long as such reserves or other appropriate provisions, if any, as shall be required by GAAP shall have been made for any such contested amounts;
- (v) Liens incurred in the ordinary course of business in connection with workers' compensation, unemployment insurance and other types of social security, or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, government contracts, trade contracts, performance and return-of-money bonds and other similar obligations (exclusive of obligations for the payment of borrowed money or other indebtedness), so long as (x) no foreclosure, sale or similar proceedings have been commenced with respect to any portion of the Collateral on account thereof and (y) adequate reserves required by GAAP shall have been set aside therefor;

EXECUTION VERSION

- (vi) easements, rights-of-way, restrictions, encroachments, and other minor defects or irregularities in title, in each case which do not and will not have a Material Adverse Effect;
 - (vii) any interest or title of a lessor or sublessor under any lease of real estate permitted hereunder;
- (viii) Liens solely on any cash earnest money deposits made by the Company in connection with any letter of intent or purchase agreement permitted hereunder;
- (ix) purported Liens evidenced by the filing of precautionary UCC financing statements relating solely to operating leases of personal property entered into in the ordinary course of business;
- (x) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods used in the ordinary course of business;
- (xi) any zoning or similar Law or right reserved to or vested in any governmental office or agency to control or regulate the use of any real property;
- (xii) licenses of patents, trademarks and other Intellectual Property rights granted by the Company in the ordinary course of business;
- (xiii) Liens securing obligations of the Company incurred under leases (but not securing indebtedness for borrowed money) in the form of cash deposits made in the ordinary course of business;
- (xiv) Liens arising in favor of depository institutions as a result of set-off rights or otherwise securing obligations owed to such depository institution in connection with bank services provided to the Company or its Subsidiaries;
- (xv) Liens on cash deposits securing obligations owed to an issuer of a letter of credit at the request of the Company of any of its Subsidiaries;
- (xvi) Liens on any assets of the Note Parties securing any Indebtedness, including Capital Lease Obligations, incurred to finance the acquisition of any fixed or capital assets (including any real property assets) so long as such Liens encumber only the assets acquired or financed with the proceeds of such Indebtedness and do not attach to any other assets of any Note Party;
 - (xvii) Liens on cash deposit accounts and cash and cash equivalents delivered or pledged to secure letters of credit;
 - (xviii) Liens on Indebtedness permitted to be incurred in reliance on Section 7.01(a); and

- (xix) Liens arising from judgments for the payment of money in circumstances not constituting an Event of Default.
- (hhh) "Permitted Transferees" means, as to any Purchaser that is a limited or general partnership or a trust, to its partners (whether general or limited, including retired partners) or beneficiaries and to affiliated partnerships managed by the same management company or managing (general) partner or by an entity which directly or indirectly controls, is controlled by, or is under common control with, such management company or managing or general partner, or member (or retired member) of such Purchaser in accordance with limited liability company interests.
- (iii) "*Person*" (regardless of whether capitalized) means any natural person, entity, or association, including without limitation any corporation, partnership, limited partnership, limited liability company, government (or agency or subdivision thereof), trust, joint venture, or proprietorship.
- (jjj) "Principal Office" means the Note Agent's office, address or bank account as from time to time designated in writing by the Note Agent to the Company and each Holder, which shall initially be the office designated on the Note Agent's signature page to this Agreement.
- (kkk) "**Pro Rata Share**" means, as to any Secured Party at any time, the ratio at such time of (x) the aggregate principal amount of the Notes held by such Secured Party to (y) the aggregate outstanding principal amount of all Notes issued hereunder.
- (III) "Purchasers" means the Initial Purchasers and/or the Additional Purchaser, as the context may require (and "Purchaser" means any one of them).
- (mmm) "Reportable Event" means any of the events set forth in Section 4043(c) of ERISA with respect to a Plan, other than those events as to which the thirty-day notice period is waived under subsection .22, ..23, .25, .27, .28, .29, .30, .31, .32, .34, or .35 of PBGC Regulation Section 4043.
- (nnn) "*Required Holders*" means Holders representing more than fifty percent (50%) of the aggregate outstanding principal amount of the Notes (including all accrued PIK Interest) issued pursuant to this Agreement at the relevant time of reference.
- (000) "Requirements of Law" means as to any Person, provisions of the Charter Documents of such Person, or any law, treaty, code, rule, regulation, right, privilege, qualification, License or franchise, or any determination of an arbitrator or a court or other Governmental Authority, in each case applicable to such Person or any of such Person's property or to which such Person or any of such Person's property is subject or pertaining to any or all of the transactions contemplated or referred to in the Note Documents, including, but not limited to the Fair Credit Reporting Act, the Gramm-Leach-Bliley Act, the Federal Trade Commission Act and comparable state laws, the Telephone Consumer Protection Act, the Telemarketing and Consumer Fraud and Abuse Prevention Act, the CAN-SPAM Act, the Electronic Fund Transfer Act, the U.S. Card Act, and any similar laws and regulations in Canada, including federal and provincial laws.

- (ppp) "Responsible Officer" means the chief financial officer, general counsel, secretary, controller or other authorized officer of the Note Parties set forth on an incumbency certificate delivered to the Collateral Agent from time to time.
- (qqq) "Restricted Payment" means as to any Person (i) any dividend or other distribution (whether in cash, Capital Stock or other property) with respect to or on account of any shares of any Capital Stock of such Person or (ii) any payment by such Person on account of the purchase, redemption, retirement, defeasance, surrender, cancellation, termination or acquisition of any Capital Stock of such Person or any claim respecting the purchase or sale of any Capital Stock of such Person.
- (rrr) "Sanctioned Entity" means (a) an agency of the government of, (b) an organization directly or indirectly controlled by, or (c) a person resident in a country that is subject to a sanctions program identified on the list maintained by OFAC and available at http://www.treas.gov/offices/enforcement/ofac/programs, or as otherwise published from time to time as such program may be applicable to such agency, organization or person.
- (sss) "Sanctioned Person" means a person named on the list of Specially Designated Nationals or Blocked Persons maintained by OFAC available at http://www.treas.gov/offices/enforcement/ofac/sdn/index.html, or as otherwise published from time to time.
 - (ttt) "Secured Debt Cap" has the meaning ascribed thereto in Section 7.1(a).
 - (uuu) "Secured Parties" means the Purchasers, Holders, the Agents, and their respective successors and permitted assigns.
 - (vvv) "Securities" has the meaning ascribed thereto in Section 8.1.
- (www) "Security Agreement" means the Security Agreement, dated as the date hereof, made by the Note Parties in favor of the Collateral Agent, substantially in the form attached hereto as Exhibit D (as amended, restated, supplemented and/or otherwise modified from time to time).
- (xxx) "Single Employer Plan" shall mean any Plan that is covered by Title IV of ERISA or Section 412 of the Code or Section 302 of ERISA, other than a Multiemployer Plan, that is maintained or contributed to by the Company or any Commonly Controlled Entity or to which the Company or a Commonly Controlled Entity has or may have an obligation to contribute, and such plan for the six-year period immediately following the latest date on which the Company or a Commonly Controlled Entity maintained, contributed to or had an obligation to contribute to (or is deemed under Section 4069 of ERISA to have maintained or contributed to or to have had an obligation to contribute to, or otherwise to have liability with respect to) such plan. For the purposes of this definition, "Commonly Controlled Entity" means a person or an entity, whether or not incorporated, that is under common control with the Company within the meaning of Section 4001 of ERISA or is part of a group that includes the Company and that is treated as a single employer under Section 414 of the Code.

- (yyy) "Solvent" means, with respect to any Person, that such Person (i) owns and will own assets the fair saleable value of which are greater than the amount that will be required to pay the probable liabilities of its then existing debts as they become absolute and matured considering all financing alternatives and potential asset sales reasonably available to it, (ii) has capital that is not unreasonably small in relation to its business as presently conducted or after giving effect to any contemplated transaction and (iii) does not intend to incur and does not believe that it will incur debts beyond its ability to pay such debts as they become due. In computing the amount of contingent or unliquidated liabilities at any time, such liabilities shall be computed at the amount that, in light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability.
- (272) "Subsidiary" means, with respect to any Person, any other Person of which an aggregate of more than fifty percent (50%) of the outstanding Capital Stock having ordinary voting power to elect a majority of the board of directors of such other Person is at the time, directly or indirectly, owned legally or beneficially by such Person or one or more Subsidiaries of such Person, or a combination thereof, or with respect to which any such Person has the right to vote or designate the vote of more than fifty percent (50%) of such Capital Stock whether by proxy, agreement, operation of law or otherwise. Unless the context otherwise requires, each reference to a Subsidiary shall mean a Subsidiary of the Company.
- (aaaa) "Taxes" means any present or future United States federal, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp occupation, premium, windfall profits, environmental (including taxes under former Code §59A), customs duties, capital stock, franchise profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on-minimum, estimated, or other taxes, levies, assessments, fees or other charges imposed by any Governmental Authority, including any interest, penalty, or addition thereto, whether disputed or not.
- (bbb) "Total Assets" shall mean the total assets of the Company and its Subsidiaries on a consolidated basis, as shown on the applicable consolidated balance sheet of the Company and its Subsidiaries and computed in accordance with GAAP. Total Assets shall be calculated after giving effect to the transaction giving rise to the need to calculate Total Assets.
 - (cccc) "UCC" means Uniform Commercial Code.
- (dddd) "Unfunded Pension Liability" of any Plan means the amount, if any, by which the value of the accumulated plan benefits under the Plan, determined on a plan termination basis in accordance with actuarial assumptions at such time consistent with those prescribed by the PBGC for purposes of Section 4044 of ERISA, exceeds the fair market value of all Plan assets allocable to such liabilities under Title IV of ERISA (excluding any accrued but unpaid contributions).
- (eeee) "Weighted Average Life to Maturity" means, when applied to any Indebtedness at any date, the number of years obtained by dividing: (a) the sum of the products obtained by multiplying (i) the amount of each then remaining installment, sinking fund, serial maturity or other required payments of principal, including payment at final maturity, in respect

thereof, by (ii) the number of years (calculated to the nearest one-twelfth (1/12th)) that will elapse between such date and the making of such payment by (b) the then outstanding principal amount of such Indebtedness.

2. TERMS OF THE SECURED CONVERTIBLE PROMISSORY NOTES.

2.1 The Initial Notes. Subject to the satisfaction (or waiver in accordance with Section 10.9) of the conditions precedent set forth in Section 4.1, the Company shall issue and sell to each Initial Purchaser, and each Initial Purchaser shall purchase from the Company, an Initial Note on the Initial Closing Date in a principal amount equal to the amount opposite such Initial Purchaser's name set forth in the Schedule of Purchasers under the header 'Initial Purchasers' attached hereto as Schedule 2. By no later than 11:00 a.m. (New York time) on the Initial Closing Date (i) each Initial Purchaser shall wire transfer same day funds in U.S. dollars, at the Note Agent's Principal Office, in the amount of such Initial Purchaser's Initial Note and (ii) the Company shall issue and deliver to each such Initial Purchaser an Initial Note in favor of such Initial Purchaser payable in the principal amount of such Initial Purchaser's Initial Note.

2.2 The Additional Notes.

- (a) Subject to the satisfaction (or waiver in accordance with Section 10.9) of the conditions precedent set forth in Section 4.2, at any time and from time to time after the Initial Closing Date, the Company shall issue and sell to each Person that executes a counterpart signature page to this Agreement in the form attached hereto as Exhibit B (individually, an "Additional Purchaser" and collectively, the "Additional Purchasers"), and each such Additional Purchaser shall purchase from the Company, an Additional Note in a principal amount to be mutually agreed between such Additional Purchaser and the Company. Each additional issuance, sale and purchase of Additional Notes as provided in this Section 2.2 shall take place on a date to be mutually agreed by the Company and such Additional Purchaser (each, an "Additional Closing Date"); provided that (i) each Additional Closing Date shall have occurred on or prior to April 26, 2021, (ii) all issuances, sales and purchases of Additional Notes on an Additional Closing Date shall be made on substantially identical terms and conditions as the Initial Notes, shall, to the extent permitted by law, be fungible for tax purposes with the Initial Notes and may only be amended pursuant to Section 10.9; (iii) the purchase price of each Additional Note shall be increased by the PIK Interest (as defined in the Note) and Cash Interest (as defined in the Note) that has accrued since the Initial Closing Date on the Initial Notes (it being understood and agreed that interest on all the Notes shall accrue PIK Interest and be payable on the same dates) and (iv) in no event shall the initial aggregate principal amount of all Notes issued hereunder exceed \$215,000,000.
- (b) Any Additional Notes issued, sold and purchased pursuant to this Section 2.2 shall be deemed to be "Notes" for all purposes under this Agreement. By no later than 11:00 a.m. (New York time) on an Additional Closing Date (i) each Additional Purchaser shall wire transfer same day funds in U.S. dollars, at the Note Agent's Principal Office, in the amount of such Additional Purchaser's Additional Note and (ii) the Company shall issue and deliver to each such Additional Purchaser an Additional Note in favor of such Additional Purchaser payable in the principal amount of such Additional Purchaser's Additional Note. The Schedule of Purchasers may be amended by the Company without the consent of the Purchasers to include any Additional Purchasers upon the execution by such Additional Purchasers of a counterpart signature page hereto in the form attached hereto as Exhibit B.

3. SECURITY.

To secure the performance and payment in full of the Obligations and the Guaranteed Obligations (as applicable), each Note Party has granted to the Collateral Agent, for the benefit of the Secured Parties, a first-priority Lien on the Collateral (subject to Permitted Liens) pursuant to the Security Agreement.

4. CONDITIONS PRECEDENT TO EACH CLOSING DATE

4.1 Conditions Precedent to the Initial Closing Date

The obligation of the Company to issue and to sell, and of each Initial Purchaser to purchase, Initial Notes the Initial Closing Date are subject to the satisfaction (or waiver in accordance with Section 10.9) of the following conditions precedent on or prior to the Initial Closing Date:

- (a) receipt by each Purchaser of counterparts to this Agreement, the Notes, the Guaranty, the Security Agreement, any Intellectual Property Security Agreement and each other Collateral Document required to be delivered on the Initial Closing Date, each duly executed by each party thereto;
- (b) receipt by each Purchaser of evidence that the Existing Credit Agreement have been or concurrently with the Initial Closing Date will be terminated, all amounts owing thereunder have been or concurrently with the Initial Closing Date will be paid in full and all guarantees thereunder have been or concurrently with the Initial Closing Date will be released;
- (c) receipt by each Purchaser of a certificate signed by a Responsible Officer of each Note Party attaching (i) true, correct and complete copies of the Charter Documents of each Note Party as in effect on the Initial Closing Date, certified, with respect to the Charter Documents set forth in clause (ii) of the definition thereof, as of a recent date by the Secretary of State of the state of incorporation or organization (as applicable) of such Note Party, (iii) a certificate of good standing of each Note Party, certified as of a recent date by the Secretary of State of the state of incorporation or organization (as applicable) of such Note Party, (iii) the names of the Responsible Officers of each Note Party authorized to sign the Note Documents and their true signatures and (iv) true, correct and complete copy of resolutions duly adopted by the board of directors or similar governing body of each Note Party authorizing the execution, delivery and performance of this Agreement and the other Note Documents;
- (d) receipt by each Purchaser of a certificate of solvency, executed by the chief financial officer (or equivalent) of the Company, substantially in the form attached hereto as Exhibit F;
- (e) receipt by each Purchaser of a certificate signed by a Responsible Officer of the Company certifying that the condition specified in <u>clause (I)</u> has been satisfied;

EXECUTION VERSION

- (f) receipt by each Purchaser of the results of customary tax, judgment lien, intellectual property and UCC lien searches with respect to each Note Party, dated a recent date prior to the Initial Closing Date;
- (g) the Purchasers and the Agents shall have received an opinion of Cooley LLP, counsel to the Company, in form and substance reasonably satisfactory to the Purchasers;
- (h) the filing of a financing statement with respect to each Note Party adequately covering the Collateral in a form appropriate for filing under the UCC in the UCC filing office applicable to each such Note Party;
- (i) all authorizations, consents, approvals or permits, if any, of any governmental authority or regulatory body that are required in connection with the lawful issuance and sale of the Notes pursuant to this Agreement shall have been duly obtained and shall be effective on and as of the Initial Closing Date;
- (j) there shall be no claim, action, suit, investigation, litigation or proceeding, pending or, to the knowledge of the Company and its Subsidiaries, threatened, in any court or before any governmental authority with respect to the Note Documents or any transactions contemplated thereby or hereby;
- (k) the representations and warranties made pursuant to Section 5 hereof shall be true and correct in all material respects (or if a representation or warranties is qualified by materiality or material adverse effect, in all respects) on and as of the Initial Closing Date with the same effect as though made on and as of such date, except to the extent such representations and warranties specifically relate to an earlier date, in which case such representations and warranties shall have been true and correct in all material respects (or if a representation or warranties is qualified by materiality or material adverse effect, in all respects) on and as of such earlier date;
- (l) immediately before and after giving effect to the issuance of the Initial Notes on the Initial Closing Date, no Event of Default shall exist;
- (m) the Company shall have paid all outstanding fees and expenses of the Agents, Shipman & Goodwin LLP, counsel to the Agents, and as otherwise required pursuant to Section 10(a);
- (n) receipt by the Company and the Note Agent of an Administrative Questionnaire and executed copies of IRS Form W-9 or appropriate IRS Form W-8, as applicable, as well as any reasonably necessary supporting information, certifying that payments to the Initial Purchasers under Initial Notes are exempt from U.S. federal withholding Tax; and
- (o) the Company shall have delivered to the Purchasers such other documents and instruments relating to the transactions contemplated by this Agreement as the Purchasers or their counsel may reasonably request (which request shall have been made no later than 3 days prior to the Initial Closing Date).

- **4.2** Conditions Precedent to each Additional Closing Date. The obligation of the Company to issue and to sell, and of each Additional Purchaser to purchase, Additional Notes on an Additional Closing Date are subject to the satisfaction (or waiver in accordance with Section 10.8) of the following conditions precedent on or prior to each such Additional Closing Date:
- (a) each Additional Purchaser shall have received copies of each of the documents set forth in <u>clauses (a)</u> through (h) and <u>clause (o)</u> of <u>Section 4.1</u> delivered to the Initial Purchasers on the Initial Closing Date;
- (b) the representations and warranties made pursuant to Section 5 hereof shall be true and correct in all material respects (or if a representation or warranties is qualified by materiality or material adverse effect, in all respects) on and as of the applicable Additional Closing Date with the same effect as though made on and as of such date, except to the extent such representations and warranties specifically relate to an earlier date, in which case such representations and warranties shall have been true and correct in all material respects (or if a representation or warranties is qualified by materiality or material adverse effect, in all respects) on and as of such earlier date;
- (c) immediately before and after giving effect to the issuance of Additional Notes on the applicable Additional Closing Date, no Event of Default (as defined in the Notes) shall exist; and
- (d) receipt by the Company and the Note Agent of an Administrative Questionnaire and of executed copies of IRS Form W-9 or appropriate IRS Form W-8, as applicable, as well as any reasonably necessary supporting information, certifying that payments to the Additional Purchasers under Additional Notes are exempt from U.S. federal withholding Tax.

5. REPRESENTATIONS AND WARRANTIES OF THE NOTE PARTIES

Each Note Party hereby represents and warrants to each Initial Purchaser and the Agents as of the Initial Closing Date and to each Additional Purchaser and the Agents as of the applicable Additional Closing Date as follows:

5.1 Organization, Good Standing and Qualification.

(a) Organization; Good Standing; Qualification. Each Note Party and each of its Subsidiaries: (i) is duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation or formation, (ii) has all requisite corporate or limited liability company power and authority to own and operate its property, to lease the property it operates as lessee and to conduct the business in which it is currently, or is currently proposed to be, engaged; (iii) is duly qualified as a foreign entity, licensed and in good standing under the laws of each jurisdiction where its ownership, lease or operation of property or the conduct of its business requires such qualification, except where the failure to be so qualified could not reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect, and (iv) has the corporate or limited liability company power and authority to execute, deliver and perform its obligations under each Note Document to which it is or will be a party and to borrow hereunder. Each Note Party's present name, former names within the last five (5)

years (if any), owned and leased locations, place of formation, tax identification number and organizational identification number are correctly set forth in <u>Schedule 5.1</u> hereto, as may be updated by the Company from time to time in a written notice provided to the Collateral Agent after the Initial Closing Date

- (b) Collateral. Except for the Liens granted under the Security Agreement, each Note Party is the sole legal and equitable owner of each item of Collateral in which it purports to grant a security interest under the Security Agreement, having good and marketable title thereto, free and clear of any and all Liens except for Permitted Liens and, upon filing of UCC financing statements in the UCC filing office applicable to such Note Party, the Collateral Agent shall have a perfected security interest in the Collateral, which such security interest shall be senior to all other Liens (other than Permitted Liens) to the extent the same can be perfect by filing.
- 5.2 Corporate Power; No Contravention. The execution, delivery and performance by the Company and each Subsidiary of each Note Document to which it is or will be a party and the consummation of the transactions contemplated hereby: (a) have been duly authorized by all necessary corporate or limited liability company action; (b) do not and will not contravene or violate the terms of the Charter Documents of the Company or any of its Subsidiaries or any amendment thereto or any material Requirement of Law applicable to the Company or such Subsidiary or the Company's or such Subsidiary's assets, business or properties; (c) do not and will not (i) conflict with, contravene, result in any violation or breach of or default under any material Contractual Obligation of the Company or such Subsidiary (with or without the giving of notice or the lapse of time or both) other than any right to consent, which consents have been obtained, (ii) create in any other Person a right or claim of termination or amendment of any material Contractual Obligation of the Company or such Subsidiary, or (iii) require modification, acceleration or cancellation of any material Contractual Obligation of the Company or such Subsidiary; and (d) do not and will not result in the creation of any Lien (or obligation to create a Lien) against any property, asset or business of the Company or such Subsidiary (other than those securing the Notes).
- 5.3 Authorization. All corporate action on the part of the Company, its Subsidiaries and their respective directors and stockholders necessary for the authorization, execution, delivery and performance of this Agreement by the Company and its Subsidiaries and the performance of the Company's obligations hereunder, including the issuance and delivery of the Notes and the reservation of the equity securities issuable upon conversion of the Notes (collectively, the "Conversion Securities") has been taken or will be taken prior to the issuance of such Conversion Securities. This Agreement and the Note Documents, when executed and delivered by the Note Parties, shall constitute valid and binding obligations of the Note Parties enforceable in accordance with their terms, subject to laws of general application relating to bankruptcy, insolvency, the relief of debtors and general principles of equity and, with respect to rights to indemnity, subject to federal and state securities laws. The Conversion Securities, when issued in compliance with the provisions of this Agreement or the Notes will be validly issued, fully paid and nonassessable and free of any Liens and issued in compliance with all applicable federal and state securities laws.

5.4 Governmental Consents. All consents, approvals, orders, or authorizations of, or registrations, qualifications, designations, declarations, or filings with, any Governmental Authority, required on the part of the Note Parties in connection with the valid execution and delivery of this Agreement and the other Note Documents, the offer, sale or issuance of the Notes and the Conversion Securities issuable upon conversion of the Notes and the consummation of any other transaction contemplated hereby shall have been obtained and will be effective on the Initial Closing Date.

5.5 Compliance with Laws.

- (a) No Note Party is in violation of any Requirements of Law, any applicable statute, rule, regulation, order or restriction of any domestic government or any instrumentality or agency thereof in respect of the conduct of its business or the ownership of its properties, which violation of which would materially and adversely affect the business, assets, liabilities, financial condition or operations of the Note Parties (individually and in the aggregate).
- (b) Neither the Company nor any of its Subsidiaries is registered or required to register as an "investment company" or a company "controlled" by an "investment company" under the Investment Company Act of 1940, as amended. Neither the Company nor any of its Subsidiaries is engaged as one of its important activities in extending credit for margin stock (under Regulations X, T and U of the Federal Reserve Board of Governors). The Company and each of its Subsidiaries has complied in all material respects with applicable provisions of the Federal Fair Labor Standards Act. Neither the Company nor any of its Subsidiaries is a "holding company" or an "affiliate" of a "holding company" or a "subsidiary company" of a "holding company" as each term is defined and used in the Public Utility Holding Company Act of 2005. Neither the Company's nor any of its Subsidiaries' properties or assets has been used by the Company or such Subsidiary or, to Company's knowledge, by previous Persons, in disposing, producing, storing, treating, or transporting any hazardous substance other than in material compliance with applicable laws. The Company and each of its Subsidiaries has obtained all consents, approvals and authorizations of, made all declarations or filings with, and given all notices to, all Governmental Authorities that are necessary to continue their respective businesses as currently conducted, except to the extent that the failure to obtain, make or give any of the foregoing would not reasonably be expected to have a Material Adverse Effect.
- (c) Neither the Company nor any Subsidiary (i) is a Sanctioned Person, (ii) has any assets in Sanctioned Entities, or (iii) derives any operating income from investments in, or transactions with Sanctioned Persons or Sanctioned Entities. The proceeds of the Notes will not be used and have not been used to fund any operations in, finance any investments or activities in, or make any payments to, a Sanctioned Person or a Sanctioned Entity.
- (d) The Company and its Subsidiaries are in compliance, in all material respects, with any United States Requirements of Law relating to terrorism, sanctions or money laundering (the "Anti-Terrorism Laws"), including the United States Executive Order No. 13224 on Terrorist Financing (the "Anti-Terrorism Order") and the Patriot Act. No part of the proceeds of any Note will be used, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended or any other Anti-Terrorism Law.

- (e) No Note Party and no Subsidiary of any Note Party (i) is listed in the annex to, or is otherwise subject to the provisions of, the Anti-Terrorism Order, (ii) is owned or controlled by, or acting for or on behalf of, any person listed in the annex to, or is otherwise subject to the provisions of, the Anti-Terrorism Order or (iii) commits, threatens or conspires to commit or supports "terrorism" as defined in the Anti-Terrorism Order.
- (f) None of the funds to be provided under this Agreement will be used, directly or indirectly, (i) for any activities in violation of any applicable anti-money laundering, economic sanctions and anti-bribery laws and regulations laws and regulations or (ii) for any payment to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.
- 5.6 Compliance with Other Instruments. The Company is not in violation or default of any term of its articles of incorporation, or of any provision of any mortgage, indenture or contract to which it is a party and by which it is bound or of any judgment, decree, order or writ, other than such violation(s) that would not have a Material Adverse Effect on the Company. The execution, delivery and performance of this Agreement and the other Note Documents, and the consummation of the transactions contemplated hereby and thereby will not result in any material violation or be in conflict with, or constitute, with or without the passage of time and giving of notice, either a default under any such material provision, instrument, judgment, decree, order or writ or an event that results in the creation of any material Lien upon any assets of the Company or the suspension, revocation, impairment, forfeiture, or nonrenewal of any material permit, license, authorization or approval applicable to the Company, its business or operations or any of its material assets or properties. Without limiting the foregoing, the Company has obtained all waivers reasonably necessary with respect to any preemptive rights, rights of first refusal or similar rights, including any notice or offering periods provided for as part of any such rights, in order for the Company to consummate the transactions contemplated hereunder without any third party obtaining any rights to cause the Company to offer or issue any securities of the Company as a result of the consummation of the transactions contemplated hereunder.
- **5.7 Solvency.** As of the Initial Closing Date both immediately before and after giving effect to the transactions contemplated by the Note Documents, the Company is individually, and the Company and its Subsidiaries on a consolidated basis are, Solvent.
- 5.8 Offering. The offer, issue, and sale of the Notes and the Conversion Securities are and will be exempt from the registration and prospectus delivery requirements of the Act, and no qualification under the Trust Indenture Act of 1939, as amended, and the rules and regulations of the Securities and Exchange Commission promulgated thereunder is required in connection with, the issuance of the Notes and the Conversion Securities.

- **5.9** No "Bad Actor" Disqualification. The Company has exercised reasonable care to determine whether any Company Covered Person (as defined below) is subject to any of the "bad actor" disqualifications described in Rule 506(d)(1)(i) through (viii), as modified by Rules 506(d)(2) and (d) (3), under the Act ("Disqualification Events"). To the Company's knowledge, no Company Covered Person is subject to a Disqualification Event. The Company has complied, to the extent required, with any disclosure obligations under Rule 506(e) under the Act.
- 5.10 Litigation. There are no legal actions, suits, proceedings, claims or disputes pending or, to the knowledge of any Note Party, threatened in writing, at law, in equity, in arbitration or before any Governmental Authority against or affecting the Company or its Subsidiaries that could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect; (b) there is no injunction, writ, temporary restraining order, decree or any order or determination of any nature by any arbitrator, court or other Governmental Authority purporting to enjoin or restrain the execution, delivery or performance of the Note Documents or which relates to the assets or the business of the Company or its Subsidiaries; and (c) there is no litigation, claim, audit, dispute, review, proceeding or investigation currently pending or threatened in writing against the Company or its Subsidiaries for any violation or alleged violation of any Requirements of Law, and neither the Company nor any Subsidiary has received written notice of any threat of any suit, action, claim, dispute, investigation, review or other proceeding pursuant to or involving any Requirements of Law.

5.11 Taxes.

- (a) Except to the extent that the failure to do so could not reasonably be expected to have a Material Adverse Effect, the Company and each of its Subsidiaries has timely filed all United States federal and state income and other material tax returns that it was required to file, in each case with due regard for any extension of time within which to file such tax return. Except to the extent that the failure to do so could not reasonably be expected to have a Material Adverse Effect, all Taxes due and payable by the Company or its Subsidiaries have been paid, in each case with due regard for any extension of time within which to file such tax return, other than any Taxes the amount or validity of which is being actively contested by Company or its Subsidiaries in good faith and by appropriate proceedings and with respect to which adequate reserves or other appropriate provision, if any, as shall be required in conformity with GAAP shall have been made or provided therefor. There are no Liens, other than Permitted Liens, on any of the assets of the Company or its Subsidiaries that arose in connection with any failure (or alleged failure) to pay any Tax. To the knowledge of the Company, no claim has been made by a Governmental Authority in a jurisdiction where the Company and its Subsidiaries do not file tax returns that the Company or any of its Subsidiaries is or may be subject to taxation by that jurisdiction that could reasonably be expected to have a Material Adverse Effect.
- (b) There is no action, suit, proceeding, investigation, examination, audit, or claim now pending or, to the knowledge of the Company, threatened in writing by any Governmental Authority regarding any Taxes relating to the Company or its Subsidiaries that could reasonably be expected to have a Material Adverse Effect.
- 5.12 Financial Condition. The Note Parties have furnished the Purchasers with true, correct and complete copies of (i) the audited consolidated balance sheets of the Company and

its Subsidiaries as of December 31, 2018 and 2019 and the related consolidated statements of income or operations, shareholders' equity and cash flows for each such fiscal year and (ii) the unaudited consolidated balance sheets of the Company and its Subsidiaries as of December 31, 2020 and the related consolidated statements of income or operations and cash flows for such fiscal year (collectively, the "Financial Statements"). The Financial Statements fairly present, in all material respects, the financial position of the Company and its Subsidiaries on a consolidated basis, as of the respective dates thereof, and the results of operations and cash flows thereof, as of the respective dates or for the respective periods set forth therein, and are in conformity with the past historical practices of the Note Parties, with GAAP consistently applied during the periods involved. As of the dates of the Financial Statements, neither the Company nor any Subsidiary had any known obligation, Indebtedness or liability (whether accrued, absolute, contingent or otherwise, and whether due or to become due), which was not reflected or reserved against in the balance sheets which are part of the Financial Statements, except for those incurred in the ordinary course of business and which are fully reflected on the books of account of the Company or its Subsidiaries, as applicable.

- **5.13** Subsidiaries. Except as set forth on Schedule 5.13, the Company does not have any Subsidiaries.
- 5.14 Capitalization. As of the Initial Closing Date, without giving effect to the transactions contemplated hereby and in the other Note Documents, the outstanding capitalization of the Company and its Subsidiaries is as set forth on Schedule 5.14. All of the issued and outstanding Capital Stock of the Company has been, and Capital Stock of the Company issuable upon the exercise of outstanding securities when issued will be, duly authorized and validly issued and are fully paid and nonassessable. All outstanding Capital Stock of the Company's Subsidiaries are 100% owned by the Company or one of its Subsidiaries free and clear of all Liens other than Permitted Liens. Except as set forth in the Charter Documents (as in effect on the Initial Closing Date), the issuance of the foregoing Capital Stock is not and has not been subject to preemptive rights in favor of any Person other than such rights that have been waived and will not result in the issuance of any additional Capital Stock of the Company or the triggering of any downround or similar rights contained in any options warrants, debentures or other securities or agreements of the Company or any of its Subsidiaries. On the Initial Closing Date, except as set forth on Schedule 5.14, there are no outstanding securities convertible into or exchangeable for Capital Stock of the Company or any of its Subsidiaries or options, warrants or other rights to purchase or subscribe for Capital Stock of the Company or any of its Subsidiaries, or contracts, commitments, agreements, understandings or arrangements of any kind to which the Company or any of its Subsidiaries is a party relating to the issuance of any Capital Stock of the Company or any of its Subsidiaries, or any such convertible or exchangeable securities or any such options, warrants or rights. On the Initial Closing Date, except as set forth on Schedule 5.14, neither the Company nor any of its Subsidiaries has any obligation, whether mandatory or at the option of any other Person, at any time to redeem or repurchase any Capital Stock of the Company or any of its Subsidiaries, pursuant to the terms of their respective Charter Documents or otherwise. All securities of the Company and its Subsidiaries (including all shares of the Company's common stock, securities, options and warrants to purchase shares of the Company's common stock (both outstanding as well as those that are no longer outstanding)), have been and were issued and granted pursuant to an exception from the Act and otherwise in compliance, in all material respects, with all securities and other applicable laws.

- **5.15 Private Offering**. No form of general solicitation or general advertising was used by the Company or its Subsidiaries or their respective representatives in connection with the offer or sale of the Notes to the Purchasers pursuant to this Agreement.
- 5.16 Broker's, Finder's or Similar Fees. Except as set forth in that certain Engagement Letter by and between the Company and GreensLedge Capital Markets LLC, dated on or about of April 19, 2021, there are no brokerage commissions, finder's fees or similar fees or commissions payable by the Company or its Subsidiaries in connection with the transactions based on any agreement, arrangement or understanding with the Company or its Subsidiaries or any action taken by the Company or its Subsidiaries. Notwithstanding the foregoing or any other provision herein, no Purchaser shall be liable for any brokerage commission, finder's fees or similar fees or commissions.
- **5.17 Indebtedness.** Schedule 5.17 lists the amount of all Indebtedness of the Company and its Subsidiaries (other than Indebtedness under this Agreement) that is in existence immediately before the Initial Closing Date and will remain outstanding after the Initial Closing Date.

6. AFFIRMATIVE COVENANTS OF THE NOTE PARTIES

Each Note Party shall, and shall cause each of its Subsidiaries to:

- **6.1 Reporting Obligations**. Deliver to the Note Agent (for prompt distribution to the Purchasers as specified in the applicable Administrative Questionnaire (or as otherwise specified by a Purchaser to the Note Agent in writing from time to time)):
- (a) as soon as available, but not later than 150 days after the end of each fiscal year of the Company, a copy of the audited consolidated balance sheets of the Company and its Subsidiaries as at the end of such fiscal year and the related consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal year, in each case, prepared in accordance with GAAP, setting forth in each case in comparative form the figures for the previous fiscal year, and accompanied by a report of any "Big Four" or, upon Required Holders' written consent (not to be unreasonably withheld), any other independent certified public accounting firm, which report shall contain an unqualified opinion (without any (A) "going concern" or like qualification or exception, (B) qualification or exception as to the scope of such audit, or (C) qualification which relates to the treatment or classification of any item and which, as a condition to the removal of such qualification, would require an adjustment to such item (except in the case of clauses (A) and (B) above as it relates to the Obligations during the last twelve months before the Maturity Date)), stating that such consolidated financial statements present fairly in all material respects the financial position for the periods indicated in conformity with GAAP applied on a basis consistent with prior years; and
- (b) as soon as available, but not later than 45 after the end of each fiscal quarter of each fiscal year (other than the fourth fiscal quarter), the unaudited consolidated balance sheets of the Company and its Subsidiaries as at the end of such fiscal quarter and the

related unaudited consolidated statements of income and cash flows of the Company and its Subsidiaries for such fiscal quarter and for the period from the beginning of the then current fiscal year to the end of such fiscal quarter, all certified on behalf of the Company by a Responsible Officer of the Company as being complete and correct and fairly presenting, in all material respects, the financial position and the results of operations of the Company and its Subsidiaries, subject to normal year-end adjustments and absence of footnote disclosures.

(c) promptly, and in any event within three (3) Business Days after the Company or any other Note Party becomes aware of or has knowledge of any event or condition that constitutes a Default or Event of Default, provide written notice of such event or condition and a statement of the curative action that the Company proposes to take with respect thereto.

Delivery of any reports, information and documents under this Section 6.2, as well as any other reports, information and documents pursuant to this Agreement, to the Note Agent is for informational purposes only and the Note Agent's receipt of the same shall not constitute constructive notice of any information contained therein or determinable from information contained therein, including the Company's compliance with any of its covenants hereunder (as to which the Note Agent is entitled to rely exclusively on certificates of a Responsible Officer of the Company). The Note Agent shall have no responsibility or liability for the filing, timeliness or content of any report required under this Section 6.2 or any other reports, information and documents required under this Agreement.

6.2 Taxes and Claims.

- (a) Timely file complete and correct United States federal and state income and applicable foreign, state and local tax returns required by law, in each case with due regard for any extension of time within which to file such tax returns, and pay when due all Taxes in each case, except to the extent that the failure to so file or pay could not reasonably be expected to have a Material Adverse Effect; provided that the Company and its Subsidiaries shall not be required to pay any such Taxes which are being contested in good faith by appropriate proceedings and with respect to which adequate reserves have been set aside in accordance with GAAP, which deferment of payment is permissible so long as no Lien, other than a Permitted Lien has been entered and the Company's and its Subsidiaries' title to, and its and their right to use, its and their respective Properties are not materially adversely affected thereby.
- (b) Pay all stamp, court or documentary, intangible, recording, filing or similar Taxes, if any, that arise solely in connection with the issuance of the Notes except to the extent such Taxes arise as a result of a transfer of a Note to a person other than the initial Holder of the Notes. The obligations of the Company under this Section 6.2(b) shall survive the payment of the Obligations and the termination of the Note Documents.

6.3 Insurance.

(a) Maintain with reputable insurance companies insurance in such amounts and covering such risks as is consistent with sound business practice, including, without limitation, property and casualty insurance on all of its property, general liability insurance, workers compensation insurance and business interruption insurance. Each Note Party will, and will cause each of its Subsidiaries to, furnish to the Collateral Agent upon reasonable request full information as to the insurance carried by it.

- (b) At all times keep its property which is subject to the Lien of the Collateral Agent insured in favor of the Collateral Agent, and all policies or certificates (or certified copies thereof) with respect to such insurance. At or prior to the Closing Date, the Company shall furnish certificates of insurance issued on applicable ACORD Forms with respect to property and liability insurance for the Company. Each Note Party will, and will cause each of its Subsidiaries to, notify the Collateral Agent and the Purchasers, promptly, upon receipt of a notice of termination, cancellation, or non-renewal from its insurance company of any such policy.
- (c) If the Company shall fail to maintain all insurance in accordance with this <u>Section 6.3</u> or to timely pay or cause to be paid the premium(s) on any such insurance, or if the Company shall fail to deliver all certificates with respect thereto, the Collateral Agent shall have the right (but shall be under no obligation) to procure such insurance or pay such premiums, and the Company agrees to reimburse the Purchasers, on demand, for all costs and expenses relating thereto.
- 6.4 Compliance with Laws. Comply with any and all Requirements of Law to which it may be subject including, without limitation, all Environmental Laws, and obtain any and all Licenses necessary to the ownership of its property or to the conduct of its businesses, except, in each case, where failure to do so could not reasonably be expected to have a Material Adverse Effect. Each Note Party will, and will cause each of its Subsidiaries to, timely satisfy all material assessments, fines, costs and penalties imposed by any Governmental Authority against such Person or any property of such Person except to the extent such assessments, fines, costs, or penalties are being contested in good faith by appropriate proceedings and for which the Company or such Subsidiary has set aside on its books adequate reserves in accordance with GAAP.
- 6.5 Maintenance of Properties. Do all things necessary to maintain, preserve, protect and keep its property (other than property that is obsolete, surplus, or no longer used or useful in the ordinary conduct of its business) in good repair, working order and condition (ordinary wear and tear and casualty and condemnation excepted), make all necessary and proper repairs, renewals and replacements such that its business can be carried on in connection therewith and be properly conducted at all times and pay and discharge when due the cost of repairs and maintenance to its property, and pay all rentals when due for all real estate leased by such Person.
- 6.6 Employee Benefit Plans. (a) Keep in full force and effect any and all Plans which are presently in existence or may, from time to time, come into existence under ERISA and not withdraw from any such Plans, unless such withdrawal can be effected or such Plans can be terminated without material liability to the Company or its Subsidiaries, (b) make contributions to all such Plans in a timely manner and in a sufficient amount to comply in all material respects with the standards of ERISA, including, without limitation, the minimum funding standards of ERISA, (c) comply in all material respects with all requirements of ERISA, (d) notify the Purchasers promptly upon receipt by the Company or any Subsidiary of any notice

concerning the imposition of any withdrawal liability or of the institution of any proceeding or other action which may result in the termination of any such Plans by the PBGC or the appointment of a trustee to administer such Plans, (c) promptly advise the Purchasers of the occurrence of any Reportable Event or non-exempt prohibited transaction (as defined in ERISA) with respect to any such Plans of which Company becomes aware, and (f) amend any Plan that is intended to be qualified within the meaning of Section 401 of the Code to the extent necessary to keep the Plan qualified and to cause the Plan to be administered and operated in a manner that does not cause the Plan to lose its qualified status.

6.7 Environmental. Use and operate all of its facilities and Properties in material compliance with all Environmental Laws, keep all necessary Licenses in effect and remain in material compliance therewith, and handle all Hazardous Materials in material compliance with all applicable Environmental Laws.

6.8 Intellectual Property.

- (a) Each Note Party will take the steps described in this Section 6.8 with respect to all new or acquired Intellectual Property to which the Company or any Guarantor is now or later becomes entitled that is necessary in the conduct of such Person's business. The Company acknowledges and agrees that the Secured Parties shall have no duties with respect to any Intellectual Property or Licenses of the Company or its Subsidiaries.
- (b) The Note Parties shall have the duty, with respect to Intellectual Property that is necessary in the conduct of such Person's business (i) to prosecute diligently any trademark application or service mark application that is part of the trademarks pending as of the date hereof or hereafter, (ii) to prosecute diligently any patent application that is part of the patents pending as of the date hereof or hereafter, and (iii) to take all reasonable and necessary action to preserve and maintain all of the Note Parties' trademarks, patents, copyrights, Licenses, and its rights therein, including paying all maintenance fees and filing of applications for renewal, affidavits of use, and affidavits of non-contestability, except in the case of clauses (i) and (ii), where the Company, in its reasonable opinion, determines that the costs for engaging in such prosecution activities exceeds the likely benefit of continued prosecution and, except in the of case (iii), where the Company, in its reasonable opinion, determines that the costs of preserving and maintaining exceeds the value the Company obtains from such preservation and maintenance. Each Note Party will require all employees, consultants, and contractors of such Note Party who were involved in the creation or development of such Intellectual Property to sign agreements containing assignment to the Company or such Guarantor of Intellectual Property rights created or developed and obligations of confidentiality. No Note Party shall abandon any Intellectual Property or License that is necessary in the conduct of the Company's or such Guarantor's business.
- (c) Each Note Party will (i) promptly file, at such Note Party's sole cost and expense, any application to register any Intellectual Property with the United States Patent and Trademark Office or the United States Copyright Office if such registration is necessary in connection with the conduct of such Note Party's business, (ii) concurrently with the delivery of financial statements pursuant to Sections 6.1(a) or (b), deliver supplements to Schedule 4(c) to the Security Agreement and any other documents reasonably necessary for the Collateral Agent to

record its security interest in such Intellectual Property with the United States Patent and Trademark Office or the United States Copyright Office and (iii) promptly after the delivery of the documents required by <u>clause (ii)</u> above, upon the reasonable request of the Collateral Agent, execute and deliver in favor of the Collateral Agent one or more Intellectual Property Security Agreements to further evidence the Purchasers' Lien on such Note Party's Intellectual Property.

- 6.9 Use of Proceeds. Use the proceeds of the Notes (i) to fund the purchase of certain equipment and other capital assets, (ii) to prepay in full all amounts owing under the Existing Credit Agreement, (iii) to pay any fees and expenses incurred by the Company in connection with the transactions contemplated hereby and (iv) for general corporate purposes not in violation of any applicable laws. The Company shall not use any proceeds of the sale of the Notes hereunder to, directly or indirectly, purchase or carry any "margin stock" (as defined in Regulation U) or to extend credit to others for the purpose of purchasing or carrying any "margin stock", in either case, in violation of the provisions of Regulation T, U or X of the Board of Governors of the Federal Reserve System.
- 6.10 Subsidiaries. If any Note Party creates, forms or acquires any Subsidiary (other than an Excluded Subsidiary) on or after the date of this Agreement, such Note Party will, and will cause such Subsidiary to, (a) within 30 days (which 30 days may be extended by the Collateral Agent at the direction of the Required Holders) of the creation, formation or acquisition of such new Subsidiary, cause such Subsidiary to join the Guaranty as a "Guarantor" and the Security Agreement as a "Grantor" and (b) take all such actions and execute and deliver, or cause to be executed and delivered, all such documents, instruments, agreements, and certificates as are similar to those described in Sections 4.1(c) and 4.1(g) or that are necessary or desirable to protect, evidence or perfect the security interest of the Collateral Agent in a manner similar to the Liens and assets granted by the existing Note parties under the existing Collateral Documents either by executing and delivering to the Collateral Agent a counterpart or supplement to the existing Collateral Documents or such new documents as are necessary or desirable to evidence, grant or perfect a first priority lien in such assets in favor of Collateral Agent, for the benefit of Secured Parties (including, without limitation, any pledges of Capital Stock (other than with respect to Excluded Property (as defined in the Security Agreement)). The Agents shall be authorized to execute and deliver such documents upon receipt of a certificate from a Responsible Officer of the Borrower stating that such documents are authorized or permitted under the Note Documents.
- **6.11 Piggyback Registration Rights**. The Company shall provide the Holders of the Notes with customary "piggyback" registration rights (with respect to the shares of Common Stock issued upon conversion of the Notes) on all registration statements of the Company, subject to the right, however, of the Company and its underwriters to reduce the number of shares proposed to be registered at the underwriter's discretion.
- **6.12 Further Assurances**. Each Note Party will take any action reasonably requested by any Holder in order to effectuate the purposes and terms contained in this Agreement or any other Note Document.

7. NEGATIVE COVENANTS OF THE NOTE PARTIES

Each Note Party shall not, and shall cause each of its Subsidiaries not to:

7.1 Liens, Restricted Payments and Dispositions. Prior to the occurrence of a Conversion Event, so long as the Obligations remain outstanding:

(a) Liens. Create, assume or suffer to exist any Liens on any assets of the Note Parties securing debt for borrowed money in excess of an amount at any time outstanding equal to the greater of (x) the sum of (I) the aggregate principal amount of the Notes at any time outstanding (disregarding, for the purposes of this clause (I) any and all PIK Interest (as defined in the Notes) that has been added to the aggregate principal amount of the Notes) plus (II) \$50,000,000 and (y) \$265,000,000 (such clause (y), the "Secured Debt Cap"); provided that (x) in no event shall the aggregate principal amount of the Notes at any time outstanding (disregarding, for the purposes of this proviso any and all PIK Interest (as defined in the Notes) that has been added to the aggregate principal amount of the Notes) exceed at any time \$215,000,000, (y) the Collateral Agent and the authorized representative with respect to any Indebtedness that is secured on a pari passu basis with the Liens on the Collateral securing the Obligations that is permitted to be secured in reliance on this clause (a) shall have entered into an intercreditor agreement providing for equal and ratable lien priority and perfection for the holders and providers of such Indebtedness at the direction of and in form and substance reasonably satisfactory to Required Holders (the "Intercreditor Agreement") and (z) notwithstanding the foregoing, the Note Parties shall be permitted to any incur letters of credit and any Indebtedness, including Capital Lease Obligations, incurred to finance the acquisition of any fixed or capital assets (including, for the avoidance of doubt, any real property assets) so long as such Liens shall encumber only the assets acquired or financed with the proceeds of such Indebtedness (or, in the case of any letters of credit, cash collateral) and do not attach to any other assets of any Note Party, which letters of credit and Indebtedness set forth in this clause (z) shall, for the avoidance of doubt, be excluded from the calculation of t

(b) Restricted Payments.

(i) Pay or make any Restricted Payment to its direct or indirect equityholders; <u>provided</u> that, the foregoing shall not restrict or prohibit any Subsidiary from paying or making any Restricted Payments, directly or indirectly, to the Company, and shall not restrict or prohibit any Restricted Payments, directly or indirectly, from the Company to its direct or indirect equityholders at such times and in such amounts as are necessary to permit:

(1) such equityholder (A) to pay general administrative costs and expenses (including audit and tax fees payable to third party auditors and tax advisors, customary compensation and reasonable costs and expenses of the board of directors or similar managing body of such entity incurred in the ordinary course of business, and franchise taxes and other fees, taxes and expenses required to maintain such entity's existence), (B) to discharge its, its Subsidiaries' and its direct and indirect owners' income tax liabilities and (C) in the case that the Company is treated as a flow-through entity for U.S. income tax purposes, its direct and indirect owners to discharge their respective U.S. federal, state and local and non-U.S. income tax obligations associated with their ownership of the Company, in each case, so long the amount of any such Restricted Payment is applied for such purpose; and

- (2) (x) purchases or cash payments in lieu of fractional shares of Capital Stock arising out of stock dividends, splits, combinations or conversions of Capital Stock in the ordinary course of business and (y) purchases or cash payments in lieu of fractional shares upon conversion of the Notes.
- (ii) From and after the occurrence of a Conversion Event, any Restricted Payments made or paid to the direct or indirect equityholders of the Company (other than the Restricted Payments of the type referred to in the proviso to <u>clause (i)</u> above) shall also be made or paid to the Holders in accordance with their Pro Rata Share of the Notes on an as-converted basis as if such Notes had been converted pursuant to their terms at the time of the applicable Restricted Payment.
- (c) Asset Dispositions. Consummate any Asset Dispositions; provided that the foregoing shall not restrict or prohibit (A) sales of inventory in the ordinary course of business; (B) the use of cash or Cash Equivalents in a manner not prohibited by the Note Documents; (C) licenses, sublicenses, leases or subleases granted to third parties in the ordinary course of business not interfering with the business of the Company and its Subsidiaries and which do not materially impair the value of the property so licensed, sublicensed, leased or subleased; (D) dispositions of obsolete, damaged, surplus or worn-out or no longer used or useful equipment; (E) the lapse, abandonment or other dispositions of Intellectual Property that is, in the reasonable business judgment of the Company, no longer economically practicable or commercially desirable to maintain; (F) dispositions by (1) any Subsidiary to the Company, (2) the Company to any Note Party, (3) any Subsidiary to any Note Party or (4) any Subsidiary that is not a Note Party; (G) dispositions resulting from any casualty or property losses or condemnation or similar proceeding of, any property or asset of the Company or any of its Subsidiaries; (H) the sale or other disposition of any assets or property of the Company not constituting Collateral; (I) any Asset Disposition so long as such Asset Disposition is made for fair market value; (J) any individual Asset Disposition so long as the consideration therefor does not exceed \$1,000,000; and (K) any other Asset Dispositions to a non-affiliated third party so long as the aggregate consideration therefor does not exceed \$10,000,000 per fiscal year; provided, further, that notwithstanding the foregoing, any Asset Disposition described in the foregoing clause (K) shall be permissible notwithstanding the fact that the counterparty is an affiliate or a Subsidiary of the Company so long as such Asset Disposition is on fair and reasonable terms no less favorable to the Company or such affiliate or Subsidiary than would be obtain
- 7.2 Issuances of Equity. Except to the extent permitted by Section 7.1(c), no Subsidiary of the Company shall issue or sell Capital Stock in such Subsidiary.
- 7.3 Modifications of Charter Documents. The Company will not permit, and will cause each of its Subsidiaries not to permit, such Person's Charter Documents to be amended or modified in any way that could reasonably be expected to materially or adversely affect the interests of the Holders in their capacity as Secured Parties (and not, for the avoidance of doubt, as holders of Capital Stock of the Company).

7.4 Compensation or Credit Support. Other than (x) to consenting or approving Holders in connection with a consent or amendment in accordance with Section 10.9 to the extent all Holders are afforded an opportunity to enter into such consent or amendment or (y) in connection with the issuance of warrants to non-consenting Holders pursuant to Section 1(b)(i)(1) of the Notes, the Company shall not enter into any arrangement with any Holder for the purposes of providing additional compensation (in the form of interest, fees or otherwise) or credit support (in the form of additional collateral or otherwise) to such Holder in connection with the Notes without providing such additional compensation or credit support for the ratable benefit of all Holders.

8. REPRESENTATIONS AND WARRANTIES OF THE PURCHASERS

Each Initial Purchaser hereby represents and warrants as of the Initial Closing Date, and each Additional Purchaser hereby represents and warrants as of the applicable Additional Closing Date as follows:

- 8.1 Purchase for Own Account. Each Purchaser represents that it is acquiring a Note and the Conversion Securities (collectively, the "Securities") solely for its own account and beneficial interest for investment and not for sale or with a view to distribution of the Securities or any part thereof, has no present intention of selling (in connection with a distribution or otherwise), granting any participation in, or otherwise distributing the same, and does not presently have reason to anticipate a change in such intention.
- **8.2 Information and Sophistication.** Without lessening or obviating the representations and warranties of the Note Parties set forth in Section 5 hereof or in any other Note Document or the right of such Purchaser to rely thereon, each Purchaser hereby: (i) acknowledges that it has received all the information it has requested from the Company and it considers necessary or appropriate for deciding whether to acquire the Securities, (ii) represents that it has had an opportunity to ask questions and receive answers from the Company regarding the terms and conditions of the offering of the Securities and to obtain any additional information necessary to verify the accuracy of the information given the Purchaser, and (iii) further represents that it has such knowledge and experience in financial and business matters that it is capable of evaluating the merits and risk of this investment.
- **8.3 Ability to Bear Economic Risk.** Each Purchaser acknowledges that investment in the Securities involves a high degree of risk, and represents that it is able, without materially impairing its financial condition, to hold the Securities for an indefinite period of time and to suffer a complete loss of its investment.
- **8.4** Further Limitations on Disposition. Without in any way limiting the representations set forth above, each Purchaser further agrees not to make any disposition of all or any portion of the Securities unless and until:
- (a) there is then in effect a registration statement under the Act covering such proposed disposition and such disposition is made in accordance with such registration statement; or

EXECUTION VERSION

(b) the Purchaser shall have notified the Company of the proposed disposition and shall have furnished the Company with a detailed statement of the circumstances surrounding the proposed disposition, and if reasonably requested by the Company, such Purchaser shall have furnished the Company with an opinion of counsel, reasonably satisfactory to the Company, that such disposition will not require registration under the Act or any applicable state securities laws, provided that no such opinion shall be required for dispositions in compliance with Rule 144, except in unusual circumstances.

Notwithstanding the provisions of paragraphs (a) and (b) above, no such registration statement or opinion of counsel shall be necessary for a transfer by such Purchaser to Permitted Transferees, or transfers by gift, will or intestate succession to any spouse or lineal descendants or ancestors of any Purchaser who is an individual, if all transferees agree in writing to be subject to the terms hereof to the same extent as if they were Purchasers hereunder.

- 8.5 Accredited Investor Status. Each Purchaser is an "accredited investor" as such term is defined in Rule 501 under the Act.
- 8.6 No "Bad Actor" Disqualification. Each Holder represents and warrants that neither (A) such Holder nor (B) any entity that controls such Holder or is under the control of, or under common control with, such Holder, is subject to any Disqualification Event, except for Disqualification Events covered by Rule 506(d)(2)(ii) or (iii) or (d)(3) under the Act and disclosed in writing in reasonable detail to the Company. Each Holder represents that such Holder has exercised reasonable care to determine the accuracy of the representation made by such Holder in this paragraph, and agrees to notify the Company if such Holder becomes aware of any fact that makes the representation given by such Holder hereunder inaccurate.
- **8.7** Foreign Investors. If a Holder is not a United States person (as defined by Section 7701(a)(30) of the Internal Revenue Code of 1986, as amended (the "Code")), such Holder hereby represents that he, she or it has satisfied itself as to the full observance of the laws of the Holder's jurisdiction in connection with any invitation to subscribe for the Securities or any use of its Note, including (A) the legal requirements within the Holder's jurisdiction for the purchase of the Securities, (B) any foreign exchange restrictions applicable to such purchase, (C) any governmental or other consents that may need to be obtained, and (D) the income tax and other tax consequences, if any, that may be relevant to the purchase, holding, redemption, sale or transfer of the Securities. The Holder's subscription, payment for and continued beneficial ownership of the Securities will not violate any applicable securities or other laws of the Holder's jurisdiction.
- **8.8** Forward-Looking Statements. With respect to any forecasts, projections of results and other forward-looking statements and information provided to a Holder, such Holder acknowledges that such statements were prepared based upon assumptions deemed reasonable by the Company at the time of preparation. There is no assurance that such statements will prove accurate, and the Company has no obligation to update such statements.

- 8.9 GreensLedge as Placement Agent. Each Purchaser represents, acknowledges and agrees that: (A) GreensLedge Capital Markets LLC ("GreensLedge") has acted as the Company's placement agent for the Securities, (B) such Purchaser is not relying on the advice or recommendations of GreensLedge (including any affiliate, agent, advisor or representative thereof) in connection with such Purchaser's purchase of Securities, (C) GreensLedge is not acting as an underwriter or initial purchaser with respect to any Securities, (D) GreensLedge has no responsibility with respect to any marketing or other disclosure documents relating to the Securities (or the completeness of any thereof) furnished to such Purchaser, (E) GreensLedge has not made, and will not make, any representation or warranty with respect to the Company or any Securities (and such Purchaser will not rely on any statements made by GreensLedge, orally or in writing, to the contrary) and (F) GreensLedge and/or any affiliate or employee thereof may purchase or otherwise invest in the Securities.
- **8.10** Further Assurances. Each Purchaser agrees and covenants that at any time and from time to time it will promptly execute and deliver to the Company such further instruments and documents and take such further action as the Company may reasonably require in order to carry out the full intent and purpose of this Agreement and to comply with state or federal securities laws or other regulatory approvals.

9. THE AGENTS

9.1 Appointment and Authorization of the Agents. Each Purchaser hereby irrevocably appoints, designates and authorizes the Note Agent to act as the "note agent" under the Note Documents and the Collateral Agent to act as the "collateral agent" under the Note Documents and to act as the agent of (and to hold any security interest created by any Note Document for and on behalf of or on trust for) such Purchaser for purposes of acquiring, holding and enforcing any and all Liens on Collateral granted by the Company to secure any of the Obligations, and to take such other action on its behalf in accordance with the provisions of this Agreement and each other Note Document and to exercise such powers and perform such duties, in each case as are expressly delegated to the Agents by the terms of this Agreement or any other Note Document, together with such powers and discretion as are reasonably incidental thereto. Notwithstanding any provision to the contrary contained elsewhere herein or in any other Note Document, the Agents shall have no duties or responsibilities, except those expressly set forth herein, nor shall the Agents have or be deemed to have any fiduciary relationship with any Purchaser or participant, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Note Document or otherwise exist against the Agents, Without limiting the generality of the foregoing sentence, the use of the term "agent" herein and in the other Note Documents with reference to the Agents is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead, such term is used merely as a matter of market custom and is intended to create or reflect only an administrative relationship between independent contracting parties. Without limiting the generality of the foregoing, the Purchasers hereby expressly authorize the Collateral Agent to execute any and all documents (including releases) with respect to the Collateral and the rights of the Secured Parties with respect thereto, as contemplated by and in accordance with the provisions of this Agreement and the Note Documents and acknowledge and agree that any such action by Collateral Agent shall bind the Purchasers. Whether or not expressly stated in any Note Document, the rights, privileges and

immunities of the Agents shall be automatically incorporated by reference therein. Whether or not a party thereto, the Agents shall be express third party beneficiaries of the Notes, including without limitation the payment waterfall set forth therein.

9.2 Liability of Agents. No Agent-Related Person shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Note Document or the transactions contemplated hereby, or (b) be responsible in any manner to any Purchaser for any recital, statement, representation or warranty made by the Company or any officer thereof, contained herein or in any other Note Document, or in any certificate, report, statement or other document referred to or provided for in, or received by the Agents under or in connection with, this Agreement or any other Note Document, or the validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Note Document, or the creation, perfection, maintenance of perfection or priority of any Lien or security interest created or purported to be created under the Note Documents, the convertibility of the Notes and the validity or the sufficiency of the Equity Securities (as defined in the Note), or for any failure of the Company or any other party (other than Agents) to any Note Document to perform its obligations hereunder or thereunder, except to the extent such loss resulted from the gross negligence or willful misconduct of such Agent-Related Person, as determined by a final nonappealable order of a court of competent jurisdiction. The Note Agent shall have no obligation to monitor the Ledger in the absence of being providing such by the Company in accordance with the terms of the Notes, and may, in its sole discretion either: (i) conclusively rely on the Ledger most-recently delivered to it, (ii) conclusively rely on a statement by a Holder as to the outstanding principal amount of Notes held by it, or (iii) refrain from taking any action until the Note Agent receives a current Ledger from the Company. No Agent-Related Person shall be under any obligation to any Purchaser or participant to ascertain or to inquire as to the observance or performance of any of the agreements contained in, o

The Agents shall not have any duties or obligations except those expressly set forth in the Note Documents. Without limiting the generality of the foregoing, (i) the Agents shall not be subject to any fiduciary or other implied duties, regardless of whether a Default or Event of Default has occurred and is continuing and (ii) the Agents shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby that Collateral Agent is instructed in writing to exercise by the Required Holders (or such other requisite number or percentage of Holders as provided in Section 10.9).

In no event shall the Agents be responsible or liable for any failure or delay in the performance of its obligations hereunder arising out of or caused by, directly or indirectly, forces beyond its control, including, without limitation, strikes, work stoppages, accidents, acts of war or terrorism, epidemics, pandemics, civil or military disturbances, nuclear or natural catastrophes or acts of God, and interruptions, loss or malfunctions of utilities, communications or computer (software and hardware) services or the unavailability of the Federal Reserve Bank wire or telex or other wire or communication facility, it being understood that the Agents shall use reasonable best efforts which are consistent with accepted practices in the banking industry to resume performance as soon as practicable under the circumstances.

- 9.3 Reliance by the Agents. The Agents shall be entitled to rely, and shall be fully protected in relying, upon any writing, communication, signature, resolution, representation, notice, consent, certificate, affidavit, letter, facsimile or telephone message, electronic mail message, statement or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper person or persons, and upon advice and statements of legal counsel (including counsel to the Company), independent accountants and other experts selected by the Agents. The Agents shall be fully justified in failing or refusing to take any action under any Note Document unless it shall first receive such advice or concurrence of the Required Holders as it deems appropriate and, if it so requests, it shall first be indemnified to its satisfaction by the Holders against any and all loss, liability and expense which may be incurred by it by reason of taking or continuing to take any such action. The Agents shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement or any other Note Document in accordance with a request or consent of the Required Holders (or such greater number of Holders as may be expressly required hereby in any instance) and such request and any action taken or failure to act pursuant thereto shall be binding upon all the of the same; provided that the Agents shall not be required to take any action that, in its opinion or in the opinion of its counsel, may expose the Agents to liability or that is contrary to any Note Document or applicable law.
- 9.4 Notice of Default. The Agents shall not be deemed to have knowledge or notice of the occurrence of any Event of Default, unless the Agents shall have received written notice from a Purchaser referring to this Agreement, describing such Event of Default and stating that such notice is a "notice of event of default." The Agents shall take such action with respect to any Event of Default as may be directed by the Required Holders in accordance with the terms of the Notes; provided that unless and until the Agents has received any such direction, the Agents may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Event of Default as it shall deem advisable or in the best interest of the Holders.
- 9.5 Agent's Reimbursement. Each of the Holders severally agrees to reimburse the Agents, in accordance with such Holder's Pro Rata Share, for any reasonable and documented out-of-pocket expenses not reimbursed by the Company (without limiting the obligation of the Company to make such reimbursement pursuant to Section 10.10): (a) for which the Agents are entitled to reimbursement by the Company under this Agreement or any Note Document and (b) after the occurrence and during the continuance of an Event of Default, for any other reasonable and documented expenses incurred by the Agents on the Holders' behalf in connection with the enforcement of the Holders' rights under this Agreement or any Note Document; provided, however, that (x) the Agents shall not be reimbursed for any such expenses arising as a result of its gross negligence or willful misconduct, as determined by a final nonappealable order of a court of competent jurisdiction and (y) in the event that the Company reimburses the Agents for any such reasonable and documented out-of-pocket expenses, any corresponding amounts previously reimbursed by the Holders to the Agents will be promptly returned to such Holders in accordance with their Pro Rata Share.
- **9.6 Indemnification.** Each of the Holders shall severally indemnify the Agents and their officers, directors, employees, agents, attorneys, accountants, consultants and controlling Persons (to the extent not reimbursed by or on behalf of the Company pursuant to <u>Section 10.10</u> and without limiting the obligations of the Company to do so), in accordance with their

respective Pro Rata Share, from and against any and all liabilities, obligations, damages, penalties, actions, judgments, suits, losses (including accrued and unpaid Agents' fees), and reasonable and documented costs, expenses or disbursements of any kind whatsoever which may at any time be imposed on, incurred by or asserted against the Agents or such Persons relating to or arising out of this Agreement, any Note Document, the transactions contemplated hereby or thereby, or any action taken or omitted by the Agents in connection with any of the foregoing; provided, however, that the foregoing shall not extend to actions or omissions to the extent arising from gross negligence or willful misconduct of the Agents, as determined by a final nonappealable order of a court of competent jurisdiction.. The Agents shall not be under any obligation to exercise any of the rights or powers vested in it by this Agreement at the request, order or direction of any of the Holders, pursuant to any provision of this Agreement, unless the Required Holders shall have offered (and, if requested, provided) to the Agents security or indemnity satisfactory to it against the costs, expenses and liabilities which may be incurred by it therein or thereby. The undertaking in this Section 9.6 shall survive the payment of all other Obligations and the resignation of the Agents.

- 9.7 Collateral Matters. The Purchasers irrevocably agree that any Lien on any property granted to or held by the Collateral Agent under any Note Document for the benefit of the Secured Parties shall be automatically released (i) upon payment in full of all Obligations (it being understood and agreed that the conversion in full of a Note by a Holder shall be deemed, for purposes of this Section 9.6, to be a repayment of the entire outstanding principal amount (including all capitalized interest) of such Note together with any unpaid accrued interest thereon on the date of such conversion), (ii) subject to Section 10.9, if the release of such Lien is approved, authorized or ratified in writing by the Purchasers or (iii) upon the sale, transfer or other disposition of any Collateral that is not prohibited by the Note Documents. Upon request by the Collateral Agent at any time, the Purchasers will confirm in writing the Collateral Agent's authority to release particular types or items of property. In each case as specified in this Section 9.7, the Collateral Agent will promptly (and each Purchaser irrevocably authorizes the Collateral Agent to), at the Company's expense, execute and deliver to the Company such documents as the Company may reasonably request to evidence the release of such item of Collateral from the assignment and security interest granted under the Note Documents. In connection with any such release, the Collateral Agent shall be entitled to a certificate of a Responsible Officer of the Company stating that such release is authorized and permitted by the Note Documents, upon which the Collateral Agent may conclusively rely. Each party to this Agreement acknowledges and agrees that the Agents shall not have an obligation to file financing statements, amendments to financing statements, or continuation statements, or to perfect or maintain the perfection of the Collateral Agent's Lien on the Collateral.
- **9.8** Successor Agents. The Agents may resign as Agents upon fifteen (15) days' notice to the Holders. If an Agent resigns under this Agreement, the Holders shall unanimously appoint a successor representative for the Holders. If no successor representative is appointed prior to the effective date of the resignation of the Agent, the resigning Agent may appoint, after consulting with the Holders and the Company, a successor agent. Upon the acceptance of its appointment as successor representative hereunder, such successor representative shall succeed to all the rights, powers and duties of the retiring Agent, the term "Agents" shall mean such successor representative and the retiring Agent's appointment, powers and duties as Agents shall be terminated. After any retiring Agent's resignation hereunder as Agent, the provisions of this Section 9 and Section 10 shall inure to its benefit as to any actions taken or omitted to be taken

by it while it was Agent under this Agreement. If no successor representative has accepted appointment as Agent by the date which is fifteen (15) days following a retiring Agent's notice of resignation, the retiring Agent's resignation shall nevertheless thereupon become effective and the Holders shall perform all of the duties of the Agents hereunder until such time, if any, as the Holders appoint a successor representative as provided for above.

9.9 Intercreditor Agreement. The Collateral Agent is hereby authorized to enter into an Intercreditor Agreement pursuant to Section 7.1(a), and the parties hereto acknowledge that such Intercreditor Agreement is binding upon them. Each Secured Party hereby (a) agrees that it will be bound by and will take no actions contrary to the provisions of the Intercreditor Agreement, (b) authorizes and instructs the Collateral Agent, without any further consent of such Secured Party, to enter into the Intercreditor Agreement provided by the Company and accompanied by the certificate specified in Section 7.1(a) and to subject the Liens on the Collateral securing the Obligations to the provisions thereof and (c) authorizes and instructs the Collateral Agent to execute and deliver on behalf of the Secured Creditors any amendment (or amendment and restatement) or modification to the Intercreditor Agreement to provide for the incurrence of any Indebtedness permitted hereunder or such other amendments as the Required Holders may specify in writing to the Collateral Agent.

10. MISCELLANEOUS

- 10.1 Binding Agreement. The terms and conditions of this Agreement shall inure to the benefit of and be binding upon the respective successors and assigns of the parties. Nothing in this Agreement, expressed or implied, is intended to confer upon any third party any rights, remedies, obligations, or liabilities under or by reason of this Agreement, except as expressly provided in this Agreement.
- 10.2 GOVERNING LAW. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES (OTHER THAN SECTIONS 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW) THEREOF.
- 10.3 JURISDICTION AND VENUE, EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY SUBMITS TO THE NON-EXCLUSIVE JURISDICTION OF ANY U.S. FEDERAL OR NEW YORK STATE COURT SITTING IN THE BOROUGH OF NEW YORK, NEW YORK IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR ANY OTHER NOTE DOCUMENT AND EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY AGREES THAT ALL CLAIMS IN RESPECT OF SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH COURT AND IRREVOCABLY WAIVES ANY OBJECTION IT MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN SUCH A COURT OR THAT SUCH COURT IS AN INCONVENIENT FORUM. NOTHING HEREIN SHALL LIMIT THE RIGHT OF THE COLLATERAL AGENT, ANY PURCHASER OR ANY HOLDER TO BRING PROCEEDINGS AGAINST THE COMPANY IN THE COURTS OF ANY OTHER

JURISDICTION. ANY JUDICIAL PROCEEDING BY THE COMPANY AGAINST THE COLLATERAL AGENT, ANY PURCHASER, ANY HOLDER OR ANY OF THEIR AFFILIATES INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH THIS AGREEMENT OR ANY OTHER NOTE DOCUMENT SHALL BE BROUGHT ONLY IN A COURT IN THE BOROUGH OF NEW YORK, NEW YORK.

- 10.4 WAIVER OF JURY TRIAL. EACH OF THE COMPANY, THE PURCHASERS, THE HOLDERS AND THE COLLATERAL AGENT HEREBY WAIVES TRIAL BY JURY IN ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER (WHETHER SOUNDING IN TORT, CONTRACT OR OTHERWISE) IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH THIS AGREEMENT OR ANY OTHER NOTE DOCUMENT OR THE RELATIONSHIP ESTABLISHED THEREUNDER.
- 10.5 Severability. If one or more provisions of this Agreement are held to be unenforceable under applicable law, the parties agree to renegotiate such provision in good faith. In the event that the parties cannot reach a mutually agreeable and enforceable replacement for such provision, then (a) such provision shall be excluded from this Agreement, (b) the balance of this Agreement shall be interpreted as if such provision were so excluded, and (c) the balance of this Agreement shall be enforceable in accordance with its terms.
- 10.6 Counterparts. This Agreement may be executed in one or more counterparts (and by different parties hereto in different counterparts), each of which shall be deemed an original, but all of which together shall constitute a single contract. Any counterpart of a signature page to this Agreement may be delivered by facsimile, electronic mail (including ".pdf" or ".tif") or by means of an electronic signature complying with the U.S. federal ESIGN Act of 2000 or the New York Electronic Signature and Records Act or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes to the fullest extent permitted by applicable law. The Note Documents and all notices, approvals, consents, requests and any communications hereunder must be in writing (provided that any such communication sent to Agents hereunder must be in the form of a document that is signed manually or by way of a digital signature provided by DocuSign (or such other digital signature provider as specified in writing to the Agents by the authorized representative), in English. The Company agrees to assume all risks arising out of the use of using digital signatures and electronic methods to submit communications to the Note Agent, including without limitation the risk of the Note Agent acting on unauthorized instructions, and the risk of interception and misuse by third parties.
- 10.7 Headings; Interpretation. Paragraph headings used in this Agreement are included for convenience of reference only and will not modify the provisions that they precede or affect the interpretation of this Agreement or any other Note Document. The term "including" shall be interpreted to mean "including but not limited to". Any of the terms defined herein may, unless the context otherwise requires, be used in the singular or the plural, depending on the reference.

- 10.8 Notices. All notices required or permitted hereunder shall be in writing and shall be deemed effectively given: (a) upon personal delivery to the party to be notified, (b) when sent by confirmed electronic mail or facsimile if sent during normal business hours of the recipient (and if not, then on the next Business Day), (c) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid or (d) one (1) day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written verification of receipt. Notices to the Agents shall be effective upon actual receipt thereof. All communications shall be sent, if to the Note Parties, at the address set forth on the signature page of the Company, if to the Agents, at the address set forth on the signature page thereof, and, if to a Purchaser, at the address(es) set forth on the Schedule of Purchasers attached hereto as Schedule 2 or at such other address(es) as the Company or Purchaser may designate by ten (10) days' advance written notice to the other parties hereto. Except where notice is specifically required by this Agreement, no notice to or demand on the Company or any of its Subsidiaries in any case shall entitle the Company or any of its Subsidiaries to any other or further notice or demand in similar or other circumstances.
- 10.9 Amendments; Waiver. Any term of the Notes and this Agreement may be amended or waived with the written consent of the Company and the Required Holders; provided, that without the consent of each Holder, no amendment, modification, termination, or consent shall be effective if the effect thereof would (a) extend the scheduled final maturity of any Note held by any non-consenting Holder, (b) waive, reduce or postpone any scheduled repayment (but not prepayment) with respect to the Note held by any non-consenting Holder; (c) reduce the rate of interest or amount of any fees payable under any Note held by any non-consenting Holder; (d) extend the time for payment of any interest or fees payable under any Note held by any non-consenting Holder; (e) reduce the principal amount of any Note held by any non-consenting Holder; (f) amend, modify, terminate or waive any provision of this Section 10.9; (g) amend the definition of "Required Holders"; (h) release all or substantially all of the Collateral securing any Note held by any non-consenting Holder except as expressly provided in the Note Documents; (j) release any Guarantor except as expressly provided in the Note Documents or (k) consent to the assignment or transfer by the Company of any of its rights and obligations under any Note Document held by any non-consenting Holder. No amendment affecting the rights, privileges and immunities of the Agents shall be effective without the consent of such Agent.

10.10 Expenses and Indemnification.

(a) Solely to the extent the Initial Closing Date occurs (with respect to any Initial Purchaser), the Company shall pay all reasonable and documented out-of-pocket costs and expenses (including reasonable and documented attorneys' fees and expenses) incurred by such Initial Purchasers in connection with the preparation, execution and delivery of this Agreement and the other Note Documents, in an amount not to exceed \$100,000 in the aggregate. The Company shall pay all reasonable and documented out-of-pocket costs and expenses of the Agents (including reasonable and documented fees, expenses and disbursements of its outside counsel) relating to the negotiation, preparation and execution of the Note Documents, review of other documents (including for purposes of due diligence review) in connection with the transactions contemplated hereby and any amendments and waivers hereto or thereto. In

EXECUTION VERSION

addition, the Company agrees to promptly pay in full after the occurrence of an Event of Default, all costs and expenses (including, without limitation, reasonable and documented fees and disbursements of counsel, agents and professional advisers) incurred by the Holders or the Agents in enforcing any obligations of or in collecting any payments due hereunder or under the Notes by reason of such Event of Default or in connection with any refinancing or restructuring of the credit arrangements provided under this Agreement in the nature of a workout, or any insolvency or bankruptcy proceedings.

(b) In addition to the payment of expenses pursuant to Section 10.10(a), the Company (as "Indemnitor") agrees to indemnify, pay and hold the Purchasers, the Holders and the Agents, and the officers, directors, employees, agents, and affiliates of the Purchasers, the Holders and the Agents (collectively called the "Indemnitees") harmless from and against any and all other liabilities, costs, expenses, liabilities, obligations, losses, damages, penalties, actions, judgments, suits, claims and disbursements of any kind or nature whatsoever (including, without limitation, the reasonable fees and disbursements of counsel for such Indemnitees) in connection with any investigative, administrative or judicial proceeding commenced or threatened (excluding claims among Indemnitees (other than claims against an Agent acting in its capacity as such) and, with the exception of claims arising out of otherwise indemnifiable matters (e.g., actions to enforce the indemnification rights provided hereunder), and excluding claims between the Company and an Indemnitee), whether or not such Indemnitee shall be designated a party thereto, which may be imposed on, incurred by, or asserted against that Indemnitee, in any manner relating to or arising out of this Agreement, the Notes, the Note Documents or the other documents related to the transactions contemplated hereby (including, without limitation, the existence or exercise of any security rights with respect to the Collateral in accordance with the Security Agreement), the Purchasers' agreement to purchase the Notes or the use or intended use of the proceeds of any of the proceeds thereof to the Company (the "Indemnified Liabilities"); provided, that the Indemnitor shall not have any obligation to an Indemnitee hereunder with respect to an Indemnified Liability to the extent that such Indemnified Liability arises from the gross negligence or willful misconduct of that Indemnitee as determined by a final nonappealable order of a court of competent jurisdiction. Each Indemnitee shall give the Indemnitor prompt written notice of any claim that might give rise to Indemnified Liabilities setting forth a description of those elements of such claim of which such Indemnitee has knowledge; provided, that any failure to give such notice shall not affect the obligations of the Indemnitor unless (and then solely to the extent) such Indemnitor is not aware of such claim and is materially prejudiced. The Indemnitor shall have the right at any time during which such claim is pending to select counsel to defend and control the defense thereof and settle any claims for which it is responsible for indemnification hereunder (provided that the Indemnitor will not settle any such claim without (i) the appropriate Indemnitee's prior written consent, which consent shall not be unreasonably withheld or (ii) obtaining an unconditional release of the appropriate Indemnitee from all claims arising out of or in any way relating to the circumstances involving such claim) so long as in any such event the Indemnitor shall have stated in a writing delivered to the Indemnitee that, as between the Indemnitor and the Indemnitee, the Indemnitor is responsible to the Indemnitee with respect to such claim to the extent and subject to the limitations set forth herein; provided, that the Indemnitor shall not be entitled to control the defense of any claim in the event that in the reasonable opinion of counsel for the Indemnitee, there are one or more material defenses available to the Indemnitee which are not available to the Indemnitor, in which case the Indemnitee may retain separate counsel and the

Company will pay the reasonable fees and expenses of such counsel (including the reasonable fees and expenses of counsel to such Indemnitee incurred in evaluating whether such a conflict exists); provided further, that with respect to any claim as to which the Indemnitee is controlling the defense, the Indemnitor will not be liable to any Indemnitee for any settlement of any claim pursuant to this Section 10.10(b) that is effected without its prior written consent, which consent shall not be unreasonably withheld. To the extent that the undertaking to indemnify, pay and hold harmless set forth in this Section 10.10(b) may be unenforceable because it is violative of any law or public policy, the Company shall contribute the maximum portion which it is permitted to pay and satisfy under applicable law, to the payment and satisfaction of all Indemnified Liabilities incurred by the Indemnitees or any of them. The obligations of each of the parties under this Section 10.10 will survive the payment or transfer of any Note, the enforcement, amendment or waiver of any provision of this Agreement, the resignation or removal of any Agent and the termination of this Agreement.

- (c) (c) To the extent permitted by applicable law, none of the parties hereto shall assert, and each of the parties hereto hereby waives, any claim against the other parties (including their respective Affiliates, partners, stockholders, members, directors, officers, agents, employees and controlling persons), on any theory of liability for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement or any agreement or instrument contemplated hereby, the transactions contemplated hereunder, any Note Document, the Notes or the use of the proceeds thereof; provided that nothing contained in this Section 10.10(c) shall limit the Notes Parties' indemnification and reimbursement obligations to the extent set forth in this Agreement.
- 10.11 Delays or Omissions. It is agreed that no delay or omission to exercise any right, power or remedy accruing to each Purchaser, upon any breach or default of the Company under this Agreement or any other Note Document shall impair any such right, power or remedy, nor shall it be construed to be a waiver of any such breach or default, or any acquiescence therein, or of or in any similar breach or default thereafter occurring; nor shall any waiver of any single breach or default be deemed a waiver of any other breach or default thereafter occurring. It is further agreed that any waiver, permit, consent or approval of any kind or character by a Purchaser of any breach or default under this Agreement, or any waiver by such Purchaser of any provisions or conditions of this Agreement must be in writing and shall be effective only to the extent specifically set forth in writing and that all remedies, either under this Agreement, or by law or otherwise afforded to such Purchaser, shall be cumulative and not alternative.
- 10.12 Waiver of Conflicts. Each party to this Note acknowledges that Cooley LLP ("Cooley"), outside general counsel to the Company, has in the past performed and is or may now or in the future represent the Holders or the Holders' affiliates in matters unrelated to the transactions contemplated by this Note (the "Note Financing"), including representation of the Holders or the Holders' affiliates in matters of a similar nature to the Note Financing. The applicable rules of professional conduct require that Cooley inform the parties hereunder of this representation and obtain their consent. Cooley has served as outside general counsel to the Company and has negotiated the terms of the Note Financing solely on behalf of the Company. The Company and the Holders hereby (i) acknowledge that they have had an opportunity to ask for and have obtained information relevant to such representation, including disclosure of the

EXECUTION VERSION

reasonably foreseeable adverse consequences of such representation; (ii) acknowledge that with respect to the Note Financing, Cooley has represented solely the Company, and not any Holder or any stockholder, board member or employee of the Company or director, stockholder or employee of the Holder; and (iii) gives the Holder's informed consent to Cooley's representation of the Company in the Note Financing.

- 10.13 Obligations Several. The Purchasers' obligations hereunder are several and not joint obligations and no Purchaser shall have any liability to any Person for the performance or non-performance of any obligation by any other Purchaser hereunder.
- 10.14 Survival of Representations and Warranties. All of the representations and warranties made by the Note Parties and their Subsidiaries herein shall survive the execution and delivery of this Agreement, any investigation by or on behalf of any Purchaser, acceptance of the Notes and payment therefor, or termination of this Agreement
- 10.15 Entire Agreement. This Agreement and the other Note Documents constitute the entire contract among the parties hereto regarding to the subject matters addressed herein and supersede any and all previous agreements, negotiations, and discussions, oral or written, by the parties regarding the subject matters addressed herein.
- 10.16 Withholding. The Company shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement such amounts as it is required to deduct and withhold under the Code, or any Tax law, with respect to the making of such payment. To the extent that amounts are so withheld, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of whom such deduction and withholding was made. If a payment is payable (in whole or in part) in consideration other than cash and if the cash portion of any such payment is insufficient to satisfy all required Tax withholding obligations, the Company shall retain an amount of the non-cash consideration otherwise payable equal in value to the amount required to satisfy any applicable withholding taxes (as reasonably determined by the Company).
- 10.17 PATRIOT ACT. The Agents hereby notify the Company that pursuant to the requirements of the PATRIOT Act, the Company may be required to obtain, verify and record information that identifies the Company, its subsidiaries and the Guarantors, including their respective names, addresses and other information that will allow the Agent to identify, the Company, its subsidiaries and the Guarantors in accordance with the PATRIOT Act.

[Signature pages follow]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first written above.

COMPANY:

CORE SCIENTIFIC HOLDING CO.

By: /s/ Michael Trzupek Name: Michael Trzupek Title: Chief Financial Officer

Address for notices:

Attention: Michael Trzupek, Chief Financial Officer

Email:

Copy to:

Todd DuChene, General Counsel

Email:

In each case, with a copy (which shall not constitute notice)

to:

Cooley LLP 1299 Pennsylvania AVE, NW Suite 700 Washington, DC 20004

Attention: Mike Tollini
Email: mtollini@cooley.com

GUARANTORS:

CORE SCIENTIFIC, INC.

By: /s/ Michael Trzupek Name: Michael Trzupek Title: Chief Financial Officer

AMERICAN PROPERTY ACQUISITION, LLC

By: Core Scientific, Inc., its sole member

By: /s/ Michael Trzupek Name: Michael Trzupek Title: Chief Financial Officer

AMERICAN PROPERTY ACQUISITIONS I, LLC

By: American Property Acquisition, LLC, its sole member

By: Core Scientific, Inc., its sole member

By: /s/ Michael Trzupek Name: Michael Trzupek Title: Chief Financial Officer

AMERICAN PROPERTY ACQUISITIONS VII, LLC

By: American Property Acquisition, LLC, its sole member

By: Core Scientific, Inc., its sole member

By: /s/ Michael Trzupek Name: Michael Trzupek Title: Chief Financial Officer

$U.S.\ BANK\ NATIONAL\ ASSOCIATION,\ AS\ NOTE$ AGENT

By: <u>/s/ Joshua A.</u> Hahn Name: Joshua A. Hahn

Title: Vice President

Address for notices: U.S. Bank National Association West Side Flats 60 Livingston Avenue EP-MN-WS3C St. Paul, MN 55107

Attention: Joshua Hahn (Core Sciences)

In each case, with a copy (which shall not constitute notice)

Shipman & Goodwin LLP One Constitution Plaza Hartford, CT 06103

Attn: Nathan Plotkin (Core Scientific)

U.S. BANK NATIONAL ASSOCIATION, AS COLLATERAL AGENT

By: /s/ Joshua A. Hahn Name: Joshua A. Hahn Title: Vice President

Address for notices: U.S. Bank National Association West Side Flats 60 Livingston Avenue EP-MN-WS3C St. Paul, MN 55107 Attention: Joshua Hahn (Core Sciences)

In each case, with a copy (which shall not constitute notice)

Shipman & Goodwin LLP One Constitution Plaza Hartford, CT 06103 Attn: Nathan Plotkin (Core Scientific)

PURCHASER:

By: /s/ Roni Cohen-Pavon

Name: Roni Cohen-Pavon

Title: Chief Revenue Officer, Celsius Network Ltd.

KENSICO ASSOCIATES, L.P. By: Kensico Capital Management Corp, its Investment Manager

By: /s/ Israel Friedman Name: Israel Friedman Title: General Counsel & CCO

KENSICO OFFSHORE FUND MASTER, LTD By: Kensico Capital Management Corp, its Investment Manager

By: /s/ Israel Friedman Name: Israel Friedman Title: General Counsel & CCO

By: /s/ William Guthy

Name: William Guthy
Title: Trustee of the William R. Guthy Separate Property
Trust

-							
v	ш	CH	A	CI	Б.	D	

By: /s/ Ken Wormser

Name: Ken Wormser

THE KIMMEL FAMILY FOUNDATION

By: /s/ Adam Kimmel Name: Adam Kimmel Title: Director

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO TACTICAL VALUE SPN INVESTMENTS LP By: Apollo Tactical Value SPN Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO CENTRE STREET PARTNERSHIP, LP By: Apollo Centre Street Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO MOULTRIE CREDIT FUND, L.P. By: Apollo Moultrie Credit Fund Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO LINCOLN FIXED INVOME FUND, L.P. By: Apollo Lincoln Fixed Income Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO CENTRE STREET PARTNERSHIP, LP By: Apollo Centre Street Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO TACTICAL VALUE SPN INVESTMENTS LP By: Apollo Tactical Value SPN Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO LINCOLN FIXED INCOME FUND, L.P. By: Apollo Lincoln Fixed Income Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO MOULTRIE CREDIT FUND, L.P. By: Apollo Moultrie Credit Fund Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

ICG CORESCI HOLDINGS, LP

By: /s/ Robert Edelstein Name: Robert Edelstein Title: Authorized Person

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

The Michael O. Johnson Revocable Trust

By: /s/ Michael O. Johnson Name: Michael O. Johnson

Title: Trustee

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

IBEX PARTNERS (CORE) LP By: Ibex GP LLC, its General Partner

By: /s/ Brian T. Abrams Name: Brian T. Abrams Title: President

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

ICG CORESCI HOLDINGS, LP

By: /s/ Robert Edelstein Name: Robert Edelstein Title: Authorized Person

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

GREENSLEDGE MERCHANT HOLDINGS LLC

By: /s/ Ken Wormser Name: Ken Wormser Title: Managing Partner

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO CENTRE STREET PARTNERSHIP, LP By: Apollo Centre Street Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO TACTICAL VALUE SPN, INVESTMENTS LP By: Apollo Tactical Value SPN Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO MOULTRIE CREDIT FUND, L.P. By: Apollo Moultrie Credit Fund Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

APOLLO LINCOLN FIXED INCOME FUND, L.P. By: Apollo Lincoln Fixed Income Management, LLC, its investment manager

The undersigned hereby confirms that, upon execution of this signature page, it hereby agrees to be bound by all of the terms, provisions and conditions contained herein and is hereby joined as an "Additional Purchaser" and a "Purchaser" hereunder.

ADDITIONAL PURCHASER:

BIRCH GROVE CREDIT STRATEGIES MASTER FUND LP

By: /s/ Todd A. Berry
Name: Todd A. Berry
Title: Authorized Signatory

SCHEDULE 2

SCHEDULE OF PURCHASERS

NOTE PARTIES

SUBSIDIARIES

CAPITALIZATION

INDEBTEDNESS

EXHIBIT A

FORM OF SECURED CONVERTIBLE PROMISSORY NOTE

(see attached)

THIS NOTE AND THE SECURITIES ISSUABLE UPON THE CONVERSION HEREOF HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR UNDER THE SECURITIES LAWS OF ANY STATES IN THE UNITED STATES. THESE SECURITIES ARE SUBJECT TO RESTRICTIONS ON TRANSFERABILITY AND RESALE AND MAY NOT BE TRANSFERRED OR RESOLD EXCEPT PURSUANT TO SECTION 5(C) HEREOF AND AS PERMITTED UNDER THE ACT AND THE APPLICABLE STATE SECURITIES LAWS, PURSUANT TO REGISTRATION OR EXEMPTION THEREFROM. THE ISSUER OF THESE SECURITIES MAY REQUIRE AN OPINION OF COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER TO THE EFFECT THAT ANY PROPOSED TRANSFER OR RESALE IS IN COMPLIANCE WITH THE ACT AND ANY APPLICABLE STATE SECURITIES LAWS.

THIS NOTE HAS BEEN ISSUED WITH "ORIGINAL ISSUE DISCOUNT" (WITHIN THE MEANING OF SECTION 1273 OF THE INTERNAL REVENUE CODE OF 1986, AS AMENDED (THE "*CODE*")). UPON WRITTEN REQUEST, THE COMPANY WILL PROMPTLY MAKE AVAILABLE TO ANY HOLDER OF THIS NOTE THE FOLLOWING INFORMATION: (1) THE ISSUE PRICE AND ISSUE DATE OF THE NOTE, (2) THE AMOUNT OF ORIGINAL ISSUE DISCOUNT ON THE NOTE AND (3) THE YIELD TO MATURITY OF THE NOTE. HOLDERS SHOULD CONTACT THE CHIEF FINANCIAL OFFICER OF THE COMPANY AT 2800 NORTHUP WAY SUITE 220 BELLEVUE, WA 98004.

[FORM OF] SECURED CONVERTIBLE PROMISSORY NOTE

Note Series: 2021 Secured Convertible Promissory Notes

Date of Note: April 19, 2021

Initial Principal Amount of Note: \$[__], plus any PIK Interest that has accrued

and been capitalized pursuant to the terms of

this Note

For value received **CORE SCIENTIFIC HOLDING CO.**, a Delaware corporation (the "*Company*"), promises to pay to the undersigned holder or such party's successors or permitted assigns (the "*Holder*") the principal amount set forth above with interest to accrue on such outstanding principal amount at a rate of 10% *per annum*, 4% of which shall be payable in cash ("*Cash Interest*") and 6% of which shall be payable in kind by capitalizing such interest payment and increasing the outstanding principal amount of this Note by the amount thereof ("*PIK Interest*"). Interest shall accrue on the outstanding principal amount of this Note commencing on (and including) the date of original issuance thereof (or the most recent interest payment date) and continuing until the earlier to occur of (x) the date this Note is repaid or

prepaid in full and (y) the date this Note is converted in full by the Holder pursuant to Section 2 hereof. Cash Interest shall be due and payable and PIK Interest shall be capitalized quarterly in arrears on the first day of each fiscal quarter, commencing on July 1, 2021. Interest shall be computed on the basis of a year of 365 days for the actual number of days elapsed. This Note shall be automatically deemed to include any accrued PIK Interest thereon and the Company shall not be obligated to issue any subsequent Notes reflecting PIK Interest. This Note is one of the "Notes" issued pursuant to the Purchase Agreement. All capitalized terms used but not otherwise defined in this Note will have the same meanings in this Note as in the Purchase Agreement referred to below.

The following is a statement of the rights of the Holder and the conditions to which this Note is subject, and to which the Holder, by the acceptance of this Note, agrees:

2. BASIC TERMS.

- (a) Series of Notes. This secured convertible promissory note (this "Note") is issued as part of a series of notes designated by the Note Series above (collectively, the "Notes"), having an initial aggregate principal amount of up to \$215,000,000 and issued pursuant to, and to those persons and entities (collectively, the "Holders") party, as Purchasers, to, that certain Purchase Agreement, dated as of even date herewith (as amended, restated, supplemented and/or otherwise modified from time to time, the "Purchase Agreement"), by and among the Company, each Purchaser party thereto, U.S. Bank National Association, as Note Agent, and U.S. Bank National Association, as Collateral Agent for the Secured Parties. The Company shall maintain a ledger of all Holders ("Ledger"), which shall include all PIK Interest accrued and capitalized, and which shall be available (i) promptly upon request to the Agents and the Holder, (ii) promptly after any transfer or assignment of any Note, (iii) promptly upon conversion of any Note and (iv) on the tenth business day prior to any interest payment date.
 - (b) Payments.
 - (i) Voluntary Prepayments.
- (A) Prior to the occurrence of a Conversion Event, subject to the expiration of any applicable lockup period set forth under any lockup agreement to which the Holder, the Company and the Company's underwriters are a party, in the event that, following written request by the Company, the Holder does not timely consent to an amendment to the Purchase Agreement to increase the Secured Debt Cap, the Company may, upon ten (10) Business Days' prior written notice (which may be included in such written request) to any such non-consenting Holder (which notice may be conditioned or rescinded upon such events as may be specified in such notice) prepay all or any portion of each Note held by any non-consenting Holder in an amount equal to the sum of (A) the outstanding principal amount (including all accrued PIK Interest not already added to the principal amount of this Note) of the Note being prepaid at such time and (B) all accrued unpaid Cash Interest on such outstanding principal amount at such time (such amount, the "Covenant Amendment Repayment Amount"). In addition to paying the Covenant Amendment Repayment Amount, the Company shall issue to each such non-consenting Holder a warrant that will entitle the Holder to acquire, upon a Conversion Event, the number of shares of Equity Securities (or Common Stock in the event of a

Change of Control) equal to the quotient of (A) the principal amount of each Note being prepaid, divided by (B) the Applicable Conversion Price (which shall be determined, for the avoidance of doubt, on the date of the consummation of such Conversion Event). The per share exercise price for such warrants shall be the Applicable Conversion Price (which shall be determined, for the avoidance of doubt, on the date of the consummation of such Conversion Event).

- (B) On or after the occurrence of an Offering, subject to and following the expiration of any applicable lockup period set forth under any lockup agreement to which the Holder, the Company and the Company's underwriters are a party, and unless otherwise converted in full or in part pursuant to Section 2 hereof (including following delivery of the Company's notice referred to in this subsection (b)(i)(2)), the Company may at any time, upon ten (10) Business Days' prior written notice to the Holder and the Note Agent (which notice may be conditioned or rescinded upon such events as may be specified in such notice) prepay all or any portion of this Note in an amount equal to (x) the outstanding principal amount (including all accrued PIK Interest not already added to the principal amount of this Note) of this Note being prepaid at such time, together with all accrued unpaid Cash Interest on such outstanding principal amount at such time multiplied by (y) 200% (such amount, the "Repayment Amount").
- (ii) **Repayment**. Unless otherwise prepaid by the Company pursuant to Section 1(b)(i) or converted in full pursuant to Section 2 hereof, this Note shall be repaid in full on the earlier to occur of (x) April 19, 2025 (the "Maturity Date"), in an amount equal to the Repayment Amount, and (y) acceleration by the Holders representing at least twenty-five percent (25%) of the aggregate outstanding principal amount of the Notes issued pursuant to the Purchase Agreement at such time after the occurrence of an Event of Default in accordance with the terms hereof, in an amount equal to the outstanding principal (including all accrued PIK Interest not already added to the principal amount of this Note) of this Note, plus all unpaid accrued Cash Interest thereon.

(iii) Payments Generally.

- (A) This Note and the other Notes issued pursuant to the Purchase Agreement are *pari passu* in right of payment and all payments (other than conversion) under this Note and such other Notes shall be made in accordance with each Holder's Pro Rata Share. All payments shall be applied <u>first</u>, to the payment of expenses due under this Note and the other Note Documents to the Agents, <u>second</u>, to the payment of expenses due under this Note and the other Note Documents to the Holders, <u>third</u>, unpaid accrued interest of this Note and <u>fourth</u>, if the amount of payment exceeds the amount of all such expenses and accrued interest, to the payment of outstanding principal (including all accrued PIK Interest not already added to the principal amount of this Note) of this Note. The conversion of this Note by the Holder pursuant to the terms hereof shall be deemed to be a repayment of the full outstanding principal amount (including all accrued PIK Interest not already added to the principal amount of this Note) of such Note together with any unpaid accrued Cash Interest thereon on the date of such conversion.
- (B) All payments to be made by the Company shall be made without condition or deduction for any counterclaim, defense, recoupment or setoff, except with

respect to Taxes required to be deducted and/or withheld under applicable law. All payments (other than by means of conversion pursuant to the terms of the Notes), including any prepayments, of interest and principal to be made by the Company hereunder shall be made by the Company to the Note Agent, in cash, for the account of the respective Holders to which such payment is owed, at the applicable Notes Agent's Principal Office for payment and in same day funds not later than 11:00 a.m. on the date specified herein. The Notes Agent will promptly distribute to each Holder its Pro Rata Share (or other applicable share as provided herein) of such payment in like funds as received by wire transfer to each Holder at the address specified in the Holder's Administrative Questionnaire (or at such other address as the Holder may indicate in writing to the Note Agent from time to time). All payments received by the Notes Agent after 11:00 a.m. may (at the sole discretion of the Notes Agent) in each case be deemed received on the next succeeding Business Day and any applicable interest or fee shall continue to accrue.

- **(C)** If any payment to be made by the Company shall come due on a day other than a Business Day, any interest shall be payable, and any PIK Interest shall be capitalized, on the next succeeding Business Day without additional interest accruing thereon.
- (D) Whenever any payment received by the Notes Agent under this Agreement or any of the other Notes Documents is insufficient to pay in full all amounts due and payable to the Notes Agent and the Holders under or in respect of this Agreement and the other Notes Documents on any date, such payment shall be distributed by the Notes Agent and applied by the Notes Agent and the Holders in the order of priority set forth in Section 1(b)(iii)(1). If the Notes Agent receives funds for application to the Obligations of the Company under or in respect of the Notes Documents under circumstances for which the Notes Documents do not specify the manner in which such funds are to be applied, the Notes Agent may, but shall not be obligated to, elect to distribute such funds to the Holders in accordance with the Holder's Pro Rata Share of such of the outstanding Notes or other Obligations then owing to the Holder.
- (iv) **Withholding**. Notwithstanding anything to the contrary herein, the Company shall be entitled to deduct and withhold from the consideration otherwise payable under the Note such amounts as it is required to deduct and withhold under the Code, or any other tax law, with respect to the making of such payment. To the extent that amounts are so withheld, such withheld amounts shall be treated for all purposes of this Note as having been paid to the Person in respect of whom such deduction and withholding was made. If a payment is payable (in whole or in part) in consideration other than cash and if the cash portion of any such payment is insufficient to satisfy all required tax withholding obligations, the Company shall retain an amount of the non-cash consideration otherwise payable equal in value to the amount required to satisfy any applicable withholding taxes (as reasonably determined by the Company.

3. CONVERSION.

(a) Conversion upon an Offering. In the event that the Company (i) issues and sells shares of its equity securities ("Equity Securities") to investors (the "Investors") in a private offering or placement with total gross proceeds to the Company of at least \$50,000,000 (excluding the conversion of the Notes or other convertible securities issued for capital raising purposes (e.g., Simple Agreements for Future Equity)) or (ii) issues and sells Equity Securities to

Investors in an underwritten public offering pursuant to an effective registration statement filed with the Securities and Exchange Commission in accordance with the Act or under equivalent securities laws and regulations of any other jurisdiction (including, for the avoidance of doubt, any transaction involving a special purpose acquisition corporation, a "SPAC"), (iii) the Company lists its Common Stock other than shares of Common Stock not eligible for resale under Rule 144 under the Securities Act) on a national securities exchange by means of an effective registration statement on Form S-1 filed by the Company with the SEC that registers shares of existing capital stock of the Company for resale, as approved by the Company's board of directors, or (iv) the listing of the Company's Equity Securities on any stock exchange (the occurrence of each event set forth in clauses (i) through (iv) above, an "Offering" and, together with the occurrence of a Change of Control as provided in Section 2(b) below, each, a "Conversion Event"), in each case, while this Note remains outstanding, the Holder shall have the right at any time and from time to time during the period commencing on the date of consummation of such Offering until the Maturity Date, to convert, in whole or in part, the outstanding principal amount (including all accrued PIK Interest not already added to the principal amount of this Note) of this Note, together with all unpaid accrued Cash Interest thereon, into Equity Securities of the Company at a conversion price equal to the Applicable Conversion Price (as defined below) (which shall be determined, for the avoidance of doubt, on the date of the consummation of such Offering). The issuance of Equity Securities by the Company pursuant to the conversion of this Note shall be upon and subject to the same terms and conditions (other than as otherwise set forth herein) applicable to Equity Securities sold in the applicable Offering.

(b) Conversion upon a Change of Control. In the event the Company consummates a Change of Control (as defined below) while this Note remains outstanding, the Company shall repay the outstanding principal amount of this Note in an amount equal to the Repayment Amount; provided, however, that upon the written election of the Holder made not less than five (5) days prior to the Change of Control, the Company shall convert the outstanding principal amount of this Note (including all accrued PIK Interest not already added to the principal amount of this Note) and any unpaid accrued Cash Interest into common stock of the Company ("Common Stock") at a conversion price equal to the Applicable Conversion Price (which shall be determined, for the avoidance of doubt, on the date of consummation of such Change of Control) (assuming conversion of all securities convertible into Common Stock and exercise of all outstanding options and warrants, but excluding the shares of equity securities of the Company issuable upon the conversion of Notes or other convertible securities issued for capital raising purposes (e.g., Simple Agreements for Future Equity)). The Company shall give the Holder and the Note Agent written notice of any Change of Control at least ten (10) Business Days prior to the consummation thereof, which notice will contain the material terms and conditions (including price and form of consideration) of the Change of Control, the identity of the parties to the Change of Control and the intended date of the Change of Control.

For purposes of this Note, (i) a "Change of Control" means, at any time, any Person or "group" other than the Permitted Holders (within the meaning of Rules 13d-3 and 13d-5 under the Exchange Act) shall have obtained the power (whether or not exercised) to elect a majority of the members of the board of directors of the Company; (ii) "Permitted Holders" means each of BCV 55 LLC, BCV 66 LLC, BCV 77 LLC and each of their respective Affiliates; (iii) "Fair Market Value Per Common Share" means the cash and/or the value of the property, rights or

securities to be paid or distributed per share of Common Stock of the Company to the holders of Common Stock of the Company pursuant to a Change of Control or a SPAC (with the value of such property, rights or securities being determined in good faith by the board of directors of the Company); and (iv) the "Applicable Conversion Price" means a conversion price equal to (a) 80% of either (x) the cash price paid per share for Equity Securities of the Company by the Investors in an Offering (other than a SPAC) or (y) the Fair Market Value Per Common Share in a Change of Control or a SPAC, as applicable, if the Conversion Event occurs on or prior to April 19, 2022, (b) 75% of either (x) the cash price paid per share for Equity Securities of the Company by the Investors in an Offering (other than a SPAC) or (y) the Fair Market Value Per Common Share in a Change of Control or a SPAC, as applicable, if the Conversion Event occurs on or after April 19, 2022 but prior to April 19, 2023, (c) 70% of either (x) the cash price paid per share for Equity Securities of the Company by the Investors in an Offering (other than a SPAC) or (y) the Fair Market Value Per Common Share in a Change of Control or a SPAC, as applicable, if the Conversion Event occurs on or after April 19, 2023 but prior to April 19, 2024 and (d) 65% of either (x) the cash price paid per share for Equity Securities of the Company by the Investors in an Offering (other than a SPAC) or (y) the Fair Market Value Per Common Share in a Change of Control or a SPAC, as applicable, if the Conversion Event occurs on or after April 19, 2024 but prior to April 19, 2025.

(c) **Procedure for Conversion**. Upon the conversion in whole or in part of this Note into Equity Securities or Common Stock of the Company (as applicable), the Holder shall surrender this Note to the Company and deliver to the Company any documentation reasonably required by the Company (including, in the case of an Offering, all financing documents executed by the Investors in connection with such Offering). The Company shall not be required to issue or deliver the Equity Securities or Common Stock of the Company (as applicable) into which this Note may convert until the Holder has surrendered this Note to the Company and delivered to the Company any such documentation. The Company shall, as soon as practicable after the surrender of this Note and delivery to the Company of such documentation deliver to the Holder (i) a certificate or certificates for the number of shares of Equity Securities or Common Stock of the Company (as applicable) into which this Note is convertible in whole or in part, rounded downward to the nearest whole share and cash for the amounts not so converted as a result of the above-referenced downward rounding, registered in the name of such Investor or registered nominee or assignee and (ii) to the extent the Holder has converted only a portion of the outstanding principal amount of this Note, a replacement Note for the outstanding principal amount of this Note not converted. Upon the conversion in full or in part of this Note into Equity Securities or Common Stock of the Company (as applicable) pursuant to the terms hereof, in lieu of any fractional shares to which the Holder would otherwise be entitled, the Company shall pay the Holder cash equal to such fraction multiplied by the price at which this Note converts. Upon conversion of this Note in full or in part and payment of cash representing any fractional share pursuant to this Section 2(c), the Company shall be forever released from all its obligations and liabilities under this Note and this Note shall be deemed of no further for

(d) Taxes. The Company shall pay any and all stamp, stock transfer, stock issuance and other similar taxes that may be payable in respect of any issuance or delivery of shares of Equity Securities or Common Stock of the Company (as applicable) upon conversion

of this Note pursuant to this Section 2. The Company shall not, however, be required to pay any income, capital gains or similar tax of the recipient of Equity Securities or Common Stock of the Company (as applicable) or any tax which may be payable in respect of any transfer involved in the issuance and delivery of shares of Equity Securities or Common Stock of the Company (as applicable) in a name other than that in which this Note so converted was registered, and no such issuance or delivery shall be made unless and until the person or entity requesting such issuance has paid to the Company the amount of any such tax or has established, to the satisfaction of the Company, that such tax has been paid.

4. SECURED NOTE.

To secure the performance and payment in full of the Obligations, the Company has granted to the Collateral Agent, for the benefit of the Secured Parties, a first-priority Lien on the Collateral (subject to Permitted Liens) pursuant to the Security Agreement.

5. EVENTS OF DEFAULT.

If there shall be any Event of Default (as defined below) hereunder, at the option and upon the declaration of the Holders representing at least twenty-five percent (25%) of the aggregate outstanding principal amount of the Notes issued pursuant to the Purchase Agreement at such time and upon written notice to the Company (which election and notice shall not be required in the case of an Event of Default under subsection (v) or (vi) below), this Note shall accelerate and all principal (including all accrued PIK Interest not already added to the principal amount of this Note) and all unpaid accrued Cash Interest shall automatically become immediately due and payable. The occurrence of any one or more of the following shall constitute an "Event of Default":

- (i) the Company fails to pay to the Holder (i) the principal of this Note as and when due or (ii) within five (5) Business Days after the same becomes due any Cash Interest on this Note or any fees or any other Obligations;
- (ii) any representation or warranty made or deemed made by or on behalf of the Company or any of its Subsidiaries to the Holder under or in connection with the Note Documents or any certificate or information delivered in connection therewith shall be materially false when made:
- (iii) the Company and its Subsidiaries fail to observe or perform any term, covenant, or provision contained in Section 7 of the Purchase Agreement and such non-observance or non-performance shall not have been remedied or waived within thirty (30) days after the earlier of (i) the Company becoming aware of such non-observance or non-compliance or (ii) receipt by the Company of notice from a Holder of such non-observance or non-compliance;
- (iv) the Company and its Subsidiaries fail to observe or perform any other term, covenant or provision contained in the Purchase Agreement (other than those specified elsewhere in this Section 4) or any other Note Document and such non-observance or non-performance shall not have been remedied or waived within thirty (30) days after the earlier of (i) the Company becoming aware of such non-observance or non-compliance or (ii) receipt by the Company of notice from a Holder of such non-observance or non-compliance;

- (v) the Company or any Material Subsidiary shall (i) apply for or consent to the appointment of a receiver, trustee, liquidator or custodian of itself or of all or a substantial part of its property, (ii) admit in writing its inability to pay its debts generally as they mature, (iii) make a general assignment for the benefit of its or any of its creditors, (iv) be dissolved or liquidated, (v) commence a voluntary case or other proceeding seeking liquidation, reorganization or other relief with respect to itself or its debts under any bankruptcy, insolvency or other similar law now or hereafter in effect or consent to any such relief or to the appointment of or taking possession of its property by any official in an involuntary case or other proceeding commenced against it, or (vi) take any action for the purpose of effecting any of the foregoing;
- (vi) proceedings for the appointment of a receiver, trustee, liquidator or custodian of the Company or any of its Material Subsidiaries, or of all or a substantial part of the property thereof, or an involuntary case or other proceedings seeking liquidation, reorganization or other relief with respect to the Company or any of its Material Subsidiaries, if any, or the debts thereof under any bankruptcy, insolvency or other similar law now or hereafter in effect shall be commenced, or an order for relief shall be entered in any such proceeding, or such proceeding shall not be dismissed or discharged within 60 days of commencement;
- (vii) the Company fails to pay when due any principal of or interest on or any other amount payable in respect of, or breaches or defaults under any other term of, any mortgage, indenture, agreement or other instrument under which there may be outstanding any Indebtedness in an aggregate principal amount of in excess of \$10,000,000, in each case, beyond the grace period, if any, if the effect of such non-payment, breach or default is to cause such Indebtedness to become or be declared due and payable (or subject to a compulsory repurchase or redemption) prior to its stated maturity;
- (viii) any court, government, or Governmental Authority shall condemn, seize or otherwise appropriate, or take custody or control of, all or any material portion of the property of the Company and its Subsidiaries, taken as a whole;
- (ix) one or more judgments or orders for the payment of money in excess of \$10,000,000 (or the equivalent thereof in currencies other than U.S. dollars) in the aggregate shall be entered or filed against the Company or any of its Subsidiaries and shall remain undischarged, unvacated, unbonded or unstayed for a period of thirty (30) consecutive days;
- (x) (i) the occurrence of a Reportable Event with respect to any Plan, (ii) the filing of a notice of intent to terminate a Plan by the Company, any ERISA Affiliate or any Subsidiary, the institution of proceedings to terminate a Plan by the PBGC or any other Person; (iii) the withdrawal in a "complete withdrawal" or a "partial withdrawal" as defined in Sections 4203 and 4205, respectively, of ERISA by the Company, any ERISA Affiliate or any Subsidiary of the Company from any Multiemployer Plan, (iv) the incurrence of any material increase in the contingent liability of the Company or any of its Subsidiaries with respect to any "employee welfare benefit plan" as defined in Section 3(1) of ERISA which covers retired

employees and its beneficiaries or (v) the Unfunded Liabilities of all Single Employer Plans shall exceed (in the aggregate) \$10,000,000, in each such case which, either individually or in the aggregate, would be reasonably expected to result in liability to any Note Party in excess of \$10,000,000;

- (xi) The institution by the Company, any ERISA Affiliate or any Subsidiary of steps to terminate any Plan if, in order to effectuate such termination, the Company, such ERISA Affiliate or such Subsidiary, as the case may be, would be required to make a contribution to such Plan, or would incur a liability or obligation to such Plan, in excess of \$10,000,000, or the institution by the PBGC of steps to terminate any Plan, which would reasonably be expected to result in material liability to any Note Party;
- (xii) The Company or any Material Subsidiary shall (i) be the subject of any proceeding pertaining to the release by the Company, any such Material Subsidiary or any other Person of any Hazardous Material into the environment or (ii) violate any Environmental Law, which, in either case could reasonably be expected to have a Material Adverse Effect;
- (xiii) Any Collateral Document shall for any reason fail to create a valid and perfected first priority (subject to any Permitted Liens) security interest in any collateral purported to be covered thereby, except as permitted by the terms of any Collateral Document, or any Collateral Document shall fail to remain in full force or effect (other than in accordance with the terms hereof or thereof) or any action taken by the Company or any Subsidiary shall be taken to discontinue or to assert the invalidity or unenforceability of any Collateral Document;
- (xiv) any subordination or intercreditor agreement relating to any Indebtedness of any Note Party subordinated to the Obligations, or any subordination provisions of any note or other document running to the benefit of the Secured Parties in respect of such Indebtedness, shall cease for any reason to be in full force and effect other than in accordance with the terms hereof or thereof or any Note Party or any of their Subsidiaries shall so assert in writing; or
- (xv) The Note Parties shall be enjoined, restrained, or in any way prevented by court order from continuing to conduct all or any material part of the business of the Note Parties, taken as a whole.

6. MISCELLANEOUS PROVISIONS.

- (a) Waivers. The Company hereby waives demand, notice, presentment, protest and notice of dishonor.
- (b) **Further Assurances**. The Holder agrees and covenants that at any time and from time to time the Holder will promptly execute and deliver to the Company such further instruments and documents and take such further action as the Company may reasonably require in order to carry out the full intent and purpose of this Note and to comply with state or federal securities laws or other regulatory approvals.
- (c) **Transfers of Notes**. Subject to the Company's prior written consent (not to be unreasonably withheld or delayed) and to the extent permitted by applicable law, this Note

may be transferred only upon its surrender to the Company for registration of transfer, duly endorsed, or accompanied by a duly executed written instrument of transfer in form satisfactory to the Company. Thereupon, this Note shall be reissued to, and registered in the name of, the transferee, or a new Note for like principal amount and interest shall be issued to, and registered in the name of, the transferee. Interest and principal shall be paid solely to the registered holder of this Note. Such payment shall constitute full discharge of the Company's obligation to pay such interest and principal. Upon the effectiveness of any transfer pursuant to this Section 5(c) the Company shall provide an updated Ledger to the Note Agent. Any transfer of this Note in contravention of this Section 5(c) shall be void and the Company shall be entitled to continue to treat the Holder as the holder of this Note for all purposes under the Note Documents. The Company shall at all times maintain a book-entry system, which shall reflect ownership of this Note and interests therein. This Note is intended to be in "registered form" for United States federal tax purposes.

(d) Market Standoff. To the extent requested by the Company or an underwriter of securities of the Company, each Holder and any permitted transferee thereof shall not, without the prior written consent of the underwriters in an underwritten public offering pursuant to an effective registration statement filed with the Securities and Exchange Commission in accordance with the Act or under equivalent securities laws and regulations of any other jurisdiction, offer, sell, make any short sale of, grant or sell any option for the purchase of, lend, pledge, otherwise transfer or dispose of (directly or indirectly), enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership (whether any such transaction is described above or is to be settled by delivery of Securities or other securities, in cash, or otherwise), any Securities or other shares of stock of the Company then owned by such Holder or any transferee thereof, or enter into an agreement to do any of the foregoing, for up to 180 days following the consummation of any such offering. For purposes of this paragraph, "Company" includes the Company or any other direct or indirect parent entity of the Company into which the Company merges or consolidates. The Company may place restrictive legends on the certificates representing the shares subject to this paragraph and may impose stop transfer instructions with respect to the Securities and such other shares of stock of each Holder and any transferee thereof (and the shares or securities of every other person subject to the foregoing restriction) until the end of such period. Each Holder and any transferee thereof shall enter into any agreement reasonably required by the underwriters for such an offering to implement the foregoing within any reasonable timeframe so requested. The underwriters for any such offering are intended third party beneficiaries of this paragraph and shall have the right, power and authority to enforce the provisions of

(e) Waiver of Statutory Information Rights. Prior to the conversion in full of this Note, the Holder, on behalf of the Holder and all beneficial owners of the Securities now or hereafter owned by the Holder (a "Beneficial Owner"), acknowledges and agrees that that neither the Holder nor any of the Beneficial Owners will have any right to receive any information from the Company by virtue of ownership of any of the Securities. Without limiting the foregoing, prior to the conversion in full of this Note, to the fullest extent permitted by law, the Holder hereby unconditionally and irrevocably waives all rights under Section 220 of the Delaware General Corporation Law and all such similar information and/or inspection rights that

may be provided under the law of any jurisdiction, or any federal, state or foreign regulation, that are, or may become, applicable to the Company or the Company's capital stock (the "Inspection Rights") on behalf of the Holder and all Beneficial Owners. The Holder, on behalf of the Holder and all Beneficial Owners, hereby covenants and agrees that neither the Holder nor any Beneficial Owner shall directly or indirectly commence, voluntarily aid in any way, prosecute, assign, transfer, or cause to be commenced any claim, action, cause of action, or other proceeding to pursue or exercise the Inspection Rights. The Holder hereby further warrants and represents that the Holder has reviewed this waiver with its legal counsel, and that the Holder knowingly and voluntarily waives its rights otherwise provided by Section 220 of the Delaware General Corporation Law (or under similar rights under other applicable law). Notwithstanding the foregoing, Beneficial Owners that were issued Equity Securities other than by way of a conversion in connection with an Offering will not be subject to this Section 5(e). The terms of this Section 5(e) shall survive any repayment of this Note.

- (f) Amendment and Waiver. This Note and any term hereof may only be amended, waived or modified in accordance with Section 10.9 of the Purchase Agreement. Upon the effectuation of such amendment, waiver or modification with the consent of the Required Holders or each Holder, as applicable, in conformance with Section 10.9 of the Purchase Agreement, such amendment, waiver or modification shall be effective as to, and binding against the Holders of, all of the Notes, and the Company shall promptly give written notice thereof to the Holder if the Holder has not previously consented to such amendment, waiver or modification in writing; provided that the failure to give such notice shall not affect the validity of such amendment, waiver or modification.
- (g) **Binding Agreement**. The terms and conditions of this Note shall inure to the benefit of and be binding upon the respective successors and assigns of the parties. Nothing in this Note, expressed or implied, is intended to confer upon any third party any rights, remedies, obligations, or liabilities under or by reason of this Note, except as expressly provided in this Note.
- (h) Counterparts. This Note may be executed in one or more counterparts (and by different parties hereto in different counterparts), each of which shall be deemed an original, but all of which together shall constitute a single contract. Any counterpart of a signature page to this Note may be delivered by facsimile, electronic mail (including ".pdf" or ".tif") or by means of an electronic signature complying with the Electronic Signatures in Global and National Commerce Act, the New York Electronic Signature and Records Act or any other similar state laws based on the Uniform Electronic Transactions Act and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes to the fullest extent permitted by applicable law.
- (i) **Headings; Interpretation**. Paragraph headings used in this Note are included for convenience of reference only and will not modify the provisions that they precede or affect the interpretation of this Note or any other Note Document. The term "including" shall be interpreted to mean "including but not limited to".
- (j) **Notices**. All communications and notices required or permitted to be given hereunder shall be delivered in accordance with Section 10.8 of the Purchase Agreement.

(k) **GOVERNING LAW.** THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES (OTHER THAN SECTIONS 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW) THEREOF.

- (I) **Expenses, Etc.** This Note is subject to the provisions of Section 10.3 (Jurisdiction and Venue), Section 10.4 (Waiver of Jury Trial) and Section 10.10 (Expenses and Indemnification) of the Purchase Agreement, which are by this reference incorporated herein in full, *mutatis mutandis*.
- (m) **Delays or Omissions**. It is agreed that no delay or omission to exercise any right, power or remedy accruing to each Holder, upon any breach or default of the Company under this Note or any Note shall impair any such right, power or remedy, nor shall it be construed to be a waiver of any such breach or default, or any acquiescence therein, or of or in any similar breach or default thereafter occurring; nor shall any waiver of any single breach or default be deemed a waiver of any other breach or default thereafter occurring. It is further agreed that any waiver, permit, consent or approval of any kind or character by a Holder of any breach or default under this Note, or any waiver by such Holder of any provisions or conditions of this Note must be in writing and shall be effective only to the extent specifically set forth in writing and that all remedies, either under this Note, or by law or otherwise afforded to such Holder, shall be cumulative and not alternative.
- (n) **Entire Agreement**. This Note and the other Note Documents constitute the entire contract among the parties hereto regarding to the subject matters addressed herein and supersede any and all previous agreements, negotiations, and discussions, oral or written, by the parties regarding the subject matters addressed herein.
- (o) **Exculpation among Holders**. The Holder acknowledges that the Holder is not relying on any person, firm or corporation, other than the Company and its officers and board members, in making its investment or decision to invest in the Company.
- (p) **Broker's Fees**. Each party hereto represents and warrants that no agent, broker, investment banker, person or firm acting on behalf of or under the authority of such party hereto is or will be entitled to any broker's or finder's fee or any other commission directly or indirectly in connection with the transactions contemplated herein. Each party hereto further agrees to indemnify each other party for any claims, losses or expenses incurred by such other party as a result of the representation in this subsection being untrue.
- (q) Repayment Amounts. The parties hereto hereby acknowledge and agree that payment of any Repayment Amount or the Covenant Amendment Repayment Amount hereunder constitutes liquidated damages and not a penalty, the actual amount of damages to the Holder or profits lost by the Holder as a result of a prepayment or conversion of this Note would be impracticable and extremely difficult to ascertain and the Repayment Amount or the Covenant Amendment Repayment Amount hereunder is part of an arm's length transaction between sophisticated parties represented by counsel and is bargained for consideration provided

for and agreed to by mutual agreement of the Company and the Holder. THE NOTE PARTIES EXPRESSLY WAIVE THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE OR LAW THAT PROHIBITS OR MAY PROHIBIT THE COLLECTION OF THE REPAYMENT AMOUNT TO THE EXTENT SUCH REPAYMENT AMOUNT IS DUE AND PAYABLE IN ACCORDANCE WITH THIS NOTE. The Note Parties expressly agree that: (i) the Repayment Amount or the Covenant Amendment Repayment Amount, as applicable, shall be payable notwithstanding the then prevailing market rates at the time payment is made; (ii) the Note Parties shall be estopped hereafter from claiming differently than as agreed to in this paragraph; and (iii) their agreement to pay the Repayment Amount or the Covenant Amendment Repayment Amount, as applicable, is a material inducement to the Holder to purchase the Note.

- (r) **Tax Forms**. Prior to the date hereof, the Holder (and, in the event any Person becomes an assignee or participant in respect of this Note, prior to the date such Person becomes such an assignee or participant) shall have delivered to the Company executed copies of Internal Revenue Service ("**IRS**") Form W-9 or appropriate IRS Form W-8, as applicable, as well as any reasonably necessary supporting information, certifying that any payments to such Holder (or assignee or participant) under this Note are exempt from U.S. federal withholding tax. The Holder (and assignee or participant) further agrees that if any such form or certification expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify the Company in writing of its legal inability to do so
- (s) Calculations. The Company shall be responsible for making all calculations with respect to this Note, including, without limitation, accrued interest (including PIK Interest), the Fair Market Value Per Common Share, Applicable Conversion Price, and the Covenant Amendment Repayment Amount, and shall forward such calculations to the Agents and the Holder upon request.

[Signature pages follow]

IN WITNESS WHEREOF, the parties hereto have executed this Note as of the date first written above.

COMPANT:	
CORE SCIENTIFIC HOLDING CO.	
Ву:	
Name: Title:	
E-mail:	
Address:	

SIGNATURE PAGE TO CORE SCIENTIFIC HOLDING CO. CONVERTIBLE PROMISSORY NOTE

EXECUTION VERSION

HOLDER (if an entity):

Name of Holder:	
	Ву:
	Name:
	E-mail:
	Address:
	HOLDER (if an individual):
Name of Holder:	
Signature:	
	E-mail:
	Address:

SIGNATURE PAGE TO CORE SCIENTIFIC HOLDING CO. CONVERTIBLE PROMISSORY NOTE

EXHIBIT B

FORM OF COUNTERPART SIGNATURE PAGE

EXHIBIT C

FORM OF GUARANTEE

EXHIBIT D

FORM OF SECURITY AGREEMENT

EXHIBIT E

FORM OF INTELLECTUAL PROPERTY SECURITY AGREEMENT

EXHIBIT F

FORM OF SOLVENCY CERTIFICATE

EXHIBIT G

FORM OF ADMINISTRATIVE QUESTIONNAIRE

FIRST AMENDMENT TO SECURED CONVERTIBLE NOTE PURCHASE AGREEMENT

THIS FIRST AMENDMENT TO SECURED CONVERTIBLE NOTE PURCHASE AGREEMENT (this "First Amendment") is made as of April 22, 2021, by and among Core Scientific Holding Co., a Delaware corporation (the "Company"), the Guarantors party hereto, the Purchaser party hereto and U.S. Bank National Association, as note agent and as collateral agent for the Secured Parties (in such capacities, the "Agents").

RECITALS

WHEREAS, the Company has entered into that certain Secured Convertible Note Purchase Agreement, dated as of April 19, 2021 (as amended, restated, supplemented and/or otherwise modified from time to time, the "Purchase Agreement"; capitalized terms used but not otherwise defined herein having the meanings ascribed thereto therein), by and among the Company, the Guarantors from time to time party thereto, the Purchasers from time to time party thereto and the Agents; and

WHEREAS, the Company desires to amend, and the Purchaser party hereto constituting the Required Holders has agree to amend, the Purchase Agreement, on the terms set forth herein.

AGREEMENT

NOW THEREFORE, in consideration of the foregoing, the parties hereto, intending to be legally bound, hereby agree as follows:

- 1. AMENDMENT. Effective as of the First Amendment Effective Date (as defined below), the Purchase Agreement is hereby amended by amending and restating clause (i) to the proviso appearing in Section 2.2(a) thereof as follows: "(i) each Additional Closing Date shall have occurred on or about April 26, 2021, but in any event, not later than April 30, 2021".
- 2. CONDITIONS PRECEDENT. This First Amendment shall become effective on the date (the "First Amendment Effective Date") on which the Company shall have received counterparts of this First Amendment duly executed by each party hereto.

3. MISCELLANEOUS.

- 3.1 Effect of Agreement. Except as set forth expressly herein, all terms of the Purchase Agreement, as amended hereby, and the other Note Documents shall be and remain in full force and effect and shall constitute the legal, valid, binding and enforceable obligations of the parties hereto. This Amendment shall constitute a "Note Document" for all purposes of the Purchase Agreement.
- 3.2 Entire Agreement. This Note and the other Note Documents constitute the entire contract among the parties hereto regarding to the subject matters addressed herein and supersede any and all previous agreements, negotiations, and discussions, oral or written, by the parties regarding the subject matters addressed herein.

- 3.3 Successors and Assigns. The terms and conditions of this First Amendment shall inure to the benefit of and be binding upon the respective successors and assigns of the parties.
- 3.4 Expenses, Governing Law, Etc. This First Amendment is subject to the provisions of Section 10.2 (Governing Law), Section 10.3 (Jurisdiction and Venue), Section 10.4 (Waiver of Jury Trial) and Section 10.10(a) (Expenses and Indemnification) of the Purchase Agreement, which are by this reference incorporated herein in full, *mutatis mutandis*. U.S. Bank National Association is entering into this First Amendment solely in its capacity as Note Agent and Collateral Agent pursuant to the direction from the Required Holders. In acting hereunder, the Agents shall be entitled to all of the rights, privileges and immunities of the Agents in acting hereunder, including without limitation the obligations of the Holders pursuant to Section 9.3 and Section 9.6 of the Purchase Agreement.
- 3.5 Counterparts. This First Amendment may be executed in one or more counterparts (and by different parties hereto in different counterparts), each of which shall be deemed an original, but all of which together shall constitute a single contract. Any counterpart of a signature page to this First Amendment may be delivered by facsimile, electronic mail (including ".pdf" or ".tif") or by means of an electronic signature complying with the Electronic Signatures in Global and National Commerce Act, the New York Electronic Signature and Records Act or any other similar state laws based on the Uniform Electronic Transactions Act and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes to the fullest extent permitted by applicable law. The Note Parties agree to assume all risks arising out of the use of using digital signatures and electronic methods to submit communications to the Note Agent, including without limitation the risk of the Note Agent acting on unauthorized instructions, and the risk of interception and misuse by third parties.

[Signature pages follow]

IN WITNESS WHEREOF, the parties hereto have executed this First Amendment as of the date first written above.

CORE SCIENTIFIC HOLDING CO.

By: /s/ Michael Trzupek Name: Michael Trzupek Title: Chief Financial Officer

CORE SCIENTIFIC, INC.

By: /s/ Michael Trzupek Name: Michael Trzupek Title: Chief Financial Officer

AMERICAN PROPERTY ACQUISITION, LLC

BY: CORE SCIENTIFIC, INC., ITS SOLE MEMBER

By: /s/ Michael Trzupek Name: Michael Trzupek Title: Chief Financial Officer

AMERICAN PROPERTY ACQUISITIONS I, LLC

BY: AMERICAN PROPERTY ACQUISITION, LLC, ITS SOLE

MEMBER

BY: CORE SCIENTIFIC, INC., ITS SOLE MEMBER

By: /s/ Michael Trzupek Name: Michael Trzupek Title: Chief Financial Officer

AMERICAN PROPERTY ACQUISITIONS VII, LLC

BY: AMERICAN PROPERTY ACQUISITION, LLC, ITS SOLE

MEMBER

BY: CORE SCIENTIFIC, INC., ITS SOLE MEMBER

By: /s/ Michael Trzupek Name: Michael Trzupek Title: Chief Financial Officer

U.S. BANK NATIONAL ASSOCIATION, AS NOTE AGENT AND AS COLLATERAL AGENT

By: /s/ Joshua Hahn

Name: Joshua Hahn Title: Authorized Signatory

CELSIUS CORE LLC

By: /s/ Ron Deutsch Name: Ron Deutsch

Title: Chief Revenue Officer

We consent to the inclusion in this Registration Statement of Core Scientific, Inc. (f/k/a Power & Digital Infrastructure Acquisition Corp.) on Form S-1 of our reports:

- Dated January 8, 2021, except for Note 7, as to which the date is February 11, 2021, which includes an explanatory paragraph as to the Company's ability to continue as a going concern, with respect to our audit of the financial statements of Power & Digital Infrastructure Acquisition Corp. (now known as Core Scientific Inc.) as of December 31, 2020 and for the period from December 29, 2020 (inception) through December 31, 2020, which report appears in the Prospectus, which is part of this Registration Statement.
- Dated February 19, 2021 except for Note 1, 2, 3, 7 and 8, which are dated December 10, 2021, which includes a restatement paragraph and
 an explanatory paragraph as to the Company's ability to continue as a going concern, with respect to our audit of the financial statements
 of Power & Digital Infrastructure Acquisition Corp. (now known as Core Scientific Inc.) as of February 12, 2021, which report appears in
 the Prospectus, which is part of this Registration Statement.

We also consent to the reference to our Firm under the heading "Experts" in such Prospectus.

/s/ Marcum LLP

Marcum LLP New York, NY February 8, 2022

Consent of Independent Registered Public Accounting Firm

We consent to the reference to our firm under the caption "Experts" and to the use of our report dated August 11, 2021, with respect to the consolidated financial statements of Core Scientific Holding Co. included in the Registration Statement (Form S-1) and related Prospectus of Core Scientific, Inc. for the registration of up to 186,172,423 shares of its common stock, up to 14,891,667 shares of its common stock issuable upon exercise of warrants, and up to 6,266,667 warrants to purchase common stock.

/s/ Ernst & Young LLP

Seattle, Washington February 8, 2022

We consent to the inclusion in this Registration Statement of Core Scientific, Inc. on Form S-1 of our report dated July 30, 2021, with respect to our audits of the financial statements of Blockcap, Inc. as of December 31, 2020 and 2019, for the year ended December 31, 2020 and for the period from February 19, 2019 (inception) through December 31, 2019, which report appears in the Prospectus, which is part of this Registration Statement. We also consent to the reference to our Firm under the heading "Experts" in such Prospectus.

/s/ Marcum LLP

We consent to the inclusion in this Registration Statement of Core Scientific, Inc. on Form S-1 of our report dated July 30, 2021, with respect to our audits of the financial statements of BEP 888, LLC as of November 30, 2020, and for the period from June 1, 2020 (inception) through November 30, 2020, which report appears in the Prospectus, which is part of this Registration Statement. We also consent to the reference to our Firm under the heading "Experts" in such Prospectus.

/s/ Marcum LLP

We consent to the inclusion in this Registration Statement of Core Scientific, Inc. on Form S-1 of our report dated July 30, 2021, with respect to our audits of the financial statements of BEP 999, LLC as of November 30, 2020, and for the period from November 5, 2020 (inception) through November 30, 2020, which report appears in the Prospectus, which is part of this Registration Statement. We also consent to the reference to our Firm under the heading "Experts" in such Prospectus.

/s/ Marcum LLP

We consent to the inclusion in this Registration Statement of Core Scientific, Inc. on Form S-1 of our report dated July 30, 2021, with respect to our audits of the financial statements of RME Black 100, LLC as of November 30, 2020, and for the period from April 16, 2020 (inception) through November 30, 2020, which report appears in the Prospectus, which is part of this Registration Statement. We also consent to the reference to our Firm under the heading "Experts" in such Prospectus.

/s/ Marcum LLP

We consent to the inclusion in this Registration Statement of Core Scientific, Inc. on Form S-1 of our report dated July 30, 2021, with respect to our audits of the financial statements of RME Black 200, LLC as of November 30, 2020, and for the period from April 27, 2020 (inception) through November 30, 2020, which report appears in the Prospectus, which is part of this Registration Statement. We also consent to the reference to our Firm under the heading "Experts" in such Prospectus.

/s/ Marcum LLP

Calculation of Filing Fee Tables

Form S-1 (Form Type)

Core Scientific, Inc.

(Exact Name of Registrant as Specified in its Charter)

Table 1: Newly Registered Securities

	Security Type	Security Class Title	Fee Calculation or Carry Forward Rule	Amount Registered(1)	Proposed Maximum Offering Price Per Unit	Maximum Aggregate Offering Price	Fee Rate	Amount of Registration Fee
Fees to		Common Stock,						
Be Paid		par value \$0.0001 per						
	Equity	share	457(o)	14,891,667(2)	\$9.00(5)	\$134,025,003.00	0.0000927	\$12,424.12(5)
Fees to Be Paid		Common Stock, par value \$0.0001 per						
	Equity	share	457(o)	186,172,423(3)	\$9.00(5)	\$1,675,551,807.00	0.0000927	\$155,323.66(5)
Fees to Be Paid		Warrants to purchase shares						
	Equity	of common stock	457(i)	6,266,667(4)	_	_	_	(6)
	Total Offering Amounts					\$1,809,576,810.00		
	Total Fees Previously Paid					_		
	Total Fee Offsets Net Fee Due							
						\$167,747.78		

⁽¹⁾ In the event of a stock split, stock dividend or other similar transaction involving the registrant's common stock, in order to prevent dilution, the number of shares of common stock registered hereby shall be automatically increased to cover the additional shares of common stock in accordance with Rule 416(a) under the Securities Act.

⁽²⁾ Consists of (i) 6,266,667 shares of common stock issuable upon the exercise of warrants issued to the XPDI Sponsor LLC and certain funds and accounts managed by subsidiaries of BlackRock, Inc. in a private placement (the "*Private Placement Warrants*") and (ii) 8,625,000 shares of common stock issuable upon the exercise of warrants included in the publicly sold units (the "*Public Warrants*") to purchase common stock, in each case at an exercise price of \$11.50 per share.

⁽³⁾ Consists of (i) 8,625,000 Founder Shares issued in a private placement in connection with the initial public offering of XPDI, (ii) up to 6,266,667 shares of common stock issuable upon exercise of the Private Placement Warrants, (iii) up to 91,052,355 shares of common stock (including shares issuable upon the exercise of convertible securities) held by certain affiliates of our company and (iv) up to 80,228,401 shares of common stock issuable upon conversion of certain Convertible Notes.

- (4) Represents the resale of 6,266,667 Private Placement Warrants.
- (5) Estimated solely for the purpose of calculating the registration fee in accordance with Rule 457(c) under the Securities Act. The price per share and aggregate offering price are based on the average of the high and low prices of the Registrant's common stock on February 3, 2022, as reported on the Nasdaq Stock Market.
- (6) In accordance with Rule 457(i), the entire registration fee for the Private Placement Warrants is allocated to the shares of common stock underlying the Private Placement Warrants, and no separate fee is payable for the Private Placement Warrants.